

MARITIME SAFETY COMMITTEE 104th session Agenda item 18 MSC 104/18/Add.1 1 November 2021 Original: ENGLISH

Ε

REPORT OF THE MARITIME SAFETY COMMITTEE ON ITS 104TH SESSION

Attached are annexes 1 to 5 and 30 to 42 to the report of the Maritime Safety Committee on its 104th session (MSC 104/18). Annexes 6 to 29 are set out in document MSC 104/18/Add.2.



LIST OF ANNEXES

- ANNEX 1 RESOLUTION MSC.491(104) AMENDMENTS TO THE PROTOCOL OF 1988 RELATING TO THE INTERNATIONAL CONVENTION ON LOAD LINES, 1966 (1988 LOAD LINES PROTOCOL)
- ANNEX 2 RESOLUTION MSC.492(104) AMENDMENTS TO THE INTERNATIONAL CODE FOR THE CONSTRUCTION AND EQUIPMENT OF SHIPS CARRYING LIQUEFIED GASES IN BULK (IGC CODE)
- ANNEX 3 DRAFT MODEL REGULATIONS ON DOMESTIC FERRY SAFETY
- ANNEX 4 UPDATED WORKPLAN FOR THE WORK ON DOMESTIC FERRY SAFETY
- ANNEX 5 DRAFT ASSEMBLY RESOLUTION ON PREVENTION AND SUPPRESSION OF PIRACY, ARMED ROBBERY AGAINST SHIPS AND ILLICIT MARITIME ACTIVITY IN THE GULF OF GUINEA
- ANNEX 30 DRAFT ASSEMBLY RESOLUTION ON PROCEDURES FOR PORT STATE CONTROL, 2021
- ANNEX 31 DRAFT ASSEMBLY RESOLUTION ON SURVEY GUIDELINES UNDER THE HARMONIZED SYSTEM OF SURVEY AND CERTIFICATION (HSSC), 2021
- ANNEX 32 DRAFT ASSEMBLY RESOLUTION ON 2021 NON-EXHAUSTIVE LIST OF OBLIGATIONS UNDER INSTRUMENTS RELEVANT TO THE IMO INSTRUMENTS IMPLEMENTATION CODE (III CODE)
- ANNEX 33 DRAFT MSC-MEPC.5 CIRCULAR ON MODEL AGREEMENT FOR THE AUTHORIZATION OF RECOGNIZED ORGANIZATIONS ACTING ON BEHALF OF THE ADMINISTRATION
- ANNEX 34 PROPOSED BIENNIAL AGENDA OF THE MARITIME SAFETY COMMITTEE FOR THE 2022-2023 BIENNIUM
- ANNEX 35 BIENNIAL STATUS REPORT OF THE MARITIME SAFETY COMMITTEE FOR THE 2020-2021 BIENNIUM
- ANNEX 36 POST-BIENNIAL AGENDA OF THE MARITIME SAFETY COMMITTEE
- ANNEX 37 BIENNIAL STATUS REPORTS OF THE SUB-COMMITTEES
- ANNEX 38 PROVISIONAL AGENDAS FOR THE FORTHCOMING SESSIONS OF THE SUB-COMMITTEES
- ANNEX 39 SUBSTANTIVE ITEMS FOR INCLUSION IN THE AGENDAS FOR MSC 105 AND MSC 106
- ANNEX 40 DRAFT ASSEMBLY RESOLUTION ON COMPREHENSIVE ACTION TO ADDRESS SEAFARERS' CHALLENGES DURING THE COVID-19 PANDEMIC
- ANNEX 41 THEMATIC PRIORITIES FOR THE ITCP COVERING THE 2022-2023 BIENNIUM
- ANNEX 42 STATEMENTS BY DELEGATIONS AND OBSERVERS

RESOLUTION MSC.491(104) (adopted on 8 October 2021)

AMENDMENTS TO THE PROTOCOL OF 1988 RELATING TO THE INTERNATIONAL CONVENTION ON LOAD LINES, 1966 (1988 LOAD LINES PROTOCOL)

THE MARITIME SAFETY COMMITTEE,

RECALLING Article 28(b) of the Convention on the International Maritime Organization concerning the functions of the Committee,

RECALLING ALSO article VI of the Protocol of 1988 relating to the International Convention on Load Lines, 1966 ("1988 Load Lines Protocol") concerning amendment procedures,

HAVING CONSIDERED, at its 104th session, amendments to the 1988 Load Lines Protocol proposed and circulated in accordance with article VI(2)(a) thereof,

1 ADOPTS, in accordance with article VI(2)(d) of the 1988 Load Lines Protocol, amendments to the 1988 Load Lines Protocol, the text of which is set out in the annex to the present resolution;

2 DETERMINES, in accordance with article VI(2)(f)(ii)(bb) of the 1988 Load Lines Protocol, that the said amendments shall be deemed to have been accepted on 1 July 2023, unless, prior to that date, more than one third of the Parties to the 1988 Load Lines Protocol, or Parties the combined merchant fleets of which constitute not less than 50% of the gross tonnage of all the merchant fleets of all Parties, have notified their objections to the amendments;

3 INVITES the Parties concerned to note that, in accordance with article VI(2)(g)(ii) of the 1988 Load Lines Protocol, the amendments shall enter into force on 1 January 2024 upon their acceptance in accordance with paragraph 2 above;

4 REQUESTS the Secretary-General, in conformity with article VI(2)(e) of the 1988 Load Lines Protocol, to transmit certified copies of the present resolution and the text of the amendments contained in the annex to all Parties to the 1988 Load Lines Protocol;

5 REQUESTS ALSO the Secretary-General to transmit copies of this resolution and its annex to Members of the Organization which are not Parties to the 1988 Load Lines Protocol.

AMENDMENTS TO THE PROTOCOL OF 1988 RELATING TO THE INTERNATIONAL CONVENTION ON LOAD LINES, 1966 (1988 LOAD LINES PROTOCOL)

Annex B ANNEXES TO THE CONVENTION AS MODIFIED BY THE PROTOCOL OF 1988 RELATING THERETO

Annex I Regulations for determining load lines

Chapter II Conditions of assignment of freeboard

Regulation 22 Scuppers, inlets and discharges

- 1 Regulation 22(1)(g) is replaced with the following:
 - "(g) Table 22.1 provides the acceptable arrangements of scuppers and discharges."

Chapter III Freeboards

Regulation 27 *Types of ships*

2 Regulation 27(13)(a) is replaced with the following:

"(13) The condition of equilibrium after flooding shall be regarded as satisfactory provided:

The final waterline after flooding, taking into account sinkage, heel and (a) trim, is below the lower edge of any opening through which progressive downflooding may take place. Such openings shall include air pipes, ventilators (even if they comply with regulation 19(4)) and openings which are closed by means of weathertight doors (even if they comply with regulation 12) or hatch covers (even if they comply with regulation 16(1) through (5)), and may exclude those openings closed by means of manhole covers and flush scuttles (which comply with regulation 18), cargo hatch covers of the type described in regulation 27(2), remotely operated sliding watertight doors, hinged watertight access doors with open/closed indication locally and at the navigation bridge, of the quickacting or single-action type that are normally closed at sea, hinged watertight doors that are permanently closed at sea, and sidescuttles of the non-opening type (which comply with regulation 23). In the case of doors separating a main machinery space from a steering gear compartment, watertight doors may be of a hinged, guick-acting type kept closed at sea whilst not in use, provided also that the lower sill of such doors is above the summer load waterline."

RESOLUTION MSC.492(104) (adopted on 8 October 2021)

AMENDMENTS TO THE INTERNATIONAL CODE FOR THE CONSTRUCTION AND EQUIPMENT OF SHIPS CARRYING LIQUEFIED GASES IN BULK (IGC CODE)

THE MARITIME SAFETY COMMITTEE,

RECALLING Article 28(b) of the Convention on the International Maritime Organization concerning the functions of the Committee,

NOTING resolution MSC.5(48), by which it adopted the International Code for the Construction and Equipment of Ships Carrying Liquefied Gases in Bulk ("the IGC Code"), which has become mandatory under chapter VII of the International Convention for the Safety of Life at Sea (SOLAS), 1974 ("the Convention"),

NOTING ALSO article VIII(b) and regulation VII/11.1 of the Convention concerning the procedure for amending the IGC Code,

HAVING CONSIDERED, at its 104th session, amendments to the IGC Code, proposed and circulated in accordance with article VIII(b)(i) of the Convention,

1 ADOPTS, in accordance with article VIII(b)(iv) of the Convention, amendments to the IGC Code, the text of which is set out in the annex to the present resolution;

2 DETERMINES, in accordance with article VIII(b)(vi)(2)(bb) of the Convention, that said amendments shall be deemed to have been accepted on 1 July 2023, unless, prior to that date, more than one third of the Contracting Governments to the Convention, or Contracting Governments the combined merchant fleets of which constitute not less than 50% of the gross tonnage of the world's merchant fleet, have notified the Secretary-General of their objections to the amendments;

3 INVITES Contracting Governments to note that, in accordance with article VIII(b)(vii)(2) of the Convention, the amendments shall enter into force on 1 January 2024 upon its acceptance in accordance with paragraph 2 above;

4 REQUESTS the Secretary-General, for the purpose of article VIII(b)(v) of the Convention, to transmit certified copies of the present resolution and the text of the amendments contained in the annex to all Contracting Governments to the Convention;

5 REQUESTS ALSO the Secretary-General to transmit copies of this resolution and its annex to Members of the Organization which are not Contracting Governments to the Convention.

AMENDMENTS TO THE INTERNATIONAL CODE FOR THE CONSTRUCTION AND EQUIPMENT OF SHIPS CARRYING LIQUEFIED GASES IN BULK (IGC CODE)

CHAPTER 2 SHIP SURVIVAL CAPABILITY AND LOCATION OF CARGO TANKS

- 2.7 Survival requirements
- 1 The existing text of paragraph 2.7.1.1 is replaced with the following:

2.7.1 In any stage of flooding:

".1 the waterline, taking into account sinkage, heel and trim, shall be below the lower edge of any opening through which progressive flooding or downflooding may take place. Such openings shall include air pipes and openings that are closed by means of weathertight doors or hatch covers and may exclude those openings closed by means of watertight manhole covers and watertight flush scuttles, small watertight cargo tank hatch covers that maintain the high integrity of the deck, remotely operated sliding watertight doors, hinged watertight access doors with open/closed indication locally and at the navigation bridge, of the quick-acting or single-action type that are normally closed at sea, hinged watertight doors that are permanently closed at sea, and sidescuttles of the non-opening type;"

DRAFT MODEL REGULATIONS ON DOMESTIC FERRY SAFETY

PREAMBLE

1 These model regulations (a) provide framework provisions on domestic ferry safety for incorporation into national law; (b) do not promulgate provisions on facilitation, security, and pollution; (c) are drafted in a way to enable easy translation and incorporation into national law; and (d) may serve as a basis for intergovernmental agreements, whether multilateral, bilateral or regional.

2 The incorporation of model regulations into national law by interested countries is purely voluntary and recommendatory and it is the prerogative of those countries to decide on their incorporation in their national law.

3 The modality through which these regulations are to be implemented is left to the competent authority, thereby encouraging, inter alia, indigenous solutions which may be low-cost yet fit-for-purpose, taking into account developments in the Organization. The Competent authority may delegate a responsibility to a recognized organization, an equivalent domestic entity or a nominated surveyor.

4 A domestic ferry when in compliance with these regulations may move from the domestic waters of one country directly to another subject to bilateral or multilateral agreement(s) between the countries concerned.

ARTICLE 1 Application

These regulations shall apply to domestic ferries which operate in domestic waters as identified by the competent authority and as shown in annex 1.

ARTICLE 2 General

1 These regulations shall be complied with by all entities associated with the construction, modification, conversion, management and operational safety of domestic ferry, and education and training of crew on board and shore personnel.

2 The competent authority may take measures to protect, preserve and promote safe and secure ways of development and construction of craft and means of transport in domestic waters.

Use of terminology

The words "shall", "should" and "may" when used in these regulations mean "mandatory", "recommendatory" and "optional", respectively.

Definitions

For the purposes of these regulations:

Certificated officer means an officer holding a valid certificate of competency issued by an accredited institution and recognized by the competent authority.

Chief Engineer means a certificated officer in possession of a certificate of competency corresponding to the rank and responsible for the efficient running and maintenance of all machinery and electrical equipment on the domestic ferry.

Competent authority means a government entity responsible for the implementation of these regulations.

Crew means any person including the master working on or operating the domestic ferry.

Dangerous goods mean any goods including personal belongings covered by the provisions of the International Maritime Dangerous Goods Code.

Deadweight tonnage means the maximum carrying capacity, in tonnes, of the domestic ferry.

Displacement means the mass of water, in tonnes, displaced at any particular draught.

Domestic ferry means a craft of any type and construction, using any means of propulsion, intended for the carriage of passengers and their belongings, including accompanied or unaccompanied freight units, over water only within domestic waters and certified as such by the competent authority.

Domestic waters mean waters in which domestic ferries may operate, clearly identified, designated and promulgated as such by the competent authority.

Draught means the vertical distance from the keel amidships to the water line.

Exception means any provision enacted by the competent authority through a decree which gives blanket relief to certain types of domestic ferries from compliance with these regulations.

Exemption means any provision enacted by the competent authority in writing which gives relief from any obligation or liability imposed by these regulations.

Gross tonnage means the measure of the overall size of a domestic ferry determined in accordance with the provisions of the present International Convention on Tonnage Measurement of Ships.

High-speed domestic ferry means a high-speed craft capable of carrying passengers and as defined in SOLAS, chapter X.

Marine casualty means an event, or a sequence of events, that has resulted in any of the following which has occurred directly in connection with the operations of a domestic ferry:

- .1 the death of, or serious injury to, a person;
- .2 the loss of a person from a domestic ferry;
- .3 the loss, presumed loss or abandonment of a domestic ferry;

- .4 material damage to a domestic ferry;
- .5 the stranding or disabling of a domestic ferry, or the involvement of a domestic ferry in a collision;
- .6 material damage to marine infrastructure external to a domestic ferry, that could seriously endanger the safety of the domestic ferry, another ship or an individual; or
- .7 severe damage to the environment, or the potential for severe damage to the environment, brought about by the damage of a domestic ferry or domestic ferries.

However, a marine casualty does not include a deliberate act or omission, with the intention to cause harm to the safety of a domestic ferry, an individual or the environment.

Marine incident means an event, or sequence of events, other than a marine casualty, which has occurred directly in connection with the operations of a domestic ferry that endangered, or, if not corrected, would endanger the safety of the domestic ferry, its occupants or any other person or the environment.

Master means an officer in possession of a certificate of competency corresponding to the rank and having command or charge of the domestic ferry.

Net tonnage means the measure of the useful capacity of a domestic ferry determined in accordance with the provisions of the present International Convention on Tonnage Measurement of Ships.

Nuclear domestic ferry means a domestic ferry provided with a nuclear power plant.

Organization means International Maritime Organization.

Passenger means any person on board the domestic ferry other than the master and crew members.

Recognized organization means an organization that has been assessed by, and found to comply with the standards acceptable to, the competent authority. A recognized organization may be authorized to carry out surveys, issue certificates and to undertake any other activity required per these regulations at the request and on behalf of the competent authority.

Seaworthy means fit to undertake the intended voyage without danger to the domestic ferry, environment or persons and complying with the provisions of these regulations in all aspects.

ARTICLE 3 Safety culture

The competent authority shall put in place effective mechanisms to promote and strengthen safety culture in all areas of domestic ferry operations.

ARTICLE 4 Surveys and certificates

1 All inspections, tests and surveys shall be carried out by the competent authority or a recognized organization, an equivalent domestic entity or a nominated surveyor. Upon satisfactory completion of the inspections, tests and surveys, certificates issued in association with such inspections, tests and surveys shall clearly and legibly state the name, functional title and contact information of the person undertaking such inspections, tests and surveys.

2 Certificates issued by the competent authority or a recognized organization, an equivalent domestic entity or a nominated surveyor shall be kept on board and readily available for inspection at short notice.

ARTICLE 5 New built domestic ferry

A new built domestic ferry shall comply with the following requirements, as appropriate:

Construction

A domestic ferry shall be issued with a safety construction certificate.

Stability standards

A domestic ferry shall comply with appropriate intact and damaged standards.

Stability management

The master shall be provided with an approved stability booklet and sufficient information in order to maintain stability and for damage control.

Fire detection system

A domestic ferry shall have adequate and approved fire detection system.

Fire extinguishing equipment

A domestic ferry shall have adequate and approved fire extinguishing equipment.

High-speed domestic ferry

A high-speed domestic ferry shall comply with SOLAS chapter X.

Inclining test

A domestic ferry shall undergo an inclining test upon completion of construction to determine its displacement and position of the centre of gravity for the lightship condition.

Life-saving equipment

A domestic ferry shall have adequate and approved life-saving equipment on board.

When determining the life-saving equipment to be carried on board a domestic ferry, the competent authority may consider the provisions of annex 2.

Load line

A domestic ferry shall be assigned a load line mark.

Markings and displays

A domestic ferry shall have the following marked or displayed, as appropriate, at a conspicuous place or places and clearly visible at all times:

- .1 name;
- .2 port of registry;
- .3 draught marks and scales;
- .4 load line mark;
- .5 deadweight tonnage;
- .6 gross tonnage;
- .7 net tonnage;
- .8 total number of passengers certified to carry;
- .9 number of passengers certified to carry on each deck;
- .10 number of life jackets required;
- .11 life-saving equipment location plan;
- .12 fire-extinguishing equipment location plan; and
- .13 navigational limits.

Nuclear domestic ferry

A nuclear domestic ferry shall comply with SOLAS chapter VIII.

Note: The domestic ferry shall retain on board certified copies of the above certificates, plans, markings and displays which shall be, where necessary, weatherproof and securely affixed to the domestic ferry at suitable place or places.

ARTICLE 6 Modification or conversion to domestic ferry

1 Any craft modified or converted to a domestic ferry shall comply with the requirements meant for new built domestic ferry.

2 Navigational routes shall be reassessed.

ARTICLE 7 Repair, modification and conversion

All repairs, modifications and conversions shall be carried out under the supervision of the competent authority or a recognized organization, an equivalent domestic entity or a nominated surveyor.

ARTICLE 8 Registration

1 The competent authority shall maintain a register of domestic ferries under its jurisdiction with the name and details of the domestic ferry, and the name and contact information of its owner.

2 All major repairs, modifications and conversions shall be recorded by the competent authority or the recognized organization, an equivalent domestic entity or a nominated surveyor in the file or in an equivalent document.

ARTICLE 9 Manning

1 The domestic ferry shall be manned will qualified, certificated, medically fit and properly rested seafarers.

2 A safe manning certificate shall be issued by the competent authority and available on board.

3 Crew shall be in possession of appropriate certification issued or recognized by the competent authority.

4 Crew shall be in possession of valid medical certificates prescribed by the competent authority prior to boarding the domestic ferry. A work arrangement schedule and record of rest hours shall be provided to ensure fitness for duty.

ARTICLE 10 Education and training

1 Education and training of shore staff and domestic ferry crew shall be provided by institutions which are duly accredited by the competent authority.

2 Education and training institutions accredited by the competent authority shall be audited by the competent authority at regular intervals to conform the institution's suitability for purpose.

3 Education and training of shore staff and domestic ferry crew may follow the training courses developed by the Organization or by the competent authority.

4 The competent authority may benefit from the provisions of explanatory notes, and guidelines or guidance prepared by the Organization when developing syllabuses for education and training of shore staff and domestic ferry crew.

ARTICLE 11 Safety management and governance

1 The competent authority shall ensure adequate provision of safety management systems and governance mechanisms ashore and on board.

2 The competent authority shall audit the safety management system to ensure its effectiveness.

3 The company shall ensure that safety management systems clearly establish that the master has the overriding authority and the responsibility to make decisions with respect to safety.

4 The company shall review safety management systems regularly and following any marine casualty or marine incident to ensure it remains effective in achieving its safety outcomes.

5 The owner, the charterer, the company operating the domestic ferry, or any other person shall not prevent or restrict the master of the domestic ferry from taking or executing any decision which, in the master's professional judgement, is necessary for safety of life at sea and protection of the marine environment. The master shall consider the checklist in annex 3 when deciding the seaworthiness of the domestic ferry.

ARTICLE 12 Occupational health and safety

The competent authority shall ensure adequate provision of occupational health and safety systems ashore and on board with emphasis on new and emerging health and safety issues.

ARTICLE 13 Navigation and radio communications

1 All equipment on board shall be certified by the competent authority or the recognized organization, an equivalent domestic entity or a nominated surveyor and be fully functional at the time of departure.

2 All domestic ferries should carry appropriate safety radio equipment which may comply with GMDSS requirements where appropriate.

Equipment

The competent authority shall establish the requirements for navigational and communication equipment for the domestic ferry taking into account its size, capacity and navigational limits.

Navigational charts

Navigational charts, including electronic versions, for the intended voyage shall be up to date.

Navigational routes

The competent authority should indicate regular navigational routes for high-speed domestic ferries.

Navigational limits

The competent authority should assign each domestic ferry navigational limits.

The domestic ferry shall not operate beyond its navigational limits other than in exceptional circumstances.

Weather

Approved weather monitoring equipment that is able to receive weather forecasts shall be fitted.

ARTICLE 14 Inspection and maintenance

The condition of the domestic ferry and its machinery and equipment shall be routinely and regularly inspected and maintained by qualified crew or operating personnel to confirm with the provisions of these regulations to ensure that the domestic ferry in all respects will remain fit to proceed without danger to the domestic ferry or persons on board.

ARTICLE 15 Stowage and securing

1 All personal belongings, baggage, cargo and vehicles shall be properly stowed and lashed before commencing the intended voyage so as to prevent as far as is practicable, throughout the voyage, damage or hazard to the domestic ferry and the persons on board, and loss of cargo overboard.

2 Dangerous goods shall not be allowed on board unless carried in compliance with relevant competent authority approved provisions.

ARTICLE 16 Embarkation and disembarkation

1 Adequate and appropriate safe gangways, fenders and safety nets shall be in place, where necessary, for the safety of the domestic ferry.

2 Adequate and safe means of access between the domestic ferry and the berth shall be provided.

ARTICLE 17 Prior to departure and arrival

1 Prior to departure from a port, the master shall confirm that the domestic ferry is in compliance with these regulations.

2 The master shall ensure that the crew is adequately rested and fit prior to the voyage.

3 Under no circumstances, the master shall, or be compelled to, take an unsafe or unseaworthy domestic ferry on any voyage.

Checklist

Prior to departure, the master shall complete a checklist, such as in annex 3, to ensure the domestic ferry's seaworthiness.

Crew and passengers

Prior to departure, all crew shall be accounted for by the master, and passengers and cargo shall be documented both on board and ashore.

The master shall ensure maximum number of passengers on each deck does not exceed assigned numbers.

Terms and conditions of carriage shall be clear and readily accessible.

Crew and passenger lists shall be clear and readily accessible.

Departure

The master shall be fully satisfied with safety matters on board and related external circumstances before taking the domestic ferry to sea.

The master shall not take an unsafe or unseaworthy domestic ferry on a voyage.

Fitness for the voyage

A valid certificate of fitness for the intended voyages issued by the competent authority shall be kept with the master.

Life-saving equipment

Before the domestic ferry leaves port and at all times during the voyage, all life-saving appliances shall be in working order and ready for immediate use.

Load line marking and draughts

The domestic ferry shall not sail with the appropriate load line mark submerged at any time during the voyage or on arrival.

Draught readings shall be taken by the master or a certificated officer and duly recorded in the logbook.

Safety briefing

Pre-departure safety briefing shall be conducted that shall include an abandon domestic ferry demonstration, donning and wearing of lifejackets and boarding of any lifesaving equipment or usage of floating devices as appropriate.

Weather bulletin

The master shall be provided with the latest weather bulletin to be encountered during the voyage.

The master shall not take the domestic ferry on the voyage in the absence of a latest weather bulletin or in receipt of an adverse weather forecast.

The competent authority shall not allow the domestic ferry to depart in or in the face of adverse weather.

Arrival

Prior to arrival the master shall conduct safety checks and make safety announcements.

The master shall be fully satisfied with safety matters on board and related external circumstances before bringing the domestic ferry to the port.

ARTICLE 18 Certificates

All certificates shall be issued by or on behalf of the competent authority or by a recognized organization, an equivalent domestic entity or a nominated surveyor.

ARTICLE 19 Exemption and exception

1 Under no circumstances shall the domestic ferry sail, or be allowed to sail, unless in full compliance with the provisions of these regulations or be in possession of a valid exemption certificate or covered by an exception decree, issued by the competent authority providing equivalent level of safety.

2 The exemption certificate shall clearly and explicitly show the reason for the exemption including the name, functional title and recognized and verifiable contact information of the authorized person issuing it. An exemption shall be verified by the master.

ARTICLE 20 Aids to navigation

1 The competent authority shall ensure provision of sufficient number of aids to navigation to facilitate safe navigation.

2 Such aids to navigation shall be fully functioning and regularly maintained.

3 Deficiencies in the functioning of the aids to navigation shall be promptly promulgated as Notices to Mariners and other appropriate means.

ARTICLE 21 Marine casualties and marine incidents

1 Any marine casualty or marine incident involving a domestic ferry shall be reported by the master to the competent authority as soon as practicable.

2 Any marine casualty or marine incident involving a domestic ferry shall be promptly investigated by the competent authority and duly recorded with emphasis on why and how it occurred.

3 The competent authority shall ensure appropriate measures are taken by all relevant parties to avoid similar or near similar future occurrences.

ARTICLE 22 Duty to render assistance

The master shall render assistance to any person in distress or who appears to be in distress at sea regardless of the nationality or status and in accordance with domestic and international law.

ARTICLE 23 Breach of regulations

The competent authority shall ensure appropriate national legal provisions are in place to take regulatory action where there is a breach of these regulations.

ARTICLE 24 Amendments

Amendments to these regulations shall be promulgated by the competent authority, according to national legislation.

ARTICLE 25 Implementation

In order to facilitate the implementation of and compliance with these regulations, the competent authority may issue guidance.

ARTICLE 26 Annexes

The annexes shall form an integral part of these regulations.

ANNEX 1 NAVIGATIONAL AREA

The competent authority shall insert:

- .1 Certified copy of domestic waters' of the country concerned chart with coordinates clearly shown.
- .2 Certified copy of the domestic ferry navigational area chart with coordinates clearly shown.

ANNEX 2 LIFE-SAVING EQUIPMENT

The life-saving equipment required by article 5 may include:

- .1 float-free Emergency Position Indicating Radio Beacon (EPIRB);
- .2 two-way Automatic Identification System (AIS);
- .3 waterproof floating hand-held VHF radio;
- .4 flares;
- .5 sufficient float-free survival craft to carry all permitted passengers plus 25%; and
- .6 sufficient approved life jackets for all permitted passengers plus 25%.

ANNEX 3 CHECKLIST

This declaration shall be signed by the Master and Chief Engineer. *Notes:*

1 Responses shall be "Yes" or "No" only.

- 2 If in doubt, the response shall be "No".
- 3 For question 8, insert "NA" if the domestic ferry is not subject to damaged stability.
- 4 If the response to any question is "No", the domestic ferry shall not sail nor the master instructed to sail, except with written permission to do so by a competent authority.

	Question	Response
1	Are the required certificates valid, including exemption certificate where required?	
2	Is the manning according to safe manning certificate?	
3	Are the crew adequately rested and fit for duties?	
4	Are there adequate and enough bunkers?	
5	Is the intended voyage within the assigned navigational area?	
6	Are the fire and bilge pumps in operational readiness?	
7	Is its stability sufficient per the stability booklet?	
8	Does it comply with the damage stability requirements?	
9	Are watertight doors and hatches closed and secured?	
10	Are the passengers distributed on various decks safely?	
11	Is it confirmed that the load line mark is not submerged?	
12	Are the draughts been taken and recorded?	
13	Is the navigational and communications equipment fully functional?	
14	Is the weather forecast confirmed for a safe journey?	
15	Are the crew and passengers accounted and reported for?	
16	Are the preparations been made for the safety briefing?	
17	Is all lifesaving equipment certified and ready for use?	
18	Is the vessel safe and seaworthy?	

Date, location, name of Master and signature

Date, location, name of Chief Engineer and signature

UPDATED WORKPLAN FOR THE WORK ON DOMESTIC FERRY SAFETY

	Event	Action	Comments
1	Interaction with teaching and research faculty at Dalian Maritime University, China (September 2019)	Shared the results of the broad analysis and basic structure of the model regulations.	Completed. Positive feedback.
2	Annual INTERFERRY conference, United Kingdom (October 2019)	Shared the results of the broad analysis and basic structure of the model regulations.	Completed. Positive feedback.
3	Lecture to students at IMLI, Malta (October 2019)	Shared the results of the broad analysis and basic structure of the model regulations.	Completed. Positive feedback.
4	Expert Group Meeting, Lagos, Nigeria (October 2019)	Event blended with regional workshop on safety of domestic ferries and non-convention ships.	Completed. Positive feedback.
5	Lecture to students at WMU, Sweden (November 2019)	Shared the results of the broad analysis and basic structure of the model regulations.	Completed. Positive feedback.
6	Expert Group Meeting (EGM) at United Nations ESCAP, Headquarters, Bangkok (March 2020)	Conducted expert group meeting on domestic ferry safety for Asia and the Pacific region jointly with ESCAP and in partnership with INTERFERRY.	Completed. Positive feedback. Bangkok Declaration (MSC 102/8/Add.1) adopted. Due to COVID-19 related restrictions, the EGM was successfully conducted in virtual mode. MSC 103 (May 2021) endorsed the Declaration.
7	MSC 101 (June 2019)	Agreed to include an item on "Measures to improve domestic ferry safety" in the agenda of MSC 102 with four sessions to complete the work and requested the Secretariat to lead the work in collaboration with interested parties.	Secretariat commenced work as requested by MSC 101.

	Event	Action	Comments
8	MSC 102 (consideration of agenda item postponed) (October 2020)	Update and submission of basic structure of draft Model Regulations on Domestic Ferry Safety (MRDFS).	Submitted documents MSC 102/8 and Add.1. Consideration of agenda item postponed to MSC 103. Secretariat continued with the development of the expanded structure of the model regulations.
9	MSC 103 (first session to consider agenda item) (May 2021)	Submission of expanded structure of draft MRDFS for consideration and approval, and updated work plan.	Completed (MSC 103/8). MSC 103 expressed general support for the MRDFS and agreed to establish a Working Group on Domestic Ferry Safety at MSC 104 to further develop the model regulations using MSC 103/8, annex 1, as basic document. Secretariat was requested to submit updated work plan for consideration by MSC 104.
10	MSC 104 (second session) (October 2021)	Envisaged approval of the MRDFS and consideration of updated work plan.	Updated work plan submitted (MSC 104/5). Following the envisaged approval of the MRDFS as further worked upon by the working group, Secretariat will commence development of explanatory manual to the MRDFS.
11	MSC 105 (third session) (scheduled for April 2022)	Envisaged adoption of MRDFS, and consideration of updated work plan as appropriate.	Secretariat will continue with the development of an explanatory manual to the MRDFS.
12	MSC 107 (fourth and final session) (first half of 2023)	Envisaged approval of explanatory manual to the MRDFS, and consideration of updated work plan, as appropriate.	Secretariat will submit explanatory manual to the MRDFS which upon approval may be circulated as MSC circular. Should the manual require further work, approval of additional sessions will be required, and work plan will be updated accordingly.
13	Technical Cooperation Committee (2023)	Progress report on the development of online training material on domestic ferry safety.	Secretariat will prepare or arrange the preparation of online training material to facilitate the implementation of the MRDFS.

	Event	Action	Comments
14	Interaction with stakeholders (ongoing)	Secretariat will remain engaged with IMLI, WMU and others to strengthen domestic ferry safety including the implementation of MRDFS.	

DRAFT ASSEMBLY RESOLUTION

PREVENTION AND SUPPRESSION OF PIRACY, ARMED ROBBERY AGAINST SHIPS AND ILLICIT MARITIME ACTIVITY IN THE GULF OF GUINEA

THE ASSEMBLY,

RECALLING Article 15(j) of the Convention on the International Maritime Organization concerning the functions of the Assembly in relation to regulations and guidelines concerning maritime safety and the prevention and control of marine pollution from ships,

CONSIDERING the relevant provisions of United Nations General Assembly resolution 55/2, adopted on 8 September 2000, on the United Nations Millennium Declaration and, in particular, section II on Peace, security and disarmament; section III on Development and poverty eradication; section IV on Protecting our common environment; and section VII on Meeting the special needs of Africa,

CONSIDERING ALSO the initiatives of the Zone of Peace and Cooperation in the South Atlantic, established by United Nations General Assembly resolution 41/11 and, in particular, the request to all relevant organizations of the United Nations to render all appropriate assistance that States may seek in their joint efforts to continue to implement the Montevideo Plan of Action, as included in United Nations General Assembly resolution 69/322,

FURTHER CONSIDERING the intensified need caused by the global pandemic to strengthen support for the Sustainable Development Goals (SDGs) through the effective implementation of wider initiatives such as:

- (a) the African Integrated Maritime Strategy for 2050 (AIMS 2050), including its relevant provisions on inter-agency and transnational cooperation and coordination on maritime safety and security; illegal oil bunkering/crude oil theft; illegal arms and drug trafficking; piracy and armed robbery at sea; maritime terrorism; human trafficking, human smuggling and asylum seekers travelling by sea;
- (b) the Economic Community of West African States (ECOWAS) Integrated Maritime Strategy (EIMS), including its strategic objectives on strengthening maritime governance, ensuring a safe and secure maritime domain and optimizing the ECOWAS maritime economy;
- the African Charter on Maritime Security and Safety and Development in Africa (Lomé Charter), including its objectives and measures to prevent and combat crimes at sea;
- (d) the African Maritime Transport Charter and the aspirations of the 2018 Sustainable Blue Economy Conference, held in Nairobi; and
- (e) the Global Maritime Security Conference (GMSC) hosted by Nigeria in 2019 on the challenges of maritime security in the Gulf of Guinea region and the development of tailored solutions,

REAFFIRMING ITS APPRECIATION of the actions taken by the Security Council of the United Nations ("the Security Council") and, in particular, the adoption of Security Council resolutions 2018 (2011) and 2039 (2012) in relation to piracy and armed robbery in the Gulf of Guinea,

NOTING IN PARTICULAR that resolution 2039 (2012) recognized the urgent need to devise and adopt effective and practical measures to counter piracy and armed robbery at sea in the Gulf of Guinea; reiterated the leadership role and responsibility of States in the region in countering the threat and addressing the underlying causes of piracy and armed robbery at sea in the Gulf of Guinea, in close cooperation with organizations in the region and their partners; emphasized the importance of building on existing national, regional and extra-regional initiatives to enhance maritime safety and security in the Gulf of Guinea; and welcomed the initiatives already taken by States in the region and regional organizations, including the Economic Community of Central African States (ECCAS), ECOWAS, the Gulf of Guinea Commission (GGC), and the Maritime Organisation for West and Central Africa (MOWCA), to enhance maritime safety and security in the Gulf of Guinea,

ALSO NOTING the presidential statement of the Security Council on 25 April 2016 condemning murders, kidnappings and hostage-taking occurring in the Gulf of Guinea; expressing its deep concern over piracy and armed robbery at sea in the area, and stressing the importance of a comprehensive approach – led by States of the region, with international support – to address the problem and its root causes,

NOTING FURTHER the presidential statement of the Security Council on 3 February 2021 expressing concern at the threat that piracy and armed robbery in the Gulf of Guinea pose to international navigation, security and development in the region and welcoming the work of the United Nations Office for West Africa and the Sahel (UNOWAS) on regional cooperation in response; underlining the need for security efforts to be aligned with political objectives, to promote the restoration of civilian security, the establishment of effective governance to deliver essential services, the administration of justice and accountability, and humanitarian assistance, and the revival of local economies to provide livelihood opportunities for surging youth populations,

REAFFIRMING ITS SUPPORT for, and continued development of, the Code of Conduct concerning the repression of piracy, armed robbery against ships and illicit maritime activity in west and central Africa (the Yaoundé Code of Conduct architecture), which has been adopted by 25 States, comprising:

- (a) the development of domestic laws and regulations, where these are not in place, criminalizing piracy and armed robbery at sea;
- (b) the development of a regional framework to counter piracy and armed robbery at sea, including information-sharing and operational coordination mechanisms in the region; and
- (c) the development and strengthening of domestic laws and regulations, as appropriate, to implement relevant international agreements addressing the safety and security of navigation, in accordance with international law,

RECALLING that the General Assembly of the United Nations, at its seventy-fifth session, adopted, on 31 December 2020, resolution 75/239 on Oceans and the Law of the Sea including its paragraphs 133, 134, 135, 138, 145, 146, 149, 151 and 153 which, inter alia:

- (a) emphasized the importance of promptly reporting incidents to enable accurate information on the scope of the problem of piracy and armed robbery against ships at sea and, in the case of armed robbery against ships at sea, by affected vessels to the coastal State; underlined the importance of effective information-sharing with States potentially affected by incidents of piracy and armed robbery against ships at sea; noted with appreciation the important role of the International Maritime Organization and the important contribution of the Information Sharing Centre of the Regional Cooperation Agreement on Combating Piracy and Armed Robbery against Ships in Asia, based in Singapore, which aspires to be recognized as a centre of excellence within its purpose and mandate; and noted the Maritime Domain Awareness for Trade Gulf of Guinea mechanism;
- (b) urged all States, in cooperation with the International Maritime Organization, to actively combat piracy and armed robbery at sea by adopting measures, including those relating to assistance with capacity-building through training of seafarers, port staff and enforcement personnel in the prevention, reporting and investigation of incidents, by bringing the alleged perpetrators to justice, in accordance with international law, and by adopting national legislation, as well as by providing enforcement vessels and equipment and guarding against fraudulent ship registration; and
- (c) encouraged States to ensure effective implementation of international law applicable to combating piracy, as reflected in the Convention, called upon States to take appropriate steps under their national law to facilitate, in accordance with international law, the apprehension and prosecution of those who are alleged to have committed acts of piracy, including the financing or facilitation of such acts, also taking into account other relevant instruments that are consistent with the Convention, and encouraged States to cooperate, as appropriate, with a view to developing their national legislation in this regard,

RECALLING ALSO article 100 of the United Nations Convention on the Law of the Sea (UNCLOS), which requires all States to cooperate to the fullest possible extent in the repression of piracy on the high seas or in any other place outside the jurisdiction of any State, article 101 which provides for the definition of piracy and the *Code of Practice for the Investigation of Crimes of Piracy and Armed Robbery Against Ships* (resolution A.1025(26) which provides for the definition of armed robbery against ships,

RECALLING FURTHER article 105 of UNCLOS, which, inter alia, provides that, on the high seas or in any other place outside the jurisdiction of any State, every State may seize a pirate ship or aircraft, or a ship or aircraft taken by piracy and under the control of pirates, and arrest the persons and seize the property on board,

REAFFIRMING:

- (a) resolution A.545(13) on *Measures to prevent acts of piracy and armed robbery against ships*, adopted on 17 November 1983;
- (b) resolution A.683(17) on *Prevention and suppression of acts of piracy and armed robbery against ships*, adopted on 6 November 1991;
- (c) resolution A.738(18) on *Measures to prevent and suppress piracy and armed robbery against ships*, adopted on 4 November 1993;

- (d) resolution A.923(22) on *Measures to prevent the registration of "phantom" ships*, adopted on 29 November 2001;
- (e) resolution A.1025(26) on Code of practice for the investigation of the crimes of piracy and armed robbery against ships, adopted on 2 December 2009;
- (f) resolution A.1069(28) on *Prevention and suppression of piracy, armed robbery against ships and illicit maritime activity in the Gulf of Guinea*, adopted on 5 February 2014;
- (g) resolution A.1117(30) on *IMO ship identification number scheme* adopted on 6 December 2017; and
- (h) resolution A.1142(31) on *Measures to prevent the fraudulent registration and fraudulent registries of ships*, adopted on 4 December 2019,

NOTING with satisfaction the actions taken by the Council, the Legal Committee, the Maritime Safety Committee, the Technical Cooperation Committee, the Facilitation Committee and the Secretary-General aiming at preventing and suppressing acts of piracy and armed robbery against ships,

BEING AWARE of the serious safety and security concerns of the industry and the seafaring community as a result of the attacks against ships sailing in the Gulf of Guinea,

MINDFUL OF the grave danger to life and the serious risks to navigational safety and the environment that attacks by pirates, armed robbers and other criminals may cause,

REITERATING its full respect for the sovereignty, sovereign rights, jurisdiction and territorial integrity of all States and the relevant provisions of international law, in particular UNCLOS,

HAVING CONSIDERED, in the light of the situation prevailing in the Gulf of Guinea, the recommendations of the Council at its thirty-fourth extraordinary session,

1 CONDEMNS AND DEPLORES piracy, armed robbery against ships and other illicit maritime activity, irrespective of where such acts have occurred or may occur;

2 EXPRESSES deep sympathy for the loss of seafarers' lives during attacks and while in captivity; for their plight while held hostage in appalling conditions; and for their families;

3 ACKNOWLEDGES the leadership role and responsibility of the States of the Gulf of Guinea to counter piracy, armed robbery against ships and illicit activity in their region;

4 EXPRESSES deep appreciation for the assistance already provided by the Organization to States in the Gulf of Guinea to help them repress piracy, armed robbery against ships and other illicit maritime activities in their region;

- 5 EXPRESSES ALSO deep appreciation for:
 - the initiatives of the G7++ Friends of the Gulf of Guinea (G7++FoGG) in collaboration with the Organization involving all stakeholders to support the effective implementation and the sustainability of the Yaoundé Code of Conduct, including its Interregional Coordination Centre (ICC);

- (b) the continuous efforts made by the region to curb piracy and armed robbery against ships in the Gulf of Guinea, including drafting anti-piracy laws, the Deep Blue project, and the establishment of the GoG Maritime Collaboration Forum (GoG-MCF/SHADE GoG) working with law enforcement and the Interregional Coordination Centre (ICC Yaoundé), utilizing useful platforms working in the region, such as the G7++ Friends of the Gulf of Guinea (FOGG) in support of the Framework;
- (c) the pilot case of the Coordinated Maritime Presences (CMP) concept in the Gulf of Guinea recently launched by the European Union which aims to provide further support to address the security challenges in the region affecting freedom of navigation, and to increase cooperation and partnership with the coastal States of the Gulf of Guinea as well as regional and maritime security organizations of the Yaoundé Architecture and stresses the need to ensure consistency with international law, in particular the rights of coastal States in the area;
- (d) the contribution by the Maritime Domain Awareness for Trade Gulf of Guinea (MDAT-GoG) in support of the Yaoundé process to maintain coherent maritime situational awareness in the central and west African maritime areas, with the ability to inform and support industry to avoid, deter, delay and respond to piracy attacks and armed robbery against ships; and the reports to inform the Organization in relation to this issue;
- (e) the efforts undertaken by the industry stakeholders to engage in proactive dialogue with Member States in the region to address maritime insecurity;
- (f) the efforts of those who have contributed towards resolving cases involving ships that had been hijacked and seafarers who had been held hostage; and
- (g) the financial contributions made to the IMO West and Central Africa Maritime Security Trust Fund, the in-kind contributions made to further the implementation of the Code of Conduct concerning the repression of piracy, armed robbery against ships and illicit maritime activity in west and central Africa and the work done by the Secretary-General and the Secretariat towards the implementation of the provisions therein;

6 APPEALS to States in the region, in close cooperation with international and regional organizations, and other interested States to take all measures possible within the provisions of international law, to ensure that:

- (a) all acts or attempted acts of piracy, armed robbery against ships and other illicit maritime activities are terminated forthwith and any plans for committing such acts are also forthwith abandoned; and
- (b) any hijacked ships, kidnapped seafarers held on them and any other persons on board such ships are immediately and unconditionally released and that no harm is caused to them;

7 URGES Governments to cooperate with and assist States in the Gulf of Guinea to develop their national and regional capabilities to improve maritime governance in waters under their jurisdiction; to prevent piracy, armed robbery against ships and other illicit maritime activities in accordance with international law, in particular UNCLOS; and to assist States to build capacity to interdict and bring to justice those who commit crimes. Such assistance might include strengthening of the legal frameworks, including anti-piracy laws and enforcement regulations; the training of national maritime law enforcement agencies; promoting anti-piracy and law enforcement coordination and cooperation procedures between and among States, regions, organizations and industry; and the sharing of information;

8 REQUESTS Governments to instruct national rescue coordination centres, information-sharing centres or other agencies involved on receipt of a report of an attack, to promptly initiate the transmission^{*} of relevant advice and warnings (through the World-Wide Navigation Warning Service; the International Enhance Group Call Service; or any other appropriate means) to ships sailing in the Gulf of Guinea, so as to warn shipping in the immediate area of the attack;

9 FURTHER URGES companies, masters and seafarers to continue the diligent application of existing IMO guidance and the effective implementation of Best Management Practices (BMP) West Africa (WA) which provides guidance to assist companies and seafarers to assess the risks associated with voyages through the Gulf of Guinea and mitigate any potential threats to their safety and well-being;

10 CALLS UPON Governments, in cooperation with the Organization and as may be requested by Governments of the region, to assist these efforts and to consider making financial contributions to the IMO West and Central Africa Maritime Security Trust Fund;

- 11 REQUESTS the Secretary-General to:
 - (a) transmit a copy of the present resolution to the Secretary-General of the United Nations for consideration and any further action he may deem appropriate; and to express to him, and through him to the Security Council, the gratitude and appreciation of the Organization for the actions he and the Security Council have taken on this issue to date;
 - (b) continue monitoring the situation in relation to threats to ships sailing in the Gulf of Guinea and report to the Council, the Maritime Safety Committee, the Legal Committee, the Technical Cooperation Committee and the Facilitation Committee, as and when appropriate, on developments and any further actions that may be required;
 - (c) continue to consult with interested Governments, industry and organizations in establishing a process and means by which technical assistance can be provided for the coastal States of the Gulf of Guinea to enhance their capacity in order to give effect to the present resolution, as appropriate;
 - (d) continue the efforts of the Organization to contribute to an effective response to acts and attempted acts of piracy, armed robbery against ships and illicit maritime activity, in particular in the Gulf of Guinea, including by promoting cooperation, coordination and avoidance of duplication of efforts among the States and organizations that provide or seek to provide assistance to the States in the region; and

Resolution MSC.305(87) on Guidelines on operational procedures for the promulgation of Maritime Safety Information concerning acts of piracy and piracy countermeasure operations.

(e) undertake, as and when necessary, any other actions which would promote the implementation of the provisions of this resolution or would be conducive to the efforts of those seeking to implement its provisions;

12 REQUESTS the Maritime Safety Committee to review the provisions of this resolution as well as any salient provisions of related resolutions which have been or may be adopted by the Security Council in this respect and to develop, where and when necessary, guidance and recommendations so as to enable Member Governments and the shipping industry to implement its provisions, taking into account current and emerging trends and practices;

13 REQUESTS the Legal Committee, the Technical Cooperation Committee and the Facilitation Committee to assist, as appropriate, the Maritime Safety Committee in pursuing the objectives of this resolution;

14 REQUESTS ALSO the Council to continue to monitor the situation in relation to acts and attempted acts of piracy and armed robbery against ships, illicit maritime activity and threats to vessels navigating in the Gulf of Guinea; and to initiate any actions which it may deem necessary, including coordinating the work of the competent Committees of the Organization to ensure the protection of seafarers and vessels navigating in those waters, and to ensure appropriate cooperation with other organizations and entities tasked with relevant activities.

DRAFT ASSEMBLY RESOLUTION

PROCEDURES FOR PORT STATE CONTROL, 2021

THE ASSEMBLY,

RECALLING Article 15(j) of the Convention on the International Maritime Organization regarding the functions of the Assembly in relation to regulations and guidelines concerning maritime safety and the prevention and control of marine pollution from ships,

RECALLING ALSO resolution A.1138(31), by which it adopted *Procedures for port State control, 2019 (*hereafter referred to as the "Procedures"), following successive revocation of resolutions A.1119(30), A.1052(27), A.882(21), A.787(19), A.742(18), A.597(15) and A.466(XII),

RECOGNIZING that efforts by port States have greatly contributed to enhanced maritime safety and security, and prevention of marine pollution,

RECOGNIZING ALSO the need to update the Procedures to take account of the amendments to IMO instruments which have entered into force or have become effective since the adoption of resolution A.1138(31),

HAVING CONSIDERED the recommendations made by the Maritime Safety Committee, at its 104th session and the Marine Environment Protection Committee, at its seventy-seventh session,

1 ADOPTS the *Procedures for port State control, 2021*, as set out in the annex to the present resolution;

2 INVITES Governments, when exercising port State control, to implement the aforementioned Procedures;

3 REQUESTS the Maritime Safety Committee and the Marine Environment Protection Committee to keep the Procedures under review and to amend them as necessary;

4 REVOKES resolution A.1138(31).

PROCEDURES FOR PORT STATE CONTROL, 2021

Contents

CHAPTER 1 – GENERAL

- 1.1 Purpose
- 1.2 Application
- 1.3 Introduction
- 1.4 Provision for port State control
- 1.5 Ships of non-Parties
- 1.6 Ships below convention size
- 1.7 Definitions
- 1.8 Professional profile of PSCOs
- 1.9 Qualification and training requirements of PSCOs

CHAPTER 2 – PORT STATE INSPECTIONS

- 2.1 General
- 2.2 Initial inspections
- 2.3 General procedural guidelines for PSCOs
- 2.4 Clear grounds
- 2.5 More detailed inspections

CHAPTER 3 – CONTRAVENTION AND DETENTION

- 3.1 Identification of a substandard ship
- 3.2 Submission of information concerning deficiencies
- 3.3 Port State action in response to alleged substandard ships
- 3.4 Responsibilities of port State to take remedial action
- 3.5 Guidance for the detention of ships
- 3.6 Suspension of inspection
- 3.7 Procedures for rectification of deficiencies and release

CHAPTER 4 – REPORTING REQUIREMENTS

- 4.1 Port State reporting
- 4.2 Flag State reporting
- 4.3 Reporting of allegations under MARPOL

CHAPTER 5 – REVIEW PROCEDURES

5.1 Report of comments

APPENDICES

- Appendix 1 Code of good practice for port State control officers conducting inspections within the framework of the regional memoranda of understanding and agreement on port State control
- Appendix 2 Guidelines for the detention of ships
- Appendix 3 Guidelines for investigations and inspections carried out under MARPOL Annex I
- Appendix 4 Guidelines for investigations and inspections carried out under MARPOL Annex II
- Appendix 5 Guidelines for discharge requirements under MARPOL Annexes I and II
- Appendix 6 Guidelines for more detailed inspections of ship structural and equipment requirements
- Appendix 7 Guidelines for control of operational requirements
- Appendix 8 Guidelines for port State control officers on the ISM Code
- Appendix 9 Guidelines for port State control related to LRIT
- Appendix 10 Guidelines for port State control under TONNAGE 1969
- Appendix 11 Guidelines for port State control officers on certification of seafarers, manning and hours of rest
- Appendix 12 List of certificates and documents
- Appendix 13 Report of inspection in accordance with procedures for port State control
- Appendix 14 Report of deficiencies not fully rectified or only provisionally rectified
- Appendix 15 Report of action taken to the notifying authority
- Appendix 16 Format for the Report of contravention of MARPOL (article 6)
- Appendix 17 Comments by flag State on detention report
- Appendix 18 Guidelines for port State control under MARPOL Annex VI
- Appendix 19 List of instruments relevant to port State control procedures

CHAPTER 1 – GENERAL

1.1 PURPOSE

This document is intended to provide basic guidance on the conduct of port State control inspections in support of the control provisions of relevant conventions and parts of the *IMO Instruments Implementation Code* (III Code) (resolution A.1070(28)) and afford consistency in the conduct of these inspections, the recognition of deficiencies of a ship, its equipment, or its crew, and the application of control procedures.

1.2 APPLICATION

- 1.2.1 These Procedures apply to ships falling under the provisions of:
 - .1 the International Convention for the Safety of Life at Sea, 1974, as amended (SOLAS 1974);
 - .2 the Protocol of 1988 relating to the International Convention for the Safety of Life at Sea, 1974, as amended (SOLAS PROT 1988);
 - .3 the International Convention on Load Lines, 1966, as amended (LL1966);
 - .4 the Protocol of 1988 relating to the International Convention on Load Lines, 1966, as amended (LL PROT 1988);
 - .5 the International Convention for the Prevention of Pollution from Ships, 1973, as modified by the 1978 and 1997 Protocols, as amended (MARPOL);
 - .6 the International Convention on Standards of Training, Certification and Watchkeeping for Seafarers, 1978, as amended (STCW 1978);
 - .7 the International Convention on Tonnage Measurement of Ships, 1969, as amended (TONNAGE 1969);
 - .8 the International Convention on the Control of Harmful Anti-fouling Systems on Ships, 2001 (AFS 2001);
 - .9 the Convention on the International Regulations for Preventing Collisions at Sea, 1972, as amended (COLREG 1972);
 - .10 the International Convention on Civil Liability for Oil Pollution Damage, 1969 (CLC 1969);
 - .11 the Protocol of 1992 to amend the International Convention on Civil Liability for Oil Pollution Damage, 1969, as amended (CLC PROT 1992);
 - .12 the International Convention on Civil Liability for Bunker Oil Pollution Damage, 2001 (BUNKERS 2001);
 - .13 the International Convention for the Control and Management of Ships' Ballast Water and Sediments, 2004, as amended (BWM 2004); and
 - .14 the Nairobi International Convention on the Removal of Wrecks, 2007 (NAIROBI WRC 2007),

hereafter referred to as the relevant conventions.

1.2.2 Ships of non-Parties should be given no more favourable treatment (see section 1.5).

1.2.3 For ships below convention size, Parties should apply the procedures in section 1.6.

1.2.4 When exercising port State control, Parties should only apply those provisions of the conventions which are in force and which they have accepted.

1.2.5 Where the provisions of the relevant conventions are not specific, the port State control officer (PSCO) should in principle accept the design arrangement approved by the flag State and when appropriate consult with the flag Administration.

1.2.6 The PSCO should be aware that the provisions of relevant conventions permit Administrations to grant exemptions, allow equivalents^{*} and approve alternative design and arrangements (ADA). When an Exemption Certificate is issued in accordance with the relevant provisions of a convention, provided it includes the correct reference to the exemption provision and the requirement to which it relates, or the ship carries the approved ADA documentation (e.g. SOLAS 1974 regulation II-1/55.4.2), port State authorities should interpret this as meaning that the ship complies with the provisions of the convention. Port State authorities should check, whenever possible, with the Administration should there be any doubt whether an exemption, equivalence or ADA has been granted.

1.2.7 Notwithstanding paragraph 1.2.4, in relation to voluntary early implementation of amendments to SOLAS 1974 and related mandatory instruments, Parties should take into account the *Guidelines on the voluntary early implementation of amendments to the 1974 SOLAS Convention and related mandatory instruments* (MSC.1/Circ.1565).

1.2.8 If a port State exercises control based on:

- .1 the International Labour Organization (ILO) Maritime Labour Convention, 2006, as amended (MLC 2006), guidance on the conduct of such inspections is given in the ILO publication "*Guidelines for port State control officers carrying out inspections under the Maritime Labour Convention, 2006, as amended*"; or
- .2 the ILO Convention No.147, Merchant Shipping (Minimum Standards) Convention, 1976, or the Protocol of 1996 to the Merchant Shipping (Minimum Standards) Convention, 1976, guidance on the conduct of such inspections is given in the ILO publication *"Inspection of labour conditions on board ship: Guidelines for procedure"*.

1.3 INTRODUCTION

1.3.1 Under the provisions of the relevant conventions set out in section 1.2 above, the Administration (i.e. the Government of the flag State) is responsible for promulgating laws and regulations and for taking all other steps which may be necessary to give the relevant conventions full and complete effect so as to ensure that, from the point of view of safety of life and pollution prevention, a ship is fit for the service for which it is intended and seafarers are qualified and fit for their duties.

Any Administration which allows, in substitution, a fitting, material, appliance or apparatus, or type thereof, or provision, shall communicate to the Organization particulars thereof together with a report on any trials made and the Organization shall circulate such particulars to other Contracting Governments for the information of their officers (e.g. SOLAS 1974, regulation I/5).

1.3.2 The nature of international shipping means that ships may not frequently call at ports in their flag State. It is therefore common to find that such flag States appoint the nominated surveyors at foreign ports and authorize recognized organizations (ROs) in accordance with the provisions of various conventions.

1.3.3 The following control procedures should be regarded as complementary to national measures taken by flag State Administrations in their countries and abroad and are intended to provide a common and consistent approach to the performance of port State control inspections and control measures taken as a consequence of the detection of serious deficiencies. These Procedures are also intended to provide assistance to flag State Administrations in securing compliance with convention provisions in safeguarding the safety of crew, passengers and ships, and ensuring the prevention of pollution.

1.4 PROVISION FOR PORT STATE CONTROL

SOLAS 1974 regulations I/19, IX/6.2, XI-1/4 and XI-2/9, as modified by SOLAS PROT 1988; article 21 of LL 1966, as modified by LL PROT 1988; articles 5 and 6, regulation 11 of Annex I, regulation 16.9 of Annex II, regulation 9 of Annex III, regulation 14 of Annex IV, regulation 9 of Annex V and regulation 10 of Annex VI of MARPOL; article X of STCW 1978; article 12 of TONNAGE 1969, article 11 of AFS 2001 and article 9 of BWM 2004 provide for control procedures to be followed by a Party to a relevant convention with regard to foreign ships visiting their ports. The authorities of port States should make effective use of these provisions for the purposes of identifying deficiencies, if any, in such ships which may render them substandard (see section 3.1) and ensuring that remedial measures are taken.

1.5 SHIPS OF NON-PARTIES

1.5.1 Article I(3) of SOLAS PROT 1988, article I(3) of LL PROT 1988, article 5(4) of MARPOL, article X(5) of STCW 1978, article 3(3) of AFS 2001 and article 3(3) of BWM 2004 provide that no more favourable treatment is to be given to the ships of countries which are not Party to the relevant convention. All Parties should, as a matter of principle, apply these Procedures to ships of non-Parties in order to ensure that equivalent surveys and inspections are conducted and an equivalent level of safety and protection of the marine environment is ensured.

1.5.2 As ships of non-Parties are not provided with SOLAS, Load Lines, MARPOL, AFS or BWM certificates, as applicable, or the crew members may not hold STCW certificates, the port State control officer (PSCO), taking into account the principles established in these Procedures, should be satisfied that the ship and crew do not present a danger to those on board or an unreasonable threat of harm to the marine environment. If the ship or crew has some form of certification other than that required by a convention, the PSCO may take the form and content of this documentation into account in the evaluation of that ship. The conditions of and on such a ship and its equipment and the certification of the crew and the flag State's minimum manning standard should be subject to such restrictions as are necessary to obtain a comparable level of safety and protection of the marine environment.

1.6 SHIPS BELOW CONVENTION SIZE

1.6.1 In the exercise of their functions, PSCOs should be guided by any certificates and other documents issued by or on behalf of the flag State Administration. In such cases, the PSCOs should limit the scope of inspection to the verification of compliance with those certificates and documents.
1.6.2 To the extent a relevant instrument is not applicable to a ship below convention size, the PSCO's task should be to assess whether the ship is of an acceptable standard in regard to safety and the environment. In making that assessment, the PSCO should take due account of such factors as the length and nature of the intended voyage or service, the size and type of the ship, the equipment provided and the nature of the cargo.

1.7 DEFINITIONS

1.7.1 **Bulk carrier:** While noting the definitions in SOLAS 1974, regulations IX/1.6 and XII/1.1 and resolution MSC.277(85), for the purposes of port State control, PSCOs should be guided by the ship's type indicated in the ship's certificates in determining whether a ship is a bulk carrier and recognize that a ship which is not designated as a bulk carrier as the ship type on the ship certificate may carry certain bulk cargo as provided for in the above instruments.

1.7.2 **Clear grounds:** Evidence that the ship, its equipment, or its crew do not correspond substantially with the requirements of the relevant conventions or that the master or crew members are not familiar with essential shipboard procedures relating to the safety of ships or the prevention of pollution. Examples of clear grounds are included in section 2.4.

1.7.3 **Deficiency:** A condition found not to be in compliance with the requirements of the relevant convention.

1.7.4 **Detention:** Intervention action taken by the port State when the condition of the ship or its crew does not correspond substantially with the relevant conventions to ensure that the ship will not sail until it can proceed to sea without presenting a danger to the ship or persons on board, or without presenting an unreasonable threat of harm to the marine environment, whether or not such action will affect the normal schedule of the departure of the ship.

1.7.5 **Initial inspection:** A visit on board a ship to check the validity of the relevant certificates and other documents, the overall condition of the ship, its equipment and its crew (see also section 2.2).

1.7.6 **More detailed inspection:** An inspection conducted when there are "clear grounds", as defined under paragraph 1.7.2.

1.7.7 **Nearest appropriate and available repair yard**: A port where follow-up action can be taken, and it is in, or closest to, the port of detention or the port where the ship is authorized to proceed taking into account the cargo on board.

1.7.8 **Port State control officer (PSCO):** A person duly authorized by the competent authority of a Party to a relevant convention to carry out port State control inspections, and responsible exclusively to that Party.

1.7.9 **Recognized organization (RO):** An organization which meets the relevant conditions set forth in the Code for Recognized Organizations (RO Code) (MSC.349(92) and MEPC.237(65)), and has been assessed and authorized by the flag State Administration in accordance with provisions of the RO Code to provide the necessary statutory services and certification to ships entitled to fly its flag.

1.7.10 **Stoppage of an operation:** Formal prohibition against a ship to continue an operation due to an identified deficiency or deficiencies which, singly or together, render the continuation of such operation hazardous.

1.7.11 **Substandard ship:** A ship whose hull, machinery, equipment or operational safety is substantially below the standards required by the relevant convention or whose crew is not in conformity with the safe manning document.

1.7.12 **Valid certificates:** A certificate that has been issued, electronically or on paper, directly by a Party to a relevant convention or on its behalf by an RO, contains accurate and effective dates, meets the provisions of the relevant convention and to which the particulars of the ship, its crew and its equipment correspond.

1.8 PROFESSIONAL PROFILE OF PSCOs

1.8.1 Port State control should be carried out only by qualified PSCOs who fulfil the qualifications and training specified in section 1.9.

1.8.2 When the required professional expertise cannot be provided by the PSCO, the PSCO may be assisted by any person with the required expertise, as acceptable to the port State.

1.8.3 PSCOs and persons assisting them should be free from any commercial, financial, and other pressures and have no commercial interest in the port of inspection, the ships inspected, ship repair facilities or any support services in the port or elsewhere, nor should PSCOs be employed by or undertake work on behalf of ROs or classification societies.

1.8.4 A PSCO should carry a personal document in the form of an identity card issued by the port State and indicating that the PSCO is authorized to carry out the control.

1.9 QUALIFICATION AND TRAINING REQUIREMENTS OF PSCOs

1.9.1 The PSCO should be an experienced officer qualified as flag State surveyor.

1.9.2 The PSCO should be able to communicate in English with the key crew.

1.9.3 Training should be provided for PSCOs to give the necessary knowledge of the provisions of the relevant conventions which are relevant to the conduct of port State control, taking into account the latest IMO Model Courses for port State control.

1.9.4 In specifying the qualifications and training requirements for PSCOs, the Administration should take into account, as appropriate, which of the internationally agreed instruments are relevant for control by the port State and the variety of types of ships which may enter its ports.

1.9.5 PSCOs carrying out inspections of operational requirements should be qualified as a master or chief engineer and have appropriate seagoing experience, or have qualifications from an institution recognized by the Administration in a maritime-related field and have specialized training to ensure adequate competence and skill, or be a qualified officer of the Administration with an equivalent level of experience and training, for performing inspections of the relevant operational requirements.

1.9.6 Periodic seminars for PSCOs should be held in order to update their knowledge with respect to instruments related to port State control.

CHAPTER 2 – PORT STATE INSPECTIONS

2.1 GENERAL

2.1.1 In accordance with the provisions of the relevant conventions, Parties may conduct inspections by PSCOs of foreign ships in their ports.

- 2.1.2 Such inspections may be undertaken:
 - .1 on the initiative of the Party;
 - .2 at the request of, or on the basis of information regarding a ship provided by, another Party; or
 - .3 on the basis of information regarding a ship provided by a member of the crew, a professional body, an association, a trade union or any other individual with an interest in the safety of the ship, its crew and passengers, or the protection of the marine environment.

2.1.3 Whereas Parties may entrust surveys and inspections of ships entitled to fly their own flag either to inspectors nominated for this purpose or to ROs, they should be aware that, under the relevant conventions, foreign ships are subject to port State control, including boarding, inspection, remedial action and possible detention, only by officers duly authorized by the port State. This authorization of PSCOs may be a general grant of authority or may be specific on a case-by-case basis.

2.1.4 All possible efforts should be made to avoid a ship being unduly detained or delayed. If a ship is unduly detained or delayed, it should be entitled to compensation for any loss or damage suffered.

2.2 INITIAL INSPECTIONS

2.2.1 In the pursuance of control procedures under the relevant conventions, which, for instance, may arise from information given to a port State regarding a ship, a PSCO may proceed to the ship and, before boarding, gain, from its appearance in the water, an impression of its standard of maintenance from such items as the condition of its paintwork, corrosion or pitting or unrepaired damage.

2.2.2 At the earliest possible opportunity, the PSCO should ascertain the type of ship, year of build and size of the ship for the purpose of determining which provisions of the conventions are applicable.

2.2.3 On boarding and introduction to the master or the responsible ship's officer, the PSCO should examine the ship's relevant certificates and documents required by the relevant conventions, as listed in appendix 12, part A. PSCOs should note the following:

- .1 certificates may be in hard copy or electronic form;
- .2 where the ship relies upon electronic certificates:
 - .1 the certificates and website used to access them should conform with the *Guidelines for the use of electronic certificates* (FAL.5/Circ.39/Rev.2 and Corr.1);

- .2 specific verification instructions are to be available on the ship; and
- .3 viewing such certificates on a computer is considered as meeting the requirement that certificates be "on board";
- .3 when examining International Tonnage Certificates, the PSCO should be guided by appendix 10; and
- .4 when examining certificates or documentary evidence of seafarers issued in accordance with STCW 1978, the PSCO should be guided by appendix 11; the list of certificates or documentary evidence required under STCW 1978 is also found in table B-I/2 of the STCW Code.

2.2.4 After the certificate and document check, the PSCO should check the overall condition of the ship, including its equipment, navigational bridge, forecastle, cargo holds/areas, engine-room and pilot transfer arrangements and verify that any outstanding deficiency from the previous PSC inspection has been rectified.

2.2.5 If the certificates required by the relevant conventions are valid and the PSCO's general impression and visual observations on board confirm a good standard of maintenance, the PSCO should generally confine the inspection to reported or observed deficiencies, if any.

2.2.6 In pursuance of control procedures under chapter IX of SOLAS 1974 in relation to the International Safety Management Code (ISM Code), the PSCO should utilize the guidelines in appendix 8.

2.2.7 If, however, the PSCO from general impressions or observations on board has clear grounds for believing that the ship, its equipment or its crew do not substantially meet the requirements, taking into account paragraph 1.2.6, the PSCO should proceed to a more detailed inspection, taking into consideration sections 2.4 and 2.5. In forming such an impression, the PSCO should utilize the guidelines in relevant appendices.

2.3 GENERAL PROCEDURAL GUIDELINES FOR PSCOs

2.3.1 The PSCO should observe the *Code of good practice for port State control officers* (MSC-MEPC.4/Circ.2), as shown in appendix 1, use professional judgement in carrying out all duties and consider consulting others as deemed appropriate.

2.3.2 When boarding a ship, the PSCO should present to the master or to the representative of the owner, if requested to do so, the PSCO identity card. This card should be accepted as documented evidence that the PSCO in question is duly authorized by the Administration to carry out port State control inspections.

2.3.3 If the PSCO has clear grounds for carrying out a more detailed inspection, the master should be immediately informed of these grounds and advised that, if so desired, the master may contact the Administration or, as appropriate, the RO responsible for issuing the certificate and invite their presence on board.

2.3.4 In the case that an inspection is initiated based on a report or complaint, especially if it is from a crew member, the source of the information should not be disclosed.

2.3.5 When exercising control, all possible efforts should be made to avoid a ship being unduly detained or delayed. It should be borne in mind that the main purpose of port State control is to prevent a substandard ship proceeding to sea. The PSCO should exercise professional judgement to determine whether to detain a ship until the deficiencies are corrected or to allow it to sail with certain deficiencies, having regard to the particular circumstances of the intended voyage.

2.3.6 It should be recognized that all equipment is subject to failure and spares or replacement parts may not be readily available. In such cases, undue delay should not be caused if, in the opinion of the PSCO, safe alternative arrangements have been made.

2.3.7 Where the grounds for detention are the result of accidental damage suffered to a ship, no detention order should be issued, provided that:

- .1 due account has been given to the convention requirements regarding notification to the flag State Administration, the nominated surveyor or the RO responsible for issuing the relevant certificate;
- .2 prior to entering a port, the master or company has submitted to the port State authority details of the circumstances of the accident and the damage suffered and information about the required notification of the flag State Administration;
- .3 appropriate remedial action, to the satisfaction of the port State authority, is being taken by the ship; and
- .4 the port State authority has ensured, having been notified of the completion of the remedial action, that deficiencies which were clearly hazardous to safety, health or environment have been rectified.

2.3.8 Since detention of a ship is a serious matter involving many issues, it may be in the best interest of the PSCO to act together with other interested parties (see paragraph 4.1.3). For example, the officer may request the owner's representatives to provide proposals for correcting the situation. The PSCO should also consider cooperating with the flag State Administration's representatives or the RO responsible for issuing the relevant certificates, and consulting them regarding their acceptance of the owner's proposals and their possible additional requirements. Without limiting the PSCO's discretion in any way, the involvement of other parties could result in a safer ship, avoid subsequent arguments relating to the circumstances of the detention and prove advantageous in the case of litigation involving "undue delay".

2.3.9 Where deficiencies cannot be remedied at the port of inspection, the PSCO may allow the ship to proceed to another port, subject to any appropriate conditions determined. In such circumstances, the PSCO should ensure that the competent authority of the next port of call and the flag State are notified.

2.3.10 Detention reports to the flag State should be in sufficient detail for an assessment to be made of the severity of the deficiencies giving rise to the detention.

2.3.11 The company or its representative have a right of appeal against a detention taken by the authority of a port State. The appeal should not cause the detention to be suspended. The PSCO should properly inform the master of the right of appeal.

2.3.12 To ensure consistent enforcement of port State control requirements, PSCOs should carry an extract of section 2.3 (General procedural guidelines for PSCOs) for ready reference when carrying out any port State control inspections.

2.3.13 PSCOs should also be familiar with the detailed guidelines given in the appendices to these Procedures.

2.4 CLEAR GROUNDS

2.4.1 When a PSCO inspects a foreign ship which is required to hold a convention certificate, and which is in a port or an offshore terminal under the jurisdiction of the port State, any such inspection should be limited to verifying that there are on board valid certificates and other relevant documentation and the PSCO forming an impression of the overall condition of the ship, its equipment and its crew, unless there are "clear grounds" for believing that the condition of the ship or its equipment does not correspond substantially with the particulars of the certificates.

- 2.4.2 "Clear grounds" to conduct a more detailed inspection include but are not limited to:
 - .1 the absence of principal equipment or arrangements required by the relevant conventions, taking into account paragraph 1.2.6;
 - .2 evidence from a review of the ship's certificates that a certificate or certificates are invalid;
 - .3 evidence that certificates and documents required by the relevant conventions and listed in appendix 12, part A are not on board, incomplete, not maintained or are falsely maintained;
 - .4 evidence from the PSCO's general impressions and observations that serious hull or structural deterioration or deficiencies exist that may place at risk the structural, watertight or weathertight integrity of the ship;
 - .5 evidence from the PSCO's general impressions or observations that serious deficiencies exist in the safety, pollution prevention or navigational equipment;
 - .6 information or evidence that the master or crew is not familiar with essential shipboard operations relating to the safety of ships or the prevention of pollution, or that such operations have not been carried out;
 - .7 indications that key crew members may not be able to communicate with each other or with other persons on board;
 - .8 the emission of false distress alerts not followed by proper cancellation procedures; and
 - .9 receipt of a report or complaint containing information that a ship appears to be substandard.

2.5 MORE DETAILED INSPECTIONS

2.5.1 If the ship does not carry valid certificates, or if the PSCO, from general impressions or observations on board, has clear grounds for believing that the condition of the ship or its equipment does not correspond substantially with the particulars of the certificates or that the master or crew is not familiar with essential shipboard procedures, a more detailed inspection, as described in this chapter, should be carried out, utilizing relevant appendices.

2.5.2 Support during the more detailed inspection could be found in the documents mentioned in appendix 12, part B, where applicable.

2.5.3 It is not envisaged that all of the equipment and procedures outlined in this chapter would be checked during a single port State control inspection, unless the condition of the ship or the familiarity of the master or crew with essential shipboard procedures necessitates such

a detailed inspection. In addition, these procedures are not intended to impose the seafarer certification programme of the port State on a ship entitled to fly the flag of another Party to STCW 1978 or to impose control procedures on foreign ships in excess of those imposed on ships of the port State.

CHAPTER 3 – CONTRAVENTION AND DETENTION

3.1 IDENTIFICATION OF A SUBSTANDARD SHIP

3.1.1 In general, a ship is regarded as substandard if the hull, machinery, equipment or operational safety and the protection of the environment is substantially below the standards required by the relevant conventions or if the crew is not in conformity with the safe manning document, owing to, inter alia:

- .1 the absence of principal equipment or arrangement required by the conventions, taking into account paragraph 1.2.6;
- .2 non-compliance of equipment or arrangement with relevant specifications of the conventions, taking into account paragraph 1.2.6;
- .3 substantial deterioration of the ship or its equipment;
- .4 insufficiency of operational proficiency, or unfamiliarity with essential operational procedures by the crew; and
- .5 insufficiency of manning or insufficiency of certification of seafarers.

3.1.2 If these evident factors as a whole or individually pose a danger to the ship or persons on board or present an unreasonable threat of harm to the marine environment if it were allowed to proceed to sea, it should be regarded as a substandard ship. The PSCO should also take into account the guidelines in appendix 2.

3.2 SUBMISSION OF INFORMATION CONCERNING DEFICIENCIES

3.2.1 Information that a ship appears to be substandard could be submitted to the appropriate authorities of the port State (see section 3.3) by a member of the crew, a professional body, an association, a trade union or any other individual with an interest in the safety of the ship, its crew and passengers, or the protection of the marine environment.

3.2.2 This information should be submitted in writing to permit proper documentation of the case and of the alleged deficiencies. If the information is passed verbally, the filing of a written report should be required, identifying, for the purposes of the port State's records, the individual or body providing the information. The attending PSCO may collect this information and submit it as part of the PSCO's report if the originator is unable to do so.

3.2.3 Information which may cause an investigation should be submitted as early as possible, giving adequate time to the authorities to act as necessary.

3.2.4 Each Party to the relevant convention should determine which authorities should receive information on substandard ships and initiate action. Measures should be taken to ensure that information submitted to the wrong department is promptly passed on by such department to the appropriate authority for action.

3.3 PORT STATE ACTION IN RESPONSE TO ALLEGED SUBSTANDARD SHIPS

3.3.1 On receipt of information about an alleged substandard ship or alleged pollution risk, the authorities should immediately investigate the matter and take the action required by the circumstances in accordance with the preceding sections.

3.3.2 Authorities which receive information about a substandard ship that could give rise to detention should forthwith notify any maritime, consular and/or diplomatic representatives of the flag State in the area of the ship and request them to initiate or cooperate with investigations. Likewise, the RO which has issued the relevant certificates on behalf of the flag State should be notified. These provisions will not, however, relieve the authorities of the port State, being a Party to a relevant convention, of the responsibility for taking appropriate action in accordance with its powers under the relevant conventions.

3.3.3 If the port State receiving information is unable to take action because there is insufficient time or no PSCOs can be made available before the ship sails, the information should be passed to the authorities of the country of the next appropriate port of call, to the flag State and also to the RO in that port, where appropriate.

3.4 RESPONSIBILITIES OF PORT STATE TO TAKE REMEDIAL ACTION

If a PSCO determines that a ship can be regarded as substandard as specified in section 3.1 and appendix 2, the port State should immediately ensure that corrective action is taken to safeguard the safety of the ship and passengers and/or crew and eliminate any threat of harm to the marine environment before permitting the ship to sail.

3.5 GUIDANCE FOR THE DETENTION OF SHIPS

Notwithstanding the fact that it is impracticable to define a ship as substandard solely by reference to a list of qualifying defects, guidance for the detention of ships is given in appendix 2.

3.6 SUSPENSION OF INSPECTION

3.6.1 In exceptional circumstances where, as a result of a more detailed inspection, the overall condition of a ship and its equipment, also taking into account the crew conditions, is found to be obviously substandard, the PSCO may suspend an inspection.

3.6.2 Prior to suspending an inspection, the PSCO should have recorded detainable deficiencies in the areas set out in appendix 2, as appropriate.

3.6.3 The suspension of the inspection may continue until the responsible parties have taken the steps necessary to ensure that the ship complies with the requirements of the relevant instruments.

3.6.4 In cases where the ship is detained and an inspection is suspended, the port State authority should notify the responsible parties without delay. The notification should include information about the detention, and state that the inspection is suspended until that authority has been informed that the ship complies with all relevant requirements.

3.7 PROCEDURES FOR RECTIFICATION OF DEFICIENCIES AND RELEASE

3.7.1 The PSCO should endeavour to secure the rectification of all deficiencies detected.

3.7.2 In the case of deficiencies which are clearly hazardous to safety or the environment, the PSCO should, except as provided in paragraph 3.7.3, ensure that the hazard is removed

before the ship is allowed to proceed to sea. For this purpose, appropriate action should be taken, which may include detention or a formal prohibition of a ship to continue an operation due to established deficiencies which, individually or together, would render the continued operation hazardous.

3.7.3 Where deficiencies which caused a detention, as referred to in paragraph 3.7.2, cannot be remedied in the port of inspection, the port State authority may allow the ship concerned to proceed to the nearest appropriate repair yard available, as chosen by the master and agreed to by that authority, provided that the conditions agreed between the port State authority and the flag State are complied with. Such conditions will ensure that the ship should not sail until it can proceed without risk to the safety of the passengers or crew, or risk to other ships, or without presenting an unreasonable threat of harm to the marine environment. Such conditions may include confirmation from the flag State that remedial action has been taken on the ship in question. In such circumstances the port State authority should notify the authority as appropriate. Notification to authorities should be made in the form shown in appendix 14. The authority receiving such notification should inform the notifying authority of action taken and may use the form shown in appendix 15.

3.7.4 On the condition that all possible efforts have been made to rectify all other deficiencies, except those referred to in paragraphs 3.7.2 and 3.7.3, the ship may be allowed to proceed to a port where any such deficiencies can be rectified.

3.7.5 If a ship referred to in paragraph 3.7.3 proceeds to sea without complying with the conditions agreed to by the authority of the port of inspection, that port State authority should immediately alert the next port, if known, the flag State and all other authorities it considers appropriate.

3.7.6 If a ship referred to in paragraph 3.7.3 does not call at the nominated repair port, the port State authority of the repair port should immediately alert the flag State and detaining port State, which may take appropriate action, and notify any other authority it considers appropriate.

CHAPTER 4 – REPORTING REQUIREMENTS

4.1 PORT STATE REPORTING

4.1.1 Port State authorities should ensure that, at the conclusion of an inspection, the master of the ship is provided with a document showing the results of the inspection, details of any action taken by the PSCO, and a list of any corrective action to be initiated by the master and/or company. Such reports should be made in accordance with the format in appendix 13.

4.1.2 Where, in the exercise of port State control, a Party denies a foreign ship entry to the ports or offshore terminals under its jurisdiction, whether or not as a result of information about a substandard ship, it should forthwith provide the master and flag State with reasons for the denial of entry.

4.1.3 In the case of a detention, at least an initial notification should be made to the flag State Administration as soon as practicable (see paragraphs 2.3.8 and 3.3.2). If such notification is made verbally, it should be subsequently confirmed in writing. As a minimum, the notification should include details of the ship's name, the IMO number, copies of Forms A and B as set out in appendix 13, time of detention and copies of any detention order. Likewise, the ROs which have issued the relevant certificates on behalf of the flag State should be notified, where appropriate. The parties above should also be notified in writing of the release

of detention. As a minimum, this information should include the ship's name, the IMO number, the date and time of release and a copy of Form B as set out in appendix 13.

4.1.4 If the ship has been allowed to sail with known deficiencies, the authorities of the port State should communicate all the facts to the authorities of the country of the next appropriate port of call, to the flag State, and to the RO, where appropriate.

4.1.5 Parties to a relevant convention, when they have exercised control giving rise to detention, should submit to the Organization reports in accordance with SOLAS 1974 regulation I/19, article 11 of MARPOL, or article X(3) of STCW 1978. Such deficiency reports should be made in accordance with the form given in appendices 13 or 16, as appropriate, or may be submitted electronically by the port State or a regional PSC regime.

4.1.6 Copies of such deficiency reports should, in addition to being forwarded to the Organization, be sent by the port State without delay to the authorities of the flag State and, where appropriate, to the RO which had issued the relevant certificate. Deficiencies found which are not related to the relevant conventions, or which involve ships of non-Parties or below convention size, should be submitted to flag States and/or to appropriate organizations but not to IMO.

4.1.7 Relevant telephone numbers and addresses of flag States' headquarters to which reports should be sent as outlined above, as well as addresses of flag State offices which provide inspection services should be provided to the Organization.^{*}

4.2 FLAG STATE REPORTING

4.2.1 On receiving a report on detention, the flag State and, where appropriate, the RO through the flag State Administration, should, as soon as possible, inform the Organization of remedial action taken in respect of the detention, which may be submitted electronically by the flag State to the Global Integrated Ship Information System (GISIS) or in a format shown in appendix 17.

4.2.2 Relevant telephone numbers and addresses of port State control offices, headquarters and those who provide inspection services should be provided to the Organization.

4.3 **REPORTING OF ALLEGATIONS UNDER MARPOL**

4.3.1 A report on alleged deficiencies or on alleged contravention of the discharge provisions relating to the provisions of MARPOL should be forwarded to the flag State as soon as possible, preferably no later than 60 days after the observation of the deficiencies or contravention. Such reports may be made in accordance with the format in appendices 13 or 16, as appropriate. If a contravention of the discharge provisions is suspected, then the information should be supplemented by evidence of violations which, as a minimum, should include the information specified in parts 2 and 3 of appendices 3 and 4 of these Procedures.

4.3.2 On receiving a report on alleged deficiencies or alleged contravention of the discharge provisions, the flag State and, where appropriate, the RO through the flag State Administration, should, as soon as possible, inform the Party submitting the report of immediate action taken in respect of the alleged deficiencies or contravention. That Party and the Organization should,

^{*} Such addresses are available in MSC-MEPC.6/Circ.19 (*National contact points for safety and pollution prevention and response*), which may be amended, the IMO Internet home page and the GISIS module on contact points (http://gisis.imo.org/Public).

upon completion of such action, be informed of the outcome and details, where appropriate, be included in the mandatory annual report to the Organization.

CHAPTER 5 – REVIEW PROCEDURES

5.1 **REPORT OF COMMENTS**

5.1.1 In the interest of making information regarding deficiencies and remedial measures generally available, a summary of such reports should be made by the Organization in a timely manner in order that the information can be disseminated in accordance with the Organization's procedures to all Parties to the relevant conventions. In the summary of deficiency reports, an indication should be given of flag State action or whether a comment by the flag State concerned is outstanding.

5.1.2 The appropriate committee should periodically evaluate the summary of the deficiency reports in order to identify measures that may be necessary to ensure more consistent and effective application of IMO instruments, paying close attention to the difficulties reported by Parties to the relevant conventions, particularly in respect of developing countries in their capacity as port States.

5.1.3 Recommendations to address such difficulties, when recognized by the appropriate committee, should, where appropriate, be incorporated into the relevant IMO instrument and any modifications relating to the procedures and obligations should be made in the port State documentation.

Appendix 1

CODE OF GOOD PRACTICE FOR PORT STATE CONTROL OFFICERS CONDUCTING INSPECTIONS WITHIN THE FRAMEWORK OF THE REGIONAL MEMORANDA OF UNDERSTANDING AND AGREEMENT ON PORT STATE CONTROL (MSC-MEPC.4/Circ.2)

Introduction

1 This Code provides guidelines regarding the standards of integrity, professionalism and transparency that regional port State control (PSC) regimes expect of all port State control officers (PSCOs) who are involved in or associated with port State control inspections.

Objective

2 The objective of this Code is to assist PSCOs in conducting their inspections to the highest professional level. PSCOs are central to achieving the aims of the regional PSC regime. They are the daily contact with the shipping world. They are expected to act within the law, within the rules of their Government and in a fair, open, impartial and consistent manner.

Fundamental principles of the Code

3 The Code of good practice encompasses three fundamental principles against which all actions of PSCOs are judged: integrity, professionalism and transparency. These are defined as follows:

- .1 integrity is the state of moral soundness, honesty and freedom from corrupting influences or motives;
- .2 professionalism is applying accepted professional standards of conduct and technical knowledge. For PSCOs, standards of behaviour are established by the maritime authority and the general consent of the port State members; and
- .3 transparency implies openness and accountability.

4 The list of the actions and behaviour expected of PSCOs in applying these principles is set out in the annex to this appendix.

5 Adhering to professional standards provides greater credibility to PSCOs and places more significance on their findings.

6 Nothing in the Code shall absolve PSCOs from complying with the specific requirements of the PSC instruments and applicable national laws.

Annex

CODE OF GOOD PRACTICE FOR PORT STATE CONTROL OFFICERS

Actions and behaviour of PSCOs

PSCOs should:

1 use their professional judgement in carrying out their duties;

Respect

- 2 remember that a ship is a home as well as a workplace for the ship's personnel and not unduly disturb their rest or privacy;
- 3 comply with any ship housekeeping rules such as removing dirty shoes or work clothes;
- 4 not be prejudiced by the race, gender, religion or nationality of the crew when making decisions and treat all personnel on board with respect;
- 5 respect the authority of the master or their deputy;
- 6 be polite but professional and firm as required;
- 7 never become threatening, abrasive or dictatorial or use language that may cause offence;
- 8 expect to be treated with courtesy and respect;

Conduct of inspections

- 9 comply with all health and safety requirements of the ship and their Administration, e.g. wearing of personal protective clothing, and not take any action or cause any action to be taken which could compromise the safety of the PSCO or the ship's crew;
- 10 comply with all security requirements of the ship and wait to be escorted around the ship by a responsible person;
- 11 present their identity cards to the master or the representative of the owner at the start of the inspection;
- 12 explain the reason for the inspections; however, where the inspection is triggered by a report or complaint they must not reveal the identity of the person making the complaint;
- 13 apply the procedures of PSC and the convention requirements in a consistent and professional way and interpret them pragmatically when necessary;
- 14 not try to mislead the crew, for example by asking them to do things that are contrary to the relevant conventions;

- 15 request the crew to demonstrate the functioning of equipment and operational activities, such as drills, and not make tests themselves;
- 16 seek advice when they are unsure of a requirement or of their findings rather than making an uninformed decision, for example by consulting colleagues, publications, the flag Administration, the recognized organization;
- 17 where it is safe to do so accommodate the operational needs of the port and the ship;
- 18 explain clearly to the master the findings of the inspection and the corrective action required and ensure that the report of inspection is clearly understood;
- 19 issue to the master a legible and comprehensible report of inspection before leaving the ship;

Disagreements

- 20 deal with any disagreement over the conduct or findings of the inspection calmly and patiently;
- 21 advise the master of the complaints procedure in place if the disagreement cannot be resolved within a reasonable time;
- 22 advise the master of the right of appeal and relevant procedures in the case of detention;

Integrity

- 23 be independent and not have any commercial interest in their ports and the ships they inspect or companies providing services in their ports. For example, PSCOs should not be employed, even on an occasional basis, by companies which operate ships in their ports or PSCOs should not have an interest in the repair companies in their ports;
- 24 be free to make decisions based on the findings of their inspections and not on any commercial considerations of the port;
- 25 always follow the rules of their Administrations regarding the acceptance of gifts and favours, e.g. meals on board;
- firmly refuse any attempts of bribery and report any blatant cases to the maritime authority;
- 27 not misuse their authority for benefit, financial or otherwise; and

Updating knowledge

28 update their technical knowledge regularly.

Appendix 2

GUIDELINES FOR THE DETENTION OF SHIPS

1 Principles governing rectification of deficiencies or detention of a ship

1.1 In taking a decision concerning the rectification of a deficiency or detention of a ship, the port State control officer (PSCO) will have to take into consideration the results of the more detailed inspection carried out in accordance with paragraph 2.5 of the procedures and guidelines contained in this appendix.

1.2 The PSCO will exercise professional judgement in determining whether to detain the ship until the deficiencies are rectified or to allow the ship to sail with certain deficiencies without unreasonable danger to safety, health, or the environment, having also considered the particular circumstances of the intended voyage.

2 Detention related to minimum safe manning and STCW certification

Before detaining a ship for the reasons of not operating at appropriate established minimum safe manning and STCW certification, the following will have to be considered, giving due regard to the points listed under areas under STCW 1978:

- .1 length and nature of the intended voyage or service;
- .2 whether or not the deficiency poses a danger to ships, persons on board or the environment;
- .3 whether or not appropriate hours of rest for the crew have been recorded and there is evidence that the minimum hours of rest have repeatedly not been kept;
- .4 ship's size and type and equipment provided; and
- .5 nature of cargo.

3 **Procedures for the detention of ships of all sizes**

3.1 When exercising professional judgement as to whether or not a ship should be detained, the PSCO will apply the following criteria:

- .1 timing: ships which are unsafe to proceed to sea will be detained upon the first inspection, irrespective of the time the ship will stay in port; and
- .2 re-inspection criterion: the ship will be detained if the deficiencies on a ship are sufficiently serious to merit a PSCO returning to the ship to be satisfied that they have been rectified before the ship sails.

3.2 The need for the PSCO to return to the ship classifies the seriousness of the deficiencies.

3.3 When deciding whether the deficiencies found in a ship are sufficiently serious to merit detention, the PSCO should assess whether:

- .1 the ship has relevant, valid documentation; and
- .2 the ship has the crew required in the minimum safe manning document or equivalent.

3.4 During inspection, the PSCO should further assess whether the ship and/or crew, throughout its forthcoming voyage, is able to:

- .1 navigate safely;
- .2 safely handle, carry and monitor the condition of the cargo;
- .3 operate the engine-room safely;
- .4 maintain proper propulsion and steering;
- .5 fight fires effectively in any part of the ship if necessary;
- .6 abandon ship speedily and safely and effect rescue if necessary;
- .7 prevent pollution of the environment;
- .8 maintain adequate stability;
- .9 maintain adequate watertight integrity;
- .10 communicate in distress situations if necessary; and
- .11 provide safe and healthy conditions on board.

3.5 If the result of any of these assessments is negative, taking into account all deficiencies found, the ship should be strongly considered for detention. A combination of deficiencies of a less serious nature may also warrant the detention of the ship.

4 General

The lack of valid certificates as required by the relevant conventions may warrant the detention of ships. However, ships flying the flag of States not a Party to a convention or not having implemented another relevant instrument are not entitled to carry the certificates provided for by the convention or other relevant instrument. Therefore, absence of the required certificates should not by itself constitute a reason to detain these ships; however, in applying the "no more favourable treatment" clause, substantial compliance with the provisions and criteria specified in these Procedures must be required before the ship sails.

5 Detainable deficiencies

To assist the PSCO in the use of these Guidelines, there follows a list of deficiencies, grouped under relevant conventions and/or codes, which are considered to be of such a serious nature that they may warrant the detention of the ship involved. This list is not considered exhaustive, but is intended to give examples of relevant items. However, the detainable deficiencies in the area of STCW 1978, listed below, are the only grounds for detention under this Convention.

Areas under SOLAS 1974

- 1 Failure of proper operation of propulsion and other essential machinery, as well as electrical installations.
- 2 Insufficient cleanliness of engine-room, excess amount of oily-water mixture in bilges, insulation of piping including exhaust pipes in engine-room contaminated by oil, and improper operation of bilge pumping arrangements.
- 3 Failure of the proper operation of emergency generator, lighting, batteries and switches.
- 4 Failure of proper operation of the main and auxiliary steering gear.
- 5 Absence, failure, insufficient capacity or serious deterioration of personal life-saving appliances, survival craft and launching and recovery arrangements (see also MSC.1/Circ.1490/Rev.1).
- 6 Absence, non-compliance or substantial deterioration to the extent that it cannot comply with its intended use of fire detection system, fire alarms, fire-fighting equipment, fixed fire-extinguishing installation, ventilation valves, fire dampers and quick-closing devices.
- 7 Absence, substantial deterioration or failure of proper operation of the cargo deck area fire protection on tankers.
- 8 Absence, non-compliance or serious deterioration of lights, shapes or sound signals.
- 9 Absence or failure of the proper operation of the radio equipment for distress and safety communication.
- 10 Absence or failure of the proper operation of navigation equipment, taking the relevant provisions of SOLAS 1974 regulation V/16.2 into account.
- 11 Absence of corrected navigational charts, and/or all other relevant nautical publications necessary for the intended voyage, taking into account that electronic charts may be used as a substitute for the charts.
- 12 Absence of non-sparking exhaust ventilation for cargo pump-rooms.
- 13 Serious deficiency in the operational requirements listed in appendix 7.
- 14 Number, composition or certification of crew not corresponding with safe manning document.
- 15 Non-implementation or failure to carry out the enhanced survey programme in accordance with SOLAS 1974 regulation XI-1/2 and the International Code on the Enhanced Programme of Inspections during Surveys of Bulk Carriers and Oil Tankers, 2011 (2011 ESP Code), as amended.
- Absence or failure of a voyage data recorder (VDR), when its use is compulsory.

Areas under the IBC Code

- 1 Transport of a substance not mentioned in the Certificate of Fitness or missing cargo information.
- 2 Missing or damaged high-pressure safety devices.
- 3 Electrical installations not intrinsically safe or not corresponding to the Code requirements.
- 4 Sources of ignition in hazardous locations.
- 5 Contravention of special requirements.
- 6 Exceeding of maximum allowable cargo quantity per tank.
- 7 Insufficient heat protection for sensitive products.
- 8 Pressure alarms for cargo tanks not operable.
- 9 Transport of substances to be inhibited without valid inhibitor certificate.

Areas under the IGC Code

- 1 Transport of a substance not mentioned in the Certificate of Fitness or missing cargo information.
- 2 Missing closing devices for accommodations or service spaces.
- 3 Bulkhead not gastight.
- 4 Defective air locks.
- 5 Missing or defective quick-closing valves.
- 6 Missing or defective safety valves.
- 7 Electrical installations not intrinsically safe or not corresponding to the Code requirements.
- 8 Ventilators in cargo area not operable.
- 9 Pressure alarms for cargo tanks not operable.
- 10 Gas detection plant and/or toxic gas detection plant defective.
- 11 Transport of substances to be inhibited without valid inhibitor certificate.

Areas under LL 1966 and LL PROT 1988

- 1 Significant areas of damage or corrosion or pitting of plating and associated stiffening in decks and hull affecting seaworthiness or strength to take local loads, unless properly authorized temporary repairs for a voyage to a port for permanent repairs have been carried out.
- 2 A recognized case of insufficient stability.
- 3 The absence of sufficient and reliable information, in an approved form, which by rapid and simple means enables the master to arrange for the loading and ballasting of the ship in such a way that a safe margin of stability is maintained at all stages and at varying conditions of the voyage, and that the creation of any unacceptable stresses in the ship's structure is avoided.
- 4 Absence, substantial deterioration or defective closing devices, hatch closing arrangements and watertight/weathertight doors.
- 5 Overloading.
- 6 Absence of, or impossibility to read, draught marks and/or Load Line Marks.
- 7 The means of freeing water from the deck not in satisfactory or operational condition.

Areas under MARPOL Annex I

- 1 Absence, serious deterioration or failure of proper operation of the oily-water filtering equipment, the oil discharge monitoring and control system or the 15 ppm alarm arrangements.
- 2 Remaining capacity of slop and/or sludge tank insufficient for the intended voyage.
- 3 Oil Record Book not available.
- 4 Unauthorized discharge bypass fitted.
- 5 Failure to meet the requirements of regulation 20.4 or alternative requirements specified in regulation 20.7.
- 6 Oily bilge water and/or oil residue accumulated in machinery spaces.

Areas under MARPOL Annex II

- 1 Absence of Procedures and Arrangements Manual (P and A Manual).
- 2 Cargo is not categorized.
- 3 No Cargo Record Book available.
- 4 Unauthorized discharge bypass fitted.

Areas under MARPOL Annex III and dangerous goods carriage requirements

- 1 Absence of a valid Document of Compliance for carriage of dangerous goods (if required).
- 2 Absence of a Dangerous Cargo Manifest or detailed stowage plan before departure of the ship.
- 3 Stowage and segregation provisions of the IMDG Code chapters 7.1, 7.2, 7.4, 7.5 and 7.6 are not met.
- 4 Ship is carrying dangerous goods not in compliance with the Document of Compliance for carriage of dangerous goods of the ship.
- 5 Ship is carrying damaged or leaking dangerous goods packages.
- 6 Ship's personnel assigned to specific duties related to the cargo are not familiar with those duties, any dangers posed by the cargo and with the measures to be taken in such a context.

Areas under MARPOL Annex IV

- 1 Absence of valid International Sewage Pollution Prevention Certificate.
- 2 Sewage treatment plant not approved and certified by the Administration.
- 3 Failure of sewage treatment plant.
- 4 Ship's personnel not familiar with disposal/discharge requirements of sewage.

Areas under MARPOL Annex V

- 1 Absence of garbage management plan.
- 2 No garbage record book available.
- 3 Ship's personnel not familiar with disposal/discharge requirements of garbage management plan.

Areas under MARPOL Annex VI

- 1 Absence of valid International Air Pollution Prevention Certificate (IAPP Certificate) and where relevant Engine International Air Pollution Prevention Certificates (EIAPP Certificates) and Technical Files.
- 2 A marine diesel engine, with a power output of more than 130 kW, which is installed on board a ship constructed on or after 1 January 2000, or a marine diesel engine having undergone a major conversion on or after 1 January 2000, which does not comply with the NO_x Technical Code 2008, as amended.
- 3 The sulphur content of any fuel oil used on board ships exceeds the limit of 0.5% m/m on and after 1 January 2020.

- 4 The sulphur content of any fuel used on board exceeds 0.1% m/m while operating within a SO_x emission control area as per the provisions of regulation 14.
- 5 Emission reduction by equivalent arrangements are not met.
- 6 An incinerator installed on board the ship on or after 1 January 2000 does not comply with requirements contained in appendix IV to the Annex, or the standard specifications for shipboard incinerators developed by the Organization (resolution MEPC.244(66)).
- 7 Ship's personnel are not familiar with essential procedures regarding the operation of air pollution prevention equipment.
- 8 Absence of valid IEEC (International Energy Efficiency Certificate).
- 9 Absence of a Statement of Compliance related to fuel oil consumption reporting on board.

Areas under STCW 1978

- 1 Failure of seafarers to hold a certificate, to have an appropriate certificate, to have a valid dispensation or to provide documentary proof that an application for an endorsement has been submitted to the Administration.
- 2 Failure to comply with the applicable safe manning requirements of the Administration.
- 3 Failure of navigational or engineering watch arrangements to conform to the requirements specified for the ship by the Administration.
- 4 Absence in a watch of a person qualified to operate equipment essential to safe navigation, safety radiocommunications or the prevention of marine pollution.
- 5 Inability to provide for the first watch at the commencement of a voyage and for subsequent relieving watches persons who are sufficiently rested and otherwise fit for duty.

Areas under AFS 2001

- 1 Absence of a valid International Anti-Fouling System Certificate or a Declaration on Anti-Fouling System.
- 2 Sampling proves it is non-compliant within the port's jurisdiction.

Areas which may not warrant a detention, but where, for example, cargo operations have to be suspended

Failure of the proper operation (or maintenance) of inert gas systems, cargo related gear or machinery should be considered sufficient grounds to stop cargo operation.

Appendix 3

GUIDELINES FOR INVESTIGATIONS AND INSPECTIONS CARRIED OUT UNDER MARPOL ANNEX I

PART 1

INSPECTION OF IOPP CERTIFICATE, SHIP AND EQUIPMENT

1 Ships required to carry an IOPP Certificate

1.1 On boarding and introduction to the master or responsible ship's officer, the port State control officer (PSCO) should examine the International Oil Pollution Prevention Certificate (IOPP Certificate), including the attached Supplement - Record of Construction and Equipment for (ships other than) oil tankers, and the Oil Record Book (ORB). The ORB may be presented in an electronic format. A declaration from the Administration should be viewed in order to accept this electronic record book. If a declaration cannot be provided, a hard copy record book will need to be presented for examination.

1.2 The certificate carries the information on the type of ship and the dates of surveys and inspections. As a preliminary check it should be confirmed that the dates of surveys and inspections are still valid. Furthermore, it should be established if the ship carries an oil cargo and whether the carriage of such oil cargo is in conformity with the certificate (see also paragraph 1.11 of the Record of Construction and Equipment for Oil Tankers).

1.3 Through examining the Record of Construction and Equipment, the PSCO may establish how the ship is equipped for the prevention of marine pollution.

1.4 If the certificate is valid and the general impression and visual observations on board confirm a good standard of maintenance, the PSCO should generally confine the inspection to reported deficiencies, if any.

1.5 If, however, the PSCO from general impressions or observations on board has clear grounds for believing that the condition of the ship or its equipment does not correspond substantially with the particulars of the certificate, a more detailed inspection should be initiated.

1.6 The inspection of the engine-room should begin with forming a general impression of the state of the engine-room, the presence of traces of oil in the engine-room bilges and the ship's routine for disposing of oil-contaminated water from the engine-room spaces.

1.7 Next, a closer examination of the ship's equipment as listed in the IOPP Certificate may take place. This examination should also confirm that no unapproved modifications have been made to the ship and its equipment.

1.8 Should any doubt arise as to the maintenance or the condition of the ship or its equipment, then further examination and testing may be conducted as considered necessary. In this respect reference is made to annex 3 to the *Survey Guidelines under the Harmonized System of Survey and Certification (HSSC), 2021* (resolution A.[...](32)), as may be amended.

1.9 The PSCO should bear in mind that a ship may be equipped over and above the requirements of MARPOL Annex I. If such equipment is malfunctioning, the flag State should be informed. This alone however should not cause a ship to be detained unless the discrepancy presents an unreasonable threat of harm to the marine environment.

1.10 In cases of oil tankers, the inspection should include the cargo tank and pump-room area of the ship and should begin with forming a general impression of the layout of the tanks, the cargoes carried, and the routine of cargo slops disposal.

2 Ships of non-Parties to MARPOL Annex I and other ships not required to carry an IOPP Certificate

2.1 As this category of ships is not provided with an IOPP Certificate, the PSCO should be satisfied with regard to the construction and equipment standards relevant to the ship on the basis of the requirements set out in MARPOL Annex I.

2.2 In all other respects the PSCO should be guided by the procedures for ships referred to in section 1 above.

2.3 If the ship has some form of certification other than the IOPP Certificate, the PSCO may take the form and content of this documentation into account in the evaluation of that ship.

3 Control

In exercising the control functions, the PSCO should use professional judgement to determine whether to detain the ship until any noted deficiencies are corrected or to allow it to sail with certain deficiencies which do not pose an unreasonable threat of harm to the marine environment. In doing this, the PSCO should be guided by the principle that the requirements contained in MARPOL Annex I, in respect of construction and equipment and the operation of ships, are essential for the protection of the marine environment and that departure from these requirements could constitute an unreasonable threat of harm to the marine environment.

PART 2

CONTRAVENTION OF DISCHARGE PROVISIONS

1 Experience has shown that information furnished to the flag State as envisaged in appendix 5 of these Procedures is often inadequate to enable the flag State to cause proceedings to be brought in respect of the alleged violation of the discharge requirements. This appendix is intended to identify information which is often needed by a flag State for the prosecution of such possible violations.

2 It is recommended that, in preparing a port State report on deficiencies, where contravention of the discharge requirements is involved, the authorities of the coastal or port State be guided by the itemized list of possible evidence as shown in part 3 of this appendix. It should be borne in mind in this connection that:

- .1 the report aims to provide the optimal collation of obtainable data; however, even if all the information cannot be provided, as much information as possible should be submitted; and
- .2 it is important for all the information included in the report to be supported by facts which, when considered as a whole, would lead the port or coastal State to believe a contravention had occurred.

3 In addition to the port State report on deficiencies, a report should be completed by a port or coastal State on the basis of the itemized list of possible evidence. It is important that these reports are supplemented by documents such as:

- .1 a statement by the observer of the pollution; in addition to the information required under section 1 of part 3 of this appendix, the statement should include considerations which lead the observer to conclude that none of any other possible pollution sources is in fact the source;
- .2 statements concerning the sampling procedures both of the slick and on board; these should include location where and time when samples were taken, identity of person(s) taking the samples and receipts identifying the persons having custody and receiving transfer of the samples;
- .3 reports of analyses of samples taken of the slick and on board; the reports should include the results of the analyses, a description of the method employed, reference to or copies of scientific documentation attesting to the accuracy and validity of the method employed, and names of persons performing the analyses and their experience;
- .4 a statement by the PSCO on board together with the PSCO's rank and organization;
- .5 statements by persons being questioned;
- .6 statements by witnesses; all observations, photographs and documentation should be supported by a signed verification of their authenticity; all certifications, authentications or verifications shall be executed in accordance with the laws of the State which prepares them; all statements should be signed and dated by the person making the statement and, if possible, by a witness to the signing; the names of the persons signing statements should be printed in legible script above or below the signature;
- .7 photographs of the oil slick; and
- .8 copies or printouts of relevant recordings, etc., pages of ORBs, logbooks, discharge.

4 The report referred to in paragraphs 2 and 3 should be sent to the flag State. If the coastal State observing the pollution and the port State carrying out the investigation on board are not the same, the State carrying out the latter investigation should also send a copy of its findings to the State observing the pollution and requesting the investigation.

PART 3

ITEMIZED LIST OF POSSIBLE EVIDENCE ON ALLEGED CONTRAVENTION OF THE MARPOL ANNEX I DISCHARGE PROVISIONS

1 Action on sighting oil pollution

1.1 Particulars of ship or ships suspected of contravention

- .1 Name of ship
- .2 Reasons for suspecting the ship
- .3 Date and time (UTC) of observation or identification
- .4 Position of ship
- .5 Flag and port of registry
- .6 Type (e.g. tanker, cargo ship, passenger ship, fishing vessel), size (estimated tonnage) and other descriptive data (e.g. superstructure colour and funnel mark)
- .7 Draught condition (loaded or in ballast)
- .8 Approximate course and speed
- .9 Position of slick in relation to ship (e.g. astern, port, starboard)
- .10 Part of the ship from which side discharge was seen emanating
- .11 Whether discharge ceased when ship was observed or contacted by radio

1.2 Particulars of slick

- .1 Date and time (UTC) of observation if different from paragraph 1.1.3
- .2 Position of oil slick in longitude and latitude if different from paragraph 1.1.4
- .3 Approximate distance in nautical miles from the nearest land
- .4 Approximate overall dimension of oil slick (length, width and percentage thereof covered by oil)
- .5 Physical description of oil slick (direction and form, e.g. continuous, in patches or in windrows)
- .6 Appearance of oil slick (indicate categories)
 - Category A: Barely visible under most favourable light condition
 - Category B: Visible as silvery sheen on water surface

- Category C: First trace of colour may be observed
- Category D: Bright band of colour
- Category E: Colours begin to turn dull
- Category F: Colours are much darker
- .7 Sky conditions (bright sunshine, overcast, etc.), lightfall and visibility (kilometres) at the time of observation
- .8 Sea state
- .9 Direction and speed of surface wind
- .10 Direction and speed of current

1.3 Identification of the observer(s)

- .1 Name of observer
- .2 Organization with which observer is affiliated (if any)
- .3 Observer's status within the organization
- .4 Observation made from aircraft/ship/shore/otherwise
- .5 Name or identity of ship or aircraft from which observation was made
- .6 Specific location of ship, aircraft, place on shore or otherwise from which observation was made
- .7 Activity engaged in by observer when observation was made, e.g. patrol, voyage, flight (en route from ... to ...)

1.4 Method of observation and documentation

- .1 Visual
- .2 Conventional photographs
- .3 Remote sensing records and/or remote sensing photographs
- .4 Samples taken from slick
- .5 Any other form of observation (specify)
- **Note:** A photograph of the discharge should preferably be in colour. Photographs can provide the following information: that a material on the sea surface is oil; that the quantity of oil discharged does constitute a violation of the Convention; that the oil is being, or has been, discharged from a particular ship; and the identity of the ship.

Experience has shown that the aforementioned can be obtained with the following three photographs:

- details of the slick taken almost vertically down from an altitude of less than 300 m with the sun behind the photographer;
- an overall view of the ship and slick showing oil emanating from a particular ship; and
- details of the ship for the purposes of identification.

1.5 Other information if radio contact can be established

- .1 Master informed of pollution
- .2 Explanation of master
- .3 Ship's last port of call
- .4 Ship's next port of call
- .5 Name of ship's master and owner
- .6 Ship's call sign

2 Investigation on board

2.1 Inspection of IOPP Certificate

- .1 Name of ship
- .2 Distinctive number or letters
- .3 Port of registry
- .4 Type of ship
- .5 Date and place of issue
- .6 Date and place of endorsement
- *Note*: If the ship is not issued an IOPP Certificate, as much as possible of the requested information should be given.

2.2 Inspection of Supplement of the IOPP Certificate

- .1 Applicable paragraphs of sections 2, 3, 4, 5 and 6 of the Supplement (non-oil tankers)
- .2 Applicable paragraphs of sections 2, 3, 4, 5, 6, 7, 8, 9 and 10 of the Supplement (oil tankers)

Note: If the ship does not have an IOPP Certificate, a description should be given of the equipment and arrangements on board, designed to prevent marine pollution.

2.3 Inspection of Oil Record Book (ORB)

- .1 Copy or print out sufficient pages of the ORB part I to cover a period of 30 days prior to the reported incident
- .2 Copy or print out sufficient pages of the ORB part II (if on board) to cover a full loading/unloading/ballasting and tank cleaning cycle of the ship. Also copy the tank diagram

2.4 Inspection of logbook

- .1 Last port, date of departure, draught forward and aft
- .2 Current port, date of arrival, draught forward and aft
- .3 Ship's position at or near the time the incident was reported
- .4 Spot check if positions mentioned in the logbook agree with positions noted in the ORB

2.5 Inspection of other documentation on board

Other documentation relevant for evidence (if necessary, make copies) such as:

- .1 recent ullage sheets
- .2 records of monitoring and control equipment

2.6 Inspection of ship

- .1 Ship's equipment in accordance with the Supplement of the IOPP Certificate
- .2 Samples taken. State location on board
- .3 Traces of oil in vicinity of overboard discharge outlets
- .4 Condition of engine-room and contents of bilges
- .5 Condition of oily water separator, filtering equipment and alarm, stopping or monitoring arrangements
- .6 Contents of sludge and/or holding tanks
- .7 Sources of considerable leakage on oil tankers

The following additional evidence may be pertinent:

- .8 Oil on surface of segregated or dedicated clean ballast
- .9 Condition of pump-room bilges

- .10 Condition of crude oil washing (COW) system
- .11 Condition of inert gas (IG) system
- .12 Condition of monitoring and control system
- .13 Slop tank contents (estimate quantity of water and of oil)

2.7 Statements of persons concerned

If the ORB – part I has not been properly completed, information on the following questions may be pertinent:

- .1 Was there a discharge (accidental or intentional) at the time indicated on the incident report?
- .2 Is the bilge discharge controlled automatically?
- .3 If so, at what time was this system last put into operation and at what time was this system last put on manual mode?
- .4 If not, what were the date and time of the last bilge discharge?
- .5 What was the date of the last disposal of residue and how was disposal effected?
- .6 Is it usual to effect discharge of bilge water directly to the sea, or to store bilge water first in a collecting tank? Identify the collecting tank.
- .7 Have oil fuel tanks recently been used as ballast tanks?

If the ORB – part II has not been properly completed, information on the following questions may be pertinent:

- .8 What was the cargo/ballast distribution in the ship on departure from the last port?
- .9 What was the cargo/ballast distribution in the ship on arrival in the current port?
- .10 When and where was the last loading effected?
- .11 When and where was the last unloading effected?
- .12 When and where was the last discharge of dirty ballast?
- .13 When and where was the last cleaning of cargo tanks?
- .14 When and where was the last COW operation and which tanks were washed?
- .15 When and where was the last decanting of slop tanks?

- .16 What is the ullage in the slop tanks and the corresponding height of interface?
- .17 Which tanks contained the dirty ballast during the ballast voyage (if ship arrived in ballast)?
- .18 Which tanks contained the clean ballast during the ballast voyage (if ship arrived in ballast)?

In addition, the following information may be pertinent:

- .19 Details of the present voyage of the ship (previous ports, next ports, trade)
- .20 Contents of oil fuel and ballast tanks
- .21 Previous and next bunkering, type of oil fuel
- .22 Availability or non-availability of reception facilities for oily wastes during the present voyage
- .23 Internal transfer of oil fuel during the present voyage

In the case of oil tankers, the following additional information may be pertinent:

- .24 The trade the ship is engaged in, such as short/long distance, crude or product or alternating crude/product, lightering service, oil/dry bulk
- .25 Which tanks are clean and dirty
- .26 Repairs carried out or envisaged in cargo tanks

Miscellaneous information:

- .27 Comments in respect of condition of ship's equipment
- .28 Comments in respect of pollution report
- .29 Other comments

3 Investigation ashore

3.1 Analyses of oil samples

Indicate method and results of the samples' analyses.

3.2 Further information

Additional information on the ship, obtained from oil terminal staff, tank cleaning contractors or shore reception facilities may be pertinent.

Note: Any information under this heading is, if practicable, to be corroborated by documentation such as signed statements, invoices, receipts.

4 Information not covered by the foregoing

5 Conclusion

- .1 Summing up of the investigator's technical conclusions.
- .2 Indication of applicable provisions of MARPOL Annex I which the ship is suspected of having contravened.
- .3 Did the results of the investigation warrant the filing of a deficiency report?

PART 4

GUIDELINES FOR IN-PORT INSPECTION OF CRUDE OIL WASHING PROCEDURES

1 Preamble

1.1 *Guidelines for the in-port inspection of crude oil washing (COW) procedures*, as called for by resolution 7 of the International Conference on Tanker Safety and Pollution Prevention, 1978, are required to provide a uniform and effective control of crude oil washing to ensure compliance of ships at all times with the provisions of MARPOL.

1.2 The design of the crude oil washing installation is subject to the approval of the flag Administration. However, although the operational aspect of crude oil washing is also subject to the approval of the same Administration, it might be necessary for a port State authority to see to it that continuing compliance with agreed procedures and parameters is ensured.

1.3 The COW Operations and Equipment Manual has been so specified that it contains all the necessary information relating to the operation of crude oil washing on a particular tanker. The objectives of the inspection would then be to ensure that the provisions of the Manual dealing with safety procedures and with pollution prevention are being strictly adhered to.

1.4 The method of the inspection is at the discretion of the port State authority and may cover the entire operation or only those parts of the operation which occur when the PSCO is on board.

1.5 Inspection will be governed by articles 5 and 6 of MARPOL.

2 Inspections

2.1 A port State should make the appropriate arrangements so as to ensure compliance with requirements governing the crude oil washing of oil tankers. This is not, however, to be construed as relieving terminal operators and shipowners of their obligations to ensure that the operation is undertaken in accordance with the regulations.

2.2 The inspection may cover the entire operation of crude oil washing or only certain aspects of it. It is thus in the interest of all concerned that the ship's records with regard to the COW operations are maintained at all times so that a PSCO may verify those operations undertaken prior to the inspection.

3 Ship's personnel

3.1 The person in charge and the other nominated persons who have responsibility in respect of the crude oil washing operation should be identified. They must, if required, be able

to show that their qualifications meet the requirements, as appropriate, of paragraphs 5.2 and 5.3 of the *Revised specifications for the design, operation and control of crude oil washing systems* (resolution A.446(XI)), as amended.

3.2 The verification may be accomplished by reference to the individual's discharge papers, testimonials issued by the ship's operator or by certificates issued by a training centre approved by an Administration. The numbers of such personnel should be at least as stated in the Manual.

4 Documentation

The following documents should be available for inspection:

- .1 the IOPP Certificate and the Record of Construction and Equipment, to determine:
 - .1 whether the ship is fitted with a crude oil washing system as required in regulation 33 of MARPOL Annex I;
 - .2 whether the crude oil washing system is according to and complying with the requirements of regulations 33 and 35 of MARPOL Annex I;
 - .3 the validity and date of the Operations and Equipment Manual; and
 - .4 the validity of the Certificate;
- .2 the approved Manual;
- .3 the ORB; and
- .4 the Cargo Ship Safety Equipment Certificate to confirm that the inert gas system conforms to regulations contained in chapter II-2 of SOLAS 1974.

5 Inert gas system

5.1 Inert gas system regulations require that instrumentation shall be fitted for continuously indicating and permanently recording at all times when inert gas is being supplied, the pressure and the oxygen content of the gas in the inert gas supply main. Reference to the permanent recorder would indicate if the system had been operating before and during the cargo discharge in a satisfactory manner.

5.2 If conditions specified in the Manual are not being met then the washing must be stopped until satisfactory conditions are restored.

5.3 As a further precautionary measure, the oxygen level in each tank to be washed is to be determined at the tank. The meters used should be calibrated and inspected to ensure that they are in good working order. Readings from tanks already washed in port prior to inspection should be available for checking. Spot checks on readings may be instituted.

6 Electrostatic generation

It should be confirmed either from the cargo log or by questioning the person in charge that the presence of water in the crude oil is being minimized as required by paragraph 6.7 of the

revised Specifications for the design, operation and control of crude oil washing systems (resolution A.446(XI)), as amended.

7 Communication

It should be established that effective means of communication exist between the person in charge and the other persons concerned with the COW operation.

8 Leakage on deck

PSCOs should ensure that the COW piping system has been operationally tested for leakage before cargo discharge and that the test has been noted in the ship's ORB.

9 Exclusion of oil from engine-room

It should be ascertained that the method of excluding cargo oil from the machinery space is being maintained by inspecting the isolating arrangements of the tank washing heater (if fitted) or of any part of the tank washing system which enters the machinery space.

10 Suitability of the crude oil

In judging the suitability of the oil for crude oil washing, the guidance and criteria contained in section 9 of the COW Operations and Equipment Manual should be taken into account.

11 Checklist

It should be determined from the ship's records that the pre-crude oil wash operational checklist was carried out and all instruments functioned correctly. Spot checks on certain items may be instituted.

12 Wash programmes

12.1 Where the tanker is engaged in a multiple port discharge, the ORB would indicate if tanks were crude oil washed at previous discharge ports or at sea. It should be determined that all tanks which will or may be used to contain ballast on the forthcoming voyage will be crude oil washed before the ship departs from the port. There is no obligation to wash any tank other than ballast tanks at a discharge port except that each of these other tanks must be washed at least in accordance with paragraph 6.1 of the revised *Specifications for the design, operation and control of crude oil washing systems* (resolution A.446(XI)), as amended. The ORB should be inspected to check that this is being complied with.

12.2 All crude oil washing must be completed before a ship leaves its final port of discharge.

12.3 If tanks are not being washed in one of the preferred orders given in the Manual, the PSCO should determine that the reason for this and the proposed order of tank washing are acceptable.

12.4 For each tank being washed it should be ensured that the operation is in accordance with the Manual in that:

.1 the deck mounted machines and the submerged machines are operating either by reference to indicators, the sound patterns or other approved methods;

- .2 the deck mounted machines, where applicable, are programmed as stated;
- .3 the duration of the wash is as required; and
- .4 the number of tank washing machines being used simultaneously does not exceed that specified.

13 Stripping of tanks

13.1 The minimum trim conditions and the parameters of the stripping operations are to be stated in the Manual.

13.2 All tanks which have been crude oil washed are to be stripped. The adequacy of the stripping is to be checked by hand dipping at least in the aftermost hand dipping location in each tank or by such other means provided and described in the Manual. It should be ascertained that the adequacy of stripping has been checked or will be checked before the ship leaves its final port of discharge.

14 Ballasting

14.1 Tanks that were crude oil washed at sea will be recorded in the ORB. These tanks must be left empty between discharge ports for inspection at the next discharge port. Where these tanks are the designated departure ballast tanks they may be required to be ballasted at a very early stage of the discharge. This is for operational reasons and also because they must be ballasted during cargo discharge, if hydrocarbon emission is to be contained on the ship. If these tanks are to be inspected when empty, then this must be done shortly after the tanker berths. If a PSCO arrives after the tanks have begun accepting ballast, then the sounding of the tank bottom would not be available. However, an examination of the surface of the ballast water is then possible. The thickness of the oil film should not be greater than that specified in paragraph 4.2.10(b) of the revised *Specifications for the design, operation and control of crude oil washing systems* (resolution A.446(XI)), as amended.

14.2 The tanks that are designated ballast tanks will be listed in the Manual. It is, however, left to the discretion of the master or responsible officer to decide which tanks may be used for ballast on the forthcoming voyage. It should be determined from the ORB that all such tanks have been washed before the tanker leaves its last discharge port. It should be noted that where a tanker back-loads a cargo of crude oil at an intermediate port into tanks designated for ballast, then it should not be required to wash those tanks at that particular port but at a subsequent port.

14.3 It should be determined from the ORB that additional ballast water has not been put into tanks which have not been crude oil washed during previous voyages.

14.4 It should be verified that the departure ballast tanks are stripped as completely as possible. Where departure ballast is filled through cargo lines and pumps these must be stripped either into another cargo tank or ashore by the special small diameter line provided for this purpose.

14.5 The methods to avoid vapour emission where locally required will be provided in the Manual and they must be adhered to. The PSCO should ensure that this is being complied with.

14.6 The typical procedures for ballasting listed in the Manual must be observed. The PSCO should ensure this is being complied with.

14.7 When departure ballast is to be shifted, the discharge into the sea must be in compliance with regulations 15 and 34 of MARPOL Annex I. The ORB should be inspected to ensure that the ship is complying with this.

Appendix 4

GUIDELINES FOR INVESTIGATIONS AND INSPECTIONS CARRIED OUT UNDER MARPOL ANNEX II

PART 1

INSPECTION OF CERTIFICATE (COF OR NLS CERTIFICATE), SHIP AND EQUIPMENT

1 Ships required to hold a Certificate

1.1 On boarding and after introducing themself to the master or responsible ship's officer, the port State control officer (PSCO) should examine the Certificate of Fitness (COF) or NLS Certificate and Cargo Record Book (CRB). The CRB may be presented in an electronic format. A declaration from the Administration should be viewed in order to accept this electronic record book. If a declaration cannot be provided, a hard copy record book will need to be presented for examination.

1.2 The Certificate includes information on the type of ship, the dates of surveys and a list of the products which the ship is certified to carry.

1.3 As a preliminary check, the Certificate's validity should be confirmed by verifying that the Certificate is properly completed and signed and that required surveys have been performed. In reviewing the Certificate, particular attention should be given to verifying that only those noxious liquid substances which are listed on the Certificate are carried and that these substances are in tanks approved for their carriage.

1.4 The CRB should be inspected to ensure that the records are up to date. The PSCO should check whether the ship left the previous port(s) with residues of noxious liquid substances on board which could not be discharged into the sea. The book could also have relevant entries from the appropriate authorities in the previous ports. If the examination reveals that the ship was permitted to sail from its last unloading port under certain conditions, the PSCO should ascertain that such conditions have been or will be adhered to. If the PSCO discovers an operational violation in this respect, the flag State should be informed by means of a deficiency report.

1.5 If the Certificate is valid and the PSCO's general impressions and visual observations on board confirm a good standard of maintenance, the PSCO should, provided that the CRB entries do not show any operational violations, confine the inspection to reported deficiencies, if any.

1.6 If, however, the PSCO's general impressions or observations on board show clear grounds for believing that the condition of the ship, its equipment, or its cargo and slops handling operations do not correspond substantially with the particulars of the Certificate, the PSCO should proceed to a more detailed inspection:

- .1 Initially this requires an examination of the ship's approved P and A Manual.
- .2 The more detailed inspection should include the cargo and pump-room areas of the ship and should begin with forming a general impression of the layout of the tanks, the cargoes carried, pumping and stripping conditions and cargo.
- .3 Next, a closer examination of the ship's equipment as shown in the P and A Manual may take place. This examination should also confirm that no unapproved modifications have been made to the ship and its equipment.
- .4 Should any doubt arise as to the maintenance or the condition of the ship or its equipment, further examination and testing may be conducted as may be necessary. In this respect reference is made to the *Survey Guidelines under the Harmonized System of Survey and Certification, 2021* (resolution A.[...](32)), as may be amended.

1.7 The PSCO should bear in mind that a ship may be equipped over and above the requirements of MARPOL Annex II. If such equipment is malfunctioning the flag State should be informed. This alone, however, should not cause a ship to be detained unless the malfunction presents an unreasonable threat of harm to the marine environment.

2 Ships of non-Parties to the Convention

2.1 As this category of ship is not provided with a COF or NLS Certificate as required by MARPOL Annex II, the PSCO should be satisfied with regard to the construction and equipment standards relevant to the ship on the basis of the requirements set out in MARPOL Annex II and the Standards for Procedures and Arrangements.

2.2 In all other respects, the PSCO should be guided by the procedures for ships referred to in section 1 above (i.e. ships required to hold a Certificate).

2.3 If the ship has some form of certification other than the required Certificate, the PSCO may take the form and content of this document into account in the evaluation of that ship. Such a form of certification, however, is only of value to the PSCO if the ship has been provided with a P and A Manual.

3 Control

In exercising the control functions, the PSCO should use professional judgement to determine whether to detain the ship until any noted deficiencies are rectified or to allow it to sail with certain deficiencies which do not pose an unreasonable threat of harm to the marine environment. In doing this, the PSCO should be guided by the principle that the requirements contained in MARPOL Annex II, in respect of construction and equipment and the operation of ships, are essential for the protection of the marine environment and that departure from these requirements could constitute an unreasonable threat of harm to the marine environment.

PART 2

CONTRAVENTION OF DISCHARGE PROVISIONS

1 With illegal discharges, past experience has shown that information furnished to the flag State is often inadequate to enable the flag State to cause proceedings to be brought in respect of the alleged violation of the discharge requirements. This appendix is intended to identify information which will be needed by a flag State for the prosecution of violations of the discharge provisions under MARPOL Annex II.

2 It is recommended that in preparing a port State report on deficiencies, where contravention of the discharge requirements is involved, the authorities of a coastal or port State should be guided by the itemized list of possible evidence as shown in part 3 of this appendix. It should be borne in mind in this connection that:

- .1 the report aims to provide the optimal collation of obtainable data; however, even if all the information cannot be provided, as much information as possible should be submitted;
- .2 it is important for all the information included in the report to be supported by facts which, when considered as a whole, would lead the port or coastal State to believe a contravention has occurred; and
- .3 the discharge may have been oil, in which case part 2 to appendix 3 of this resolution applies (Guidelines for investigation and inspections carried out under MARPOL Annex I).

3 In addition to the port State report on deficiencies, a report should be completed by a port or coastal State, on the basis of the itemized list of possible evidence. It is important that these reports are supplemented by documents such as:

- .1 a statement by the observer of the pollution; in addition to the information required under section 1 of part 3 of this appendix, the statement should include considerations which have led the observer to conclude that none of any other possible pollution sources is in fact the source;
- .2 statements concerning the sampling procedures both of the slick and on board; these include location where and time when samples were taken, identity of person(s) taking the samples and receipts identifying the persons having custody and receiving transfer of the samples;
- .3 reports of analyses of samples taken of the slick and on board; the reports should include the results of the analyses, a description of the method employed, reference to or copies of scientific documentation attesting to the accuracy and validity of the method employed and names of persons performing the analyses and their experience;
- .4 a statement by the PSCO on board together with the PSCO's rank and organization;
- .5 statements by persons being questioned;
- .6 statements by witnesses;
- .7 photographs of the slick; and
- .8 copies or printouts of relevant pages of the CRB, logbooks, discharge recordings, etc.

4 All observations, photographs and documentation should be supported by a signed verification of their authenticity. All certifications, authentications or verifications shall be executed in accordance with the laws of the State which prepares them. All statements should be signed and dated by the person making the statement and, if possible, by a witness to the signing. The names of the persons signing statements should be printed in legible script above or below the signature. 5 The report referred to in paragraphs 2 and 3 should be sent to the flag State. If the coastal State observing the pollution and the port State carrying out the investigation on board are not the same, the State carrying out the latter investigation should also send a copy of its findings to the State observing the pollution and requesting the investigation.

PART 3

ITEMIZED LIST OF POSSIBLE EVIDENCE ON ALLEGED CONTRAVENTION OF THE MARPOL ANNEX II DISCHARGE PROVISIONS

1 Action on sighting pollution

1.1 Particulars of ship or ships suspected of contravention

- .1 Name of ship and IMO Number
- .2 Reasons for suspecting the ship
- .3 Date and time (UTC) of observation or identification
- .4 Position of ship
- .5 Flag and port of registry
- .6 Type, size (estimated tonnage) and other descriptive data (e.g. superstructure colour and funnel mark)
- .7 Draught condition (loaded or in ballast)
- .8 Approximate course and speed
- .9 Position of slick in relation to ship (e.g. astern, port, starboard)
- .10 Part of the ship from which discharge was seen emanating
- .11 Whether discharge ceased when ship was observed or contacted by radio

1.2 Particulars of slick

- .1 Date and time (UTC) of observation if different from item 1.1.3
- .2 Position of slick in longitude and latitude if different from item 1.1.4
- .3 Approximate distance in nautical miles from the nearest land
- .4 Depth of water according to sea chart
- .5 Approximate overall dimension of slick (length, width and percentage thereof covered)
- .6 Physical description of slick (direction and form, e.g. continuous, in patches or in windrows)

- .7 Colour of slick
- .8 Sky conditions (bright sunshine, overcast, etc.), lightfall and visibility (kilometres) at the time of observation
- .9 Sea state
- .10 Direction and speed of surface wind
- .11 Direction and speed of current

1.3 Identification of the observer(s)

- .1 Name of observer
- .2 Organization with which observer is affiliated (if any)
- .3 Observer's status within the organization
- .4 Observation made from aircraft, ship, shore or otherwise
- .5 Name or identity of ship or aircraft from which observation was made
- .6 Specific location of ship, aircraft, place on shore or otherwise from which observation was made
- .7 Activity engaged in by observer when observation was made, e.g. patrol, voyage, flight (enroute from ... to ...)

1.4 Method of observation and documentation

- .1 Visual
- .2 Conventional photographs
- .3 Remote sensing records and/or remote sensing photographs
- .4 Samples taken from slick
- .5 Any other form of observation (specify)
- *Note*: A photograph of the discharge should preferably be in colour. The best results may be obtained with the following three photographs:
 - details of the slick taken almost vertically down from an altitude of less than 300 metres with the sun behind the photographer;
 - an overall view of the ship and slick showing a substance emanating from the particular ship; and
 - details of the ship for the purposes of identification.

1.5 Other information if radio contact can be established

- .1 Master informed of pollution
- .2 Explanation of master
- .3 Ship's last port of call
- .4 Ship's next port of call
- .5 Name of ship's master and owner
- .6 Ship's call sign

2 Investigation on board

2.1 Inspection of the Certificate (COF or NLS Certificate)

- .1 Name of ship and IMO Number
- .2 Distinctive number or letters
- .3 Port of registry
- .4 Type of ship
- .5 Date and place of issue
- .6 Date and place of endorsement
- .7 List of Annex II substances the ship is certified to carry
- .8 Limitation as to tanks in which these substances may be carried

2.2 Inspection of P and A Manual

- .1 Ship equipped with an efficient stripping system
- .2 Residue quantities established at survey

2.3 Inspection of CRB

Copy or print out sufficient pages of the CRB to cover a full loading/unloading/ballasting and tank cleaning cycle of the ship. Also copy the tank diagram.

2.4 Inspection of logbook

- .1 Last port, date of departure, draught forward and aft
- .2 Current port, date of arrival, draught forward and aft
- .3 Ship's position at or near the time the incident was reported

.4 Spot check if times entered in the CRB in respect of discharges correspond with sufficient distance from the nearest land, the required ship's speed and with sufficient water depth

2.5 Inspection of other documentation on board

Other documentation relevant for evidence (if necessary make copies) such as:

- cargo documents of cargo presently or recently carried, together with relevant information on required unloading temperature, viscosity and/or melting point;
- records of temperature of substances during unloading; and
- records of monitoring equipment if fitted.

2.6 Inspection of ship

- .1 Ship's equipment in accordance with the P and A Manual
- .2 Samples taken; state location on board
- .3 Sources of considerable leakage
- .4 Cargo residues on surface of segregated or dedicated clean ballast
- .5 Condition of pump-room bilges
- .6 Condition of monitoring system
- .7 Slop tank contents (estimate quantity of water and residues)

2.7 Statements of persons concerned (if the CRB has not been properly completed, information on the following questions may be pertinent)

- .1 Was there a discharge (accidental or intentional) at the time indicated on the incident report?
- .2 Which tanks are going to be loaded in the port?
- .3 Which tanks needed cleaning at sea? Had the tanks been prewashed?
- .4 When and where were these cleaned?
- .5 Residues of which substances were involved?
- .6 What was done with the tank washing slops?
- .7 Was the slop tank, or cargo tank used as a slop tank, discharged at sea?
- .8 When and where was the discharge effected?
- .9 What are the contents of the slop tank or cargo tank used as slop tank?

- .10 Which tanks contained the dirty ballast during the ballast voyage (if ship arrived in ballast)?
- .11 Which tanks contained the clean ballast during the ballast voyage (if ship arrived in ballast)?
- .12 Details of the present voyage of the ship (previous ports, next ports, trade)
- .13 Difficulties experienced with discharge to shore reception facilities
- .14 Difficulties experienced with efficient stripping operations
- .15 Which tanks are clean or dirty on arrival?
- .16 Repairs carried out or envisaged in cargo tanks

Miscellaneous information

- .17 Comments in respect of condition of ship's equipment
- .18 Comments in respect of pollution report
- .19 Other comments

3 Investigation ashore

3.1 Analyses of samples

Indicate method and results of the samples' analyses.

3.2 Further information

Additional information on the ship, obtained from terminal staff, tank cleaning contractors or shore reception facilities, may be pertinent.

Note: Any information under this heading is, if practicable, to be corroborated by documentation such as signed statements, invoices, receipts.

3.3 Information from previous unloading port terminal

- .1 Confirmation that the ship was unloaded, stripped or prewashed in accordance with its P and A Manual
- .2 The nature of difficulties, if any
- .3 Restrictions by authorities under which the ship was permitted to sail
- .4 Restrictions in respect of shore reception facilities

4 Information not covered by the foregoing

5 Conclusion

- .1 Summing up of the investigator's conclusions
- .2 Indication of applicable provisions of MARPOL Annex II which the ship is suspected of having contravened
- .3 Did the results of the investigation warrant the filing of a deficiency report?

PART 4

PROCEDURES FOR INSPECTION OF UNLOADING, STRIPPING AND PREWASHING OPERATIONS (MAINLY IN UNLOADING PORTS)

1 Introduction

The PSCO or the surveyor authorized by the Administration exercising control in accordance with regulation 16 of MARPOL Annex II should be thoroughly acquainted with MARPOL Annex II and the custom of the port as of relevance to cargo handling, tank washing, cleaning berths, prohibition of lighters alongside, etc.

2 Documentation

The documentation required for the inspection referred to in this appendix consists of:

- .1 COF or NLS Certificate;
- .2 cargo plan and shipping document;
- .3 P and A Manual; and
- .4 CRB.

3 Information by ship's staff

3.1 Of relevance to the PSCO or the surveyor appointed or authorized by the Administration is the following:

- .1 the intended loading and unloading programme of the ship;
- .2 whether unloading and stripping operations can be effected in accordance with the P and A Manual and if not the reason why it cannot be done;
- .3 the constraints, if any, under which the efficient stripping system operates (i.e. back pressure, ambient air temperature, malfunctioning, etc.); and
- .4 whether the ship requests an exemption from the prewashing and the discharge of residues in the unloading port.

3.2 When tank washing is required without the use of water the PSCO or the surveyor appointed or authorized by the Administration is to be informed about the tank washing procedure and disposal of residues.

3.3 When the CRB is not up to date, any information on prewash and residue disposal operations outstanding should be supplied.

4 Information from terminal staff

Terminal staff should supply information on limitations imposed upon the ship in respect of back pressure and/or reception facilities.

5 Control

5.1 On boarding and introduction to the master or responsible ship officers, the PSCO or the surveyor appointed or authorized by the Administration should examine the necessary documentation.

5.2 The documentation may be used to establish the following:

- .1 noxious liquid substances to be unloaded, their categories and stowage (cargo plan, P and A Manual);
- .2 details of efficient stripping system, if fitted (P and A Manual);
- .3 tanks which require prewashing with disposal of tank washings to reception facilities (shipping document and cargo temperature);
- .4 tanks which require prewashing with disposal of tank washings either to reception facilities or into the sea (P and A Manual, shipping document and cargo temperature);
- .5 prewash operations and/or residue disposal operations outstanding (CRB); and
- .6 tanks which may not be washed with water due to the nature of substances involved (P and A Manual).

5.3 In respect of the prewash operations referred to under paragraph 5.2, the following information is of relevance (P and A Manual):

- .1 pressure required for tank washing machines;
- .2 duration of one cycle of the tank washing machine and quantity of water used;
- .3 washing programmes for the substances involved;
- .4 required temperature of washing water; and
- .5 special procedures.

5.4 The PSCO or the surveyor authorized by the Administration, in accordance with regulation 16 of MARPOL Annex II, should ascertain that unloading, stripping and/or prewash operations are carried out in conformity with the information obtained in accordance with paragraph 2 (Documentation) of this part. If this cannot be achieved, alternative measures should be taken to ensure that the ship does not proceed to sea with more than the quantities of residue specified in regulation 12 of MARPOL Annex II, as applicable. If the residue quantities cannot be reduced by alternative measures the PSCO or the surveyor appointed or authorized by the Administration should inform the port State Administration.

5.5 Care should be taken to ensure that cargo hoses and piping systems of the terminal are not drained back to the ship.

5.6 If a ship is exempted from certain pumping efficiency requirements under regulation 4.4 of MARPOL Annex II or requests an exemption from certain stripping or prewashing procedures under regulation 13.4 of MARPOL Annex II, the conditions for such exemption set out in the said regulations should be observed. These concern:

- .1 regulations 4.2 and 4.3: the ship is constructed before 1 July 1986 and is exempted from the requirement for reducing its residue quantities to specified limits of regulation 12 (i.e. category X or Y substances 300 litres and category Z substances 900 litres); this is subject to the conditions of regulation 4.3 that whenever a cargo tank is to be washed or ballasted, a prewash is required with disposal of prewash slops to shore reception facilities; the COF or NLS Certificate should have been endorsed to the effect that the ship is solely engaged in restricted voyages;
- .2 regulation 4.4: the ship is never required to ballast its cargo tanks and tank washing is only required for repair or dry-docking; the COF or NLS Certificate should indicate the particulars of the exemption; each cargo tank should be certified for the carriage of only one named substance;
- .3 regulation 13.4.1: cargo tanks will not be washed or ballasted prior to the next loading;
- .4 regulation 13.4.2: cargo tanks will be washed and prewash slops will be discharged to reception facilities in another port; it should be confirmed in writing that an adequate reception facility is available at that port for such purpose; and
- .5 regulation 13.4.3: the cargo residues can be removed by ventilation.

5.7 The PSCO or the surveyor appointed or authorized by the Administration must endorse the CRB under section J whenever an exemption under regulation 13.4 referred to in paragraph 5.6 above has been granted, or whenever a tank having unloaded category X substances has been prewashed in accordance with the P and A Manual.

5.8 Alternatively, for category X substances, regulation 13.6.1.1 of MARPOL Annex II, residual concentration should be measured by the procedures which each port State authorizes. In this case the PSCO or the surveyor authorized by the Administration must endorse in the CRB under section K whenever the required residual concentration has been achieved.

5.9 In addition to paragraph 5.7 above, the PSCO or the surveyor authorized by the Administration shall endorse the CRB whenever the unloading, stripping or prewash of category Y and Z substances, in accordance with the P and A Manual, has actually been witnessed.

5.10 With reference to endorsements 5.7, 8, 9 if the ship has implemented an electronic record book, the shipowner may request these endorsements using a standalone form or request of a copy of the surveyor's report to accompany the electronic record book entry.

(State)

EXAMPLE FORM OF A CARGO RECORD BOOK ENDORSEMENT

Cargo Record Book Endorsement

(Official Seal)

Issued under the authority of the Government of:

(full designation of the country)

by _____

(Organization, company, government agency authorized)

Name of Ship:	Distinctive Letters:	Number	or	Port of Registry:
Gross Tonnage:	IMO Number:	*		

Port:		
Tank(s):	Substance(s):	Category(ies):
	Yes	No
Tank(s), pump(s) and piping system(s) emptied?		
Prewash carried out in accordance with the PA Manual?		
Tank washings resulting from prewash been discharged ashore and is the tank empty?		
Exemption granted from mandatory prewash?		

Reasons for exemption: ______

THIS IS TO ENDORSE:

That, in accordance with regulation 16 of MARPOL Annex II, the entries into the Cargo Record Book according to regulation 13.6 of MARPOL Annex II have been made and operations have been carried out in accordance with the Procedures and Arrangements manual.

^{*} Refer to *IMO* ship identification number scheme (resolution A.1117(30)).

(Date)

(Name and Signature of authorized surveyor)

(Seal or stamp of the issuing authority, as appropriate)

Appendix 5

GUIDELINES FOR DISCHARGE REQUIREMENTS UNDER MARPOL ANNEXES I AND II

1 Introduction

1.1 Regulations 15 and 34 of MARPOL Annex I prohibit the discharge into the sea of oil and regulation 13 of Annex II prohibits the discharge into the sea of noxious liquid substances except under precisely defined conditions. A record of these operations shall be completed, where appropriate, in the form of an Oil or Cargo Record Book as applicable and shall be kept in such a place as to be readily available for inspection at all reasonable times.

1.2 The regulations referred to above provide that whenever visible traces of oil are observed on or below the surface of the water in the immediate vicinity of a ship or of its wake, a Party should, to the extent that it is reasonably able to do so, promptly investigate the facts bearing on the issue of whether or not there has been a violation of the discharge provisions.

1.3 The conditions under which noxious liquid substances are permitted to be discharged into the seas include quantity, quality and position limitations, which depend on category of substance and sea area.

1.4 An investigation into an alleged contravention should therefore aim to establish whether a noxious liquid substance has been discharged and whether the operations leading to that discharge were in accordance with the ship's Procedures and Arrangements Manual (P and A Manual).

1.5 Recognizing the likelihood that many of the violations of the discharge provisions will take place outside the immediate control and knowledge of the flag State, article 6 of MARPOL provides that Parties shall cooperate in the detection of violations and the enforcement of the provisions using all appropriate and practicable measures of detection and environmental monitoring, and adequate procedures for reporting and gathering evidence. MARPOL also contains a number of more specific provisions designed to facilitate that cooperation.

1.6 Several sources of information about possible violations of the discharge provisions can be indicated. These include:

- .1 reports by masters: article 8 and Protocol I of MARPOL require, inter alia, a ship's master to report certain incidents involving the discharge or the probability of a discharge of oil or oily mixtures, or noxious liquid substances or mixtures containing such substances;
- .2 reports by official bodies: article 8 of MARPOL requires furthermore that a Party issue instructions to its maritime inspection vessels and aircraft and to other appropriate services to report to its authorities incidents involving the discharge or the probability of a discharge of oil or oily mixtures, or noxious liquid substances or mixtures containing such substances;
- .3 reports by other Parties: article 6 of MARPOL provides that a Party may request another Party to inspect a ship; the Party making the request shall supply sufficient evidence that the ship has discharged oil or oily mixtures, noxious liquid substances or mixtures containing such substances, or that the ship has departed from the unloading port with residues of noxious liquid substances in excess of those permitted to be discharged into the sea; and

.4 reports by others: it is not possible to list exhaustively all sources of information concerning alleged contravention of the discharge provisions; Parties should take all circumstances into account when deciding upon investigating such reports.

1.7 Action which can be taken by States other than the flag or port States that have information on discharge violations (hereinafter referred to as coastal States):

- .1 coastal States that are Parties to MARPOL, upon receiving a report of pollution by oil or noxious liquid substances allegedly caused by a ship, may investigate the matter and collect such evidence as can be collected; for details of the desired evidence, reference is made to appendices 3 and 4;
- .2 if the investigation referred to under sub-paragraph .1 above discloses that the next port of call of the ship in question lies within its jurisdiction, the coastal State should also take port State action as set out in paragraphs 2.1 to 2.6 below;
- .3 if the investigation referred to in sub-paragraph .1 above discloses that the next port of call of the ship in question lies within the jurisdiction of another Party, then the coastal State should in appropriate cases furnish the evidence to that other Party and request that Party to take port State action in accordance with paragraphs 2.1 to 2.6 below; and
- .4 in either case referred to in sub-paragraphs .2 and .3 above and if the next port of call of the ship in question cannot be ascertained, the coastal State shall inform the flag State of the incident and of the evidence obtained.

2 Port State action

2.1 Parties shall appoint or authorize officers to carry out investigations for the purpose of verifying whether a ship has discharged oil or noxious liquid substances in violation of the provisions of MARPOL.

2.2 Parties may undertake such investigations on the basis of reports received from sources indicated in paragraph 1.6 above.

2.3 These investigations should be directed towards the gathering of sufficient evidence to establish whether the ship has violated the discharge requirements. Guidelines for the optimal collation of evidence are given in appendices 3 and 4.

2.4 If the investigations provide evidence that a violation of the discharge requirements took place within the jurisdiction of the port State, that port State shall either cause proceedings to be taken in accordance with its law, or furnish to the flag State all information and evidence in its possession about the alleged violation. When the port State causes proceedings to be taken, it shall inform the flag State.

2.5 Details of the report to be submitted to the flag State are set out in appendix 16.

2.6 The investigation might provide evidence that pollution was caused through damage to the ship or its equipment. This might indicate that a ship is not guilty of a violation of the discharge requirements of MARPOL Annex I or Annex II provided that:

- .1 all reasonable precautions have been taken after the occurrence of the damage or discovery of the discharge for the purpose of preventing or minimizing the discharge; and
- .2 the owner or the master did not act either with intent to cause damage or recklessly and with knowledge that damage would probably result.

2.7 However, action by the port State as set out in chapter 3 of these Procedures may be called for.

3 Inspection of crude oil washing (COW) operations

3.1 Regulations 18, 33 and 35 of MARPOL Annex I inter alia require that crude oil washing of cargo tanks be performed on certain categories of crude carriers. A sufficient number of tanks shall be washed in order that ballast water is put only in cargo tanks which have been crude oil washed. The remaining cargo tanks shall be washed on a rotational basis for sludge control.

3.2 Port State authorities may carry out inspections to ensure that crude oil washing is performed by all crude carriers either required to have a COW system or where the owner or operator chooses to install a COW system in order to comply with regulation 18 of MARPOL Annex I. In addition, compliance should be ensured with the operational requirements set out in the *Revised specifications for the design, operation and control of crude oil washing systems* (resolution A.446(XI), as amended). This can best be done in the ports where the cargo is unloaded.

3.3 Parties should be aware that the inspection referred to in paragraph 3.2 may also lead to the identification of a pollution risk, necessitating additional action by the port State as set out in chapter 3 of these Procedures.

3.4 Detailed guidelines for in-port inspections of crude oil washing procedures have been approved and published by IMO (Crude Oil Washing Systems, revised edition, 2000) and are set out in part 4 of appendix 3.

4 Inspection of unloading, stripping and prewash operations

4.1 Regulation 16 of MARPOL Annex II requires Parties to MARPOL to appoint or authorize surveyors for the purpose of implementing the regulation.

4.2 The provisions of regulation 16 are aimed at ensuring in principle that a ship having unloaded, to the maximum possible extent, noxious liquid substances of category X, Y or Z, proceeds to sea only if residues of such substances have been reduced to such quantities as may be discharged into the sea.

4.3 Compliance with these provisions is in principle ensured in the case of categories X, Y and Z substances through the application of a prewash in the unloading port and the discharge of prewash residue water mixtures to reception facilities, except that, in the case of non-solidifying and low viscosity categories Y and Z substances, requirements for the efficient stripping of a tank to negligible quantities apply in lieu of the application of a prewash. Alternatively, for a number of substances ventilation procedures may be employed for removing cargo residues from a tank.

4.4 Regulation 16.6 permits the Government of the receiving Party to exempt a ship proceeding to a port or terminal under the jurisdiction of another Party from the requirement to prewash cargo tanks and discharge residue/water mixtures to a reception facility.

4.5 Existing chemical tankers engaged on restricted voyages may by virtue of regulation 4.3 of MARPOL Annex II be exempted from the quantity limitation requirements of regulations 12.1 to 12.3. If a cargo tank is to be ballasted or washed, a prewash is required after unloading category Y or Z substances and prewash residue water mixtures must be discharged to shore reception facilities. The exemption should be indicated on the certificate.

4.6 A ship whose constructional and operational features are such that ballasting of cargo tanks is not required and cargo tank washing is only required for repairs or dry-docking may by virtue of regulation 4.4 be exempted from the provisions of regulation 12 of MARPOL Annex II, provided that all conditions mentioned in regulation 4.4 are complied with. Accordingly, the certificate of the ship should indicate that each cargo tank is only certified for the carriage of one named substance. It should also indicate the particulars of the exemption granted by the Administration in respect of pumping, piping and discharge arrangements.

4.7 Detailed instructions on efficient stripping and prewash procedures are included in a ship's P and A Manual. The Manual also contains alternative procedures to be followed in case of equipment failure.

4.8 Parties should be aware that the inspection referred to in paragraphs 1.3 and 1.4 above may lead to the identification of a pollution risk or of a contravention of the discharge provisions, necessitating port State action as set out in chapter 3 of these Procedures.

4.9 For details in respect of inspections under this section, reference is made to appendix 4.

Appendix 6

GUIDELINES FOR MORE DETAILED INSPECTIONS OF SHIP STRUCTURAL AND EQUIPMENT REQUIREMENTS

1 Introduction

If the port State control officer (PSCO) from general impressions or observations on board has clear grounds for believing that the ship might be substandard, the PSCO should proceed to a more detailed inspection, taking the following considerations into account.

2 Structure

2.1 The PSCO's impression of hull maintenance and the general state on deck, the condition of such items as ladderways, guard rails, pipe coverings and areas of corrosion or pitting should influence the PSCO's decision as to whether it is necessary to make the fullest possible examination of the structure with the ship afloat. Significant areas of damage or corrosion or pitting of plating and associated stiffening in decks and hull affecting seaworthiness or strength to take local loads, may justify detention. It may be necessary for the underwater portion of the ship to be checked. In reaching a decision, the PSCO should have regard to the seaworthiness and not the age of the ship, making an allowance for fair wear and tear over the minimum acceptable scantlings. Damage not affecting seaworthiness will not constitute grounds for judging that a ship should be detained, nor will damage that has been temporarily but effectively repaired for a voyage to a port for permanent repairs. However, in this assessment of the effect of damage, the PSCO should have regard to the location of crew accommodation and whether the damage substantially affects its habitability.

2.2 The PSCO should pay particular attention to the structural integrity and seaworthiness of bulk carriers and oil tankers and note that these ships must undergo the enhanced programme of inspection during surveys under the provision of SOLAS 1974 regulation XI-1/2.

2.3 The PSCO's assessment of the safety of the structure of those ships should be based on the Survey Report File carried on board. This file should contain reports of structural surveys, condition evaluation reports (translated into English and endorsed by or on behalf of the Administration), thickness measurement reports and a survey planning document. The PSCO should note that there may be a short delay in the update of the Survey Report File following survey. Where there is doubt that the required survey has taken place, the PSCO should seek confirmation from the RO.

2.4 If the Survey Report File necessitates a more detailed inspection of the structure of the ship or if no such report is carried, special attention should be given by the PSCO, as appropriate, to hull structure, piping systems in way of cargo tanks or holds, pump-rooms, cofferdams, pipe tunnels, void spaces within the cargo area and ballast tanks.

2.5 For bulk carriers, PSCOs should inspect holds' main structure for any obviously unauthorized repairs. For bulk carriers, the PSCO should verify that the bulk carrier booklet has been endorsed, the water level alarms in cargo holds are fitted, and where applicable, that any restrictions imposed on the carriage of solid bulk cargoes have been recorded in the booklet and the bulk carrier loading triangle is permanently marked.

3 Machinery spaces

3.1 The PSCO should assess the condition of the machinery and of the electrical installations such that they are capable of providing sufficient continuous power for propulsion and for auxiliary services.

3.2 During inspection of the machinery spaces, the PSCO should form an impression of the standard of maintenance. Frayed, disconnected or inoperative quick-closing valve wires, disconnected or inoperative extended control rods or machinery trip mechanisms, missing valve hand wheels, evidence of chronic steam, water and oil leaks, dirty tank tops and bilges or extensive corrosion of machinery foundations are pointers to an unsatisfactory organization of the systems' maintenance. A large number of temporary repairs, including pipe clips or cement boxes, will indicate reluctance to make permanent repairs.

3.3 While it is not possible to determine the condition of the machinery without performance trials, general deficiencies, such as leaking pump glands, dirty water gauge glasses, inoperable pressure gauges, rusted relief valves, inoperative or disconnected safety or control devices, evidence of repeated operation of diesel engine scavenge belt or crankcase relief valves, malfunctioning or inoperative automatic equipment and alarm systems, and leaking boiler casings or uptakes, would warrant inspection of the engine-room logbook and investigation into the record of machinery failures and accidents and a request for running tests of machinery.

3.4 If one electrical generator is out of commission, the PSCO should investigate whether power is available to maintain essential and emergency services and should conduct tests.

3.5 If evidence of neglect becomes evident, the PSCO should extend the scope of an investigation to include, for example, tests on the main and auxiliary steering gear arrangements, overspeed trips, circuit breakers.

3.6 It must be stressed that while detection of one or more of the above deficiencies would afford guidance to a substandard condition, the actual combination is a matter for professional judgement in each case.

4 Conditions of assignment of load lines

It may be that the PSCO has concluded that a hull inspection is unnecessary but, if dissatisfied on the basis of observations on deck, with items such as defective hatch closing arrangements, corroded air pipes and vent coamings, the PSCO should examine closely the conditions of assignment of load lines, paying particular attention to closing appliances, means of freeing water from the deck and arrangements concerned with the protection of the crew.

5 Life-saving appliances

5.1 The effectiveness of life-saving appliances depends heavily on good maintenance by the crew and their use in regular drills. The lapse of time since the last survey for a Safety Equipment Certificate can be a significant factor in the degree of deterioration of equipment if it has not been subject to regular inspection by the crew. Apart from failure to carry equipment required by a convention or obvious defects such as holed lifeboats, the PSCO should look for signs of disuse of, obstructions to, or defects with survival craft launching and recovery equipment, which may include paint accumulation, seizing of pivot points, absence of greasing, condition of blocks and falls, condition of lifeboat lifting hook attachment to the lifeboat hull and improper lashing or stowing of deck cargo.

5.2 Should such signs be evident, the PSCO would be justified in making a detailed inspection of all life-saving appliances. Such an examination might include the lowering of survival craft, a check on the servicing of liferafts, the number and condition of lifejackets and lifebuoys and ensuring that the pyrotechnics are still within their period of validity. It would not normally be as detailed as that for a renewal of the Safety Equipment Certificate and would concentrate on essentials for safe abandonment of the ship, but in an extreme case could progress to a full Safety Equipment Certificate inspection. The provision and functioning of effective overside lighting, means of alerting the crew and passengers and provision of illuminated routes to assembly points and embarkation positions should be given importance in the inspection.

6 Fire safety

6.1 Ships in general: The poor condition of fire and wash deck lines and hydrants and the possible absence of fire hoses and extinguishers in accommodation spaces might be a guide to a need for a close inspection of all fire safety equipment. In addition to compliance with convention requirements, the PSCO should look for evidence of a higher fire risk than normal; this might be brought about by a poor standard of cleanliness in the machinery space, which together with significant deficiencies of fixed or portable fire-extinguishing equipment could lead to a judgement of the ship being substandard. Queries on the method of structural protection should be addressed to the flag Administration and the PSCO should generally confine the inspection to the effectiveness of the arrangements provided.

6.2 Passenger ships: The PSCO should initially form an opinion of the need for inspection of the fire safety arrangements on the basis of consideration of the ship under the previous headings and, in particular, that dealing with fire safety equipment. If the PSCO considers that a more detailed inspection of fire safety arrangements is necessary, the PSCO should examine the fire control plan on board in order to obtain a general picture of the fire safety measures provided in the ship and consider their compliance with convention requirements for the year of build. Queries on the method of structural protection should be addressed to the flag Administration and the PSCO should generally confine the inspection to the effectiveness of the arrangements provided.

6.3 The spread of fire could be accelerated if fire doors are not readily operable. The PSCO should inspect for the operability and securing arrangements of those doors in the main zone bulkheads and stairway enclosures and in boundaries of high fire risk spaces, such as main machinery rooms and galleys, giving particular attention to those retained in the open position. Attention should also be given to main vertical zones which may have been compromised through new construction. An additional hazard in the event of fire is the spread of smoke through ventilation systems. Spot checks might be made on dampers and smoke flaps to ascertain the standard of operability. The PSCO should also ensure that ventilation fans can be stopped from the master controls and that means are available for closing main inlets and outlets of ventilation systems.

6.4 Attention should be given to the effectiveness of escape routes by ensuring that vital doors are not kept locked and that alleyways and stairways are not obstructed. Regarding the minimum width of external escape routes, the arrangements approved by the flag Administrations should be accepted.

6.5 The arrangements for the location of manually operated call points as approved by the flag Administrations should be accepted.

7 Regulations for preventing collisions at sea

A vital aspect of ensuring safety of life at sea is full compliance with the collision regulations. Based on observations on deck, the PSCO should consider the need for close inspection of lanterns and their screening and means of making sound and distress signals.

8 Cargo Ship Safety Construction Certificate

The general condition of the ship may lead the PSCO to consider matters other than those concerned with safety equipment and assignment of load lines, but nevertheless associated with the safety of the vessel, such as the effectiveness of items associated with the Cargo Ship Safety Construction Certificate, which can include pumping arrangements, means for shutting off air and oil supplies in the event of fire, alarm systems and emergency power supplies.

9 Cargo Ship Safety Radio Certificates

The validity of the Cargo Ship Safety Radio Certificates and associated Record of Equipment (Form R) may be accepted as proof of the provision and effectiveness of its associated equipment, but the PSCO should ensure that appropriate certificated personnel are carried for its operation and for listening periods. Requirements for maintenance of radio equipment are contained in SOLAS 1974 regulation IV/15. The radio log or radio records should be examined. Where considered necessary, operational checks may be carried out.

10 Means of access to ship

10.1 Prior to boarding a ship, the PSCO should assess the means of embarkation on and disembarkation from the ship. The PSCO should be guided by SOLAS 1974 regulation II-1/3-9, noting its application to ships constructed on or after 1 January 2010, but also noting that paragraph 3 of this regulation applies to all ships and requires that:

- .1 the means of embarkation and disembarkation shall be inspected and maintained in suitable condition for their intended purpose, taking into account any restrictions related to safe loading; and
- .2 all wires used to support the means of embarkation and disembarkation shall be maintained as specified in SOLAS 1974 regulation III/20.4.

10.2 In regard to the maintenance of the means of embarkation and disembarkation, the PSCO should refer to the *Guidelines for construction, installation, maintenance and inspection/survey of means of embarkation and disembarkation* (MSC.1/Circ.1331).

10.3 During the inspection, the PSCO should also ensure that the pilot transfer arrangements comply with SOLAS 1974 regulation V/23 and the *Unified interpretation of SOLAS regulation V/23* (MSC.1/Circ.1375/Rev.1 and MSC.1/Circ.1495/Rev.1).

11 Equipment in excess of convention or flag State requirements

Equipment on board which is expected to be relied on in situations affecting safety or pollution prevention must be in operating condition. If such equipment is inoperative and is in excess of the equipment required by an appropriate convention and/or the flag State, it should be repaired, removed or, if removal is not practicable, clearly marked as inoperative and secured.

Appendix 7

GUIDELINES FOR CONTROL OF OPERATIONAL REQUIREMENTS

PART 1

INSPECTION PROCESSES

1 Introduction

1.1 When, during a port State control inspection, the port State control officer (PSCO) has clear grounds according to section 2.2 of the present Procedures, the following onboard operational procedures may be checked in accordance with this resolution.

1.2 However, in exercising controls recommended in these Guidelines, the PSCO should not include any operational tests or impose physical demands which, in the judgement of the master, could jeopardize the safety of the ship, crew, passengers, control officers or cargo. Prior to requiring any practical operational control, the PSCO should review training and drill records and should inspect, as appropriate, the associated safety equipment and its maintenance records. For example, an enclosed space entry drill may be sufficiently verified without an actual enclosed space entry by verifying drill records, maintenance records, physical inspection and physical demonstrations by crew of breathing apparatus, safety harnesses and atmosphere testing instruments.

1.3 When carrying out operational control, the PSCO should ensure, as far as possible, no interference with normal shipboard operations, such as loading and unloading of cargo and ballasting, which is carried out under the responsibility of the master, nor should the PSCO require demonstration of operational aspects which would unnecessarily delay the ship.

1.4 Having assessed the extent to which operational requirements are complied with, the PSCO then has to exercise professional judgement to determine whether the operational proficiency of the crew as a whole is of a sufficient level to allow the ship to sail without danger to the ship or persons on board, or without presenting an unreasonable threat of harm to the marine environment.

1.5 When assessing the crew's ability to conduct an operational drill, the mandatory minimum requirements for familiarization and basic safety training for seafarers, as stated in STCW 1978, as amended, shall be used as a benchmark.

1.6 Definitions and abbreviations

The definitions and abbreviations used in this appendix are those of section 1.7 of the Procedures supplemented as follows:

Operational control: A control inspection to confirm the master and crew are familiar with essential shipboard procedures with respect to the safety of the ship and crew and protection of the environment and are able to apply such procedures. It includes a check on the effectiveness of communication and interaction and familiarity of the crew, including the human interface.

Functional test: A test of an item to prove the correct operation and function of equipment. Functional tests may be carried out during an initial or more detailed inspection.

2 Clear grounds

2.1 Clear grounds are defined in section 1.7.2 of the Procedures

2.2 In addition to the general examples of clear grounds in section 2.4 of the Procedures, clear grounds related to operation requirements are listed in appendix 11 section 6.3.2 items 7-13.

3 More detailed Inspection for operational requirements

3.1 A more detailed inspection should assess the ability of relevant crew to operate essential shipboard equipment that is relevant to their role. The responsible crew member, must be able to operate such equipment independent of others and care must be taken to ensure they are not coached through the process when asked to demonstrate their understanding.

3.2 A more detailed inspection should assess the familiarity of crew to essential shipboard procedures relevant to their role, the safety of the ship and the protection of the environment.

3.3 The PSCO should make an overall assessment of the effectiveness of communication and interaction and familiarity of the crew, including the human interface.

3.4 The PSCO can use the items in section 5 below as guidance in assessing the ability of the master or crew member to operate the ship. The desired outcome is to effectively assess compliance with operational requirements in order that corrective action(s) may be applied where necessary.

3.5 Drills

A more detailed inspection may include drills. Where drills are to be conducted these should be carried out at a safe pace. PSCO(s) should not expect to see operational activities including drills conducted in real time. Care should be taken to ensure that all crew familiarizes themselves with their duties and with the equipment. If necessary, drills should be stopped or suspended if the PSCO(s) considers that the crew are carrying out unsafe practices or if there is a real emergency. In addition, the following should be considered:

- .1 the PSCO(s) should devise the emergency scenario on which a drill will be based in conjunction with the master. Experience has shown that the best assessment is achieved when the PSCO(s) devises and controls the scenario, (in collaboration with the master) since there is then an element of uncertainty on the part of the ship's officers as to how a drill will progress and is more realistic to the actual onboard situation facing crew members in a critical situation; and
- .2 it is essential that meetings are held between the PSCOs and key members of the ship's personnel before and after any operational activity involving multiple crew members. An initial briefing should be used to explain in general terms how the activity will be conducted and should also enable the ship's staff to recognise the PSCOs who are witnessing the activity, it is recommended that all PSCOs witnessing the drill wear distinctive high visibility clothing to distinguish them from crew members.

3.6 Meeting on inspection outcomes and findings with regard to operational requirements

At the conclusion of the inspection a meeting should held with the master to ensure there is a common understanding of the outcomes and any findings of the detailed inspection, to identify any shortcomings and, if appropriate, where operational activity did not meet the required standard.

4 Communication

4.1 The PSCO may determine if the key crew members are able to communicate with each other, and with passengers, as appropriate, in such a way that the safe operation of the ship is not impaired, especially in emergency situations.

4.2 The PSCO may ask the master which languages are used as the working languages and may verify whether the language has been recorded in the logbook.

4.3 The PSCO may ensure that the key crew members are able to understand each other during the inspection or drills. The crew members assigned to assist passengers should be able to give the necessary information to the passengers in case of an emergency.

4.4 Language difficulty between the PSCO(s) and non-English-speaking crews can make it difficult to put across the intentions for the conduct of the inspection and any associated drills. Care needs to be exercised when an unsatisfactory inspection outcome is found to ensure there is a differentiation between the miscommunication between the PSCO and the crew and failure of operational requirements.

4.5 Passenger ships constructed on or after 1 July 2010 shall have on board a safety centre. The safety centre shall either be a part of the navigation bridge or be located in a separate space adjacent but having direct access to the navigation bridge.

4.6 The PSCO should verify effective means of communication between the safety centre, the central control station, the navigation bridge, the engine control room, the storage room(s) for fire extinguishing system(s) and fire equipment lockers are provided.

5 Assessing the ship with respect to Operational Requirements

5.1 If any of the following are found during a more detailed inspection a detention of the ship may be considered:

- .1 failure of deck officers and crew to monitor cargo loading operations and take precautions appropriate to that cargo;
- .2 lack of awareness of the operation of, and limitations of, navigation equipment or how to test such equipment (including navigation lights);
- .3 deck officers unable to demonstrate the operation of essential navigation equipment such as ECDIS and integrated navigations systems. This includes the monitoring and interrogating alarms on such systems;
- .4 there is evidence that the ship's navigation has been carried out in an unsafe manner including, but not limited to:

- .1 failure to monitor the ships position in accordance with shipboard procedures;
- .2 failure to verify the accuracy of position-fixing through use of multiple means of obtaining fixes;
- .3 failure to properly plan and assess a voyage; and
- .4 navigating the ship into danger or into restricted areas;
- .5 deck officers unfamiliar with the operation and testing of radio communications equipment and/or the mechanism by which marine safety information is provided to the ship;
- .6 relevant officers and crew unfamiliar with the locations of the starting positions or the starting operation of the firefighting equipment such as the emergency fire pump or the release system for the fixed fire-fighting system;
- .7 relevant officers and crew lack awareness of the location, operation and coverage area of ventilation stops in the accommodation, engine room and other protected areas;
- .8 officers and crew unaware of the location of fire alarm indicators in the accommodation and in the engine room;
- .9 relevant officers and crew not aware of the location and operation of the fuel cut off quick closing valves for main engine and auxiliary engines;
- .10 relevant officers and crew unaware of the operation of lifesaving equipment and how to effectively test such equipment;
- .11 relevant officers and crew unfamiliar with the operation of equipment, or procedures, intended to prevent maritime pollution; or
- .12 evidence of unsafe operations that pose a risk to life and the environment.

5.2 Observation by PSCO must be directly related to compliance with Convention requirements. In relating the deficiency, it is critical to note that having the necessary equipment installed and operational does not provide a capability as required by Convention unless the master and crew are familiar with the operation of the equipment and associated procedures as required by STCW Section A-I/4.4. Examples of deficiencies and relevant convention references are shown below:

.1 engineer officer unable to demonstrate the operation of fuel oil valves provided in accordance with SOLAS regulation II-2/4.2.2.3.4 from outside the machinery space;

Note 1: This would be related to SOLAS regulation XI-1/4

.2 engineer officer unable to demonstrate the operation of the Sewage treatment plant required by regulation 2 of MARPOL Annex IV; and

Note 1: This would be related to regulation 14 of MARPOL Annex IV.

Note 2: Where the Sewage Treatment Plant was found to be unserviceable or sewage had been discharged into the sea this should also be related as evidence of the failure of Operational Requirements.

.3 (on a ship subject to SOLAS regulation V/19.2.10) deck officer unable to demonstrate the process of planning and conducting a navigational passage and unable to demonstrate how to determine the ship position using ECDIS.

Note 1: This could be related to SOLAS regulation XI-1/4, or section A-I/4.4 of STCW. Depending on the nature and scope of the issues either could be used, noting SOLAS has a broader scope.

6 Detailed guidance on assessing compliance with operational requirements

6.1 Detailed guidance on areas to be inspected is provided in part 2 of this appendix. Detailed guidance is divided into means of assessing compliance day to day activities and emergency preparedness. An assessment of compliance in respect of both should be undertaken where the circumstances warrant it.

6.2 The PSCO should consider requesting a drill be conducted where ship's records indicate that the specified drill has not been conducted in accordance with the Convention requirements.

7 Witnessing and assessment of drills

7.1 If a drill will involve passengers, it is prudent to provide as much notice as possible before the start of the drill to enable the Master to inform the passengers about the drill. The information should be broadcast by public announcements in all relevant languages for the route concerned. The announcement should be repeated during the drill with appropriate intervals. The completion of the drill should be announced to the passengers.

7.2 During the conduct of a drill, the PSCO should consider questioning the crew members, particularly those assigned to assist any passengers, in order to get an impression of the safety awareness on board the ship.

- 7.3 When witnessing a drill, the PSCO should seek:
 - .1 confirmation that the crew follow what is required of them by the muster list;
 - .2 confirmation that there are sufficient personnel assigned to the various parties to cope with the duties given to them;
 - .3 confirmation that there is an effective means of communication between the party, the party leader and the bridge, and that relevant information is being exchanged;
 - .4 confirmation of the efficiency of the crew working as a team. This would be based on questioning of personnel and observation of their actions, the response times should be noted of the various parties in assembling at their stations and the reaction of the parties to unplanned events should also be noted;
 - .5 confirmation that key members of the crew are able to understand each other;

- .6 confirmation of the efficiency of the equipment used, for example:
 - .1 that the fire alarms are audible and efficient;
 - .2 that the fire and watertight doors close as required; and
 - .3 that items of personal fire-fighting equipment appear well maintained; and
- .7 confirmation that the response time was considered fast enough (taking into account safety of the drill as indicated in paragraph 2.5.4 of this appendix), considering the size of the ship and the locations of fire, personnel and fire-fighting equipment.
- 7.4 In the case of evacuation or abandon ship drills:
 - .1 confirmation that the escape arrangements for passengers/crew from lower decks are adequate, that the assembly or muster stations are clearly indicated, that the crew are familiar with the layout of the ship and are able to respond to changes in circumstances, for example directing passengers so as to avoid a smoke filled area; and
 - .2 confirmation that the boat lowering party is proficient and that boats are lowered and ready for embarkation with ancillary equipment deployed.

7.5 If the PSCO determines that the crew are unfamiliar with their duties or incapable of safely operating the life-saving and fire-fighting equipment, the PSCO should halt the drill, notify the master that the drill was unsuccessful and use their professional judgement to establish the next steps, noting the likelihood that this will establish "clear grounds" for a more detailed inspection.

7.6 Having assessed the extent to which operational requirements are complied with, the PSCO(s) should then exercise their professional judgement to determine whether the operational familiarity of the crew as a whole is of sufficient level to allow the ship to sail without danger to the ship or persons on board, or presenting an unreasonable threat of harm to the marine environment.

8 Detention under operational requirements

8.1 Paragraph 3.1.1 and sub-paragraph 3.1.1.4 of the Procedures identify a substandard ship as being one where operational safety is substantially below the standards required by the relevant convention and specifically, in the case of operational requirements, where there is:

"insufficiency of operational proficiency, or unfamiliarity of essential operational procedures by the crew"

8.2 In such cases the relevant operational requirements provisions of conventions require the port State to take such action as necessary to bring ships into compliance where it is found that the master and/or crew are unfamiliar with essential shipboard procedures. The following provisions are relevant:

- .1 SOLAS regulation XI-1/4;
- .2 MARPOL Annex I, regulation 11;

- .3 MARPOL Annex II, regulation 16.9;
- .4 MARPOL Annex III, regulation 9;
- .5 MARPOL Annex IV, regulation 14;
- .6 MARPOL Annex V, regulation 9;
- .7 MARPOL Annex VI, regulation 10; and
- .8 STCW, Article X and regulation I/4 and section A-I/4.

PART 2

GUIDANCE ON SPECIFIC INSPECTION ACTIVITIES

1 Introduction

This section provides detailed guidance on specific inspection activities described in part 1 with respect to the assessment of compliance with operation requirements in relation to day-to-day activities.

1.1 Bridge operation

1.1.1 The PSCO may determine if officers in charge of a navigational watch are familiar with bridge control and navigational equipment, changing the steering mode from automatic to manual and vice versa, and the ship's manoeuvring characteristics.

1.1.2 All officers in charge of a navigational watch should have knowledge of the location and operation of all safety and navigational equipment. Moreover, this officer(s) should be familiar with procedures which apply to the navigation of the ship in all circumstances and should be aware of all information available.

1.1.3 The PSCO may also verify the familiarity of the officers with all the information available to them such as manoeuvring characteristics of the ship, life-saving signals, up-to-date nautical publications, checklists concerning bridge procedures, instructions and manuals.

1.1.4 The Permit to Operate High-Speed Craft (HSC) includes limitations of the maximum significant wave height (and wind force for hovercraft) within which the craft may operate. When carrying out inspections of HSC, PSCOs may verify by the logbook and the weather records whether these limitations have been respected. PSCOs may find that a voyage had to be completed when worse weather conditions than permitted were encountered and not expected according to the weather forecast, but a new voyage should not commence in such conditions.

1.1.5 The PSCO may verify the familiarity of the officers with procedures such as periodic tests and checks of equipment, preparations for arrival and departure, changeover of steering modes, signalling, communications, alarm system, manoeuvring, emergencies and logbook entries.

1.2 Cargo operation

1.2.1 The PSCO may determine if ship's personnel assigned to specific duties related to the cargo and cargo equipment are familiar with those duties, any dangers posed by the cargo and with the measures to be taken in such a context. This will require the availability of all relevant cargo information as required by SOLAS 1974 regulation VI/2.

1.2.2 With respect to the carriage of solid bulk cargoes, the PSCO should verify, as appropriate, that cargo loading is performed in accordance with a ship's loading plan and unloading in accordance with a ship's unloading plan agreed by the ship and the terminal, taking into account the information provided by the loading instrument, where fitted.

1.2.3 The PSCO, when appropriate, may determine whether the responsible crew members are familiar with the relevant provisions of the International Maritime Solid Bulk Cargoes Code (IMSBC Code), particularly those concerning moisture limits and trimming of the cargo. Additionally, it is expected that the responsible crew members have appropriate knowledge of the recommendatory IMO Code of Safe Practice for Ships Carrying Timber Deck Cargoes (2011 TDC Code) and the Code of Safe Practice for Cargo Stowage and Securing (CSS Code) (non-mandatory, except mandatory sub-chapter 1.9), as amended.

1.2.4 Some solid materials transported in bulk can present a hazard during transport because of their chemical nature or physical properties. Section 2 of the IMSBC Code gives general precautions. Section 4 of the IMSBC Code contains the obligation imposed on the shipper to provide all necessary information to ensure safe transport of the cargo. The PSCO may determine whether all relevant details, including all relevant certificates of tests, have been provided to the master by the shipper.

1.2.5 For some cargoes, such as cargoes which are subject to liquefaction, special precautions are given (see section 7 of the IMSBC Code). The PSCO may determine whether all precautions are met with special attention to the stability of those ships engaged in the transport of cargoes subject to liquefaction and solid hazardous waste in bulk.

1.2.6 Officers responsible for cargo handling and operation and key crew members of oil tankers, chemical tankers and liquefied gas carriers should be familiar with the cargo and cargo equipment and with the safety measures as stipulated in the relevant sections of the IBC and IGC Codes.

1.2.7 For the carriage of grain in bulk, reference is made to part C of chapter VI of SOLAS 1974 and the mandatory International Code for the Safe Carriage of Grain in Bulk (Grain Code).

1.2.8 The PSCO may determine whether the operations and loading manuals include all the relevant information for safe loading and unloading operations in port as well as in transit conditions.

1.3 Operation of machinery

1.3.1 The PSCO may determine if responsible ship's personnel are familiar with their duties related to operating essential machinery, such as:

- .1 emergency and standby sources of electrical power;
- .2 auxiliary steering gear;

- .3 bilge and fire pumps; and
- .4 any other equipment essential in emergency situations.

1.3.2 The PSCO may verify whether the responsible ship's personnel are familiar with, inter alia:

- .1 emergency generator:
 - .1 actions which are necessary before the engine can be started;
 - .2 different possibilities to start the engine in combination with the source of starting energy; and
 - .3 procedures when the first attempts to start the engine fail; and
- .2 standby generator engine:
 - .1 possibilities to start the standby engine, automatic or by hand;
 - .2 blackout procedures; and
 - .3 load-sharing system.

1.3.3 The PSCO may verify whether the responsible ship's personnel are familiar with, inter alia:

- .1 which type of auxiliary steering gear system applies to the ship;
- .2 how it is indicated which steering gear unit is in operation; and
- .3 what action is needed to bring the auxiliary steering gear into operation.

1.3.4 The PSCO may verify whether the responsible ship's personnel are familiar with, inter alia:

- .1 bilge pumps:
 - .1 number and location of bilge pumps installed on board the ship (including emergency bilge pumps);
 - .2 starting procedures for all these bilge pumps;
 - .3 appropriate valves to operate; and
 - .4 most likely causes of failure of bilge pump operation and their possible remedies; and
- .2 fire pumps:
 - .1 number and location of fire pumps installed on board the ship (including the emergency fire pump);
 - .2 starting procedures for all these pumps; and

.3 appropriate valves to operate.

1.3.5 The PSCO may verify whether the responsible ship's personnel are familiar with, inter alia:

- .1 starting and maintenance of lifeboat engine and/or rescue boat engine;
- .2 local control procedures for those systems which are normally controlled from the navigating bridge;
- .3 use of the emergency and fully independent sources of electrical power of radio installations;
- .4 maintenance procedures for batteries;
- .5 emergency stops, fire detection system and alarm system operation of watertight and fire doors (stored energy systems); and
- .6 change of control from automatic to manual for cooling water and lube oil systems for main and auxiliary engines.

1.4 Manuals, instructions, etc.

1.4.1 The PSCO may determine if the appropriate crew members are able to understand the information given in manuals, instructions, etc. relevant to the safe condition and operation of the ship and its equipment, and if they are aware of the requirements for maintenance, periodic testing, training, drills and recording of logbook entries.

1.4.2 The following information, inter alia, should be provided on board and PSCOs may determine whether it is in a language or languages understood by the crew and whether crew members concerned are aware of the contents and are able to respond accordingly:

- .1 instructions concerning the maintenance and operation of all the equipment and installations on board for the fighting and containment of fire should be kept under one cover, readily available in an accessible position;
- .2 clear instructions to be followed in the event of an emergency should be provided for every person on board;
- .3 illustrations and instructions in appropriate languages should be posted in passenger cabins and be conspicuously displayed at muster stations and other passenger spaces to inform passengers of their muster station, the essential action they must take in an emergency and the method of donning lifejackets;
- .4 posters and signs should be provided on or in the vicinity of survival craft and their launching controls and shall illustrate the purpose of controls and the procedures for operating the appliance and give relevant instructions or warnings;
- .5 instructions for onboard maintenance of life-saving appliances;
- .6 training manuals should be provided in each crew mess room and recreation room or in each crew cabin; the training manual, which may comprise several

volumes, should contain instructions and information, in easily understood terms illustrated wherever possible, on the life-saving appliances provided in the ship and on the best method of survival; and

.7 SOPEP in accordance with regulation 37 of MARPOL Annex I, or SMPEP for noxious liquid substances in accordance with regulation 17 of MARPOL Annex II, where applicable; and.8 stability booklet, associated stability plans, stability information and approved stability instrument for tankers.

1.5 Oil and oily mixtures from machinery spaces

1.5.1 The PSCO may determine if all operational requirements of MARPOL Annex I have been met, taking into account:

- .1 the quantity of oil residues generated;
- .2 the capacity of the sludge and bilge water holding tank; and
- .3 the capacity of the oily water separator.

1.5.2 An inspection of the ORB should be made. The PSCO may determine if reception facilities have been used and note any alleged inadequacy of such facilities.

1.5.3 The PSCO may determine whether the responsible officer is familiar with the handling of sludge and bilge water. The relevant items from the guidelines for systems for handling oily wastes in machinery spaces of ships may be used as guidance. Taking into account the above, the PSCO may determine if the ullage of the sludge tank is sufficient for the expected generated sludge during the next intended voyage. The PSCO may verify that, in respect of ships for which the Administration has waived the requirements of regulations 14(1) and (2) of MARPOL Annex I, all oily bilge water is retained on board for subsequent discharge to a reception facility.

1.5.4 When reception facilities in other ports have not been used because of inadequacy, the PSCO should advise the master to report the inadequacy of the reception facility to the ship's flag State, in conformity with the *Format for reporting alleged inadequacies of port reception facilities* (MEPC.1/Circ.834/Rev.1, appendix 1 of the annex), as may be amended.

1.6 Loading, unloading and cleaning procedures for cargo spaces of tankers

1.6.1 The PSCO may determine if all operational requirements of MARPOL Annexes I or II have been met, taking into account the type of tanker and the type of cargo carried, including the inspection of the ORB and/or CRB. The PSCO may determine if the reception facilities have been used and note any alleged inadequacy of such facilities.

1.6.2 For the control on loading, unloading and cleaning procedures for tankers carrying oil, reference is made to paragraphs 3.1 to 3.4 of appendix 5 where guidance is given for the inspection of crude oil washing (COW) operations. In appendix 3, the PSCO may find detailed guidelines for in-port inspection of crude oil washing procedures.

1.6.3 For the control on loading, unloading and cleaning procedures for tankers carrying noxious liquid substances, reference is made to paragraphs 4.1 to 4.9 of appendix 5 where guidance is given for the inspection of unloading, stripping and prewash operations. In appendix 4 more detailed guidelines for these inspections are given.

1.6.4 When reception facilities in other ports have not been used because of inadequacy, the PSCO should advise the master to report the inadequacy of the reception facility to the ship's flag State, in conformity with MEPC.1/Circ.834/Rev.1, as may be amended.

1.6.5 The Garbage Record Book may be presented in an electronic format. A declaration from the Administration should be viewed in order to accept this electronic record book. If a declaration cannot be provided, a hard copy record book will need to be presented for examination.

1.6.6 When a ship is permitted to proceed to the next port with residues of noxious liquid substances on board in excess of those permitted to be discharged into the sea during the ship's passage, it should be ascertained that the residues can be received by that port. At the same time, that port should be informed, if practicable.

1.7 Dangerous goods and harmful substances in packaged form

1.7.1 The PSCO may determine if the required shipping documents for the carriage of dangerous goods and harmful substances carried in packaged form are provided on board and whether the dangerous goods and harmful substances are properly stowed and segregated and the crew members are familiar with the essential action to be taken in an emergency involving such packaged cargo (see SOLAS 1974 regulation VII/3).

1.7.2 Ship types and cargo spaces of ships of over 500 gross tonnage built on or after 1 September 1984 and ship types and cargo spaces of ships of less than 500 gross tonnage built on or after 1 February 1992 are to fully comply with the requirements of SOLAS 1974 chapter II-2. Administrations may reduce the requirements for cargo ships of less than 500 gross tonnage but such reductions shall be recorded in the Document of Compliance. A Document of Compliance is not required for ships which only carry class 6.2, class 7 or dangerous goods in limited quantities and excepted quantities.

1.7.3 MARPOL Annex III contains requirements for the carriage of harmful substances in packaged form which are identified in the IMDG Code as marine pollutants. Cargoes which are determined to be marine pollutants should be labelled and stowed in accordance with MARPOL Annex III.

1.7.4 The PSCO may determine whether a Document of Compliance is on board and whether the ship's personnel are familiar with this document provided by the Administration as evidence of compliance of construction and equipment with the requirements. Additional control may consist of:

- .1 checking whether the dangerous goods have been stowed on board in conformity with the Document of Compliance, using the dangerous goods manifest or the stowage plan, required by SOLAS 1974 chapter VII; this manifest or stowage plan may be combined with the one required under MARPOL Annex III;
- .2 checking whether inadvertent pumping of leaking flammable or toxic liquids is not possible in case these substances are carried in under-deck cargo spaces; or
- .3 determining whether the ship's personnel are familiar with the relevant provisions of the Medical First Aid Guide and Emergency Procedures for Ships Carrying Dangerous Goods.

1.8 Garbage

1.8.1 The PSCO may determine if all operational requirements of MARPOL Annex V have been met. The PSCO may determine if the reception facilities have been used and note any alleged inadequacy of such facilities.

1.8.2 The 2017 Guidelines for the implementation of MARPOL Annex V (resolution MEPC.295(71)), as may be amended, are to assist ship operators complying with the requirements set forth in Annex V and domestic laws.

- 1.8.3 The PSCO may determine whether:
 - .1 ship's personnel are aware of these Guidelines, in particular section 2 on "Garbage management"; and
 - .2 ship's personnel are familiar with the disposal and discharge requirements under MARPOL Annex V inside and outside a special area and are aware of the areas determined as special areas under MARPOL Annex V.

1.8.4 When reception facilities in other ports have not been used because of inadequacy, the PSCO should advise the master to report the inadequacy of the reception facility to the ship's flag State, in conformity with MEPC.1/Circ.834/Rev.1, as may be amended.

1.9 Sewage

- 1.9.1 The PSCO may determine:
 - .1 if all operational requirements of MARPOL Annex IV have been met; the PSCO may determine if the sewage treatment system, comminuting and disinfecting system or holding tank has been used and note any alleged inadequacy of the system or holding tank; and
 - .2 that appropriate ship's personnel are familiar with the correct operation of the sewage treatment system, comminuting and disinfecting system or holding tank.

1.9.2 The PSCO may determine whether appropriate ship's personnel are familiar with the discharge requirements of regulation 11 of MARPOL Annex IV.

1.9.3 When reception facilities in other ports have not been used because of inadequacy, the PSCO should advise the master to report the inadequacy of the reception facility to the ship's flag State, in conformity with the waste reception facility reporting requirements (MEPC.1/Circ.834/Rev.1, as may be amended).

1.10 Air pollution prevention

The PSCO may determine whether:

- .1 the master or crew is familiar with the procedures to prevent emissions of ozone-depleting substances and sulphur when equivalent arrangements are in place;
- .2 the master or crew is familiar with the proper operation and maintenance of diesel engines, in accordance with their Technical Files;

- .3 the master or crew has undertaken the necessary fuel changeover procedures or equivalent, associated with demonstrating compliance within a SO_x emission control area;
- .4 the master or crew is familiar with the garbage screening procedure to ensure that prohibited garbage is not incinerated;
- .5 the master or crew is familiar with the operation of the shipboard incinerator, as required by regulation 16.2 of MARPOL Annex VI, within the limit provided in appendix IV to the Annex, in accordance with the operational manual;
- .6 the master or crew recognizes the regulation of emissions of volatile organic compounds (VOCs), when the ship is in ports or terminals under the jurisdiction of a Party to the 1997 Protocol to MARPOL in which VOCs emissions are to be regulated, and is familiar with the proper operation of a vapour collection system approved by the Administration (in case the ship is a tanker as defined in regulation 2.27 of MARPOL Annex VI); and
- .7 the master or crew is familiar with bunker delivery procedures in respect of bunker delivery notes and retained samples as required by regulation 18 of MARPOL Annex VI.

2 Introduction

This section provides detailed guidance on specific inspection activities described in part 1 with respect to the assessment of preparedness for emergencies and drills.

2.1 Muster list

2.1.1 The PSCO may determine if the crew members are aware of their duties indicated in the muster list and that they are familiar with the duties assigned to them and are aware of the locations where they should perform their duties, this is done by asking the crew relevant questions. This could be done prior to the drill or during the drill, for instance questioning of stairway guides on a passenger ship.

2.1.2 To determine whether the muster list is up to date, the PSCO(s) may require an up-to-date crew list.

2.1.3 The PSCO may ensure that muster lists are exhibited in conspicuous places throughout the ship, including the navigational bridge, the engine-room and the crew accommodation spaces. When determining if the muster list is in accordance with the regulations, the PSCO may verify whether:

- .1 the muster list shows the duties assigned to the different members of the crew;
- .2 the muster list specifies which officers are assigned to ensure that life-saving and fire appliances are maintained in good condition and are ready for immediate use;
- .3 the muster list specifies the substitutes for key persons who may become disabled, taking into account that different emergencies may call for different actions;

- .4 the muster list shows the duties assigned to crew members in relation to passengers in case of emergency; and
- .5 the format of the muster list used on passenger ships is approved and is drawn up in the language or languages required by the ship's flag State and in the English language.

2.1.4 To determine whether the muster list is up to date, the PSCO may require an up-to-date crew list, if available, to verify this.

2.1.5 The PSCO may determine whether the duties assigned to crew members manning the survival craft (lifeboats or liferafts) are in accordance with the regulations and verify that a deck officer or certificated person is placed in charge of each survival craft to be used. However, the Administration (of the flag State), having due regard to the nature of the voyage, the number of persons on board and the characteristics of the ship, may permit persons practised in the handling and operation of liferafts to be placed in charge of liferafts in lieu of persons qualified as above. A second-in-command shall also be nominated in the case of lifeboats.

2.1.6 Every motorized survival craft shall have a person assigned who is capable of operating the engine and carrying out minor adjustments.

2.2 Communication during drills

2.2.1 The PSCO(s) may determine if the key crew members are able to communicate with each other, and with passengers, as appropriate, in such a way that the safe operation of the ship is not impaired, especially in emergency situations.

- 2.2.2 For drills, key crew members could be but are not limited to:
 - .1 bridge team including GMDSS operators who must also be able to communicate with the shore and other vessels;
 - .2 fire parties;
 - .3 damage control parties;
 - .4 boat preparation parties; or
 - .5 passenger muster personnel on passenger ships.

2.2.3 The PSCO(s) should verify the working language of the vessel. The crew members assigned to assist passengers should be able to give the necessary information to the passengers in case of an emergency.

2.2.4 The PSCO(s) should determine, if UHF or VHF handheld radios are being used for drills, that the crew are familiar with the equipment, that they are aware of reception dead zones/areas and what alternative communication methods are available.

2.2.5 When drills are being conducted the PSCO(s) should establish that there are sufficient personnel on the bridge to make decisions, navigate the ship as necessary and deal with the considerable amount of communication that is likely.

2.2.6 When a ship is in difficulty it is likely that shore-based organizations, such as the operator of the ship and regional rescue coordination centres, will need to be involved. The PSCO should confirm the master and crew are aware of procedures where shore-based communication is required and how such communication can be established.

2.3 Search and rescue plan

For passenger ships, the PSCO may verify that there is on board an approved plan for cooperation with appropriate search and rescue services in the event of an emergency.

2.4 Fire and abandon ship drills

2.4.1 The PSCO witnessing a fire and abandon ship drill should ensure that the crew members are familiar with their duties and the proper use of the ship's installations and equipment.

2.4.2 When setting a drill scenario, witnessing the drill and finally assessing the standard of the drill, it is important to emphasize that the PSCO is not looking for an exceptional drill, particularly on cargo ships. The main points for the PSCO to be satisfied are:

- .1 In the event of a shipboard emergency can the crew organize themselves into an effective team to tackle the emergency?
- .2 Can the crew communicate effectively?
- .3 Is the master in control and is information flowing to/from the command centre?
- .4 In the event of the situation getting out of hand can the crew safely abandon the ship?

2.4.3 It is important that when setting the scenario the PSCO clearly explains to the master exactly what is required and expected during the drill, bearing in mind there may be language difficulties. PSCOs should not be intimidating, not interfere during the drill nor offer advice. The PSCO should stand back and observe only, making appropriate notes. It is important to emphasize that the PSCO's role is not to teach or train but to witness.

2.4.4 Drills should be carried out at a safe speed. PSCOs should not expect to see operational drills conducted in real time. During drills, care should be taken to ensure that everybody familiarizes themself with their duties and with the equipment. If necessary, drills should be stopped if the PSCO considers that the crew are carrying out unsafe practices or if there is a real emergency.

2.5 Fire drills

2.5.1 The PSCO may witness a fire drill carried out by the crew assigned to these duties on the muster list. After consultation with the master of the vessel, one or more specific locations of the ship may be selected for a simulated fire. A crew member may be sent to the location(s) and activate a fire alarm system or use other means to give the alarm.

2.5.2 At the location the PSCO can describe the fire indication to the crew member and observe how the report of fire is relayed to the bridge or damage control centre. At this point most ships will sound the crew alarm to summon the fire-fighting parties to their stations. The PSCO should observe the fire-fighting party arriving on the scene, breaking out their equipment and fighting the simulated fire. Team leaders should be giving orders as appropriate
to their crews and passing the word back to the bridge or damage control centre on the conditions. The fire-fighting crews should be observed for proper donning and use of their equipment. The PSCO should make sure that all the gear is complete. Merely mustering the crew with their gear is not acceptable. Crew response to personnel injuries can be checked by selecting a crew member as a simulated casualty. The PSCO should observe how the word is passed and the response of stretcher and medical teams. Handling a stretcher properly through narrow passageways, doors and stairways is difficult and takes practice.

2.5.3 The drill should, as far as practicable, be conducted as if there were an actual emergency.

2.5.4 Those crew members assigned to other duties related to a fire drill, such as the manning of the emergency generators, the CO_2 room, the sprinkler and emergency fire pumps, should also be involved in the drill. The PSCO may ask these crew members to explain their duties and, if possible, to demonstrate their familiarity.

2.5.5 On passenger ships, special attention should be paid to the duties of those crew members assigned to the closing of manually operated doors and fire dampers. These closing devices should be operated by the responsible persons in the areas of the simulated fire(s) during the drill. Crew members not assigned to the fire-fighting teams are generally assigned to locations throughout the passenger accommodations to assist in passenger evacuation. These crew members should be asked to explain their duties and the meaning of the various emergency signals and asked to point out the two means of escape from the area, and where the passengers are to report. Crew members assigned to assist passengers should be able to communicate at least enough information to direct a passenger to the proper muster and embarkation stations.

2.6 Abandon ship drills

2.6.1 After consultation with the master, the PSCO may require an abandon ship drill for one or more survival craft. The essence of this drill is that the survival craft are manned and operated by the crew members assigned to them on the muster list. If possible, the PSCO should include the rescue boat(s) in this drill. SOLAS 1974 chapter III gives specific requirements on abandon ship training and drills, of which the following principles are particularly relevant.

2.6.2 The drill should, as far as practicable, be conducted as if there were an actual emergency.

- 2.6.3 The abandon ship drill should include:
 - .1 summoning crew, and passengers where applicable, to the muster station(s) with the required alarm and ensuring that they are aware of the order to abandon ship as specified in the muster list;
 - .2 reporting to the stations and preparing for the duties described in the muster list;
 - .3 checking that crew, and passengers where applicable, are suitably dressed;
 - .4 checking that lifejackets are correctly donned;
 - .5 lowering at least one lifeboat after the necessary preparation for launching;
 - .6 starting and operating the lifeboat engine;

- .7 operating the davits used for launching liferafts;
- .8 conducting a mock search and rescue of passenger trapped in their staterooms (if applicable);
- .9 giving instructions in the use of radio life-saving appliances;
- .10 testing emergency lighting and low location lights if applicable for mustering and abandonment; and
- .11 if the ship is fitted with marine evacuation systems, exercising the procedures required for the deployment of such systems up to the point immediately preceding actual deployment.

2.6.4 If the lifeboat lowered during the drill is not the rescue boat, the rescue boat should be lowered as well, taking into account that it is boarded and launched in the shortest possible time. The PSCO should ensure that crew members are familiar with the duties assigned to them during abandon ship operations and that the crew member in charge of the survival craft has complete knowledge of the operation and equipment of the survival craft. Care needs to be taken when requiring a ship to lower lifeboats. The number of persons inside the lifeboats during launching for the purpose of a drill should be at the master's discretion, noting that SOLAS 1974 does not require persons in the lifeboat during lowering and recovery. The purpose of this is to reduce the risk of accidents during launching and recovery; however, this must be balanced out with the risk of embarking/disembarking while the boat is still in the water, if the boat is to be taken away and run.

2.6.5 Each survival craft should be stowed in a state of continuous readiness so that two crew members can carry out preparations for embarking and launching in less than five minutes.

2.7 Enclosed space entry and rescue drills

2.7.1 After consultation with the master, the PSCO may require an enclosed space entry and rescue drill. The essence of this drill is to confirm that crew members are familiar with the procedure to enter an enclosed space and to rescue personnel safely, can demonstrate an enclosed space entry and rescue drill, and can communicate effectively when entering an enclosed space in case of planned entry and/or an emergency situation.

2.7.2 The place of the drill can be selected at an assumed enclosed space; it is not necessary to select an actual enclosed space.

2.7.3 The PSCO should check the structure of the enclosed space, the scenarios of the drills and the responsible officers listed on the muster list where applicable.

- 2.7.4 The enclosed space entry and rescue drill should include:
 - .1 checking and use of personal protective equipment required for entry;
 - .2 checking and use of communication equipment and procedures;
 - .3 checking and use of instruments for measuring the atmosphere in enclosed spaces;
 - .4 checking and use of rescue equipment and procedures; and

.5 instructions in first aid and resuscitation techniques.

2.8 Emergency steering drills

2.8.1 After consultation with the master, the PSCO may require an emergency steering drill. The essence of this drill is to confirm crew members are familiar with the procedure for emergency steering.

2.8.2 The PSCO may check the procedure and means of communication at both the navigation bridge and the steering gear room.

- 2.8.3 The emergency steering drills should include:
 - .1 direct control within the steering gear compartment;
 - .2 communication procedure with the navigational bridge; and
 - .3 operation of alternative power supplies where applicable.

2.9 Damage Control Plan and Shipboard Oil Pollution Emergency Plan (SOPEP) or Shipboard Marine Pollution Emergency Plans (SMPEP)

2.9.1 The PSCO may determine if a damage control plan is provided on a passenger ship and whether the crew members are familiar with their duties and the proper use of the ship's installations and equipment for damage control purposes. The same applies with regard to SOPEP on all ships and SMPEP where applicable.

2.9.2 The PSCO may determine if the officers of the ship are aware of the contents of the damage control booklet, which should be available to them, or of the damage control plan.

2.9.3 The officers may be asked to explain the action to be taken in various damage conditions.

2.9.4 The officers may also be asked to explain about the boundaries of the watertight compartments, the openings therein with the means of closure and position of any controls thereof and the arrangements for the correction of any list due to flooding.

2.9.5 The officers should have a sound knowledge of the effect of trim and stability of their ship in the event of damage to and consequent flooding of a compartment and countermeasures to be taken.

2.10 Fire control plan

2.10.1 The PSCO may determine if a fire control plan or booklet is provided, whether the crew members are familiar with the information given in the fire control plan or booklet, and whether, for tankers, crew members are familiar with the approved stability instrument.

2.10.2 The PSCO may verify that fire control plans are permanently exhibited for the guidance of the ship's officers. Alternatively, booklets containing the information about the fire control plan may be supplied to each officer, and one copy should at all times be available on board in an accessible position. Plans and booklets should be kept up to date, any alterations being recorded therein as soon as possible.

2.10.3 The PSCO may determine that the responsible officers, especially those who are assigned to related duties on the muster list, are aware of the information provided by the fire control plan or booklet and how to act in case of a fire.

2.10.4 The PSCO may ensure that the officers in charge of the ship are familiar with the principal structural members which form part of the various fire sections and the means of access to the different compartments.

GUIDELINES FOR PORT STATE CONTROL OFFICERS ON THE ISM CODE

1 GENERAL

1.1 The International Safety Management Code (ISM Code) was adopted by the Assembly at its eighteenth session by resolution A.741(18) and was amended by resolutions MSC.104(73), MSC.179(79), MSC.195(80), MSC.273(85) and MSC.353(92). The ISM Code has been made mandatory through SOLAS 1974 regulation IX/3.

1.2 The Administration is responsible for verifying compliance with the requirements of the ISM Code and issuing Documents of Compliance to companies and Safety Management Certificates to ships. This verification is carried out by the Administration or a recognized organization (RO).

1.3 Port State control officers (PSCOs) do not perform safety management audits. ISM auditing is the responsibility of the flag State and the company and does not fall under the scope of port State control. PSCOs conduct inspections of ships, which are a sampling process and give a snapshot of the vessel on a particular day.

1.4 The SMS documentation is in the ship's working language, which may not be understood by the PSCO. The procedure may not be harmonized if the PSCO is only able to review the SMS documentation on those ships where they can understand the language.

2 GOALS AND PURPOSE

2.1 The Guidelines provide guidance to PSCOs for the harmonized application of related technical or operational deficiencies found in relation to the ISM Code during a PSC inspection.

3 APPLICATION

3.1 The ISM Code applies to the following types of ships engaged in international voyages:

- .1 all passenger ships including passenger high-speed craft;
- .2 oil tankers, chemical tankers, gas carriers, bulk carriers and cargo high-speed craft of 500 gross tonnage and above; and
- .3 other cargo ships and self-propelled mobile offshore drilling units (MODUs) of 500 gross tonnage and above.

3.2 For establishing the applicability SOLAS 1974 chapter IX and the ISM Code, "gross tonnage" means the gross tonnage of the ship as determined under the provisions of TONNAGE 1969, and as stated on the International Tonnage Certificate of the ship.

3.3 The ISM Code does not apply to government-operated ships used for non-commercial purposes.

4 RELEVANT DOCUMENTATION

4.1 Applicable documentation for these Guidelines is as follows:

- .1 SOLAS 1974;
- .2 ISM Code;
- .3 Copy of the Interim DOC or copy of the DOC;
- .4 Interim SMC or SMC; and
- .5 MSC/Circ.1059-MEPC/Circ.401, as may be amended.

5 DEFINITIONS AND ABBREVIATIONS

- SOLAS: International Convention for the Safety of Life at Sea, 1974, as amended
- ISM Code: International Safety Management Code

"The International Management Code for the Safe Operation of Ships and for Pollution Prevention", as adopted by resolution A.741(18), as amended

Procedures for *Procedures for port State control, 2021*, as adopted by resolution port State *A.*[...](32), as may be amended control:

- Company: The owner of the ship or any other organization or person such as the manager, or the bareboat charterer, who has assumed the responsibility for operation of the ship from the shipowner and who, on assuming such responsibility, has agreed to take over all duties and responsibility imposed by the Code
- Administration: The Government of the State whose flag the ship is entitled to fly
- DOC: Document of Compliance

A document issued to a company which complies with the requirements of the ISM Code

SMC: Safety management certificate

A document issued to a ship which signifies that the company and its shipboard management operate in accordance with the approved safety management system

SMS: Safety management system

A structured and documented system enabling company personnel to implement effectively the company safety and environmental protection policy

Objective Quantitative or qualitative information, records or statements of Evidence: fact pertaining to safety or to the existence and implementation of a safety management system element, which is based on observation, measurement or test and which can be verified

- Valid certificate: A certificate that has been issued, electronically or on paper, directly by a Party to a relevant convention or on its behalf by a recognized organization, and contains accurate and effective dates, meets the provisions of the relevant convention, and with which the particulars of the ship, its crew and its equipment correspond
- PSC: Port State control
- PSCO: Port State control officer
- RO: Recognized organization

An organization recognized by the Administration

- MODU: Mobile offshore drilling unit
- ISM-related: A technical and/or operational deficiency which has been assessed by the PSCO to be objective evidence of a failure, or lack of effectiveness, of the implementation of the ISM Code, and which is marked as "ISM-related" in the inspection report

ISM deficiency: A deficiency that is cited against the ISM Code

6 INSPECTION OF SHIP

6.1 Initial inspection

6.1.1 Initial inspection should be carried out in accordance with the *Procedures for port State control.*

6.1.2 During the initial PSC inspection, the PSCO should verify that the ship carries the ISM certificates according to the provisions of chapter IX of SOLAS 1974 and the ISM Code by examining the copy of the DOC and the SMC, for which the following points are to be considered:

- .1 A copy of the DOC should be on board. However, according to the provisions of SOLAS 1974, the copy of the DOC is not required to be authenticated or certified. The copy of the DOC should have the required endorsements.
- .2 The SMC is not valid unless the operating company holds a valid DOC for that ship type. The ship type in the SMC should be included in the DOC and the company's particulars should be the same on both the DOC and the SMC. The SMC should have the required endorsements.
- .3 The validity of an Interim DOC should not exceed a period of 12 months. The validity of an Interim SMC should not exceed a period of six months. In special cases, the Administration, or at the request of the Administration another Government, may extend the validity of the Interim SMC for a period which should not exceed six months from the date of expiry.
- .4 ROs may issue a short-term DOC or SMC not exceeding five months, while the full-term certificate is being prepared in accordance with their internal procedures. If a renewal verification has been completed and a new SMC cannot be issued or placed on board the ship before the expiry date of the existing certificate, the Administration or RO may endorse the existing

certificate. Such a certificate should be accepted as valid for a further period which should not exceed five months from the expiry date.

- .5 If a ship at the time when an SMC expires is not in a port in which SMC verification is to be carried out, the Administration may extend the period of validity of the SMC but this extension should be granted only for the purpose of allowing the ship to complete its voyage to the port in which SMC verification is to be carried out, and then only in cases where it appears proper and reasonable to do so.
- .6 No SMC should be extended for a period of longer than three months, and the ship to which an extension is granted should not, on its arrival in the port in which SMC verification is to be carried out, be entitled by virtue of such extension to leave that port without having a new SMC. When the renewal verification is completed, the new SMC should be valid until a date not exceeding five years from the expiry date of the existing SMC before the extension was granted.
- .7 If no technical or operational-related deficiencies are found during an initial inspection carried out in accordance with the *Procedures for port State control* and guidelines, there is no need to consider the ISM aspect.

6.2 Clear grounds

6.2.1 Since the PSCO is not carrying out a safety management audit of the SMS during a PSC inspection, the term "clear grounds" is not applicable in this context.

6.2.2 Clear grounds and the subsequent more detailed inspection only exist for technical or operational deficiencies.

6.3 More detailed inspection

6.3.1 If a more detailed inspection for technical or operational-related deficiencies is carried out, this should be done in accordance with the *Procedures for port State control*. Any technical and/or operational deficiencies found during this inspection should be individually or collectively considered by the PSCO, using their professional judgement, to indicate that either:

- .1 these do not show a failure, or lack of effectiveness, of the implementation of the ISM Code; or
- .2 there is a failure, or lack of effectiveness, of the implementation of the ISM Code.

6.3.2 If an outstanding ISM-related deficiency from a previous PSC inspection exists and the current PSC inspection is more than three months later, the PSCO will verify, during the present PSC inspection, the effectiveness of any corrective action taken by the company by examining the areas of the technical and/or operational deficiencies of the previous PSC inspection report which led to the issuance of the ISM deficiency.

7 FOLLOW-UP ACTION

7.1 Technical, operational and ISM Code deficiencies

7.1.1 The principles outlined in the *Procedures for port State control* with regard to reporting and rectification of technical or operational deficiencies, and detention and release of the ship are applicable.

- 7.1.2 If there are technical or operational deficiencies reported:
 - .1 which, whether detainable or non-detainable, do not show a failure, or lack of effectiveness, of the implementation of the ISM Code, no ISM deficiency should be reported in the PSC inspection report;
 - .2 of which at least one non-detainable deficiency indicates a failure, or lack of effectiveness, of the implementation of the ISM Code, a non-detainable ISM deficiency will be reported in the PSC inspection report with the requirement of corrective action within three months;
 - .3 which individually do not lead to a detention but collectively warrant the detention of the ship indicating a serious failure, or lack of effectiveness, of the implementation of the ISM Code, ISM deficiency will be reported in the PSC inspection report with the requirement that a safety management audit has to be carried out by the Administration or the RO before the ship may be released from its detention; and
 - .4 of which at least one detainable deficiency indicates a serious failure, or lack of effectiveness, of the implementation of the ISM Code, a detainable ISM deficiency will be reported in the PSC inspection report with the requirement that a safety management audit has to be carried out by the Administration or the RO before the ship may be released from detention.
 - *Note:* Where the PSCO considers that one or more technical and/or operational deficiencies are related to the ISM Code, this should be recorded as only one ISM deficiency.

7.1.3 The PSCO will verify the effectiveness of any corrective action as described in section 6.3.2. If examination of the areas in relation to an ISM deficiency with the requirement corrective action within three months is found not satisfactory, a new detainable ISM deficiency with the requirement that a safety management audit has to be carried out by the Administration or the RO will be raised. In this case the PSCO should apply the following procedure:

- .1 record one or more technical/operational deficiencies, detainable or not, in the same area(s) which led to the issuance of the previous ISM deficiency;
- .2 mark the deficiency or deficiencies "ISM-related" and add in the additional comments the following text: "This deficiency shows non-effective implementation of the ISM Code in the areas where the ISM deficiency or deficiencies were found during the PSC inspection on _____"; and
- .3 record a new detainable ISM deficiency with the requirement that a safety management audit has to be conducted by the Administration or the RO before the ship may be released from detention.

7.2 Deficiencies not warranting detention

Minor typing errors in the DOC, the Interim DOC, the SMC, or Interim SMC should be recorded in the PSC inspection report as a technical deficiency with the certificates and no ISM deficiency should be recorded.

7.3 Deficiencies warranting detention

The following are deficiencies which may warrant detention:

- .1 deficiencies of a technical and/or operational nature which individually or collectively provide objective evidence of a serious failure, or lack of effectiveness, of the implementation of the ISM Code;
- .2 there is no SMC, Interim SMC and/or copy of the DOC or Interim DOC on board the ship;
- .3 there is no valid SMC or Interim SMC on board;
- .4 the SMC intermediate verification is overdue;
- .5 the SMC has expired and there is no objective evidence of an extension issued by the Administration; or the SMC has been withdrawn by the Administration;
- .6 the DOC or Interim DOC has expired or been withdrawn;
- .7 the ship type as indicated on the SMC or Interim SMC is not listed on the DOC or Interim DOC;
- .8 evidence of the DOC annual verification is not available on board;
- .9 the certificate number on the copy of the DOC and the endorsement pages are not the same; and
- .10 the company name, the company address or the issuing Government authority on the DOC or Interim DOC is not the same as on the SMC or Interim SMC.

8 REPORTING

8.1 Technical and operational-related deficiencies

8.1.1 All technical and/or operational deficiencies should be recorded as an individual deficiency in the PSC inspection report according to the *Procedures for port State control*.

8.1.2 A technical deficiency with the defective item DOC/SMC or Interim DOC/SMC should be recorded in the PSC inspection report under the deficiency code addressing the DOC or SMC respectively.

8.2 ISM deficiency

Where the PSCO has considered the technical and/or operational deficiencies found and concluded these provide objective evidence of a failure, serious failure or lack of effectiveness of the implementation of the ISM Code, an ISM deficiency should be recorded in the PSC inspection report.

GUIDELINES FOR PORT STATE CONTROL RELATED TO LRIT

1 PURPOSE

These Guidelines are intended to provide basic guidance to port State control officers (PSCOs) to verify compliance with the requirements of SOLAS 1974 for Long-Range Identification and Tracking (LRIT).

2 APPLICATION

2.1 LRIT equipment is required by the provisions of SOLAS 1974 regulation V/19-1, and the *Revised performance standards and functional requirements for the Long-Range Identification and Tracking of ships* (resolution MSC.263(84)), as amended, and requires all passenger ships, cargo ships (including high-speed craft) over 300 gross tonnage and mobile offshore drilling units (MODUs) to send LRIT position information at least every six hours. Ships fitted with an automatic identification system (AIS) and operated exclusively within sea area A1 are not required to comply with LRIT. Sea area A1 is defined by SOLAS 1974 regulation IV/2.1.12 as "an area within the radiotelephone coverage of at least one VHF coast station in which continuous DSC alerting is available, as may be defined by a Contracting Government".

2.2 SOLAS Contracting Governments are expected to maintain an LRIT data centre, either on a national basis, or on a regional or cooperative basis with other flag States, and notify IMO of it. In turn, these LRIT data centres will forward, upon request, LRIT information from ships entitled to fly their flags, to other SOLAS Contracting Governments through the International LRIT Data Exchange. Port States are entitled to request LRIT information from foreign ships that have indicated their intention to enter a port, port facility or place under its jurisdiction.

2.3 In most cases a stand-alone Inmarsat C or Inmarsat mini-C terminal used for GMDSS or ship security alert system will function as the LRIT terminal, but other equipment may be employed for the LRIT function (example, Inmarsat D+ or Iridium).

3 INSPECTION OF SHIPS REQUIRED TO CARRY LRIT EQUIPMENT

3.1 Initial inspection

3.1.1 The PSCO should first establish the sea area the ship is certified to operate in. This verification should ensure that the ship is subject to the LRIT regulation in relation to its ship type and tonnage. After the certificate check, the PSCO should verify that:

- .1 the Record of Equipment (Form E, P or C) indicates LRIT as required, if applicable; and*
- .2 the equipment identified by the ship's representative as the designated LRIT terminal is switched on.[†]

^{*} A Record of Equipment is required for cargo ships greater than 500 gross tonnage and passenger ships.

[†] In exceptional circumstances and for the shortest duration possible, LRIT is capable of being switched off or may transmit less frequently (SOLAS 1974 regulation V/19-1.7.2 and resolution MSC.263(84), paragraph 4.4.1).

- 3.1.2 In case of recent transfer of flag, the PSCO may further ensure that:
 - .1 a conformance test report has been re-issued if the new flag State does not recognize the issuing body of the existing conformance test report; or
 - .2 a new conformance test has been carried out by the application service provider (ASP) on behalf of the Administration before issuance of a new test report and certificate.

3.2 Clear grounds

Conditions which may warrant a more detailed inspection of equipment used for LRIT may comprise the following:

- .1 defective main or emergency source of energy;
- .2 information or indication that LRIT equipment is not functioning properly;
- .3 ship does not hold conformance test report; and
- .4 the "record of navigational activities" indicates that the LRIT installation has been switched off and that this has not been reported to the flag Administration as required by SOLAS 1974 regulation V/19-1.7.2.

3.3 More detailed inspection

3.3.1 In case of doubt or reports of malfunctioning of the LRIT installation, the flag Administration may be contacted to determine if the ship's LRIT information has been reliably relayed to the LRIT data centre.

3.3.2 If any issues are identified at the initial inspection, a more detailed inspection of equipment used for LRIT may comprise the following:

- .1 verification of the power supply, which should be connected to the main source of energy and the emergency source of energy – there is no requirement for an uninterrupted power source; if LRIT is part of the GMDSS radio-installation, the power supply should conform to GMDSS regulations;
- .2 inspection of the "record of navigational activities" log to establish if and when the installation has been switched off and if this has been reported to the flag Administration (SOLAS 1974 regulation V/19-1.7.2 and resolution MSC.263(84), paragraph 4.4.1); and
- .3 ensuring that any conformance test report is issued on behalf of the flag State, even by itself or by an authorized application service provider (see MSC.1/Circ.1377/Rev.11 and updated versions as shown in GISIS), available for a ship that has an LRIT installation.

4 Deficiencies warranting detention

4.1 A PSCO should use professional judgement to determine whether to detain the ship until any noted deficiencies are corrected or to permit a vessel to sail with deficiencies.*

4.2 In order to assist the PSCO in the use of these Guidelines, the following deficiencies should be considered to be of such nature that they may warrant the detention of a ship:

- .1 absence of a valid LRIT conformance test report; and
- .2 the master or the responsible officer is not familiar with essential shipboard operational procedures relating to LRIT.

4.3 Taking into account the guidance found in the *Guidance on the implementation of the LRIT system* (MSC.1/Circ.1298), PSCOs are also advised that ships should not be detained if the LRIT installation on board works but the shore-side installation or organization is not able to receive, relay or process the information.

4.4 PSCOs are advised that a flag State may issue a short-term certificate; this could happen if, following a successful inspection for the issuance of a conformance test report, the ASP has not been able to issue a document yet, or if the ASP is not able to perform a conformance test in due time upon the request of the shipowner.

^{*} SOLAS 1974 regulation V/16.2: "while all reasonable steps shall be taken to maintain the equipment required by this chapter in efficient working order, malfunctions of that equipment shall not be considered as making the ship unseaworthy or as a reason for delaying the ship in ports where repair facilities are not readily available, provided suitable arrangements are made by the master to take the inoperative equipment or unavailable information into account in planning and executing a safe voyage to a port where repairs can take place."

GUIDELINES FOR PORT STATE CONTROL UNDER TONNAGE 1969

1 The International Convention on Tonnage Measurement of Ships, 1969 (TONNAGE 1969), which came into force on 18 July 1982, applies to:

- .1 new ships, i.e. ships the keels of which were laid on or after 18 July 1982; and
- .2 existing ships, i.e. ships the keels of which were laid before 18 July 1982, as from 18 July 1994,

except that for the purpose of application of SOLAS 1974, MARPOL and STCW 1978, the following interim schemes indicated in paragraph 2 may apply.

2 In accordance with the interim schemes adopted by the Organization,* the Administration may, at the request of the shipowner, use the gross tonnage determined in accordance with national rules prior to the coming into force of TONNAGE 1969 for the following ships:

- .1 for the purpose of SOLAS 1974:
 - .1 ships the keels of which were laid before 1 January 1986;
 - .2 in respect of SOLAS 1974 regulation IV/3, ships the keels of which were laid on or after 1 January 1986 but before 18 July 1994; and
 - .3 cargo ships of less than 1,600 tons gross tonnage (as determined under the national tonnage rules) the keels of which were laid on or after 1 January 1986 but before 18 July 1994; and
- .2 for the purpose of MARPOL, ships of less than 400 tons gross tonnage (as determined under the national tonnage rules) the keels of which were laid before 18 July 1994.

3 For ships to which the above interim schemes apply, a statement to the effect that the gross tonnage has been measured in accordance with the national tonnage rules should be included in the "REMARKS" column of the International Tonnage Certificate and in the footnote to the figure of the gross tonnage in the relevant SOLAS 1974 and MARPOL certificates.

4 The port State control officer (PSCO) should take the following actions as appropriate when deficiencies are found in relation to TONNAGE 1969:

- .1 if a ship does not hold a valid International Tonnage Certificate, the ship loses all privileges of TONNAGE 1969, and the flag State should be informed without delay;
- .2 if the required remarks and footnote are not included in the relevant certificates on ships to which the interim schemes apply, this deficiency should be notified to the master; and

^{*} Resolutions A.494(XII) in respect of SOLAS 1974, A.540(13) in respect of STCW 78, and A.541(13) in respect of MARPOL.

.3 if the main characteristics of the ship differ from those entered on the International Tonnage Certificate, so as to lead to an increase in the gross tonnage or net tonnage, the flag State should be informed without delay.

5 The control provisions of article 12 of TONNAGE 1969 do not include the provision for detention of a ship holding a valid International Tonnage Certificate.

GUIDELINES FOR PORT STATE CONTROL OFFICERS ON CERTIFICATION OF SEAFARERS, MANNING AND HOURS OF REST

1 GENERAL

The International Convention for the Safety of Life at Sea (SOLAS 1974) was adopted in 1974 and entered into force in 1980. Similarly, the International Convention on Standards of Training, Certification and Watchkeeping for Seafarers (STCW 1978) was adopted in 1978 and entered into force in 1984. Both have been amended several times since their entry into force.

2 GOALS AND PURPOSE

These Guidelines are intended to provide guidance for a harmonized approach to port State control (PSC) inspections in compliance with SOLAS 1974 regulation V/14 (manning) and STCW 1978 regulation I/2 (seafarer certification) and chapter VIII (hours of rest).

3 APPLICATION

3.1 SOLAS 1974 regulation V/14.2 only applies to ships covered by chapter I of SOLAS 1974. STCW 1978, as amended, applies to seafarers serving on board seagoing ships. The STCW Code is divided into a mandatory part A and a non-mandatory part B. Part B of the STCW Code is not applicable during the inspection.

3.2 All passenger ships regardless of size and all other ships of 500 gross tonnage or more should have a minimum safe manning document or equivalent on board issued by the flag State.

3.3 Any new or single deficiency which is either a deficiency related to SOLAS 1974, STCW 1978 or other IMO conventions, should preferably be registered with these conventions' references.

4 RELEVANT DOCUMENTATION

The documentation required for the inspection referred to in these Guidelines consists of:

Seafarer certification

- .1 certificate of competency;
- .2 certificate of proficiency;
- .3 endorsement attesting the recognition of a certificate (flag State endorsement);
- .4 documentary evidence (passenger ships only);
- .5 medical certificate;

Manning

.6 minimum safe manning document;

.7 muster list;

Hours of rest

- .8 table of ship working arrangements and/or watch schedule; and
- .9 records of daily hours of rest.

5 DEFINITIONS AND ABBREVIATIONS

5.1 Certificate of Competency means a certificate issued and endorsed for masters, officers and Global Maritime Distress and Safety System (GMDSS) radio operators in accordance with the provisions of chapters II, III, IV or VII of STCW 1978 and entitling the lawful holder thereof to serve in the capacity and perform the functions involved at the level of responsibility specified therein.

5.2 Certificate of Proficiency means a certificate, other than a certificate of competency issued to a seafarer, stating that the relevant requirements of training, competencies or seagoing service in STCW 1978 have been met.

5.3 Documentary evidence means documentation, other than a Certificate of Competency or Certificate of Proficiency, used to establish that the relevant requirements of STCW 1978, as amended, have been met. The only documentary evidence required under STCW 1978, as amended, is issued to personnel meeting the mandatory minimum requirements for the training and qualifications of masters, officers, ratings and other personnel on passenger ships (regulation V/2).

5.4 The following abbreviations have been used:

- .1 CoC (Certificate of Competency);
- .2 CoP (Certificate of Proficiency); and
- .3 MSMD (minimum safe manning document).

6 INSPECTION OF SHIP

6.1 **Pre-boarding preparation**

6.1.1 Taking into account the type, size, engine power and other particulars of the ship, the port State control officer (PSCO) should be aware of the relevant requirements of SOLAS 1974 regulation V/14 and STCW 1978.

6.1.2 The PSCO should be aware that resolutions are non-mandatory documents and not applicable during a PSC inspection.

6.1.3 The PSCO should also identify if the flag State is a Party to STCW 1978, as amended. If the flag State is not a Party to the Convention or is a Party but not listed in MSC.1/Circ.1163, as amended, a more detailed inspection should be carried out.

6.2 Initial inspection

Seafarer certificates and documents

6.2.1 The PSCO should examine the applicable documents, found in section 4.

6.2.2 The inspection should be limited to verification that seafarers serving on board, who are required to be certificated, hold the appropriate CoC, CoP and documentary evidence issued in accordance with chapters II, III, IV, V, VI and VII of STCW 1978, as amended, as well as their relevant flag State endorsement, valid dispensation, or documentary proof that an application for an endorsement has been submitted to the flag State Administration, where applicable. These documents are evidence of having successfully completed all required training and that the required standard of competence has been achieved.

6.2.3 During the verification of the seafarers' certificates and documents, the PSCO should confirm that they are applicable to the ship's characteristics, operation and their position on board.

6.2.4 In accordance with the provision of article VI, paragraph 2 of STCW 1978, certificates for masters and officers should be endorsed by the issuing Administration in the form prescribed in regulation I/2 of the annex to the Convention.

6.2.5 The certificates may be issued as one certificate with the required endorsement incorporated. If so incorporated, the form used should be that set forth in section A-I/2, paragraph 1 of the STCW Code.

6.2.6 The endorsement may also be issued as a separate document. If so, the form used should be that set out in section A-I/2, paragraph 2 of the STCW Code.

6.2.7 However, Administrations may use a format for certificates and endorsements different from those given in section A-I/2 of the STCW Code, provided that, at a minimum, the required information is provided in Roman characters and Arabic figures. Permitted variations to the format are set out in section A-I/2, paragraph 4 of the STCW Code.

6.2.8 Certificates and endorsements issued as separate documents should each be assigned a unique number, except that endorsements attesting the issuance of a certificate may be assigned the same number as the certificate concerned, provided that number is unique.

6.2.9 Certificates and endorsements issued as separate documents should include a date of expiry. The date of expiry on an endorsement issued as a separate document should not exceed 5 years from the date of issue and may never exceed the date of expiry on the certificate.

6.2.10 A CoP issued to a master or an officer in accordance with regulation V/1-1 or V/1-2, as well as a CoC that has been issued by a State other than the flag State of the ship in which the seafarer is engaged, is required to be recognized by the ship's flag State. If the PSCO identifies that the flag State has recognized a CoC or CoP from a Party not listed in MSC.1/Circ.1163, as amended, clarification should be sought from the flag Administration. According to regulation I/10, paragraph 4 of STCW 1978, certificates issued by or under the authority of a non-Party shall not be recognized by the ship's flag State Administration.

6.2.11 An Administration which recognizes under regulation I/10 a CoC or CoP issued to masters and officers should endorse that certificate to attest to its recognition. The form of the endorsement should be that found in section A-I/2, paragraph 3 of the STCW Code.

6.2.12 Incorrect wording or missing information may be a cause for suspicion regarding fraudulent certificates or endorsements.

6.2.13 Endorsements attesting to the recognition of a certificate should each be assigned a unique number; however, they may be assigned the same number as the certificate concerned, provided that number is unique.

6.2.14 Endorsements attesting to the recognition of a certificate should include a date of expiry. The date of expiry on an endorsement attesting to the recognition may never exceed the date of expiry on the certificate being recognized.

6.2.15 The capacity in which the holder of a certificate is authorized to serve should be identified in the form of endorsement in terms identical to those used in the applicable safe manning requirements of the Administration. This may result in slight variations of terminology between the original CoC and the endorsement to the recognition.

6.2.16 Seafarers must have their original CoC on board as well as any original endorsements to the recognition. An endorsement attesting the recognition of a certificate should not entitle a seafarer to serve in a higher capacity than the original CoC.

6.2.17 If circumstances require it, a flag State Administration may permit a seafarer to serve for a period not exceeding three months on ships entitled to fly its flag while holding a valid CoC issued by another party and valid for service on that party's ships. If such a situation exists, documentary proof must be readily available that an application for endorsement has been made to the Administration of the flag State. This is often referred to as the confirmation of receipt of application (CRA). This provision allows Administrations to permit seafarers to serve on their ships while the application for recognition is being processed.

6.2.18 If an endorsement to attest recognition or certificate of competency has expired or has not been issued or documentary proof of application for endorsement is not readily available, the PSCO should consider whether or not the ship can comply with STCW 1978 regulation I/4.1.2 regarding the numbers and certificates on board being in compliance with the applicable safe manning requirements of the flag State. This may be considered a deficiency in accordance with regulation I/4.2.4 and rectified before departure or detention may be applied. The officer carrying out the control should forthwith inform, in writing, the master of the ship and the Consul or, in his absence, the nearest diplomatic representative or the maritime authority of the State whose flag the ship is entitled to fly, so that appropriate action may be taken.

6.2.19 In cases of suspected intoxication of masters, officers and/or other seafarers while performing designated safety, security and marine environmental protection duties, the appropriate authorities of the port and flag State should be notified in accordance with chapters 3 and 4 of the *Procedures for port State control*.

6.2.20 Seafarers should have a valid medical certificate and have completed applicable familiarization on board the ship. If such crew members are assigned to any designated safety, security or pollution prevention duties, they must be trained and qualified for such duties in accordance with the applicable chapter of the STCW Code.

6.2.21 In accordance with section A-VI/1, paragraph 5 of the STCW Code, the flag State Administration may exempt the seafarers engaged on ships other than passenger ships of more than 500 gross tonnage on international voyages and tankers from some of the requirements of that section.

Manning

- 6.2.22 The PSCO should examine the applicable documents, found in section 4.
- 6.2.23 The guiding principles for port State control of the manning of a foreign ship should be:
 - .1 verification that the numbers and certificates of the seafarers serving on board are in conformity with the applicable safe manning requirements of the flag State; and
 - .2 verification that the vessel and its personnel conform to the international provisions as laid down in SOLAS 1974 and STCW 1978.

6.2.24 If a ship is manned in accordance with an MSMD or equivalent document issued by the flag State, the PSCO should accept that the ship is safely manned unless the document has clearly been issued without regard to the principles contained in the relevant instruments, in which case the PSCO should consult the flag State Administration.

6.2.25 If the flag State Administration has not issued a safe manning document or equivalent due to the ship's size the PSCO should examine the CoC, CoP and their relevant flag State endorsement for the crew and compare with the requirements of STCW 1978. Regarding the number of seafarers, the PSCO should then use his or her professional judgement, taking into account chapter VIII of STCW 1978 and the STCW Code and the duration and area of the next voyage, to determine if it can be undertaken safely. The PSCO should note the number of seafarers on board during the previous voyage as another indicator of standard manning levels for the ship. The PSCO should consult the flag State Administration if additional information is necessary.

6.2.26 If an endorsement to attest recognition has expired or has not been issued or documentary proof of application for endorsement (CRA) is not readily available, the PSCO should consider whether the ship can comply with the applicable safe manning requirements of the flag State Administration. In cases where the PSCO finds that additional information is necessary, the flag State Administration should be consulted.

6.2.27 If the flag State does not respond to the request, this should be considered as clear grounds for a more detailed inspection to ensure that the number and composition of the crew are in accordance with the principles laid down in paragraph 6.2.23 above. The ship should only be allowed to proceed to sea if it is safe to do so, taking into account the criteria for detention indicated in section 7.3. In any such case, the minimum standards to be applied should be no more stringent than those applied to ships flying the flag of the port State.

Hours of rest

6.2.28 All persons who are assigned duty as officer in charge of a watch or as a rating forming part of a watch and those whose duties involve designated safety, security and environmental protection duties shall be provided with a rest period of not less than:

- .1 a minimum of 10 hours of rest in any 24-hour period; and
- .2 77 hours in any 7-day period.

6.2.29 The hours of rest may be divided into no more than two periods, one of which shall be at least 6 hours in length, and the intervals between consecutive periods of rest shall not exceed 14 hours.

6.2.30 The PSCO should examine the applicable documents, found in section 4, specifically the watch schedule and the records of daily hours of rest. The PSCO may inspect the seafarer's personal copy of his or her records pertaining to the hours of rest being held by the seafarer on board in order to verify that the records are accurate.

6.2.31 The watch schedule shall be in a standardized format,^{*} easily accessible to the crew and posted in the working language or languages of the ship and in English.

6.2.32 Daily hours of rest shall be maintained in a standardized format,^{*} in the working language or languages of the ship and in English.

6.2.33 The PSCO should consider that seafarers who are on call, such as when a machinery space is unattended, are to be provided with an adequate compensatory rest period if the normal period is disturbed by call-outs to work.

6.2.34 While assessing hours of rest, the PSCO should take into account any emergency conditions encountered which required a seafarer to perform additional hours of work for the immediate safety of the ship. In such cases, the master should be consulted for an explanation of the events and how impacted seafarers were provided with an adequate period of rest.

6.2.35 Flag State Administrations may provide exceptions to the requirements of paragraphs 6.2.28.2 and 6.2.29 above for no more than two consecutive weeks provided that the rest period for the seafarer is not less than 70 hours in any 7-day period.

6.3 Clear grounds

6.3.1 Clear grounds are defined in section 1.7.2 of the *Procedures for port State control*.

6.3.2 In addition to the general examples of clear grounds in section 2.4 of the Procedures, the specific occurrences below, as outlined in paragraph 1.3 of regulation I/4 of STCW 1978, are considered as factors leading to a more detailed inspection:

- .1 the ship has been involved in a collision, grounding or stranding; or
- .2 there has been a discharge of substances from the ship when under way, at anchor or at berth which is illegal under any international convention; or
- .3 the ship has been manoeuvred in an erratic or unsafe manner whereby routeing measures adopted by IMO or safe navigation practices and procedures have not been followed; or
- .4 the ship is otherwise being operated in such a manner as to pose a danger to persons, property, or the environment, or a compromise to security.

6.4 More detailed inspection

- 6.4.1 The PSCO should:
 - .1 verify that seafarers are sufficiently rested and otherwise fit for duty for the first watch at the commencement of the intended voyage and for subsequent relieving watches; this may be done by comparing records of daily hours of

^{*} The IMO/ILO Guidelines for the development of tables of seafarers' shipboard working arrangements and formats of records of seafarers' hours of work or hours of rest may be used.

rest with the requirements in STCW 1978 for an appropriate period, which should at least include, whenever possible, the seven-day period immediately prior to departure; the rest period must reflect actual hours worked;

- .2 verify a sufficient number of certificates from all departments to demonstrate that the vessel and the composition of the crew complies with the MSMD and requirements of STCW 1978; and
- .3 verify that navigational or engineering watch arrangements conform to the requirements specified for the ship in the MSMD by the flag State and the requirements of STCW 1978 regulation VIII/2 and STCW Code section A-VIII/2.

6.4.2 An assessment of seafarers can only be conducted by the port State if there are clear grounds for believing that the ability of the seafarers of the ship to maintain watchkeeping and security standards, as appropriate, as required by STCW 1978 is not being maintained because any of the situations mentioned in paragraphs 6.3.2.1 to 6.3.2.4 have occurred:

- .1 the assessment procedure provided in STCW 1978 regulation I/4, paragraph 1.3, should take the form of a verification that members of the crew who are required to be competent do in fact possess the necessary skills related to the occurrence;
- .2 it should be borne in mind when making this assessment that onboard procedures are relevant to the International Safety Management (ISM) Code and that the provisions of STCW 1978 are confined to the competence to safely execute those procedures;
- .3 control procedures under STCW 1978 should be confined to the standards of competence of the individual seafarers on board and their skills related to watchkeeping as defined in part A of the STCW Code. Onboard assessment of competency should commence with verification of the certificates of the seafarers;
- .4 notwithstanding verification of the certificate, the assessment under STCW 1978 regulation I/4, paragraph 1.3 can require the seafarer to demonstrate the related competency at the place of duty. Such demonstration may include verification that operational requirements in respect of watchkeeping standards have been met and that there is a proper response to emergency situations within the seafarer's level of competence;
- .5 in the assessment, only the methods for demonstrating competence together with the criteria for its evaluation and the scope of the standards given in part A of the STCW Code should be used. In cases where there is doubt about knowledge of operational use of equipment, the relevant officer or crew member should be asked to perform a functional test. Failure to perform a functional test could indicate the lack of familiarization or competency; and
- .6 assessment of competency related to security should be conducted for those seafarers with specific security duties only in case of clear grounds, as provided for in chapter XI-2 of SOLAS 1974, by the competent security authority. In all other cases, it should be confined to the verification of the certificates and/or endorsements of the seafarers.

7 FOLLOW-UP ACTION

7.1 Possible action

Possible action to be considered by the PSCO for the control in compliance with SOLAS 1974 or STCW 1978 may be dealt with in the following ways:

- .1 exercise of control with regard to the documentation concerning the ship; and
- .2 exercise of control with regard to the documentation for individual seafarers on board.

7.2 Possible deficiencies

The following is a non-exhaustive list of possible deficiencies:

Seafarers' documentation:

- .1 no CoC, CoP, flag State endorsements or proof that an application for an endorsement has been submitted (STCW 1978 regulations I/4.2.1 and I/10);
- .2 special training requirements: mandatory basic or advanced training or endorsement not presented;
- .3 no evidence of basic training, or other certificate of proficiency, if not included in a qualification certificate held (STCW 1978 regulations VI/1, VI/1.2, VI/3, VI/4 and VI/6); and
- .4 information or evidence that the master or crew is not familiar with essential shipboard operations relating to the safety of ships or the prevention of pollution, or that such operations have not been carried out;

Manning:

- .5 no MSMD or the manning (number or qualification) not in accordance with the MSMD (SOLAS 1974 regulation V/14 and STCW 1978 regulation I/4.2.2); and
- .6 unqualified person on duty (STCW 1978 regulation I/4.2.4);

Hours of rest:

- .7 watch schedule not posted or not being followed (STCW 1978 regulations I/4.2.3 and I/4.2.5 and STCW Code A-VIII/1.5);
- .8 the absence of a table of shipboard working arrangements or of records of rest of seafarers (STCW Code A-VIII/1.7);
- .9 the records of hours of rest are inaccurate or incomplete (STCW Code A-VIII/1.7); and

.10 the watchkeeper is receiving less than 10 hours rest in any 24-hour period (i.e. working in excess of 14 hours) or 77 hours rest in any 7-day period (STCW Code A-VIII/1).

7.3 Deficiencies that may warrant detention

7.3.1 Deficiencies which may be deemed to pose a danger to persons, property or the environment, as specified in paragraph 2 of regulation I/4 of STCW 1978, as amended:

- .1 failure of seafarers to hold a certificate, to have an appropriate certificate, to have a valid dispensation or to provide documentary proof that an application for an endorsement has been submitted to the Administration in accordance with regulation I/10, paragraph 5;
- .2 failure to comply with the applicable safe manning requirement of the Administration;
- .3 failure of navigational or engineering watch arrangements to conform to the requirements specified for the ship by the Administration;
- .4 absence in a watch of a person qualified to operate equipment essential to safe navigation, safety radiocommunications or the prevention of marine pollution; and
- .5 inability to provide, for the first watch at the commencement of a voyage and for subsequent relieving watches, persons who are sufficiently rested and otherwise fit for duty.

7.3.2 Failure to correct any of the deficiencies, in so far as it has been determined by the PSCO that they pose a danger to persons, property or the environment, shall be the only grounds under STCW 1978, as amended, on which a ship may be detained.

7.3.3 Examples of detainable deficiencies according to SOLAS 1974 and STCW 1978 are listed below:

Ship-related:

- .1 MSMD or equivalent not presented (SOLAS 1974 regulation V/14.2); and
- .2 records of daily hours of rest are not on board (STCW Code A-VIII/1.7); and

Seafarers' documentation:

- .3 not available or serious discrepancy in the CoC (STCW 1978 regulation I/4.2.1);
- .4 absence in watch of a radio operator (general/restricted GMDSS); certificates and endorsement not available (STCW 1978 regulations I/4.2.1, I/4.2.2, I/4.2.3, I/4.2.4 and II/1.2.1);
- .5 documentation for personnel with designated safety, security and marine environmental duties not available (STCW 1978 regulations I/4.2.1, I/4.2.2, I/4.2.3 and I/4.2.4);

- .6 expired certificates (STCW 1978 regulation I/4.2.1), and for medical certificates also refer to STCW 1978 regulations I/9.6 and I/9.7; and
- .7 evidence that a certificate has been fraudulently obtained or the holder of a certificate is not the person to whom that certificate was originally issued.

7.4 Actions to be considered

Ship-related

7.4.1 If the actual number of crew or composition does not conform to the manning document, the port State should request the flag State for advice as to whether or not the ship should be allowed to sail with the actual number of crew and composition of crew. Such a request and response should be by the most expedient means and either party may request the communication in writing. If the actual crew number or composition is not brought into compliance with the MSMD or the flag State does not advise that the ship may sail, the ship may be considered for detention after the criteria set out in section 7.3 have been taken into account.

- 7.4.2 Before detaining the ship the PSCO should consider the following:
 - .1 length and nature of the intended voyage or service;
 - .2 whether or not the deficiency poses a danger to ships, persons on board or the environment;
 - .3 whether or not appropriate rest periods of the crew can be observed;
 - .4 size and type of ship and equipment provided; and
 - .5 nature of cargo.

Deficiency-related

7.4.3 When the manning is not in accordance with the MSMD and no flag State endorsements or no "documentary proof of application" can be presented, the port State should consult the flag State whenever possible, taking into account time differences or other conditions. However, if it is not possible to establish contact with the flag State, the port State should forthwith inform, in writing, the master of the ship and the Consul or, in their absence, the nearest diplomatic representative or the maritime authority of the State whose flag the ship is entitled to fly, so that appropriate action may be taken.

7.4.4 In cases where an unqualified seafarer has been on duty and/or the watch schedule has not been followed, the flag State should be informed and this could be considered as an ISM deficiency.

7.4.5 In cases where there is a seafarer on duty who is not qualified to carry out an operation, that particular operation should be stopped immediately.

8 NOTE ON REPORTING DEFICIENCIES

The PSCO should be aware that, in addition to SOLAS 1974 and STCW 1978, there may be other applicable international instruments. The PSCO should decide which one is the most appropriate.

Annex

Table B-I/2

List of certificates or documentary evidence required under STCW 1978

Refer to table B-I/2 of the STCW Code, as amended

LIST OF CERTIFICATES AND DOCUMENTS

PART A

List of certificates and documents which to the extent applicable should be checked as a minimum during the inspection referred to in paragraph 2.2.3 (as appropriate):

- 1 International Tonnage Certificate (TONNAGE 1969 article 7);
- 2 Reports of previous port State control inspections;
- 3 Passenger Ship Safety Certificate (SOLAS 1974 regulation I/12);
- 4 Cargo Ship Safety Construction Certificate (SOLAS 1974 regulation I/12);
- 5 Cargo Ship Safety Equipment Certificate (SOLAS 1974 regulation I/12);
- 6 Cargo Ship Safety Radio Certificate (SOLAS 1974 regulation I/12);
- 7 Cargo Ship Safety Certificate (SOLAS 1974 regulation I/12);
- 8 Exemption Certificate (SOLAS 1974 regulation I/12);
- 9 Minimum safe manning document (SOLAS 1974 regulation V/14.2);
- 10 International Load Line Certificate (1966) (LL 1966/LL PROT 1988 article 16.1);
- 11 International Load Line Exemption Certificate (LL 1966/LL PROT 1988 article 16.2);
- 12 International Oil Pollution Prevention Certificate (MARPOL Annex I regulation 7.1);
- 13 International Pollution Prevention Certificate for the Carriage of Noxious Liquid Substances in Bulk (NLS) (MARPOL Annex II regulation 9.1);
- 14 International Sewage Pollution Prevention Certificate (MARPOL Annex IV regulation 5.1 and MEPC.1/Circ.408);
- 15 International Air Pollution Prevention Certificate (MARPOL Annex VI regulation 6.1);
- 16 International Energy Efficiency Certificate (MARPOL Annex VI regulation 6);
- 17 International Ballast Water Management Certificate (BWM 2004 article 9.1(a) and regulation E-2);
- 18 International Anti-Fouling System Certificate (AFS 2001 annex 4 regulation 2);
- 19 Declaration on AFS (AFS 2001 annex 4 regulation 5);
- 20 International Ship Security Certificate or Interim International Ship Security Certificate (ISPS Code part A/19 and appendices);

- 21 Certificates for masters, officers or ratings (STCW 1978 article VI and regulation I/2, and STCW Code section A-I/2);
- 22 Copy of Document of Compliance or a copy of the Interim Document of Compliance (SOLAS 1974 regulation IX/4.2 and ISM Code paragraphs 13 and 14);
- 23 Safety Management Certificate or an Interim Safety Management Certificate (SOLAS 1974 regulation IX/4.3 and ISM Code paragraphs 13 and 14);
- 24 International Certificate of Fitness for the Carriage of Liquefied Gases in Bulk, or the Certificate of Fitness for the Carriage of Liquefied Gases in Bulk, whichever is appropriate (IGC Code section 1.4 or GC Code section 1.6);
- 25 International Certificate of Fitness for the Carriage of Dangerous Chemicals in Bulk, or the Certificate of Fitness for the Carriage of Dangerous Chemicals in Bulk, whichever is appropriate (IBC Code section 1.5 or BCH Code section 1.6);
- 26 International Certificate of Fitness for the Carriage of INF Cargo (SOLAS 1974 regulation VII/16 and INF Code section 1.3);
- 27 Certificate of insurance or other financial security in respect of civil liability for oil pollution damage (CLC 69/92 article VII.2);
- 28 Certificate of insurance or other financial security in respect of civil liability for bunker oil pollution damage (BUNKERS 2001 article 7.2);
- 29 Certificate of Insurance or other Financial Security in respect of Liability for the Removal of Wrecks (Nairobi WRC 2007 article 12);
- 30 High-Speed Craft Safety Certificate and Permit to Operate High-Speed Craft (SOLAS 1974 regulation X/3.2 and 1994/2000 HSC Code paragraph 1.8.1 and section 1.9);
- 31 Document of Compliance with the special requirements for ships carrying dangerous goods (SOLAS 1974 regulation II-2/19.4);
- 32 Document of authorization for the carriage of grain and grain loading manual (SOLAS 1974 regulation VI/9 and Grain Code section 3);
- 33 Condition Assessment Scheme (CAS) Statement of Compliance, CAS Final Report and Review Record (MARPOL Annex I regulations 20 and 21; resolution MEPC.94(46), as amended by resolutions MEPC.99(48), MEPC.112(50), MEPC.131(53), MEPC.155(55) and MEPC.236(65));
- 34 Continuous Synopsis Record (SOLAS 1974 regulation XI-1/5);
- 35 Oil Record Book, parts I and II (MARPOL Annex I regulations 17 and 36;
- 36 Cargo Record Book (MARPOL Annex II regulation 15);
- 37 Garbage Record Book (MARPOL Annex V regulation 10);
- 38 Garbage Management Plan (MARPOL Annex V regulation 10 and resolution MEPC.220(63));

- 39 Logbook and the recordings of the tier and on/off status of marine diesel engines (MARPOL Annex VI regulation 13.5.3);
- 40 Logbook for fuel oil changeover (MARPOL Annex VI regulation 14.6);
- 41 Ozone-depleting Substances Record Book (MARPOL Annex VI regulation 12.6);
- 42 Ballast Water Record Book (BWM 2004 article9.1 (b) and regulation B-2);
- 43 Fixed gas fire-extinguishing systems cargo spaces Exemption Certificate and any list of cargoes (SOLAS 1974 regulation II-2/10.7.1.4);
- 44 Dangerous goods manifest or stowage plan (SOLAS 1974 regulations VII/4 and VII/7-2 and MARPOL Annex III regulation 5);
- 45 For oil tankers, the record of oil discharge monitoring and control system for the last ballast voyage (MARPOL Annex I regulation 31.2);
- 46 Search and rescue cooperation plan for passenger ships trading on fixed routes (SOLAS 1974 regulation V/7.3);
- 47 For passenger ships, List of operational limitations (SOLAS 1974 regulation V/30.2);
- 48 Nautical charts and nautical publications (SOLAS 1974 regulations V/19.2.1.4 and V/27);
- 49 Records of hours of rest and table of shipboard working arrangements (STCW Code section A-VIII/1.5 and 1.7, ILO Convention No.180 articles 5.7 and 8.1 and MLC 2006 Standards A.2.3.10 and A.2.3.12); and
- 50 Unattended machinery spaces (UMS) evidence (SOLAS 1974 regulation II-I/46.3).

Part B

List of other certificates and documents which to the extent applicable are required to be on board (as appropriate):

- 1 Construction drawings (SOLAS 1974 regulation II-1/3-7);
- 2 Ship Construction File (SOLAS 1974 regulation II-1/3-10);
- 3 Manoeuvring booklet and information (SOLAS 1974 regulation II-1/28);
- 4 Stability information (SOLAS 1974 regulations II-1/5 and II-1/5-1, and LL 1966/LL PROT 1988 regulation 10);
- 5 Subdivision and stability information (MARPOL Annex I regulation 28);
- 6 Damage control plans and booklets (SOLAS 1974 regulation II-1/19 and MSC.1/Circ.1245, as amended);
- 7 Ship Structure Access Manual (SOLAS 1974 regulation II-1/3-6);

- 8 Enhanced survey report files (in case of bulk carriers or oil tankers) (SOLAS 1974 regulation XI-1/2 and 2011 ESP Code paragraphs 6.2 and 6.3 of annex A, part A and part B, and annex B, part A and part B);
- 9 Cargo Securing Manual (SOLAS 1974 regulation VI/5.6 and VII/5 and MSC.1/Circ.1353/Rev.1);
- 10 Bulk carrier booklet (SOLAS 1974 regulations VI/7.2 and XII/8 and BLU Code);
- 11 Loading/unloading plan for bulk cargoes (SOLAS 1974 regulation VI/7.3);
- 12 Cargo information (SOLAS 1974 regulations VI/2 and XII/10 and MSC/Circ.663);
- 13 Fire control plan/booklet (SOLAS 1974 regulations II-2/15.2.4 and II-2/15.3.2);
- 14 Fire safety operational booklet (SOLAS 1974 regulation II-2/16.2);
- 15 Fire safety training manual (SOLAS 1974 regulation II-2/15.2.3);
- 16 Training manual (SOLAS 1974 regulation III/35);
- 17 Onboard training, drills and maintenance records (SOLAS 1974 regulations II-2/15.2.2.5, III/19.3, III/19.5, III/20.6 and III/20.7);
- 18 Ship-specific plans and procedures for recovery of persons from the water (SOLAS 1974 regulation III/17-1, resolution MSC.346(91) and MSC.1/Circ.1447);
- 19 Decision support system for masters (Passenger ships) (SOLAS 1974 regulation III/29);
- 20 International Code of Signals and a copy of Volume III of IAMSAR Manual (SOLAS 1974 regulation V/21);
- 21 Records of navigational activities (SOLAS 1974 regulations V/26 and V/28.1);
- 22 Ship Security Plan and associated records (SOLAS 1974 regulation XI-2/9 and ISPS Code part A/9 and 10);
- 23 Engine International Air Pollution Prevention Certificate (NO_X Technical Code 2008 paragraph 2.1.1.1);
- 24 EEDI Technical File (MARPOL Annex VI regulation 20);
- 25 Technical Files (NO_X Technical Code 2008 paragraph 2.3.4);
- 26 Record Book of Engine Parameters (NO_X Technical Code paragraph 2.3.7);
- 27 Type approval certificate of incinerator (MARPOL Annex VI regulation 16.6);
- 28 Manufacturer's operating manual for incinerators (MARPOL Annex VI regulation 16.7);
- 29 Fuel oil changeover procedure for fuel oil changeover (MARPOL Annex VI regulation 14.6);

- 30 Bunker delivery notes and representative sample (MARPOL Annex VI regulations 18.6 and 18.8.1);
- 31 Shipboard Oil Pollution Emergency Plan (SOPEP) (MARPOL Annex I regulation 37.1 and resolution MEPC.54(32), as amended by resolution MEPC.86(44));
- 32 Shipboard Marine Pollution Emergency Plan for Noxious Liquid Substances (MARPOL Annex II regulation 17);
- 33 Ship Energy Efficiency Management Plan (MARPOL Annex VI regulation 22, MEPC.1/Circ.795);
- 34 STS operation plan and records of STS operations (MARPOL Annex I regulation 41);
- 35 Procedures and Arrangements Manual (chemical tankers) (MARPOL Annex II regulation 14.1; resolution MEPC.18(22), as amended by resolution MEPC.62(35));
- 36 VOC Management Plan (MARPOL Annex VI regulation 15.6);
- 37 Ballast Water Management Plan (BWM 2004 regulation B-1 and resolution MEPC.127(53), as amended);
- 38 LRIT conformance test report (SOLAS 1974 regulation V/19-1.6 and MSC.1/Circ.1307);
- 39 Copy of the certificate of compliance issued by the testing facility, stating the date of compliance and the applicable performance standards of VDR (voyage data recorder) (SOLAS 1974 regulation V/18.8);
- 40 AIS test report (SOLAS 1974 regulation V/18.9 and MSC.1/Circ.1252);
- 41 Noise survey report (SOLAS 1974 regulation II-1/3-12);
- 42 Oil discharge monitoring and control (ODMC) operational manual (MARPOL Annex I regulation 31; resolution A.496(XII); resolution A.586(14), as amended by resolution MEPC.24(22); and resolution MEPC.108(49), as amended by resolution MEPC.240(65));
- 43 Crude Oil Washing Operation and Equipment Manual (MARPOL Annex I regulation 35 and resolution MEPC.81(43));
- 44 Material Safety Data Sheets (MSDS) (SOLAS 1974 regulation VI/5-1 and resolution MSC.286(86));
- 45 Record of AFS (AFS 2001 annex 4 regulation 2);
- 46 Coating Technical File (SOLAS 1974 regulation II-1/3-2); and
- 47 Maintenance plans (SOLAS 1974 regulations II-2/14.2.2, II-2/14.3 and II-2/14.4).

For reference:

- 1 Certificate of Registry or other document of nationality (UNCLOS article 91);
- 2 Certificates as to the ship's hull strength and machinery installations issued by the classification society in question (only to be required if the ship maintains its class with a classification society);
- 3 Cargo Gear Record Book (ILO Convention No.32 article 9.2(4) and ILO Convention No.152 article 25);
- 4 Certificates loading and unloading equipment (ILO Convention No.134 article 4.3(e) and ILO Convention No.32 article 9(4));
- 5 Medical certificates (ILO Convention No.73 or MLC 2006 Standard A1.2);
- 6 Records of hours of work or rest of seafarers (ILO Convention No.180 part II article 8.1 or MLC 2006 Standard A.2.3.12);
- 7 Maritime Labour Certificate (MLC 2006 regulation 5.1.3);
- 8 Declaration of Maritime Labour compliance on board (parts I and II) (MLC 2006 regulation 5.1.3);
- 9 Seafarers' employment agreements (MLC 2006 Standard A 2.1);
- 10 Certificate of insurance or financial security for repatriation of seafarers (MLC 2006, regulation 2.5); and
- 11 Certificate of insurance or financial security for shipowners' liability (MLC 2006 regulation 4.2).

REPORT OF INSPECTION IN ACCORDANCE WITH PROCEDURES FOR PORT STATE CONTROL^{*}

FORM A

(Reporting authority) (Address) (Telephone) (Telefax)				Copy to:	Master Head office PSCO
(Em	iai)	lf shij	o is detair Flag St IMO Recogi	ned, copy [.] ate nized orga	to: nization, if applicable
1	Name of reporting authority		2 Nam	ne of ship .	
3	Flag of ship		4 Type	e of ship	
5	Call sign		6 IMO	number	
7	Gross tonnage		8 Dea	dweight (w	/here applicable)
9	Year of build		10 Date	e of inspec	tion
11	Place of inspection		12 Clas	sification s	society
13	13 Date of release from detention ^{**}				
14	Particulars of ISM company (d	etails or IMO C	Company	Number)**	
15	Relevant certificate(s)**				
	a) Title	b) Issuing a	uthority	c)	Dates of issue and expiry
1					
2					
3					
4					
5					
6					
7					
8					
9					
10					
11					
12					

d) Information on last intermediate or annual survey**

	Date	Surveying authority		Place	
1					
2					
3					
4					
5					
6					
7					
8					
9					
10					
11					
12					
16 D	eficiencies	No	Yes (see atta	ached FORM B)	
17 Pe	enalty imposed	No	Yes	Amount:	
18 SI	nip detained	No	Yes ***		
19 Supporting documentation		No	Yes (see anr	nex)	
Issuing office			Name		
			(duly auth	authority)	
Telep	hone				
Telefax			Signature		

This report must be retained on board for a period of two years and must be available for consultation by port State control officers at all times.

^{*} This inspection report has been issued solely for the purposes of informing the master and other port States that an inspection by the port State, mentioned in the heading, has taken place. This inspection report cannot be construed as a seaworthiness certificate in excess of the certificate the ship is required to carry.

^{**} To be completed in the event of a detention.

^{***} Masters, shipowners and/or operators are advised that detailed information on a detention may be subject to future publication.

REPORT OF INSPECTION IN ACCORDANCE WITH PROCEDURES FOR PORT STATE CONTROL

FORM B

(Reporting authority) (Address) (Telephone) (Telefax) (Email)	Copy to: Master Head office PSCO
lf sh	hip is detained, copy to: Flag State IMO Recognized organization, if applicable
2 Name of ship	6 IMO number
10 Date of inspection	11 Place of inspection

20 Na	ature of deficiency ¹	Convention ²	21 Action taken ³	22 ISM-related
•				
-				

Signature

¹ This inspection was not a full survey and deficiencies listed may not be exhaustive. In the event of a detention, it is recommended that a full survey is carried out and all deficiencies are rectified before an application for re-inspection is made.

² To be completed in the event of a detention.

³ Actions taken include: ship detained/released, flag State informed, classification society informed, next port informed.

REPORT OF DEFICIENCIES NOT FULLY RECTIFIED OR ONLY PROVISIONALLY RECTIFIED

In accordance with the provision of paragraph 3.7.3 of Procedures for port State control (resolution A.[...](32))

(Copy to maritime authority of next port of call, flag Administration, or other certifying authority as appropriate)

1	From (country/region)	2	Port	
3	To (country/region)	4	Port	
5	Name of ship	6	Date departed	
7	Estimated place and time of arrival			
8	IMO number	9	Flag of ship and POR	
10	Type of ship	11	Call sign	
12	Gross tonnage	13	Year of build	
14	Issuing authority of relevant certificate(s)			
15	Nature of deficiencies to be rectified	16	Suggested action (including action at next port of call)	
17 Action taken				
Rep	orting authority	0	ffice	
Nam	ne	Τe	elefax/email	
(duly authorized PSCO of reporting authority)				

Date

Signature
REPORT OF ACTION TAKEN TO THE NOTIFYING AUTHORITY

In accordance with the provision of paragraph 3.7.3 of Procedures for port State control (resolution A.[...](32))

(by telefax/email and/or mail)

1	To:	(Name) (Position) (Authority) Telephone Date:	Tele	efax/email
2	From	n: (Name) (Position) (Authority) Telephone	Tele	efax/email
3	Name	of ship		
4	Call si	ign	5	IMO Number
6	Port o	f inspection		
7	Date of	of inspection		
8	Actic	on taken		
	a) [Deficiencies	b)	Action taken
9	Next p	port	(Da	te)
10	Suppo	orting documentation No Yes	(Se	e attached)
			Sigi	nature

FORMAT FOR THE REPORT OF CONTRAVENTION OF MARPOL (article 6)

PROCEDURES FOR PORT STATE CONTROL

(resolution A.[...](32))

(Issuing (Addres (Teleph (Telefax (Email)	g authority) ss) ione) x)		Copy to:	Master
1	Reporting country			
2	Name of ship			
3	Flag of ship			
4	Type of ship			
5	Call sign	6	IMO numb	er
7	Gross tonnage	8	Deadweigl (where ap	ht propriate)
9	Year of build	10	Classificat	ion society
11	Date of incident	12	Place of in	cident
13	Date of investigation			

14 In case of contravention of discharge provisions, a report may be completed in addition to a port State report on deficiencies. This report should be in accordance with parts 2 and 3 of appendix 3 and/or parts 2 and 3 of appendix 4, as applicable, and should be supplemented by documents such as:

- .1 a statement by the observer of the pollution;
- .2 the appropriate information listed under section 1 of part 3 of appendices 3 and 4 to the Procedures; the statement should include considerations which lead the observer to conclude that none of any other possible pollution sources is in fact the source;
- .3 statements concerning the sampling procedures both of the slick and on board; these should include location where and time when samples were taken, identity of person(s) taking the samples and receipts identifying the persons having custody and receiving transfer of the samples;

- .4 reports of analyses of samples taken of the slick and on board; the reports should include the results of the analyses, a description of the method employed, reference to or copies of scientific documentation attesting to the accuracy and validity of the method employed and names of persons performing the analyses and their experience;
- .5 if applicable, a statement by the PSCO on board together with the PSCO's rank and organization;
- .6 statements by persons being questioned;
- .7 statements by witnesses;
- .8 photographs of the slick; and
- .9 copies or printouts of relevant pages of Oil/Cargo Record Books, logbooks, discharge recordings, etc.

Name and title (duly authorized contravention investigation official)

Signature

COMMENTS BY FLAG STATE ON DETENTION REPORT

Did you receive the notification of detention? (tick the box if the answer is "yes")

Action taken

a)	Deficiencies	b) Cause	c)	Action taken

Additional Information:

GUIDELINES FOR PORT STATE CONTROL UNDER MARPOL ANNEX VI

Chapter 1 GENERAL

1.1 This document is intended to provide basic guidance on the conduct of port State control inspections for compliance with MARPOL Annex VI (hereinafter referred to as "the Annex") and afford consistency in the conduct of these inspections, the recognition of deficiencies and the application of control procedures.

Chapter 2 INSPECTIONS OF SHIPS REQUIRED TO CARRY THE IAPP CERTIFICATE AND/OR THE IEE CERTIFICATE

2.1 Initial inspections

2.1.1 The port State control officer (PSCO) should ascertain the status of the ship as regards application of regulations 20 and 21 of the Annex, the ship's tonnage, the date of ship construction and the date of installation of equipment on board which are subject to the provisions of the Annex, in order to confirm which regulations of the Annex are applicable.

2.1.2 On boarding and introduction to the master or responsible ship's officer, the PSCO should examine the following documents, where applicable:

- .1 the International Air Pollution Prevention Certificate (IAPP Certificate) (regulation VI/6), including its Supplement;
- .2 the International Energy Efficiency Certificate (IEE Certificate) (regulation VI/6) including its Supplement;
- .3 the Engine International Air Pollution Prevention Certificate (EIAPP Certificate) (paragraph 2.2 of the NO_X Technical Code) including its Supplement, for each applicable marine diesel engine;
- .4 the Technical File (paragraph 2.3.4 of the NO_X Technical Code) for each applicable marine diesel engine;
- .5 depending on the method used for demonstrating NO_X compliance for each applicable marine diesel engine:
 - .1 the Record Book of Engine Parameters for each marine diesel engine (paragraph 6.2.2.7 of the NO_X Technical Code) demonstrating compliance with regulation VI/13 by means of the marine diesel engine parameter check method; or
 - .2 documentation relating to the simplified measurement method; or
 - .3 documentation related to the direct measurement and monitoring method;
- .6 for a ship to which regulation VI/13.5.1 applies for a particular NO_X Tier III emission control area and that has one or more installed marine diesel engines

certified to both Tier II and Tier III or which has one or more marine diesel engines certified to Tier II only,^{*} the required logbook and the recordings for the tier and on/off status of those marine diesel engines while the ship is within an applicable NO_X Tier III emission control area;

- .7 the Approved Method File (regulation VI/13.7);
- .8 the written procedures covering fuel oil change over operations (in a working language or languages understood by the crew) where separate fuel oils are used in order to achieve compliance (regulation VI/14.6);
- .9 the approved documentation relating to exceptions and/or exemptions granted under regulation VI/3;
- .10 the approved documentation (SECC where issued, ETM, OMM, SECP) and relating to any installed Exhaust Gas Cleaning System (EGCS) or equivalent means, to reduce SO_x emissions (regulation VI/4);
- .11 EGCS monitoring records, checking they have been retained and show compliance. Additionally, checking that the EGCS Record Book including nitrate discharge data and performance records,[†] or approved alternative, has been duly maintained;
- .12 the bunker delivery notes (BDNs) and representative samples or records thereof (regulation VI/18);
- .13 the copy of the type approval certificate of applicable shipboard incinerator (resolutions MEPC.76(40) or MEPC.244(66));
- .14 the Ozone-depleting Substances Record Book (regulation VI/12.6);
- .15 the VOC Management Plan (regulation VI/15.6);
- .16 any notification to the ship's flag Administration issued by the master or officer in charge of the bunker operation together with any available commercial documentation relevant to non-compliant bunker delivery, regulation VI/18.2;
- .17 if the ship has not been able to obtain compliant fuel oil, the notification to the ship's flag Administration and the competent authority of the relevant port of destination as set out in the appendix;
- .18 the Ship Energy Efficiency Management Plan (SEEMP) including, where applicable, the methodology that will be used to collect the data required by regulation 22A of the Annex and the associated Confirmation of Compliance in respect of that methodology; and

Unified Interpretation to regulation 13.5.3 set out in MEPC.1/Circ.795/Rev.5.

In assessing the emission ratio and discharge water records the PSCO should be mindful that such factors as transient engine operation or analyser performance outputs may result in isolated "spikes" in the recorded output which, while these measurements in themselves may be above the required emission ratio or discharge water limit values, do not indicate that overall the EGCS was not being operated and controlled as required and hence should not be taken as evidence of non-compliance with the requirements.

.19 for the year 2019 and onwards that the ship has, no later than 1 June of each following year, the Statement) of Compliance – Fuel Oil Consumption Reporting.

The Record Books referenced in sub-paragraphs .1, .5, .10 and .13 above may be presented in an electronic format. A declaration from the Administration should be viewed in order to accept this Electronic Record Book. If a declaration cannot be provided, a hard copy Record Book will need to be presented for examination.

2.1.3 As a preliminary check, the IAPP Certificate's validity should be confirmed by verifying that the Certificate is properly completed and signed and that required surveys have been performed.

2.1.4 Through examining the Supplement to the IAPP Certificate, the PSCO may establish how the ship is equipped for the prevention of air pollution.

2.1.5 In the case where the bunker delivery note or the representative sample as required by regulation VI/18 presented to the ship are not in compliance with the relevant requirements (the BDN is set out in appendix V of MARPOL Annex VI), the master or officer in charge of the bunker operation may have documented that through a Notification to the ship's flag Administration with copies to the port authority under whose jurisdiction the ship did not receive the required documentation pursuant to the bunkering operation and to the bunker deliverer.

2.1.6 In addition, if the BDN shows compliant fuel, but the master has independent test results of the fuel oil sample taken by the ship during the bunkering which indicates non-compliance, the master may have documented that through a Notification to the ship's flag Administration with copies to the competent authority of the relevant port of destination, the Administration under whose jurisdiction the bunker deliverer is located and to the bunker deliverer.

2.1.7 In all cases, a copy may be retained on board the ship, together with any available commercial documentation, for the subsequent scrutiny of port State control.

2.1.8 As a preliminary check, the IEE Certificate's validity should be confirmed by verifying that the Certificate is properly completed and signed.

2.2 Initial inspection on ships equipped with equivalent means of SO_x compliance.

- 2.2.1 On ships equipped with equivalent means of compliance, the PSCO will look at:
 - .1 evidence that the ship has received an appropriate approval for any installed equivalent means (approved, under trial or being commissioned);
 - .2 evidence that the ship is using an equivalent means, as identified on the Supplement of the IAPP certificate, for fuel oil combustion units on board or that compliant fuel oil is used in equipment not so covered; and
 - .3 BDNs on board^{*} which indicate that the fuel oil is intended to be used in combination with an equivalent means of SO_x compliance or the ship is subject to a relevant exemption to conduct trials for SO_x emission reduction and control technology research.

^{*} Resolution MEPC.305(73) on *Prohibition on the carriage of non-compliant fuel oil for combustion purposes for propulsion or operation on board a ship* is not applicable to fuel oil carried as cargo or for ships fitted with an approved equivalent means of compliance.

2.2.2 In the case where an EGCS is not in compliance with the relevant requirements for other than transitory periods and isolated spikes in the recorded output, the master or officer in charge may have documented that through a Notification to the ship's flag Administration with copies to the competent authority of the relevant port of destination, and present those corrective actions taken in order to rectify the situation in accordance with the guidance given in the EGCS Technical Manual. If a malfunction occurs in the instrumentation for the monitoring of emission to air or the monitoring of washwater discharge to sea, the ship may have alternative documentation demonstrating compliance.*

2.3 Initial inspection within an ECA

2.3.1 When a ship is inspected in a port in an ECA designated for SO_X emission control, the PSCO should look at:

- .1 evidence of fuel oil delivered to and used on board with a sulphur content of not more than 0.10% m/m through the BDNs and appropriate onboard records including records of bunkering operations as set out in the Oil Record Book Part 1 (regulations VI/18.5 and VI/14.4); and
- .2 for those ships using separate fuel oils for compliance with regulation VI/14, evidence of a written procedure (in a working language or languages understood by the crew) and records of changeover to fuel oil with a sulphur content of not more than 0.10% m/m before entering the ECA such that compliant fuel was being used while sailing in the entire ECA as required in regulation VI/14.6.

2.3.2 When a ship to which regulation VI/13.5.1 applies for a particular NO_X Tier III emission control area is inspected in a port in that area, the PSCO should look at:

- .1 the records in respect of the tier and on/off status, together with any changes to that status while within that NO_x Tier III emission control area, which are to be logged as required by regulation VI/13.5.3 in respect of an installed marine diesel engine certified to both Tier II and Tier III or which is certified to Tier II only;[†] and
- .2 the status of an installed marine diesel engine which is certified to both Tier II and Tier III showing that that engine was operating in its Tier III condition on entry into that NO_X Tier III emission control area and that status was maintained at all times while that marine diesel engine was in operation within that area; or
- .3 the records related to the conditions associated with an exemption granted under regulation VI/13.5.4, checking they have been logged as required by that exemption and that the terms and duration of that exemption have been complied with as required.

^{*} MEPC.1/Circ.883 on Guidance on indication of ongoing compliance in the case of the failure of a single monitoring instrument, and recommended actions to take if the Exhaust Gas Cleaning System (EGCS) fails to meet the provisions of the 2015 EGCS Guidelines (resolution MEPC.259(68)): ships should have documented notification of system non-compliance to relevant authorities as in paragraph 2.2.2.

[†] Unified Interpretation to regulation 13.5.3 set out in MEPC.1/Circ.795/Rev.5.

2.4 Initial inspection outside an ECA or first port after transiting an ECA

2.4.1 When a ship is inspected in a port outside the ECA the PSCO will look to the same documentation and evidence as during inspections in ports inside the ECA. The PSCO should, in particular, look at:

- .1 evidence that the sulphur content of the fuel oil is in accordance with regulation VI/14.1* through the BDNs and appropriate onboard records including records of bunkering operations as set out in the Oil Record Book Part 1 (regulations VI/18.5 and VI/14.4); and
- .2 evidence of a written procedure (in a working language or languages understood by the crew) and records of changeover from fuel oil with a sulphur content of not more than 0.10% m/m after leaving the ECA such that compliant fuel was being used while sailing in the entire ECA.

2.4.2 When a ship to which regulation VI/13.5.1 applies for a particular NO_X Tier III emission control area is inspected in a port outside that area, the PSCO should look at the records required by 2.3.2.1 and 2.3.2.2 or 2.3.2.3 to ensure that the relevant requirements were complied with for the whole period of time the ship was operating in that area.

2.5 Outcome of initial inspection

2.5.1 If the certificates and documents are valid and appropriate and, after an inspection of the ship to check that the overall condition of the ship meets generally accepted international rules and standards, the PSCO's general impressions and observations on board confirm a good standard of maintenance, the inspection should be considered satisfactorily concluded.

2.5.2 If, however, the PSCO's general impressions or observations on board give clear grounds (see paragraph 2.5.3) for believing that the condition of the ship or its equipment does not correspond substantially with the particulars of the certificates or the documents, the PSCO should proceed to a more detailed inspection.

- 2.5.3 "Clear grounds" to conduct a more detailed inspection include:
 - .1 evidence that certificates required by the Annex are missing or clearly invalid;
 - .2 evidence that documents required by the Annex are missing or clearly invalid;
 - .3 the absence or malfunctioning of equipment or arrangements specified in the certificates or documents;
 - .4 the presence of equipment or arrangements not specified in the certificates or documents;
 - .5 evidence from the PSCO's general impressions or observations that serious deficiencies exist in the equipment or arrangements specified in the certificates or documents;

^{*} Resolution MEPC.305(73) on *Prohibition on the carriage of non-compliant fuel oil for combustion purposes for propulsion or operation on board a ship* is not applicable to fuel oil carried as cargo or for ships fitted with an approved equivalent means of compliance.

- .6 information or evidence that the master or crew are not familiar with essential shipboard operations relating to the prevention of air pollution, or that such operations have not been carried out;
- .7 evidence of inconsistency between information in the bunker delivery note and paragraph 2.3 of the Supplement to the IAPP certificate;
- .8 evidence that an equivalent means has not been used as required; or
- .9 evidence, for example by fuel calculators, that the quantity of bunkered compliant fuel oil is inconsistent with the ship's voyage plan; and
- .10 receipt of a report or complaint containing information that the ship appears to be non-compliant including but not limited to information from remote sensing surveillance of SO_X emissions or portable fuel oil sulphur content measurement devices indicating that a ship appears to use non-compliant fuel while in operation/under way;
- .11 evidence that the tier and/or on/off status of applicable installed marine diesel engines has not been maintained correctly or as required;
- .12 receipt of a report or complaint containing information that one or more of the installed marine diesel engines has not been operated in accordance with the provisions of the respective Technical File or the requirements relevant to a particular NO_X Tier III emission control area; and
- .13 receipt of a report or complaint containing information that the conditions attached to an exemption granted under regulation VI/13.5.4 have not been complied with.

2.6 More detailed inspections

- 2.6.1 The PSCO should verify that:
 - .1 there are effectively implemented maintenance procedures for the equipment containing ozone-depleting substances; and
 - .2 there are no deliberate emissions of ozone-depleting substances.

2.6.2 In order to verify that each installed marine diesel engine with a power output of more than 130 kW is approved by the Administration in accordance with the NO_x Technical Code and maintained appropriately, the PSCO should pay particular attention to the following:

- .1 examine such marine diesel engines to be consistent with the EIAPP Certificate and its Supplement, Technical File and, if applicable, Record Book of Engine Parameters or Onboard Monitoring Manual and related data;
- .2 examine marine diesel engines specified in the Technical Files to verify that no unapproved modifications, which may affect NO_X emission, have been made to the marine diesel engines;
- .3 in the case of an installed marine diesel engine certified to Tier III, check that the required records, if applicable, in accordance with regulation VI/13.5.1 or in the Technical File, including those required by 2.3.6 of the NO_x Technical

Code, have been maintained as necessary and that the marine diesel engine, including any NO_x control device and associated ancillary systems and equipment, including, where fitted, bypass arrangements, is maintained in accordance with the associated Technical File and is in good order;

- .4 if applicable, examine whether the conditions attached to an exemption granted under regulation VI/13.5.4 have been complied with as required;
- .5 examine marine diesel engines with a power output of more than 5,000 kW and a per cylinder displacement at or above 90 litres installed on a ship constructed on or after 1 January 1990 but prior to 1 January 2000 to verify that they are certified, if so required, in accordance with regulation VI/13.7;
- .6 in the case of ships constructed before 1 January 2000, verify that any marine diesel engine which has been subject to a major conversion, as defined in regulation VI/13, has been approved by the Administration; and
- .7 emergency marine diesel engines intended to be used solely in case of emergency are still in use for this purpose.

2.6.3 The PSCO should check and verify whether fuel oil complies with the provisions of regulation VI/14 taking into account appendix VI^{*} of MARPOL Annex VI.

2.6.4 The PSCO should pay attention to the record required in regulation VI/14.6 in order to identify the sulphur content of fuel oil used by the ship depending on the area of trade, or that other equivalent approved means have been applied as required, the fuel oil consumed in and outside the ECA, and that there is enough fuel in compliance with regulation VI/14 to reach the next port destination.

2.6.5 Where EGCS is used, the PSCO should check that it has been installed and operated, together with its monitoring systems, in accordance with the associated approved documentation according to the survey procedures as established in the OMM.

2.6.6 If the ship is equipped with an EGCS as an equivalent means of SO_x compliance, the PSCO should verify that the system is properly functioning, is in operation, there are continuous-monitoring systems with tamper-proof data recording and processing devices,[†] if applicable, and the records demonstrate the necessary compliance when set against the limits given in the approved documentation and applies to relevant fuel combustion units on board. Checking can include but is not limited to emissions ratio, pH, PAH, turbidity readings as limit values given in ETM-A or ETM-B and operation parameters as listed in the system documentation.

2.6.7 If the ship is a tanker, as defined in regulation VI/2.21, the PSCO should verify that the vapour collection system approved by the Administration, taking into account MSC/Circ.585, is installed, if required under regulation VI/15.

2.6.8 If the ship is a tanker carrying crude oil, the PSCO should verify that there is on board an approved VOC Management Plan.

2.6.9 The PSCO should verify that prohibited materials are not incinerated.

^{*} Amendments to MARPOL VI, appendix VI, Verification procedures for a MARPOL Annex VI fuel oil sample.

⁺ Equivalent emission values for emission abatement methods are 4.3 and 21.7 SO2 (ppm)/CO2 (% v/v) for marine fuels with a sulphur content of 0.10 and 0.50 (% m/m) respectively.

2.6.10 The PSCO should verify that shipboard incineration of sewage sludge or sludge oil in boilers or marine power plants is not undertaken while the ship is inside ports, harbours or estuaries (regulation VI/16.4).

2.6.11 The PSCO should verify that the shipboard incinerator, if required by regulation VI/16.6.1, is approved by the Administration. For these units, it should be verified that the incinerator is properly maintained, therefore the PSCO should examine whether:

- .1 the shipboard incinerator is consistent with the certificate of shipboard incinerator;
- .2 the operational manual, in order to operate the shipboard incinerator within the limits provided in appendix IV to the Annex, is provided; and
- .3 the combustion chamber flue gas outlet temperature is monitored at all times the unit is in operation (regulation VI/16.9).

2.6.12 The PSCO should verify whether the ship has been subject to a major conversion (regulation VI/2.24) or there have been changes to the ship in respect of aspects which are covered by the EEDI Technical File.

2.6.13 If there are clear grounds as defined in paragraph 2.5.3, the PSCO may examine operational or reporting procedures by confirming that:

- .1 the master or crew are familiar with the procedures to prevent emissions of ozone-depleting substances;
- .2 the master or crew are familiar with the proper operation and maintenance of marine diesel engines, in accordance with their Technical Files or Approved Method file, as applicable, and with due regard for emission control areas for NO_X control;
- .3 the master or crew are familiar with fuel oil bunkering procedures in connection to the respective bunker delivery notes and onboard records including the Oil Record Book Part 1 (regulations VI/18.5 and VI/14.4) and retained samples as required by regulation VI/18;
- .4 the master or crew are familiar with the correct operation of an EGCS or other equivalent means on board together with any applicable monitoring and recording, and record keeping requirements;
- .5 the master or crew are familiar and have undertaken the necessary fuel oil changeover procedures, or equivalent, associated with demonstrating compliance within an emission control area;
- .6 the master or crew are familiar with the garbage screening procedure to ensure that prohibited garbage is not incinerated;
- .7 the master or crew are familiar with the operation of the shipboard incinerator, as required by regulation VI/16.6, within the limits provided in appendix IV to the Annex, in accordance with its operational manual;
- .8 the master or crew are familiar with the regulation of emissions of VOCs, when the ship is in ports or terminals under the jurisdiction of a Party to the 1997 Protocol to MARPOL 73/78 in which VOCs emissions are to be

regulated, and are familiar with the proper operation of a vapour collection system approved by the Administration (in case the ship is a tanker as defined in regulation VI/2.21); and

.9 the master or crew are familiar with the application of the VOC Management Plan, if applicable.

2.7 Detainable deficiencies

2.7.1 In exercising his or her functions, the PSCO should use professional judgement to determine whether to detain the ship until any noted deficiencies are corrected or to allow it to sail with certain deficiencies which do not pose an unreasonable threat of harm under the scope of the Annex provided they will be addressed in a timely manner. In doing this, the PSCO should be guided by the principle that the requirements contained in the Annex, with respect to the construction, equipment and operation of the ship, are essential for the protection of the marine environment, navigational safety or human health and that departure from these requirements could constitute an unreasonable threat of harm to the protection aspects mentioned and should be avoided.

2.7.2 In order to assist the PSCO in the use of these Guidelines, there follows a list of deficiencies which are considered, taking into account the provisions of regulation VI/3, to be of such a serious nature that they may warrant the detention of the ship involved:

- .1 absence of valid IAPP Certificate, EIAPP Certificates or Technical Files, if applicable;
- .2 absence of valid IEE Certificate, EEDI Technical File or SEEMP;
- .3 absence of a valid Statement of Compliance Fuel Oil Consumption Reporting covering the year 2019 and onwards from 1 June of each following year;
- .4 a marine diesel engine, with a power output of more than 130 kW, which is installed on board a ship constructed on or after 1 January 2000, or a marine diesel engine having undergone a major conversion on or after 1 January 2000, which does not conform to its Technical File, or where the required records have not been maintained as necessary or where it has not met the applicable requirements of the particular NO_x Tier III emission control area in which it is operating;
- .5 a marine diesel engine, with a power output of more than 5,000 kW and a per cylinder displacement at or above 90 litres, which is installed on board a ship constructed on or after 1 January 1990 but prior to 1 January 2000, and an approved method for that engine has been certified by an Administration and was commercially available, for which an approved method is not installed after the first renewal survey specified in regulation VI/13.7.2;
- .6 on ships not equipped with equivalent means of SO_x compliance, based on the methodology of sample analysis in accordance with appendix VI^{*} of MARPOL Annex VI, the sulphur content of any fuel oil being used or carried for use on board exceeds the applicable limit required by regulation VI/14. If the master claims that it was not possible to bunker compliant fuel oil, the

^{*} Amendments to MARPOL VI, appendix VI, Verification procedures for a MARPOL Annex VI fuel oil sample.

PSCO should take into account the provisions of regulation VI/18.2 (see the appendix).

- .7 on ships equipped with equivalent means of SO_x compliance, absence of an appropriate approval for the equivalent means, which applies to relevant fuel combustion units on board. With regard to combustion units not connected to an EGCS, the sulphur content of any fuel oil being used on these combustion units exceeds the limits stipulated in regulation VI/14, taking into account the provisions of regulation VI/18.2 (see the appendix).
- .8 non-compliance with the relevant requirements while operating within an emission control area for SO_x and particulate matter control;
- .9 an incinerator installed on board the ship on or after 1 January 2000 does not comply with requirements contained in appendix IV to the Annex, or the standard specifications for shipboard incinerators developed by the Organization (resolutions MEPC.76(40) and MEPC.244(66)); and
- .10 the master or crew are not familiar with essential procedures regarding the operation of air pollution prevention equipment or reporting requirements as defined in paragraph 2.6.12 above.

Chapter 3 INSPECTIONS OF SHIPS OF NON-PARTIES TO THE ANNEX AND OTHER SHIPS NOT REQUIRED TO CARRY THE IAPP CERTIFICATE OR THE IEE CERTIFICATE

3.1 Ships of non-Parties and ships not required to carry the IAPP Certificate.

3.1.1 As this category of ships is not provided with the IAPP Certificate, the PSCO should judge whether the condition of the ship and its equipment satisfies the requirements set out in chapter 3 of the Annex. In this respect, the PSCO should take into account that, in accordance with article 5(4) of MARPOL, no more favourable treatment is to be given to ships of non-Parties.

3.1.2 In all other respects the PSCO should be guided by the procedures for ships referred to in chapter 2 and should be satisfied that the ship and crew do not present a danger to those on board or an unreasonable threat of harm to the marine environment.

3.1.3 If the ship has a form of certification other than the IAPP Certificate, the PSCO may take such documentation into account in the evaluation of the ship.

3.2 Ships of non-Parties and ships not required to carry the IEE Certificate.

3.2.1 As ships of non-Parties are not provided with the IEE Certificate, the PSCO may examine equivalent documentation issued by that non-Party showing that the ship is of a design no less energy efficient than that required by Chapter 4 of the Annex. In addition, the ship should have on board an energy efficiency management plan equivalent to that required for the SEEMP. Such ships are not required to have documentation and procedures covering Fuel Oil Consumption Reporting and hence will not have a Statement of Compliance – Fuel Oil Consumption Reporting.

3.2.2 Ships of Parties which are not required to carry the IEE Certificate are not required to have a SEEMP or to have documentation and procedures covering Fuel Oil Consumption Reporting and hence will not have a Statement of Compliance – Fuel Oil Consumption Reporting.

NON-AVAILABILITY OF COMPLIANT FUEL OIL CLAIMED

In case non-availability of compliant fuel oil is claimed the master/owner must present a record of actions taken to attempt to bunker compliant fuel oil and provide evidence:

- .1 of attempts to purchase compliant fuel oil in accordance with its voyage plan;
- .2 if the fuel oil was not made available where expected, that attempts were made to locate alternative sources for such fuel oil; and
- .3 that despite best efforts to obtain compliant fuel oil no such fuel oil was made available for purchase.

Best efforts to procure compliant fuel oil include, but are not limited to, investigating alternative sources of fuel oil prior to commencing the voyage or en route.

The ship should not be required to deviate from its intended voyage or to unduly delay the voyage in order to achieve compliance.

If the ship provides the information, as above, the port State should take into account all relevant circumstances and the evidence presented to determine the appropriate action to take, including not taking control measures.

The master/owner may provide evidence as below to support their claim (not exhaustive):

- .1 a copy (or description) of the ship's voyage plan, including the ship's port of origin and port of destination;
- .2 the time the ship first received notice it would be conducting a voyage involving transit/arrival in the port and the ship's location when it first received such notice;
- .3 a description of the actions taken to attempt to achieve compliance, including a description of all attempts that were made to locate alternative sources of compliant fuel oil, and a description of the reason why compliant fuel was not available (e.g. compliant fuel oil was not available at ports on the "intended voyage", fuel oil supply disruptions at port);
- .4 the cost of compliant fuel is not considered to be a valid basis for claiming non-availability of fuel;
- .5 names and addresses of the fuel oil suppliers contacted and the dates on which contact was made;
- .6 in cases of fuel oil supply disruption, the name of the port at which the ship was scheduled to receive compliant fuel oil and the name of the fuel supplier that is reporting the non-availability of compliant fuel oil;
- .7 the availability of compliant fuel oil at the next port-of-call and plans to obtain that fuel oil; and

.8 if applicable, identification and description of any operational constraints that prevented use of compliant fuel oil, e.g. with respect to viscosity or other fuel oil parameters.

If, despite best efforts, it was not possible to procure compliant fuel oil the master/owner must notify the port State control authorities in the port of arrival and the flag Administration (regulation VI/18.2.4).

LIST OF INSTRUMENTS RELEVANT TO PORT STATE CONTROL PROCEDURES

Instrument (related to)	Name	IMO Body	Remark	Year (adopted/ approved)
AFS				
MEPC.104(49)	Guidelines for brief sampling of anti-fouling systems on ships			18 July 2003
MEPC.208(62)	2011 Guidelines for inspection of anti-fouling systems on ships			15 July 2011
Ballast Water				
MEPC.173(58)	Guidelines for ballast water sampling (G2)	MEPC/PPR	To be read in conjunction with MEPC.252(67) and BWM.2/Circ.42/Rev .1	10 October 2008
MEPC.252(67)	Guidelines for port State control under the BWM Convention	MEPC/III	To be read in conjunction with MEPC.173(58) and BWM.2/Circ.42/Rev .1	17 October 2014
MEPC.279(70)	2016 Guidelines for approval of ballast water management systems (G8)	MEPC		28 October 2016
MEPC.297(72)	Amendments to regulation B-3 (Implementation schedule of ballast water management for ships)	MEPC		13 April 2018
BWM.2/Circ.42/Rev.2	Guidance on ballast water sampling and analysis for trial use in accordance with the BWM Convention and Guidelines (G2)	MEPC/PPR	To be read in conjunction with MEPC.173(58) and MEPC.252(67)	9 December 2020

Instrument (related to)	Name	IMO Body	Remark	Year (adopted/ approved)
Bulk				
A.797(19)	Safety of ships carrying solid bulk cargoes	CCC		23 November 1995
A.862(20)	Code of practice for safe loading and unloading of bulk carriers	CCC	As amended by MSC.238(82) and MSC.304(87)	27 November 1997
MSC/Circ.656	Safety of ships carrying solid bulk cargoes	CCC		6 June 1994
MSC/Circ.1117	Guidance for checking the structure of bulk carriers	SDC		24 June 2004
MSC.1/Circ.1464/Rev.1, Corr.1 and Corr.2	Unified interpretations of the provisions of SOLAS chapters II-1 and XII, of the Technical provisions for means of access for inspections (resolution MSC.158(78)) and of the Performance standards for water level detectors on bulk carriers and single hold cargo ships other than bulk carriers (resolution MSC.188(79))	SDC	As amended by MSC.1/Circ.1507 of 5 June 2015	24 October 2013 14 November 2013 9 June 2017
Certificates				
MSC/Circ.606	Port State concurrence with SOLAS exemptions	111		12 February 1993
MSC/Circ.1012 MEPC/Circ.384	Endorsement of certificates with the date of completion of the survey on which they are based			26 June 2001
FAL.2/Circ.131- MEPC.1/Circ.873- MSC.1/Circ.1586- LEG.2/Circ.3 and Corr.1	List of certificates and documents required to be carried on board ships, 2017	111		19 July 2017 4 June 2018
MSC-MEPC.4/Circ.1	Retention of original records/documents on board ships	111		26 September 2005
MSC-MEPC.5/Circ.6	Guidance on the timing of replacement of existing certificates by the certificates issued after the entry into force of amendments to certificates in IMO instruments	111		6 August 2009

Instrument (related to)	Name	IMO Body	Remark	Year (adopted/ approved)
MARPOL				
MEPC.259(68)	2015 Guidelines for exhaust gas cleaning systems	MEPC/PPR	Read in conjunction with MEPC68/21/Add.1/ Corr.2	15 May 2015
MSC.286(86)	Recommendations for material safety data sheets (MSDS) for MARPOL Annex I oil cargo and oil fuel	PPR		5 June 2009
MEPC.312(74)	Guidelines for the use of electronic record books under MARPOL	MEPC		17 May 2019
MEPC.320(74)	2019 Guidelines for consistent implementation of the 0.50% sulphur limit under MARPOL Annex VI	MEPC		17 May 2019
MSC.465(101)	Recommended interim measures to enhance the safety of ships relating to the use of oil fuel	MSC		14 June 2019
MEPC/Circ.479 and Corr.1	Guidelines for port State control officers while checking compliance with the Condition Assessment Scheme (CAS)	MEPC/III		24 August 2005 6 October 2005
MEPC.1/Circ.508	Bunker delivery note and fuel oil sampling	MEPC/III		9 May 2006

Instrument (related to)	Name	IMO Body	Remark	Year (adopted/ approved)
MEPC.1/Circ.516	Public access to the condition assessment scheme (CAS) database	MEPC		5 May 2006
MEPC.1/Circ.637	Fuel oil availability and quality	MEPC		17 November 2008
MEPC.1/Circ.640	Interim Guidance on the use of the Oil Record Book concerning voluntary declaration of quantities retained on board in oily bilge water holding tanks and heating of oil residue (sludge)	SSE		4 November 2008
MEPC.1/Circ.675/Rev.1	Discharge of cargo hold washing water in the Gulfs area, Mediterranean Sea area and wider Caribbean Region under MARPOL Annex V	MEPC		26 March 2010
MEPC.1/Circ.834/Rev.1	Consolidated guidance for port reception facility providers and users	MEPC		1 March 2018
MEPC.1/Circ.864/Rev.1	2019 Guidelines for on board sampling for the verification of the sulphur content of the fuel oil used on board ships	MEPC		21 May 2019
MEPC.1/Circ.881	Guidance for port State control on contingency measures for addressing non-compliant fuel oil	MEPC		21 May 2019
MEPC.1/Circ.882	Early application of the verification procedures for a MARPOL Annex VI fuel oil sample (regulation 18.82 or regulation 14.8)	MEPC		16 July 2019
MEPC.1/Circ.883	Guidance on indication of ongoing compliance in the case of the failure of a single monitoring instrument, and recommended actions to take if the Exhaust Gas Cleaning System (EGCS) fails to meet the provisions of the 2015 EGCS Guidelines (resolution MEPC.259(68))	PPR		21 May 2019
MSC-MEPC.4/Circ.3	Blanking of bilge discharge piping systems in port	MSC/MEPC		19 December 2008

Instrument (related to)	Name	IMO Body	Remark	Year (adopted/ approved)
PSC activities				
MSC.1/Circ.1191	Further reminder of the obligation to notify flag States when exercising control and compliance measures	MSC/III		30 May 2006
MSC.1/Circ.1199	Interim Guidance on compliance of ships carrying dry cargoes in bulk with requirements of SOLAS chapters II-1, III, IX, XI-1 and XII	SDC		31 May 2006
MSC.1/Circ.1221	Validity of type approval certification for marine products	111		11 December 2006
MSC.1/Circ.1565	Guidelines on the voluntary early implementation of amendments to the 1974 SOLAS Convention and related mandatory instruments			15 June 2017
MSC/Circ.1011 MEPC/Circ.383	Measures to improve port State control procedures	111		26 June 2001
MSC-MEPC.2/Circ.2	IMO requirements on carriage of publications on board ships	III/NCSR		1 June 2006
MSC-MEPC.4/Circ.2	Code of good practice for port State control officers	111		1 November 2007
Security				
MSC.159(78)	Interim Guidance on control and compliance measures to enhance maritime security	MSC/III		21 May 2004
MSC/Circ.1097	Guidance relating to the implementation of SOLAS chapter XI-2 and the ISPS Code	MSC		6 June 2003

MSC 104/18/Add.1 Annex 30, page 136

Instrument Name I (related to)	IMO Body	Remark	Year (adopted/ approved)
MSC/Circ.1113 Guidance to port State control officers on the non-security related elements of the 2002 SOLAS amendment		Automatic Identification Systems (AIS) and ship's identification number, and, Continuous Synopsis Record (CSR) Read in conjunction with A.959(23), adopted on 5 December 2003, Format and guidelines for the maintenance of the Continuous Synopsis Record (CSR), as amended by MSC.198(80), adopted on 20 May 2005	7 June 2004

Instrument (related to)	Name	IMO Body	Remark	Year (adopted/ approved)
MSC/Circ.1156	Guidance on the access of public authorities, emergency response services and pilots on board ships to which SOLAS chapter XI-2 and the ISPS Code apply	MSC	Read in conjunction with MSC.1/Circ.1342 of 27 May 2010, Reminder in connection with shore leave and access to ships (and MSC.1/Circ.1194 of 30 May 2006)	23 May 2005
MSC.1/Circ.1235	Guidelines on security-related training and familiarization for shipboard personnel	HTW		21 October 2007
MSC.1/Circ.1342	Reminder in connection with shore leave and access to ships	MSC		27 May 2010
SOLAS				
A.1047(27)	Principles of minimum safe manning	MSC/HTW		30 November 2011
MSC/Circ.592	Carriage of dangerous goods – Acceptance of the document of compliance	CCC		21 April 1992
MSC/Circ.811	Identification of float-free arrangements for liferafts	SSE		8 July 1997
MSC/Circ.887	Interpretation of the term "other strategic points" in SOLAS regulation III/50 and LSA Code section VII/7.2	SSE		21 December 1998
MSC/Circ.907	Application of SOLAS regulation III/28.2 concerning helicopter landing areas on non ro-ro passenger ships	SSE		17 June 1999

Instrument (related to)	Name	IMO Body	Remark	Year (adopted/ approved)
MSC/Circ.955	Servicing of life-saving appliances and radiocommunication equipment under the Harmonized System of Survey and Certification (HSSC)			23 June 2000
MSC/Circ.1016	Application of SOLAS regulation III/26 concerning fast rescue boats and means of rescue systems on ro-ro passenger ships	SSE	Complemented by MSC/Circ.1094 of 17 June 2003, Application of SOLAS regulation III/26 concerning fast rescue boat systems on ro-ro passenger ships	26 June 2001
MSC/Circ.1107	Application of SOLAS regulation II-1/3-6 on Access to and within spaces in, and forward of, the cargo area of oil tankers and bulk carriers and application of the Technical provisions for means of access for inspections	SDC		25 May 2004
MSC.1/Circ.1326 and Corr.1	Clarification of SOLAS regulation III/19	SSE	Note: SOLAS regulation III/19.3.3.3 as referred to in the circular should be read as SOLAS regulation III/19.3.4.3 as the 2013 amendments to SOLAS renumbered paragraph 19.3.3.3 as 19.3.4.3	11 June 2009 13 August 2009

Instrument (related to)	Name	IMO Body	Remark	Year (adopted/ approved)
MSC.1/Circ.1331	Guidelines for construction, installation, maintenance and inspection/survey of means of embarkation and disembarkation	SSE		11 June 2009
MSC.1/Circ.1402	Safety of pilot transfer arrangements			14 June 2011
STCW				
STCW.7/Circ.22	Advice for port State control officers, recognized organizations and recognized security organizations clarifying training and certification requirements for ship security officers and seafarers with designated security duties	MSC/HTW		25 February 2014
STCW.7/Circ.24/Rev.1	Guidance for Parties, Administrations, port State control authorities, recognized organizations and other relevant parties on the requirements under the STCW Convention, 1978, as amended	III/HTW		16 June 2017
MSC/Circ.635	Tonnage measurement of certain ships relevant to the International Convention on Standards of Training, Certification and Watchkeeping for Seafarers, 1978	HTW/SDC		15 June 1994
MSC/Circ.1089	Guidance on recommended anti-fraud measures and forgery prevention measures for seafarers' certificate	III/HTW		6 June 2003
MSC.1/Circ.1208	Promoting and verifying continued familiarization of GMDSS operators on board ships	HTW		22 May 2006

ANNEX 31

DRAFT ASSEMBLY RESOLUTION

SURVEY GUIDELINES UNDER THE HARMONIZED SYSTEM OF SURVEY AND CERTIFICATION (HSSC), 2021

THE ASSEMBLY,

RECALLING Article 15(j) of the Convention on the International Maritime Organization concerning the functions of the Assembly in relation to regulations and guidelines regarding maritime safety and the prevention and control of marine pollution from ships,

RECALLING ALSO the adoption by:

- (a) the International Conference on the Harmonized System of Survey and Certification, 1988 of the Protocol of 1988 relating to the International Convention for the Safety of Life at Sea, 1974 and of the Protocol of 1988 relating to the International Convention on Load Lines, 1966, which, inter alia, introduced the harmonized system of survey and certification into the International Convention for the Safety of Life at Sea, 1974 and the International Convention on Load Lines, 1966, respectively;
- (b) resolution MEPC.39(29) of amendments to introduce the harmonized system of survey and certification into the International Convention for the Prevention of Pollution from Ships, 1973, as modified by the Protocol of 1978 relating thereto (MARPOL);
- (c) resolution MEPC.132(53) of amendments to introduce the harmonized system of survey and certification into MARPOL Annex VI;
- (d) the International Conference on Ballast Water Management for Ships of the International Convention for the Control and Management of Ships' Ballast Water and Sediments, 2004, which included the harmonized system of survey and certification; and
- (e) the resolutions given below of amendments to introduce the harmonized system of survey and certification into:
 - (i) the International Code for the Construction and Equipment of Ships Carrying Dangerous Chemicals in Bulk (IBC Code) (resolutions MEPC.40(29) and MSC.16(58));
 - (ii) the International Code for the Construction and Equipment of Ships Carrying Liquefied Gases in Bulk (IGC Code) (resolution MSC.17(58)); and
 - (iii) the Code for the Construction and Equipment of Ships Carrying Dangerous Chemicals in Bulk (BCH Code) (resolutions MEPC.41(29) and MSC.18(58)),

RECALLING FURTHER resolution A.1140(31), by which it adopted the *Survey Guidelines under the Harmonized System of Survey and Certification (HSSC), 2019* (hereafter referred to as the "Survey Guidelines"), following successive revocation of resolutions A.1120(30), A.1104(29), A.1076(28), A.1053(27), A.1020(26), A.997(25), A.948(23) and A.746(18), replacing the guidelines adopted by resolutions A.560(14), MEPC.11(18) and MEPC.25(23),

RECOGNIZING the need for the Survey Guidelines to be further revised to take into account the amendments to the IMO instruments referred to above which have entered into force or become effective since the adoption of resolution A.1140(31),

HAVING CONSIDERED the recommendations made by the Maritime Safety Committee, at its [104th] session, and the Marine Environment Protection Committee, at its [seventy-seventh] session,

1 ADOPTS the Survey Guidelines under the Harmonized System of Survey and Certification (HSSC), 2021, as set out in the annex to the present resolution;

2 INVITES Governments carrying out surveys required by relevant IMO instruments to apply the provisions of the annexed Survey Guidelines;

3 REQUESTS the Maritime Safety Committee and the Marine Environment Protection Committee to keep the Survey Guidelines under review and to amend them as necessary;

4 REVOKES resolution A.1140(31).

ANNEX

SURVEY GUIDELINES UNDER THE HARMONIZED SYSTEM OF SURVEY AND CERTIFICATION (HSSC), 2021

(These Guidelines take into account relevant IMO mandatory instruments and amendments thereto entering into force before or on 31 December 2021)

Table of contents

GENERAL

- 1 INTRODUCTION
- 2 TYPES OF SURVEY
- 3 APPLICATION AND ARRANGEMENT OF THE GUIDELINES
- 4 DESCRIPTION OF THE VARIOUS TYPES OF SURVEY
- (I) 4.1 Initial surveys
- (A) 4.2 Annual surveys
- (In) 4.3 Intermediate surveys
- (P) 4.4 Periodical surveys
- (R) 4.5 Renewal surveys
- (B) 4.6 Inspections of the outside of the ship's bottom of cargo ships
- (Ad) 4.7 Additional surveys
 - 4.8 Completion of surveys
 - 5 AMPLIFICATION OF TERMS AND CONDITIONS
 - 5.1 Definition of related items
 - 5.2 Extending to five years a certificate issued for less than five years
 - 5.3 Extending the period between inspections of the outside of the ship's bottom
 - 5.4 Definition of "short voyage"
 - 5.5 Application of "special circumstances"
 - 5.6 Revalidation of certificates
 - 5.7 Meaning of "any five-year period"
 - 5.8 Surveys required after transfer of the ship to the flag of another State
 - 5.9 Recommended conditions for extending the period of validity of a certificate
 - 5.10 Inspection of the outside of a passenger ship's bottom
 - 5.11 Survey of radio installations
 - 5.12 Survey of the automatic identification system (AIS)
 - 5.13 Survey for ships intended to operate in polar waters

Annex 1

SURVEY GUIDELINES UNDER THE 1974 SOLAS CONVENTION, AS MODIFIED BY THE 1988 PROTOCOL RELATING THERETO

- (E) 1 GUIDELINES FOR SURVEYS FOR THE CARGO SHIP SAFETY EQUIPMENT CERTIFICATE
- (EI) 1.1 Initial surveys
- (EA) 1.2 Annual surveys
- (EP) 1.3 Periodical surveys
- (ER) 1.4 Renewal surveys
- (C) 2 GUIDELINES FOR SURVEYS FOR THE CARGO SHIP SAFETY CONSTRUCTION CERTIFICATE
- (CI) 2.1 Initial surveys
- (CA) 2.2 Annual surveys
- (CIn) 2.3 Intermediate surveys
- (CR) 2.4 Renewal surveys
- (B) 3 GUIDELINES FOR THE INSPECTION OF THE OUTSIDE OF THE SHIP'S BOTTOM OF CARGO SHIPS
- (R) 4 GUIDELINES FOR SURVEYS FOR THE CARGO SHIP SAFETY RADIO CERTIFICATE
- (RI) 4.1 Initial surveys
- (RP) 4.2 Periodical surveys
- (RR) 4.3 Renewal surveys
- (P) 5 GUIDELINES FOR SURVEYS FOR THE PASSENGER SHIP SAFETY CERTIFICATE
- (PI) 5.1 Initial surveys
- (PR) 5.2 Renewal surveys

Annex 2

SURVEY GUIDELINES UNDER THE 1966 LL CONVENTION, AS MODIFIED BY THE 1988 PROTOCOL RELATING THERETO

- (L) 1 GUIDELINES FOR SURVEYS FOR THE INTERNATIONAL LOAD LINE CERTIFICATE OR INTERNATIONAL LOAD LINE EXEMPTION CERTIFICATE
- (LI) 1.1 Initial surveys
- (LA) 1.2 Annual surveys
- (LR) 1.3 Renewal surveys

Annex 3

SURVEY GUIDELINES UNDER THE MARPOL CONVENTION

- (O) 1 GUIDELINES FOR SURVEYS FOR THE INTERNATIONAL OIL POLLUTION PREVENTION CERTIFICATE
- (OI) 1.1 Initial surveys
- (OA) 1.2 Annual surveys
- (OIn) 1.3 Intermediate surveys
- (OR) 1.4 Renewal surveys
- (N) 2 GUIDELINES FOR SURVEYS FOR THE INTERNATIONAL POLLUTION PREVENTION CERTIFICATE FOR THE CARRIAGE OF NOXIOUS LIQUID SUBSTANCES IN BULK
- (NI) 2.1 Initial surveys
- (NA) 2.2 Annual surveys
- (NIn) 2.3 Intermediate surveys
- (NR) 2.4 Renewal surveys
- (S) 3 GUIDELINES FOR SURVEYS FOR THE INTERNATIONAL SEWAGE POLLUTION PREVENTION CERTIFICATE
- (SI) 3.1 Initial surveys
- (SR) 3.2 Renewal surveys
- (A) 4 GUIDELINES FOR SURVEYS FOR THE INTERNATIONAL AIR POLLUTION PREVENTION CERTIFICATE AND THE NO_X TECHNICAL CODE
- (AI) 4.1 Initial surveys
- (AA) 4.2 Annual surveys
- (Aln) 4.3 Intermediate surveys
- (AR) 4.4 Renewal surveys

Annex 4

SURVEY GUIDELINES UNDER THE INTERNATIONAL CONVENTION FOR THE CONTROL AND MANAGEMENT OF SHIPS' BALLAST WATER AND SEDIMENTS, 2004

- (B) 1 GUIDELINES FOR SURVEYS FOR THE INTERNATIONAL BALLAST WATER MANAGEMENT CERTIFICATE
- (BI) 1.1 Initial surveys
- (BA) 1.2 Annual surveys
- (BIn) 1.3 Intermediate surveys
- (BR) 1.4 Renewal surveys
- (BAd) 1.5 Additional surveys

Annex 5

SURVEY GUIDELINES UNDER MANDATORY CODES

- (D) 1 GUIDELINES FOR SURVEYS FOR THE INTERNATIONAL CERTIFICATE OF FITNESS FOR THE CARRIAGE OF DANGEROUS CHEMICALS IN BULK AND THE CERTIFICATE OF FITNESS FOR THE CARRIAGE OF DANGEROUS CHEMICALS IN BULK
- (DI) 1.1 Initial surveys
- (DA) 1.2 Annual surveys
- (DIn) 1.3 Intermediate surveys
- (DR) 1.4 Renewal surveys
- (G) 2 GUIDELINES FOR SURVEYS FOR THE INTERNATIONAL CERTIFICATE OF FITNESS FOR THE CARRIAGE OF LIQUEFIED GASES IN BULK
- (GI) 2.1 Initial surveys
- (GA) 2.2 Annual surveys
- (GIn) 2.3 Intermediate surveys
- (GR) 2.4 Renewal surveys
- (W) 3 GUIDELINES FOR SURVEYS FOR THE POLAR SHIP CERTIFICATE ADDITIONAL TO SOLAS CERTIFICATES
- (WI) 3.1 Initial surveys
- (WA) 3.2 Annual surveys
- (WIn) 3.3 Intermediate surveys
- (WP) 3.4 Periodical surveys
- (WR) 3.5 Renewal surveys
- Appendix 1 SUMMARY OF AMENDMENTS TO MANDATORY INSTRUMENTS REFLECTED IN THE SURVEY GUIDELINES UNDER THE HSSC
- Appendix 2 THE HARMONIZED SYSTEM OF SURVEY AND CERTIFICATION DIAGRAMMATIC ARRANGEMENT

GENERAL

1 INTRODUCTION

1.1 These Guidelines supersede the guidelines adopted by resolution A.1140(31), and take account of the harmonized system of survey and certification in the following instruments:

- .1 International Convention for the Safety of Life at Sea, 1974 (1974 SOLAS Convention, or SOLAS 74), as modified by the Protocol of 1988 relating thereto, as amended (SOLAS 74/88);
- .2 International Convention on Load Lines, 1966 (1966 LL Convention, or LLC 66), as modified by the Protocol of 1988 relating thereto, as amended (LLC 66/88);
- .3 International Convention for the Prevention of Pollution from Ships, 1973, as modified by the Protocol of 1978 relating thereto, and as further amended by the Protocol of 1997, as amended (MARPOL);
- .4 International Convention for the Control and Management of Ships' Ballast Water and Sediments, 2004, as amended (BWM Convention));
- .5 International Code for the Construction and Equipment of Ships Carrying Dangerous Chemicals in Bulk, as amended (IBC Code);
- .6 International Code for the Construction and Equipment of Ships Carrying Liquefied Gases in Bulk, as amended (IGC Code);
- .7 Code for the Construction and Equipment of Ships Carrying Dangerous Chemicals in Bulk, as amended (BCH Code); and
- .8 International Code for Ships Operating in Polar Waters (Polar Code).

1.2 These Guidelines take into account the amendments to statutory instruments which entered into force before or on 31 December 2021 (see appendix 1), and contain the following:

- .1 Survey Guidelines under the 1974 SOLAS Convention, as modified by the Protocol of 1988 relating thereto (annex 1);
- .2 Survey Guidelines under the 1966 LL Convention, as modified by the Protocol of 1988 relating thereto (annex 2);
- .3 Survey Guidelines under the MARPOL Convention (annex 3);
- .4 Survey Guidelines under the International Convention for the Control and Management of Ships' Ballast Water and Sediments, 2004 (annex 4); and
- .5 Survey Guidelines under mandatory codes (annex 5).

1.3 The harmonized system, a diagrammatic arrangement of which is given in appendix 2, provides for:

- .1 a one-year standard interval between surveys, based on initial, annual, intermediate, periodical and renewal surveys, as appropriate, except for MARPOL Annex IV, which is based on initial and renewal surveys;
- .2 a scheme providing the necessary flexibility to execute each survey, with provision for:
 - .1 completion of the renewal survey within three months before the expiry date of the existing certificate with no loss of its period of validity; and
 - .2 a "time window" of six months from three months before to three months after the anniversary date of the certificate for annual, intermediate and periodical surveys;
 - .3 items associated with the cargo ship safety construction intermediate survey that are additional to the requirements of the annual survey may, alternatively, be carried out at the second or third annual survey or between these surveys;
- .3 in no case does the survey carried out under 1.3.2.3 remove the need to carry out the required annual surveys.
- .4 a maximum period of validity of five years for all cargo ship certificates;
- .5 a maximum period of validity of 12 months for the Passenger Ship Safety Certificate;
- .6 a system for the extension of certificates limited to three months, enabling a ship to complete its voyage, or one month for ships engaged on short voyages;
- .7 when an extension has been granted, the period of validity of the new certificate starting from the expiry date of the existing certificate before its extension;
- .8 a flexible system for inspection of the outside of the ship's bottom on the following conditions:
 - .1 a minimum of two inspections during any five-year period of validity of the Cargo Ship Safety Construction Certificate or the Cargo Ship Safety Certificate; and
 - .2 the interval between any two such inspections should not exceed 36 months;
- .9 a Cargo Ship Safety Certificate under SOLAS 74/88, as an alternative to separate Cargo Ship Safety Construction, Cargo Ship Safety Equipment and Cargo Ship Safety Radio Certificates; and

.10 a flexible system concerning the frequency and the period of validity of certificates, subject to the minimum pattern of surveys being maintained.

1.4 In implementing the harmonized system, the following principal changes made to the survey and certification requirements of SOLAS 74/88 have been taken into account:

- .1 unscheduled inspections are no longer included and annual surveys are mandatory for cargo ships;
- .2 intervals between the periodical surveys of equipment covered by the Cargo Ship Safety Equipment Certificate are alternately two and three years instead of two years;
- .3 intermediate surveys are required for all ships under the Cargo Ship Safety Construction Certificate;
- .4 inspection of the outside of the ship's bottom is required for all cargo ships;
- .5 intermediate surveys for the Cargo Ship Safety Construction Certificate are completed within three months of either the second or third anniversary date; items associated with the cargo ship safety construction intermediate survey that are additional to the requirements of the annual survey may, alternatively, be carried out at the second or third annual survey or between these surveys;
- .6 all cargo ship certificates may be issued for any period of validity up to and including five years;
- .7 there is provision for a Cargo Ship Safety Certificate; and
- .8 the extension provisions have been reduced from five months to three months to enable a ship to complete its voyage and the extension for one month for a period of grace is limited to ships engaged on short voyages.

1.5 With regard to LLC 66/88, the principal changes to the requirements for survey and certification are the introduction of similar extension provisions (see 1.4.8) and linking of the period of validity of the new certificate to the expiry date of the previous certificate (see 1.3.6).

1.6 With regard to MARPOL and the IBC, IGC and BCH Codes, the main changes are the linking of the period of validity of the new certificate to the expiry date of the previous certificate (see 1.3.6), the holding of the intermediate survey within three months of the second or third anniversary date and the introduction of the same extension provisions (see 1.4.8).

2 TYPES OF SURVEY

The types of survey used in the harmonized system are as follows:

(I) 2.1 *An initial survey* is a complete inspection before a ship is put into service of all the items relating to a particular certificate, to ensure that the relevant requirements are complied with and that these items are satisfactory for the service for which the ship is intended.

- (P) 2.2 *A periodical survey* is an inspection of the items relating to the particular certificate to ensure that they are in a satisfactory condition and fit for the service for which the ship is intended.
- (R) 2.3 *A renewal survey* is the same as a periodical survey but also leads to the issue of a new certificate.
- (In) 2.4 *An intermediate survey* is an inspection of specified items relevant to the particular certificate to ensure that they are in a satisfactory condition and fit for the service for which the ship is intended.
- (A) 2.5 *An annual survey* is a general inspection of the items relating to the particular certificate to ensure that they have been maintained and remain satisfactory for the service for which the ship is intended.
- (B) 2.6 An inspection of the outside of the ship's bottom is an inspection of the underwater part of the ship and related items to ensure that they are in a satisfactory condition and fit for the service for which the ship is intended.
- (Ad) 2.7 *An additional survey* is an inspection, either general or partial according to the circumstances, to be made after:
 - .1 a repair resulting from investigations or whenever any important repairs or renewals are made; or,
 - .2 change, replacement, or significant repair of the structure, equipment, systems, fittings, arrangements and material (BWM Convention regulation E-1.1.5).
 - 2.8 List of types of survey in conventions and codes
- (I) 2.8.1 *Initial surveys*

SOLAS 74/88 regulations I/7(a)(i) and 7(b)(i) regulations I/8(a)(i) and 8(b)(i) regulations I/9(a)(i) and 9(b)(i) regulations I/10(a)(i) and 10(b)(i) LLC 66/88 article 14(1)(a) MARPOL Annex I regulation 6.1.1 MARPOL Annex II regulation 8.1.1 MARPOL Annex IV regulation 8.1.1 MARPOL Annex IV regulation 4.1.1 MARPOL Annex VI regulation 5.1.1 BWM Convention regulation E-1.1.1 IBC Code regulation 1.5.2.1.1 IGC Code regulation 1.4.2.1 BCH Code regulation 1.6.2.1.1

(P) 2.8.2 *Periodical surveys*

SOLAS 74/88 regulations I/8(a)(iii) and 8(b)(ii) regulations I/9(a)(iii) and 9(b)(ii)

(R) 2.8.3 Renewal surveys

SOLAS 74/88 regulations I/7(a)(ii) and 7(b)(ii) regulations I/8(a)(ii) and 8(b)(ii)
regulations I/9(a)(ii) and 9(b)(ii) regulations I/10(a)(ii) and 10(b)(ii)

LLC 66/88 article 14(1)(b) MARPOL Annex I regulation 6.1.2 MARPOL Annex II regulation 8.1.2 MARPOL Annex IV regulation 4.1.2 MARPOL Annex VI regulation 5.1.2 BWM Convention regulation E-1.1.2 IBC Code regulation 1.5.2.1.2 IGC Code regulation 1.4.2.2 BCH Code regulation 1.6.2.1.2

(In) 2.8.4 Intermediate surveys

SOLAS 74/88 regulations I/10(a)(iii) and 10(b)(iii) MARPOL Annex I regulation 6.1.3 MARPOL Annex II regulation 8.1.3 MARPOL Annex VI regulation 5.1.3 BWM Convention regulation E-1.1.3 IBC Code regulation 1.5.2.1.3 IGC Code regulation 1.4.2.3 BCH Code regulation 1.6.2.1.3

(A) 2.8.5 Annual surveys

SOLAS 74/88 regulations I/8(a)(iv), 8(b)(iii), and10(a)(iv) and 10(b)(iv) LLC 66/88 article 14(1)(c) MARPOL Annex I regulation 6.1.4 MARPOL Annex II regulation 8.1.4 MARPOL Annex VI regulation 5.1.4 BWM Convention regulation E-1.1.4 IBC Code regulation 1.5.2.1.4 IGC Code regulation 1.4.2.4 BCH Code regulation 1.6.2.1.4

(B) 2.8.6 Inspection of the outside of the ship's bottom

SOLAS 74/88 regulations I/10(a)(v) and 10(b)(v)

(Ad) 2.8.7 Additional surveys

SOLAS 74/88 regulations I/7(a)(iii) and 7(b)(iii) regulation I/8(a)(v) regulation I/9(a)(iv) regulation I/10(a)(vi)

MARPOL Annex I regulation 6.1.5 MARPOL Annex II regulation 8.1.5 MARPOL Annex IV regulation 4.1.3 MARPOL Annex VI regulation 5.1.5 BWM Convention regulation E-1.1.5 IBC Code regulation 1.5.2.1.5 IGC Code regulation 1.4.2.5 BCH Code regulation 1.6.2.1.5

3 APPLICATION AND ARRANGEMENT OF THE GUIDELINES

3.1 The Guidelines provide a general framework upon which Administrations will be able to base their arrangements for carrying out surveys. It is recognized that survey provisions contained in the Guidelines are not necessarily applicable to all types and sizes of ship.

3.2 While the Guidelines are intended to cover instruments listed in 1.1, they should be applied, as appropriate, to drilling rigs and other platforms covered by MARPOL Annex I regulation 39 and Annex VI regulation 5.

3.3 A description of the various types of survey is given in section 4 and, as shown on the contents page, this is followed by the detailed requirements for the various surveys for each of the certificates.

3.4 When appropriate, the detailed requirements for the various surveys contain a section that is applicable to all cargo ships followed by a section that only applies to specific ship types.

3.5 While the convention or code references are included, where possible, it should be noted that, in general, it has not been possible to indicate where there are differing requirements dependent upon the ship's year of build. Consequently, care should be taken in applying specific requirements, particularly where there have been amendments that are only applicable to ships built after a certain date.

3.6 Although also part of the requirements for the Cargo Ship Safety Construction Certificate, a separate section is provided for inspection of the outside of the ship's bottom.

3.7 SOLAS 74/88 regulation I/12(v) provides for a Cargo Ship Safety Certificate to be issued as an alternative to the Cargo Ship Safety Equipment Certificate, the Cargo Ship Safety Construction Certificate and the Cargo Ship Safety Radio Certificate. Consequently, the surveys for the issue and renewal of the Cargo Ship Safety Certificate should be in accordance with the certificates it replaces and, similarly, the annual and intermediate surveys should be the same as those required for the replaced certificates and the appropriate sections of the Cargo Ship Safety Certificate, endorsed accordingly.

3.8 On the left-hand side of each item to be surveyed may be found two or three letters in brackets, the first indicating the certificate to which the survey relates, as follows:

- (E) for the Cargo Ship Safety Equipment Certificate;
- (C) for the Cargo Ship Safety Construction Certificate;
- (R) for the Cargo Ship Safety Radio Certificate;
- (L) for the International Load Line Certificate;
- (O) for the International Oil Pollution Prevention Certificate;
- (N) for the International Pollution Prevention Certificate for Carriage of Noxious Liquid Substances in Bulk;
- (S) for the International Sewage Pollution Prevention Certificate;
- (A) for the International Air Pollution Prevention Certificate;

- (D) for the International Certificate of Fitness for the Carriage of Dangerous Chemicals in Bulk or the Certificate of Fitness for the Carriage of Dangerous Chemicals in Bulk;
- (G) for the International Certificate of Fitness for the Carriage of Liquefied Gases in Bulk;
- (P) for the Passenger Ship Safety Certificate;
- (W) for the Polar Ship Certificate;
- (B) for the International Ballast Water Management Certificate;

and the second and third letters the type of survey, as follows:

- (I) for the initial survey;
- (A) for the annual survey;
- (In) for the intermediate survey;
- (P) for the periodical survey;
- (R) for the renewal survey;
- (B) for inspection of the outside of the ship's bottom;
- (Ad) for an additional survey.

Accordingly, "(EI)", "(OIn)" and "(PR)", for example, indicate the initial survey for the Cargo Ship Safety Equipment Certificate, the intermediate survey for the International Oil Pollution Prevention Certificate and the renewal survey for the Passenger Ship Safety Certificate, respectively.

3.9 For the application of these Guidelines, the following guidance on terms used in the survey requirements is provided:

- .1 "Examining", except where used in "examining the plans" or "examining the design", should be understood as an examination, using appropriate techniques, of the components, system or appliance in question for satisfactory provision, arrangement and condition and for any signs of defects, deterioration or damage. The extent of this examination shall be adapted by the surveyor considering the type of survey performed (e.g. initial, annual, renewal) and the actual maintenance condition of the ship and its equipment;
- .2 "Testing" should be understood as a functional test of the system or appliance in question, to confirm its satisfactory operation and performance for its intended use.
- 3.10 The amplification of various terms and conditions is given in section 5.

4 DESCRIPTION OF THE VARIOUS TYPES OF SURVEY

(I) 4.1 Initial surveys

4.1.1 Frequency

The initial survey, as required by the relevant regulations (see 2.8.1), should be held before the ship is put in service, or when a new instrument applies to an existing ship, and the appropriate certificate is issued for the first time.

4.1.2 General

4.1.2.1 The initial survey should include a complete inspection, with tests when necessary, of the structure, machinery and equipment to ensure that the requirements relevant to the particular certificate are complied with and that the structure, machinery and equipment are fit for the service for which the ship is intended.

4.1.2.2 The initial survey should consist of:

- .1 an examination of the plans, diagrams, specifications, calculations and other technical documentation to verify that the structure, machinery and equipment comply with the requirements relevant to the particular certificate;
- .2 an inspection of the structure, machinery and equipment to ensure that the materials, scantlings, construction and arrangements, as appropriate, are in accordance with the approved plans, diagrams, specifications, calculations and other technical documentation and that the workmanship and installation are in all respects satisfactory; and
- .3 a check that all the certificates, record books, operating manuals and other instructions and documentation specified in the requirements relevant to the particular certificate have been placed on board the ship.

4.1.3 Examination of plans and designs

An application for an initial survey should be accompanied by plans and designs referred to in sections 1, 2, 4 and 5 of annex 1 and in annexes 2, 3, 4 and 5, as appropriate, together with:

- .1 the particulars of the ship;
- .2 any exemptions sought; and
- .3 any special conditions.

(A) 4.2 Annual surveys

4.2.1 Frequency

The annual survey, as required by the relevant regulations (see 2.8.5) and as shown diagrammatically in appendix 2, should be held within three months before or after each anniversary date of the certificate.

4.2.2 General

4.2.2.1 An annual survey should enable the Administration to verify that the condition of the ship, its machinery and equipment is being maintained in accordance with the relevant requirements.

4.2.2.2 In general, the scope of the annual survey should be as follows:

- .1 it should consist of a certificate examination, a visual examination of a sufficient extent of the ship and its equipment, and certain tests to confirm that their condition is being properly maintained;
- .2 it should also include a visual examination to confirm that no unapproved modifications have been made to the ship and its equipment;
- .3 the content of each annual survey is given in the respective guidelines; the thoroughness and stringency of the survey should depend upon the condition of the ship and its equipment; and
- .4 should any doubt arise as to the maintenance of the condition of the ship or its equipment, further examination and testing should be conducted as considered necessary.

4.2.3 Where an annual survey has not been carried out within the due dates, reference should be made to 5.6.

(In) 4.3 Intermediate surveys

4.3.1 *Frequency*

The intermediate survey, as required by the relevant regulations (see 2.8.4) and as shown diagrammatically in appendix 2, should be held within three months before or after the second anniversary date or within three months before or after the third anniversary date of the appropriate certificate and should take the place of one of the annual surveys. Items associated with the cargo ship safety construction intermediate survey that are additional to the requirements of the annual survey may, alternatively, be carried out at the second or third annual survey or between these surveys.

4.3.2 General

4.3.2.1 The intermediate survey should be an inspection of items relevant to the particular certificate to ensure that they are in a satisfactory condition and are fit for the service for which the ship is intended.

4.3.2.2 When specifying items of hull and machinery for detailed examination, due account should be taken of any continuous survey schemes that may be applied by classification societies.

4.3.2.3 Where an intermediate survey has not been carried out within the due dates, reference should be made to 5.6.

(P) 4.4 Periodical surveys

4.4.1 Frequency

The periodical survey, as required by the relevant regulations (see 2.8.2) and as shown diagrammatically in appendix 2, should be held within three months before or after the second anniversary date or within three months before or after the third anniversary date in the case of the cargo ship safety equipment certificate and should take the place of one of the annual surveys; in the case of the cargo ship safety radio certificate, it should be held within three months before or after each anniversary date.

4.4.2 General

4.4.2.1 The periodical survey should consist of an inspection, with tests when necessary, of the equipment to ensure that requirements relevant to the particular certificate are complied with and that they are in a satisfactory condition and are fit for the service for which the ship is intended.

4.4.2.2 The periodical survey should also consist of a check that all the certificates, record books, operating manuals and other instructions and documentation specified in the requirements relevant to the particular certificate are on board the ship.

4.4.2.3 Where a periodical survey has not been carried out within the due dates, reference should be made to 5.6.

(R) 4.5 Renewal surveys

4.5.1 Frequency

The renewal survey, as required by the relevant regulations (see 2.8.3) and as shown diagrammatically in the appendix 2, should be held before the appropriate certificate is renewed.

The cargo ship safety construction renewal survey may be commenced at the fourth annual survey and may be progressed during the succeeding year with a view to completion by the fifth anniversary date. The survey items of the fourth annual survey should not be credited to the completion of the renewal survey.

4.5.2 General

4.5.2 1 The renewal survey should consist of an inspection, with tests when necessary, of the structure, machinery and equipment to ensure that the requirements relevant to the particular certificate are complied with and that they are in a satisfactory condition and are fit for the service for which the ship is intended.

4.5.2.2 The renewal survey should also consist of a check that all the certificates, record books, operating manuals and other instructions and documentation specified in the requirements relevant to the particular certificate are on board the ship.

4.5.2.3 Concurrent crediting to both intermediate and renewal safety construction surveys for surveys of spaces should not be acceptable.

(B) 4.6 Inspections of the outside of the ship's bottom of cargo ships

4.6.1 Frequency

There should be a minimum of two inspections of the outside of the ship's bottom during any five-year period (see 5.7), except where SOLAS 74/88 regulation I/14(e) or (f) is applicable. One such inspection should be carried out on or after the fourth annual survey in conjunction with the renewal of the Cargo Ship Safety Construction Certificate or the Cargo Ship Safety Certificate. Where the Cargo Ship Safety Construction Certificate or the Cargo Ship Safety Certificate has been extended under SOLAS 74/88 regulation I/14(e) or (f), this five-year period may be extended to coincide with the validity of the certificate. In all cases the interval between any two such inspections should not exceed 36 months.

4.6.2 General

4.6.2.1 The inspection of the outside of the ship's bottom and the survey of related items (see 5.1) should include an inspection to ensure that they are in a satisfactory condition and fit for the service for which the ship is intended.¹

4.6.2.2 Inspections of the outside of the ship's bottom should normally be carried out with the ship in dry dock. However, consideration may be given to alternate inspections being carried out with the ship afloat. Special consideration should be given before ships of 15 years of age and over other than bulk carriers and oil tankers are permitted to have such surveys afloat. Inspection of the outside of the ship's bottom of bulk carriers and oil tankers of 15 years of age and over should be carried out with the ship in dry dock. Inspections with the ship afloat should only be carried out when the conditions are satisfactory and the proper equipment and suitably trained staff are available. For ships subject to enhanced survey, the provisions of paragraph 2.2.2² of the applicable part of annex A or B of the International Code on the Enhanced Programme of Inspections during Surveys of Bulk Carriers and Oil Tankers, 2011 (2011 ESP Code) adopted by resolution A.1049(27), as amended, should apply.

4.6.3 Where an inspection of the ship's bottom has not been carried out before the due dates, reference should be made to 5.6.

(Ad) 4.7 Additional surveys

4.7.1 Frequency

The additional survey, as required by the relevant regulations (see 2.8.7), should be held:

- 1 when required after an investigation; or
- 2 whenever any important repairs or renewals are made; or
- 3 when ballast water management systems (BWMS) are retrofitted on an existing ship to which an International Ballast Water Management Certificate was previously issued.

¹ Refer to the *Guidelines for pre-planning of surveys in dry dock of ships which are not subject to the enhanced programme of inspections* (MSC.1/Circ.1223).

² "2.2.2 For ships of 15 years of age and over, inspection of the outside of the ship's bottom should be carried out with the ship in dry-dock. For ships less than 15 years of age, alternate inspections of the ship's bottom not conducted in conjunction with the renewal survey may be carried out with the ship afloat. Inspection of the ship afloat should only be carried out when the conditions are satisfactory and the proper equipment and suitably qualified staff is available."

4.7.2 General

Whenever an accident occurs to a ship or a defect is discovered which affects the safety or integrity of the ship or the efficiency or completeness of its equipment, the master or owner should make a report at the earliest opportunity to the Administration, the nominated surveyor or recognized organization responsible for issuing the relevant certificate. The Administration, the nominated surveyor or recognized organization responsible for issuing the relevant certificate should then initiate an investigation to determine whether a survey, as required by the regulations applicable to the particular certificate, is necessary. This additional survey, which may be general or partial according to the circumstances, should be such as to ensure that the repairs and any renewals have been effectively made and that the ship and its equipment continue to be fit for the service for which the ship is intended. In case of installations of BWMS on existing ships the survey shall be such as to ensure that this retrofit, replacement, or significant repair has been effectively made, so that the ship complies with the requirements of the BWM Convention.

4.8 Completion of surveys

If a survey shows that the condition of the ship or its equipment does not correspond 4.8.1 substantially with the particulars of the certificate or is not fit to proceed to sea without danger to the ship, or persons on board, or without presenting unreasonable threat of harm to the environment, the officer of the Administration, nominated surveyor or recognized organization should be guided by the requirements of SOLAS 74/88 regulation I/6(c), MARPOL Annex I regulation 6.3.3, MARPOL Annex II regulation 8.2.5, MARPOL Annex IV regulation 4.5, MARPOL Annex VI regulation 5.3.3, IBC Code regulation 1.5.1.4, IGC Code regulation 1.4.1.4 and BCH Code regulation 1.6.1.3, or in the case of the BWM Convention the survey shows that the ship's ballast water management does not conform to the particulars of the Certificate required under regulations E-2 or E-3, or is such that the ship is not fit to proceed to sea without presenting a threat of harm to the environment, human health, property or resources, the surveyor should be guided by regulation E-1.6. These instruments require that corrective action be taken immediately and the Administration notified in due course. In cases where the corrective action has not been undertaken the relevant certificate should be withdrawn and the Administration notified immediately. If the ship is in the port of another Party, the appropriate authorities of the port State should also be notified immediately.

4.8.2 Although the Polar Code does not contain specific requirements, if the ship is trading in polar waters and a survey shows that the condition of the ship or its equipment does not correspond substantially with the particulars of the Polar Ship Certificate, or the ship is not fit to operate in polar waters without danger to the ship, or persons on board, or without presenting unreasonable threat of harm to the environment, the officer of the Administration, nominated surveyor or recognized organization should, nevertheless, be guided by 4.8.1. The validity of the Polar Ship Certificate does not affect the validity of other certificates.

4.8.3 Although LLC 66/88 does not contain specific requirements, if a load line survey shows that the condition of the ship or its equipment does not correspond substantially with the particulars of the certificate or is not fit to proceed to sea without danger to the ship, or persons on board, or without presenting unreasonable threat of harm to the environment, the officer of the Administration, nominated surveyor or recognized organization should, nevertheless, be guided by 4.8.1.

4.8.4 If a survey shows that the condition of the ship and its equipment correspond substantially with the particulars of the certificate, and the ship is fit to proceed to sea without danger to the ship, or persons on board, and without presenting unreasonable threat of harm to the environment, but deficiencies exist that cannot be rectified at the time of survey, the following guidance should be given:

- .1 a condition should be issued, detailing any relevant requirements or conditions with assigned due date for the time needed to rectify the deficiencies, with relevant information being kept available on board; if and as required by the Administration, the relevant certificates should be issued with the appropriate expiry dates; and
- .2 the Administration should be notified, as appropriate, according to the agreement with the nominated surveyor or the recognized organization.

5 AMPLIFICATION OF TERMS AND CONDITIONS

5.1 Definition of related items

Reference: SOLAS 74/88 regulation I/10(b)(v).

Related items mean those items which may only be inspected when the ship is in dry dock or undergoing an in-water examination of the outside of its bottom. For oil tankers, chemical tankers and gas carriers, this may mean that the ship has to be specially prepared by, for example, being cleaned and gas freed. Then the survey of items such as the internal examination of cargo tanks, as referred to in (Cln) 2.3.2 and (Cln) 2.3.3 in annex 1 may be undertaken at the same time.

5.2 Extending to five years a certificate issued for less than five years

References: SOLAS 74/88 regulation I/14(c), LLC 66/88 article 19(3), MARPOL Annex I regulation 10.3, MARPOL Annex II regulation 10.3, MARPOL Annex IV regulation 8.3, MARPOL Annex VI regulation 9.3, BWM Convention regulations E-5.5 and E-5.6, IBC Code regulation 1.5.6.3, IGC Code regulation 1.4.6.3, BCH Code regulation 1.6.6.3, and Polar Code part I-A /regulation 1.3.6.

Where a certificate has been issued for a period of less than five years, it is permissible under these regulations or article to extend the certificate so that its maximum period of validity is five years provided that the pattern of surveys for a certificate with a five-year period of validity is maintained (see appendix 2). This means that, for example, if a request is made to extend a two-year Cargo Ship Safety Equipment Certificate to five years, then a periodical and two further annual surveys, as detailed in SOLAS 74/88 regulation I/8, would be required. Also, for example, if it was intended to extend a four-year Cargo Ship Safety Construction Certificate to five years, an additional annual survey would be required, as detailed in SOLAS 74/88 regulation I/10. Where a certificate has been so extended, it is still permissible to also extend the certificate under SOLAS 74/88 regulations I/14(e) and (f), LLC 66/88 articles 19(5) and (6), MARPOL Annex I regulations 10.5 and 10.6, MARPOL Annex II regulations 10.5 and 10.6, MARPOL Annex IV regulations 8.5 and 8.6, MARPOL Annex VI regulations 9.5 and 9.6, BWM Convention regulation E-5.2.2, IBC Code regulations 1.5.6.5 and 1.5.6.6, IGC Code regulations 1.4.6.5 and 1.4.6.6, and BCH Code regulations 1.6.6.5 and 1.6.6.6, when no additional surveys would be required but, of course, the new certificate issued after the renewal survey would date from the five-year expiry of the existing certificate, in accordance with SOLAS 74/88 regulation I/14(b)(ii), LLC 66/88 article 19(2)(b), MARPOL Annex I regulation 10.2.2, MARPOL Annex II regulation 10.2.2, MARPOL Annex IV regulation 8.2.2, MARPOL Annex VI regulation 9.2.2, BWM Convention regulation E-5.3, IBC Code regulation 1.5.6.2.2, IGC Code regulation 1.4.6.2.2, BCH Code regulation 1.6.6.2.2 and Polar Code part I-A/regulation 1.3.6.

5.3 Extending the period between inspections of the outside of the ship's bottom

Reference: SOLAS 74/88 regulation I/10(a)(v).

This permits the period of five years in which two inspections of the ship's bottom are to be carried out to be extended when the Cargo Ship Safety Construction Certificate is extended under regulation I/14(e) and (f). However, no extension should be permitted on the period of 36 months between any two such inspections. If the first ship's bottom inspection is carried out between 24 and 27 months, the 36-month limitation may prevent the certificate being extended by the periods permitted in regulation I/14(e) and (f).

5.4 Definition of "short voyage"

References: SOLAS 74/88 regulation I/14(f), LLC 66/88 article 19(6), MARPOL Annex I regulation 10.6, MARPOL Annex II regulation 10.6, MARPOL Annex IV regulation 8.6, MARPOL Annex VI regulation 9.6, BWM Convention regulation E-5.3, IBC Code regulation 1.5.6.6, IGC Code regulation 1.4.6.6, BCH Code regulation 1.6.6.6 and Polar Code part I-A/ regulation 1.3.6.

For the purpose of these regulations or article, a "short voyage" means a voyage where neither the distance from the port in which the voyage begins to the final port of destination nor the return voyage exceeds 1,000 miles.

5.5 Application of "special circumstances"

References: SOLAS 74/88 regulation I/14(g), LLC 66/88 article 19(7), MARPOL Annex I regulation 10.7, MARPOL Annex II regulation 10.7, MARPOL Annex IV regulation 8.7, MARPOL Annex VI regulation 9.7, BWM Convention regulation E-5.7, IBC Code regulation 1.5.6.7, IGC Code regulation 1.4.6.7, BCH Code regulation 1.6.6.7 and Polar Code part I-A/ regulation 1.3.6.

The purpose of these regulations or article is to permit Administrations to waive the requirement that a certificate issued following a renewal survey that is completed after the expiry of the existing certificate should be dated from the expiry date of the existing certificate. The special circumstances when this could be permitted are where the ship has been laid up or has been out of service for a considerable period because of a major repair or modification. While the renewal survey would be as extensive as if the ship had continued in service, the Administration should consider whether additional surveys or examinations are required depending on how long the ship was out of service and the measures taken to protect the hull and machinery during this period. Where this regulation is invoked, it is reasonable to expect an examination of the outside of the ship's bottom to be held at the same time as the renewal survey when it would not be necessary to include any special requirements for cargo ships for the continued application of SOLAS 74/88 regulation I/10(a)(v).

5.6 Revalidation of certificates

References: SOLAS 74/88 regulation I/14(i)(i), LLC 66/88 article 19(9)(c), MARPOL Annex I regulation 10.9.1, MARPOL Annex II regulation 10.9.1, MARPOL Annex IV regulation 8.8.1, MARPOL Annex VI regulation 9.9.1, BWM Convention E-5.9.3, IBC Code regulation 1.5.6.9.1, IGC Code regulation 1.4.6.9.1, BCH Code regulation 1.6.6.9.1 and Polar Code part I-A/regulation 1.3.6.

A certificate ceases to be valid if the periodical, intermediate or annual survey, as appropriate, or the inspection of the outside of the ship's bottom is not completed within the periods specified in the relevant regulation or article. The validity of the certificate should be restored by carrying out the appropriate survey which, in such circumstances, should consist of the requirements of the survey that was not carried out, but its thoroughness and stringency should have regard to the time this survey was allowed to lapse. The Administration concerned should then ascertain why the survey was allowed to lapse and consider further action.

5.7 Meaning of "any five-year period"

Reference: SOLAS 74/88 regulation I/10(a)(v).

Any five-year period is the five-year period of validity of the Cargo Ship Safety Construction Certificate or the Cargo Ship Safety Certificate.

5.8 Surveys required after transfer of the ship to the flag of another State

The certificates cease to be valid when a ship transfers to the flag of another State and it is required that the Government of the State to which the ship transfers should not issue new certificates until it is fully satisfied that the ship is being properly maintained and that there have been no unauthorized changes made to the structure, machinery and equipment. When so requested, the Government of the State whose flag the ship was formerly entitled to fly is obliged to forward, as soon as possible, to the new Administration copies of certificates carried by the ship before the transfer and, if available, copies of the relevant survey reports and records, such as record of safety equipment and conditions of assignment for load line. When fully satisfied by an inspection that the ship is being properly maintained and that there have been no unauthorized changes, in order to maintain the harmonization of the surveys the new Administration may give due recognition to initial and subsequent surveys carried out by, or on behalf of, the former Administration and issue new certificates having the same expiry date as the certificates that ceased to be valid because of the change of flag.

5.9 Recommended conditions for extending the period of validity of a certificate

5.9.1 In SOLAS 74/88 and other mandatory IMO instruments the following provision applies: If a ship at the time when a certificate expires is not in a port in which it is to be surveyed, the Administration may extend the period of validity of a certificate but this extension should be granted only for the purpose of allowing the ship to complete its voyage to the port in which it is to be surveyed, and then only in cases *where it appears proper and reasonable to do so.* No certificate should be extended for a period longer than three months, and a ship to which an extension is granted should not, on its arrival in the port in which is to be surveyed, be entitled by virtue of such extension to leave that port without having a new certificate.

5.9.2 If a ship is in a port where the required survey cannot be completed, and where the Convention allows the Administration to extend the certificate when it is proper and reasonable to do so, the Administration should be guided by the following:

- .1 an additional survey, equivalent to at least the same scope of an annual survey required by the relevant certificate(s) should be carried out;
- .2 the renewal survey should be carried out to the maximum extent possible;
- .3 in cases where a dry-docking is required, but cannot be carried out, an underwater inspection of the ship's bottom should be carried out;

- .4 in cases where an underwater inspection is not possible (e.g. poor water visibility, draught restrictions, excessive current, refusal by the port Authority), an internal inspection of the ship's bottom structure, to the maximum extent practicable, should be carried out;
- .5 the ship should be allowed to sail directly to a named final agreed cargo discharge port and then directly to a named agreed port to complete the survey and/or dry-docking;
- .6 the extension period should be for the minimum amount of time needed to complete the survey and/or dry-docking under the relevant certificate(s);
- .7 the condition of the ship found by the surveys indicated above should be considered in determining the duration, distance and operational restrictions, if any, of the voyage needed to complete the survey and/or dry-docking; and
- .8 the extension period of the relevant statutory certificate(s) should not exceed the period of validity of the certificate which may be issued to document compliance with the structural, mechanical and electrical requirements of the recognized classification society.

5.10 Inspection of the outside of a passenger ship's bottom

5.10.1 A minimum of two of the inspections of the outside of the ship's bottom during any five-year period should be conducted in dry dock. In all cases, the maximum interval between any two dry-dock bottom inspections should not exceed 36 months.

5.10.2 Where acceptable to the Administration, the minimum number of inspections in dry dock of the outside of the bottom of a passenger ship which is not a ro-ro passenger ship in any five-year period may be reduced from two to one.³ In such cases the interval between consecutive inspections in dry dock should not exceed 60 months.

Note: The definition of "any five-year period" is the five-year period of validity of the International Load Line Certificate.

5.10.3 Inspections of the ship's bottom required for the renewal survey that are not conducted in dry dock may be carried out with the ship afloat. The bottom inspection, regardless of method, should be carried out within the allowable time window for the Passenger Ship Safety Certificate renewal survey (i.e. within the three-month time window before the expiry date of the certificate). Additionally, inspections of the outside of the ship's bottom conducted afloat should only be carried out when the conditions are satisfactory and the proper equipment and suitably qualified staff are available. Rudder bearing clearances specified in (PR) 5.2.2.1 need not be taken at the afloat inspections.

5.10.4 Special consideration should be given to ships 15 years of age or over before being permitted to credit inspections afloat.

5.10.5 If a survey in dry dock is not completed within the maximum intervals referred to above, the Passenger Ship Safety Certificate should cease to be valid until the survey in dry dock is completed.

³ Refer to the Guidelines for the assessment of technical provisions for the performance of an in-water survey in lieu of bottom inspection in dry-dock to permit one dry-dock examination in any five-year period for passenger ships other than ro-ro passenger ships (MSC.1/Circ.1348).

5.11 Survey of radio installations

The survey of the radio installations, including those used in life-saving appliances, should always be carried out by a qualified radio surveyor who has necessary knowledge of the requirements of SOLAS 74, the International Telecommunication Union's Radio Regulations and the associated performance standards for radio equipment. The radio survey should be carried out using suitable test equipment capable of performing all the relevant measurements required by these Guidelines. On satisfactory completion of the survey, the radio surveyor should forward a report of the survey, which should also state the organization he or she represents, to the authorities responsible for the issue of the ship's Cargo Ship Safety Radio Certificate or Passenger Ship Safety Certificate.

5.12 Survey of the automatic identification system (AIS)

The survey of the automatic identification system should always be carried out by a qualified radio surveyor who has necessary knowledge of the requirements of SOLAS 74, the International Telecommunication Union's Radio Regulations and the associated performance standards for radio equipment. The survey of the automatic identification system should be carried out using suitable test equipment capable of performing all the relevant measurements required by and in accordance with the *Guidelines on annual testing of the automatic identification system (AIS)* (MSC.1/Circ.1252).

5.13 Surveys for ships intended to operate in polar waters4

5.13.1 In accordance with SOLAS 74/88 regulations XIV/2.1 and 3.1, MARPOL Annex I regulation 47, MARPOL Annex II regulation 22, MARPOL Annex IV regulation 18 and MARPOL Annex V regulation 14, the Polar Code is a standalone instrument, providing requirements additional to SOLAS 74/88 and MARPOL for ships intended to operate in polar waters. The requirements of the Polar Code should be surveyed in the context of the surveys under SOLAS 74/88 and MARPOL, but do not form separate survey types.

5.13.2 For MARPOL Annexes I and II,⁵ compliance with the Polar Code should be indicated on the International Oil Pollution Prevention Certificate and, where applicable, the International Pollution Prevention Certificate for the Carriage of Noxious Liquid Substances in Bulk. For SOLAS 74/88, the Polar Ship Certificate should be issued (Polar Code, paragraph 1.3) and endorsed for compliance with the Polar Code. The Polar Ship Certificate should be perceived as a certificate additional to SOLAS certificates for ships intended to operate in polar waters, and the validities of other certificates are not affected by it when a ship stays outside of polar areas.

5.13.3 Although there is no dedicated type of survey associated with the Polar Ship Certificate, the following types of survey are applicable in the Survey Guidelines in annex 5 and contain survey items specific to the Polar Ship Certificate:

.1 the initial survey would confirm the survey scopes of the items related to part I-A of the Polar Code with respect to the initial surveys of safety construction, safety equipment and safety radio of cargo ships or the initial survey of the Passenger Ship Safety Certificate;

⁴ Refer to the Unified interpretations of SOLAS regulation XIV/2.2 and paragraphs 1.3.2 and 1.3.6, part I-A of the Polar Code (MSC.1/Circ.1562).

⁵ Refer to the *Guidance for issuing revised certificates, manuals and record books under Annexes I, II and V of MARPOL for compliance with environment-related requirements of the Polar Code* (MEPC.1/Circ.856 and Corr.1).

- .2 the annual survey would confirm the survey scopes of the items related to part I-A of the Polar Code with respect to the periodical safety radio survey and the annual safety construction and safety equipment surveys of cargo ships;
- .3 the intermediate survey would confirm the survey scopes of the items related to part I-A of the Polar Code with respect to the intermediate safety construction survey of cargo ships;
- .4 the periodical survey (second or third year of its validity) would confirm the survey scopes of the items related to part I-A of the Polar Code with respect to the periodical safety equipment survey of cargo ships; and
- .5 the renewal survey would confirm the survey scopes of the items related to part I-A of the Polar Code with respect to the renewal surveys of safety construction, safety equipment and safety radio of cargo ships or the renewal survey of the Passenger Ship Safety Certificate.

5.13.4 For a Polar Ship Certificate issued to a cargo ship, the endorsements of the certificate for annual survey, intermediate survey and periodical survey would confirm the satisfactory completion of the survey scopes as stated respectively in subparagraphs 5.13.3.2 to 5.13.3.4.

5.13.5 For a Polar Ship Certificate issued to a passenger ship, only the endorsements in accordance with SOLAS 74/88 regulations I/14(d) and I/14(e) or I/14(f) are applicable.

5.13.6 After the relevant surveys under the related SOLAS certificates together with the survey items relevant to the Polar Code are carried out, the relevant SOLAS certificates should then be endorsed prior to the Polar Ship Certificate being issued/endorsed.

5.13.7 In the event that the Polar Ship Certificate has ceased to be valid or has expired for a period of time and other certificates remain valid, the Administration may require, as deemed appropriate, a survey which addresses only the survey items additional to the survey items under the related SOLAS certificates for reinstating the validity of the Polar Ship Certificate. The duration of the related SOLAS certificates should remain unchanged.

Annex 1

SURVEY GUIDELINES UNDER THE 1974 SOLAS CONVENTION, AS MODIFIED BY THE 1988 PROTOCOL RELATING THERETO

- (E) 1 GUIDELINES FOR SURVEYS FOR THE CARGO SHIP SAFETY EQUIPMENT CERTIFICATE
- (EI) **1.1** Initial surveys see part "General" section 4.1.
- (EI) 1.1.1 For the life-saving appliances and the other equipment of cargo ships the examination of plans and designs should consist of:
- (EI) 1.1.1.1 examining the plans for the fire pumps including the emergency fire pump,⁶ if applicable, fire mains, hydrants, hoses and nozzles and the international shore connection (SOLAS 74/00/14 regs.II-2/10.2 and 10.4.4 and FSS Code chs.2 and 12);
- (EI) 1.1.1.2 checking the provision, specification and arrangements of the fire extinguishers (SOLAS 74/00 reg.II-2/10.3) (SOLAS 74/88 reg.II-2/6);
- (EI) 1.1.1.3 checking the provision, specification and arrangements of the fire-fighters' outfits including their self-contained compressed air breathing apparatus, emergency escape breathing devices (EEBDs), onboard means of recharging breathing apparatus cylinders used during drills or a suitable number of spare cylinders to replace those used, and provision of two-way portable radiotelephone apparatus of an explosion-proof type or intrinsically safe (SOLAS 74/00/12 regs.II-2/10.10, 13.3.4, 13.4.3 and 15.2.2; FSS Code ch.3) (SOLAS 74/88 reg.II-2/17) (BCH Code, ch.III, part E);
- (EI) 1.1.1.4 examining the plans for the fire-extinguishing arrangements in the machinery spaces (SOLAS 74/00/12/14/16 regs.II-2/10.4 and 10.5 (except 10.5.5); FSS Code chs.5, 6 and 7) (SOLAS 74/88 reg.II-2/7);
- (EI) 1.1.1.5 examining the plans for the special arrangements in the machinery spaces (SOLAS 74/00 regs.II-2/5.2, 8.3 and 9.5) (SOLAS 74/88 reg.II-2/11);
- (EI) 1.1.1.6 checking the provision of a fixed fire detection and fire alarm system for machinery spaces including periodically unattended machinery spaces and enclosed spaces containing incinerators (SOLAS 74/00/10 regs.II-2/7.2, 7.3 and 7.4; FSS Code ch.9) (SOLAS 74/88 regs.II-2/13 and 14);
- (EI) 1.1.1.7 checking the provision of a fixed fire detection and fire alarm system and/or a sprinkler, fire detection and fire alarm system in accommodation and service spaces and control stations (SOLAS 74/00 regs.II-2/7.2, 7.3, 7.5.5, 7.7 and 10.6.2; FSS Code chs.8 and 9) (SOLAS 74/88 reg.II-2/52);

⁶ Refer to the Unified interpretation of chapter 12 of the International Code for Fire Safety Systems (MSC.1/Circ.1388).

- (EI) 1.1.1.8 checking the provision of a fire-extinguishing system for spaces containing paint and/or flammable liquids and deep-fat cooking equipment in accommodation and service spaces (SOLAS 74/00 regs.II-2/10.6.3 and 10.6.4; FSS Code chs.5 and 7) (SOLAS 74/88 regs.II-2/18.7) (BCH Code ch.III, part E);
- (EI) 1.1.1.9 examining the arrangements for remote closing of valves for oil fuel, lubricating oil and other flammable oils (SOLAS 74/00 reg.II-2/4.2.2.3.4) (SOLAS 74/88 reg.II-2/15.2.5);
- (EI) 1.1.1.10 examining the plans for the fire protection arrangements in cargo spaces for general cargo and dangerous goods (SOLAS 74/00/14 regs.II-2/10.7.1, 10.7.2 and 19) (SOLAS 74/88 regs.II-2/53 and 54);
- (EI) 1.1.1.11 examining the plans for the fire protection arrangements for ships designed to carry containers on or above the weather deck, as applicable, including provision of the water mist lance, and as appropriate, mobile water monitors and all necessary hoses, fittings and required fixing hardware together with the requirements additional to fire fire mains, fire hoses fire hydrants pumps, and (SOLAS 74/00/14 reg.II 2/10.7.3);
- (EI) 1.1.1.12 examining the plans for the fire protection arrangements in vehicle, special category and ro-ro spaces, including the fire safety arrangements for vehicle carriers carrying motor vehicles with compressed hydrogen or natural gas in their tanks for their own propulsion as cargo, as applicable (SOLAS 74/00/14 reg.II-2/20 (except 20.2.2 and 20.5) and 20-1; FSS Code chs.5, 6, 7, 9 and 10) (SOLAS 74/88 regs.II-2/37, 38 and 53);
- (EI) 1.1.1.13 checking navigation bridge visibility (SOLAS 74/00 reg.V/22);
- (EI) 1.1.1.14 examining the plans for the helicopter facilities, including foam firefighting appliances when appropriate (SOLAS 74/00/16 reg.II-2/18; FSS Code ch.17) (SOLAS 74/88 reg.II-2/18.8);
- (EI) 1.1.1.15 examining the plans for the special arrangements for the carriage of dangerous goods, when appropriate, including water supplies, electrical equipment and wiring, fire detection including sample extraction smoke detection systems, where applicable, ventilation, bilge pumping, personnel protection and any water spray system (SOLAS 74/00 reg.II-2/19 (except 19.3.8, 19.3.10 and 19.4); FSS Code chs.9 and 10) (SOLAS 74/88 reg.II-2/54);
- (EI) 1.1.1.16 examining the provision and disposition of the survival craft and rescue boats and, where applicable, marine evacuation systems (MESs) (SOLAS 74/88 regs.III/11 to 16, 31 and 33);
- (EI) 1.1.1.17 examining, where applicable, the approved documentation for the alternative design and arrangements (SOLAS 74/00/06 regs.II-2/17 and III/38);

- (EI) 1.1.1.18 examining the design of the survival craft, including their construction equipment, fittings, release mechanisms and recovery appliances and embarkation and launching arrangements (SOLAS 74/96/06/11 regs.III/4,16, 31, 32 to 33; LSA Code sections 3.2, 4.1 to 4.9, 6.1 and 6.2);
- (EI) 1.1.1.19 checking that the life-saving appliances are of an international or vivid reddish orange, or a comparably highly visible colour on all parts where this will assist detection at sea (LSA Code section 1.2.2.6);
- (EI) 1.1.1.20 examining the design of the rescue boats, including their equipment and launching and recovery appliances and arrangements (SOLAS 74/00 regs.III/17 and 31; LSA Code sections 5.1 and 6.1);
- (EI) 1.1.1.21 examining the provision, specification and stowage of two-way VHF radiotelephone apparatus and search and rescue locating devices (SOLAS 74/88/08 reg.III/6);
- (EI) 1.1.1.22 examining the provision, specification and stowage of the distress flares and the line-throwing appliance and the provision of onboard communications equipment and the general alarm system (SOLAS 74/00 regs.II-2/12.1 and 12.2, and regs.III/6 and 18; and LSA Code sections 3.1, 7.1 and 7.2);
- (EI) 1.1.1.23 examining the provision, specification and stowage of the lifebuoys, including those fitted with self-igniting lights, self-activating smoke signals and buoyant lines, lifejackets,⁷ immersion suits and anti-exposure suits (SOLAS 74/00/06 regs.III/7 and 32; LSA Code sections 2.1 to 2.5 and 3.1 to 3.3);
- (EI) 1.1.1.24 checking that immersion suits designed to be worn in conjunction with a lifejacket are suitably marked (LSA Code section 2.3.1);
- (EI) 1.1.1.25 examining the plans for the lighting of the muster and embarkation stations and the alleyways, stairways and exits giving access to the muster and embarkation stations, including the supply from the emergency source of power (SOLAS 74/88 regs.II-1/43 and III/11);
- (EI) 1.1.1.26 examining the plans for the positioning of, and the specification for, the navigation lights, shapes and sound signalling equipment (COLREG 1972, rules 20 to 24, 27 to 30 and 33);
- (EI) 1.1.1.27 examining the plans relating to the bridge design and arrangement of navigational systems and equipment and bridge procedures (SOLAS 74/00 regs.V/15 and 19);
- (EI) 1.1.1.28 checking the provision and specification of the following navigation equipment as appropriate: daylight signalling lamp, magnetic compass, transmitting heading device, gyro compass, gyro compass repeaters, radar installation(s), automatic identification system, electronic plotting aid, automatic tracking aid(s) or automatic radar plotting aid(s), echo-sounding device, speed and distance measuring device(s), rudder angle indicator, propeller rate of revolution indicator, variable-pitch

⁷ SOLAS regulation III/7.2.1.5 should be considered.

propeller pitch and operational mode indicator, rate-of-turn indicator, heading or track control system, GNSS receiver, terrestrial radio navigation system and sound reception system, means of communication with emergency steering position, a pelorus or compass bearing device, means for correcting heading and bearings, a bridge navigational watch alarm system (BNWAS) as applicable and an electronic chart display and information system (ECDIS) including backup arrangements as applicable (SOLAS 74/00/09/13 reg.V/19);

- (EI) 1.1.1.29 checking the provision and specification of voyage data recorder (SOLAS 74/00 reg.V/20);
- (EI) 1.1.1.30 checking the provision and specification of the long-range identification and tracking system (SOLAS 74/04 reg.V/19-1);
- (EI) 1.1.1.31 checking the plans and specification for the pilot transfer arrangement, the pilot ladders, the combination arrangements, where applicable, the access to the ship's deck and the associated equipment and lighting (SOLAS 74/88/10 reg.V/23); and
- (EI) 1.1.1.32 checking the provision of means of embarkation and disembarkation from ships for use in port and in port-related operations, such as gangways and accommodation ladders (SOLAS 74/08 reg.II-1/3-9).
- (EI) 1.1.2 For the examination of plans and designs of the life-saving appliances and the other equipment of cargo ships, the additional requirements for tankers should consist of:
- (EI) 1.1.2.1 examining the plans for the cargo tank protection (SOLAS 74/00/15 regs.II-2/4.5.3, 4.5.5, 4.5.6 and 10.8; FSS Code chs.14 and 15) (SOLAS 74/88 regs.II-2/60 and 62);
- (EI) 1.1.2.2 examining the plans for gas measurement in double hull spaces and double bottom spaces, including the fitting of permanent gas sampling lines, where appropriate (SOLAS 74/10 reg.II-2/4.5.7.2);
- (EI) 1.1.2.3 examining, for oil tankers of 20,000 tonnes deadweight and above, the plans for the fixed hydrocarbon gas detection system for measuring hydrocarbon gas concentrations in all ballast tanks and void spaces of double hull and double bottom spaces adjacent to the cargo tanks, including the forepeak tank and any other tanks and spaces under the bulkhead deck adjacent to cargo tanks (SOLAS 74/10 reg.II-2/4.5.7.3 and FSS Code ch.16); and
- (EI) 1.1.2.4 examining the plans for protection of the cargo pump-rooms (SOLAS 78/00 regs.II-2/4.5.10 and 10.9) (SOLAS 74/88 reg.II-2/63).
- (EI) 1.1.3 For the examination of plans and designs of the life-saving appliances and the other equipment of cargo ships using natural gas as fuel other than ships covered by the IGC Code, the additional requirements should consist of:
- (EI) 1.1.3.1 examining the plans, for the fire detection and alarm system and fire-fighting arrangements (IGF Code paras.11.4, 11.5, 11.6 and 11.7).

- (EI) 1.1.4 For the life-saving appliances and the other equipment of cargo ships the survey during construction and after installation should consist of:
- (EI) 1.1.4.1 examining the fire pumps and fire main and the disposition of the hydrants, hoses and nozzles and the international shore connection and checking that each fire pump, including the emergency fire pump, can be operated separately so that two jets of water are produced simultaneously from different hydrants at any part of the ship while the required pressure is maintained in the fire main; and testing that the emergency fire pump has the required capacity, and if the emergency fire pump is the main supply of water for any fixed fire-extinguishing system, checking that the emergency fire pump has the capacity for this system⁸ (SOLAS 74/00/14 reg.II-2/10.2; FSS Code chs.2 and 12) (SOLAS 74/88 regs.II-2/4 and 19);
- (EI) 1.1.4.2 for ships designed to carry containers on or above the weather deck, as applicable, testing the water mist lance, and as appropriate, the mobile water monitors and examining all necessary hoses, fittings and required fixing hardware, and testing that the mobile water monitors are capable of being securely fixed to the ship structure ensuring safe and effective operation, and testing that the mobile water monitor jets reach the top tier of containers with all required monitors and water jets from fire hoses operated simultaneously (SOLAS 74/00/14 reg.II-2/10.7.3);
- (EI) 1.1.4.3 examining the provision and disposition of the fire extinguishers (SOLAS 74/00 reg.II-2/10.3; FSS Code ch.4) (SOLAS 74/88 reg.II-2/17);
- (EI) 1.1.4.4 examining the fire-fighters' outfits including their self-contained compressed air breathing apparatus, emergency escape breathing devices (EEBDs), onboard means of recharging breathing apparatus cylinders used during drills or the provision of a suitable number of spare cylinders to replace those used, and provision of two-way portable radiotelephone apparatus of an explosion-proof type or intrinsically safe (SOLAS 74/00/12 regs.II-2/10.10, 13.3.4, 13.4.3 and 15.2.2; FSS Code ch.3) (SOLAS 74/88 reg.II-2/17) (BCH Code ch.III, part E);
- (EI) 1.1.4.5 checking the operational readiness and maintenance of fire-fighting systems (SOLAS 74/00 reg.II-2/14.1) (SOLAS 74/88 reg.II-2/21);
- (EI) 1.1.4.6 examining the fixed fire-fighting system for the machinery, cargo, vehicle, special category and ro-ro spaces, as appropriate, and confirming that the installation tests have been satisfactorily completed means operation and that its of are clearly marked (SOLAS 74/00/08/12/14 regs.II-2/10.4, 10.5, 10.7.1, 10.7.2 and 20.6.1; FSS Code chs.5 to 7) (SOLAS 74/88 regs.II-2/7 and 53);
- (EI) 1.1.4.7 checking that fixed carbon dioxide fire-extinguishing systems for the protection of machinery spaces and cargo pump-rooms, where applicable, are provided with two separate controls, one for opening of the gas piping and one for discharging the gas from the storage container, each of them located in a release box clearly identified for the particular space (SOLAS 74/08 reg.II-2/10.4.1.5);

⁸ Refer to the Unified interpretation of chapter 12 of the International Code for Fire Safety Systems (MSC.1/Circ.1388).

- (EI) 1.1.4.8 examining the fire-extinguishing and special arrangements in the machinery spaces and confirming, as far as practicable and as appropriate, the operation of the remote means of control provided for the opening and closing of the skylights, the release of smoke, the closure of the funnel and ventilation openings, the closure of power-operated and other doors, the stopping of ventilation and boiler forced and induced draught fans and the stopping of oil fuel and other pumps that discharge flammable liquids (SOLAS 74/00/12/14/16 regs.II-2/5.2, 8.3, 9.5 and 10.5) (SOLAS 74/88 regs.II-2/7 and 11);
- (EI) 1.1.4.9 examining any fixed fire detection and alarm system and any automatic sprinkler, fire detection and fire alarm system, and any sample extraction smoke detection system, and confirming that installation tests have been satisfactorily completed (SOLAS 74/00/10 regs.II-2/7.2, 7.3, 7.4, 7.5.1, 7.5.5, 19.3.3 and 20.4; FSS Code chs.8, 9 and 10) (SOLAS 74/88 regs.II-2/11, 13, 14, 53 and 54);
- (EI) 1.1.4.10 examining the fire-extinguishing system for spaces containing paint and/or flammable liquids and deep-fat cooking equipment in accommodation and service spaces and confirming that installation tests have been satisfactorily completed and that its means of operation are clearly marked (SOLAS 74/00regs.II-2/10.6.3 and 10.6.4; FSS Code chs.4 to 7) (SOLAS 74/88 reg.II-2/18.7) (BCH Code ch.III, part E);
- (EI) 1.1.4.11 examining the arrangements for oil fuel, lubricating oil and other flammable oils and testing the remote closing of valves for oil fuel, lubricating oil and other flammable oils and the operation of the remote means of closing the valves on the tanks that contain oil fuel, lubricating oil and other flammable oils (SOLAS 74/00 reg.II-2/4.2.2.3.4) (SOLAS 74/88 reg.II-2/15.2.5);
- (EI) 1.1.4.12 examining the fire protection arrangements in cargo vehicle and ro-ro spaces, including the fire safety arrangements for vehicle carriers carrying motor vehicles with compressed hydrogen or natural gas in their tanks for their own propulsion as cargo, as applicable, and confirming, as far as practicable and as appropriate, the operation of the means for closing the various openings (SOLAS 74/00/14 regs.II-2/10.7.1, 10.7.2, 20.2.1, 20.3, 20.6.2, 20-1.2.1, 20-1.3, and 20-1.4) (SOLAS 74/88 reg.II-2/53);
- EI) 1.1.4.13 examining the portable gas detectors suitable for the detection of the gas fuel, for vehicle carriers carrying motor vehicles with compressed hydrogen or natural gas in their tanks for their own propulsion as cargo (SOLAS 74/14 regs.II-2/20-1.2.1 and 20-1.5);
- (EI) 1.1.4.14 examining, where applicable, the alternative design and arrangements for fire safety or life-saving appliances and arrangements, in accordance with the test and inspection requirements, if any, specified in the approved documentation (SOLAS 74/00/06 regs.II-2/17 and III/38);
- (EI) 1.1.4.15 examining the helicopter facilities, including foam firefighting appliances when appropriate (SOLAS 74/00/16 reg.II-2/18; FSS Code ch.17) (SOLAS 74/88 reg.II-2/18.8);
- (EI) 1.1.4.16 examining, when appropriate, the special arrangements for carrying dangerous goods, including checking the electrical equipment and wiring, the ventilation, the provision of protective clothing and portable

appliances and the testing of the water supply, bilge pumping and any water spray system (SOLAS 74/00/08 reg.II-2/19 (except 19.3.8, 19.3.10 and 19.4); FSS Code chs.9 and 10) (SOLAS 74/88 reg.II-2/54);

- (EI) 1.1.4.17 checking that the life-saving appliances are of international or vivid reddish orange, or a comparably highly visible colour on all parts where this will assist detection at sea (LSA Code section 1.2.2.6);
- (EI) 1.1.4.18 checking the provision and disposition of the survival craft, where applicable, marine evacuation systems and rescue boats (SOLAS 74/88 regs.III/11 to 16 and 31; LSA Code section 6.2);
- (EI) 1.1.4.19 deployment of 50% of the MES after installation (LSA Code paragraph 6.2.2.2);
- (EI) 1.1.4.20 examining each survival craft, including its equipment; for liferafts provided for easy side-to-side transfer, verifying that they are less than 185 kg (SOLAS 74/88 reg.III/31; LSA Code sections 2.5, 3.1 to 3.3 and 4.1 to 4.9) (SOLAS 74/00 reg.III/31.1);
- (EI) 1.1.4.21 examining the embarkation arrangements for each survival craft and the testing of each launching appliance, including overload tests, tests to establish the lowering speed and the lowering of each survival craft to the water with the ship at its lightest seagoing draught, and, where applicable, launching underway at 5 knots, checking the recovery of each lifeboat (SOLAS 74/00 regs.III/11, 12, 13, 16, 31 and 33; LSA Code section 6.1);
- (EI) 1.1.4.22 examining the embarkation arrangements for each marine evacuation device, where applicable, and the launching arrangements, including inspection for lack of side shell opening between the embarkation station and waterline, review of distance to the propeller and other life-saving appliances and ensuring that the stowed position is protected from heavy weather damage, as much as practicable (SOLAS 74/00 reg.III/15; LSA Code section 6.2);
- (EI) 1.1.4.23 examining each rescue boat, including its equipment; for inflatable rescue boats, confirming that they are stowed in a fully inflated condition (SOLAS 74/88 regs.III/14 and 31; LSA Code sections 2.5, 5.1 and 6.1);
- (EI) 1.1.4.24 examining the embarkation and recovery arrangements for each rescue boat and testing each launching and recovery appliance, including overload tests, tests to establish the lowering and recovery speeds and ensuring that each rescue boat can be lowered to the water and recovered with the ship at its lightest seagoing draught, launching underway at 5 knots (SOLAS 74/88 regs.III/14, 17 and 31; LSA Code section 6.1);
- (EI) 1.1.4.25 testing that the engine of the rescue boat(s) and of each lifeboat, when so fitted, start satisfactorily and operate both ahead and astern (SOLAS 74/00 reg.III/19);

- (EI) 1.1.4.26 confirming that there are posters or signs in the vicinity of survival craft and their launching stations and containers, brackets, racks and other similar stowage locations for life-saving equipment (SOLAS 74/88 regs.III/9 and 20);
- (EI) 1.1.4.27 examining the provision and stowage and checking the operation of portable onboard communications equipment, if provided, and two-way VHF radiotelephone apparatus⁹ and search and rescue locating devices (SOLAS 74/88/08 regs.II-2/12.2 and III/6);
- (EI) 1.1.4.28 examining the provision and stowage of the distress flares and the line-throwing appliance, checking the provision and operation of fixed onboard communications equipment, if provided, and testing the means of operation of the general alarm system (SOLAS 74/00 regs.III/6 and 18; LSA Code sections 3.1, 7.1 and 7.2);
- (EI) 1.1.4.29 examining the provision, disposition and stowage of the lifebuoys, including those fitted with self-igniting lights, self-activating smoke signals and buoyant lines, lifejackets,¹⁰ immersion suits and anti-exposure suits (SOLAS 74/00/06 regs.III/7 and 32; LSA Code sections 2.1 to 2.5 and 3.1 to 3.3);
- (EI) 1.1.4.30 checking the lighting of the muster and embarkation stations and the alleyways, stairways and exits giving access to the muster and embarkation stations, including when supplied from the emergency source of power (SOLAS 74/88 regs.II-1/43 and III/11);
- (EI) 1.1.4.31 examining the provision and positioning and checking the operation of, as appropriate, the navigation lights, shapes and sound signalling equipment (COLREG 1972, rules 20 to 24, 27 to 30 and 33);
- (EI) 1.1.4.32 checking that the minimum safe distances from the steering and standard magnetic compasses for all electrical equipment are complied with (SOLAS 74/00 regs.V/17 and 19);
- (EI) 1.1.4.33 checking the electromagnetic compatibility of electrical and electronic equipment on or in the vicinity of the bridge (SOLAS 74/00 reg.V/17);
- (EI) 1.1.4.34 checking, as appropriate, the provision and operation of the following shipborne navigational systems equipment (SOLAS 74/00 reg.V/19):
- (EI) 1.1.4.34.1 the magnetic compass, including examining the siting, movement and illumination and a pelorus or compass bearing device (SOLAS 74/00 reg.V/19);
- (EI) 1.1.4.34.2 nautical charts and nautical publications necessary for the intended voyage are available and have been updated, and, where ECDIS is used, the electronic charts have been updated and the required backup system is provided and updated (SOLAS 74/00/09 reg.V/19);

⁹ Expired primary batteries, which complied with the requirements of paragraph 12.6 of *Revised Recommendation on performance standards for survival craft portable Two-Way VHF radiotelephone apparatus* (resolution MSC.149 (77)), could be used for no other purpose than to examine and check the operation of two-way VHF radiotelephone apparatus.

¹⁰ SOLAS regulation III/7.2.1.5 should be considered.

- (EI) 1.1.4.34.3 global navigation satellite system receiver or terrestrial radionavigation system;
- (EI) 1.1.4.34.4 sound-reception system, when bridge is totally enclosed;
- (EI) 1.1.4.34.5 means of communication to emergency steering position, where provided;
- (EI) 1.1.4.34.6 spare magnetic compass;
- (EI) 1.1.4.34.7 daylight signalling lamp;
- (EI) 1.1.4.34.8 echo-sounding device;
- (EI) 1.1.4.34.9 radar(s), including examining the waveguide and cable runs for routeing and protection and the display unit confirming lighting, correct operation of all controls, and functions;
- (EI) 1.1.4.34.10 electronic plotting aid, automatic tracking aid or automatic radar plotting aid as appropriate, using the appropriate test facilities;
- (EI) 1.1.4.34.11 speed and distance measuring devices "through the water" and "over the ground";
- (EI) 1.1.4.34.12 transmitting heading device providing heading information to radar, plotting aids and automatic identification system equipment;
- (EI) 1.1.4.34.13 automatic identification system;
- (EI) 1.1.4.34.14 gyrocompass, including examining the alignment of the master and all repeaters;
- (EI) 1.1.4.34.15 rudder angle indicator;
- (EI) 1.1.4.34.16 propeller rate of revolution indicator;
- (EI) 1.1.4.34.17 propeller, operational mode, thrust, and pitch indicator;
- (EI) 1.1.4.34.18 rate-of-turn indicator;
- (EI) 1.1.4.34.19 heading or track control system;
- (EI) 1.1.4.34.20 BNWAS;
- (EI) 1.1.4.35 checking for the provision and operation of the voyage data recorder (SOLAS 74/00 reg.V/20);
- (EI) 1.1.4.36 checking the record of the voyage data recorder annual performance test (SOLAS 74/00 reg.V/18);
- (EI) 1.1.4.37 checking navigation bridge visibility (SOLAS 74/00 reg.V/22);
- (EI) 1.1.4.38 checking that a valid conformance test report of the long-range identification and tracking system is available on board (SOLAS 74/04 reg.V/19-1);

- (EI) 1.1.4.39 checking the provision of the pilot transfer arrangement, the access to the ship's deck and the associated equipment and lighting, checking the operation of the pilot ladders and the combination arrangements (SOLAS 74/00/10 reg.V/23);
- (EI) 1.1.4.40 checking the provision of means of embarkation and disembarkation from ships for use in port and in port-related operations, such as gangways and accommodation ladders (SOLAS 74/08 reg.II-1/3-9); and checking, when appropriate, the provision of an appropriate instrument for measuring the concentration of gas or oxygen in the air together with detailed instructions for its use (SOLAS 74/08 reg.VI/3).
- (EI) 1.1.5 For the life-saving appliances and the other equipment of cargo ships for the additional requirements for tankers the survey during construction and after installation should consist of:
- (EI) 1.1.5.1 checking the deck foam system, including the supplies of foam concentrate, and testing that the minimum number of jets of water at the required pressure in the fire main is obtained (see (EI) 1.1.3.1) when the system is in operation (SOLAS 74/00 reg.II-2/10.8; FSS Code ch.15) (SOLAS 74/88 reg.II-2/61);
- (EI) 1.1.5.2 examining the inert gas system (SOLAS 74/00/14 reg.II-2/4.5.5; FSS Code ch.15) (SOLAS 74/88 reg.II-2/62) and in particular:
- (EI) 1.1.5.2.1 examining externally for any sign of gas or effluent leakage;
- (EI) 1.1.5.2.2 confirming the proper operation of both inert gas blowers;
- (EI) 1.1.5.2.3 observing the operation of the scrubber-room ventilation system;
- (EI) 1.1.5.2.4 checking the deck water seal for automatic filling and draining, and the arrangements for protecting the system against freezing;
- (EI) 1.1.5.2.5 where a double block and bleed valve is installed, checking the automatic operations of the block and the bleed valves upon loss of power;
- (EI) 1.1.5.2.6 where two shut-off valves in series with a venting valve in between are used as non-return devices, checking the automatic operation of the venting valve, and the alarm for faulty operation of the valves;
- (EI) 1.1.5.2.7 examining the operation of all remotely operated or automatically controlled valves and, in particular, the flue gas isolating valves;
- (EI) 1.1.5.2.8 observing a test of the interlocking feature of soot blowers;
- (EI) 1.1.5.2.9 observing that the gas pressure-regulating valve automatically closes when the inert gas blowers are secured;
- (EI) 1.1.5.2.10 checking the means for separating the cargo tank not being inerted from the inert gas main;

- (EI) 1.1.5.2.11 checking the alarms of the two oxygen sensors positioned in the space or spaces containing the inert gas system;
- (EI) 1.1.5.2.12 checking, as far as practicable, the following alarms and safety devices of the inert gas system using simulated conditions where necessary:
- (EI) 1.1.5.2.12.1 high oxygen content of gas in the inert gas main;
- (EI) 1.1.5.2.12.2 low gas pressure in the inert gas main;
- (EI) 1.1.5.2.12.3 low pressure in the supply to the deck water seal;
- (EI) 1.1.5.2.12.4 high temperature of gas in the inert gas main;
- (EI) 1.1.5.2.12.5 low water pressure or low water-flow rate;
- (EI) 1.1.5.2.12.6 accuracy of portable and fixed oxygen-measuring equipment by means of calibration gas;
- (EI) 1.1.5.2.12.7 high water level in the scrubber;
- (EI) 1.1.5.2.12.8 failure of the inert gas blowers;
- (EI) 1.1.5.2.12.9 failure of the power supply to the automatic control system for the gas regulating valve and to the instrumentation for continuous indication and permanent recording of pressure and oxygen content in the inert gas main;
- (EI) 1.1.5.2.12.10 high pressure of gas in the inert gas main;
- (EI) 1.1.5.2.13 checking the proper operation of the inert gas system on completion of the checks listed above;
- (EI) 1.1.5.3 examining the fixed fire-fighting system for the cargo pump-room, confirming that the installation tests have been satisfactorily completed and that its means of operation are clearly marked (SOLAS 74/00 reg.II-2/10.9; FSS Code chs.5, 6, 7 and 8, as applicable) and, when appropriate, checking the operation of the remote means for closing the various openings;
- (EI) 1.1.5.4 examining the protection of the cargo pump-rooms and confirming that the installation tests have been satisfactorily completed (SOLAS 74/00 reg.II-2/4.5.10) (SOLAS 74/88 regs.II-2/55 to 58);
- (EI) 1.1.5.5 examining, for all tankers, the arrangements for cargo tank protection, (SOLAS 74/00/10/15 regs.II-2/4.5.3, 4.5.6 and 10.8; FSS Code chs.14 and 15) (SOLAS 74/88 regs.II-2/60 and 62);
- (EI) 1.1.5.6 checking, for all tankers, the provision of at least one portable instrument for measuring oxygen and one for measuring flammable vapour concentrations, together with a sufficient set of spares, and suitable means for the calibration of these instruments (SOLAS 74/10 reg.II-2/4.5.7.1);
- (EI) 1.1.5.7 examining the arrangements for gas measurement in double hull spaces and double bottom spaces, including the fitting of permanent gas sampling lines, where appropriate (SOLAS 74/10 reg.II-2/4.5.7.2); and

- (EI) 1.1.5.8 examining, for oil tankers of 20,000 tonnes deadweight and above, the fixed hydrocarbon gas detection system for measuring hydrocarbon gas concentrations in all ballast tanks and void spaces of double hull and double bottom spaces adjacent to the cargo tanks, including the forepeak tank and any other tanks and spaces under the bulkhead deck adjacent to cargo tanks, and confirming that the installation tests have been satisfactorily completed (SOLAS 74/10 reg.II-2/4.5.7.3 and FSS Code ch.16).
- (EI) 1.1.6 For the life-saving appliances and the other equipment of cargo ships, concerning the additional requirements for the ships using natural gas as fuel other than ships covered by the IGC Code, the survey during construction and after installation should consist of:
- (EI) 1.1.6.1 examining the arrangements for fire protection and fire extinction (IGF Code ch.11);
- (EI) 1.1.6.2 examining the fire pump capacity and working pressure in relation to water spray system, if the water spray system is part of the fire main system (IGF Code para.11.4.1);
- (EI) 1.1.6.3 examining the isolating valves of the fire main, when the fuel storage tank(s) is located on the open deck (IGF Code para.11.4.2);
- (EI) 1.1.6.4 examining the water spray system arrangement for fuel storage tanks(s) on open deck including remote operation (IGF Code para.11.5);
- (EI) 1.1.6.5 examining the fixed dry chemical powder fire-extinguishing system for the bunkering station area (IGF Code para.11.6.1);
- (EI) 1.1.6.6 examining the portable dry powder extinguisher (IGF Code para.11.6.2); and
- (EI) 1.1.6.7 examining the fixed fire detection and alarm system (IGF Code para.11.7).
- (EI) 1.1.7 For the life-saving appliances and the other equipment of cargo ships the check that the required documentation has been placed on board should consist of:
- (EI) 1.1.7.1 confirming that the fire control plans are permanently exhibited or, alternatively, emergency booklets have been provided and that a duplicate of the plans or the emergency booklet are available in a prominently marked enclosure external to the ship's deckhouse (SOLAS 74/00 reg.II-2/15.2.4) (SOLAS 74/88 reg.II-2/20);
- (EI) 1.1.7.2 confirming that maintenance plans have been provided (SOLAS 74/00 regs.II-2/14.2.2 and 14.4);
- (EI) 1.1.7.3 confirming that the training manuals and the fire safety operational booklets have been provided (SOLAS 74/00/14 regs.II-2/15.2.3, 16.2 and 16.3);

- (EI) 1.1.7.4 confirming that, where applicable, the approved documentation for the alternative design and arrangement is on board (SOLAS 74/00/06 regs.II-2/17 and III/38);
- (EI) 1.1.7.5 confirming, where appropriate, that the ship is provided with a document indicating compliance with the special requirement for carrying dangerous goods (SOLAS 74/00/08, reg.II-2/19.4) (SOLAS 74/88 reg.II-2/54(3));
- (EI) 1.1.7.6 confirming that emergency instructions are available for each person on board, that the muster list is posted in conspicuous places and that they are in a language understood by the persons on board (SOLAS 74/00/17 regs.III/8 and 37);
- (EI) 1.1.7.7 confirming that ship-specific plans and procedures for recovery of persons from the water have been provided (SOLAS 74/12 reg.III/17-1);
- (EI) 1.1.7.8 confirming that the training manual and training aids for the life-saving appliances have been provided and are available in the working language of the ship (SOLAS 74/00 reg.III/35);
- (EI) 1.1.7.9 confirming that the instructions for onboard maintenance of the life-saving appliances have been provided (SOLAS 74/88 reg.III/36);
- (EI) 1.1.7.10 confirming that a table or curve of residual deviations for the magnetic compass has been provided, and that a diagram of the radar installations shadow sectors is displayed (SOLAS 74/00 reg.V/19);
- (EI) 1.1.7.11 checking that operational and, where appropriate, maintenance manuals for all navigational equipment are provided (SOLAS 74/00 reg.V/16);
- (EI) 1.1.7.12 checking that records are provided, identifying any pilot ladders placed into service (SOLAS 74/10 reg.V/23.2.4);
- (EI) 1.1.7.13 checking that the charts and nautical publications necessary for the intended voyage are available and have been updated (SOLAS 74/88 reg.V/27);
- (EI) 1.1.7.14 checking that the International Code of Signals and an up-to-date copy of Volume III of the International Aeronautical and Maritime Search and Rescue (IAMSAR) Manual have been provided (SOLAS 74/00/02 reg.V/21);
- (EI) 1.1.7.15 checking that arrangements are provided to maintain records of navigational activities and daily reporting (SOLAS 74/00/03 reg.V/28);
- (EI) 1.1.7.16 checking that an illustrated table describing the life-saving signals to be used by ships, aircraft or persons in distress is available (SOLAS 74/00 reg.V/29); and
- (EI) 1.1.7.17 confirming that a continuous synopsis record is provided (SOLAS 74/02 reg.XI-1/5).

(EI)	1.1.8	For the life-saving appliances and the other equipment of cargo ships, concerning the additional requirements for tankers the check that the required documentation has been placed on board should consist of:
(EI)	1.1.8.1	confirming, when appropriate, that the instruction manuals for the inert gas system have been provided (FSS Code ch.15 paragraph 2.2.5) (SOLAS 74/88 reg.II-2/62.21); and
(EI)	1.1.8.2	confirming that the operating and maintenance instructions for the fixed hydrocarbon gas detection system are provided (SOLAS 74/10 reg II-2/4 5 7 3 and ESS Code ch 16).
(EI)	1.1.9	For the life-saving appliances and the other equipment of cargo ships the completion of the initial survey should consist of:
(EI)	1.1.9.1	after a satisfactory survey, the Cargo Ship Safety Equipment Certificate and its associated Record of Equipment (Form E) should be issued.
(EA)	1.2	Annual surveys – see part "General" section 4.2.
(EA)	1.2.1	For the life-saving appliances and the other equipment of cargo ships the examination of current certificates and other records should consist of:
(EA)	1.2.1.1	checking the validity, as appropriate, of the Cargo Ship Safety Equipment Certificate, the Cargo Ship Safety Radio Certificate and the Cargo Ship Safety Construction Certificate or the Cargo Ship Safety Certificate;
(EA)	1.2.1.2	checking, as appropriate, the validity of the Polar Ship Certificate;
(EA)	1.2.1.3	checking the validity of the Safety Management Certificate (SMC) and that a copy of the Document of Compliance (DOC) is on board;
(EA)	1.2.1.4	checking the validity of the International Ship Security Certificate;
(EA)	1.2.1.5	checking the validity of the International Load Line Certificate or International Load Line Exemption Certificate;
(EA)	1.2.1.6	checking the validity of the International Oil Pollution Prevention Certificate;
(EA)	1.2.1.7	checking the certificates of class, if the ship is classed with a classification society;
(EA)	1.2.1.8	checking, when appropriate, the validity of the International Certificate of Fitness for the Carriage of Dangerous Chemicals in Bulk or the Certificate of Fitness for the Carriage of Dangerous Chemicals in Bulk;
(EA)	1.2.1.9	checking, when appropriate, the validity of the International Certificate of Fitness for the Carriage of Liquefied Gases in Bulk;
(EA)	1.2.1.10	checking, when appropriate, the validity of the International Pollution Prevention Certificate for the Carriage of Noxious Liquid Substances in Bulk;
(EA)	1.2.1.11	checking, when appropriate, the validity of the International Sewage Pollution Prevention Certificate;

- (EA) 1.2.1.12 checking, when appropriate, the validity of the International Air Pollution Prevention Certificate;
- (EA) 1.2.1.13 confirming, when appropriate, the validity of the International Energy Efficiency Certificate (MARPOL Annex VI, regs.6.4 and 6.5);
- (EA) 1.2.1.14 confirming, when appropriate, that confirmation of compliance for the SEEMP part II is provided to and retained on board the ship (MARPOL Annex VI, reg. 5.4.5)¹¹;
- (EA) 1.2.1.15 confirming, when appropriate, the validity of the Statements of Compliance related to fuel oil consumption reporting (MARPOL Annex VI, regs.6.6 and 6.7);
- (EA) 1.2.1.16 checking, when appropriate, the validity of the International Ballast Water Management Certificate;
- (EA) 1.2.1.17 checking that the ship's complement complies with the Minimum Safe Manning Document (SOLAS 74/00/12 reg.V/14) (SOLAS 74/88 reg.V/13(b));
- (EA) 1.2.1.18 checking that the master, officers and ratings are certificated as required by the STCW Convention;
- (EA) 1.2.1.19 checking the manning and supervision of survival craft (SOLAS 74/00 reg.III/10);
- (EA) 1.2.1.20 confirming that, where applicable, the approved documentation for the alternative design and arrangements is on board (SOLAS 74/00/06 regs.II-2/17 and III/38);
- (EA) 1.2.1.21 checking whether any new equipment has been fitted and, if so, confirming that it has been approved before installation and that any changes are reflected in the appropriate certificate;
- (EA) 1.2.1.22 confirming that the fire control plans are permanently exhibited or, alternatively, that emergency booklets have been provided and that a duplicate of the plans or the emergency booklet are available in a prominently marked enclosure external to the ship's deckhouse (SOLAS 74/00 reg.II-2/15.2.4) (SOLAS 74/88 reg.II-2/20);
- (EA) 1.2.1.23 confirming that the maintenance plans have been provided (SOLAS 74/00 regs.II-2/14.2.2 and 14.4);
- (EA) 1.2.1.24 confirming that the training manuals and the fire safety operational booklets have been provided (SOLAS 74/00/14 regs.II-2/15.2.3, 16.2 and 16.3);
- (EA) 1.2.1.25 checking whether any fire has occurred on board necessitating the operation of the fixed fire-extinguishing systems or the portable fire extinguishers since the last survey;

¹¹ Refer to the Sample Format of Confirmation of Compliance, Early Submission of the SEEMP part II on the ship fuel oil consumption data collection plan and its timely verification pursuant to regulation 5.4.5 of MARPOL Annex VI (MEPC.1/Circ.876).

- (EA) 1.2.1.26 checking, when appropriate, that the ship is provided with a document indicating compliance with the special requirements for carrying dangerous goods (SOLAS 74/00/08 reg.II-2/19.4) (SOLAS 74/88 reg.II-2/54(3));
- (EA) 1.2.1.27 confirming, when appropriate, that there is a special list, manifest or stowage plan for the carriage of dangerous goods (SOLAS 74/88 reg.VII/5(3));
- (EA) 1.2.1.28 confirming, when appropriate, that the instruction manuals for the inert gas system have been provided and checking from the records of the pressure and oxygen content that the inert gas system is being operated correctly (FSS Code ch.15) (SOLAS 74/88 reg.II-2/62);
- (EA) 1.2.1.29 confirming that, where applicable, a factual statement has been provided on board by the lifeboat release and retrieval system manufacturer or one of their representatives that confirms the successful completion of the overhaul examination of an existing lifeboat release and retrieval system found to be compliant with paragraphs 4.4.7.6.4 to 4.4.7.6.6 of the LSA Code, or, alternatively, that a statement of acceptance of the installation of a replacement release and retrieval system to an existing available (SOLAS 74/11 reg.III/1.5; lifeboat is LSA Code section 4.4.7.6);
- (EA) 1.2.1.30 checking that logbook entries are being made (SOLAS 74/00/12 regs.III/19 and 20) and in particular:
- (EA) 1.2.1.30.1 the date when the last full muster of the crew for boat and fire drill took place, and the date when the last enclosed space entry and rescue drills took place;
- (EA) 1.2.1.30.2 the records indicating that the lifeboat equipment was examined at that time and found to be complete;
- (EA) 1.2.1.30.3 the last occasion when the lifeboats were swung out and when each one was lowered into the water;
- (EA) 1.2.1.30.4 the records indicating that crew members have received the appropriate onboard training;
- (EA) 1.2.1.30.5 the records indicating that on voyages where passengers are scheduled to be on board for more than 24 h, musters of newly embarked passengers have taken place prior to or immediately upon departure;
- (EA) 1.2.1.31 confirming that the training manual and training aids for the life-saving appliances are available on board in the working language of the ship (SOLAS 74/00 reg.III/35);
- (EA) 1.2.1.32 confirming that the checklist and instructions for onboard maintenance of the life-saving appliances are on board (SOLAS 74/00 reg.III/36);
- (EA) 1.2.1.33 confirming that a table or curve of residual deviations for the magnetic compass has been provided, the compass deviation book has been properly maintained and a diagram of the radar installations shadow sectors is displayed (SOLAS 74/00 reg.V/19);
- (EA) 1.2.1.34 checking that operational and, where appropriate, maintenance manuals for all navigational equipment are provided (SOLAS 74/00 reg.V/16);

- (EA) 1.2.1.35 checking that nautical charts and nautical publications necessary for the intended voyage are available and have been updated, and, where electronic systems are used, the required backup system is provided (SOLAS 74/00 regs.V/19 and 27);
- (EA) 1.2.1.36 checking that the International Code of Signals and an up-to-date copy of Volume III of the International Aeronautical and Maritime Search and Rescue (IAMSAR) Manual have been provided (SOLAS 74/00/02 reg.V/21);
- (EA) 1.2.1.37 checking that records are maintained identifying any pilot ladders placed into service and any repair effected (SOLAS 74/10 reg.V/23.2.4);
- (EA) 1.2.1.38 checking that an illustrated table describing the life-saving signals to be used by ships, aircraft or persons in distress is available (SOLAS 74/00 reg.V/29);
- (EA) 1.2.1.39 checking that records of navigational activities and daily reporting have been maintained (SOLAS 74/00/03 reg.V/28);
- (EA) 1.2.1.40 confirming that a continuous synopsis record is provided (SOLAS 74/02 reg.XI-1/5); and
- (EA) 1.2.1.41 confirming the availability of the International Anti-fouling System Certificate (AFS 2001 annex 4 reg.2), when applicable.
- (EA) 1.2.2 For the life-saving appliances and the other equipment of cargo ships the annual survey should consist of:
- (EA) 1.2.2.1 examining the fire pumps, fire main, hydrants, hoses and nozzles and the international shore connection and checking that each fire pump, including the emergency fire pump, can be operated separately so that two jets of water are produced simultaneously from different hydrants at any part of the ship while the required pressure is maintained in the fire main (SOLAS 74/00/14 reg.II-2/10.2; FSS Code chs.2 and 12) (SOLAS 74/88 regs.II-2/4 and 19);
- (EA) 1.2.2.2 for ships designed to carry containers on or above the weather deck, as applicable, examining the water mist lance and, as appropriate, the mobile water monitors and all necessary hoses, fittings and required fixing hardware (SOLAS 74/00/14 reg.II-2/10.7.3);
- (EA) 1.2.2.3 checking the provision and randomly examining the condition of the portable and non-portable fire extinguishers (SOLAS 74/00 reg.II-2/10.3; FSS Code ch.4) (SOLAS 74/88 reg.II-2/6);
- (EA) 1.2.2.4 confirming that the fire-fighters' outfits including their self-contained compressed air breathing apparatus and emergency escape breathing devices (EEBDs) are complete and in good condition, that the cylinders, including the spare cylinders, of any required self-contained breathing apparatus are suitably charged, and that onboard means of recharging breathing apparatus cylinders used during drills or a suitable number of spare cylinders to replace those used are provided, and provision of two-way portable radiotelephone apparatus of an explosion-proof type or intrinsically safe (SOLAS 74/00/12 regs.II-2/10.10, 13.3.4, 13.4.3 and 15.2.2; FSS Code ch.3) (SOLAS 74/88 reg.II-2/17) (BCH Code, ch.III, part E);

- (EA) 1.2.2.5 checking the operational readiness and maintenance of fire-fighting systems (SOLAS 74/00 reg.II-2/14) (SOLAS 74/88/91 reg.II-2/21);
- (EA) 1.2.2.6 examining the fixed fire-fighting system for the machinery, cargo, vehicle, special category and ro-ro spaces, as appropriate, and confirming that its means of operation is clearly marked (SOLAS 74/00/12/14 regs.II-2/10.4, 10.5, 10.7.1, 10.7.2 and 20.6.1; FSS Code chs.5 to 7) (SOLAS 74/88 regs.II-2/7 and 53);
- (EA) 1.2.2.7 examining the fire-extinguishing and special arrangements in the machinery spaces and confirming, as far as practicable and as appropriate, the operation of the remote means of control provided for the opening and closing of the skylights, the release of smoke, the closure of the funnel and ventilation openings, the closure of power-operated and other doors, the stopping of ventilation and boiler forced and induced draught fans and the stopping of oil fuel and other pumps that discharge flammable liquids (SOLAS 74/00/12/14/16 regs.II-2/5.2, 8.3, 9.5 and 10.5) (SOLAS 74/88 regs.II-2/7 and 11);
- (EA) 1.2.2.8 checking that fixed carbon dioxide fire-extinguishing systems for the protection of machinery spaces and cargo pump-rooms, where applicable, are provided with two separate controls, one for opening of the gas piping and one for discharging the gas from the storage container, each of them located in a release box clearly identified for the particular space (SOLAS 74/08 reg.II-2/10.4.1.5);
- (EA) 1.2.2.9 examining, as far as possible, and testing, as feasible, any fire detection and alarm system and any sample extraction smoke detection system (SOLAS 74/00/10 regs.II-2/7.2, 7.3, 7.4, 7.5.1, 7.5.5, 19.3.3 and 20.4; FSS Code chs.9 and 10) (SOLAS 74/88 regs.II-2/11, 13, 14, 53 and 54);
- (EA) 1.2.2.10 examining the fire-extinguishing systems for spaces containing paint and/or flammable liquids and deep-fat cooking equipment in accommodation and service spaces (SOLAS 74/00 regs.II-2/10.6.3 and 10.6.4; FSS Code chs.5 to 7) (SOLAS 74/88 reg.II-2/18.7) (BCH Code ch.III, part E);
- (EA) 1.2.2.11 examining the helicopter facilities, including foam firefighting appliances when appropriate (SOLAS 74/00/16 reg.II-2/18; FSS Code ch.17) (SOLAS 74/88 reg.II-2/18.8);
- (EA) 1.2.2.12 examining the arrangements for oil fuel, lubricating oil and other flammable oils and testing the remote closing of valves for oil fuel, lubricating oil and other flammable oils and the operation of the remote means of closing the valves on the tanks that contain oil fuel, lubricating oil and other flammable oils (SOLAS 74/00 reg.II-2/4.2.2.3.4) (SOLAS 74/88 reg.II-2/15.2.5);
- (EA) 1.2.2.13 examining and testing of the general emergency alarm system (SOLAS 74/88 reg.III/20);
- (EA) 1.2.2.14 examining the fire protection arrangements in cargo, vehicle and ro-ro spaces, including the fire safety arrangements for vehicle carriers carrying motor vehicles with compressed hydrogen or natural gas in their tanks for their own propulsion as cargo, as applicable, and confirming, as far as practicable and as appropriate, the operation of the means of

control provided for closing the various openings (SOLAS 74/00/14 regs.II-2/10.7.1, 10.7.2, 20.2.1, 20.3, 20.6.2, 20-1.2.1, 20-1.3, and 20-1.4) (SOLAS 74/88 reg.II-2/53);

- (EA) 1.2.2.15 examining and testing the portable gas detectors suitable for the detection of the gas fuel, for vehicle carriers carrying motor vehicles with compressed hydrogen or natural gas in their tanks for their own propulsion as cargo (SOLAS 74/00/14 regs.II-2/20-1.2 and 20-1.5);
- (EA) 1.2.2.16 examining, where applicable, the alternative design and arrangements for fire safety or life-saving appliances and arrangements, in accordance with the test, inspection and maintenance requirements, if any, specified in the approved documentation (SOLAS 74/00/06 regs.II-2/17 and III/38);
- (EA) 1.2.2.17 examining, when appropriate, the special arrangements for carrying dangerous goods, including checking the electrical equipment and wiring, the ventilation, the provision of protective clothing and portable appliances and the testing of the water supply, bilge pumping and any water spray system (SOLAS 74/00/08 reg.II-2/19 (except 19.3.8, 19.3.10 and 19.4)) (SOLAS 74/88 reg.II-2/54);
- (EA) 1.2.2.18 checking that emergency instructions are available for each person on board, that copies of the suitably updated muster list are posted in conspicuous places, and that they are in a language understood by all persons on board, and confirming that there are posters or signs in the vicinity of survival craft and their launching stations (SOLAS 74/00/17 regs.III/8, 9 and 37);
- (EA) 1.2.2.19 checking that the life-saving appliances are of an international or vivid reddish orange, or a comparably highly visible colour on all parts where this will assist detection at sea (LSA Code section 1.2.2.6);
- (EA) 1.2.2.20 examining each survival craft, including its equipment and, when fitted, the on-load release mechanism and hydrostatic lock and, for inflatable liferafts, the hydrostatic release unit and float-free arrangements; checking that the hand-held flares are not out of date (SOLAS 74/00 regs.III/16, 20 and 31; LSA Code sections 2.5, 3.1 to 3.3, 4.1.5, 4.4.7 and 4.4.8);
- (EA) 1.2.2.21 for liferafts provided for easy side-to-side transfer, verifying that they are less than 185 kg (SOLAS 74/00 reg.III/31.1);
- (EA) 1.2.2.22 checking that the falls used in launching appliances have been periodically inspected and have been renewed as necessary in the past five years (SOLAS 74/00 reg.III/20);
- (EA) 1.2.2.23 examining the embarkation arrangements and launching appliances for each survival craft; each lifeboat should be lowered to the embarkation position or, if the stowage position is the embarkation position, lowered a short distance and, if practicable, one of the survival craft should be lowered to the water; the operation of the launching appliances for davit-launched liferafts should be demonstrated (SOLAS 74/00 regs.III/11, 12, 13, 16, 20 and 31; LSA Code section 6.1);

- (EA) 1.2.2.24 checking that a thorough examination of launching appliances, including the dynamic testing of the winch brake, and servicing of lifeboat and rescue boat on-load release gear, including fast rescue boat release gear, free-fall lifeboat release systems and davit-launched liferaft automatic release hooks, and that a thorough examination and operational test of lifeboat and rescue boat including fast rescue boat, has been carried out in accordance with the Requirements for maintenance, thorough examination, operational testing, overhaul and repair of lifeboats and rescue boats, launching appliances and release gear (resolution MSC.402(96)),. The operational testing of free-fall lifeboat release systems shall be performed either by free-fall launch with only the operating crew on board or by a test without launching the lifeboat carried out in accordance with the Requirements for maintenance, thorough examination, operational testing, overhaul and repair of lifeboats and rescue boats, launching appliances and release gear (SOLAS 74/00/12/16 reg.III/ 20.11);
- (EA) 1.2.2.25 examining each rescue boat, including its equipment; for inflatable rescue boats, confirming that they are stowed in a fully inflated condition (SOLAS 74/88 reg.III/14 and 31; LSA Code sections 2.5 and 5.1);
- (EA) 1.2.2.26 confirming that there are posters or signs in the vicinity of the survival craft, their launching stations and containers, brackets, racks and other similar stowage locations for life-saving equipment (SOLAS 74/00 regs.III/9 and 20);
- (EA) 1.2.2.27 examining the embarkation and recovery arrangements for each rescue boat; if practicable, the rescue boat(s) should be lowered to the water and its recovery demonstrated (SOLAS 74/00 regs.III/14, 17 and 31; LSA Code section 6.1);
- (EA) 1.2.2.28 testing that the engine of the rescue boat(s) and of each lifeboat, when so fitted, start satisfactorily and operate both ahead and astern;
- (EA) 1.2.2.29 examining and checking the operation of two-way VHF radiotelephone apparatus¹² and search and rescue locating devices (SOLAS 74/88/08 reg.III/6);
- (EA) 1.2.2.30 examining the line-throwing appliance and checking that its rockets and the ship's distress signals are not out of date, and examining and checking the operation of onboard communications equipment and the general emergency alarm system (SOLAS 74/00 regs.II-2/12.2 and III/6 and 18; LSA Code sections 3.1, 7.1 and 7.2);
- (EA) 1.2.2.31 examining the provision, disposition, stowage and condition of the lifebuoys, including those fitted with self-igniting lights, self-activating smoke signals and buoyant lines, lifejacket¹³ and their whistles and lights, immersion suits and anti-exposure suits and checking that their

¹² Expired primary batteries, which complied with the requirements of paragraph 12.6 of *Revised Recommendation on performance standards for survival craft portable Two-Way VHF radiotelephone apparatus* (resolution MSC.149 (77)), could be used for no other purpose than to examine and check the operation of two-way VHF radiotelephone apparatus.

¹³ SOLAS regulation III/7.2.1.5 should be considered.

associated batteries are not out of date (SOLAS 74/88/06 regs.III/7 and 32, LSA Code sections 2.1 to 2.5);

- (EA) 1.2.2.32 checking that immersion suits designed to be worn in conjunction with a lifejacket are suitably marked (LSA Code section 2.3.1);
- (EA) 1.2.2.33 checking the lighting of the muster and embarkation stations and the alleyways, stairways and exits giving access to the muster and embarkation stations, including when supplied from the emergency source of power (SOLAS 74/88 regs.II-1/42 or 43 and III/11);
- (EA) 1.2.2.34 checking that the required navigation lights, shapes and sound signalling equipment are in order (COLREG 1972, rules 20 to 24, 27 to 30 and 33);
- (EA) 1.2.2.35 checking that the following items of navigation equipment are in working order, as appropriate: daylight signalling lamp, magnetic compass, transmitting heading device, gyro compass, gyro compass repeaters, radar installation(s), electronic plotting aid, automatic tracking aid(s) or automatic radar plotting aid(s), echo-sounding device, speed and distance measuring device(s), rudder angle indicator, propeller rate of revolution indicator, variable-pitch propeller pitch and operational mode indicator, rate-of-turn indicator, heading or track control system, Global Navigation Satellite System (GNSS) receiver, terrestrial radio navigation system and sound reception system, means of communication with emergency steering position, a pelorus or compass bearing device, means for correcting heading and bearings, BNWAS as applicable and ECDIS including backup arrangements, as applicable; items that cannot be checked with the ship in port should be verified from records (SOLAS 74/00/09/13 reg.V/19);
- (EA) 1.2.2.36 checking that the International Code of Signals is available (SOLAS 74/00 reg.V/21);
- (EA) 1.2.2.37 checking the rotational deployment of the marine evacuation system (MES) (SOLAS 74/88 reg.III/20.8.2; LSA Code section 6.2.2.2);
- (EA) 1.2.2.38 checking the provision, specification, operation and annual performance test of the voyage data recorder, where fitted (SOLAS 74/00/04 reg.V/20);
- (EA) 1.2.2.39 checking the provision and operation of the automatic identification system, where fitted, and whether the annual test has been carried out and a copy of the test report is on board (SOLAS 74/00/04/10 regs.V/18.9 and 19);
- (EA) 1.2.2.40 checking that a valid conformance test report of the long-range identification and tracking system is available on board, where fitted (SOLAS 74/04 reg.V/19-1);
- (EA) 1.2.2.41 checking the provision and specification of the pilot ladders and pilot transfer arrangements (SOLAS 74/00/10 reg.V/23);
- (EA) 1.2.2.42 checking that the means of embarkation and disembarkation from ships for use in port and in port-related operations, such as gangways and

accommodation ladders, are in satisfactory condition, as applicable (SOLAS 74/08 reg.II-1/3-9);

- (EA) 1.2.2.43 checking, when appropriate, the provision of an appropriate instrument for measuring the concentration of gas or oxygen in the air together with detailed instructions for its use (SOLAS 74/08 reg.VI/3); and
- (EA) 1.2.2.44 confirming that ship-specific plans and procedures for recovery of persons from the water have been provided (SOLAS 74/12 reg.III/17-1).
- (EA) 1.2.3 For the life-saving appliances and the other equipment of cargo ships, concerning the additional requirements for tankers the annual survey should consist of:
- (EA) 1.2.3.1 checking the deck foam system, including the supplies of foam concentrate and testing that the minimum number of jets of water at the required pressure in the fire main is obtained (see (EA) 1.2.2.1) when the system is in operation (SOLAS 74/00 reg.II-2/10.8; FSS Code ch.14) (SOLAS 74/88 reg.II-2/61);
- (EA) 1.2.3.2 examining the inert gas system (SOLAS 74/00/14 reg.II-2/4.5.5; FSS Code ch.15) (SOLAS 74/88 reg.II-2/62), and in particular:
- (EA) 1.2.3.2.1 examining externally for any sign of gas or effluent leakage;
- (EA) 1.2.3.2.2 confirming the proper operation of both inert gas blowers;
- (EA) 1.2.3.2.3 observing the operation of the scrubber-room ventilation system;
- (EA) 1.2.3.2.4 checking the deck water seal for automatic filling and draining, and the arrangements for protecting the system against freezing;
- (EA) 1.2.3.2.5 where a double block and bleed valve is installed, checking the automatic operations of the block and the bleed valves upon loss of power;
- (EA) 1.2.3.2.6 where two shut-off valves in series with a venting valve in between are used as non-return devices, checking the automatic operation of the venting valve, and the alarm for faulty operation of the valves;
- (EA) 1.2.3.2.7 examining the operation of all remotely operated or automatically controlled valves and, in particular, the flue gas isolating valves;
- (EA) 1.2.3.2.8 observing a test of the interlocking feature of soot blowers;
- (EA) 1.2.3.2.9 observing that the gas pressure regulating valve automatically closes when the inert gas blowers are secured;
- (EA) 1.2.3.2.10 checking the means for separating the cargo tank not being inerted from the inert gas main;
- (EA) 1.2.3.2.11 checking the alarms of the two oxygen sensors positioned in the space or spaces containing the inert gas system;
- (EA) 1.2.3.2.12 checking, as far as practicable, the following alarms and safety devices of the inert gas system using simulated conditions where necessary:
- (EA) 1.2.3.2.12.1 high oxygen content of gas in the inert gas main;
- (EA) 1.2.3.2.12.2 low gas pressure in the inert gas main;
- (EA) 1.2.3.2.12.3 low pressure in the supply to the deck water seal;
- (EA) 1.2.3.2.12.4 high temperature of gas in the inert gas main;
- (EA) 1.2.3.2.12.5 low water pressure or low water-flow rate;
- (EA) 1.2.3.2.12.6 accuracy of portable and fixed oxygen-measuring equipment by means of calibration gas;
- (EA) 1.2.3.2.12.7 high water level in the scrubber;
- (EA) 1.2.3.2.12.8 failure of the inert gas blowers;
- (EA) 1.2.3.2.12.9 failure of the power supply to the automatic control system for the gas regulating valve and to the instrumentation for continuous indication and permanent recording of pressure and oxygen content in the inert gas main;
- (EA) 1.2.3.2.12.10 high pressure of gas in the inert gas main;
- (EA) 1.2.3.3 checking, when practicable, the proper operation of the inert gas system on completion of the checks listed above (FSS Code ch.15) (SOLAS 74/88 reg.II-2/62);
- (EA) 1.2.3.4 examining the fixed fire-fighting system for the cargo pump-rooms (SOLAS 74/00 reg.II-2/10.9) (SOLAS 74/88 reg.II-2/63) and confirming, as far as practicable and when appropriate, the operation of the remote means for closing the various openings;
- (EA) 1.2.3.5 checking for all tankers the provision of at least one portable instrument for measuring oxygen and one for measuring flammable vapour concentrations, together with a sufficient set of spares, and suitable means for the calibration of these instruments (SOLAS 74/10 reg.II-2/4.5.7.1);
- (EA) 1.2.3.6 examining the arrangements for gas measurement in double hull spaces and double bottom spaces, including the fitting of permanent gas sampling lines, where appropriate (SOLAS 74/10 reg.II-2/4.5.7.2);
- (EA) 1.2.3.7 examining, as far as possible, and testing the fixed hydrocarbon gas detection system (SOLAS 74/10 reg.II-2/4.5.7.3 and FSS Code ch.16);
- (EA) 1.2.3.8 checking the condition and operation of water spray and air supply systems that are in totally enclosed lifeboats and have self-contained air support systems (LSA Code sections 4.4 and 4.6 to 4.9);
- (EA) 1.2.3.9 checking the protection of the cargo pump-room (SOLAS 74/00 reg.II-2/4.5.10), and in particular:
- (EA) 1.2.3.9.1 checking temperature sensing devices for bulkhead glands and alarms;
- (EA) 1.2.3.9.2 checking the interlock between lighting and ventilation;
- (EA) 1.2.3.9.3 checking the gas detection system; and

(EA)	1.2.3.9.4	checking bilge level monitoring devices and alarms.
(EA)	1.2.4	For the life-saving appliances and the other equipment of cargo ships, concerning the additional requirements for the ships using natural gas as fuel other than ships covered by the IGC Code, the annual survey should consist of:
(EA)	1.2.4.1	examining the arrangements for fire protection and fire extinction (IGF Code ch.11);
(EA)	1.2.4.2	examining the fire pump capacity and working pressure in relation to the water spray system, if the water spray system is part of the fire main system (IGF Code para.11.4.1);
(EA)	1.2.4.3	examining the isolating valves of the fire main, when the fuel storage tank(s) is located on the open deck (IGF Code para.11.4.2);
(EA)	1.2.4.4	examining the water spray system for cooling, fire protection and crew protection (IGF Code para.11.5);
(EA)	1.2.4.5	examining the water spray system arrangement for fuel storage tanks(s) on open deck including remote operation (IGF Code para.11.5);
(EA)	1.2.4.6	examining the fixed fire detection and alarm system (IGF Code para.11.7);
(EA)	1.2.4.7	examining the fixed dry chemical powder fire-extinguishing system for the bunkering station area (IGF Code para.11.6.1); and
(EA)	1.2.4.8	examining the portable dry powder extinguisher (IGF Code para.11.6.2).
(EA)	1.2.5	For the life-saving appliances and the other equipment of cargo ships the completion of the annual survey should consist of:
(EA)	1.2.5.1	after a satisfactory survey, the Cargo Ship Safety Equipment Certificate should be endorsed; and
(EA)	1.2.5.2	if a survey shows that the condition of a ship or its equipment is unsatisfactory, see part "General", section 4.8.
(EP)	1.3	Periodical surveys – see part "General" section 4.4.
(EP)	1.3.1	For the life-saving appliances and the other equipment of cargo ships the examination of current certificates and other records should consist of:
(EP)	1.3.1.1	the provisions of (EA) 1.2.1.
(EP)	1.3.2	For the life-saving appliances and the other equipment of cargo ships the periodical survey should consist of:
(EP)	1.3.2.1	the provisions of (EA) 1.2.2;
(EP)	1.3.2.2	confirming, during the examination of the fixed fire-fighting system for the machinery, cargo, vehicle, special category and ro-ro spaces, that, as appropriate, any foam compounds and the CO_2 capacity have been checked and that the distribution pipework has been proved clear

(SOLAS 74/00/12/14 regs.II-2/10.4, 10.5, 10.7.1, 10.7.2 and 20.6.1; FSS Code chs.5 to 7) (SOLAS 74/88 regs.II-2/7 and 53);

- (EP) 1.3.2.3 testing the operation of the remote means of control provided for the opening and closing of the skylights, the release of smoke, the closure of the funnel and ventilation openings, the closure of power-operated and other doors, the stopping of ventilation and boiler forced and induced draught fans and the stopping of oil fuel and other pumps that discharge flammable liquids (SOLAS 74/00/14 regs.II-2/5.2, 8.3, 9.5 and 10.5) (SOLAS 74/88 reg.II-2/11);
- (EP) 1.3.2.4 testing any fire detection and alarm system and any sample extraction smoke detection system (SOLAS 74/00/10 regs.II-2/7.2, 7.3, 7.4, 7.5.5, 19.3.3 and 20.4; FSS Code chs.9 and 10) (SOLAS 74/88 regs.II-2/11, 13, 14, 53 and 54);
- (EP) 1.3.2.5 testing, as feasible, the fire-extinguishing system for spaces containing paint and/or flammable liquids and deep-fat cooking equipment in accommodation and service spaces (SOLAS 74/00 regs.II-2/10.6.3 and 10.6.4; FSS Code chs.5 to 7) (SOLAS 74/88 reg.II-2/18.7);
- (EP) 1.3.2.6 testing the remote closing of valves for oil fuel, lubricating oil and other flammable oils and the operation of the remote means of closing the valves on the tanks that contain oil fuel, lubricating oil and other flammable oils (SOLAS 74/00 reg.II-2/4.2.2.3.4) (SOLAS 74/88 reg.II-2/15.2.5);
- (EP) 1.3.2.7 testing the operation of the means of control provided for closing the various openings for the cargo, vehicle, special category and ro-ro spaces (SOLAS 74/00/14 regs.II-2/5.2 and 20.3) (SOLAS 74/88 reg.II-2/53); and
- (EP) 1.3.2.8 testing, as feasible, the helicopter facilities, including foam firefighting appliances when appropriate (SOLAS 74/00/16 reg.II-2/18; FSS Code ch.17) (SOLAS 74/88 reg.II-2/18.8).
- (EP) 1.3.3 For the life-saving appliances and the other equipment for the additional requirements for tankers the periodical survey should consist of:
- (EP) 1.3.3.1 the provisions of (EA) 1.2.3; and
- (EP) 1.3.3.2 confirming during the examination of the fixed fire-fighting system for the cargo pump-rooms that, as appropriate, any foam compounds have been checked and that the distribution pipework has been proved clear (SOLAS 74/00 reg.II-2/10.9; FSS Code chs.5 to 7) (SOLAS 74/88 reg.II-2/63) and checking the operation of the remote means for closing the various openings.
- (EP) 1.3.4 For the life-saving appliances and the other equipment of cargo ships, concerning the additional requirements for the ships using natural gas as fuel other than ships covered by the IGC Code, the periodical survey should consist of:
- (EP) 1.3.4.1 the provisions of (EA) 1.2.4.

	/	
(EP)	1.3.5	For the life-saving appliances and the other equipment of cargo ships the completion of the periodical survey should consist of:
(EP)	1.3.5.1	after a satisfactory survey, the cargo Ship Safety Equipment Certificate should be endorsed; and
(EP)	1.3.5.2	if a survey shows that the condition of a ship or its equipment is unsatisfactory, see part "General", section 4.8.
(ER)	1.4	Renewal surveys – see part "General" section 4.5
(ER)	1.4.1	For the life-saving appliances and the other equipment of cargo ships the examination of current certificates and other records should consist of:
(ER)	1.4.1.1	the provisions of (EA) 1.2.1, except for the validity of the Cargo Ship
(ER)	1.4.2	For the life-saving appliances and the other equipment of cargo ships the renewal survey should consist of:
(ER)	1.4.2.1	the provisions of (EP) 1.3.2; and
(ER)	1.4.2.2	for ships designed to carry containers on or above the weather deck, as applicable, testing that the mobile water monitors can be securely fixed to the ship structure ensuring safe and effective operation, and testing that the mobile water monitor jets reach the top tier of containers with all required monitors and water jets from fire hoses operated simultaneously (SOLAS 74/00/14 reg.II-2/10.7.3).
(ER)	1.4.3	For the life-saving appliances and the other equipment of cargo ships, concerning the additional requirements for tankers the renewal survey should consist of:
(ER)	1.4.3.1	the provisions of (EP) 1.3.3; and
(ER)	1.4.3.2	examining the deck water seal for the inert gas system internally and checking the condition of the non-return valve (FSS Code ch.15, paragraphs 2.2.3.1 and 2.3.1.6.2) (SOLAS 74/88 reg.II-2/62).
(ER)	1.4.4	For the life-saving appliances and the other equipment of cargo ships, concerning the additional requirements for the ships using natural gas as fuel other than ships covered by the IGC Code, the renewal survey should consist of:
(ER)	1.4.4.1	the provisions of (EP) 1.3.4.
(ER)	1.4.5	For the life-saving appliances and the other equipment of cargo ships the completion of the renewal survey should consist of:
(ER)	1.4.5.1	after a satisfactory survey, the Cargo Ship Safety Equipment Certificate should be issued.

- (C) 2 GUIDELINES FOR SURVEYS FOR THE CARGO SHIP SAFETY CONSTRUCTION CERTIFICATE
- (CI) **2.1** Initial surveys see part "General", section 4.1.
- (CI) 2.1.1 For the hull, machinery and equipment of cargo ships the examination of plans and designs should consist of:
- (CI) 2.1.1.1 examining the plans for the hull (SOLAS 74/88 regs.II-1/11, 12-1, 14, 18 and 19) (SOLAS 74/06/17 regs.II-1/9, 10, 11, 12, 13-1, 15, 15-1, 16 and 16-1);
- (CI) 2.1.1.2 examining plans to verify that bulk carriers of 150 m in length and above, where appropriate, meet the applicable structural requirements of an organization recognized by the Administration, or national standards of the Administration, conforming to the functional requirements of the Goal-based Ship Construction Standards for Bulk Carriers and Oil Tankers (SOLAS 74/10 reg.II-1/3-10);
- (CI) 2.1.1.3 examining the plans for the bilge pumping and drainage systems (SOLAS 74/88 reg.II-1/21) (SOLAS 74/05/08/09/17 regs.II-1/35-1 and II-2/20.6.1.4);
- (CI) 2.1.1.4 examining the stability information and the damage control plans (SOLAS 74/88/00 regs.II-1/22, 23-1 and 25-8) (SOLAS 74/06/08/17 regs.II-1/5, 5-1 and 19; IS Code chs.1, 2 and 3);
- (CI) 2.1.1.5 examining the ballasting arrangements (SOLAS 74/06/17 reg.II-1/20);
- (CI) 2.1.1.6 examining the plans for the machinery installation (SOLAS 74/88 regs.II-1/26 to 36);
- (CI) 2.1.1.7 examining the plans for the electrical installation (SOLAS 74/88 regs.II-1/40, 41, 43, 44 and 45);
- (CI) 2.1.1.8 examining, where applicable, the approved documentation for the alternative design and arrangements (SOLAS 74/00/15 regs.II-1/55 and II-2/17 and IGF Code, ch.2);
- (CI) 2.1.1.9 examining the plans for the periodically unattended machinery spaces (SOLAS 74/00 reg.II-2/4.2.5) (SOLAS 74/88 regs.II-1/46 to 53);
- (CI) 2.1.1.10 examining the plans for the structural fire protection, including ventilation systems, in accommodation and service spaces, control stations and machinery spaces and oil fuel and lubricating oil systems (SOLAS 74/00/12/14 regs.II-2/4.2.2, 4.2.2.3, 4.2.2.4, 4.2.2.5, 4.4, 5.2, 5.3.1, 5.3.2, 6.2, 6.3, 7.5.5, 7.7, 8.2, 8.4, 9.2.1, 9.3, 9.5, 9.7.1, 9.7.2, 9.7.3, 9.7.5.2, 9.7.6, 11.2, 11.3, 11.4, 11.5 and 17) (SOLAS 74/88 regs.II-2/42 to 52 (except 45 and 51));
- (CI) 2.1.1.11 examining the plans for the structural fire protection, including ventilation systems, in cargo spaces (SOLAS 74/00/15/17 regs.II-2/5.2, 9.7.1, 9.7.2, 9.7.3, 9.7.6, 11.2, 11.3, 11.5, 19.3.8, 19.3.10, 20.2.1, 20.3, 20-1.2.1, 20-1.3 and 20-1.4) (SOLAS 74/88 regs.II-2/42 to 54);

(CI)	2.1.1.12	examining the plans for the means of escape (SOLAS 74/00/14 regs.II-2/13.2, 13.3.1, 13.3.3, 13.4.2 and 13.6; FSS Code ch.13 paragraph 3) (SOLAS 74/88 reg.II-2/45);
(CI)	2.1.1.13	examining the plans for the arrangements for gaseous fuel for domestic purposes (SOLAS 74/00 reg.II-2/4.3) (SOLAS 74/88 reg.II-2/51);
(CI)	2.1.1.14	examining the arrangements for the openings in the shell plating below the freeboard deck (SOLAS 74/06/17 reg.II-1/15);
(CI)	2.1.1.15	examining the plans for helicopter facilities for ships fitted with such facilities (SOLAS 74/00/16 reg.II-2/18) (SOLAS 74/88 reg.II-2/18.8);
(CI)	2.1.1.16	examining the Cargo Securing Manual for ships carrying cargoes other than solid and liquid bulk cargoes, cargo units and cargo transport units (SOLAS 74/98/02_reg.VI/5.6);
(CI)	2.1.1.17	checking for the loading booklet for carriage of cargoes in bulk (SOLAS 74/00 reg.VI/7);
(CI)	2.1.1.18	examining the loading instrument for bulk carriers of 150 m in length and upwards (SOLAS 74/97/04 reg.XII/11);
(CI)	2.1.1.19	confirming that bulk carriers, when appropriate, meet the requirements of damage stability and structural strength with its cargo hold(s) flooded, including other structural requirements (SOLAS 74/97/04 regs.XII/3, 4, 5 and 6);
(CI)	2.1.1.20	examining the functionality of bilge well alarms to all cargo holds and conveyor tunnels (SOLAS 74/97/04 reg.XII/9);
(CI)	2.1.1.21	confirming that the ship is constructed in accordance with the requirements of a recognized classification society, or one with equivalent national standards (SOLAS 74/00 reg.II-1/3-1);
(CI)	2.1.1.22	confirming that a corrosion prevention system is fitted, when appropriate, in dedicated seawater ballast tanks arranged in ships and double-side skin spaces arranged in bulk carriers of 150 m in length and upwards (SOLAS 74/04/06 reg.II-1/3-2);
(CI)	2.1.1.23	examining, for oil tankers and bulk carriers when appropriate, the Ship Structure Access Manual (SOLAS 74/00/02/04 reg.II-1/3-6(4));
(CI)	2.1.1.24	for bulk carriers, checking the arrangements for hold, ballast and dry space water level detectors and their audible and visual alarms (SOLAS 74/02 reg.XII/12);
(CI)	2.1.1.25	for bulk carriers, checking the arrangements for availability of draining and pumping systems forward of the collision bulkhead (SOLAS 74/02 reg.XII/13);
(CI)	2.1.1.26	examining the calculation and drawings for the sufficient safe working load of towing and mooring equipment to enable the safe conduct of all towing and mooring operation in normal operation of the ship (SOLAS 74/04 reg.II-1/3-8); and

- (CI) 2.1.1.27 checking the provision of means to prevent blockage of drainage arrangements, for closed vehicle and ro-ro spaces and special category spaces where fixed pressure water-spraying systems are used (SOLAS 74/08 reg.II-2/20.6.1.5).
- (CI) 2.1.2 For the hull, machinery and equipment of cargo ships, concerning the examination of plans and designs the additional requirements for oil tankers, chemical tankers and gas carriers should consist of:
- (CI) 2.1.2.1 examining the plans for the steering gear (SOLAS 74/14 reg.II-1/29);
- (CI) 2.1.2.2 examining the plans for the electrical installation (SOLAS 74/00 reg.II-1/43) (SOLAS 74/88 reg.II-1/45);
- (CI) 2.1.2.3 examining the plans for the structural fire protection (SOLAS 74/00/15 regs.II-2/1.6, 4.5.1, 4.5.2, 4.5.9, 9.2.4, 9.3, 9.4, 9.5, 9.6.3 and 11.6) (SOLAS 74/88 regs.II-2/55 to 58);
- (CI) 2.1.2.4 examining the plans for the cargo tank venting, cargo tank purging and gas freeing and other ventilation arrangements and protection of the cargo tank structure against pressure or vacuum (SOLAS 74/00/15 regs.II-2/4.5.3, 4.5.4, 4.5.6, 4.5.8, 11.6 and 16.3) (SOLAS 74/88 reg.II-2/59);
- (CI) 2.1.2.5 examining the plans of access to bow (SOLAS 74/00/04 reg.II-1/3-3);
- (CI) 2.1.2.6 examining the plans for emergency towing, for tankers of not less than 20,000 tonnes deadweight (SOLAS 74/00/04 reg.II-1/3-4);
- (CI) 2.1.2.7 checking the access to spaces in the cargo area of oil tankers (SOLAS 74/88/92/00 reg.II-1/12-2) (SOLAS 74/04 reg.II-1/3-6); and
- (CI) 2.1.2.8 examining plans to verify that oil tankers of 150 m in length and above, where appropriate, meet the applicable structural requirements of an organization recognized by the Administration, conforming to the functional requirements of the Goal-based Ship Construction Standards for Bulk Carriers and Oil Tankers (SOLAS 74/10 reg.II-1/3-10).
- (CI) 2.1.3 For the hull, machinery and equipment of cargo ships, concerning the examination of plans and designs additional requirements for the ships using natural gas as fuel other than ships covered by the IGC Code should consist of:
- (CI) 2.1.3.1 examining the plans for the fuel containment systems, control of vapour space of liquefied gas fuel tanks, vapour detection, gauging, loading limits for liquefied gas fuel tanks and other special requirements (IGF Code chs.5, 6, 7, 8 and 15));
- (CI) 2.1.3.2 examining the plans for the ship arrangements (IGF Code ch.5);
- (CI) 2.1.3.3 examining the plans for piping systems (IGF Code chs.5, 6, 7 and 9);
- (CI) 2.1.3.4 examining the plans for the pressure control (IGF Code ch.6);

(CI)	2.1.3.5	examining the plans for the environmental control (IGF Code ch.6);
(CI)	2.1.3.6	examining the plans for machinery installation (IGF Code ch.10);
(CI)	2.1.3.7	examining the plans for fire protection (IGF Code section 11.3);
(CI)	2.1.3.8	examining the plans for the ventilation systems (IGF Code chs.12 and 13);
(CI)	2.1.3.9	examining the plans for the electrical installations (IGF Code chs.12 and 14); and
(CI)	2.1.3.10	examining the plans for the control, monitoring and safety systems (IGF Code ch.15).
(CI)	2.1.4	For the hull, machinery and equipment of cargo ships the survey during construction and after installation should consist of:
(CI)	2.1.4.1	confirming that the collision bulkhead is watertight up to the freeboard deck, that the valves fitted on the pipes piercing the collision bulkhead are operable from above the freeboard deck and that there are no doors, manholes, ventilation ducts or any other openings (SOLAS 74/88 reg.II-1/11) (SOLAS 74/06/17 reg.II-1/12);
(CI)	2.1.4.2	confirming in accordance with the survey plan that bulk carriers of 150 m in length and above, where appropriate, meet the applicable structural requirements of an organization recognized by the Administration, or national standards of the Administration, conforming to the functional requirements of the Goal-based Ship Construction Standards for Bulk Carriers and Oil Tankers (SOLAS 74/10 reg.II-1/3.10);
(CI)	2.1.4.3	confirming that the subdivision bulkheads are constructed and tested as watertight up to the freeboard deck, as applicable (SOLAS 74/88 reg.II-1/14) (SOLAS 74/06/17 reg.II-1/10 and 11);
(CI)	2.1.4.4	confirming that each watertight door has been tested (SOLAS 74/88 reg.II-1/18) (SOLAS 74/06/17 reg.II-1/16);
(CI)	2.1.4.5	confirming that the arrangements for operating any watertight doors are generally in accordance with the requirements for passenger ships and carrying out similar tests (see (PI) 5.1.2.6 to (PI) 5.1.2.8) (SOLAS 74/88 reg.II-1/15) (SOLAS 74/06 reg.II-1/13-1);
(CI)	2.1.4.6	confirming by a hose or flooding test the watertightness of watertight decks and trunks, tunnels and ventilators (SOLAS 74/88 reg.II-1/19) (SOLAS 74/06/17 reg.II-1/16-1);
(CI)	2.1.4.7	checking the ballasting arrangements (SOLAS 74/06/17 reg.II-1/20);
(CI)	2.1.4.8	confirming that each bilge pump and the bilge pumping system provided for each watertight compartment are working efficiently (SOLAS 74/88 reg.II-1/21) (SOLAS 74/05/17 reg.II-1/35-1);

- (CI) 2.1.4.9 confirming that the drainage system of enclosed cargo spaces situated on the freeboard deck is working efficiently (SOLAS 74/88 reg.II-1/21) (SOLAS 74/05/17 reg.II-1/35-1);
- (CI) 2.1.4.9.1 examining visually the drainage facilities for blockage or other damage and confirming the provision of means to prevent blockage of drainage arrangements, for closed vehicle and ro-ro spaces and special category spaces where fixed pressure water-spraying systems are used (SOLAS 74/08 reg.II-2/20.6.1.5);
- (CI) 2.1.4.10 conducting an inclining test, when this is required (SOLAS 74/88 reg.II-1/22) (SOLAS 74/06/17 reg.II-1/5);
- (CI) 2.1.4.11 confirming that the machinery, boilers and other pressure vessels, associated piping systems and fittings are installed and protected so as to reduce to a minimum any danger to persons on board, due regard being given to moving parts, hot surfaces and other hazards (SOLAS 74/00/15 reg.II-2/4.2 (except 4.2.2.3.4 relating to remote closing of valves included in safety equipment)) (SOLAS 74/88 regs.II-1/26, 32, 33 and 34) (SOLAS 74/88/06 reg.II-2/15 (except 15.2.5));
- (CI) 2.1.4.12 confirming that the normal operation of the propulsion machinery can be sustained or restored even though one of the essential auxiliaries becomes inoperative (SOLAS 74/88 reg.II-1/26);
- (CI) 2.1.4.13 confirming that means are provided so that the machinery can be brought into operation from the dead ship condition without external aid (SOLAS 74/88 reg.II-1/26);
- (CI) 2.1.4.14 confirming that the boilers, all parts of the machinery, all steam, hydraulic, pneumatic and other systems and their associated fittings which are under internal pressure have been subjected to the appropriate tests, including a pressure test as may be specified in the requirements of the Administration or the classification societies (SOLAS 74/88 reg.II-1/26);
- (CI) 2.1.4.15 confirming that means are provided to ensure that the safe speed is not exceeded where there is the risk of machinery overspeeding (SOLAS 74/88 reg.II-1/27);
- (CI) 2.1.4.16 confirming that, where practicable, means are provided to protect against overpressure in the parts of main, auxiliary and other machinery that are subject to internal pressure and may be subject to dangerous overpressure (SOLAS 74/88 reg.II-1/27);
- (CI) 2.1.4.17 confirming that, when required, crankcase explosion relief devices are fitted to internal combustion engines and that they are arranged so as to minimize the possibility of injury to personnel (SOLAS 74/88 reg.II-1/27);
- (CI) 2.1.4.18 confirming that main turbine propulsion machinery and, where applicable, main internal combustion propulsion machinery and auxiliary machinery are provided with automatic shut-off arrangements in the case of failures, such as lubricating oil supply failure, which could rapidly

lead to a complete breakdown, serious damage or explosion (SOLAS 74/88 reg.II-1/27);

- (CI) 2.1.4.19 confirming and recording the ability of the machinery to reverse the direction of the thrust of the propeller in sufficient time and to bring the ship to rest within a reasonable distance, including the effectiveness of any supplementary means of manoeuvring or stopping the ship¹⁴ (SOLAS 74/88 reg.II-1/28);
- (CI) 2.1.4.20 confirming that the main and auxiliary steering gear are so arranged that the failure of one of them does not render the other inoperative¹⁵ (SOLAS 74/88/14 reg.II-1/29);
- (CI) 2.1.4.21 confirming that, where appropriate, essential components of the steering gear are permanently lubricated or provided with lubrication fittings (SOLAS 74/88/14 reg.II-1/29);
- (CI) 2.1.4.22 confirming that relief valves are fitted to any part of a steering gear hydraulic system which can be isolated and in which pressure can be generated from the power source or from external forces and that these relief valves are set to a pressure not exceeding the design pressure (SOLAS 74/88/14 reg.II-1/29);
- (CI) 2.1.4.23 confirming that the main steering gear is capable of steering the ship at maximum ahead service speed and is capable of putting the rudder over from 35° on one side to 35° on the other side with the ship at its deepest seagoing draught and running ahead at maximum ahead service speed and, under the same conditions, from 35° on either side to 30° on the other side in not more than 28 s,¹⁶ or, where demonstration at the deepest seagoing draught is impracticable, with alternative permissible sea trial loading condition¹⁷ (SOLAS 74/88 reg.II-1/29);
- (CI) 2.1.4.24 confirming that the auxiliary steering gear is capable of steering the ship at navigable speed and of being brought speedily into action in an emergency and that it is capable of putting the rudder over from 15° on one side to 15° on the other side in not more than 60 s with the ship at its deepest seagoing draught and running ahead at one half of the maximum ahead service speed or 7 knots, whichever is the greater,¹⁸

¹⁴ For ships fitted with alternative propulsion and steering arrangements other than traditional arrangements, such as but not limited to azimuthing propulsors or water jet propulsion systems, refer to the *Unified interpretations of SOLAS regulations II-1/28, II-1/29 and II-1/30* (MSC.1/Circ.1416/Rev.1).

¹⁵ For ships fitted with alternative propulsion and steering arrangements other than traditional arrangements, such as but not limited to azimuthing propulsors or water jet propulsion systems, refer to the *Unified interpretations of SOLAS regulations II-1/28, II-1/29 and II-1/30* (MSC.1/Circ.1416/Rev.1).

¹⁶ For ships fitted with alternative propulsion and steering arrangements other than traditional arrangements, such as but not limited to azimuthing propulsors or water jet propulsion systems, refer to the *Unified interpretations of SOLAS regulations II-1/28, II-1/29 and II-1/30* (MSC.1/Circ.1416/Rev.1).

¹⁷ Refer to the Unified interpretations of SOLAS regulations II-1/29.3 and II-1/29.4 (MSC.1/Circ.1536).

¹⁸ For ships fitted with alternative propulsion and steering arrangements other than traditional arrangements, such as but not limited to azimuthing propulsors or water jet propulsion systems, refer to the Unified interpretations of SOLAS regulations II-1/28, II-1/29 and II-1/30 (MSC.1/Circ.1416/Rev.1).

or, where this is impracticable, with an alternative permissible sea trial loading condition¹⁹ (SOLAS 74/88/14 reg.II-1/29);

- (CI) 2.1.4.25 confirming that the main and auxiliary steering gear power units restart automatically when power is restored after a power failure, that they are capable of being brought into operation from a position on the navigating bridge and that, in the event of a power failure to any one of the steering gear power units, an audible and visual alarm is given on the navigating bridge (SOLAS 74/88/14 reg.II-1/29);
- (CI) 2.1.4.26 confirming that, where the main steering gear comprises two or more identical power units and an auxiliary steering gear is not fitted, a defect can be isolated so that steering capability can be maintained or speedily regained after a single failure in its piping system or in one of the power units²⁰ (SOLAS 74/88/14 reg.II-1/29);
- (CI) 2.1.4.27 confirming that the control systems for the main steering gear from both the navigating bridge and the steering gear compartment are operating satisfactorily (SOLAS 74/8814 reg.II-1/29);
- (CI) 2.1.4.28 confirming that, where the main steering gear comprises two or more identical power units and an auxiliary steering gear is not fitted, the two independent control systems from the navigating bridge are operating satisfactorily (SOLAS 74/88/14 reg.II-1/29);
- (CI) 2.1.4.29 confirming that the control system for the auxiliary steering gear in the steering gear compartment and, if this gear is power-operated, from the navigating bridge are operating satisfactorily and that the latter is independent of the control system for the main steering gear (SOLAS 74/88/14 reg.II-1/29);
- (CI) 2.1.4.30 confirming that the control system for any main and auxiliary steering gear control system operable from the navigating bridge is capable of being brought into operation from a position on the navigating bridge, that means are provided in the steering gear compartment for disconnecting it from the steering gear that it serves, and that an audible and visual alarm is given on the navigating bridge in the event of a failure of electrical power supply (SOLAS 74/88/14 reg.II-1/29);
- (CI) 2.1.4.31 confirming that the electric power circuits and steering gear control systems, together with their associated components, cables and pipes, are separated, as far as practicable, throughout their length (SOLAS 74/88/14 reg.II-1/29);
- (CI) 2.1.4.32 confirming that the means of communication between the bridge and the steering gear compartment is operating satisfactorily and that, with ships having emergency steering positions, a telephone or other means of communication for relaying heading information and supplying visual

¹⁹ Refer to the Unified interpretations of SOLAS regulations II-1/29.3 and II-1/29.4 (MSC.1/Circ.1536).

For ships fitted with alternative propulsion and steering arrangements other than traditional arrangements, such as but not limited to azimuthing propulsors or water jet propulsion systems, refer to the Unified interpretations of SOLAS regulations II-1/28, II-1/29 and II-1/30 (MSC.1/Circ.1416/Rev.1).

compass readings to the emergency steering position are provided (SOLAS 74/88/14 reg.II-1/29) (SOLAS 74/00 reg.V/19);

- (CI) 2.1.4.33 confirming that the angular position of the rudder is indicated independently of the steering control system on the navigating bridge if the main steering gear is power-operated and that this angular position is given in the steering gear compartment (SOLAS 74/88/14 reg.II-1/29) (SOLAS 74/00 reg.V/19);
- (CI) 2.1.4.34 confirming that with a hydraulic power-operated steering gear the audible and visual low-level alarms on the navigating bridge and in the machinery space for each hydraulic fluid reservoir are operating satisfactorily and that at least one power actuating system including the reservoir can be recharged from a position within the steering gear compartment by means of a fixed storage tank (to which a contents gauge is fitted) with fixed piping (SOLAS 74/88/14 reg.II-1/29);
- (CI) 2.1.4.35 confirming that the steering gear compartment is readily accessible, that it is separated, as far as practicable, from machinery spaces and that it is provided with suitable arrangements to ensure working access to steering gear machinery and controls under safe conditions (SOLAS 74/88/14 reg.II-1/29);
- (CI) 2.1.4.36 confirming that with electric and electro-hydraulic steering gear the means are provided for indicating on the navigating bridge and at a main machinery control position that the motors are running and that the overload alarm and alarm for the loss of a phase in a three-phase supply located at the main machinery control position are operating satisfactorily (SOLAS 74/88 reg.II-1/30);
- (CI) 2.1.4.37 confirming that the main and auxiliary machinery essential for propulsion and the safety of the ship are provided with the effective means for its operation and control (SOLAS 74/88 reg.II-1/31);
- (CI) 2.1.4.38 confirming that appropriate means are provided where it is intended that the propulsion machinery should be remotely controlled from the navigating bridge, including, where necessary, the control, monitoring, reporting, alert and safety actions (SOLAS 74/00/02 reg.II-1/31);
- (CI) 2.1.4.39 confirming that arrangements to operate main and other machinery from a machinery control room are satisfactory (SOLAS 74/88 reg.II-1/31);
- (CI) 2.1.4.40 confirming that, in general, means are provided for manually overriding automatic controls and that a failure does not prevent the use of the manual override (SOLAS 74/88 reg.II-1/31);
- (CI) 2.1.4.41 confirming that oil-fired and exhaust gas boilers, unfired steam generators, steam pipe systems and air pressure systems are fitted with the appropriate safety features (SOLAS 74/88 regs.II-1/32, 33 and 34);
- (CI) 2.1.4.42 confirming the operation of the ventilation for the machinery spaces (SOLAS 74/88 reg.II-1/35);

- (CI) 2.1.4.43 when appropriate, confirming that the measures to prevent noise in machinery spaces are effective (SOLAS 74/88 reg.II-1/36 and SOLAS 74/12/16 reg.II-1/3-12.2); or confirming that the ship was constructed to reduce onboard noise and to protect personnel from noise in accordance with the Code on Noise Levels on board Ships, adopted by resolution MSC.337(91), as amended (SOLAS 74/12/16 reg.II-1/3-12);
- (CI) 2.1.4.44 confirming that the engine-room telegraph giving visual indication of the orders and answers both in the machinery space and on the navigating bridge is operating satisfactorily (SOLAS 74/88 reg.II-1/37);
- (CI) 2.1.4.45 confirming that the second means of communication between the navigation bridge and machinery space is also operating satisfactorily and that appropriate means are provided to any other positions from which the engines are controlled (SOLAS 74/88 reg.II-1/37);
- (CI) 2.1.4.46 confirming that the engineer's alarm is clearly audible in the engineers' accommodation (SOLAS 74/88 reg.II-1/38);
- (CI) 2.1.4.47 confirming that precautions, taken to prevent any oil that may escape under pressure from any pump, filter or heater from coming into contact with heated surfaces, are efficient (SOLAS 74/00 reg.II-2/4.2.2.3);
- (CI) 2.1.4.48 confirming that the means of ascertaining the amount of oil contained in any oil tank are in good working condition (SOLAS 74/00 reg.II-2/4.2.2.3.5);
- (CI) 2.1.4.49 confirming that the devices provided to prevent overpressure in any oil tank or in any part of the oil system, including the filling pipes, are in good working condition (SOLAS 74/00 reg.II-2/4.2.2.4);
- (CI) 2.1.4.50 confirming that forepeak tanks are not intended for carriage of oil fuel, lubrication oil and other flammable oils (SOLAS 74/00 reg.II-2/4.2.2.3);
- (CI) 2.1.4.51 confirming that the electrical installations, including the main source of power and lighting systems, are installed in accordance with the approved plans (SOLAS 74/88 regs.II-1/40 and 41);
- (CI) 2.1.4.52 confirming that a self-contained emergency source of electrical power has been provided and that the appropriate systems are satisfactorily supplied (SOLAS 74/88 reg.II-1/43);
- (CI) 2.1.4.53 confirming that the starting arrangements of each emergency generating set are satisfactory (SOLAS 74/88 reg.II-1/44);
- (CI) 2.1.4.54 confirming that precautions have been provided against shock, fire and other hazards of electrical origin (SOLAS 74/88 reg.II-1/45);
- (CI) 2.1.4.55 confirming that the arrangements for periodically unattended machinery spaces are satisfactory (SOLAS 74/88 regs.II-1/46 to 53) and in particular:
- (CI) 2.1.4.55.1 checking the fire precautions and testing alarms, as appropriate;

MSC 104/18/Add.1 Annex 31, page 60

- (CI) 2.1.4.55.2 checking the means for the protection against flooding;
- (CI) 2.1.4.55.3 checking the means to control the propulsion from the navigating bridge;
- (CI) 2.1.4.55.4 ensuring that a means of vocal communication between the main machinery control room or its control position, as appropriate, and the navigating bridge and engineer officer's accommodation is provided and is effective;
- (CI) 2.1.4.55.5 checking that an alarm system is provided with random testing of functions;
- (CI) 2.1.4.55.6 checking that means are provided to automatically shut down machinery or boiler operations in the event of serious malfunction, and testing the alarms;
- (CI) 2.1.4.55.7 ensuring that special requirements for the machinery, boiler and electrical installations, as appropriate, are provided;
- (CI) 2.1.4.56 examining, where applicable, the alternative design and arrangements for machinery or electrical installations or low-flashpoint fuel storage and distribution systems, or fire safety, in accordance with the test and inspection requirements, if any, specified in the approved documentation (SOLAS 74/00/06/15 regs.II-1/55 and II-2/17 and IGF Code ch.2);
- (CI) 2.1.4.57 confirming that all aspects of the structural fire protection, including the ventilation systems, in accommodation and service spaces, control stations and machinery spaces are installed in accordance with the approved plans, testing the operation of fire dampers of ventilation ducts and the means of closing the main inlets and outlets of all ventilation systems and proving that the power ventilation is capable of being stopped from outside the space served (SOLAS 74/00/12/14 regs.II-2/4.4, 5.2, 5.3.1, 5.3.2, 6.2, 6.3, 7.5.5, 7.7, 8.2, 8.4, 9.2.1, 9.3, 9.4.2, 9.5, 9.7.1, 9.7.2, 9.7.3, 9.7.5.2, 9.7.6, 11.2, 11.3, 11.4 and 11.5) (SOLAS 74/88 regs.II-2/42 to 44, 46 to 50 and 52);
- (CI) 2.1.4.58 confirming that all aspects of the structural fire protection, including the ventilation systems, in cargo spaces are installed in accordance with the approved plans, testing the operation of fire dampers of ventilation ducts and the means of closing the main inlets and outlets of all ventilation systems and proving that the power ventilation is capable of being stopped from outside the space served (SOLAS 74/00/15/17 regs.II-2/5.2.1, 9.7.1, 9.7.2, 9.7.3, 9.7.6, 11.2, 11.3, 11.5, 19.3.8, 19.3.10, 20.2.1, 20.3, 20-1.2.1, 20-1.3 and 20-1.4) (SOLAS 74/88 regs.II-2/42 to 44, 46 to 50 and 52 to 54);
- (CI) 2.1.4.59 confirming that stairways and ladders are so arranged as to provide a means of escape from all accommodation spaces and from spaces in which the crew is normally employed, other than machinery spaces, to the open deck and thence to the lifeboats and liferafts (SOLAS 74/00 regs.II-2/13.2, 13.3.1, 13.3.3 and13.6; FSS Code ch.13 paragraph 3) (SOLAS 74/88 reg.II-2/45) and in particular that:

- (CI) 2.1.4.59.1 at all levels of accommodation there are provided at least two widely separated means of escape from each restricted space or group of spaces;
- (CI) 2.1.4.59.2 below the lowest open deck the main means of escape is a stairway (the second being a trunk or a stairway);
- (CI) 2.1.4.59.3 above the lowest open deck the means of escape are stairways or doors to an open deck or a combination of them;
- (CI) 2.1.4.59.4 the radiotelegraph station has direct access to the open deck or is provided with two means of access or egress, one of which is a porthole or window of sufficient size;
- (CI) 2.1.4.60 confirming that two widely separated means of escape and, when appropriate, a fire shelter from the lower part of the space, are provided from each machinery space of Category A and that suitable escape routes are provided from other machinery spaces, and that two means of escape are provided for machinery control rooms and for main workshops located within the machinery space of Category A, as applicable (SOLAS 74/00/14 reg.II-2/13.4.2; FSS Code ch.13 paragraph 3) (SOLAS 74/88 reg.II-2/45);
- (CI) 2.1.4.61 examining the arrangements for gaseous fuel for domestic purposes (SOLAS 74/00 reg.II-2/4.3);
- (CI) 2.1.4.62 confirming, when appropriate, that all aspects of the helicopter facilities are installed in accordance with the approved plans (SOLAS 74/00/16 reg.II-2/18) (SOLAS 74/88 reg.II-2/18.8);
- (CI) 2.1.4.63 confirming that installed materials do not contain asbestos²¹ (SOLAS 74/00/09 reg.II-1/3-5);
- (CI) 2.1.4.64 confirming, for bulk carriers, that dedicated seawater ballast tanks have an efficient corrosion protection system such as hard coating (SOLAS 74/00 reg.II-1/3-2);
- (CI) 2.1.4.65 confirming that dedicated seawater ballast tanks arranged in ships and double side skin spaces arranged in bulk carriers of 150 m in length and upward when appropriate have been coated in accordance with resolution MSC.215(82), as amended (SOLAS 74/00/06 reg.II-1/3-2);
- (CI) 2.1.4.66 prior to the review of the coating technical file:
- (CI) 2.1.4.66.1 checking that the Technical Data Sheet and Statement of Compliance or Type Approval Certificate comply with the Standard;
- (CI) 2.1.4.66.2 checking that the coating identification on representative containers is consistent with the coating identified in the Technical Data Sheet;
- (CI) 2.1.4.66.3 checking that the inspector is qualified in accordance with the qualification standards;

²¹ Refer to the *Unified interpretation of SOLAS regulation II-1/3-5* (MSC.1/Circ.1379 and MSC.1/Circ.1426/Rev.1).

- (CI) 2.1.4.66.4 checking that the inspector's reports of surface preparation and the coating's application indicate compliance with the manufacturer's Technical Data Sheet and Statement of Compliance or Type Approval Certificate; and
- (CI) 2.1.4.66.5 monitoring the implementation of the coating inspection requirements;
- (CI) 2.1.4.67 reviewing the coating technical file (SOLAS 74/00/06/10 regs.II-1/3-2 and II-1/3-11; MSC.215(82), as amended, and MSC.288(87), as amended);
- (CI) 2.1.4.68 confirming for oil tankers and bulk carriers, when appropriate, the provision of means of access to cargo and other spaces in accordance with the arrangements in the Ship Structures Access Manual (SOLAS 74/00/02/04 reg.II-1/3-6, SOLAS 74/10 reg.II-1/3-10 and MSC.287(87));
- (CI) 2.1.4.69 for bulk carriers, examining and testing the hold, ballast and dry space water level detectors and their audible and visual alarms (SOLAS 74/02 reg.XII/12);
- (CI) 2.1.4.70 for bulk carriers, checking the arrangements for availability of draining and pumping systems forward of the collision bulkhead (SOLAS 74/02 reg.XII/13);
- (CI) 2.1.4.71 confirming, for bulk carriers, that the loading instrument is on board and functioning (SOLAS 74/97/04 reg.XII/11);
- (CI) 2.1.4.72 confirming that the ship's identification number is permanently marked (SOLAS 74/02 reg.XI-1/3);
- (CI) 2.1.4.73 confirming that the towing and mooring equipment is properly marked with any restriction associated with its safe operation (SOLAS 74/04 reg.II-1/3-8); and
- (CI) 2.1.4.74 confirming, where applicable, that an appropriate portable atmosphere testing instrument or instruments²² is on board, and that suitable means are provided for the calibration of all such instruments;²³ and checking the appropriateness of the testing and calibration (SOLAS 74/14 reg.XI-1/7).
- (CI) 2.1.5 For the hull, machinery and equipment of cargo ships, concerning the additional requirements for oil tankers the survey during construction and after installation should consist of:
- (CI) 2.1.5.1 confirming, when appropriate, that the main steering gear comprises the necessary two or more identical power units and the requisite arrangements to regain steering capability in the event of the prescribed single failure (SOLAS 74/88/14 reg.II-1/29);

²² Refer to the Guidelines to facilitate the selection of portable atmosphere testing instruments for enclosed spaces as required by SOLAS regulation XI-1/7 (MSC.1/Circ.1477).

²³ Refer to the Unified interpretations of SOLAS regulation XIV/2.2 and paragraphs 1.3.2 and 1.3.6, part I-A of the Polar Code (MSC.1/Circ.1562).

- (CI) 2.1.5.2 confirming in accordance with the survey plan that oil tankers of 150 m in length and above, where appropriate, meet the applicable structural requirements of an organization recognized by the Administration, or national standards of the Administration, conforming to the functional requirements of the Goal-based Ship Construction Standards for Bulk Carriers and Oil Tankers (SOLAS 74/10 reg.II-1/3-10);
- (CI) 2.1.5.3 confirming that a hull return system of distribution and earthed distribution system are not used (SOLAS 74/88 reg.II-1/45);
- (CI) 2.1.5.4 confirming that all aspects of the location of spaces and the structural fire protection, including the special arrangements when the ship is a combination carrier, are in accordance with the approved plans (SOLAS 74/00/12 regs.II-2/1.6, 4.5.1, 4.5.2, 4.5.9, 9.2.4, 9.3 and 9.6.3) (SOLAS 74/88 regs.II-2/55 to 58);
- (CI) 2.1.5.5 confirming that permanent approved gastight lighting enclosures for illuminating cargo pump-rooms, having adequate strength and not impairing the integrity and gas tightness of the bulkheads or decks, are fitted in bulkheads and decks separating cargo pump-rooms and other spaces (SOLAS 74/00 reg.II-2/4.5.2.5) (SOLAS 74/88 reg.II-2/58.5);
- (CI) 2.1.5.6 confirming that all aspects of the cargo tank venting, cargo tank purging and gas freeing and other ventilation arrangements and protection of the cargo tank structure against pressure or vacuum are in accordance with the approved plans (SOLAS 74/00/15 regs.II-2/4.5.3, 4.5.4, 4.5.6, 4.5.8 and 11.6) (SOLAS 74/88 regs.II-2/59 and 62.13.1 to 62.13.3);
- (CI) 2.1.5.7 confirming that access to bow is arranged in accordance with approved plans (SOLAS 74/00/04 reg.II-1/3-3);
- (CI) 2.1.5.8 confirming, for tankers of not less than 20,000 tonnes deadweight, that emergency towing is arranged in accordance with approved plans (SOLAS 74/00/04 reg.II-1/3-4);
- (CI) 2.1.5.9 confirming, when appropriate, that dedicated seawater ballast tanks have an efficient corrosion protection system such as hard coating (SOLAS 74/00/06 reg.II-1/3-2);
- (CI) 2.1.5.10 confirming that all cargo oil tanks in crude oil tankers have either:
- (CI) 2.1.5.10.1 been coated in accordance with MSC.288(87), as amended; or
- (CI) 2.1.5.10.2 been protected by alternative means of corrosion protection or utilization of approved corrosion-resistant material (steel) in accordance with MSC.289(87) (SOLAS 74/10 reg.II-1/3-11).
- (CI) 2.1.6 For the hull, machinery and equipment of cargo ships, concerning the additional requirements for chemical tankers and gas carriers, the survey during construction and after installation should consist of:
- (CI) 2.1.6.1 the provisions of (CI) 2.1.5 except (CI) 2.1.5.2.

(CI)	2.1.7	For the hull, machinery and equipment of cargo ships, concerning the
		additional requirements for the ships using natural gas as fuel other than
		ships covered by the IGC Code, the survey during construction and after
		installation should consist of:

- (CI) 2.1.7.1 confirming that the arrangement of the accommodation, the fuel containment system, service and machinery spaces are in accordance with the approved plans and that control, monitoring and safety systems are satisfactory (IGF Code chs.4, 5, 6, 8, 9 and 15);
- (CI) 2.1.7.2 confirming the inert gas system is satisfactory (IGF Code ch.6);
- (CI) 2.1.7.3 confirming that the fuel containment systems are arranged and installed in accordance with the approved plans, internally examining the fuel containment systems and ensuring that the appropriate testing is carried out (IGF Code chs.6 and 16);
- (CI) 2.1.7.4 examining of machinery installations (IGF Code ch.10);
- (CI) 2.1.7.4.1 ventilation systems;
- (CI) 2.1.7.4.2 dual-fuel engines;
- (CI) 2.1.7.4.3 gas-only engines;
- (CI) 2.1.7.4.4 multi-fuel engines;
- (CI) 2.1.7.4.5 main and auxiliary boilers;
- (CI) 2.1.7.4.6 gas turbines;
- (CI) 2.1.7.5 confirming that the fire protection is installed in accordance with the approved plans (IGF Code ch.11.3);
- (CI) 2.1.7.6 confirming the ventilation arrangements are satisfactory (IGF Code chs.12 and 13); and
- (CI) 2.1.7.7 examining the electrical installations with particular reference to the certified safe type equipment fitted in gas-dangerous spaces and zones (IGF Code chs.12 and 14).
- (CI) 2.1.8 For the hull, machinery and equipment of cargo ships the check that the required documentation has been placed on board should consist of:
- (CI) 2.1.8.1 confirming that the stability information and the damage control plans and damage control booklets have been provided (SOLAS 74/88 regs.II-1/22 and 23-1) (SOLAS 74/06/17 regs.II-1/5-1 and 19);
- (CI) 2.1.8.2 checking, where applicable, that the noise survey report as required by the Code on Noise Levels on Board Ships is available on board (SOLAS 74/12/16 reg.II-1/3-12);

- (CI) 2.1.8.3 confirming that the manoeuvring booklet has been provided and that the manoeuvring information has been displayed on the navigating bridge (SOLAS 74/88 reg.II-1/28);
- (CI) 2.1.8.4 confirming that the approved Cargo Securing Manual for ships carrying cargoes other than solid and liquid bulk cargoes, cargo units and cargo transport units is provided on board (SOLAS 74/98/02 reg.VI/5.6);
- (CI) 2.1.8.5 confirming, for oil tankers and bulk carriers when appropriate, that the Ship Structure Access Manual is on board (SOLAS 74/00/02/04 reg.II-1/3-6(4));
- (CI) 2.1.8.6 confirming that a set of as-built construction drawings is available on board (SOLAS 74/04 reg.II-1/3-7);
- (CI) 2.1.8.7 confirming when appropriate that a coating technical file reviewed by the Administration has been provided on board (SOLAS 74/00/06/10 regs.II-1/3-2 and 3-11);
- (CI) 2.1.8.8 checking the provision of a ship-specific emergency towing procedure (SOLAS 74/08 reg.II-1/3-4);
- (CI) 2.1.8.9 confirming, for oil tankers and bulk carriers of 150 m in length and above, that the Ship Construction File has been provided (SOLAS 74/10 reg.II-1/3-10 and MSC.290(87));
- (CI) 2.1.8.10 confirming, when appropriate, that a technical file verified by the Administration has been provided on board (SOLAS 74/10 reg.II-1/3 11 and MSC.289(87)); and
- (CI) 2.1.8.11 confirming that, where applicable, the approved documentation for the alternative design and arrangements is on board (SOLAS 74/00/06/15 regs.II-1/55 and II-2/17 and IGF Code ch.2).
- (CI) 2.1.9 For the hull, machinery and equipment of cargo ships the completion of the initial survey should consist of:
- (CI) 2.1.9.1 after a satisfactory survey, the Cargo Ship Safety Construction Certificate should be issued.
- (CA) **2.2 Annual surveys** see part "General", section 4.2.
- (CA) 2.2.1 For the hull, machinery and equipment of cargo ships the examination of current certificates and other records should consist of:
- (CA) 2.2.1.1 checking the validity, as appropriate, of the Cargo Ship Safety Equipment Certificate, the Cargo Ship Safety Radio Certificate and the Cargo Ship Safety Construction Certificate or the Cargo Ship Safety Certificate;
- (CA) 2.2.1.2 checking, as appropriate, the validity of the Polar Ship Certificate;
- (CA) 2.2.1.3 checking the validity of the Safety Management Certificate (SMC) and that a copy of the Document of Compliance (DOC) is on board;

(CA)	2.2.1.4	checking the validity of the International Ship Security Certificate;
(CA)	2.2.1.5	checking the validity of the International Load Line Certificate or International Load Line Exemption Certificate;
(CA)	2.2.1.6	checking the validity of the International Oil Pollution Prevention Certificate;
(CA)	2.2.1.7	checking the certificates of class, if the ship is classed with a classification society;
(CA)	2.2.1.8	checking, when appropriate, the validity of the International Certificate of Fitness for the Carriage of Dangerous Chemicals in Bulk or the Certificate of Fitness for the Carriage of Dangerous Chemicals in Bulk;
(CA)	2.2.1.9	checking, when appropriate, the validity of the International Certificate of Fitness for the Carriage of Liquefied Gases in Bulk;
(CA)	2.2.1.10	checking, when appropriate, the validity of the International Pollution Prevention Certificate for the Carriage of Noxious Liquid Substances in Bulk;
(CA)	2.2.1.11	checking, when appropriate, the validity of the International Sewage Pollution Prevention Certificate;
(CA)	2.2.1.12	checking, when appropriate, the validity of the International Air Pollution Prevention Certificate;
(CA)	2.2.1.13	confirming, when appropriate, the validity of the International Energy Efficiency Certificate (MARPOL Annex VI, regs.6.4 and 6.5);
(CA)	2.2.1.14	confirming, when appropriate, that confirmation of compliance for the SEEMP part II is provided to and retained on board the ship (MARPOL Annex VI, reg. 5.4.5) ²⁴ ;
(CA)	2.2.1.15	confirming, when appropriate, the validity of the Statements of Compliance related to fuel oil consumption reporting (MARPOL Annex VI, regs.6.6 and 6.7);
(CA)	2.2.1.16	checking, when appropriate, the validity of the International Ballast Water Management Certificate;
(CA)	2.2.1.17	checking that the ship's complement complies with the Minimum Safe Manning Document (SOLAS 74/00/12 reg.V/14) (SOLAS 74/88 reg.V/13(b));
(CA)	2.2.1.18	checking that the master, officers and ratings are certificated as required by the STCW Convention;

²⁴ Refer to the Sample Format of Confirmation of Compliance, Early Submission of the SEEMP part II on the ship fuel oil consumption data collection plan and its timely verification pursuant to regulation 5.4.5 of MARPOL Annex VI (MEPC.1/Circ.876).

- (CA) 2.2.1.19 checking, where applicable, that the noise survey report as required by the Code on Noise Levels on Board Ships is available on board (SOLAS 74/12/16 reg.II-1/3-12);
- (CA) 2.2.1.20 confirming that, where applicable, the approved documentation for the alternative design and arrangements is on board (SOLAS 74/00/06/15 regs.II-1/55 and II-2/17);
- (CA) 2.2.1.21 checking whether any new equipment has been fitted and, if so, confirm that it has been approved before installation and that any changes are reflected in the appropriate certificate;
- (CA) 2.2.1.22 checking the provision of a ship-specific emergency towing procedure (SOLAS 74/08 reg.II-1/3-4);
- (CA) 2.2.1.23 confirming that the stability information, including damage stability, where applicable, and the damage control plans and damage control booklets are on board (SOLAS 74/88 regs.II-1/22, 23 and 25) (SOLAS 74/06/17 regs.II-1/5-1 and 19);
- (CA) 2.2.1.24 confirming from the logbook entries, where applicable, that the watertight doors or ramps fitted internally to subdivide large cargo spaces required to be closed during navigation are being kept closed (SOLAS 74/06/17 reg.II-1/24);
- (CA) 2.2.1.25 confirming that the manoeuvring booklet is on board and that the manoeuvring information is displayed on the navigating bridge (SOLAS 74/88 reg.II-1/28);
- (CA) 2.2.1.26 checking by the logbook entries that the testing and the emergency drills of the steering gear have been carried out (SOLAS 74/00 reg.V/26) (SOLAS 74/88 reg.V/19);
- (CA) 2.2.1.27 checking that the routine surveys of the boilers and other pressure vessels, as determined by the Administration, have been carried out as required and that safety devices, such as the boiler safety valves, have been tested;
- (CA) 2.2.1.28 checking that, as appropriate, the hull and machinery has been presented for survey in accordance with the continuous survey scheme approved by the Administration or a classification society;
- (CA) 2.2.1.29 confirming, when appropriate, that a complete file of the enhanced survey reports and the Condition Evaluation Report are on board;²⁵
- (CA) 2.2.1.30 confirming that suitable Material Safety Data Sheets are available on board;
- (CA) 2.2.1.31 confirming, for bulk carriers, that the loading/unloading booklet required in SOLAS regulation VI/7.2 is on board (SOLAS 74/97/04 reg.XII/8.1);

²⁵ Refer to the International Code on the Guidelines on the Enhanced Programme of Inspections during Surveys of Bulk Carriers and Oil Tankers, 2011 (resolution A.1049(27)), as amended.

- (CA) 2.2.1.32 confirming, that bulk carriers of 150 m in length and upwards of single side skin construction designed to carry solid bulk cargoes having a density of 1,780 kg/m³ and above, constructed before 1 July 1999, have, after the implementation date given in SOLAS 74/94/97 reg.XII/3, sufficient stability and strength to withstand flooding of the foremost cargo hold (SOLAS 74/97/04 regs.XII/3, 4 and 6);
- (CA) 2.2.1.33 confirming that approved Cargo Securing Manual for ships carrying cargoes other than solid and liquid bulk cargoes, cargo units and cargo transport units is on board (SOLAS 74/98/02 reg.VI/5.6);
- (CA) 2.2.1.34 confirming that the loading booklet for carriage of cargoes in bulk is on board (SOLAS 74/00 reg.VI/7);
- (CA) 2.2.1.35 confirming, for oil tankers and bulk carriers when appropriate, that the Ship Structure Access Manual is on board (SOLAS 74/00/02, reg.II-1/3-6(4));
- (CA) 2.2.1.36 confirming that structural alterations performed, if any, have been approved by the classification society and reported on the as-built drawings kept on board (SOLAS 74/04 reg.II-1/3-7);
- (CA) 2.2.1.37 confirming, when appropriate, that the coating technical file is available on board and maintained (SOLAS 74/00/06/10 regs.II-1/3-2 and 3-11);
- (CA) 2.2.1.38 confirming, when appropriate, that the maintenance of the protective coating is included in the overall ship's maintenance system (SOLAS 74/00/06 reg.II-1/3-2);
- (CA) 2.2.1.39 confirming, where appropriate, for crude oil tankers, that a technical file verified by the Administration has been provided on board (SOLAS 74/10 reg.II-1/3-11 and MSC.289(87));
- (CA) 2.2.1.40 confirming, for oil tankers and bulk carriers of 150 m in length and above, that the Ship Construction File is available and updated, where applicable²⁶ (SOLAS 74/10 reg.II-1/3-10 and MSC.287(87)); and
- (CA) 2.2.1.41 confirming the availability of the International Anti-fouling System Certificate (AFS 2001 annex 4 reg.2), when applicable.
- (CA) 2.2.2 For the hull,²⁷ machinery and equipment of cargo ships the annual survey should consist of:
- (CA) 2.2.2.1 examining, in general and as far as can be seen, the hull and its closing appliances;
- (CA) 2.2.2.2 examining the anchoring and mooring equipment as far as can be seen; for ships built after 1 January 2007, confirming that the towing and

²⁶ Refer also to the International Code on the Enhanced Programme of Inspections during Surveys of Bulk Carriers and Oil Tankers, 2011 (resolution A.1049(27)), as amended.

²⁷ Refer also to annex A to the *International Code on the Enhanced Programme of Inspections during Surveys of Bulk Carriers and Oil Tankers, 2011* (resolution A.1049(27)), as amended.

mooring equipment is properly marked with any restriction associated with its safe operation (SOLAS 74/04 reg.II-1/3-8);

- (CA) 2.2.2.3 examining, for bulk carriers of 150 m and above, where appropriate, the ship's structure in accordance with the Ship Construction File, taking into account identified areas that need special attention (SOLAS 74/10 reg.II-1/3-10 and MSC.287(87));
- (CA) 2.2.2.4 examining the collision and the other watertight bulkheads as far as can be seen (SOLAS 74/88 regs.II-1/11 and 14) (SOLAS 74/06/17 regs.II-1/10, 11 and 12);
- (CA) 2.2.2.5 examining and testing (locally and remotely) all the watertight doors in watertight bulkheads (SOLAS 74/88 reg.II-1/18) (SOLAS 74/06/17 reg.II-1/16);
- (CA) 2.2.2.6 examining the arrangements for closing openings in the shell plating below the freeboard deck (SOLAS 74/06/17 reg.II-1/15);
- (CA) 2.2.2.7 checking the ballasting arrangements (SOLAS 74/06/17 reg.II-1/20);
- (CA) 2.2.2.8 examining each bilge pump and confirming that the bilge pumping system for each watertight compartment is satisfactory (SOLAS 74/88 reg.II-1/21) (SOLAS 74/05/17 reg.II-1/35-1);
- (CA) 2.2.2.9 confirming that the drainage from enclosed cargo spaces situated on the freeboard deck is satisfactory (SOLAS 74/88 reg.II-1/21) (SOLAS 74/05/17 reg.II-1/35-1);
- (CA) 2.2.2.9.1 examining visually the drainage facilities for blockage or other damage and confirming the provision of means to prevent blockage of drainage arrangements, for closed vehicle and ro-ro spaces and special category spaces where fixed pressure water-spraying systems are used (SOLAS 74/08 reg.II-2/20.6.1.5);
- (CA) 2.2.2.10 confirming that the machinery, boilers and other pressure vessels, associated piping systems and fittings are installed and protected so as to reduce to a minimum any danger to persons on board, due regard being given to moving parts, hot surfaces and other hazards (SOLAS 74/00/15 reg.II-2/4.2 (except 4.2.2.3.4 relating to remote closing of valves included in safety equipment)) (SOLAS 74/88 regs.II-1/26, 32, 33 and 34) (SOLAS 74/88/06 reg.II-2/15 (except 15.2.5));
- (CA) 2.2.2.11 confirming that the normal operation of the propulsion machinery can be sustained or restored even though one of the essential auxiliaries becomes inoperative (SOLAS 74/88 reg.II-1/26);
- (CA) 2.2.2.12 confirming that means are provided so that the machinery can be brought into operation from the dead ship condition without external aid (SOLAS 74/88 reg.II-1/26);
- (CA) 2.2.2.13 carrying out a general examination of the machinery, the boilers, all steam, hydraulic, pneumatic and other systems and their associated fittings to see whether they are being properly maintained and with

particular attention to the fire and explosion hazards (SOLAS 74/88 regs.II-1/26 and 27);

- (CA) 2.2.2.14 examining and testing the operation of main and auxiliary steering arrangements, including their associated equipment and control systems (SOLAS 74/88/14 reg.II-1/29);
- (CA) 2.2.2.15 confirming that the means of communication between the navigation bridge and steering gear compartment and the means of indicating the angular position of the rudder are operating satisfactorily (SOLAS 74/88/14 reg.II-1/29) (SOLAS 74/00 reg.V/19);
- (CA) 2.2.2.16 confirming that with ships having emergency steering positions there are means of relaying heading information and, when appropriate, of supplying visual compass readings to the emergency steering position (SOLAS 74/88/14 reg.II-1/29 and SOLAS 74/00 reg.V/19 or the SOLAS 74/88 reg.V/12 in force prior to 1 July 2002 as appropriate);
- (CA) 2.2.2.17 confirming that the various alarms required for hydraulic power-operated, electric and electro-hydraulic steering gears are operating satisfactorily and that the re-charging arrangements for hydraulic power-operated steering gears are being maintained (SOLAS 74/88/14 reg.II-1/29 and SOLAS 74/88 reg.II-1/30);
- (CA) 2.2.2.18 examining the means for the operation of the main and auxiliary machinery essential for the propulsion and the safety of the ship, including, when applicable, the means of remotely controlling the propulsion machinery from the navigating bridge (including the control, monitoring, reporting, alert and safety actions) and the arrangements to operate the main and other machinery from a machinery control room (SOLAS 74/88/00/02 reg.II-1/31);
- (CA) 2.2.2.19 confirming the operation of the ventilation for the machinery spaces (SOLAS 74/88 reg.II-1/35);
- (CA) 2.2.2.20 when appropriate, confirming that the measures to prevent noise in machinery spaces are effective (SOLAS 74/88 reg.II-1/36 and SOLAS 74/12/16 reg.II-1/3-12.2); or confirming that the ship was constructed to reduce onboard noise and to protect personnel from noise in accordance with the Code on Noise Levels on Board Ships, adopted by resolution MSC.337(91), as amended (SOLAS 74/12/16 reg.II-1/3-12);
- (CA) 2.2.2.21 confirming that the engine-room telegraph, the second means of communication between the navigation bridge and the machinery space and the means of communication with any other positions from which the engines are controlled are operating satisfactorily (SOLAS 74/88 reg.II-1/37);
- (CA) 2.2.2.22 confirming that the engineer's alarm is clearly audible in the engineers' accommodation (SOLAS 74/88 reg.II-1/38);
- (CA) 2.2.2.23 examining, as far as practicable, visually and in operation, the electrical installations, including the main source of power and the lighting systems (SOLAS 74/88 regs.II-1/40 and 41);

- (CA) 2.2.2.24 confirming, as far as practicable, the operation of the emergency source(s) of electrical power including their starting arrangements, the systems supplied and, when appropriate, their automatic operation (SOLAS 74/88 regs.II-1/43 and 44);
- (CA) 2.2.2.25 examining, in general, that the precautions provided against shock, fire and other hazards of electrical origin are being maintained (SOLAS 74/88 reg.II-1/45);
- (CA) 2.2.2.26 examining the arrangements for periodically unattended machinery spaces (SOLAS 74/88 regs.II-1/46 to 53) and, in particular, the random testing of alarm, automatic and shutdown functions;
- (CA) 2.2.2.27 examining, where applicable, the alternative design and arrangements for machinery or electrical installations, low-flashpoint fuel storage and distribution systems, or fire safety, in accordance with the test, inspection and maintenance requirements, if any, specified in the approved documentation (SOLAS 74/00/06/15 regs.II-1/55 and II-2/17 and IGF Code ch.2);
- (CA) 2.2.2.28 confirming, as far as practicable, that no changes have been made in the structural fire protection, examining any manual and automatic fire doors and proving their operation, testing the fire dampers of ventilation ducts and the means of closing the main inlets and outlets of all ventilation systems and testing the means of stopping power ventilation systems from outside the space served (SOLAS 74/00/12/15/17 regs.II-2/4.4, 5.2, 5.3.1, 5.3.2, 6.2, 6.3, 7.5.5, 7.7, 8.2, 8.3, 8.4, 9.2.1, 9.2.3, 9.3, 9.4.2, 9.5, 9.7.1, 9.7.2, 9.7.3, 9.7.5.2, 9.7.6, 11.2, 11.3, 11.4, 11.5, 19.3.8, 19.3.10, 20.2.1, 20.3, 20-1.2.1, 20-1.3 and 20-1.4) (SOLAS 74/88 regs.II-2/42 to 44, 46 to 50 and 52);
- (CA) 2.2.2.29 confirming that the means of escape from accommodation, machinery and other spaces are satisfactory (SOLAS 74/00/14 regs.II-2/13.2, 13.3.1, 13.3.3, 13.4.2 and 13.6) (SOLAS 74/88 reg.II-2/45);
- (CA) 2.2.2.30 examining the helicopter facilities (SOLAS 74/00/16 reg.II-2/18) (SOLAS 74/88 reg.II-2/18.8);
- (CA) 2.2.2.31 examining the arrangements for gaseous fuel for domestic purposes (SOLAS 74/00 reg.II-2/4.3) (SOLAS 74/88 reg.II-2/51);
- (CA) 2.2.2.32 examining visually the condition of any expansion joints in seawater systems;
- (CA) 2.2.2.33 confirming, when appropriate and as far as is practicable when examining internal spaces on oil tankers and bulk carriers, that the means of access to cargo and other spaces remain in good condition (SOLAS 74/00/02 reg.II-1/3-6);
- (CA) 2.2.2.34 confirming that no new materials containing asbestos were installed on board²⁸ (SOLAS 74/00/04/09 reg.II-1/3-5);

²⁸ Refer to the Unified interpretation of SOLAS regulation II-1/3-5 (MSC.1/Circ.1379 and MSC.1/Circ.1426/Rev.1).

(CA)	2.2.2.35	examining the functionality of bilge well alarms to all cargo holds and conveyor tunnels (SOLAS 74/97/04 reg.XII/9);
(CA)	2.2.2.36	for bulk carriers, examining the hold, ballast and dry space water level detectors and their audible and visual alarms (SOLAS 74/02 reg.XII/12);
(CA)	2.2.2.37	for bulk carriers, checking the arrangements for availability of draining and pumping systems forward of the collision bulkhead (SOLAS 74/02 reg.XII/13);
(CA)	2.2.2.38	confirming that the ship's identification number is permanently marked (SOLAS 74/02 reg.XI-1/3);
(CA)	2.2.2.39	confirming, where applicable, that an appropriate portable atmosphere testing instrument or instruments ²⁹ is on board, and that suitable means are provided for the calibration of all such instruments; ³⁰ and checking the appropriateness of the testing and calibration (SOLAS 74/14 reg.XI-1/7);
(CA)	2.2.2.40	for single hull, single hold cargo ships, examining the cargo hold water level detector and its audible and visual alarm (SOLAS 74/04 reg.II-1/23-3) (SOLAS 74/06 reg.II-1/25);
(CA)	2.2.2.41	confirming that the coating system in dedicated SWB tanks in ships and double side skin spaces arranged in bulk carriers of 150 m in length and upward when appropriate is maintained and that maintenance, repair and partial recoating are recorded in the coating technical file (SOLAS 74/00/06 reg.II-1/3-2);
(CA)	2.2.2.42	confirming, for bulk carriers constructed before 1 July 1999 with restrictions imposed with respect to the carriage of cargoes with a density of 1,780 kg/m ³ and above, that a triangle is permanently marked at midship (SOLAS 74/97/04 reg.XII/8.3); and
(CA)	2.2.2.43	confirming, for bulk carriers, that the loading instrument is on board and functioning (SOLAS 74/97/04 reg.XII/11).
(CA)	2.2.3	For the hull, ³¹ machinery and equipment of cargo ships, concerning the additional requirements for oil tankers, the annual survey should consist of:
(CA)	2.2.3.1	confirming, when appropriate, that the requisite arrangements to regain steering capability in the event of the prescribed single failure are being maintained (SOLAS 74/88/14 reg.II-1/29);
(CA)	2.2.3.2	examining the cargo tank openings, including gaskets, covers, coamings and screens;

²⁹ Refer to the *Guidelines to facilitate the selection of portable atmosphere testing instruments for enclosed spaces* as required by SOLAS regulation XI-1/7 (MSC.1/Circ.1477).

³⁰ Refer to the Unified interpretations of SOLAS regulation XIV/2.2 and paragraphs 1.3.2 and 1.3.6, part I-A of the Polar Code (MSC.1/Circ.1562).

³¹ Refer also to annex B of the International Code on the Enhanced Programme of Inspections during Surveys of Bulk Carriers and Oil Tankers, 2011 (resolution A.1049(27)), as amended.

- (CA) 2.2.3.3 examining the cargo tank pressure/vacuum valves and devices to prevent the passage of flame (SOLAS 74/00/15 reg.II-2/11.6);
- (CA) 2.2.3.4 examining the devices to prevent the passage of flame on vents to all bunker, oily-ballast and oily-slop tanks and void spaces, as far as practicable;
- (CA) 2.2.3.5 examining the cargo tank venting, cargo tank purging and gas freeing and other ventilation systems (SOLAS 74/00/15 regs.II-2/4.5.3, 4.5.4, 4.5.6 and 4.5.8) (SOLAS 74/88 reg.II-2/59);
- (CA) 2.2.3.6 examining the cargo, crude oil washing, ballast and stripping systems both on deck and in the cargo pump-rooms and the bunker system on deck;
- (CA) 2.2.3.7 confirming that all electrical equipment in dangerous zones is suitable for such locations, is in good condition and is being properly maintained;
- (CA) 2.2.3.8 confirming that potential sources of ignition in or near the cargo pump-room are eliminated, such as loose gear, combustible materials, etc., that there are no signs of undue leakage and that access ladders are in good condition;
- (CA) 2.2.3.9 examining all pump-room bulkheads for signs of oil leakage or fractures and, in particular, the sealing arrangements of all penetrations of cargo pump-room bulkheads;
- (CA) 2.2.3.10 examining, as far as practicable, the cargo, bilge, ballast and stripping pumps for undue gland seal leakage, verification of proper operation of electrical and mechanical remote operating and shutdown devices and operation of cargo pump-room bilge system, and checking that pump foundations are intact;
- (CA) 2.2.3.11 confirming that the pump-room ventilation system is operational, ducting intact, dampers operational and screens clean;
- (CA) 2.2.3.12 verifying that installed pressure gauges on cargo discharge lines and level indicator systems are operational;
- (CA) 2.2.3.13 examining access to bow arrangement (SOLAS 74/00/04 reg.II-1/3-3);
- (CA) 2.2.3.14 examining the towing arrangement for tankers of not less than 20,000 tonnes deadweight (SOLAS 74/00/04 reg.II-1/3-4);
- (CA) 2.2.3.15 confirming that the corrosion prevention system fitted to dedicated ballast water tanks of oil tankers and bulk carriers when appropriate is maintained (SOLAS 74/00 reg.II-1/3-2);
- (CA) 2.2.3.16 confirming that the coating system in cargo oil tanks of crude oil tankers, when appropriate, is maintained and that in-service maintenance and repair activities are recorded in the coating technical file (SOLAS 74/10 reg.II-1/3-11 and MSC.288(87), as amended);
- (CA) 2.2.3.17 examining the emergency lighting in all cargo pump-rooms of tankers constructed after 1 July 2002 (SOLAS 74/00 reg.II-1/43); and

- (CA) 2.2.3.18 examining, for oil tankers of 150 m in length and above, where appropriate, the ship's structure in accordance with the Ship Construction File, taking into account identified areas that need special attention, and verifying that the Ship Construction File is updated, where applicable32 (SOLAS reg.II-1/3-10 and MSC.287(87)).
- (CA) 2.2.4 For the hull, machinery and equipment of cargo ships, concerning the additional requirements for chemical tankers and gas carriers, the annual survey should consist of:
- (CA) 2.2.4.1 the provisions of (CA) 2.2.3 except (CA) 2.2.3.16 and (CA) 2.2.3.18.
- (CA) 2.2.5 For the hull, machinery and equipment of cargo ships concerning the additional requirements for using natural gas as fuel other than ships covered by the IGC Code, the annual survey should consist of:
- (CA) 2.2.5.1 examining the logbooks and operating records with regard to correct functioning of the gas detection systems, fuel supply/gas systems, etc. (IGF Code ch.16);
- (CA) 2.2.5.2 confirming the manufacturer/builder instructions and manuals covering the operations, safety and maintenance requirements and occupational health hazards relevant to fuel storage, fuel bunkering, and fuel supply and associated systems for the use of the fuel, are provided on board the vessel (IGF Code chs.6 and 18);
- (CA) 2.2.5.3 confirming gas detection and other leakage detection equipment in compartments containing fuel storage, fuel bunkering, and fuel supply equipment or components or associated systems, including indicators and alarms, are in satisfactory operating condition (IGF Code chs.6 and 15);
- (CA) 2.2.5.4 confirming the satisfactory operation of the control, monitoring and automatic shutdown systems of the fuel supply and bunkering systems (IGF Code ch.15);
- (CA) 2.2.5.5 confirming the availability of test and calibration records of the gas detection systems (IGF Code ch.15);
- (CA) 2.2.5.6 examining piping, hoses, emergency shutdown valves, remote operating valves, relief valves, means for inerting, machinery and equipment for fuel storage, fuel bunkering, and fuel supply such as venting, compressing, refrigerating, liquefying, heating, cooling or otherwise handling the fuel (IGF Code chs.5, 6, 8, 9, 10 and 15);
- (CA) 2.2.5.7 testing the shutdown of ESD protected machinery spaces operationally, as far as practicable (IGF Code ch.5);
- (CA) 2.2.5.8 confirming stopping of pumps and compressors upon emergency shutdown of the system (IGF Code chs.6, 10 and 15);
- (CA) 2.2.5.9 examining the ventilation system, including portable ventilating equipment where fitted, for spaces containing fuel storage, fuel bunkering, and fuel supply units or components or associated systems,

³² Refer to annex B of the International Code on the Enhanced Programme of Inspections during Surveys of Bulk Carriers and Oil Tankers, 2011 (resolution A.1049(27)), as amended.

including air locks, pump-rooms, compressor rooms, fuel preparation rooms, fuel valve rooms, control rooms and spaces containing gas burning equipment (IGF Code chs.12 and 13);

- (CA) 2.2.5.10 testing, as far as practicable, alarms, such as differential pressure and loss of pressure alarms (IGF Code ch.15);
- (CA) 2.2.5.11 examining portable and fixed drip trays and insulation (IGF Code ch.5);
- (CA) 2.2.5.12 examining electrical equipment including electrical bonding arrangements and bulkhead/deck penetrations including access openings in hazardous areas (IGF Code chs.5, 12 and 14);
- (CA) 2.2.5.13 examining the condition and arrangement of fuel storage, bunkering and supply systems including external examination of storage tank (including secondary barrier if fitted) and relief valves if accessible, verification of satisfactory operation of tank monitoring system, examination and testing of installed bilge alarms and means of drainage (IGF Code chs.6, 8 and 15);
- (CA) 2.2.5.14 testing of the remote and local closing of the installed main tank valve (IGF Code chs.6 and 10);
- (CA) 2.2.5.15 examining bunkering stations and the fuel bunkering system including operation of the fuel bunkering control, monitoring and shutdown systems (IGF Code ch.8);
- (CA) 2.2.5.16 examining the Ship-shore link (SSL) or equivalent means for automatic and manual ESD communication to the bunkering source (IGF Code para.8.5.7);
- (CA) 2.2.5.17 examining the fuel supply system including the fuel supply system control, monitoring and shutdown systems (IGF Code chs.9 and 15);
- (CA) 2.2.5.18 testing of the remote and local closing of the master fuel valve for each engine compartment (IGF Code chs.5, 9 and 15);
- (CA) 2.2.5.19 checking the records about drills and emergency exercises (IGF Code ch.17); and
- (CA) 2.2.5.20 checking the pre-bunkering verification records according to the bunker safety checklist (IGF Code ch.18).
- (CA) 2.2.6 For the hull, machinery and equipment of cargo ships the completion of the annual survey should consist of:
- (CA) 2.2.6.1 after a satisfactory survey, the Cargo Ship Safety Construction Certificate should be endorsed; and
- (CA) 2.2.6.2 if a survey shows that the condition of a ship or its equipment is unsatisfactory, see part "General", section 4.8.
- (Cln) **2.3** Intermediate surveys see part "General", section 4.3
- (CIn) 2.3.1 For the hull, machinery and equipment of cargo ships the examination of current certificates and other records should consist of:

MSC 104/18/Add.1 Annex 31, page 76

- (CIn) 2.3.1.1 the provisions of (CA) 2.2.1.
- (CIn) 2.3.2 For the hull,³³ machinery and equipment of cargo ships the intermediate survey should consist of:
- (CIn) 2.3.2.1 the provisions of (CA) 2.2.2;
- (Cln) 2.3.2.2 for ships over 5 years of age, an internal examination of representative spaces used for water ballast;
- (CIn) 2.3.2.3 for ships over 10 years of age, other than gas carriers and ships engaged in the carriage of dry cargoes only, an internal examination of selected cargo spaces; and
- (CIn) 2.3.2.4 for ships over 15 years of age, engaged in the carriage of dry cargoes only, an internal examination of selected cargo spaces.
- (CIn) 2.3.3 For the hull,³⁴ machinery and equipment of cargo ships for the additional requirements for oil tankers the intermediate survey should consist of:
- (Cln) 2.3.3.1 the provisions of (CA) 2.2.3;
- (CIn) 2.3.3.2 should there be any doubt as to its condition when examining the various piping systems, the piping may be required to be pressure tested, gauged or both; particular attention is to be paid to repairs such as welded doublers; and
- (CIn) 2.3.3.3 testing the insulation resistance of electrical circuits in dangerous zones such as cargo pump-rooms and areas adjacent to cargo tanks, but in cases where a proper record of testing is maintained, consideration should be given to accepting recent readings.
- (CIn) 2.3.4 For the hull, machinery and equipment of cargo ships, concerning the additional requirements for chemical tankers and gas carriers, the intermediate survey should consist of:
- (Cln) 2.3.4.1 the provisions of (CA) 2.2.3 except (CA) 2.2.3.16 and (CA) 2.2.3.18.
- (CIn) 2.3.5 For the hull, machinery and equipment of cargo ships concerning the additional requirements for using natural gas as fuel other than ships covered by the IGC Code, the intermediate survey should consist of:
- (Cln) 2.3.5.1 the provisions of (CA) 2.2.5; and
- (CIn) 2.3.5.2 testing gas detectors, temperature sensors, pressure sensors, level indicators, and other arrangement of control, monitoring and safety of fuel supply systems including proper response of the fuel safety system upon fault conditions (IGF Code ch.15).

³³ Refer also to annex A to the International Code on the Enhanced Programme of Inspections during Surveys of Bulk Carriers and Oil Tankers, 2011 (resolution A.1049(27)), as amended.

³⁴ Refer also to annex B to the International Code on the Enhanced Programme of Inspections during Surveys of Bulk Carriers and Oil Tankers, 2011 (resolution A.1049(27)), as amended.

- (CIn) 2.3.6 For the hull, machinery and equipment of cargo ships the completion of the intermediate survey should consist of:
- (Cln) 2.3.6.1 after a satisfactory survey, the Cargo Ship Safety Construction Certificate should be endorsed; and
- (Cln) 2.3.6.2 if a survey shows that the condition of a ship or its equipment is unsatisfactory, see part "General", section 4.8.
- (CR) **2.4 Renewal surveys** see part "General", section 4.5
- (CR) 2.4.1 For the hull, machinery and equipment of cargo ships the examination of current certificates and other records should consist of:
- (CR) 2.4.1.1 the provisions of (CA) 2.2.1, except for the validity of the Cargo Ship Safety Construction Certificate.
- (CR) 2.4.2 For the hull,³⁵ machinery and equipment of cargo ships the renewal survey should consist of:
- (CR) 2.4.2.1 the provisions of (Cln) 2.3.2;
- (CR) 2.4.2.2 examination of sea valves and their connections to the hull; and
- (CR) 2.4.2.3 examination of anchoring and mooring equipment, for which purpose the anchors should be lowered and raised using the windlass.
- (CR) 2.4.3 For the hull,³⁶ machinery and equipment of cargo ships, concerning the additional requirements for oil tankers, the renewal survey should consist of:
 (CR) 2.4.3.1 the provisions of (CIn) 2.3.3.
- (CR) 2.4.4 For the hull, machinery and equipment of cargo ships, concerning the additional requirements for chemical tankers and gas carriers, the renewal survey should consist of:
- (CR) 2.4.4.1 the provisions of (CA) 2.2.3 except (CA) 2.2.3.16 and (CA) 2.2.3.18.
- (CR) 2.4.5 For the hull, machinery and equipment of cargo ships, concerning the additional requirements for bulk carriers the renewal survey should consist of the provisions of (CI) 2.1.4.68 and 2.1.4.70.
- (CR) 2.4.6 For the hull, machinery and equipment of cargo ships concerning the additional requirements for ships using natural gas as fuel other than ships covered by the IGC Code, the renewal survey should consist of:
- (CR) 2.4.6.1 the provisions of (CIn) 2.3.5;

³⁵ Refer also to annex A of the International Code on the Enhanced Programme of Inspections during Surveys of Bulk Carriers and Oil Tankers, 2011 (resolution A.1049(27)), as amended.

³⁶ Refer also to annex B of the International Code on the Enhanced Programme of Inspections during Surveys of Bulk Carriers and Oil Tankers, 2011 (resolution A.1049(27)), as amended.

- (CR) 2.4.6.2 examining the storage tanks and all associated piping for fuel storage, fuel bunkering, and fuel supply such as venting, compressing, refrigerating, liquefying, heating, storing, burning or otherwise handling the fuel and liquid nitrogen installations, and requiring removal of insulation from the piping and opening for examination and hydrostatic test of suspected pipeline as necessary, and leak test of complete piping after reassembly (IGF Code chs.5, 6, 7, 8, 9 and 10);
- (CR) 2.4.6.3 examining emergency shutdown valves, check valves, block and bleed valves, master gas valves, remote operating valves, isolating valves for pressure relief valves in the fuel storage, fuel bunkering, and fuel supply piping systems, with randomly selected valves being opened for examination (IGF Code chs.5, 6, 7, 9, 15 and 16);
- (CR) 2.4.6.4 examining pressure relief valves connected to fuel storage tanks and connected pipes and venting system, with PRV being opened for examination, adjusted and function tested (IGF Code ch.6);
- (CR) 2.4.6.5 examining and testing pressure relief valves in fuel supply/bunker lines, including valves being opened for internal examination and testing; the number of valves being opened up for internal examination and being tested should include all PRVs that were not internally examined and tested in the past 5 years and a random selection of PRVs that were internally examined and tested in the past 5 years provided satisfactory records of overhaul and testing of these PRVs are available (IGF Code ch.6);
- (CR) 2.4.6.6 examining pressure/vacuum relief valves or devices for interbarrier spaces and hold spaces, with the valves being opened, examined, tested and readjusted as necessary (IGF Code ch.6);
- (CR) 2.4.6.7 examining fuel storage tanks internally in accordance with an approved survey plan (IGF Code ch.6);
- (CR) 2.4.6.8 examining and testing of spill protection and water spray systems, for portable liquefied gas fuel tanks located on open deck (IGF Code para.6.5.2);
- (CR) 2.4.6.9 examining and testing the thermal oxidation system if any (IGF Code para.6.9.4);
- (CR) 2.4.6.10 examining and NDE testing the low temperature steel shielding at the bunker station if any (IGF Code para.8.3.1.6);
- (CR) 2.4.6.11 examining fuel pumps, compressors, process pressure vessels, inert gas generators, heat exchangers and other components used in connection with fuel handling (IGF Code chs.5, 6, 8, 9, 10 and 15);
- (CR) 2.4.6.12 examining electrical equipment including the physical condition of electrical cables and supports, intrinsically safe, explosion proof, or increased safety features of electrical equipment, including functional tests of pressurized electrical equipment and associated alarms, testing of electrical equipment for de-energization which is not certified for use

in hazardous areas and insulation resistance test of circuits passing through a hazardous zone (IGF Code chs.12 and 14); and

- (CR) 2.4.6.13 examining and testing gas detectors, temperature sensors, pressure sensors, level indicators, and other equipment providing input to the fuel safety system, including verification of the response upon fault conditions, and the calibrations of pressure, temperature and level indicating equipment in accordance with the manufacturer's requirements (IGF Code ch.15).³⁷
- (CR) 2.4.7 After a satisfactory survey, the Cargo Ship Safety Construction Certificate should be issued.
- (B) 3 GUIDELINES FOR THE INSPECTION OF THE OUTSIDE OF THE SHIP'S BOTTOM OF CARGO SHIPS
- (CB) 3.1 For the inspection of the outside of the ship's bottom of cargo ships the inspection should consist of:
- (CB) 3.1.1 examining the ship's shell including bottom and bow plating, keel, bilge keels, stem, stern frame and rudder;
- (CB) 3.1.2 noting the clearances measured in the rudder bearings;
- (CB) 3.1.3 examining the propeller and shaft seals, as far as practicable;
- (CB) 3.1.4 noting the clearance measured in the propeller shafts, as far as practicable;
- (CB) 3.1.5 examining sea chests and strainers; and
- (CB) 3.1.6 the survey of related items inspected at the same time (see part "General" section 5.1).
- (CB) 3.2 For the inspection of the outside of the ship's bottom of cargo ships the completion of the inspection should consist of:
- (CB) 3.2.1 after a satisfactory survey, the Cargo Ship Safety Construction Certificate should be endorsed; and
- (CB) 3.2.2 if a survey shows that the condition of a ship or its equipment is unsatisfactory, see part "General" section 4.8.
- (R) 4 GUIDELINES FOR SURVEYS FOR THE CARGO SHIP SAFETY RADIO CERTIFICATE
- (RI) **4.1** Initial surveys see part "General" section 4.1
- (RI) 4.1.1 For the radio installations, including those used in life-saving appliances, of cargo ships the examination of plans and designs should consist of:
- (RI) 4.1.1.1 establishing the sea areas declared for operation, the equipment installed to fulfil the functional requirements for the sea areas of operation, the methods adopted to ensure the availability of the functional requirements

³⁷ Refer to the *Unified interpretations of the IGF Code* (MSC.1/Circ.1591).

and the arrangements for supply of an emergency source of energy (if any) (SOLAS 74/88 regs.II-1/43 and IV/1 to 15);

- (RI) 4.1.1.2 establishing which radio equipment is to be surveyed and, if duplication of equipment is used as a means of ensuring the availability of the functional requirements, establishing which is the "basic equipment" and which the "duplicated equipment" (SOLAS 74/88 reg.IV/15) (additional radiocommunications equipment provided other than for SOLAS compliance should be noted);
- (RI) 4.1.1.3 confirming all SOLAS equipment complies with appropriate performance standards not inferior to those adopted by IMO (SOLAS 74/88 reg.IV/14);
- (RI) 4.1.1.4 examining the plans for the provision and position of the radio installation, including sources of energy and antennas (SOLAS 74/88 regs.II-1/43, IV/6, IV/14 and V/19); and
- (RI) 4.1.1.5 examining the plans for the provision and positioning of the radio life-saving appliances (SOLAS 74/88 reg.III/6).
- (RI) 4.1.2 For the radio installations, including radio life-saving appliances, of cargo ships the survey during construction and after installation should consist of:
- (RI) 4.1.2.1 examining the position, physical and electromagnetic protection and illumination of each radio installation (SOLAS 74/88 reg.IV/6);
- (RI) 4.1.2.2 confirming the provision of equipment for the radio installation with due regard to the declared sea areas in which the ship will trade and the declared means of maintaining availability of functional requirements (SOLAS 74/88 regs.III/6, IV/7 to 11, 14 and 15);
- (RI) 4.1.2.3 confirming the ability to initiate the transmission of ship-to-shore distress alerts by at least two separate and independent means, each using a different radiocommunication service, from the position from which the ship is normally navigated (SOLAS 74/88/06 regs.IV/4, 7 to 11);
 (RI) 4.1.2.4 examining all antennas, including:
- (RI) 4.1.2.4.1 visually checking all antennas, including recognized mobile satellite service antennas, and feeders for satisfactory siting and absence of defects (SOLAS 74/88/18 reg.IV/14);
- (RI) 4.1.2.4.2 checking insulation and safety of all antennas;
- (RI) 4.1.2.5 examining the reserve source of energy, including:
- (RI) 4.1.2.5.1 checking there is sufficient capacity to operate the basic or duplicated equipment for 1 hour or 6 hours, as appropriate (SOLAS 74/88 reg.IV/13);
- (RI) 4.1.2.5.2 if the reserve source of energy is a battery:
- (RI) 4.1.2.5.2.1 checking its siting and installation (SOLAS 74/88 reg.IV/13);

- (RI) 4.1.2.5.2.2 where appropriate, checking its condition by specific gravity measurement or voltage measurement;
- (RI) 4.1.2.5.2.3 with the battery off charge, and the maximum required radio installation load connected to the reserve source of energy, checking the battery voltage and discharge current;
- (RI) 4.1.2.5.2.4 checking that the charger or chargers are capable of recharging the reserve battery within 10 hours (SOLAS 74/88 reg.IV/13);
- (RI) 4.1.2.5.2.5 checking that information on ship's position is provided continuously and automatically to all two-way communication equipment (SOLAS 74/88 reg.IV/18);
- (RI) 4.1.2.6 examining the VHF transceiver(s), including:
- (RI) 4.1.2.6.1 checking for operation on channels 6, 13 and 16 (SOLAS 74/88 regs.IV/7 and 14);
- (RI) 4.1.2.6.2 checking frequency tolerance, transmission line quality and radio frequency power output (SOLAS 74/88 reg.IV/14);
- (RI) 4.1.2.6.3 checking for correct operation of all controls including priority of control units (SOLAS 74/88 reg.IV/14);
- (RI) 4.1.2.6.4 checking that the equipment operates from the main, emergency (if provided) and reserve sources of energy (SOLAS 74/88 reg.IV/13);
- (RI) 4.1.2.6.5 checking the operation of the VHF control unit(s) or portable VHF equipment provided for navigational safety (SOLAS 74/88 reg.IV/6);
- (RI) 4.1.2.6.6 checking for correct operation by on-air contact with a coast station or other ship;
- (RI) 4.1.2.7 examining the VHF DSC controller and channel 70 Digital Selective Calling (DSC) watch receiver, including:
- (RI) 4.1.2.7.1 performing an off-air check confirming the correct Maritime Mobile Service Identity is programmed in the equipment (SOLAS 74/88 reg.IV/14);
- (RI) 4.1.2.7.2 checking for correct transmission by means of a routine or test call to a coast station, other ship, onboard duplicate equipment or special test equipment;
- (RI) 4.1.2.7.3 checking for correct reception by means of a routine or test call from a coast station, other ship, onboard duplicate equipment or special test equipment;
- (RI) 4.1.2.7.4 checking the audibility of the VHF/DSC alarm;
- (RI) 4.1.2.7.5 checking that the equipment operates from the main, emergency (if provided) and reserve sources of energy (SOLAS 74/88 reg.IV/13);

(RI)	4.1.2.8	examining the MF/HF radiotelephone equipment, including:
(RI)	4.1.2.8.1	checking that the equipment operates from the main, emergency (if provided) and reserve sources of energy (SOLAS 74/88 reg.IV/13);
(RI)	4.1.2.8.2	checking the antenna tuning in all appropriate bands;
(RI)	4.1.2.8.3	checking that the equipment is within frequency tolerance on all appropriate bands (SOLAS 74/88 reg.IV/14);
(RI)	4.1.2.8.4	checking for correct operation by contact with a coast station and/or measuring transmission line quality and radio frequency output;
(RI)	4.1.2.8.5	checking receiver performance by monitoring known stations on all appropriate bands;
(RI)	4.1.2.8.6	if control units are provided outside the navigating bridge, checking that the control unit on the bridge has first priority for the purpose of initiating distress alerts (SOLAS 74/88 regs.IV/9, 10, 11 and 14);
(RI)	4.1.2.9	examining the HF radiotelex equipment, including:
(RI)	4.1.2.9.1	checking that the equipment operates from the main, emergency (if provided) and reserve sources of energy (SOLAS 74/88 reg.IV/13);
(RI)	4.1.2.9.2	confirming that the correct selective calling number is programmed in the equipment;
(RI)	4.1.2.9.3	checking correct operation by inspection of recent hard copy or by a test with a coast radio station (SOLAS 74/88 regs.IV/10 and 11);
(RI)	4.1.2.10	examining the MF/HF DSC controller(s), including:
(RI)	4.1.2.10.1	checking that the equipment operates from the main, emergency (if provided) and reserve sources of energy (SOLAS 74/88 reg.IV/13);
(RI)	4.1.2.10.2	confirming that the correct Maritime Mobile Service Identity is programmed in the equipment;
(RI)	4.1.2.10.3	checking the off-air self-test programme;
(RI)	4.1.2.10.4	checking operation by means of a test call on MF and/or HF to a coast radio station if the rules of the berth permit the use of MF/HF transmissions (SOLAS 74/88 regs.IV/9, 10 and 11);
(RI)	4.1.2.10.5	checking the audibility of the MF/HF DSC alarm;
(RI)	4.1.2.11	examining the MF/HF DSC watch receiver(s), including:
(RI)	4.1.2.11.1	confirming that only distress and safety DSC frequencies are being monitored (SOLAS 74/88 regs.IV/9 to 12);
(RI)	4.1.2.11.2	checking that a continuous watch is being maintained while keying MF/HF radio transmitters (SOLAS 74/88 reg.IV/12);
(RI)	4.1.2.11.3	checking for correct operation by means of a test call from a coast station or other ship;
- (RI) 4.1.2.12 examining the recognized mobile satellite service ship earth station(s), including:
- (RI) 4.1.2.12.1 checking that the equipment operates from the main, emergency (if provided) and reserve sources of energy, and that where an uninterrupted supply of information from the ship's navigational or other equipment is required ensuring such information remains available in the event of failure of the ship's main or emergency source of electrical power (SOLAS 74/88 regs.IV/13 and 14);
- (RI) 4.1.2.12.2 checking the distress function by means of an approved test procedure where possible (SOLAS 74/88 regs.IV/10, 12 and 14);
- (RI) 4.1.2.12.3 checking for correct operation by inspection of recent hard copy or by test call;
- (RI) 4.1.2.13 if appropriate, examining the NAVTEX equipment (SOLAS 74/88 regs.IV/7, 12 and 14), including:
- (RI) 4.1.2.13.1 checking for correct operation by monitoring incoming messages or inspecting recent hard copy;
- (RI) 4.1.2.13.2 running the self-test programme if provided;
- (RI) 4.1.2.14 examining the Enhanced Group Call equipment (SOLAS 74/88 regs.IV/7 and 14), including:
- (RI) 4.1.2.14.1 checking for correct operation and area by monitoring incoming messages or by inspecting recent hard copy;
- (RI) 4.1.2.14.2 running the self-test programme if provided;
- (RI) 4.1.2.15 if appropriate, examining the radio equipment for receipt of maritime safety information by HF Narrow Band Direct Printing (NBDP) (SOLAS 74/88 regs.IV/7, 12 and 14), including:
- (RI) 4.1.2.15.1 checking for correct operation by monitoring incoming messages or inspecting recent hard copy;
- (RI) 4.1.2.15.2 running the self-test programme if provided;
- (RI) 4.1.2.16 examining the 406 MHz satellite Emergency Position Indicating Radio Beacon (EPIRB) (SOLAS 74/88 regs.IV/7 and 14), including:
- (RI) 4.1.2.16.1 checking position and mounting for float free operation;
- (RI) 4.1.2.16.2 carrying out visual inspection for defects;
- (RI) 4.1.2.16.3 carrying out the self-test routine;
- (RI) 4.1.2.16.4 checking that the unique beacon identification code is clearly marked on the outside of the equipment and, where possible, decoding the unique beacon identification code confirming it is correct;

(RI)	4.1.2.16.5	checking that the unique beacon identification code programmed in the EPIRB corresponds with the unique beacon identification code assigned by or on behalf of the Administration;
(RI)	4.1.2.16.6	checking that the MMSI number if encoded in the beacon corresponds with the MMSI number assigned to the ship;
(RI)	4.1.2.16.7	checking the battery expiry date;
(RI)	4.1.2.16.8	if provided, checking the hydrostatic release and its expiry date;
(RI)	4.1.2.16.9	checking the emission on operational frequencies, coding and registration on the 406 MHz signal without transmission of a distress call to the satellite;
(RI)	4.1.2.16.10	checking that the EPIRB has been subject to maintenance at intervals not exceeding five years at an approved shore-based maintenance facility (SOLAS 74/04 reg.IV/15.9);
(RI)	4.1.2.16.11	if possible, checking the emission on operational frequencies, coding and registration on the 121.5 MHz homing signal without transmission of a distress call to the satellite;
(RI)	4.1.2.17	examining the two-way VHF radiotelephone apparatus ³⁸ (SOLAS 74/88 reg.III/6), including:
(RI)	4.1.2.17.1	checking for correct operation on Channel 16 and one other by testing with another fixed or portable VHF installation (SOLAS 74/88 reg.IV/14);
(RI)	4.1.2.17.2	checking the battery charging arrangements where re-chargeable batteries are used;
(RI)	4.1.2.17.3	checking the expiry date of primary batteries where used;
(RI)	4.1.2.17.4	where appropriate, checking any fixed installation provided in a survival craft;
(RI)	4.1.2.18	examining the search and rescue locating device(s) (SOLAS 74/88/08 regs.III/6, IV/7 and 14), including:
(RI)	4.1.2.18.1	checking the position and mounting;
(RI)	4.1.2.18.2	monitoring response on ship's 9 GHz radar;
(RI)	4.1.2.18.3	checking the battery expiry date; and
(RI)	4.1.2.19	examining the test equipment and spares carried to ensure carriage is adequate in accordance with the sea areas in which the ship trades and the declared options for maintaining availability of the functional requirements (SOLAS 74/88 reg.IV/15).

³⁸ Expired primary batteries, which complied with the requirements of paragraph 12.6 of *Revised Recommendation on performance standards for survival craft portable Two-Way VHF radiotelephone apparatus* (resolution MSC.149 (77)), could be used for no other purpose than to examine and check the operation of two-way VHF radiotelephone apparatus.

- (RI) 4.1.3 For the radio installations, including those used in life-saving appliances, the check that documentation, etc., has been placed on board should consist of:
- (RI) 4.1.3.1 checking for a valid radio licence issued by the flag Administration (ITU RR Article 24);
- (RI) 4.1.3.2 checking the radio operator's certificates of competence (SOLAS 74/88 reg.IV/16 and ITU RR Article 56);
- (RI) 4.1.3.3 checking the radio record (log) (SOLAS 74/88 reg.IV/17 and ITU RR App.11);
- (RI) 4.1.3.4 checking the carriage of up-to-date ITU publications (ITU RR App.11);
- (RI) 4.1.3.5 checking the carriage of operating manuals for all equipment (SOLAS 74/88 reg.IV/15); and
- (RI) 4.1.3.6 checking the carriage of service manuals for all equipment when at-sea maintenance is the declared option (SOLAS 74/88 reg.IV/15).
- (RI) 4.1.4 For the radio installations, including those used in life-saving appliances, of cargo ships the completion of the initial survey should consist of:
- (RI) 4.1.4.1 the surveyor preparing and forwarding a survey report, indicating clearly the organization he or she represents, to the relevant authorities, detailing results of the survey and recording omissions and deficiencies; if satisfied, the relevant authorities should issue a Cargo Ship Safety Radio Certificate and the associated Record of Equipment (form R).
- (RP) **4.2 Periodical surveys** see part "General" section 4.4
- (RP) 4.2.1 For radio installations, including radio life-saving appliances, on cargo ships the examination of current certificates and other records should consist of:
- (RP) 4.2.1.1 checking the validity, as appropriate, of the Cargo Ship Safety Equipment Certificate, the Cargo Ship Safety Radio Certificate and the Cargo Ship Safety Construction Certificate or the Cargo Ship Safety Certificate;
- (RP) 4.2.1.2 checking, as appropriate, the validity of the Polar Ship Certificate;
- (RP) 4.2.1.3 checking that the ship's complement complies with the Minimum Safe Manning Document (SOLAS 74/00/12 reg.V/14) (SOLAS 74/88 reg.V/13(b));
- (RP) 4.2.1.4 checking the validity of the International Ship Security Certificate;
- (RP) 4.2.1.5 checking the validity of the International Load Line Certificate or International Load Line Exemption Certificate;
- (RP) 4.2.1.6 checking the validity of the International Oil Pollution Prevention Certificate;

(RP)	4.2.1.7	checking the certificates of class, if the ship is classed with a classification society;
(RP)	4.2.1.8	checking, when appropriate, the validity of the International Certificate of Fitness for the Carriage of Dangerous Chemicals in Bulk or the Certificate of Fitness for the Carriage of Dangerous Chemicals in Bulk;
(RP)	4.2.1.9	checking, when appropriate, the validity of the International Certificate of Fitness for the Carriage of Liquefied Gases in Bulk;
(RP)	4.2.1.10	checking, when appropriate, the validity of the International Pollution Prevention Certificate for the Carriage of Noxious Liquid Substances in Bulk;
(RP)	4.2.1.11	checking, when appropriate, the validity of the International Sewage Pollution Prevention Certificate;
(RP)	4.2.1.12	checking, when appropriate, the validity of the International Air Pollution Prevention Certificate;
(RP)	4.2.1.13	confirming, when appropriate, the validity of the International Energy Efficiency Certificate (MARPOL Annex VI, regs.6.4 and 6.5);
(RP)	4.2.1.14	confirming, when appropriate, that confirmation of compliance for the SEEMP part II is provided to and retained on board the ship (MARPOL Annex VI, reg. 5.4.5); ³⁹
(RP)	4.2.1.15	confirming, when appropriate, the validity of the Statements of Compliance related to fuel oil consumption reporting (MARPOL Annex VI, regs.6.6 and 6.7);
(RP)	4.2.1.16	checking, when appropriate, the validity of the International Ballast Water Management Certificate;
(RP)	4.2.1.17	checking that the ship's complement complies with the Minimum Safe Manning Document (SOLAS 74/88 reg.V/13(b));
(RP)	4.2.1.18	checking that adequate information is on board to enable the equipment to be properly operated and maintained;
(RP)	4.2.1.19	checking that the master, officers and ratings are certificated as required by the STCW Convention;
(RP)	4.2.1.20	confirming that any new equipment has been properly approved before installation and that no changes have been made such as would affect the validity of the certificate;
(RP)	4.2.1.21	confirming that a record has been kept in the period since the last survey to the satisfaction of the Administration and as required by the Radio Regulations (SOLAS 74/88 reg.IV/17);

³⁹ Refer to the Sample Format of Confirmation of Compliance, Early Submission of the SEEMP part II on the ship fuel oil consumption data collection plan and its timely verification pursuant to regulation 5.4.5 of MARPOL Annex VI (MEPC.1/Circ.876).

- (RP) 4.2.1.22 checking documentary evidence that the actual capacity of the battery has been proved in port within the last 12 months (SOLAS 74/88 reg.IV/13);
- (RP) 4.2.1.23 confirming that the provisions of (RI) 4.1.3 have been met;
- (RP) 4.2.1.24 checking that the annual test has been carried out for the Satellite EPIRB and, if applicable, shore-based maintenance has been carried out at intervals not exceeding five years (SOLAS 74/04 reg.IV/15); and
- (RP) 4.2.1.25 confirming the availability of the International Anti-fouling System Certificate (AFS 2001 annex 4 reg.2) when applicable.
- (RP) 4.2.2 For radio installations, including radio life-saving appliances, of cargo ships the periodical survey should consist of:
- (RP) 4.2.2.1 the provisions of (RI) 4.1.2.
- (RP) 4.2.3 For radio installations, including those used in radio life-saving appliances, of cargo ships the completion of the periodical survey should consist of:
- (RP) 4.2.3.1 after a satisfactory survey, endorsing the Cargo Ship Safety Radio Certificate; and
- RP) 4.2.3.2 if a survey shows that the condition of a ship or its equipment is unsatisfactory, see part "General" section 4.8.
- (RR) **4.3 Renewal surveys** see part "General" section 4.5
- (RR) 4.3.1 For the radio installations, including those used in life-saving appliances, of cargo ships the examination of current certificates and other records should consist of:
- (RR) 4.3.1.1 the provisions of (RP) 4.2.1, except for the validity of the Cargo Ship Safety Radio Certificate.
- (RR) 4.3.2 For the radio installations, including those used in radio life-saving appliances, of cargo ships the renewal survey should consist of:
- (RR) 4.3.2.1 the provisions of (RI) 4.1.2.
- (RR) 4.3.3 For the radio installations, including those used in radio life-saving appliances, of cargo ships the completion of the renewal survey should consist of:
- (RR) 4.3.3.1 after a satisfactory survey, issuing the Cargo Ship Safety Radio Certificate as per the provisions of (RI) 4.1.4.
- (P) 5 GUIDELINES FOR SURVEYS FOR THE PASSENGER SHIP CERTIFICATE
- (PI) **5.1** Initial surveys see part "General" section 4.1.

(PI)	5.1.1	For the hull, machinery and equipment of passenger ships the examination of plans and designs should consist of:
(PI)	5.1.1.1	examining the subdivision and stability (SOLAS 74/88/95 regs.II-1/4 to 8, 8-1, 8-2, 8-3, 13 and 16) (SOLAS 74/06/08/17 regs.II-1/5 to 8-1, 14 and 18; IS Code chs.1, 2 and 3) (SOLAS 74/12/17/18 regs.II-1/8-1);
(PI)	5.1.1.2	examining the ballasting arrangements (SOLAS 74/88 reg.II-1/9) (SOLAS 74/06/17 reg.II-1/20);
(PI)	5.1.1.3	examining the arrangement of the bulkheads, their construction and the openings therein, including the disposition and means of operation of the watertight doors (SOLAS 74/88 regs.II-1/10, 14, and 15) (SOLAS 74/06/17 regs.II-1/10, 11 12 and 13);
(PI)	5.1.1.4	examining the arrangement of the double bottoms (SOLAS 74/88 reg.II-1/12) (SOLAS 74/06/17 reg.II-1/9);
(PI)	5.1.1.5	examining the arrangements for the openings in the shell plating below the margin line or the bulkhead deck as applicable, the construction of the watertight doors, side scuttles, watertight decks, trunks, etc., and the watertight integrity above the margin line or the bulkhead deck as applicable (SOLAS 74/88 regs.II-1/17, 18, 19 and 20) (SOLAS 74/06/17 regs.II-1/15, 16, 16-1 and 17);
(PI)	5.1.1.6	examining the plans for the bilge pumping and drainage systems (SOLAS 74/88 regs.II-1/21 and 39) (SOLAS 74/05/09/17 reg.II-1/35-1 and SOLAS 74/08 reg.II-2/20.6.1.4);
(PI)	5.1.1.7	examining, when appropriate, the means of indicating the status of any bow doors and the leakage therefrom (SOLAS 74/88 reg.II-1/23-2) (SOLAS 74/06 reg.II-1/17-1);
(PI)	5.1.1.8	examining the plans for the machinery installation (SOLAS 74/88 regs.II-1/26 to 36 and 54);
(PI)	5.1.1.9	examining the plans for the electrical installation (SOLAS 74/88 regs.II-1/39, 40, 41, 42, 44 and 45);
(PI)	5.1.1.10	checking, when appropriate, the provision of supplementary emergency lighting (SOLAS 74/88 reg.II-1/42-1);
(PI)	5.1.1.11	examining, where applicable, the approved documentation for the alternative design and arrangements (SOLAS 74/00/06/15 regs.II-1/55, II-2/17 and III/38 and IGF Code ch.2);
(PI)	5.1.1.12	examining the plans for the fire pumps, including the emergency fire pump ⁴⁰ if applicable, fire mains, hydrants, hoses and nozzles and the international shore connection (SOLAS 74/88 reg.II-1/39 and SOLAS 74/00/14 reg.II-2/10.2; FSS Code chs.2 and 12) (SOLAS 74/88 reg.II-1/39 and regs.II-2/4 and 19);

⁴⁰ Refer to the Unified interpretation of chapter 12 of the International Code for Fire Safety Systems (MSC.1/Circ.1388).

- (PI) 5.1.1.13 examining the plans for the fire-extinguishing arrangements in the machinery spaces (SOLAS 74/00/12/14/16 regs.II-2/10.4 and 10.5; FSS Code chs.5, 6 and 7) (SOLAS 74/88 reg.II-2/7);
- (PI) 5.1.1.14 checking the provision and specification of the fire extinguishers and the fire-fighters' outfits including their self-contained compressed air breathing apparatus, and provision of two-way portable radiotelephone apparatus of an explosion-proof type or intrinsically safe (SOLAS 74/88 regs.II-2/6 and 17) (SOLAS 74/00/12 reg.II-2/10.10);
- (PI) 5.1.1.15 for passenger ships constructed on or after 1 July 2010, checking the provision of a suitably located means for fully recharging breathing air cylinders (SOLAS 74/08 reg.II-2/10.10.2);
- (PI) 5.1.1.16 examining the plans for the fire-extinguishing and special arrangements in the machinery spaces (SOLAS 74/88 regs.II-1/39 and regs.II-2/7 and 11);
- (PI) 5.1.1.17 examining the arrangements for oil fuel, lubricating oil and other flammable oils (SOLAS 74/00 reg.II-2/4.2.3) (SOLAS 74/88 reg.II-2/15);
- (PI) 5.1.1.18 examining the plans for the structural fire protection, including the means of escape (SOLAS 74/00/12/15/16/17 regs.II-2/4.4.4, 5.2, 5.3, 7.5, 7.8.2, 8.4, 8.5, 9, 10.6, 11, 13, 17, 20 and 20-1; FSS Code ch.13, sections 1 and 2) (SOLAS 74/88 regs.II-2/23 to 36);
- (PI) 5.1.1.19 examining the plans for the protection of special category spaces and other cargo spaces (SOLAS 74/88 regs.II-2/37, 38 and 39) (SOLAS 74/00/06/10/15/17 regs.II-2/ 7.6, 9, 10.7.1, 10.7.2 and 20; FSS Code chs.9 and 10);
- (PI) 5.1.1.20 examining the plans for the fire protection arrangements for passenger ships designed to carry containers on or above the weather deck, as applicable, including the water mist lance (SOLAS 74/00/14 reg.II 2/10.7.3);
- (PI) 5.1.1.21 examining the plans for the fixed fire detection and alarm system, and any automatic sprinkler, fire detection and fire alarm system, as applicable, in machinery spaces, including enclosed spaces containing incinerators, accommodation and service spaces and control spaces (SOLAS 74/00/06/10 reg.II-2/7 (except 7.5.5, 7.6 and 7.9); FSS Code chs.8, 9 and 10) (SOLAS 74/88 reg.II-2/40);
- (PI) 5.1.1.22 examining the plans for the crew alarm and the public address system or other effective means of communication (SOLAS 74/00/06 reg.II-2/7.9; FSS Code ch.9; LSA Code ch.7) (SOLAS 74/88 reg.II-2/40);
- (PI) 5.1.1.23 examining the plans for the helicopter facilities, including foam firefighting appliances when appropriate (SOLAS 74/00/16 reg.II-2/18; FSS Code ch. 17) (SOLAS 74/88 reg.II-2/18.8);
- (PI) 5.1.1.24 examining the plans for the special arrangements for the carriage of dangerous goods, when appropriate, including water supplies, electrical equipment and wiring, fire detection sample extraction smoke detection

system, bilge pumping and personnel protection (SOLAS 74/88 regs.II-2/41 and 54) (SOLAS 74/00/08 reg.II-2/19; FSS Code chs.9 and 10);

- (PI) 5.1.1.25 examining the provision and disposition of the survival craft and rescue boats and the arrangements for mustering passengers (SOLAS 74/00 regs.III/11 to 17, 21 and 24);
- (PI) 5.1.1.26 examining the design of the survival craft, including their construction, equipment, fittings, release mechanisms and recovery appliances and embarkation and launching arrangements (SOLAS 74/88 regs.III/ 20 to 24, 36, 38 to 44 and 48) (SOLAS 74/06 reg.III/4) (LSA Code sections 3.2, 4.1 to 4.6, 6.1 to 6.2);
- (PI) 5.1.1.27 examining the design of the rescue boats, including their equipment and launching and recovery appliances and arrangements (SOLAS 74/88 regs.III/16, 20, 47 and 48);
- (PI) 5.1.1.28 examining the provision, specification and stowage of two-way VHF radiotelephone apparatus and search and rescue locating devices (SOLAS 74/88/08 reg.III/6.2.2);
- (PI) 5.1.1.29 examining the provision, specification and stowage of the distress flares and the line-throwing appliance and the provision of onboard communications equipment and the general alarm system (SOLAS 74/88 regs.III/6, 17, 35, 49 and 50);
- (PI) 5.1.1.30 examining the provision, specification and stowage of the lifebuoys, including those fitted with self-igniting lights, self-activating smoke signals and buoyant lines, lifejackets, immersion suits and thermal protective aids (SOLAS 74/88/06 regs.III/7, 21, 22 and 26);
- (PI) 5.1.1.31 examining the plans for the lighting of the muster and embarkation stations and the alleyways, stairways and exits giving access to the muster and embarkation stations, including the supply from the emergency source of power (SOLAS 74/88 regs.II-1/42 and III/11);
- (PI) 5.1.1.32 examining the plans for the positioning of, and the specification for, the navigation lights, shapes and sound signalling equipment (COLREG 1972, rules 20 to 24, 27 to 30 and 33);
- (PI) 5.1.1.33 examining the plans relating to the bridge design and arrangement of navigational systems and equipment and bridge procedures (SOLAS 74/00 reg.V/15);
- (PI) 5.1.1.34 checking the provision and specification of the following navigation equipment as appropriate: daylight signalling lamp, magnetic compass, transmitting heading device, gyro compass, gyro compass repeaters, radar installation(s), automatic identification system, electronic plotting aid, automatic tracking aid(s) or automatic radar plotting aid(s), echo-sounding device, speed and distance indicator, rudder angle indicator, propeller rate of revolution indicator, variable pitch propeller pitch and operational mode indicator, rate-of-turn indicator, heading or track control system, GNSS receiver, terrestrial radio navigation system and sound reception system, a pelorus or compass bearing device, means for correcting heading and bearings, a BNWAS as applicable and

ECDIS including backup arrangements as applicable (SOLAS 74/00/09/13 reg.V/19);

- (PI) 5.1.1.35 checking the provision and specification of the voyage data recorder (SOLAS 74/00 reg.V/20);
- (PI) 5.1.1.36 checking navigation bridge visibility (SOLAS 74/00 reg.V/22);
- (PI) 5.1.1.37 checking for the provision and specification of the long-range identification and tracking system (SOLAS 74/04 reg.V/19-1);
- (PI) 5.1.1.38 checking the plans and specification of the pilot transfer arrangement, the pilot ladders, the combination arrangements, where applicable, the access to the ship's deck and the associated equipment and lighting and pilot transfer arrangements (SOLAS 74/00/10 reg.V/23);
- (PI) 5.1.1.39 establishing the sea areas declared for operation, the equipment installed to fulfil the functional requirements for the sea areas of operation, the methods adopted to ensure the availability of the functional requirements and the arrangements for supply of an emergency source of energy (if any) (SOLAS 74/88 regs.II-1/42 and IV/1 to 15);
- (PI) 5.1.1.40 establishing which radio equipment is to be surveyed and, if duplication of equipment is used as a means of ensuring the availability of the functional requirements, establishing which is the "basic equipment" and which the "duplicated equipment" (SOLAS 74/88 reg.IV/15) (additional radiocommunication equipment provided other than for SOLAS compliance should be noted);
- (PI) 5.1.1.41 confirming that all SOLAS equipment complies with appropriate performance standards not inferior to those adopted by IMO (SOLAS 74/88 reg.IV/14);
- (PI) 5.1.1.42 examining the plans for the provision and positioning of the radio installation including sources of energy and antennas (SOLAS 74/88 regs.II-1/42, IV/6 and 14);
- (PI) 5.1.1.43 examining the plans for the provision and positioning of the radio life-saving appliances (SOLAS 74/88 reg.III/6);
- (PI) 5.1.1.44 if applicable, checking that a list of all limitations on the operation of a passenger ship is kept on board and updated;
- (PI) 5.1.1.45 checking the provision of means of embarkation and disembarkation from ships for use in port and in port-related operations, such as gangways and accommodation ladders (SOLAS 74/08 reg.II-1/3-9);
- (PI) 5.1.1.46 checking the provision of means to prevent blockage of drainage arrangements, for closed vehicle and ro-ro spaces and special category spaces where fixed pressure water-spraying systems are used (SOLAS 74/08 reg.II-2/20.6.1.5);

(PI)	5.1.1.47	for passenger ships constructed on or after 1 July 2010, checking the provision of a safety centre (SOLAS 74 reg.II-2/23) and associated ventilation requirements (SOLAS 74/06 reg.II-2/8.2);
(PI)	5.1.1.48	for passenger ships constructed on or after 1 July 2010 and having a length of 120 m or more or having three or more main vertical zones, confirming that design criteria for the ship's safe return to port and for systems to remain operational after a fire casualty have been documented and that safe areas have been designated (SOLAS 74/06 regs.II-2/21 and 22); and
(PI)	5.1.1.49	where applicable, examining the Cargo Securing Manual for ships carrying cargoes other than solid and liquid bulk cargoes, cargo units and cargo transport units (SOLAS 74/98/02 reg.VI/5.6).
(PI)	5.1.2	For the hull, machinery and equipment of passenger ships, concerning the examination of plans and designs additional requirements for the passenger ships using natural gas as fuel other than ships covered by the IGC Code should consist of:
(PI)	5.1.2.1	examining the plans for the fuel containment systems, control of vapour space of the liquefied gas fuel tanks, vapour detection, gauging, loading limits for the liquefied gas fuel tanks and other special requirements (IGF Code chs.5, 6, 7, 8 and 15));
(PI)	5.1.2.2	examining the plans for the ship arrangements (IGF Code ch.5);
(PI)	5.1.2.3	examining the plans for piping systems (IGF Code chs.5, 6, 7 and 9);
(PI)	5.1.2.4	examining the plans for the pressure control (IGF Code ch.6);
(PI)	5.1.2.5	examining the plans for the environmental control (IGF Code ch.6);
(PI)	5.1.2.6	examining the plans for machinery installation (IGF Code ch.10);
(PI)	5.1.2.7	examining the plans for fire protection and fire extinction equipment (IGF Code ch.11);
(PI)	5.1.2.8	examining the plans, for the fire detection and alarm system and fire-fighting arrangements (IGF Code paras.11.4, 11.5, 11.6 and 11.7);
(PI)	5.1.2.9	examining the plans for the ventilation systems (IGF Code chs.12 and 13);
(PI)	5.1.2.10	examining the plans for the electrical installations (IGF Code chs.12 and 14); and
(PI)	5.1.2.11	examining the plans for the control, monitoring and safety systems (IGF Code ch.15).
(PI)	5.1.3	For the hull, machinery and equipment of passenger ships the survey during construction and after installation should consist of:
(PI)	5.1.3.1	examining the outside of the ship's bottom, including the bottom and bow plating, keel, bilge keels, stem, stern frame, the rudder, sea chests and strainers (SOLAS 74/88 reg.l/7(b)(i));

- (PI) 5.1.3.2 confirming the arrangements on which the calculations for subdivision and stability are based, and checking the subdivision load lines (SOLAS 74/88 /95 regs.II-1/4 to 8, 13 and 16) (SOLAS 74/06/08/17 regs.II-1 /6, 7, 7-1, 7-2, 7-3, 8, 9,14, 18) (SOLAS 74/12/17/18 reg.II-1/8-1);
- (PI) 5.1.3.3 confirming the provision of operational information to the master for safe return to port after a flooding casualty by onboard stability computer or shore-based support (SOLAS 74/12/17/18 reg.II-1/8-1);
- (PI) 5.1.3.4 checking the ballasting arrangements (SOLAS 74/88 reg.II-1/9) (SOLAS 74/06/17 reg.II-1/20);
- (PI) 5.1.3.5 confirming that dedicated seawater ballast tanks have an approved coating system when appropriate (SOLAS 74/00/06 reg.II-1/3-2);
- (PI) confirming the arrangement of the bulkheads, their construction and 5.1.3.6 the openings therein, confirming that the collision bulkhead is watertight up to the freeboard deck, that the valves fitted on the pipes piercing the collision bulkhead are operable from above the freeboard deck and that there are no doors, manholes, ventilation ducts or any other openings, confirming that the other bulkheads, as required for the ship's subdivision, are watertight up to the bulkhead deck and confirming the construction of the watertight doors and that they have regs.II-1/10, been tested (SOLAS 74/88 14. 15 and 18) (SOLAS 74/06/17 regs.II-1/10, 11, 12, 13 and 16);
- (PI) 5.1.3.7 confirming that the watertight integrity has been maintained where pipes, scuppers, etc., pass through subdivision watertight bulkheads (SOLAS 74/88 reg.II-1/15) (SOLAS 74/06/17 reg.II-1/13);
- (PI) 5.1.3.8 confirming that a diagram is provided on the navigating bridge showing the location of the watertight doors together with indicators showing whether the doors are open or closed and confirming that the watertight doors and their means of operation have been installed in accordance with the approved plans (SOLAS 74/88 reg.II-1/15) (SOLAS 74/06/17 reg.II-1/13);
- (PI) 5.1.3.9 testing the operation of the watertight doors both from the navigating bridge in the event of an emergency and locally at the door itself (SOLAS 74/88 reg.II-1/15) (SOLAS 74/06/17 reg.II-1/13) and, in particular, that they are:
- (PI) 5.1.3.9.1 operable locally from each side of the bulkhead;
- (PI) 5.1.3.9.2 provided with devices giving an indication of whether the door is open or closed at all remote operating positions;
- (PI) 5.1.3.9.3 provided with an audible alarm that is distinct from any other alarm in the area and, when appropriate, an intermittent visual signal;
- (PI) 5.1.3.9.4 provided with control handles on each side of the bulkhead so that a person may hold both handles in the open position and pass safely through the watertight door without accidentally setting the power closing mechanism into operation;

(PI)	5.1.3.10	testing the remote hand-operation to close the power-operated sliding watertight door from an accessible position above the bulkhead deck (SOLAS 74/88 reg.II-1/15) (SOLAS 74/06/17 reg.II-1/13);
(PI)	5.1.3.11	confirming that the watertight doors and their indicating devices are operable in the event of a failure of the main and emergency sources of power (SOLAS 74/88 reg.II-1/15) (SOLAS 74/06/17 reg.II-1/13);
(PI)	5.1.3.12	checking, when appropriate, any watertight doors, that are not required to be closed remotely and are fitted in watertight bulkheads dividing 'tween deck spaces, and confirming that a notice is affixed concerning their closure (SOLAS 74/88 reg.II-1/15) (SOLAS 74/06/17 reg.II-1/13);
(PI)	5.1.3.13	confirming that a notice is affixed to any portable plates on bulkheads in machinery spaces concerning their closure and, if appropriate, testing any power-operated watertight door fitted in lieu (SOLAS 74/88 reg.II-1/15) (SOLAS 74/06/17 reg.II-1/13);
(PI)	5.1.3.14	confirming the arrangements for closing side scuttles and their deadlights, also scuppers, sanitary discharges and similar openings and other inlets and discharges in the shell plating below the bulkhead deck (SOLAS 74/06/17 reg.II-1/15);
(PI)	5.1.3.15	confirming that valves for closing the main and auxiliary sea inlets and discharges in the machinery spaces are readily accessible and indicators showing the status of the valves are provided (SOLAS 74/06/17 reg.II-1/15);
(PI)	5.1.3.16	confirming that gangway, cargo and fuelling ports fitted below the bulkhead deck can be effectively closed and that the inboard end of any ash or rubbish chutes are fitted with an effective cover (SOLAS 74/06/17 reg.II-1/13);
(PI)	5.1.3.17	confirming by a hose or flooding test the watertightness of watertight decks and trunks, tunnels and ventilators (SOLAS 74/88 reg.II-1/19) (SOLAS 74/06/17 reg.II-1/16-1);
(PI)	5.1.3.18	confirming the arrangements to maintain the watertight integrity above the bulkhead deck (SOLAS 74/06/17 regs.II-1/17 and 17-1);
(PI)	5.1.3.19	confirming the arrangements for the bilge pumping and that each bilge pump and pumping system provided for each watertight compartment is working efficiently (SOLAS 74/88 reg.II-1/21) (SOLAS 74/05/17 reg.II-1/35-1);
(PI)	5.1.3.20	confirming that the drainage system of enclosed cargo spaces situated on the freeboard deck is working efficiently (SOLAS 74/88 reg.II-1/21) (SOLAS 74/05/17 reg.II-1/35-1);
(PI)	5.1.3.20.1	examining visually the drainage facilities for blockage or other damage and confirming the provision of means to prevent blockage of drainage arrangements, for closed vehicle and ro-ro spaces and special category spaces where fixed pressure water-spraying systems are used (SOLAS 74/08 reg.II-2/20.6.1.5);

(PI) 5.1.3.21 conducting an inclining test (SOLAS 74/88 reg.II-1/22) (SOLAS 74/06/17 reg.II-1/5);

- (PI) 5.1.3.22 checking, when appropriate, the means of indicating the status of any bow doors and any leakage therefrom (SOLAS 74/88 reg.II-1/23-2) (SOLAS 74/06 reg.II-1/17-1);
- (PI) 5.1.3.23 confirming that the arrangement for monitoring special category spaces or ro-ro spaces, when fitted, is satisfactory (SOLAS 74/06/17 reg.II-1/23);
- (PI) 5.1.3.24 confirming that the machinery, boilers and other pressure vessels, associated piping systems and fittings are installed and protected so as to reduce to a minimum any danger to persons on board, due regard being given to moving parts, hot surfaces and other hazards (SOLAS 74/88 reg.II-1/26);
- (PI) 5.1.3.25 confirming that the normal operation of the propulsion machinery can be sustained or restored even though one of the essential auxiliaries becomes inoperative (SOLAS 74/88 reg.II-1/26);
- (PI) 5.1.3.26 confirming that means are provided so that the machinery can be brought into operation from the dead ship condition without external aid (SOLAS 74/88 reg.II-1/26);
- (PI) 5.1.3.27 confirming that the boilers, all parts of the machinery, all steam, hydraulic, pneumatic and other systems and their associated fittings which are under internal pressure have been subjected to the appropriate tests, including a pressure test (SOLAS 74/88 reg.II-1/26);
- (PI) 5.1.3.28 confirming that means are provided to ensure that the safe speed is not exceeded where there is the risk of machinery overspeeding (SOLAS 74/88 reg.II-1/27);
- (PI) 5.1.3.29 confirming that, where practicable, means are provided to protect against overpressure in the parts of main, auxiliary and other machinery that are subject to internal pressure and may be subject to dangerous overpressure (SOLAS 74/88 reg.II-1/27);
- (PI) 5.1.3.30 confirming that, when required, crankcase explosion relief devices are fitted to internal combustion engines and that they are arranged so as to minimize the possibility of injury to personnel (SOLAS 74/88 reg.II-1/27);
- (PI) 5.1.3.31 confirming that main turbine propulsion machinery and, where applicable, main internal combustion propulsion machinery and auxiliary machinery are provided with automatic shut-off arrangements in the case of failures, such as lubricating oil supply failure, which could rapidly lead to a complete breakdown, serious damage or explosion (SOLAS 74/88 reg.II-1/27);
- (PI) 5.1.3.32 confirming and recording the ability of the machinery to reverse the direction of the thrust of the propeller in sufficient time and to bring the ship to rest within a reasonable distance, including the effectiveness of any supplementary means of manoeuvring or stopping the ship⁴¹ (SOLAS 74/88 reg.II-1/28);

⁴¹ For ships fitted with alternative propulsion and steering arrangements other than traditional arrangements, such as but not limited to azimuthing propulsors or water jet propulsion systems, refer to the *Unified interpretations of SOLAS regulations II-1/28, II-29 and II-30* (MSC.1/Circ.1416/Rev.1).

- (PI) 5.1.3.33 confirming that the main and auxiliary steering gear are so arranged that the failure of one of them does not render the other inoperative⁴² (SOLAS 74/88/14 reg.II-1/29);
- (PI) 5.1.3.34 confirming that, where appropriate, essential components of the steering gear are permanently lubricated or provided with lubrication fittings (SOLAS 74/88/14 reg.II-1/29);
- (PI) 5.1.3.35 confirming that relief valves are fitted to any part of a steering gear hydraulic system which can be isolated and in which pressure can be generated from the power source or from external forces and that these relief valves are set to a pressure not exceeding the design pressure (SOLAS 74/88/14 reg.II-1/29);
- (PI) 5.1.3.36 confirming that the main steering gear is capable of steering the ship at maximum ahead service speed and is capable of putting the rudder over from 35° on one side to 35° on the other side with the ship at its deepest seagoing draught and running ahead at maximum ahead service speed and, under the same conditions, from 35° on either side to 30° on the other side in not more than 28 s,⁴³ or, where demonstration at the deepest seagoing draught is impracticable, with an alternative permissible sea trial loading condition⁴⁴ (SOLAS 74/88/14 reg.II-1/29);
- (PI) 5.1.3.37 confirming that the auxiliary steering gear is capable of steering the ship at navigable speed and of being brought speedily into action in an emergency and that it is capable of putting the rudder over from 15° on one side to 15° on the other side in not more than 60 s with the ship at its deepest seagoing draught and running ahead at one half of the maximum ahead service speed or 7 knots, whichever is the greater,⁴⁵ or, where this is impracticable, with an alternative permissible sea trial loading condition⁴⁶ (SOLAS 74/88/14 reg.II-1/29);
- (PI) 5.1.3.38 confirming that the main or auxiliary steering gear power units restart automatically when power is restored after a power failure, that they are capable of being brought into operation from a position on the navigating bridge and that, in the event of a power failure to any one of the steering gear power units, an audible and visual alarm is given on the navigating bridge (SOLAS 74/88/14 reg.II-1/29);
- (PI) 5.1.3.39 confirming that, where the main steering gear comprises two or more identical power units and an auxiliary steering gear is not fitted, a defect can be isolated so that steering capability can be maintained or speedily

⁴² For ships fitted with alternative propulsion and steering arrangements other than traditional arrangements, such as but not limited to azimuthing propulsors or water jet propulsion systems, refer to the *Unified interpretations of SOLAS regulations II-1/28, II-29 and II-30* (MSC.1/Circ.1416/Rev.1).

⁴³ For ships fitted with alternative propulsion and steering arrangements other than traditional arrangements, such as but not limited to azimuthing propulsors or water jet propulsion systems, refer to the *Unified interpretations of SOLAS regulations II-1/28, II-29 and II-30* (MSC.1/Circ.1416/Rev.1).

⁴⁴ Refer to the Unified interpretations of SOLAS regulations II-1/29.3 and II-1/29.4 (MSC.1/Circ.1536).

⁴⁵ For ships fitted with alternative propulsion and steering arrangements other than traditional arrangements, such as but not limited to azimuthing propulsors or water jet propulsion systems, refer to the *Unified interpretations of SOLAS regulations II-1/28, II-29 and II-30* (MSC.1/Circ.1416/Rev.1).

⁴⁶ Refer to the *Unified interpretations of SOLAS regulations II-1/29.3 and II-1/29.4* (MSC.1/Circ.1536).

regained after a single failure in its piping system or in one of the power units^{*} (SOLAS 74/88/14 reg.II-1/29);

- (PI) 5.1.3.40 confirming that the control systems for the main steering gear from both the navigating bridge and the steering gear compartment are operating satisfactorily (SOLAS 74/88/14 reg.II-1/29);
- (PI) 5.1.3.41 confirming that, where the main steering gear comprises two or more identical power units and an auxiliary steering gear is not fitted, the two independent control systems from the navigating bridge are operating satisfactorily (SOLAS 74/88/14 reg.II-1/29);
- (PI) 5.1.3.42 confirming that the control system for the auxiliary steering gear, in the steering gear compartment and, if this gear is power-operated, from the navigating bridge, is operating satisfactorily and that the latter is independent of the control system for the main steering gear (SOLAS 74/88/14 reg.II-1/29);
- (PI) 5.1.3.43 confirming that the control system for any main and auxiliary steering gear control system operable from the navigating bridge is capable of being brought into operation from a position on the navigating bridge, that means are provided in the steering gear compartment for disconnecting it from the steering gear that it serves and that an audible and visual alarm is given on the navigating bridge in the event of a failure of electrical power supply (SOLAS 74/88/14 reg.II-1/29);
- (PI) 5.1.3.44 confirming that the electric power circuits and steering gear control system, together with their associated components, cables and pipes, are separated, as far as practicable, throughout their length (SOLAS 74/88/14 reg.II-1/29);
- (PI) 5.1.3.45 confirming that the means of communication between the bridge and the steering gear is operating satisfactorily and that, with ships having emergency steering positions, a telephone or other means of communication for relaying heading information and supplying visual compass readings to the emergency steering position are provided (SOLAS 74/88/14 reg.II-1/29) (SOLAS 74/00 reg.V/19);
- (PI) 5.1.3.46 confirming that the angular position of the rudder is indicated independently of the steering control system on the navigating bridge if the main steering gear is power-operated and that this angular position is given in the steering gear compartment (SOLAS 74/88/14 reg.II-1/29) (SOLAS 74/00 reg.V/19);
- (PI) 5.1.3.47 confirming that with a hydraulic power-operated steering gear the audible and visual low-level alarms on the navigating bridge and in the machinery space for each hydraulic fluid reservoir are operating satisfactorily and that at least one power actuating system including the reservoir can be recharged from a position within the steering gear compartment by means of a fixed storage tank to which a contents gauge is fitted with fixed piping (SOLAS 74/88/14 reg.II-1/29);

- (PI) 5.1.3.48 confirming that the steering gear compartment is readily accessible, that it is separated, as far as practicable, from machinery spaces and is provided with suitable arrangements to ensure working access to steering gear machinery and controls under safe conditions (SOLAS 74/88/14 reg.II-1/29);
- (PI) 5.1.3.49 confirming that with electric and electro-hydraulic steering gear, the means for indicating, on the navigating bridge and at a main machinery control position, that the motors are running and that the overload alarm and alarm for the loss of a phase in a three-phase supply located at the main machinery control position are operating satisfactorily (SOLAS 74/88 reg.II-1/30);
- (PI) 5.1.3.50 confirming that the main and auxiliary machinery essential for propulsion and the safety of the ship are provided with the effective means for its operation and control (SOLAS 74/88 reg.II-1/31);
- (PI) 5.1.3.51 confirming that appropriate means are provided where it is intended that the propulsion machinery should be remotely controlled from the navigating bridge, including, where necessary, the control, monitoring, reporting, alert and safety actions (SOLAS 74/00/02 reg.II-1/31);
- (PI) 5.1.3.52 confirming that arrangements to operate main and other machinery from a machinery control room are satisfactory (SOLAS 74/88 reg.II-1/31);
- (PI) 5.1.3.53 confirming that, in general, means are provided for manually overriding automatic controls and that a failure does not prevent the use of the manual override (SOLAS 74/88 reg.II-1/31);
- (PI) 5.1.3.54 confirming that oil-fired and exhaust gas boilers, unfired steam generators, steam pipe systems and air pressure systems are fitted with the appropriate safety features (SOLAS 74/88 regs.II-I/32, 33 and 34);
- (PI) 5.1.3.55 confirming the operation of the ventilation for the machinery spaces (SOLAS 74/88 reg.II-I/35);
- (PI) 5.1.3.56 when appropriate, confirming that the measures to prevent noise in machinery spaces are effective (SOLAS 74/88 reg.II-I/36 and SOLAS 74/12/16 reg.II-1/3-12.2); or confirming that the ship was constructed to reduce onboard noise and to protect personnel from noise in accordance with the Code on Noise Levels on Board Ships, adopted by resolution MSC.337(91), as amended (SOLAS 74/12/16 reg.II-1/3-12);
- (PI) 5.1.3.57 confirming that the engine-room telegraph giving visual indication of the orders and answers both in the machinery space and on the navigating bridge is operating satisfactorily (SOLAS 74/88 regulation II-1/37);
- (PI) 5.1.3.58 confirming that the second means of communication between the navigation bridge and machinery space is also operating satisfactorily and that appropriate means are provided to any other positions from which the engines are controlled (SOLAS 74/88 regulation II-1/37);

- (PI) 5.1.3.59 confirming that the engineer's alarm is clearly audible in the engineers' accommodation (SOLAS 74/88 regulation II-1/38);
- (PI) 5.1.3.60 confirming that precautions, taken to prevent any oil that may escape under pressure from any pump, filter or heater from coming into contact with heated surfaces, are efficient;
- (PI) 5.1.3.61 confirming that the means of ascertaining the amount of oil contained in any oil tank are in satisfactory working condition (SOLAS 74/88 reg.II-2/15) (SOLAS 74/00 reg.II-2/4.2.2.3.5);
- (PI) 5.1.3.62 confirming that the devices provided to prevent overpressure in any oil tank or in any part of the oil system, including the filling pipes, are in satisfactory working condition (SOLAS 74/88 reg.II-2/15) (SOLAS 74/00 reg.II-2/4.2.2.4);
- (PI) 5.1.3.63 confirming that forepeak tanks are not intended for carriage of oil fuel, lubrication oil and other flammable oils;
- (PI) 5.1.3.64 confirming that the electrical installations, including the main source of power and lighting systems, are installed in accordance with the approved plans (SOLAS 74/88 regs.II-1/40 and 41);
- (PI) 5.1.3.65 confirming that a self-contained emergency source of electrical power has been provided and that the appropriate systems are satisfactorily supplied (SOLAS 74/88 reg.II-1/42);
- (PI) 5.1.3.66 confirming that the starting arrangements of each emergency generating set are satisfactory (SOLAS 74/88 reg.II-1/44);
- (PI) 5.1.3.67 checking, when appropriate, the disposition of, and testing, the supplementary emergency lighting (SOLAS 74/88 reg.II-1/42-1);
- (PI) 5.1.3.67.1 for passenger ships, constructed on or after 1 July 2010, confirming provision of supplementary lighting in all cabins, and checking that such lighting automatically illuminates and remains on for a minimum of 30 min when power to the normal cabin lighting is lost (SOLAS 74/06/10 reg.II-1/41.6);
- (PI) 5.1.3.67.2 for passenger ships constructed on or after 1 July 2010, checking the provision of smoke detectors in cabins, which, when activated, are capable of emitting, or cause to be emitted, an audible alarm within the space where they are located (SOLAS 74/06 regs.II-2/7.5.2 and 7.5.3.1);
- (PI) 5.1.3.68 confirming that precautions have been provided against shock, fire and other hazards of electrical origin (SOLAS 74/88 reg.II-1/45);
- (PI) 5.1.3.69 confirming, when appropriate, that the arrangements for the machinery spaces being periodically unattended are satisfactory (SOLAS 74/88 reg.II-1/54);
- (PI) 5.1.3.70 examining, where applicable, the alternative design and arrangements for machinery or electrical installations, low-flashpoint fuel storage and distribution systems, fire safety, or life-saving appliances and

arrangements, in accordance with the test and inspection requirements, if any, specified in the approved documentation (SOLAS 74/00/06/15 regs.II-1/55, II-2/17 and III/38 and IGF Code ch.2);

- (PI) 5.1.3.71 examining the fire pumps and fire main and the disposition of the hydrants, hoses and nozzles and the international shore connection and checking that each fire pump, including the emergency fire pump, can be operated separately so that two jets of water are produced simultaneously from different hydrants at any part of the ship while the required pressure is maintained in the fire main and testing that the emergency fire pump, if applicable, has the required capacity, and, if the emergency fire pump is the main supply of water for any fixed fire-extinguishing system, checking that the emergency fire pump has the capacity for this system⁴⁷ (SOLAS 74/88 regs.II-2/4 and 19; SOLAS 74/00/14 reg.II-2/10.2; FSS Code chs.2 and 12);
- (PI) 5.1.3.72 for passenger ships designed to carry containers on or above the weather deck, as applicable, examining the water mist lance (SOLAS 74/00/14 reg.II-2/10.7.3);
- (PI) 5.1.3.73 examining the provision and disposition of the fire extinguishers (SOLAS 74/00 reg.II-2/10.3; FSS Code ch.4) (SOLAS 74/88 reg.II-2/17);
- (PI) 5.1.3.74 examining the fire-fighters' outfits including their self-contained compressed air breathing apparatus, and emergency escape breathing devices (EEBDs); confirming that they are complete and in satisfactory condition and that the cylinders, including the spare cylinders, of the self-contained breathing apparatus, are suitably charged, and that onboard means of recharging breathing apparatus cylinders used during drills or a suitable number of spare cylinders to replace those used are provided, and provision of two-way portable radiotelephone apparatus of an explosion-proof type or intrinsically safe. (SOLAS 74/00/08/12 regs.II-2/10.10, 13.3.4, 13.4.3 and 15.2.2; FSS Code ch.3) (SOLAS 74/88 reg.II-2/17);
- (PI) 5.1.3.75 checking the operational readiness and maintenance of fire-fighting systems (SOLAS 74/00 reg.II-2/14) (SOLAS 74/88 regs.II-2/6, 17 and 21);
- (PI) 5.1.3.76 examining the fixed fire-fighting system for the machinery, cargo, special category and vehicle spaces, as appropriate, and confirming that the installation tests have been satisfactorily completed and that its means of operation are clearly marked (SOLAS 74/00/12/14 regs.II-2/10.4, 10.5, 10.7.1, 10.7.2 and 20.6.1; FSS Code ch.5 to 7) (SOLAS 74/88 regs.II-2/7 and 53);
- (PI) 5.1.3.77 examining the fire-extinguishing and special arrangements in the machinery spaces and confirming, as far as practicable and as appropriate, the operation of the remote means of control provided for the opening and closing of the skylights, the release of smoke, the closure of the funnel and ventilation openings, the closure of power-operated and other doors, the stopping of ventilation and boiler forced and induced draught fans and the stopping of oil fuel and other pumps that discharge

⁴⁷ Refer to the Unified interpretation of chapter 12 of the International Code for Fire Safety Systems (MSC.1/Circ.1388).

flammable liquids (SOLAS 74/00/12/14/16 regs.II-2/5.2, 8.3, 9.5 and 10.5) (SOLAS 74/88 regs.II-2/7 and 11);

- (PI) 5.1.3.78 checking that fixed carbon dioxide fire-extinguishing systems for the protection of machinery spaces, where applicable, are provided with two separate controls, one for opening the gas piping and one for discharging the gas from the storage container, each of them located in a release box clearly identified for the particular space (SOLAS 74/08 reg.II-2/10.4; FSS Code ch.5.2.2.2);
- (PI) 5.1.3.79 examining the arrangements for oil fuel, lubricating oil and other flammable oils and testing the remote closing of valves for oil fuel, lubricating oil and other flammable oils and the operation of the remote means of closing the valves on the tanks that contain oil fuel, lubricating oil and other flammable oils (SOLAS 74/88/06 reg.II-2/15) (SOLAS 74/00/15 reg.II-2/4.2);
- (PI) 5.1.3.80 examining any fire detection and alarm system and confirming that installation tests have been satisfactorily completed (SOLAS 74/88 regs.II-2/11, 12, 13, 14, 36 and 41);
- (PI) 5.1.3.81 confirming that all aspects of installation of the structural fire protection, including the structure, fire integrity, protection of stairways and lifts, cabin balconies, openings in "A" and "B" Class divisions, ventilation systems and windows and side scuttles, and the use of combustible material are in accordance with the approved plans (SOLAS 74/00/04/12/15/17 regs.II-2/4.4.4, 5.2, 5.3, 7.5, 7.8.2, 8.4, 8.5, 9, 10.6, 11, 13, 17, 20, 20-1 and FSS Code ch.13 sections 1 and 2) (SOLAS 74/88 regs.II-2/23 to 35);
- (PI) 5.1.3.82 testing any manual and automatic fire doors, including the means of closing the openings in "A" and "B" Class divisions (SOLAS 74/88 regs.II-2/30 and 31);
- (PI) 5.1.3.83 testing the means of closing the main inlets and outlets of all ventilation smoke extraction systems and proving that the power ventilation is capable of being stopped from outside the space served (SOLAS 74/88 reg.II-2/32);
- (PI) 5.1.3.84 confirming that stairways and ladders are so arranged as to provide a means of escape to the lifeboat and liferaft and liferaft embarkation deck from all passenger and crew spaces and from those spaces in which the crew is normally employed (SOLAS 74/00/16 regs.II- 2/13.3.2 and 13.7) and in particular that:
- (PI) 5.1.3.84.1 below the bulkhead deck there are two means of escape from each watertight compartment, one being independent of watertight doors;
- (PI) 5.1.3.84.2 above the bulkhead deck there are two means of escape from each vertical zone or similar such area, one leading directly to a stairway forming a vertical escape;

- (PI) 5.1.3.84.3 the radiotelegraph station, if provided, has direct access to the open deck or is provided with two means of access or egress, one of which is a porthole or window of sufficient size;
- (PI) 5.1.3.85 confirming that the means of escape from any special category spaces are generally in accordance with (PI) 5.1.3.84 (SOLAS 74/88 reg.II-2/28);
- (PI) 5.1.3.86 confirming that in the machinery spaces there are two widely separated means of escape leading to the lifeboat and liferaft embarkation decks, including, when from a space below the bulkhead deck, a continuous fire shelter, and that two means of escape are provided for the main workshop located within the machinery space, as applicable (SOLAS 74/00/14 reg.II-2/13.4.1; FSS Code chapter 13) (SOLAS 74/88 reg.II-2/28);
- (PI) 5.1.3.87 confirming the fire protection arrangements, including fire detection and sample extraction smoke detection systems for cargo spaces for cargo and dangerous goods and testing, as appropriate, the operation of the means for closing the various openings (SOLAS 74/88 reg.II-2/39) (SOLAS 74/00/14 regs.II-2/7.6, 10.7.1 and 10.7.2; FSS Code chs.5, 9 and 10);
- (PI) 5.1.3.88 confirming the fire protection arrangements, including fire detection and sample extraction smoke detection systems, where applicable for vehicle, special category and ro-ro spaces and testing, as appropriate, the operation of the means for closing the various openings (SOLAS 74/88 regs.II-2/37, and 38) (SOLAS 74/00/15/17 reg.II-2/20 (except 20.5); FSS Code chs.5, 6, 7, 9, 10);
- (PI) 5.1.3.89 confirming and testing, as appropriate, any fixed fire detection and alarm system, and any automatic sprinkler, fire detection and fire alarm system, as applicable, in machinery spaces, including enclosed spaces containing incinerators, accommodation, service and control spaces (SOLAS 74/88 reg.II-2/40) (SOLAS 74/00/06/10 reg.II-2/7 (except 7.5.5, 7.6 and 7.9); FSS Code chs.8 and 9);
- (PI) 5.1.3.90 confirming and testing the special alarm and the public address system or other effective means of communication (SOLAS 74/88 reg.II-2/40) (SOLAS 74/00/06/10 reg.II-2/12; LSA Code ch.7);
- (PI) 5.1.3.91 for passenger ships constructed on or after 1 July 2010, confirming the provision of a fixed fire detection and fire alarm system for passenger ships capable of remotely and individually identifying each detector and manually operated call point (SOLAS 74/06 reg.II-2/7.2.4);
- (PI) 5.1.3.92 examining the helicopter facilities, including foam firefighting appliances when appropriate (SOLAS 74/00/16 reg.II-2/18; FSS Code ch.17) (SOLAS 74/88 reg.II-2/18.8);
- (PI) 5.1.3.93 examining, when appropriate, the special arrangements for carrying dangerous goods, including checking the electrical equipment and wiring, fire detection, ventilation and boundary insulation, the provision of protective clothing and portable appliances and the testing of the water supply, bilge pumping and any water spray system (SOLAS 74/88 regs.II-2/41 and 54) (SOLAS 74/00/08 reg.II-2/19);

- (PI) 5.1.3.94 checking the provision and disposition of the survival craft and rescue boats and the arrangements for mustering passengers (SOLAS 74/88 regs III/11 to 16, 20 and 24);
- (PI) 5.1.3.95 examining each survival craft, including its equipment, and that the required number of search and rescue locating devices are fitted in liferafts and those liferafts are clearly marked (SOLAS 74/88/00/02/08 regs.III/20, 21 and 26; LSA Code sections 2.3 to 2.5, 3.2 and 4.1 to 4.6);
- (PI) 5.1.3.96 examining the embarkation arrangements for each survival craft and the testing of each launching appliance, including overload tests, tests to establish the lowering speed and the lowering of each survival craft to the water with the ship at its lightest seagoing draught, checking the recovery of each lifeboat (SOLAS 74/88 regs.III/11, 12, 13, 15, 20 and 48);
- (PI) 5.1.3.97 deploying 50% of the MES after installation (LSA Code paragraph 6.2.2.2);
- (PI) 5.1.3.98 examining each rescue boat, including its equipment; for inflatable rescue boats, confirming that they are stowed in a fully inflated condition (SOLAS 74/00/04 regs.III/21 and 26.3; LSA Code section 5.1 and MSC/Circ.809);
- (PI) 5.1.3.99 examining the embarkation and recovery arrangements for each rescue boat and testing each launching and recovery appliance, including overload tests, tests to establish the lowering and recovery speeds and ensuring that each rescue boat can be lowered to the water and recovered with the ship at its lightest seagoing draught; the rescue boat(s) should be lowered to the water and its recovery demonstrated while underway at 5 knots (SOLAS 74/88 regs.III/14, 16, 17 and 20);
- (PI) 5.1.3.100 examining the arrangements for mustering passengers (SOLAS 74/88 reg.III/24);
- (PI) 5.1.3.101 testing that the engine of the rescue boat(s) and of each lifeboat, when so fitted, start satisfactorily and operate both ahead and astern (LSA Code section 4.4.6.5);
- (PI) 5.1.3.102 confirming that there are posters or signs in the vicinity of survival craft and their launching stations (SOLAS 74/88 reg.III/9);
- (PI) 5.1.3.103 examining the provision and stowage and checking the operation of two-way VHF radiotelephone apparatus⁴⁸ and search and rescue locating devices (SOLAS 74/88/08 reg.III/6);

⁴⁸ Expired primary batteries, which complied with the requirements of paragraph 12.6 of *Revised Recommendation on performance standards for survival craft portable Two-Way VHF radiotelephone apparatus* (resolution MSC.149 (77)), could be used for no other purpose than to examine and check the operation of two-way VHF radiotelephone apparatus.

- (PI) 5.1.3.104 examining the provision and stowage of the distress flares and the line-throwing appliance, checking the provision and operation of onboard communications equipment and testing the means of operation of the general alarm system, verifying that the general alarm system is audible in accommodation, normal crew working spaces and on open decks (SOLAS 74/88 reg.III/6);
- (PI) 5.1.3.105 examining the provision, disposition and stowage of the lifebuoys, including those fitted with self-igniting lights, self-activating smoke signals and buoyant lines, lifejackets,⁴⁹ immersion suits and thermal protective aids (SOLAS 74/88/06 regs.III/7, 21, 22 and 26; LSA Code section 2.1-2.5 and 3.3);
- (PI) 5.1.3.105.1 checking that the life-saving appliances are of an international or vivid reddish orange, or a comparably highly visible colour on all parts where this will assist detection at sea (LSA Code section 1.2.2.6);
- (PI) 5.1.3.105.2 checking the provision of lifejackets in three sizes (Infant, Child, Adult) and checking that they are marked by either weight or height, or by both weight and height (LSA Code section 2.2.1.1); for passenger ships on voyages less than 24 hours, checking that the number of infant lifejackets is equal to at least 2.5% of the number of passengers on board and for passenger ships on voyages 24 hours or greater, checking that infant lifejackets are provided for each infant on board (SOLAS 74/06 reg.III/7.2.1);
- (PI) 5.1.3.105.3 checking that immersion suits designed to be worn in conjunction with a lifejacket are suitably marked (LSA Code section 2.3.1);
- (PI) 5.1.3.106 checking the lighting of the muster and embarkation stations and the alleyways, stairways and exits giving access to the muster and embarkation stations, including when supplied from the emergency source of power (SOLAS 74/88 regs.II-1/42 and III/11);
- (PI) 5.1.3.107 checking that means of rescue is provided on ro-ro passenger ships (SOLAS 74/00 reg.III/26.4);
- (PI) 5.1.3.108 checking that a helicopter pick-up area is provided on ro-ro passenger ships (SOLAS 74/00 reg.III/28);
- (PI) 5.1.3.109 checking that a decision support system is provided for the master (SOLAS 74/00 reg.III/29; SOLAS 74/06 regs.II-2/21 and 22);
- (PI) 5.1.3.110 checking the electromagnetic compatibility of electrical and electronic equipment on or in the vicinity of the bridge (SOLAS 74/00 reg.V/17);
- (PI) 5.1.3.111 examining the provision and positioning and checking the operation of, as appropriate, the navigation lights, shapes and sound signalling equipment (International Regulations for Preventing Collisions at Sea in force, rules 20 to 24, 27 to 30 and 33);

⁴⁹ SOLAS Regulations III/7.2.1.1, 7.2.1.2 and 7.2.1.5 should be considered.

- (PI) 5.1.1.112 checking the provision and specification of the daylight signalling lamp (SOLAS 74/88 reg.V/11);
- (PI) 5.1.3.113 checking, as appropriate, the provision and operation of the following equipment (SOLAS 74/00 reg.V/19):
- (PI) 5.1.3.113.1 the magnetic compass, including examining the siting, movement, illumination and a pelorus or compass bearing device (SOLAS 74/00 reg.V/19);
- (PI) 5.1.3.113.2 that nautical charts and nautical publications necessary for the intended voyage are available and have been updated and, where ECDIS is used, that the electronic charts have been updated and the required backup system is provided and updated (SOLAS 74/00/09 reg.V/19);
- (PI) 5.1.3.113.3 global navigation satellite receiver or terrestrial radionavigation system;
- (PI) 5.1.3.113.4 sound reception system, when bridge is totally enclosed;
- (PI) 5.1.3.113.5 means of communication to emergency steering position, where provided;
- (PI) 5.1.3.113.6 spare magnetic compass;
- (PI) 5.1.3.113.7 daylight signalling lamp;
- (PI) 5.1.3.113.8 echo-sounding device, including examining the display for good access, viewing and lighting;
- (PI) 5.1.3.113.9 radar(s), including examining the waveguide and cable runs for routeing and protection and the display unit confirming lighting, plotting facilities, correct operation of all controls, functions and the true-motion facility if provided;
- (PI) 5.1.3.113.10 electronic plotting aid, automatic tracking aid or automatic radar plotting aid as appropriate, using the appropriate test facilities;
- (PI) 5.1.3.113.11 speed and distance measuring device;
- (PI) 5.1.3.113.12 transmitting heading device providing heading information to radar, plotting aids and automatic identification system equipment and distance devices;
- (PI) 5.1.3.113.13 heading or track control system;
- (PI) 5.1.3.113.14 BNWAS;
- (PI) 5.1.3.114 checking for the provision, specification, operation and annual performance test of the voyage data recorder (SOLAS 74/00/04 reg.V/20);
- (PI) 5.1.3.115 checking that a valid conformance test report of the long-range identification and tracking system is available on board (SOLAS 74/04 reg.V/19-1);

(PI)	5.1.3.116	checking that the International Code of Signals and an up-to-date copy of Volume III of the International Aeronautical and Maritime Search and Rescue (IAMSAR) Manual have been provided (SOLAS 74/00/02 reg.V/21);
(PI)	5.1.3.117	checking the provision of the pilot transfer arrangement, the access to the ship's deck and the associated equipment and lighting, checking the operation of the pilot ladders and combination arrangements, where applicable (SOLAS 74/00/10 reg.V/23);
(PI)	5.1.3.118	examining the position, physical and electromagnetic protection and illumination of each radio installation (SOLAS 74/88 reg.IV/6);
(PI)	5.1.3.119	confirming the provision of equipment for the radio installation with due regard to the declared sea areas in which the ship will trade and the declared means of maintaining availability of functional requirements (SOLAS 74/88 regs.III/6, IV/7 to 11, 14 and 15);
(PI)	5.1.3.120	confirming the ability to initiate the transmission of ship-to-shore distress alerts by at least two separate and independent means, each using a different radio communication service, from the position from which the ship is normally navigated (SOLAS 74/88/06 regs.IV/4, 7 to 11);
(PI)	5.1.3.121	examining all antennas, including:
(PI)	5.1.3.121.1	visually checking all antennas, including recognized mobile satellite service antennas, and feeders for satisfactory siting and absence of defects (SOLAS 74/88/18 reg.IV/14);
(PI)	5.1.3.121.2	checking insulation and safety of all antennas;
(PI)	5.1.3.122	examining the reserve source of energy, including:
(PI)	5.1.3.122.1	checking there is sufficient capacity to operate the basic or duplicated equipment for 1 hour or 6 hours, as appropriate (SOLAS 74/88 reg.IV/13);
(PI)	5.1.3.122.2	and, if the reserve source of energy is a battery:
(PI)	5.1.3.122.2.1	checking its siting and installation (SOLAS 74/88 reg.IV/13);
(PI)	5.1.3.122.2.2	where appropriate, checking its condition by specific gravity measurement or voltage measurement;
(PI)	5.1.3.122.2.3	with the battery off charge, and the maximum required radio installation load connected to the reserve source of energy, checking the battery voltage and discharge current;
(PI)	5.1.3.122.2.4	checking that the charger or chargers are capable of recharging the reserve battery within 10 hours (SOLAS 74/88 reg.IV/13);
(PI)	5.1.3.123	examining the VHF transceiver(s), including:

- (PI) 5.1.3.123.1 checking for operation on channels 6, 13 and 16 (SOLAS 74/88 regs.IV/7 and 14);
- (PI) 5.1.3.123.2 checking frequency tolerance, transmission line quality and radio frequency power output (SOLAS 74/88 reg.IV/14);
- (PI) 5.1.3.123.3 checking for correct operation of all controls including priority of control units (SOLAS 74/88 reg.IV/14);
- (PI) 5.1.3.123.4 checking that the equipment operates from the main, emergency (if provided) and reserve sources of energy (SOLAS 74/88 reg.IV/13);
- (PI) 5.1.3.123.5 checking the operation of the VHF control unit(s) or portable VHF equipment provided for navigational safety (SOLAS 74/88 reg.IV/6);
- (PI) 5.1.3.123.6 checking for correct operation by on-air contact with a coast station or other ship;
- (PI) 5.1.3.124 examining the VHF DSC controller and channel 70 DSC watch receiver, including:
- (PI) 5.1.3.124.1 performing an off-air check confirming the correct Maritime Mobile Service Identity is programmed in the equipment (SOLAS 74/88 reg.IV/14);
- (PI) 5.1.3.124.2 checking for correct transmission by means of a routine or test call to a coast station, other ship, onboard duplicate equipment or special test equipment;
- (PI) 5.1.3.124.3 checking for correct reception by means of a routine or test call from a coast station, other ship, onboard duplicate equipment or special test equipment;
- (PI) 5.1.3.124.4 checking the audibility of the VHF/DSC alarm;
- (PI) 5.1.3.124.5 checking that the equipment operates from the main, emergency (if provided) and reserve sources of energy (SOLAS 74/88 reg.IV/13);
- (PI) 5.1.3.125 examining the MF/HF radiotelephone equipment, including:
- (PI) 5.1.3.125.1 checking that the equipment operates from the main, emergency (if provided) and reserve sources of energy (SOLAS 74/88 reg.IV/13);
- (PI) 5.1.3.125.2 checking the antenna tuning in all appropriate bands;
- (PI) 5.1.3.125.3 checking the equipment is within frequency tolerance on all appropriate bands (SOLAS 74/88 reg.IV/14);
- (PI) 5.1.3.125.4 checking for correct operation by contact with a coast station and/or measuring transmission line quality and radio frequency output;
- (PI) 5.1.3.125.5 checking receiver performance by monitoring known stations on all appropriate bands;

(PI)	5.1.3.125.6	if control units are provided outside the navigating bridge, checking the control unit on the bridge has first priority for the purpose of initiating distress alerts (SOLAS 74/88 regs.IV/9, 10, 11 and 14);
(PI)	5.1.3.125.7	checking the correct operation of the radiotelephone alarm signal generating device on a frequency other than 2182 kHz;
(PI)	5.1.3.126	examining the HF radiotelex equipment, including:
(PI)	5.1.3.126.1	checking that the equipment operates from the main, emergency (if provided) and reserve sources of energy (SOLAS 74/88 reg.IV/13);
(PI)	5.1.3.126.2	confirming that the correct selective calling number is programmed in the equipment;
(PI)	5.1.3.126.3	checking correct operation by inspection of recent hard copy or by a test with a coast radio station (SOLAS 74/88 regs.IV/10 and 11);
(PI)	5.1.3.127	examining the MF/HF DSC controller(s), including:
(PI)	5.1.3.127.1	checking that the equipment operates from the main, emergency (if provided) and reserve sources of energy (SOLAS 74/88 reg.IV/13);
(PI)	5.1.3.127.2	confirming that the correct Maritime Mobile Service Identity is programmed in the equipment;
(PI)	5.1.3.127.3	checking the off-air self-test programme;
(PI)	5.1.3.127.4	checking operation by means of a test call on MF and/or HF to a coast radio station if the rules of the berth permit the use of MF/HF transmissions (SOLAS 74/88 regs.IV/9 to 11);
(PI)	5.1.3.127.5	checking the audibility of the MF/HF DSC alarm;
(PI)	5.1.3.128	examining the MF/HF DSC watch receiver(s), including:
(PI)	5.1.3.128.1	confirming that only distress and safety DSC frequencies are being monitored (SOLAS 74/88 regs.IV/9 to 12);
(PI)	5.1.3.128.2	checking that a continuous watch is being maintained while keying MF/HF radio transmitters (SOLAS 74/88 reg.IV/12);
(PI)	5.1.3.128.3	checking for correct operation by means of a test call from a coast station or other ship;
(PI)	5.1.3.129	examining the recognized mobile satellite service ship earth station(s), including:

- (PI) 5.1.3.129.1 checking that the equipment operates from the main, emergency (if provided) and reserve sources of energy, and that where an uninterrupted supply of information from the ship's navigational or other equipment is required ensuring such information remains available in the event of failure of the ship's main or emergency source of electrical power (SOLAS 74/88 regs.IV/13 and 14);
- (PI) 5.1.3.129.2 checking the distress function by means of an approved test procedure where possible (SOLAS 74/88 regs.IV/10, 12 and 14);
- (PI) 5.1.3.129.3 checking for correct operation by inspection of recent hard copy or by test call;
- (PI) 5.1.3.130 if appropriate, examining the NAVTEX equipment (SOLAS 74/88 regs.IV/7, 12 and 14), including:
- (PI) 5.1.3.130.1 checking for correct operation by monitoring incoming messages or inspecting recent hard copy;
- (PI) 5.1.3.130.2 running the self-test programme if provided;
- (PI) 5.1.3.131 examining the enhanced group call equipment (SOLAS 74/88 regs.IV/7 and 14), including:
- (PI) 5.1.3.131.1 checking for correct operation and area by monitoring incoming messages or by inspecting recent hard copy;
- (PI) 5.1.3.131.2 running the self-test programme if provided;
- (PI) 5.1.3.132 if appropriate, examining the radio equipment for receipt of maritime safety information by HF NBDP (SOLAS 74/88 regs.IV/7, 12 and 14), including:
- (PI) 5.1.3.132.1 checking for correct operation by monitoring incoming messages or inspecting recent hard copy;
- (PI) 5.1.3.132.2 running the self-test programme if provided;
- (PI) 5.1.3.133 examining the 406 MHz satellite EPIRB (SOLAS 74/88 regs.IV/7 and 14), including:
- (PI) 5.1.3.133.1 checking position and mounting for float-free operation;
- (PI) 5.1.3.133.2 carrying out visual inspection for defects;
- (PI) 5.1.3.133.3 carrying out the self-test routine;
- (PI) 5.1.3.133.4 checking that the unique beacon identification code is clearly marked on the outside of the equipment and, where possible, decoding the unique beacon identification code confirming it is correct;
- (PI) 5.1.3.133.5 checking that the unique beacon identification code programmed in the EPIRB corresponds with the unique beacon identification code assigned by or on behalf of the Administration;

MSC 104/18/Add.1 Annex 31, page 110

(PI)	5.1.3.133.6	checking that the MMSI number if encoded in the beacon corresponds with the MMSI number assigned to the ship;
(PI)	5.1.3.133.7	checking the battery expiry date;
(PI)	5.1.3.133.8	if provided, checking the hydrostatic release and its expiry date;
(PI)	5.1.3.134	examining the two-way VHF radiotelephone apparatus (SOLAS 74/88 reg.III/6), including:
(PI)	5.1.3.134.1	checking for correct operation on channel 16 and one other by testing with another fixed or portable VHF installation (SOLAS 74/88 reg.IV/14);
(PI)	5.1.3.134.2	checking the battery charging arrangements where rechargeable batteries are used (SOLAS 74/88 reg.IV/14);
(PI)	5.1.3.134.3	checking the expiry date of primary batteries where used (SOLAS 74/88 reg.IV/14);
(PI)	5.1.3.134.4	where appropriate, checking any fixed installation provided in a survival craft (SOLAS 74/88 reg.IV/14);
(PI)	5.1.3.135	examining the search and rescue locating device(s) (SOLAS 74/88/08 reg.III/6 and regs.IV/7 and 14), including:
(PI)	5.1.3.135.1	checking the position and mounting;
(PI)	5.1.3.135.2	monitoring response on ship's 9 GHz radar;
(PI)	5.1.3.135.3	checking the battery expiry date;
(PI)	5.1.3.136	examining the test equipment and spares carried to ensure carriage is adequate in accordance with the sea areas in which the ship trades and the declared options for maintaining availability of the functional requirements (SOLAS 74/88 reg.IV/15);
(PI)	5.1.3.137	checking the distress panel installed at the conning position; or, where applicable, checking an additional EPIRB is placed near the conning position (SOLAS 74/88 reg.IV/6);
(PI)	5.1.3.138	checking that positional information is provided continuously and automatically to all communications equipment included in the initial distress alert (SOLAS 74/88 reg.IV/6);
(PI)	5.1.3.139	checking the distress alarm panel installed at the conning position and its visual and aural indications of received distress alerts (SOLAS 74/88 reg.IV/6);
(PI)	5.1.3.140	checking the provision and operation of the means for two-way on-scene communication for search and rescue purposes and its operation on 121.5 MHz and 123.1 MHz from the position from which the ship is normally navigated (SOLAS 74/88 reg.IV/7);

- (PI) 5.1.3.141 confirming that the ship's identification number is permanently marked (SOLAS 74/02 reg.XI-1/3);
- (PI) 5.1.3.142 checking the provision and operation of the automatic identification system (SOLAS 74/00/04 reg.V/19);
- (PI) 5.1.3.143 for passenger ships carrying more than 36 passengers constructed on or after 1 July 2010, checking the provision of a suitably located means for fully recharging breathing air cylinders, free from contamination (SOLAS 74/08 reg.II-2/10.10.2.6);
- (PI) 5.1.3.144 confirming that installed materials do not contain asbestos⁵⁰ (SOLAS 74/09 reg.II-1/3-5);
- (PI) 5.1.3.145 confirming the provision of means of embarkation and disembarkation from ships for use in port and in port-related operations, such as gangways and accommodation ladders (SOLAS 74/08 reg.II-1/3-9);
- (PI) 5.1.3.146 for passenger ships constructed on or after 1 July 2010 and having a length of 120 m or more or having three or more main vertical zones, checking the designation of safe areas (SOLAS 74/06 reg.II-2/21);
- (PI) 5.1.3.147 for passenger ships constructed on or after 1 July 2010, checking the provision of a safety centre (SOLAS 74 reg.II-2/23) and associated ventilation requirements (SOLAS 74/06 reg.II-2/8.2); and
- (PI) 5.1.3.148 confirming, where applicable, that an appropriate portable atmosphere testing instrument or instruments⁵¹ is on board, and that suitable means are provided for the calibration of all such instruments;⁵² and checking the appropriateness of the testing and calibration (SOLAS 74/14 reg.XI-1/7).
- (PI) 5.1.4 For the hull, machinery and equipment of passenger ships using natural gas as fuel the additional requirements for the survey during construction and after installation should consist of:
- (PI) 5.1.4.1 confirming that the arrangement of the accommodation, fuel containment systems, service and machinery spaces are in accordance with the approved plans and control, monitoring and safety systems are satisfactory (IGF Code chs.4, 5, 6, 8, 9 and 15);
- (PI) 5.1.4.2 confirming the inert gas system is satisfactory (IGF Code ch.6);
- (PI) 5.1.4.3 confirming the ventilation arrangements are satisfactory (IGF Code chs.12 and 13);

⁵⁰ Refer to the *Unified interpretation of SOLAS regulation II-1/3-5* (MSC.1/Circ.1379 and MSC.1/Circ.1426/Rev.1).

⁵¹ Refer to the Guidelines to facilitate the selection of portable atmosphere testing instruments for enclosed spaces as required by SOLAS regulation XI-1/7 (MSC.1/Circ.1477).

⁵² Refer to the Unified interpretations of SOLAS regulation XIV/2.2 and paragraphs 1.3.2 and 1.3.6, part I-A of the Polar Code (MSC.1/Circ.1562)

(PI)	5.1.4.4	confirming that the fuel containment systems are arranged and installed in accordance with the approved plans, internally examining the fuel containments and ensuring that the appropriate testing is carried out (IGF Code chs.6 and 16);
(PI)	5.1.4.5	examining the electrical installations with particular reference to the certified safe type equipment fitted in gas-dangerous spaces and zones (IGF Code chs.12 and 14);
(PI)	5.1.4.6	examining the arrangements for the fire protection and fire extinction (IGF Code ch.11);
(PI)	5.1.4.7	examining the fire pump capacity and working pressure in relation to the water spray system, if the water spray system is part of the fire main system (IGF Code para.11.4.1);
(PI)	5.1.4.8	examining the isolating valves of the fire main, when the fuel storage tank or tanks are located on the open deck (IGF Code para.11.4.2);
(PI)	5.1.4.9	examining the water spray system arrangement for fuel storage tanks(s) on open deck including remote operation (IGF Code para.11.5);
(PI)	5.1.4.10	examining the fixed dry chemical powder fire-extinguishing system for the bunkering station area (IGF Code para.11.6.1);
(PI)	5.1.4.11	examining the portable dry powder extinguisher (IGF Code para.11.6.2);
(PI)	5.1.4.12	examining the fixed fire detection and alarm system (IGF Code section 11.7);
(PI)	5.1.4.13	examining the machinery installations (IGF Code ch.10);
(PI)	5.1.4.13.1	ventilation systems;
(PI)	5.1.4.13.2	dual-fuel engines;
(PI)	5.1.4.13.3	gas-only engines;
(PI)	5.1.4.13.4	multi-fuel engines;
(PI)	5.1.4.13.5	main and auxiliary boilers;
PI)	5.1.4.13.6	gas turbines.
(PI)	5.1.5	For the hull, machinery and equipment of passenger ships the check that the required documentation has been placed on board should consist of:
(PI)	5.1.5.1	confirming that the stability information and damage control plans and damage control booklets have been provided (SOLAS 74/88 regs.II-1/22 and 23) (SOLAS 74/06/17 regs.II-1/5-1 and 19);
(PI)	5.1.5.2	checking, where applicable, that the noise survey report as required by the Code on Noise Levels on Board Ships is available on board (SOLAS 74/12/16 reg.II-1/3-12);

- (PI) 5.1.5.3 checking the provision of a ship-specific emergency towing procedure (SOLAS 74/08 reg.II-1/3-4);
- (PI) 5.1.5.4 confirming that the manoeuvring booklet has been provided and that the manoeuvring information has been displayed on the navigating bridge (SOLAS 74/88 reg.II-1/28);
- (PI) 5.1.5.5 confirming that documented operating procedures for closing and securing the openings in special category spaces and ro-ro spaces are available on board (SOLAS 74/06/17 reg.II-1/23);
- (PI) 5.1.5.6 where applicable, confirming that the approved Cargo Securing Manual for ships carrying cargoes other than solid and liquid bulk cargoes, cargo units and cargo transport units is on board (SOLAS 74/98/02 reg.VI/5.6);
- (PI) 5.1.5.7 confirming that, where applicable, the approved documentation for the alternative design and arrangements is on board (SOLAS 74/00/06/15 regs.II-1/55, II-2/17 and III/38 and IGF Code ch.2);
- (PI) 5.1.5.8 confirming that the fire control plans are permanently exhibited or, alternatively, that emergency booklets have been provided to each officer and a duplicate of the plans or the emergency booklet are available in a prominently marked enclosure external to the ship's deckhouse (SOLAS 74/00 regs.II-2/15.2.4 and 15.3.2) (SOLAS 74/88 reg.II-2/20); and that the fire control plan is in the language required by the Administration;
- (PI) 5.1.5.9 confirming that the maintenance plans have been provided (SOLAS 74/88 regs.II-2/14.2.2 and 14.3);
- (PI) 5.1.5.10 confirming that the training manuals and the fire safety operational booklets have been provided (SOLAS 74/88 regs.II-2/15.2.3 and 16.2);
- (PI) 5.1.5.11 confirming, when appropriate, that the ship is provided with a document indicating compliance with the special requirements for carrying dangerous goods (SOLAS 74/00/08 reg.II-2/19.4) (SOLAS 74/88 regs.II-2/41 and 54.3);
- (PI) 5.1.5.12 confirming that emergency instructions are available for each person on board, that the muster list is posted in conspicuous places, and that they are in a language understood by the persons on board (SOLAS 74/00/17 regs.III/8 and 37);
- (PI) 5.1.5.13 confirming that ship-specific plans and procedures for recovery of persons from the water have been provided (SOLAS 74/12 reg.III/17-1);
- (PI) 5.1.5.14 confirming that the training manual for the life-saving appliances has been provided and is available in the working language of the ship (SOLAS 74/00/04 reg.III/35);
- (PI) 5.1.5.15 confirming that the checklist and instructions for MES, where appropriate, and onboard maintenance of the life-saving appliances have been provided (SOLAS 74/00 reg.III/36);

(PI)	5.1.5.16	confirming that a table or curve of residual deviations for the magnetic compass have been provided, and that a diagram of the radar installations shadow sectors is displayed (SOLAS 74/00 reg.V/19);
(PI)	5.1.5.17	checking that operational and, where appropriate, maintenance manuals for all navigational equipment are provided (SOLAS 74/00 regs.V/16 and 19);
(PI)	5.1.5.18	checking that the charts and nautical publications necessary for the intended voyage are available and have been updated (SOLAS 74/00 rates)//40 and 27):
(PI)	5.1.5.19	checking that the International Code of Signals is available where the ship is required to carry a radio installation (SOLAS 74/88 reg.V/21);
(PI)	5.1.5.20	checking that records are provided identifying any pilot ladders placed into service (SOLAS 74/10 reg.V/23.2.4);
(PI)	5.1.5.21	confirming that a list showing the operational limitations imposed on the ship is kept on board (SOLAS 74/00 reg.V/30);
(PI)	5.1.5.22	checking that an illustrated table describing the life-saving signals to be used by ships, aircraft or persons in distress is available (SOLAS 74/00 reg.V/29);
(PI)	5.1.5.23	checking the carriage of operating manuals for all equipment (SOLAS 74/88 reg.IV/15);
(PI)	5.1.5.24	checking the carriage of service manuals for all equipment when at-sea maintenance is the declared option (SOLAS 74/88 reg.IV/15);
(PI)	5.1.5.25	checking for a valid radio licence issued by the flag Administration (ITU RR Article 24);
(PI)	5.1.5.26	checking the radio operators' certificates of competence (ITU RR Article 55);
(PI)	5.1.5.27	checking the emission on operational frequencies, coding and registration on the 406 MHz signal without transmission of a distress call to the satellite;
(PI)	5.1.5.28	checking the radio log (SOLAS 74/88 reg.IV/19 in force prior to 1 February 1992 and ITU RR App.11);
(PI)	5.1.5.29	checking the carriage of up-to-date ITU publications (ITU RR App.11);
(PI)	5.1.5.30	checking that the EPIRB has been subject to maintenance at intervals not exceeding five years at an approved shore-based maintenance facility;
(PI)	5.1.5.31	if possible, checking the emission on operational frequencies, coding and registration on the 121.5 MHz homing signal without transmission of the distress call to satellite system;

- (PI) 5.1.5.32 confirming that a continuous synopsis record is provided (SOLAS 74/02 reg.XI-1/5); and
- (PI) 5.1.5.33 checking that arrangements are provided to maintain records of navigational activities and daily reporting (SOLAS 74/00/03 reg.V/28).
- (PI) 5.1.6 For the hull, machinery and equipment of passenger ships the completion of the initial survey should consist of:
- (PI) 5.1.6.1 after a satisfactory survey, issuing the Passenger Ship Safety Certificate and its associated Record of Equipment (Form P).
- (PR) **5.2 Renewal surveys** see part "General", section 4.5
- (PR) 5.2.1 For the hull, machinery and equipment of passenger ships the examination of current certificates and other records should consist of:
- (PR) 5.2.1.1 checking the validity of the International Load Line Certificate or International Load Line Exemption Certificate;
- (PR) 5.2.1.2 checking, as appropriate, the validity of the Polar Ship Certificate;
- (PR) 5.2.1.3 checking the validity of the Safety Management Certificate (SMC) and that a copy of the Document of Compliance (DOC) is on board;
- (PR) 5.2.1.4 checking the validity of the International Ship Security Certificate;
- (PR) 5.2.1.5 checking the validity of the International Oil Pollution Prevention Certificate;
- (PR) 5.2.1.6 checking the certificates of class, if the ship is classed with a classification society;
- (PR) 5.2.1.7 checking, when appropriate, the validity of the International Pollution Prevention Certificate for the Carriage of Noxious Liquid Substances in Bulk;
- (PR) 5.2.1.8 checking, when appropriate, the validity of the International Sewage Pollution Prevention Certificate;
- (PR) 5.2.1.9 checking, when appropriate, the validity of the International Air Pollution Prevention Certificate;
- (PR) 5.2.1.10 confirming, when appropriate, the validity of the International Energy Efficiency Certificate (MARPOL Annex VI, regs.6.4 and 6.5);
- (PR) 5.2.1.11 confirming, when appropriate, that confirmation of compliance for the SEEMP part II is provided to and retained on board the ship (MARPOL Annex VI, reg. 5.4.5);⁵³

⁵³ Refer to the Sample Format of Confirmation of Compliance, Early Submission of the SEEMP part II on the ship fuel oil consumption data collection plan and its timely verification pursuant to regulation 5.4.5 of MARPOL Annex VI (MEPC.1/Circ.876).

- (PR) 5.2.1.12 confirming, when appropriate, the validity of the Statements of Compliance related to fuel oil consumption reporting (MARPOL Annex VI, regs.6.6 and 6.7);
- (PR) 5.2.1.13 checking, when appropriate, the validity of the International Ballast Water Management Certificate;
- (PR) 5.2.1.14 checking that the ship's complement complies with the Minimum Safe Manning Document (SOLAS 74/00/12 reg.V/14) (SOLAS 74/88 reg.V/13(b));
- (PR) 5.2.1.15 checking that the master, officers and ratings are certificated as required by the STCW Convention;
- (PR) 5.2.1.16 checking, where applicable, that the noise survey report as required by the Code on Noise Levels on Board Ships is available on board (SOLAS 74/12/16 reg.II-1/3-12);
- (PR) 5.2.1.17 confirming that, where applicable, the approved documentation for the alternative design and arrangements is on board (SOLAS 74/00/06/15 regs.II-1/55, II-2/17 and III/38 and IGF Code ch.2);
- (PR) 5.2.1.18 checking whether any new equipment has been fitted and, if so, confirming that it has been approved before installation and that any changes are reflected in the appropriate certificate;
- (PR) 5.2.1.19 checking that the routine surveys of the boilers and other pressure vessels, as determined by the Administration, have been carried out as required and that safety devices, such as the boiler safety valves, have been tested;
- (PR) 5.2.1.20 checking that, as appropriate, the hull and machinery have been presented for survey in accordance with the continuous survey scheme approved by the Administration or a classification society;
- (PR) 5.2.1.21 confirming that the opening and the closing and locking of side scuttles positioned below the margin line or the bulkhead deck, as applicable, are being recorded in the logbook (SOLAS 74/88 reg.II-1/17), (SOLAS 74/06/17 reg.II-1/15);
- (PR) 5.2.1.22 confirming that the closure of the cargo loading doors and the opening and closing of any doors at sea required for the operation of the ship or the embarking and disembarking of passengers are being recorded in the logbook (SOLAS 74/88 reg.II-1/20-1) (SOLAS 74/06/17 reg.II-1/22);
- (PR) 5.2.1.23 confirming that the stability information and damage control plans and damage control booklets are readily available (SOLAS 74/88 regs.II-1/22 and 23) (SOLAS 74/06/17 regs.II-1/5-1 and 19);
- (PR) 5.2.1.24 confirming from the logbook entries that the openings required to be closed at sea are being kept closed and that the required operational tests and inspections of watertight doors, etc., are being carried out (SOLAS 74/88 regs.II-1/24 and 25) (SOLAS 74/06/17 regs.II-1/21 and 22);

- (PR) 5.2.1.25 confirming that documented operating procedures for closing and securing the openings in special category spaces and ro-ro spaces are available on board (SOLAS 74/06/17 reg.II-1/23);
- (PR) 5.2.1.26 confirming that the manoeuvring booklet is readily available and that the manoeuvring information is displayed on the navigating bridge (SOLAS 74/88 reg.II-1/28);
- (PR) 5.2.1.27 confirming that the fire control plans are permanently exhibited or, alternatively, that emergency booklets have been provided and a duplicate of the plans or the emergency booklet is available in a prominently marked enclosure external to the ship's deckhouse (SOLAS 74/88 reg.II-2/20);
- (PR) 5.2.1.28 confirming that the maintenance plans have been provided (SOLAS 74/00 regs.II-2/14.2.2 and 14.3);
- (PR) 5.2.1.29 confirming that the training manuals and the fire safety operational booklets have been provided (SOLAS 74/00 regs.II-2/15.2.3 and 16.2);
- (PR) 5.2.1.30 checking whether any fire has occurred on board necessitating the operation of the fixed fire-extinguishing systems or the portable fire extinguishers since the last survey and the entries into the ship's logbook;
- (PR) 5.2.1.31 checking, when appropriate, that the ship is provided with a document indicating compliance with the special requirements for carrying dangerous goods (SOLAS 74/00/08 reg.II-2/19.4) (SOLAS 74/88 reg.II-2/54.3);
- (PR) 5.2.1.32 confirming that ship-specific plans and procedures for recovery of persons from the water have been provided (SOLAS 74/12 reg.III/17-1);
- (PR) 5.2.1.33 confirming, when appropriate, that there is a special list, manifest or stowage plan for the carriage of dangerous goods (SOLAS 74/88 reg.VII/5);
- (PR) 5.2.1.34 confirming that emergency instructions are available for each person on board, that the muster list is posted in conspicuous places, and that they are in a language understood by the persons on board (SOLAS 74/00/17 regs.III/8 and 37);
- (PR) 5.2.1.35 confirming that, if applicable, a factual statement issued by the manufacturer of the lifeboat release mechanism is available, confirming the successful overhaul examination of a mechanism compliant with paragraphs 4.4.7.6.4 to 4.4.7.6.6 of the LSA Code, or, alternatively, that a statement of acceptance of the installation of a replacement release and retrieval system to an existing lifeboat is available (SOLAS 74/11 reg.III/1.5; LSA Code section 4.4.7.6);
- (PR) 5.2.1.36 checking that logbook entries are being made (SOLAS 74/00/13 regs.III/19 and 20) (SOLAS 74/17 regs.II-1/19-1), in particular:

(PR) 5.2.1.36.1 the date when the last full muster of the passengers and crew for boat and fire drill took place, the date when the last enclosed space entry and

rescue drills took place, and the date when the last damage control drills took place;

- (PR) 5.2.1.36.2 the records indicating that on voyages where passengers are scheduled to be on board for more than 24 h, musters of newly embarked passengers have taken place prior to or immediately upon departure;
- (PR) 5.2.1.36.3 the records indicating that the lifeboat equipment was examined at that time and found to be complete;
- (PR) 5.2.1.36.4 the last occasion when the lifeboats were swung out and when each one was lowered into the water;
- (PR) 5.2.1.36.5 the records indicating that crew members have received the appropriate onboard training;
- (PR) 5.2.1.37 confirming that the training manual and training aids for the life-saving appliances are available on board in the working language of the ship (SOLAS 74/00/04 reg.III/35);
- (PR) 5.2.1.38 confirming that the instructions for onboard maintenance of the life-saving appliances is on board (SOLAS 74/00 reg.III/36);
- (PR) 5.2.1.39 checking by the logbook entries that the testing and the emergency drills of the steering gear have been carried out (SOLAS 74/00 reg.V/26);
- (PR) 5.2.1.40 confirming that a table or curve of residual deviations for the magnetic compass is available and that a diagram of the radar installations shadow sectors is displayed (SOLAS 74/00 reg.V/19);
- (PR) 5.2.1.41 checking that operational and, where appropriate, maintenance manuals for all navigational equipment are provided (SOLAS 74/00 reg.V/16);
- (PR) 5.2.1.42 checking that the charts and nautical publications necessary for the intended voyage are available and have been updated (SOLAS 74/00 reg.V/27);
- (PR) 5.2.1.43 checking that the compass deviation book is properly maintained (SOLAS 74/00 reg.V/19);
- (PR) 5.2.1.44 checking that records are maintained identifying any pilot ladders placed into service and any repair effected (SOLAS 74/10 reg.V/23.2.4);
- (PR) 5.2.1.45 confirming that a list showing the operational limitations imposed on the ship is kept on board (SOLAS 74/00 reg.V/30);
- (PR) 5.2.1.46 checking that an illustrated table describing the life-saving signals to be used by ships, aircraft or persons in distress is available (SOLAS 74/00 reg.V/29);
- (PR) 5.2.1.47 confirming the provisions of (PI) 5.1.5.17 to (PI) 5.1.5.23 except (PI) 5.1.5.20;
| (PR) | 5.2.1.48 | confirming that a record has been kept in the period since the last survey |
|------|----------|--|
| | | to the satisfaction of the Administration and as required by the Radio |
| | | Regulations (SOLAS 74/88 reg.IV/17); |

- (PR) 5.2.1.49 checking documentary evidence that the actual capacity of the battery has been proved in port within the last 12 months (SOLAS 74/88 reg.IV/13);
- (PR) 5.2.1.50 if applicable, checking that a list of all limitations on the operation of a passenger ship is kept on board and updated;
- (PR) 5.2.1.51 where applicable, confirming that the approved Cargo Securing Manual for ships carrying cargoes other than solid and liquid bulk cargoes, cargo units and cargo transport units is on board (SOLAS 74/98/02 reg.VI/5.6);
- (PR) 5.2.1.52 confirming that a continuous synopsis record is provided (SOLAS 74/02 reg.XI-1/5);
- (PR) 5.2.1.53 checking that the annual test has been carried out for the Satellite EPIRB and, if applicable, that shore-based maintenance has been carried out at intervals not exceeding five years (SOLAS 74/04 reg.IV/15);
- (PR) 5.2.1.54 checking that arrangements are provided to maintain records of navigational activities and daily reporting (SOLAS 74/00/03 reg.V/28); and
- (PR) 5.2.1.55 confirming the availability of the International Anti-fouling System Certificate (AFS 2001 annex 4 reg.2), when applicable.
- (PR) 5.2.2 For the hull, machinery and equipment of passenger ships the renewal survey should consist of:
- (PR) 5.2.2.1 examining the outside of the ship's bottom, including the bottom and bow plating, keel, bilge keels, stem, stern frame, rudder, sea chests and strainers, noting the clearance measured in the rudder bearings, examining the propeller and shaft seals, as far as practicable, and noting the clearance measured in the propeller shafts (SOLAS 74/88 reg.l/7(b)(ii));
- (PR) 5.2.2.2 examining the arrangements for subdivision, including the ship's stability in the damaged condition, and checking the subdivision load lines (SOLAS 74/88 regs.II-1/4 to 8, 13 and 16) (SOLAS 74/06/12/17/18 regs.II-1/8, 8-1, 14 and 18);
- (PR) 5.2.2.3 confirming the provision of operational information to the master for safe return to port after a flooding casualty by onboard stability computer or shore-based support (SOLAS 74/12/17/18 reg.II-1/8-1);
- (PR) 5.2.2.4 checking the ballasting arrangements (SOLAS 74/88 reg.II-1/9) (SOLAS 74/06/17 reg.II-1/20);
- (PR) 5.2.2.5 confirming that dedicated seawater ballast tanks have been coated in accordance with resolution MSC.215(82), as amended, when appropriate (SOLAS 74/00/06 reg.II-1/3-2);

- (PR) 5.2.2.6 confirming when appropriate that the maintenance of the protective coating is included in the overall ship's maintenance system (SOLAS 74/00/06 reg.II-1/3-2);
- (PR) 5.2.2.7 examining the collision and other watertight bulkheads required for the ship's subdivision (SOLAS 74/88 regs.II-1/10, 14, 15 and 18) (SOLAS 74/06/17 regs.II-1/10, 11, 12, 13 and 16);
- (PR) 5.2.2.8 confirming that the watertight integrity has been maintained where pipes, scuppers, etc. pass through subdivision watertight bulkheads (SOLAS 74/88 reg.II-1/15) (SOLAS 74/06/17 reg.II-1/13);
- (PR) 5.2.2.9 confirming that a diagram is provided on the navigating bridge showing the location of the watertight doors together with indicators showing whether the doors are open or closed (SOLAS 74/88 reg.II-1/15) (SOLAS 74/06/17 reg.II-1/13);
- (PR) 5.2.2.10 testing the operation of the watertight doors both from the navigating bridge in the event of an emergency and locally at the door itself (SOLAS 74/88 reg.II-1/15) (SOLAS 74/06/17 reg.II-1/13) and, in particular, that they are:
- (PR) 5.2.2.10.1 operable locally from each side of the bulkhead;
- (PR) 5.2.2.10.2 provided with devices giving an indication of whether the door is open or closed at all remote operating positions;
- (PR) 5.2.2.10.3 provided with an audible alarm that is distinct from any other alarm in the area and, when appropriate, an intermittent visual signal;
- (PR) 5.2.2.10.4 provided with control handles on each side of the bulkhead so that a person may hold both handles in the open position and pass safely through the watertight door without accidentally setting the power closing mechanism into operation;
- (PR) 5.2.2.11 testing the remote hand-operation to close the power-operated sliding watertight door from an accessible position above the bulkhead deck (SOLAS 74/88/14 reg.II-1/15) (SOLAS 74/06/17 reg.II-1/13);
- (PR) 5.2.2.12 confirming that the watertight doors and their indicating devices are operable in the event of a failure of the main and emergency sources of power (SOLAS 74/88 reg.II-1/15) (SOLAS 74/06/17 reg.II-1/13);
- (PR) 5.2.2.13 checking, when appropriate, any watertight doors that are not required to be closed remotely, fitted in watertight bulkheads dividing 'tween deck spaces, and confirming that a notice is affixed concerning their closure (SOLAS 74/88 reg.II-1/15) (SOLAS 74/06/17 reg.II-1/13);
- (PR) 5.2.2.14 confirming that a notice is affixed to any portable plates on bulkheads in machinery spaces concerning their closure and, if appropriate, testing any power-operated watertight door fitted in lieu (SOLAS 74/88 reg.II-1/15) (SOLAS 74/06/17 reg.II-1/13);
- (PR) 5.2.2.15 examining the arrangements for closing side scuttles and their deadlights, also scuppers, sanitary discharges and similar openings and other inlets and discharges in the shell plating below the margin line (SOLAS 74/88 reg.II-1/17);

- (PR) 5.2.2.16 examining the arrangements for closing side scuttles and their deadlights, also scuppers, sanitary discharges and similar openings and other inlets and discharges in the shell plating below the bulkhead deck (SOLAS 74/06/17 reg.II-1/15);
- (PR) 5.2.2.17 confirming that valves for closing the main and auxiliary sea inlets and discharges in the machinery spaces are readily accessible and indicators showing the status of the valves are provided (SOLAS 74/88 reg.II-1/17) (SOLAS 74/06/17 reg.II-1/15);
- (PR) 5.2.2.18 confirming that gangway, cargo and coaling ports fitted below the margin line may be effectively closed and that the inboard ends of any ash or rubbish chutes are fitted with an effective cover (SOLAS 74/88 reg.II-1/17);
- (PR) 5.2.2.19 confirming that gangway, cargo and fuelling ports fitted below the bulkhead deck may be effectively closed and that the inboard ends of any ash or rubbish chutes are fitted with an effective cover (SOLAS 74/06/17 reg.II-1/15);
- (PR) 5.2.2.20 examining the arrangements to maintain the watertight integrity above the margin line or the bulkhead deck as applicable (SOLAS 74/88 reg.II-1/20) (SOLAS 74/06/17 reg.II-1/17);
- (PR) 5.2.2.21 examining the arrangements for the bilge pumping and confirming that each bilge pump and pumping system provided for each watertight compartment is working efficiently (SOLAS 74/88 reg.II-1/21) (SOLAS 74/05/17 reg.II-1/35-1);
- (PR) 5.2.2.22 confirming that the drainage system of enclosed cargo spaces situated on the freeboard deck is working efficiently (SOLAS 74/88 reg.II-1/21) (SOLAS 74/05/17 reg.II-1/35-1);
- (PR) 5.2.2.23 examining visually the drainage facilities for blockage or other damage and confirming the provision of means to prevent blockage of drainage arrangements, for closed vehicle and ro-ro spaces and special category spaces where fixed pressure water-spraying systems are used (SOLAS 74/08 reg.II-2/20.6.1.5);
- (PR) 5.2.2.24 examining, when appropriate, the means of indicating the status of any bow doors and any leakage therefrom (SOLAS 74/88 reg.II-1/23-2);
- (PR) 5.2.2.25 confirming, that the arrangement for monitoring special category spaces or ro-ro spaces, when fitted, is satisfactory (SOLAS 74/06/17 reg.II-1/23);
- (PR) 5.2.2.26 confirming that the machinery, boilers and other pressure vessels, associated piping systems and fittings are being maintained so as to reduce to a minimum any danger to persons on board, due regard being given to moving parts, hot surfaces and other hazards (SOLAS 74/88 reg.II-1/26);
- (PR) 5.2.2.27 confirming that normal operation of the propulsion machinery can be sustained or restored even though one of the essential auxiliaries becomes inoperative (SOLAS 74/88 reg.II-1/26);

- (PR) 5.2.2.28 confirming that means are provided so that the machinery can be brought into operation from the dead ship condition without external aid (SOLAS 74/88 reg.II-1/26);
- (PR) 5.2.2.29 examining, where practicable, the means provided to protect against overpressure in the parts of main, auxiliary and other machinery that is subject to internal pressure and may be subject to dangerous overpressure (SOLAS 74/88 reg.II-1/27);
- (PR) 5.2.2.30 examining, when appropriate, the crankcase explosion relief devices fitted to internal combustion engines and confirming that they are arranged so as to minimize the possibility of injury to personnel (SOLAS 74/88 reg.II-1/27);
- (PR) 5.2.2.31 confirming that the automatic shut-off arrangements fitted to the main turbine propulsion machinery and, where applicable, main internal combustion propulsion machinery and auxiliary machinery are being properly maintained (SOLAS 74/88 reg.II-1/27);
- (PR) 5.2.2.32 confirming, as far as practicable, the ability of the machinery to reverse the direction of the thrust of the propeller in sufficient time, including the effectiveness of any supplementary means of manoeuvring or stopping the ship (SOLAS 74/88 reg.II-1/28);
- (PR) 5.2.2.33 confirming that the main and auxiliary steering gear are being properly maintained, are arranged so that the failure of one does not render the other inoperative and that the auxiliary steering gear is capable of being brought speedily into action in an emergency⁵⁴ (SOLAS 74/88/14 reg.II-1/29);
- (PR) 5.2.2.34 confirming that, where appropriate, essential components of the steering gear are permanently lubricated or provided with lubrication fittings (SOLAS 74/88/14 reg.II-1/29);
- (PR) 5.2.2.35 confirming that relief valves fitted to the steering gear hydraulic system which can be isolated and in which pressure can be generated from the power source or from external forces are being maintained and are set to a pressure not exceeding the design pressure (SOLAS 74/88/14 reg.II-1/29);
- (PR) 5.2.2.36 confirming that the main or auxiliary steering gear power units restart automatically when power is restored after a power failure, that they are capable of being brought into operation from a position on the navigating bridge and that, in the event of a power failure to any one of the steering gear power units, an audible and visual alarm is given on the navigating bridge (SOLAS 74/88/14 reg.II-1/29);
- (PR) 5.2.2.37 confirming that the control systems for the main steering gear from both the navigating bridge and the steering gear compartment are operating satisfactorily (SOLAS 74/88/14 reg.II-1/29);

⁵⁴ For ships fitted with alternative propulsion and steering arrangements other than traditional arrangements, such as but not limited to azimuthing propulsors or water jet propulsion systems, refer to the *Unified interpretations of SOLAS regulations II-1/28, II-1/29 and II-1/30* (MSC.1/Circ.1416/Rev.1).

- (PR) 5.2.2.38 confirming that, where the main steering gear comprises two or more identical power units and an auxiliary steering gear is not fitted, the two independent control systems from the navigating bridge are operating satisfactorily (SOLAS 74/88/14 reg.II-1/29);
- (PR) 5.2.2.39 confirming that the control system for the auxiliary steering gear, in the steering gear compartment and, if this gear is power-operated, from the navigating bridge, are operating satisfactorily and that the latter is independent of the control system for the main steering gear (SOLAS 74/88/14 reg.II-1/29);
- (PR) 5.2.2.40 confirming that an audible and visual alarm is given on the navigating bridge in the event of a failure of electrical power supply (SOLAS 74/88/14 reg.II-1/29);
- (PR) 5.2.2.41 confirming that the means of communication between the bridge and the steering gear is operating satisfactorily and that, with ships having emergency steering positions, a telephone or other means of communication for relaying heading information and supplying visual compass readings to the emergency steering position is provided (SOLAS 74/14 regs.II-1/29 and SOLAS 74/00 reg.V/19);
- (PR) 5.2.2.42 confirming that the angular position of the rudder is indicated independently of the steering control system on the navigating bridge if the main steering gear is power-operated and that this angular position is given in the steering gear compartment (SOLAS 74/14 reg.II 1/29 and SOLAS 74/00 reg.II-1/29 and reg.V/19);
- (PR) 5.2.2.43 confirming that with a hydraulic power-operated steering gear the audible and visual low-level alarms on the navigating bridge and in the machinery space for each hydraulic fluid reservoir are operating satisfactorily and that at least one power-actuating system including the reservoir can be recharged from a position within the steering gear compartment by means of a fixed storage tank to which a contents gauge is fitted with fixed piping (SOLAS 74/88/14 reg.II-1/29);
- (PR) 5.2.2.44 confirming that the steering gear compartment is readily accessible and is provided with suitable arrangements to ensure working access to steering gear machinery and controls under safe conditions (SOLAS 74/88/14 reg.II-1/29);
- (PR) 5.2.2.45 confirming that, with electric and electro-hydraulic steering gear, the means for indicating on the navigating bridge and at a main machinery control position that the motors are running and, as far as practicable, that the overload alarm and alarm for the loss of a phase in a three-phase supply located at the main machinery control position are operating satisfactorily (SOLAS 74/88 reg.II-1/30);
- (PR) 5.2.2.46 confirming that the effective means of operation and control of the main and auxiliary machinery essential for the propulsion and the safety of the ship are being maintained, including, when appropriate, any means for remotely controlling the propulsion machinery from the navigating bridge (including the control, monitoring, reporting, alert and safety actions) (SOLAS 74/88/00/02 reg.II-1/31);

(PR)	5.2.2.47	confirming that arrangements for operating main and other machinery from a machinery control room are satisfactory (SOLAS 74/88 reg.II-1/31);
(PR)	5.2.2.48	confirming that the means provided for manually overriding automatic controls are being maintained and that a failure does not prevent the use of the manual override (SOLAS 74/88 reg.II-1/31);
(PR)	5.2.2.49	confirming that the appropriate safety features fitted to the oil-fired and exhaust gas boilers, unfired steam generators, steam pipe systems and air pressure systems are being maintained (SOLAS 74/88 regs.II-I/32, 33 and 34);
(PR)	5.2.2.50	confirming the operation of the ventilation for the machinery spaces (SOLAS 74/88 reg.II-I/35);
(PR)	5.2.2.51	when appropriate, confirming that the measures to prevent noise in machinery spaces are effective (SOLAS 74/88 reg.II-I/36 and SOLAS 74/12/16 reg.II-1/3-12.2); or confirming that the ship was constructed to reduce onboard noise and to protect personnel from noise in accordance with the Code on Noise Levels on Board Ships, adopted by resolution MSC.337(91), as amended (SOLAS 74/12/16 reg.II-1/3-12);
(PR)	5.2.2.52	confirming that the engine-room telegraph giving visual indication of the orders and answers both in the machinery space and on the navigation bridge is operating satisfactorily (SOLAS 74/88 reg.II-1/37);
(PR)	5.2.2.53	confirming that the second means of communication between the navigation bridge and machinery space is also operating satisfactorily, including any appropriate means provided to any other positions from which the engines are controlled (SOLAS 74/88 reg.II-1/37);
(PR)	5.2.2.54	confirming that the engineer's alarm is clearly audible in the engineers' accommodation (SOLAS 74/88 reg.II-1/38);
(PR)	5.2.2.55	confirming that precautions taken to prevent any oil that may escape under pressure from any pump, filter or heater from coming into contact with heated surfaces are efficient;
(PR)	5.2.2.56	confirming that the means of ascertaining the amount of oil contained in any oil tank are in satisfactory working condition (SOLAS 74/88 reg.II-2/15) (SOLAS 74/00 reg.II-2/4.2.2.3.5);
(PR)	5.2.2.57	confirming that the devices provided to prevent overpressure in any oil tank or in any part of the oil system, including the filling pipes, are in satisfactory working condition (SOLAS 74/88 reg.II-2/15) (SOLAS 74/00 reg.II-2/4.2.2.4);
(PR)	5.2.2.58	confirming that the electrical installations, including the main source of power and lighting systems, are being maintained (SOLAS 74/88 regs.II-1/40 and 41);
(PR)	5.2.2.59	confirming that the self-contained emergency source of electrical power and its associated systems are operating satisfactorily (SOLAS 74/88 reg.II-1/42);
(PR)	5.2.2.60	confirming that the starting arrangements of each emergency generating set are satisfactory (SOLAS 74/88 reg.II-1/44);

- (PR) 5.2.2.61 checking, when appropriate, the disposition of and testing the supplementary emergency lighting (SOLAS 74/88 reg.II-1/42-1);
- (PR) 5.2.2.62 for passenger ships constructed on or after 1 July 2010, checking the provision of supplementary lighting in all cabins, and checking that such lighting automatically illuminates and remains on for a minimum of 30 min when power to the normal cabin lighting is lost (SOLAS 74/06 reg.II-1/41.6);
- (PR) 5.2.2.63 confirming that precautions provided against shock, fire and other hazards of electrical origin are being maintained (SOLAS 74/88 reg.II-1/45);
- (PR) 5.2.2.64 confirming, when appropriate, that the arrangements for the machinery spaces being periodically unattended are satisfactory (SOLAS 74/88 reg.II-1/54);
- (PR) 5.2.2.65 examining, where applicable, the alternative design and arrangements for machinery or electrical installations, low-flashpoint fuel storage and distribution systems, fire safety, or life-saving appliances and arrangements, in accordance with the test, inspection and maintenance requirements, if any, specified in the approved documentation (SOLAS 74/00/06/15 regs.II-1/55, II-2/17 and III/38 and IGF Code ch.2);
- (PR) 5.2.2.66 examining the fire pumps and fire main and the disposition of the hydrants, hoses and nozzles and the international shore connection and checking that each fire pump, including the emergency fire pump, can be operated separately so that two jets of water are produced simultaneously from different hydrants at any part of the ship while the required pressure is maintained in the fire main (SOLAS 74/00/14 reg.II-2/10.2; FSS Code chs.2 and 12) (SOLAS 74/88 regs.II-2/4 and 19);
- (PR) 5.2.2.67 for passenger ships designed to carry containers on or above the weather deck, as applicable, examining the water mist lance (SOLAS 74/00/14 reg.II-2/10.7.3);
- (PR) 5.2.2.68 examining the provision and randomly examining the condition of the portable and non-portable fire extinguishers (SOLAS 74/00 reg.II-2/10.3; FSS Code ch.4) (SOLAS 74/88 reg.II-2/6);
- (PR) 5.2.2.69 examining the fixed fire-extinguishing system for machinery, cargo, special category and vehicle spaces and confirming that its means of operation are clearly marked (SOLAS 74/00/12/14 regs.II-2/10.4, 10.5, 10.7.1, 10.7.2 and 20.6.1; FSS Code chs.5 to 7) (SOLAS 74/88 regs.II-2/5, 7, 9, 10 and 53);
- (PR) 5.2.2.70 examining the special arrangements in the machinery spaces and confirming, as far as practicable and as appropriate, the operation of the remote means of control provided for the opening and closing of the skylights, the release of smoke, the closure of the funnel and ventilation openings, the closure of power-operated and other doors, the stopping of ventilation and boiler forced and induced draught fans and the stopping of oil fuel and other pumps that discharge flammable liquids (SOLAS 74/00 regs.II-2/5.2, 8.3 and 9.5) (SOLAS 74/88 reg.II-2/11);
- (PR) 5.2.2.71 checking that fixed carbon dioxide fire-extinguishing systems for the protection of machinery spaces, where applicable, are provided with two

separate controls, one for opening of the gas piping and one for discharging the gas from the storage container, each of them located in a release box clearly identified for the particular space (SOLAS 74/08 reg.II-2/10.4, FSS Code ch.5.2.2.2);

- (PR) 5.2.2.72 examining the fire-extinguishing arrangements in control stations, accommodation and service spaces (SOLAS 74/00 reg.II-2/10.6.1; FSS Code ch.8) (SOLAS 74/88 reg.II-2/36);
- (PR) 5.2.2.73 examining, when applicable, the fire-extinguishing arrangements in cabin balconies (SOLAS 74/00 reg.II-2/10.6.1);
- (PR) 5.2.2.74 examining the provision of fire-extinguishing systems for the spaces containing paint and/or flammable liquids and deep-fat cooking equipment in accommodation and service spaces (SOLAS 74/00 regs.II-2/10.6.3 and 10.6.4; FSS Code chs.5, 6 and 7) (SOLAS 74/88 reg.II-2/15.2.5));
- (PR) 5.2.2.75 examining the arrangements for oil fuel, lubricating oil and other flammable oils and testing the remote closing of valves for oil fuel, lubricating oil and other flammable oils and the operation of the remote means of closing the valves on the tanks that contain oil fuel, lubricating oil and other flammable oils (SOLAS 74/00 reg.II-2/4.2) (SOLAS 74/88 reg.II-2/15);
- (PR) 5.2.2.76 examining and testing, as far as practicable, any fire detection and fire alarm arrangements in machinery spaces, including enclosed spaces containing incinerators, if applicable, accommodation and service spaces and control spaces (SOLAS 74/00/10 reg.II-2/7 (except 7.5.5, 7.6 and 7.9); FSS Code chs.8 and 9) (SOLAS 74/88 regs.II-2/11, 12, 13, 13-1, 14, 36 and 41);
- (PR) 5.2.2.76.1 examining and testing, where applicable, any fire detection and fire alarm arrangements on cabin balconies. (SOLAS 74/00 reg.II-2/7.10);
- (PR) 5.2.2.76.2 for passenger ships constructed on or after 1 July 2010, confirming the smoke detectors in cabins, when activated, are emitting, or causing to emit, an audible alarm within the space where they are located (SOLAS 74/06 regs.II-2/7.5.2 and 7.5.3.1);
- (PR) 5.2.2.76.3 for passenger ships constructed on or after 1 July 2010, confirming detectors and manually operated call points of a fixed fire detection and fire alarm system can be remotely and individually identified (SOLAS 74/06 reg.II-2/7.2.4);
- (PR) 5.2.2.77 confirming that the fire-fighters' outfits including their self-contained compressed air breathing apparatus, and the emergency escape breathing devices (EEBDs) are complete and in good condition and that the cylinders, including the spare cylinders, of the self-contained breathing apparatus, are suitably charged, and that onboard means of recharging breathing apparatus cylinders used during drills or a suitable number of spare cylinders to replace those used are provided, and that two-way portable radiotelephone apparatus of an explosion-proof type or intrinsically safe is provided (SOLAS 74/00/12 regs.II-2/10.10, 13.3.4, 13.4.3 and 15.2.2; FSS Code ch.3) (SOLAS 74/88 reg.II-2/17);

- (PR) 5.2.2.78 checking the operational readiness and maintenance of fire-fighting systems (SOLAS 74/00 reg.II-2/14) (SOLAS 74/88/91 reg.II- 2/21);
- (PR) 5.2.2.79 confirming, as far as practicable, that no changes have been made in the structural fire protection, including the structure, fire integrity, protection of stairways and lifts, cabin balconies, openings in "A" and "B" Class divisions, ventilation systems and windows and side scuttles, and the use of combustible material (SOLAS 74/00/04/12/15/17 regs.II-2/5.2, 5.3, 6, 8.2, 8.5, 9.2.1, 9.2.2, 9.3, 9.4.1, 9.5, 9.6 (except 9.6.3), 9.7 and 11 (except 11.6)) (SOLAS 74/88 regs.II-2/11, 16, 18, 23 to 35 and 37);
- (PR) 5.2.2.80 confirming, as far as practicable, that no changes have been made in the structural fire protection in cargo spaces intended for the carriage of dangerous goods (SOLAS 74/00 regs.II-2/19.3.8 and 19.3.10) (SOLAS 74/88 regs.II-2/4, 54.2.8, 54.2.10 and 54.2.11);
- (PR) 5.2.2.81 examining and testing any manual and automatic fire doors including the means of closing the openings in "A" and "B" Class divisions (SOLAS 74/00/17 reg.II-2/9.4.1) (SOLAS 74/88 regs.II-2/30 and 31);
- (PR) 5.2.2.82 examining and testing the fire dampers of ventilation ducts and the main inlets and outlets of all ventilation systems and checking that the power ventilation is capable of being stopped from outside the space served (SOLAS 74/00/14 regs.II-2/5.2.1 and 9.7) (SOLAS 74/88 regs.II-2/16 and 32);
- (PR) 5.2.2.83 confirming that the stairways and ladders, including the low-location lighting system, arranged to provide a means of escape to the lifeboat and liferaft embarkation deck from all passenger and crew spaces and from those spaces in which the crew is normally employed are being maintained (SOLAS 74/00/16 regs.II-2/13.2, 13.3.1, 13.3.2 and 13.7; FSS Code chs.11 and 13 (except paragraph 3)) (SOLAS 74/88 reg.II-2/28);
- (PR) 5.2.2.84 confirming that the means of escape from any special category spaces and ro-ro spaces are satisfactory (SOLAS 74/00 regs.II-2/13.5 and 13.6) (SOLAS 74/88 reg.II-2/28);
- (PR) 5.2.2.85 confirming that the means of escape from the machinery spaces are satisfactory (SOLAS 74/00/14 reg.II-2/13.4.1) (SOLAS 74/88 reg.II-2/28);
- (PR) 5.2.2.86 examining the fire-extinguishing arrangements, examining and testing the fire detection and alarm systems and the sample extraction smoke detection systems, where applicable in cargo spaces for general cargo and dangerous goods and testing, as far as practicable and as appropriate, the operation of the means for closing the various openings (SOLAS 74/00/14 regs.II-2/7.6, 10.7.1 and 10.7.2; FSS Code chs.5, 9 and 10) (SOLAS 74/88 reg.II-2/39);
- (PR) 5.2.2.87 examining the fire-extinguishing arrangements, examining and testing the fire detection and alarm system and the sample extraction smoke detection system, where applicable, in vehicle, special category and ro-ro spaces, and testing, as far as practicable and as appropriate, the

operation of the means for closing the various openings (SOLAS 74/00/15/17 reg.II-2/20 (except 20.5); FSS Code chs.5, 6, 7, 9 and 10) (SOLAS 74/88 regs.II-2/37, 38 and 38-1);

- (PR) 5.2.2.88 examining and testing, as appropriate and as far as practicable, the crew alarm and the public address system or other effective means of communication (SOLAS 74/00 regs.II-2/7.9 and 12; LSA Code ch.7) (SOLAS 74/88 reg.II-2/40);
- (PR) 5.2.2.89 examining, when appropriate, the special arrangements for carrying dangerous goods, including checking the electrical equipment and wiring, ventilation, the provision of personnel protection clothing and portable appliances, testing any fire detection and alarm system and any sample extraction smoke detection system and testing, as far as practicable, the water supply, bilge pumping and any water spray system (SOLAS 74/00/08 reg.II-2/19 (except 19.3.8, 19.3.10 and 19.4); FSS Code chs.3, 4, 7, 9 and 10) (SOLAS 74/88 regs.II-2/41 and 54);
- (PR) 5.2.2.90 examining and testing, when appropriate, the helicopter facilities, including foam firefighting appliances, when appropriate (SOLAS 74/00/16 regs.II-2/18; FSS Code ch.17) (SOLAS 74/88 reg.II-2/18.8);
- (PR) 5.2.2.91 checking the requirement for passenger ships carrying more than 36 passengers and constructed before 1 October 1994 (SOLAS 74/88/92 regs.II-2/41-1 and 41-2);
- (PR) 5.2.2.92 for passenger ships constructed on or after 1 July 2010 and having a length of 120 m or more or having three or more main vertical zones, checking the designation of safe areas (SOLAS 74/06 reg.II-2/21);
- (PR) 5.2.2.93 for passenger ships constructed on or after 1 July 2010, checking the provision of a safety centre (SOLAS 74/06 reg.II-2/23) and its associated ventilation system (SOLAS 74/06 reg.II-2/8.2);
- (PR) 5.2.2.94 checking that emergency instructions are available for each person on board, the muster list is posted in conspicuous places and there are signs or posters in the vicinity of survival craft and their launching stations (SOLAS 74/96/17 regs.III/8, 9 and 37);
- (PR) 5.2.2.95 checking that the falls used in launching have been periodically inspected and have been renewed in the past five years (SOLAS 74/96/04 reg.III/20);
- (PR) 5.2.2.96 examining each survival craft, including its equipment and, when fitted, the on-load release mechanism and hydrostatic lock, and for inflatable liferafts the hydrostatic release unit and float free arrangements, including the date of servicing or replacement; checking that the hand-flares are not out of date and that the required number of search and rescue locating devices are fitted in liferafts and those liferafts are clearly marked (SOLAS 74/96/00/02/08 regs.III/20, 21, 23, 24 and 26; LSA Code sections 2.3 to 2.5, 3.2 and 4.1 to 4.6);

- (PR) 5.2.2.97 checking that the life-saving appliances are of an international or vivid reddish orange, or a comparably highly visible colour on all parts where this will assist detection at sea (LSA Code section 1.2.2.6);
- (PR) 5.2.2.98 examining the embarkation arrangements and launching appliances for each survival craft; each lifeboat should be lowered to the embarkation position or, if the stowage position is the embarkation position, lowered a short distance and, if practicable, one of the survival craft should be lowered to the water. The operation of the launching appliances for davit-launched liferafts should be demonstrated (SOLAS 74/96/04 regs.III/11, 12, 13, 15, 16, 20, 21 and 23; LSA Code sections 6.1 and 6.2);
- (PR) 5.2.2.99 checking that a thorough examination of launching appliances, including the dynamic testing of the winch brake, and servicing of lifeboat and rescue boat on-load release gear, including fast recuse boat, and davit launched liferaft automatic release hooks, and that a thorough examination and operational test of lifeboat and rescue boat including fast rescue boat, have been carried out in accordance with the Requirement for maintenance, thorough examination, operational testing, overhaul and repair of lifeboats and rescue boats, launching appliances and release gear (resolution MSC.402(96)) (SOLAS 74/00/12/16 reg.III/20.11);
- (PR) 5.2.2.100 checking the rotational deployment of MES (SOLAS 74/88 reg.III/20.8.2; LSA Code section 6.2.2.2);
- (PR) 5.2.2.101 examining each rescue boat, including its equipment; for inflatable rescue boats, confirming that they are stowed in a fully inflated condition (SOLAS 74/88/04 regs.III/14, 17, 21, 26.3 and 34);
- (PR) 5.2.2.102 examining the embarkation and recovery arrangements for each rescue boat (SOLAS 74/88 reg.III/14);
- (PR) 5.2.2.103 checking the arrangements for mustering passengers (SOLAS 74/96 regs.III/11, 24 and 25);
- (PR) 5.2.2.104 confirming that a means of rescue is provided on ro-ro passenger ships (SOLAS 74/00 regs.III/11, 26.4);
- (PR) 5.2.2.105 confirming that a helicopter pick-up area is provided on ro-ro passenger ships (SOLAS 74/00 reg.III/28);
- (PR) 5.2.2.106 confirming that a decision support system is provided for the master (SOLAS 74/88 reg.III/29) (SOLAS 74/06 regs.II-2/21 and 22);
- (PR) 5.2.2.107 testing that the engine of the rescue boat(s) and of each lifeboat, when so fitted, start satisfactorily and operate both ahead and astern;

- (PR) 5.2.2.108 examining and checking the operation of two-way VHF radiotelephone apparatus⁵⁵ and search and rescue locating devices (SOLAS 74/88/08 regs.III/6, IV/7 and 14);
- (PR) 5.2.2.109 examining the line-throwing appliance and checking that its rockets and the ship's distress signals are not out of date (SOLAS 74/96 regs.III/6, 18 and 35; LSA Code sections 3.1 and 7.1);
- (PR) 5.2.2.110 examining and checking the operation of onboard communications equipment and verifying that the general alarm system is audible in accommodation, normal crew working spaces and on open decks (SOLAS 74/96 regs.III/6, 18 and 35; LSA Code sections 3.1 and 7.1);
- (PR) 5.2.2.111 examining the provision, disposition, stowage and condition of the lifebuoys, including those fitted with self-igniting lights, self-activating smoke signals and buoyant lines, lifejackets,⁵⁶ immersion suits, anti-exposure suits and thermal protective aids and that their associated batteries are not out of date (SOLAS 74/88/06 regs.III/7, 21, 22 and 26; LSA Code sections 2.1 to 2.5 and 3.1 to 3.3);
- (PR) 5.2.2.111.1 checking the provision of lifejackets in three sizes (Infant, Child, Adult) and checking that they are marked by either weight or height, or by both weight and height (LSA Code section 2.2.1.1); for passenger ships on voyages less than 24 h, checking that the number of infant lifejackets is equal to at least 2.5% of the number of passengers on board and for passenger ships on voyages 24 h or greater, checking that infant lifejackets are provided for each infant on board (SOLAS 74/06 reg.III/7.2.1);
- (PR) 5.2.2.111.2 checking that immersion suits designed to be worn in conjunction with a lifejacket are suitably marked (LSA Code section 2.3.1);
- (PR) 5.2.2.112 checking the lighting of the muster and embarkation stations and the alleyways, stairways and exits giving access to the muster and embarkation stations, including when supplied from the emergency source of power (SOLAS 74/88 regs.II-1/42 and III/11);
- (PR) 5.2.2.113 checking that the required navigation lights, shapes and sound signalling equipment are in order (COLREG, rules 20 to 24, 27 to 30 and 33);
- (PR) 5.2.2.114 checking the provision and specification of the following navigation equipment as appropriate: daylight signalling lamp, magnetic compass, transmitting heading device, gyro compass, gyro compass repeaters, radar installation(s), electronic plotting aid, automatic tracking aid(s) or automatic radar plotting aid(s), echo-sounding device, speed and distance indicator, rudder angle indicator, propeller rate-of-revolution indicator, variable pitch propeller pitch and operational mode indicator, rate-of-turn indicator, heading or track control system, GNSS receiver,

⁵⁵ Expired primary batteries, which complied with the requirements of paragraph 12.6 of *Revised Recommendation on performance standards for survival craft portable Two-Way VHF radiotelephone apparatus* (resolution MSC.149 (77)), could be used for no other purpose than to examine and check the operation of two-way VHF radiotelephone apparatus.

⁵⁶ SOLAS regulations III/7.2.1.1, 7.2.1.2 and 7.2.1.5 should be considered.

terrestrial radio navigation system and sound reception system, a pelorus or compass bearing device, means for correcting heading and bearings, a BNWAS, as applicable, and ECDIS including backup arrangements, as applicable; items that cannot be checked with the ship in port should be verified from records (SOLAS 74/00/09/13 reg.V/19);

- (PR) 5.2.2.115 checking for the provision, specification operation and annual performance test of the voyage data recorder (SOLAS 74/00/04 reg.V/20);
- (PR) 5.2.2.116 checking that the International Code of Signals and an up-to-date copy of Volume III of the International Aeronautical and Maritime Search and Rescue (IAMSAR) Manual have been provided (SOLAS 74/00/02 reg.V/21);
- PR) 5.2.2.117 checking that a valid conformance test report of the long-range identification and tracking system is available on board, where fitted (SOLAS 74/04 reg.V/19-1);
- (PR) 5.2.2.118 checking the provision and operation of the automatic identification system, where fitted, and whether the annual test has been carried out and a copy of the test report is on board (SOLAS 74/00/04/10 regs.V/18.9 and 19);
- (PR) 5.2.2.119 checking the provision and specification of the pilot ladders and pilot transfer arrangements (SOLAS 74/00/10 reg.V/23);
- (PR) 5.2.2.120 checking the provisions of (PI) 5.1.3.118 to (PI) 5.1.3.143 and (PI) 5.1.3.145 to (PI) 5.1.3.147;
- (PR) 5.2.2.121 confirming that no new materials containing asbestos were installed on board (SOLAS 74/00/05/09 reg.II-1/3-5);⁵⁷
- (PR) 5.2.2.122 checking that the means of embarkation and disembarkation from ships for use in port and in port-related operations, such as gangways and accommodation ladders, are in satisfactory condition, as applicable (SOLAS 74/08 reg.II-1/3-9); and
- (PR) 5.2.2.123 confirming, where applicable, that an appropriate portable atmosphere testing instrument or instruments⁵⁸ is on board, and that suitable means are provided for the calibration of all such instruments;⁵⁹ and checking the appropriateness of the testing and calibration (SOLAS 74/14 reg.XI-1/7).
- (PR) 5.2.3 For the hull, machinery and equipment of passenger ships using natural gas as fuel the additional requirements for the renewal survey should consist of:

⁵⁷ Refer to the *Unified interpretation of SOLAS regulation II-1/3-5* (MSC.1/Circ.1379 and MSC.1/Circ.1426/Rev.1).

⁵⁸ Refer to the *Guidelines to facilitate the selection of portable atmosphere testing instruments for enclosed spaces as required by SOLAS regulation XI-1/7* (MSC.1/Circ.1477).

⁵⁹ Refer to the Unified interpretations of SOLAS regulation XIV/2.2 and paragraphs 1.3.2 and 1.3.6, part I-A of the Polar Code (MSC.1/Circ.1562).

- (PR) 5.2.3.1 examining the logbooks and operating records with regard to correct functioning of the gas detection systems, fuel supply/gas systems, etc. (IGF Code ch.16);
- (PR) 5.2.3.2 confirming the manufacturer/builder instructions and manuals covering the operations, safety and maintenance requirements and occupational health hazards relevant to fuel storage, fuel bunkering, and fuel supply and associated systems for the use of the fuel, are provided on board the vessel (IGF Code chs.6 and 18);
- (PR) 5.2.3.3 confirming gas detection and other leakage detection equipment in compartments containing fuel storage, fuel bunkering, and fuel supply equipment or components or associated systems, including indicators and alarms, is in satisfactory operating condition (IGF Code chs.6 and 15);
- (PR) 5.2.3.4 confirming the satisfactory operation of the control, monitoring and automatic shutdown systems of the fuel supply and bunkering systems (IGF Code ch.15);
- (PR) 5.2.3.5 confirming the availability of test and calibration records of the gas detection systems (IGF Code ch.15);
- (PR) 5.2.3.6 examining piping, hoses, emergency shutdown valves, remote operating valves, relief valves, means for inerting, machinery and equipment for fuel storage, fuel bunkering, and fuel supply such as venting, compressing, refrigerating, liquefying, heating, cooling or otherwise handling the fuel (IGF Code chs.5, 6, 8, 9, 10 and 15);
- (PR) 5.2.3.7 testing the shutdown of ESD protected machinery spaces operationally, as far as practicable (IGF Code ch.5);
- (PR) 5.2.3.8 confirming stopping of pumps and compressors upon emergency shutdown of the system (IGF Code chs.6, 10 and 15);
- (PR) 5.2.3.9 examining the ventilation system, including portable ventilating equipment where fitted, for spaces containing fuel storage, fuel bunkering, and fuel supply units or components or associated systems; and including air locks, pump-rooms, compressor rooms, fuel preparation rooms, fuel valve rooms, control rooms and spaces containing gas burning equipment (IGF Code chs.12 and 13);
- (PR) 5.2.3.10 testing, as far as practicable, alarms, such as differential pressure and loss of pressure alarms (IGF Code ch.15);
- (PR) 5.2.3.11 examining portable and fixed drip trays and insulation (IGF Code ch.5);
- (PR) 5.2.3.12 examining electrical equipment including electrical bonding arrangements and bulkhead/deck penetrations including access openings in hazardous areas (IGF Code chs.5, 12 and 14);
- (PR) 5.2.3.13 examining the condition and arrangement of fuel storage, bunkering and supply systems, including external examination of storage tank and secondary barrier if fitted and relief valves if accessible, verifying the satisfactory operation of the tank monitoring system, and examining and

testing installed bilge alarms and means of drainage (IGF Code chs.6, 8, and 15);

- (PR) 5.2.3.14 testing of the remote and local closing of the installed main tank valve (IGF Code chs.6 and 10);
- (PR) 5.2.3.15 examining bunkering stations and the fuel bunkering system including operation of the fuel bunkering control, monitoring and shutdown systems (IGF Code ch.8);
- (PR) 5.2.3.16 examining the Ship-shore link (SSL) or equivalent means for automatic and manual ESD communication to the bunkering source (IGF Code para.8.5.7);
- (PR) 5.2.3.17 examining the fuel supply system including the fuel supply system control, monitoring and shutdown systems (IGF Code chs.9 and 15);
- (PR) 5.2.3.18 testing of the remote and local closing of the master fuel valve for each engine compartment (IGF Code chs.5, 9 and 15);
- (PR) 5.2.3.19 testing gas detectors, temperature sensors, pressure sensors, level indicators, and other equipment providing input to the fuel safety system including proper response of the fuel safety system upon fault conditions (IGF Code ch.15);
- (PR) 5.2.3.20 examining the storage tanks and all associated piping for fuel storage, fuel bunkering, and fuel supply such as venting, compressing, refrigerating, liquefying, heating, storing, burning or otherwise handling the fuel and liquid nitrogen installations, and requiring removal of insulation from the piping and opening for examination and hydrostatic test of suspected pipeline as necessary, and leak test of complete piping after reassembly (IGF Code chs.5, 6, 7, 8, 9 and 10);
- (PR) 5.2.3.21 examining emergency shutdown valves, check valves, block and bleed valves, master gas valves, remote operating valves, isolating valves for pressure relief valves in the fuel storage, fuel bunkering, and fuel supply piping systems, with randomly selected valves being opened for examination (IGF Code chs.5, 6, 7, 9, 15 and 16);
- (PR) 5.2.3.22 examining pressure relief valves connected to fuel storage tanks and connected pipes and venting system, with PRV being opened for examination, adjusted and function tested (IGF Code ch.6);
- (PR) 5.2.3.23 examining and testing pressure relief valves in fuel supply/bunker lines, including valves being opened for internal examination and testing; the number of valves being opened up for internal examination and being tested should include all PRVs that were not internally examined and tested in the past 5 years and a random selection of PRVs that were internally examined and tested in the past 5 years provided satisfactory records of overhaul and testing of these PRVs are available (IGF Code ch.6);
- (PR) 5.2.3.24 examining pressure/vacuum relief valves or devices for interbarrier spaces and hold spaces, with the valves being opened, examined, tested and readjusted as necessary (IGF Code ch.6);

(PR)	5.2.3.25	examining fuel storage tanks internally in accordance with an approved survey plan (IGF Code ch.6);
(PR)	5.2.3.26	examining and testing of spill protection and water spray systems, for portable liquefied gas fuel tanks located on open deck (IGF Code para.6.5.2);
(PR)	5.2.3.27	examining and testing the thermal oxidation system if any (IGF Code para.6.9.4);
(PR)	5.2.3.28	examining and NDE testing the low temperature steel shielding at the bunker station if any (IGF Code para.8.3.1.6);
(PR)	5.2.3.29	examining fuel pumps, compressors, process pressure vessels, inert gas generators, heat exchangers and other components used in connection with fuel handling (IGF Code chs.5, 6, 8, 9, 10 and 15);
(PR)	5.2.3.30	examining electrical equipment including the physical condition of electrical cables and supports, intrinsically safe, explosion proof, or increased safety features of electrical equipment, including functional tests of pressurized electrical equipment and associated alarms, testing of electrical equipment for de-energization which is not certified for use in hazardous areas and insulation resistance test of circuits passing through hazardous zone (IGF Code chs.12 and 14);
(PR)	5.2.3.31	examining and testing gas detectors, temperature sensors, pressure sensors, level indicators, and other equipment providing input to the fuel safety system, including verification of the response upon fault conditions, and the calibrations of pressure, temperature and level indicating equipment in accordance with the manufacturer's requirements (IGF Code ch.15); ⁶⁰
(PR)	5.2.3.32	examining the arrangements for the fire protection and fire extinction (IGF Code ch.11);
(PR)	5.2.3.33	examining the fire pump capacity and working pressure in relation to the water spray system, if the water spray system is part of the fire main system (IGF Code para.11.4.1);
(PR)	5.2.3.34	examining the isolating valves of the fire main, when the fuel storage tank or tanks are located on the open deck (IGF Code para.11.4.2);
(PR)	5.2.3.35	examining the water spray system arrangement for fuel storage tanks(s) on open deck including remote operation (IGF Code section 11.5);
(PR)	5.2.3.36	examining the fixed dry chemical powder fire-extinguishing system for the bunkering station area (IGF Code para.11.6.1);
(PR)	5.2.3.37	examining the portable dry powder extinguisher (IGF Code para.11.6.2);
(PR)	5.2.3.38	examining the fixed fire detection and alarm system (IGF Code sections 11.7, 15.9);

⁶⁰ Where applicable, refer to the *Unified interpretations of the IGF Code* (MSC.1/Circ.1591).

- (PR) 5.2.3.39 checking the records about drills and emergency exercises (IGF Code ch.17); and
- (PR) 5.2.3.40 checking the pre-bunkering verification records according to the bunker safety checklist (IGF Code ch.18).
- (PR) 5.2.4 For the hull, machinery and equipment of passenger ships the completion of the renewal survey should consist of:
- (PR) 5.2.4.1 after a satisfactory survey, issuing the Passenger Ship Safety Certificate and its associated Record of Equipment (Form P).

Annex 2

SURVEY GUIDELINES UNDER THE 1966 LL CONVENTION, AS MODIFIED BY THE 1988 PROTOCOL RELATING THERETO

- (L) 1 GUIDELINES FOR SURVEYS FOR THE INTERNATIONAL LOAD LINE CERTIFICATE OR INTERNATIONAL LOAD LINE EXEMPTION CERTIFICATE
- (LI) **1.1** Initial surveys see part "General" section 4.1
- (LI) 1.1.1 For the load line the examination of plans and designs should consist of:
- (LI) 1.1.1.1 examining the structural strength at the draught corresponding to the assigned freeboard (LLC 66/88 reg.1);
- (LI) 1.1.2 examining the intact stability, and, where applicable, the damaged stability information and the loading and ballasting information that is to be supplied to the master, and, where not dispensed by the Administration, inclining experimental data (LLC 66/88/08 regs.1 and 10; IS Code chs.1, 2 and 3); and
- (LI) 1.1.1.3 determining the freeboard, including specifying and considering the conditions of assignment for the freeboard (LLC 66/88/03 regs.11 to 45).
- (LI) 1.1.2 For the load line the survey during construction and after installation should consist of:
- (LI) 1.1.2.1 checking that, as far as its strength is concerned, the ship has been constructed in accordance with the approved plans (LLC 66/88 reg.1);
- (LI) 1.1.2.2 confirming that the deck line and load line mark are properly positioned (LLC 66/88 regs.4 to 9);
- (LI) 1.1.2.3 witnessing the inclining experiment or lightweight survey (LLC 66/88/03 reg.10);
- (LI) 1.1.2.4 examining the superstructure end bulkheads and the openings therein (LLC 66/88 regs.11 and 12);
- (LI) 1.1.2.5 examining the means of securing the weathertightness of cargo hatchways, other hatchways and other openings on the freeboard and superstructure decks (LLC 66/88 regs.13 to 18);
- (LI) 1.1.2.6 examining the ventilators and air pipes, including their coamings and closing appliances (LLC 66/88 regs.19 and 20);
- (LI) 1.1.2.7 examining the watertight integrity of the closures to any openings in the ship's side below the freeboard deck (LLC 66/88 reg.21);
- (LI) 1.1.2.8 examining the scuppers, inlets and discharges (LLC 66/88/03 reg.22);
- (LI) 1.1.2.9 examining the garbage chutes (LLC 66/88/03 reg.22-1);

- (LI) 1.1.2.10 examining the spurling pipes and cable lockers (LLC 66/88/03 reg.22-2);
- (LI) 1.1.2.11 examining the side scuttles and deadlights (LLC 66/88 reg.23);
- (LI) 1.1.2.12 examining the bulwarks including the provision of freeing ports, special attention being given to any freeing ports fitted with shutters (LLC 66/88/03 regs.24 and 25);
- (LI) 1.1.2.13 examining the guardrails, gangways, walkways and other means provided for the protection of the crew and means for safe passage of the crew (LLC 66/88/03 regs.25 and 25-1);
- (LI) 1.1.2.14 examining the special requirements for ships permitted to sail with type "A" or type "B-minus" freeboards (LLC 66/88/03 regs.26 and 27); and
- (LI) 1.1.2.15 checking, when applicable, the fittings and appliances for timber deck cargoes (LLC 66/88 regs.42 to 45).
- (LI) 1.1.3 For the load line the check that certificates, etc., have been placed on board should consist of:
- (LI) 1.1.3.1 checking that the loading and ballasting information has been supplied to the master (LLC 66/88 reg.10).
- (LI) 1.1.4 For the load line the completion of the initial survey should consist of:
- (LI) 1.1.4.1 after a satisfactory survey, issuing the International Load Line Certificate or International Load Line Exemption Certificate.
- (LA) **1.2 Annual surveys** see part "General", section 4.2
- (LA) 1.2.1 For the load line the examination of current certificates and other records should consist of:
- (LA) 1.2.1.1 checking the validity, as appropriate, of the Cargo Ship Safety Equipment Certificate, the Cargo Ship Safety Radio Certificate and the Cargo Ship Safety Construction Certificate or the Cargo Ship Safety Certificate;
- (LA) 1.2.1.2 checking the validity of the Safety Management Certificate (SMC) and that a copy of the Document of Compliance (DOC) is on board;
- (LA) 1.2.1.3 checking the validity of the International Ship Security Certificate;
- (LA) 1.2.1.4 checking the validity of the International Load Line Certificate or International Load Line Exemption Certificate;
- (LA) 1.2.1.5 checking the validity of the International Oil Pollution Prevention Certificate;
- (LA) 1.2.1.6 checking the certificate of class, if the ship is classed with a classification society;

(LA)	1.2.1.7	checking, when appropriate, the validity of the International Certificate of Fitness for the Carriage of Dangerous Chemicals in Bulk or the Certificate of Fitness for the Carriage of Dangerous Chemicals in Bulk;
(LA)	1.2.1.8	checking, when appropriate, the validity of the International Certificate of Fitness for the Carriage of Liquefied Gases in Bulk;
(LA)	1.2.1.9	checking, when appropriate, the validity of the International Pollution Prevention Certificate for the Carriage of Noxious Liquid Substances in Bulk;
(LA)	1.2.1.10	checking, when appropriate, the validity of the International Sewage Pollution Prevention Certificate;
(LA)	1.2.1.11	checking, when appropriate, the validity of the International Air Pollution Prevention Certificate;
(LA)	1.2.1.12	confirming, when appropriate, the validity of the International Energy Efficiency Certificate (MARPOL Annex VI regs.6.4 and 6.5);
(LA)	1.2.1.13	confirming, when appropriate, that confirmation of compliance for the SEEMP part II is provided to and retained on board the ship (MARPOL Annex VI, reg. 5.4.5) ⁶¹ ;
(LA)	1.2.1.14	confirming, when appropriate, the validity of the Statements of Compliance related to fuel oil consumption reporting (MARPOL Annex VI, regs.6.6 and 6.7);
(LA)	1.2.1.15	checking, when appropriate, the validity of the International Ballast Water Management Certificate;
(LA)	1.2.1.16	checking that the ship's complement complies with the Minimum Safe Manning Document (SOLAS 74/00/12 reg.V/14) (SOLAS 74/88 reg.V/13(b));
(LA)	1.2.1.17	checking that the master, officers and ratings are certificated as required by the STCW Convention;
(LA)	1.2.1.18	checking whether any new equipment has been fitted and, if so, confirm that it has been approved before installation and that any changes are reflected in the appropriate certificate;
(LA)	1.2.1.19	checking that the stability and, where applicable, the loading and ballasting information is available (LLC 66/88/08 regs.1 and 10; IS Code chs.1, 2 and 3); and
(LA)	1.2.1.20	confirming the availability of the International Anti-fouling System Certificate (AFS 2001 annex 4 reg.2), when applicable.
(LA)	1.2.2	For the load line the annual survey should consist of:

⁶¹ Refer to the Sample Format of Confirmation of Compliance, Early Submission of the SEEMP part II on the ship fuel oil consumption data collection plan and its timely verification pursuant to regulation 5.4.5 of MARPOL Annex VI (MEPC.1/Circ.876).

- (LA) 1.2.2.1 checking, in general, that there has been no deterioration in the strength of the hull (LLC 66/88 reg.1);
- (LA) 1.2.2.2 checking the positions of the deck line and load line which, if necessary, are to be re-marked and re-painted (LLC 66/88 regs.4 to 9);
- (LA) 1.2.2.3 checking that no alterations have been made to the hull or superstructures that would affect the calculations determining the position of the load lines (LLC 66/88 regs.11 to 45);
- (LA) 1.2.2.4 examining the superstructure end bulkheads and the openings therein (LLC 66/88 regs.11 and 12);
- (LA) 1.2.2.5 examining the means of securing the weathertightness of cargo hatchways, other hatchways and other openings on the freeboard and superstructure decks (LLC 66/88 regs.13 to 18);
- (LA) 1.2.2.6 examining the ventilators and air pipes, including their coamings and closing appliances (LLC 66/88 regs.19 and 20);
- (LA) 1.2.2.7 examining the watertight integrity of the closures to any openings in the ship's side below the freeboard deck (LLC 66/88 reg.21);
- (LA) 1.2.2.8 examining the scuppers, inlets and discharges (LLC 66/88 reg.22);
- (LA) 1.2.2.9 examining the garbage chutes (LLC 66/88/03 reg.22-1);
- (LA) 1.2.2.10 examining the means provided to minimize water ingress through the spurling pipes and chain lockers (LLC 66/88/03 reg.22-2);
- (LA) 1.2.2.11 examining the side scuttles and deadlights (LLC 66/88 reg.23);
- (LA) 1.2.2.12 examining the bulwarks including the provision of freeing ports, special attention being given to any freeing ports fitted with shutters (LLC 66/88/03 regs.24 and 25);
- (LA) 1.2.2.13 examining the guardrails, gangways, walkways and other means provided for the protection of the crew and means for safe passage of the crew (LLC 66/88/03 regs.25 and 25-1);
- (LA) 1.2.2.14 examining the special requirements for ships permitted to sail with type "A" or type "B-minus" freeboards (LLC 66/88/03 regs.26 and 27); and
- (LA) 1.2.2.15 checking, when applicable, the fittings and appliances for timber deck cargoes (LLC 66/88 regs.42 to 45).
- (LA) 1.2.3 For the load line the completion of the annual survey should consist of:
- (LA) 1.2.3.1 after a satisfactory survey, endorsement of the International Load Line Certificate or International Load Line Exemption Certificate; and
- (LA) 1.2.3.2 if a survey shows that the condition of a ship or its equipment is unsatisfactory, see part "General", section 4.8.

(LR)	1.3	Renewal surveys – see part "General", section 4.5
(LR)	1.3.1	For the load line the examination of current certificates and other records should consist of:
(LR)	1.3.1.1	the provisions of (LA) 1.2.1, except for the validity of the International Load Line Certificate or International Load Line Exemption Certificate.
(LR)	1.3.2	For the load line the renewal survey should consist of:
(LR)	1.3.2.1	the provisions of (LA) 1.2.2;
(LR)	1.3.2.2	examining the hull to ensure that its strength is sufficient for the draught corresponding to the freeboard assigned (LLC 66/88 reg.1).
(LR)	1.3.3	For the load line the completion of the renewal survey should consist of:
(LR)	1.3.3.1	after a satisfactory survey, issuing the International Load Line Certificate or International Load Line Exemption Certificate.

Annex 3

SURVEY GUIDELINES UNDER THE MARPOL CONVENTION

- (O) 1 GUIDELINES FOR SURVEYS FOR THE INTERNATIONAL OIL POLLUTION PREVENTION CERTIFICATE
- (OI) **1.1** Initial surveys see part "General", section 4.1
- (OI) 1.1.1 For oil pollution prevention the examination of plans and designs should consist of:
- (OI) 1.1.1.1 examining the arrangements for the control of the discharge of oil and examining the plans and designs of the oil discharge monitoring and control system and oily-water separating and oil filtering equipment; confirming that pollution prevention equipment is type-approved in accordance with the relevant resolution (MARPOL 90/04/15 Annex I regs.14 & 15);
- (OI) 1.1.1.2 examining the arrangements for operation in special areas (MARPOL 90/04/15 Annex I reg.15);
- (OI) 1.1.1.3 examining the arrangements for the segregation of oil and water ballast and the prohibition of carriage of oil in the forepeak tanks or in spaces forward of the collision bulkhead (MARPOL 90/04 Annex I reg.16);
- (OI) 1.1.1.4 examining the oil residue (sludge) tank and standard discharge arrangements (MARPOL 90/04/15 Annex I regs.12 and 13, and where applicable, Polar Code part II-A/para.1.2.4);
- (OI) 1.1.1.5 examining oil fuel tank protection arrangements (MARPOL 90/04 Annex I reg.12A, and where applicable, Polar Code part II-A/para.1.2.1);
- (OI) 1.1.1.6 confirming that requirements regarding capacity and protection of oil fuel tanks are complied with (MARPOL 90/04 Annex I reg.12A, and where applicable, Polar Code part II-A/para.1.2.1); and
- (OI) 1.1.1.7 examining the shipboard oil pollution emergency plan or in the case of a chemical/product tanker the shipboard marine pollution emergency plan (MARPOL 90/04 Annex I reg.37, and where applicable, Polar Code part II-A/para.1.1.4).
- (OI) 1.1.2 For oil pollution prevention, concerning the additional requirements for oil tankers the examination of plans and designs should consist of:
- (OI) 1.1.2.1 examining the ODME Manual and the arrangements for the control of the discharge of oil and for the retention of oil on board; verifying that the ODME is type-approved in accordance with the relevant resolution (MARPOL 90/04/15 Annex I regs.29, 31 and 34);
- (OI) 1.1.2.2 examining the arrangements for operation in special areas (MARPOL 90/04/15 Annex I reg.34);
- (OI) 1.1.2.3 examining the arrangements for the segregated ballast tanks, checking their capacity and ascertaining whether the draught and trim conditions will be met (MARPOL 90/04 Annex I reg.18);

- (OI) 1.1.2.4 examining the arrangements for crude oil washing, including shadow diagrams and the Operations and Equipment Manual, checking that an inert gas system is to be fitted (MARPOL 90/04 Annex I regs.33 and 35);
- (OI) 1.1.2.5 examining, as appropriate, the arrangements for the prevention of oil pollution in the event of collision or stranding (MARPOL 90/04/14 Annex I regs.19 to 22, and where applicable, Polar Code part II A/paras.1.2.2 and 1.2.3);
- (OI) 1.1.2.6 examining the protective location of the segregated ballast spaces and the arrangements for minimizing pollution due to side and bottom damages (MARPOL 90/04 Annex I regs.18, and 24 to 26);
- (OI) 1.1.2.7 confirming, as appropriate, that arrangements are made for the maintenance and inspection of wing and double bottom tanks or spaces (MARPOL 90/04/14 Annex I reg.19);
- (OI) 1.1.2.8 examining the arrangements for cargo pump-room bottom protection (double bottom where required) (MARPOL 90/04 Annex I reg.22);
- (OI) 1.1.2.9 examining the pumping, piping and discharge arrangements (MARPOL 90/04 Annex I reg.30);
- (OI) 1.1.2.10 examining the arrangements of the oil/water interface detector (MARPOL 90/04 Annex I reg.32);
- (OI) 1.1.2.11 examining, for oil tankers of 5,000 tonnes deadweight and above delivered on or after 1 February 2002, the intact stability (MARPOL 90/04 Annex I, reg.27);
- (OI) 1.1.2.12 examining, for oil tankers of 150 gross tonnage and above delivered after 31 December 1979, the subdivision and damage stability (MARPOL 90/04 Annex I reg.28);
- (OI) 1.1.2.13 examining the accidental oil outflow performance, as applicable (MARPOL 90/04 Annex I reg.23);
- (OI) 1.1.2.14 examining, where applicable, the stability instrument (MARPOL 90/04/14 Annex I reg.28); and
- (OI) 1.1.2.15 examining, when carriage of a stability instrument is waived, the alternative means of verification for intact and damage stability (MARPOL 90/04/14 Annex I reg.3).
- (OI) 1.1.3 For oil pollution prevention the survey during construction and after installation should consist of:
- (OI) 1.1.3.1 confirming the satisfactory installation and operation of, as appropriate, oil filtering equipment and when appropriate the operation of the automatic means provided to stop the discharge of effluent and the satisfactory operation of the alarm or other installation (MARPOL 90/04/15 Annex I regs.14 and 15);

- (OI) 1.1.3.2 confirming, when applicable, that the oil content meter and its recording device are operable and that there is a sufficient supply of consumables for the recording device on board (MARPOL 90/04/15 Annex I regs.14 and 15);
- (OI) 1.1.3.3 testing, where fitted, the automatic stopping device required for discharges in Special Areas (MARPOL 90/04/15 Annex I reg.15);
- (OI) 1.1.3.4 confirming the segregation of the oil fuel and water ballast system and the non-carriage of oil in forepeak tanks (MARPOL 90/04 Annex I reg.16);
- (OI) 1.1.3.5 confirming that the oil residue (sludge) tank and its discharge arrangements are satisfactory and, when the size of the sludge tank is approved on the basis of such installations, confirming the satisfactory operation of homogenizers, sludge incinerators or other recognized means for the control of sludge (MARPOL 90/04/15 Annex I reg.12, and where applicable, Polar Code part II-A/para.1.2.4);
- (OI) 1.1.3.6 confirming the provision of the standard discharge connection (MARPOL 90/04 Annex I reg.13); and
- (OI) 1.1.3.7 confirming oil fuel tank protection arrangements (MARPOL 90/04 Annex I reg.12A, and where applicable, Polar Code part II-A/para.1.2.1).
- (OI) 1.1.4 For oil pollution prevention, concerning the additional requirements for oil tankers the survey during construction and after installation should consist of:
- (OI) 1.1.4.1 confirming that the arrangements of slop tanks or cargo tanks designated as slop tanks, and associated piping systems, are satisfactory (MARPOL 90/04/15 Annex I regs.29 and 34);
- (OI) 1.1.4.2 confirming the satisfactory installation and operation of the oil discharge monitoring and control system, including any audible or visual alarms, the automatic and manual means to stop the discharge of effluent, the starting interlock, the accuracy of the flow meter and the applicable resolution's requirements for installation survey⁶² (MARPOL 90/04/15 Annex I regs.31 and 34);
- (OI) 1.1.4.3 confirming that the oil content meter and its recording device are operable and that there is a sufficient supply of consumables for the recording device on board (MARPOL 90/04/15 Annex I regs.31 and 34);
- (OI) 1.1.4.4 confirming that the approved oil/water interface detectors are on board and are operational (MARPOL 90/04 Annex I reg.32);
- (OI) 1.1.4.5 confirming that the arrangements of pumps, pipes and valves are in accordance with the requirements for segregated ballast systems and that there are no cross-connections between the cargo and segregated ballast systems (MARPOL 90/04 Annex I reg.18);

⁶² Refer to the *Revised guidelines and specifications for oil discharge monitoring and control systems for oil tankers* (Resolution A.586(14)), as amended, or the *Revised guidelines and specifications for oil discharge monitoring and control systems for oil tankers* (resolution MEPC.108(49)), as amended, as applicable.

- (OI) 1.1.4.6 where a portable spool piece is provided for the emergency discharge of segregated ballast by connecting the segregated ballast system to a cargo pump, confirming that non-return valves are fitted on the segregated ballast connections and that the spool piece is mounted in a conspicuous position in the pump-room with a permanent notice restricting its use (MARPOL 90/04 Annex I reg.18);
- (OI) 1.1.4.7 testing ballast pipelines that pass through cargo tanks and those cargo pipelines that pass through ballast tanks to ensure there is no cross contamination (MARPOL 90/04 Annex I reg.18);
- (OI) 1.1.4.8 confirming that the crude oil washing system is installed in accordance with the approved plans (MARPOL 90/04 Annex I regs.18 and 33) and, in particular:
- (OI) 1.1.4.8.1 examining crude oil washing piping, pumps, valves and deck-mounted washing machines for signs of leakage and to check that all anchoring devices for crude oil washing piping are intact and secure;
- (OI) 1.1.4.8.2 carrying out pressure testing of the crude oil washing system to 1.5 times the working pressure;
- (OI) 1.1.4.8.3 confirming in those cases where drive units are not integral with the tank washing machines that the number of operational drive units specified in the Manual are on board;
- (OI) 1.1.4.8.4 checking that, when fitted, steam heaters for water washing can be properly isolated during crude oil washing operations, either by double shut-off valves or by clearly identifiable blanks;
- (OI) 1.1.4.8.5 checking that the prescribed means of communication between the deck watchkeeper and the cargo control position is operational;
- (OI) 1.1.4.8.6 confirming that an overpressure relief device (or other approved arrangement) is fitted to the pumps supplying the crude oil washing system;
- (OI) 1.1.4.8.7 verifying that flexible hoses for the supply of oil to the washing machines on combination carriers are of an approved type, are properly stored and are in good condition;
- (OI) 1.1.4.9 verifying the effectiveness of the crude oil washing system (MARPOL 90/04 Annex I reg.33) and, in particular:
- (OI) 1.1.4.9.1 checking that the crude oil washing machines are operable and observing the proper operation of the washing machines by means of the movement indicators and/or sound patterns or other approved methods;
- (OI) 1.1.4.9.2 checking the effectiveness of the stripping system in appropriate cargo tanks by observing the monitoring equipment and by hand-dipping or other approved means;

- (OI) 1.1.4.9.3 verifying by internal tank inspection after crude oil washing that the installation and operational procedures laid down in the Operations and Equipment Manual are satisfactory;
- (OI) 1.1.4.10 confirming that, where there is a crude oil washing system, an inert gas system has been installed and tested in accordance with the requirements of SOLAS 74/88/2000 (see (EI) 1.1.5.2 in Annex 1);
- (OI) 1.1.4.11 confirming, as appropriate, that the arrangements for the prevention of oil pollution in the event of collision or stranding are in accordance with the approved plans (MARPOL 90/04/14 Annex I regs.19 to 22, Polar Code part II-A/paras.1.2.2 and 1.2.3);
- (OI) 1.1.4.12 confirming that the piping systems associated with the discharge of dirty ballast water or oil-contaminated water are satisfactory (MARPOL 90/04 Annex I reg.30);
- (OI) 1.1.4.13 confirming that the observation and discharge control positions for visually observing the discharge of oil-contaminated water, including the testing of the communication system between the two positions, are satisfactory (MARPOL 90/04 Annex I reg.30);
- (OI) 1.1.4.14 confirming that the means of draining cargo pumps and cargo lines, including the provision of a stripping device and the connections for pumping to the slop or cargo tanks or ashore, are satisfactory (MARPOL 90/04 Annex I reg.30);
- (OI) 1.1.4.15 confirming that closing devices installed in the cargo transfer system and cargo piping, as appropriate, are satisfactory (MARPOL 90/04 Annex I regs.23 and 26);
- (OI) 1.1.4.16 confirming that the subdivision and stability arrangements, in addition to the provision of (OI) 1.1.4.15, to prevent progressive flooding are satisfactory (MARPOL 90/04 Annex I regs.23 and 26); and
- (OI) 1.1.4.17 confirming the arrangements for cargo pump-room bottom protection (double bottom where required) (MARPOL 90/04 Annex I reg.22).
- (OI) 1.1.5 For oil pollution prevention the check that the documentation has been placed on board cargo ships should consist of:
- (OI) 1.1.5.1 confirming that certificates for type approval for the oil filtering equipment and oil content meters are available (MARPOL 90/04/15 Annex I reg.14);
- (OI) 1.1.5.2 confirming that the Oil Record Book or as an electronic record book⁶³(Part I) has been provided (MARPOL 90/04/19 Annex I reg.17);
- (OI) 1.1.5.3 confirming that the shipboard oil pollution emergency plan or, in the case of a chemical/product tanker, a shipboard marine pollution emergency plan has been provided (MARPOL 90/04 Annex I reg.37, and where applicable, Polar Code part II-A/section 1.1); and

⁶³ Refer to the *Guidelines for the use of electronic record books under MARPOL* (resolution MEPC.312(74)).

- (OI) 1.1.5.4 confirming, as appropriate, that the Operating and Maintenance Manuals for the 15 ppm bilge separator and 15 ppm bilge alarm are available.
- (OI) 1.1.6 For the oil pollution prevention the check that the documentation has been placed on board oil tankers should additionally consist of:
- (OI) 1.1.6.1 confirming that, if applicable, a ship-to-ship (STS) operations plan approved by the Administration has been provided (MARPOL Annex I reg.41);
- (OI) 1.1.6.2 confirming that, if applicable, a Crude Oil Washing Operations and Equipment Manual has been provided (MARPOL 90/04 Annex I reg.35);
- (OI) 1.1.6.3 confirming that an operations manual for the oil discharge monitoring and control system has been provided together with any other documentation requested by the applicable resolution⁶⁴ (MARPOL 90/04 Annex I reg.31);
- (OI) 1.1.6.4 confirming that certificates for type approval for the oil content meters, oil discharge monitoring and control system and oil/water interface detectors are available (MARPOL 90/04 Annex I regs.31 and 32);
- (OI) 1.1.6.5 confirming that the Oil Record Book or as an electronic record book⁶⁵ (Part II) has been provided (MARPOL 90/04 Annex I reg.36);
- (OI) 1.1.6.6 confirming that the information and data concerning the subdivision and damage stability has been provided (MARPOL 90/04 Annex I reg.28);
- (OI) 1.1.6.7 confirming that the shipboard oil pollution emergency plan or in the case of a chemical/product tanker a shipboard marine pollution emergency plan has been provided (MARPOL 90/04 Annex I reg.37, and where applicable, Polar Code part II-A/section 1.1);
- (OI) 1.1.6.8 confirming, for oil tankers of 5,000 tonnes deadweight and above delivered on/after 1 February 2002, that the intact stability has been approved (MARPOL 90/04 Annex I reg.27);
- (OI) 1.1.6.9 confirming, for oil tankers of 5,000 tonnes deadweight and above, that arrangements are in place to provide prompt access to shore-based damage stability and residual structural strength computerized calculation programs (MARPOL 90/04 Annex I reg.37.4);
- (OI) 1.1.6.10 confirming, where applicable, the stability instrument has been approved and is operating satisfactorily (MARPOL 90/04/14 Annex I reg.28); and

⁶⁴ Refer to *Revised guidelines and specifications for oil discharge monitoring and control systems for oil tankers* (resolution A.586(14)), as amended, or the *Revised guidelines and specifications for oil discharge monitoring and control systems for oil tankers* (resolution MEPC.108(49)), as amended, as applicable.

⁶⁵ Refer to the *Guidelines for the use of electronic record books under MARPOL*(resolution MEPC.312(74)).

- (OI) 1.1.6.11 confirming, when carriage of a stability instrument is waived, the alternative means of verification for intact and damage stability is recorded on Form B attached to the IOPP Certificate and is being applied effectively (MARPOL 90/04/14 Annex I reg.3).
- (OI) 1.1.7 For oil pollution prevention the completion of the initial survey should consist of:
- (OI) 1.1.7.1 after a satisfactory survey, issuing the International Oil Pollution Prevention Certificate.
- (OA) **1.2 Annual surveys** see part "General", section 4.2
- (OA) 1.2.1 For oil pollution prevention the examination of current certificates and other records should consist of:
- (OA) 1.2.1.1 checking the validity, as appropriate, of the Cargo Ship Safety Equipment Certificate, the Cargo Ship Safety Radio Certificate and the Cargo Ship Safety Construction Certificate or the Cargo Ship Safety Certificate;
- (OA) 1.2.1.2 checking the validity of the International Load Line Certificate or International Load Line Exemption Certificate;
- (OA) 1.2.1.3 checking the validity of the International Oil Pollution Prevention Certificate;
- (OA) 1.2.1.4 checking the certificates of class, if the ship is classed with a classification society;
- (OA) 1.2.1.5 checking, when appropriate, the validity of the International Sewage Pollution Prevention Certificate;
- (OA) 1.2.1.6 checking, when appropriate, the validity of the International Certificate of Fitness for the Carriage of Dangerous Chemicals in Bulk or the Certificate of Fitness for the Carriage of Dangerous Chemicals in Bulk;
- (OA) 1.2.1.7 checking, when appropriate, the validity of the International Certificate of Fitness for the Carriage of Liquefied Gases in Bulk;
- (OA) 1.2.1.8 checking, when appropriate, the validity of the International Pollution Prevention Certificate for the Carriage of Noxious Liquid Substances in Bulk;
- (OA) 1.2.1.9 checking, when appropriate, the validity of the International Air Pollution Prevention Certificate;
- (OA) 1.2.1.10 confirming, when appropriate, the validity of the International Energy Efficiency Certificate (MARPOL Annex VI, regs.6.4 and 6.5);

(OA)	1.2.1.11	confirming, when appropriate, that confirmation of compliance for the SEEMP part II is provided to and retained on board the ship (MARPOL Annex VI, reg. 5.4.5) ⁶⁶ ;
(OA)	1.2.1.12	confirming, when appropriate, the validity of the Statements of Compliance related to fuel oil consumption reporting (MARPOL Annex VI, regs.6.6 and 6.7);
(OA)	1.2.1.13	checking, when appropriate, the validity of the International Ballast Water Management Certificate;
(OA)	1.2.1.14	checking, when appropriate, the validity of the Safety Management Certificate (SMC) and that a copy of the Document of Compliance (DOC) is on board;
(OA)	1.2.1.15	checking the validity of the International Ship Security Certificate;
(OA)	1.2.1.16	checking, as appropriate, the validity of the Polar Ship Certificate;
(OA)	1.2.1.17	checking that the ship's complement complies with the Minimum Safe Manning Document (SOLAS 74/00/12 reg.V/14) (SOLAS 74/88 reg.V/13(b));
(OA)	1.2.1.18	checking that the master, officers and ratings are certificated as required by the STCW Convention;
(OA)	1.2.1.19	checking whether any new equipment has been fitted and, if so, confirming that it has been approved before installation and that any changes are reflected in the appropriate certificate;
(OA)	1.2.1.20	checking the certificates for the type approval of the oil filtering equipment (MARPOL 90/04/15 Annex I regs.14 and 15);
(OA)	1.2.1.21	checking, when appropriate, that the Operating and Maintenance Manuals for the 15 ppm bilge separator and 15 ppm bilge alarm are available on board;
(OA)	1.2.1.22	verifying, if applicable, that the 15 ppm bilge alarm has been calibrated by the manufacturer or a person authorized by the manufacturer and that a valid calibration certificate is available on board; ⁶⁷
(OA)	1.2.1.23	checking whether the appropriate entries have been made in Part I of the Oil Record Book or as an electronic record book ⁶⁸ (MARPOL 90/04/19 Annex I reg.17, and where applicable, Polar Code part II- A/section 1.1);

⁶⁶ Refer to the Sample Format of Confirmation of Compliance, Early Submission of the SEEMP part II on the ship fuel oil consumption data collection plan and its timely verification pursuant to regulation 5.4.5 of MARPOL Annex VI (MEPC.1/Circ.876).

⁶⁷ Applicable for installations complying with the *Revised guidelines and specifications for pollution prevention equipment for machinery space bilges of ships* (resolution MEPC.107(49)), as amended.

⁶⁸ Refer to the *Guidelines for the use of electronic record books under MARPOL*(resolution MEPC.312(74)).

- (OA) 1.2.1.24 confirming the availability of the International Anti-fouling System Certificate (AFS 2001 annex 4 reg.2), when applicable; and
- (OA) 1.2.1.25 confirming that the oil pollution emergency plan or, in the case of a chemical/product tanker, a shipboard marine pollution emergency plan, is on board (MARPOL 90/04 Annex I reg.37, and where applicable, Polar Code part II-A/section 1.1).
- (OA) 1.2.2 For oil pollution prevention the examination of current certificates and other records for oil tankers should additionally consist of:
- (OA) 1.2.2.1 confirming that the approved Dedicated Clean Ballast Tank Operation Manual, and/or the approved Operations and Equipment Manual for the Crude Oil Washing Systems, as appropriate, is/are on board (MARPOL 90/04 Annex I regs.18 and 35);
- (OA) 1.2.2.2 confirming, when appropriate, that a CAS Statement of Compliance together with the CAS Final Report⁶⁹ are on board (MARPOL 90/04/14 Annex I, regs.20.6, 20.7 and 21.6);
- (OA) 1.2.2.3 confirming that the operating and maintenance manual for the oil discharge monitoring and control system is on board (MARPOL 90/04 Annex I reg.31);
- (OA) 1.2.2.4 confirming that a valid calibration certificate for the oil discharge monitoring equipment is available on board;⁷⁰
- (OA) 1.2.2.5 checking whether the appropriate entries have been made in Part II of the Oil Record Book or as an electronic record book⁷¹ (MARPOL 90/04/19 Annex I reg.36, and where applicable, Polar Code part II-A/section 1.1);
- (OA) 1.2.2.6 confirming that for oil tankers of 5,000 tonnes deadweight and above delivered on/after 1 February 2002 the loading conditions and intact stability information, in an approved form, is on board (MARPOL 90/04 Annex I reg.27);
- (OA) 1.2.2.7 confirming that subdivision and damage stability information in an approved form, where applicable, is on board (MARPOL 90/04/14 Annex I reg.28);
- (OA) 1.2.2.8 checking the certificates for the type approval of the oil pollution prevention equipment, such as the oil content meters and oil/water interface detectors, and sighting the records of the various oil discharge monitoring equipment, as applicable (MARPOL 90/04 Annex I reg.31);
- (OA) 1.2.2.9 checking that the ship is allowed continued operation according to the phase-out scheme of MARPOL 90/04/14 Annex I reg.20);

⁶⁹ Refer to the *Condition Assessment Scheme* (resolution MEPC.94(46)), as amended.

⁷⁰ Applicable for installations complying with the *Revised guidelines and specifications for oil discharge monitoring and control systems for oil tankers* (resolution MEPC.108(49)), as amended.

⁷¹ Refer to the *Guidelines for the use of electronic record books under MARPOL*(resolution MEPC.312(74)).

- (OA) 1.2.2.10 confirming that, if applicable, a ship-to-ship (STS) operations plan approved by the Administration has been provided (MARPOL Annex I reg.41);
- (OA) 1.2.2.11 confirming, where applicable, the approved stability instrument is available on board and operating satisfactorily (MARPOL 90/04/14 Annex I reg.28); and
- (OA) 1.2.2.12 confirming, when carriage of a stability instrument is waived, the alternative means of verification for intact and damage stability recorded on Form B attached to the IOPP Certificate is available on board and is being applied effectively (MARPOL 90/04/14Annex I reg.3).
- (OA) 1.2.3 For oil pollution prevention the annual survey should consist of:
- (OA) 1.2.3.1 examining externally the oil filtering equipment and confirming, as far as practicable, its satisfactory operation including, when appropriate, testing the operation of the automatic means provided to stop the discharge of effluent and the alarm for the oil filtering equipment (MARPOL 90/04/15 Annex I regs.14 and 15);
- (OA) 1.2.3.2 testing, where fitted, the oil filtering equipment required for discharge in special areas (MARPOL 90/04/15 Annex I reg.15);
- (OA) 1.2.3.3 confirming the segregation of oil fuel and water ballast systems and that the arrangements prohibit the carriage of oil in forepeak tanks or in spaces forward of the collision bulkhead (MARPOL 90/04 Annex I reg.16);
- (OA) 1.2.3.4 checking that the arrangement of oil residue (sludge) tank and its discharge arrangements are satisfactory and confirming that, where applicable, homogenizers, sludge incinerators or other recognized means for the control of sludge are satisfactory (MARPOL 90/04/15 Annex I reg.12 and, where applicable, Polar Code part II-A/para.1.2.4); and
- (OA) 1.2.3.5 confirming that a standard discharge connection is provided (MARPOL 90/04 Annex I reg.13).
- (OA) 1.2.4 For oil pollution prevention the annual survey of the additional requirements for oil tankers should consist of:
- (OA) 1.2.4.1 examining the oil discharge monitoring and control system and its associated equipment (MARPOL 90/04 Annex I reg.31) and, in particular:
- (OA) 1.2.4.1.1 examining externally the system and equipment and, if applicable, verifying that the instrument is properly sealed;
- (OA) 1.2.4.1.2 confirming, as far as practicable, the satisfactory operation of the oil discharge monitoring and control system including the oil content meter and, where applicable, the automatic and manual means provided to stop the discharge of effluent and the starting interlock;

- (OA) 1.2.4.1.3 observing that indicators and recording devices are operable and verifying that a sufficient supply of consumables for the recorders is on board; and
- (OA) 1.2.4.1.4 testing, as far as practicable, any audible or visual alarms fitted to the oil discharge monitoring and control system;
- (OA) 1.2.4.2 examining, as far as practicable, the oil/water interface detectors (MARPOL 90/04 Annex I reg.32);
- (OA) 1.2.4.3 confirming that no cross-connections have been fitted between the cargo and segregated ballast systems (MARPOL 90/04 Annex I reg.18);
- (OA) 1.2.4.4 where a portable spool piece is provided for the emergency discharge of segregated ballast by connecting the segregated ballast system to a cargo pump, confirming that non-return valves are fitted on the segregated ballast connections and that the spool piece is mounted in a conspicuous position in the pump-room with a permanent notice restricting its use (MARPOL 90/04 Annex I reg.18);
- (OA) 1.2.4.5 confirming by sighting that there has been no contamination with oil in the segregated ballast tanks (MARPOL 90/04 Annex I reg.18);
- (OA) 1.2.4.6 confirming, as far as practicable, that the dedicated clean ballast tank arrangement remains satisfactory (MARPOL 90/04 Annex I reg.18);
- (OA) 1.2.4.7 confirming by sighting that there has been no contamination with oil in the dedicated clean ballast tanks (MARPOL 90/04 Annex I reg.18);
- (OA) 1.2.4.8 confirming, as far as practicable, that the crude oil washing system remains satisfactory (MARPOL 90/04 Annex I reg.33) and, in particular:
- (OA) 1.2.4.8.1 examining externally the crude oil washing piping, pumps, valves and deck-mounted washing machines for signs of leakage and checking that all anchoring devices for crude oil washing piping are intact and secure;
- (OA) 1.2.4.8.2 confirming, in those cases where drive units are not integral with the tank cleaning machines, that the number of operational drive units as specified in the Manual are on board;
- (OA) 1.2.4.8.3 checking that, when fitted, steam heaters for water washing can be properly isolated during crude oil washing operations, either by double shut-off valves or clearly identifiable blanks;
- (OA) 1.2.4.8.4 checking that the prescribed means of communication between the deck watchkeeper and the cargo control position is operational;
- (OA) 1.2.4.8.5 confirming that an overpressure relief device (or other approved arrangement) is fitted to the pumps supplying the crude oil washing systems;
- (OA) 1.2.4.8.6 confirming that flexible hoses for the supply of oil to the washing machines on combination carriers are of an approved type, are properly stored and are in good condition;

- (OA) 1.2.4.9 verifying, where applicable and as far as practicable, the effectiveness of the crude oil washing system (MARPOL 90/04 Annex I reg.33) and, in particular:
- (OA) 1.2.4.9.1 checking tanks containing departure and/or arrival ballast water, as applicable, to confirm the effectiveness of the cleaning and stripping;
- (OA) 1.2.4.9.2 checking, as far as practicable, that the crude oil washing machines are operable and, when the survey is carried out during crude oil washing operations, observing the proper operation of the washing machines by means of the movement indicators and/or sound patterns or other approved methods;
- (OA) 1.2.4.9.3 checking, as far as practicable, the effectiveness of the stripping system in appropriate cargo tanks by observing the monitoring equipment and by hand-dipping or other approved means;
- (OA) 1.2.4.10 confirming that on those existing tankers operating with special ballast arrangements the arrangements are as approved and are satisfactory (MARPOL 90/04 Annex I reg.18);
- (OA) 1.2.4.11 confirming, as appropriate and as far as practicable, that the arrangements for the prevention of oil pollution in the event of collision or stranding are approved and are satisfactory (MARPOL 90/04/14 Annex I regs.19 to 22, Polar Code part II-A/paras.1.2.2 and 1.2.3);
- (OA) 1.2.4.12 examining the piping systems associated with the discharge of dirty ballast or oil-contaminated water including the part flow system, if fitted (MARPOL 90/04 Annex I reg.30);
- (OA) 1.2.4.13 testing the communication system between the observation and discharge control positions (MARPOL 90/04 Annex I reg.30);
- (OA) 1.2.4.14 examining the means of draining cargo pumps and cargo lines, including the stripping device and the connections for pumping to the slop or cargo tanks or ashore (MARPOL 90/04 Annex I reg.30); and
- (OA) 1.2.4.15 confirming for oil tankers of 5,000 tonnes deadweight and above that arrangements are in place to provide prompt access to shore-based damage stability and residual structural strength computerized calculation programs (MARPOL 90/04 Annex I reg.37.4).
- (OA) 1.2.5 For oil pollution prevention the completion of the annual survey should consist of:
- (OA) 1.2.5.1 after a satisfactory survey, endorsing the International Oil Pollution Prevention Certificate; and
- (OA) 1.2.5.2 if a survey shows that the condition of a ship or its equipment is unsatisfactory, see part "General", section 4.8.
- (Oln) **1.3** Intermediate surveys see part "General", section 4.3

- (OIn) 1.3.1 For oil pollution prevention the examination of current certificates and other records should consist of:
- (OIn) 1.3.1.1 the provisions of (OA) 1.2.1.
- (OIn) 1.3.2 For oil pollution prevention the examination of current certificates and other records for oil tankers should additionally consist of:
- (OIn) 1.3.2.1 the provisions of (OA) 1.2.2.
- (OIn) 1.3.3 For oil pollution prevention the intermediate survey should consist of:
- (OIn) 1.3.3.1 the provisions of (OA) 1.2.3;
- (OIn) 1.3.3.2 examining the oily-water separating equipment or oil filtering equipment or process unit, where fitted, including associated pumps, piping and fittings for wear and corrosion (MARPOL 90/04/15 Annex I regs.14 and 15); and
- (OIn) 1.3.3.3 examining the oil content meter (15 ppm alarm and bilge monitor) for obvious defects, deterioration or damage and checking the record of calibration of the meter when done in accordance with the manufacturer's operational and instruction manual (MARPOL 90/04/15 Annex I reg.14).
- (OIn) 1.3.4 For oil pollution prevention the intermediate survey of the additional requirements for oil tankers should consist of:
- (OIn) 1.3.4.1 the provisions of (OA) 1.2.4;
- (OIn) 1.3.4.2 examining the oil discharge monitoring and control system and the oil content meter for obvious defects, deterioration or damage, and checking the record of calibration of the meter when done in accordance with the manufacturer's operational and instruction manual (MARPOL 90/04 Annex I reg.31);
- (OIn) 1.3.4.3 confirming the satisfactory operation of the oil/water interface detectors (MARPOL 90/04 Annex I reg.32);
- (OIn) 1.3.4.4 for the crude oil washing system (MARPOL 90/04 Annex I reg.33):
- (OIn) 1.3.4.4.1 examining the crude oil washing piping outside the cargo tanks; if upon examination there is any doubt as to its condition, the piping may be required to be pressure tested, gauged or both; particular attention should be paid to any repairs such as welded doublers;
- (OIn) 1.3.4.4.2 confirming the satisfactory operation of the isolation valves to steam heaters for washing water, when fitted;
- (OIn) 1.3.4.4.3 examining at least two selected cargo tanks for the express purpose of verifying the continued effectiveness of the installed crude oil washing and stripping systems; if a tank cannot be gas freed for the safe entry of the surveyor, an internal examination of it should not be conducted, in which case the verification is to be made by another alternative method

acceptable to the Administration; an acceptable alternative would be satisfactory results during the surveys required by (OA) 1.2.4.9 (MARPOL 90/04 Annex I reg.33); and

- (OIn) 1.3.4.5 examining the manual and/or remote operation of the individual tank valves (or other similar closing devices) to be kept closed at sea (MARPOL 90/04 Annex I regs.23 and 26).
- (OIn) 1.3.5 For the oil pollution prevention the completion of the intermediate survey should consist of:
- (OIn) 1.3.5.1 after a satisfactory survey, endorsing the International Oil Pollution Prevention Certificate; and
- (OIn) 1.3.5.2 if a survey shows that the condition of a ship or its equipment is unsatisfactory, see part "General", section 4.8.
- (OR) **1.4 Renewal surveys** see part "General" section 4.5
- (OR) 1.4.1 For oil pollution prevention the examination of current certificates and other records should consist of:
- (OR) 1.4.1.1 the provisions of (OA) 1.2.1, except for the validity of the International Oil Pollution Prevention Certificate; and
- (OR) 1.4.1.2 verifying that, if applicable, the 15 ppm bilge alarm has been calibrated by the manufacturer or a person authorized by the manufacturer and that a valid calibration certificate is available on board.⁷²
- (OR) 1.4.2 For oil pollution prevention the examination of current certificates and other records for tankers should additionally consist of:
- (OR) 1.4.2.1 the provisions of (OA) 1.2.2; and
- (OR) 1.4.2.2 verifying that, if applicable, the oil discharge monitoring equipment has been calibrated and that a valid calibration certificate is available on board.⁷³
- (OR) 1.4.3 For oil pollution prevention the renewal survey should consist of:
- (OR) 1.4.3.1 the provisions of (OIn) 1.3.3;
- (OR) 1.4.3.2 confirming, if necessary by simulated test or equivalent, the satisfactory operation of the oily-water separating equipment or oil filtering equipment (MARPOL 90/04/15 Annex I reg.14);

⁷² Applicable for installations complying with the *Revised guidelines and specifications for pollution prevention equipment for machinery space bilges of ships* (resolution MEPC.107(49)), as amended.

⁷³ Applicable for installations complying with the *Revised guidelines and specifications for oil discharge monitoring and control systems for oil tankers* (resolution MEPC.108(49)), as amended.
- (OR) 1.4.3.3 confirming, if necessary by simulated test or equivalent, the satisfactory operation of the oil discharge monitoring and control system, including where practicable the automatic and manual operation of the means provided to stop the discharge of effluent (MARPOL 90/04 Annex I reg.31);
- (OR) 1.4.3.4 confirming the satisfactory operation of the alarm for the oil filtering system (MARPOL 90/04/15 Annex I reg.14); and
- (OR) 1.4.3.5 confirming the satisfactory operation of homogenizers, sludge incinerators or other recognized means for the control of sludge when the size of oil residue (sludge) tank is approved on the basis of such installations (MARPOL 90/04/15 Annex I reg.12).
- (OR) 1.4.4 For oil pollution prevention the renewal survey of the additional requirements for oil tankers should consist of:
- (OR) 1.4.4.1 the provisions of (OIn) 1.3.4;
- (OR) 1.4.4.2 confirming that the arrangements of slop tanks or cargo tanks designated as slop tanks and associated piping systems are satisfactory (MARPOL 90/04/15 Annex I regs.29 and 34);
- (OR) 1.4.4.3 confirming, if necessary by simulated test or equivalent, the satisfactory operation of the oil discharge monitoring and control system and its associated equipment, including the oil/water interface detectors (MARPOL 90/04 Annex I regs.31 and 32);
- (OR) 1.4.4.4 confirming that the arrangements of pumps, pipes and valves are in accordance with the requirements for SBT systems (MARPOL 90/04 Annex I reg.18);
- (OR) 1.4.4.5 confirming that the arrangements of pumps, pipes and valves are in accordance with the *Revised specifications for oil tankers with dedicated clean ballast tanks* (MARPOL 90/04 Annex I reg.18);
- (OR) 1.4.4.6 confirming that the crude oil washing system is in accordance with the requirements for such systems (MARPOL 90/04 Annex I reg.33) and, in particular:
- (OR) 1.4.4.6.1 carrying out pressure testing of the crude oil washing system to at least the working pressure;
- (OR) 1.4.4.6.2 examining the cargo tanks for the express purpose of verifying the continued effectiveness of the installed crude oil washing and stripping systems;
- (OR) 1.4.4.6.3 examining internally, when fitted, the isolation valves for any steam heaters;
- (OR) 1.4.4.6.4 verifying, by internal tank inspection or by another alternative method acceptable to the Administration, the effectiveness of the crude oil washing system; if the tank cannot be gas freed for the safe entry of the surveyor, an internal inspection should not be conducted; an acceptable

alternative would be satisfactory results during the surveys required by (OA) 1.2.4.9 (MARPOL 90/04 Annex I reg.33);

- (OR) 1.4.4.7 confirming that there is no leakage from those ballast pipelines passing through cargo tanks and those cargo pipelines passing through ballast tanks (MARPOL 90/04 Annex I regs.18 and 33);
- (OR) 1.4.4.8 confirming that the pumping, piping and discharge arrangements are satisfactory (MARPOL 90/04 Annex I reg.30) and, in particular:
- (OR) 1.4.4.8.1 confirming that the piping systems associated with the discharge of dirty ballast water or oil contaminated water are satisfactory;
- (OR) 1.4.4.8.2 confirming that the means of draining cargo pumps and cargo lines, including the stripping device and the connections for pumping to the slop or cargo tanks or ashore are satisfactory;
- (OR) 1.4.4.8.3 confirming that the arrangements for the part flow system, where fitted, are satisfactory;
- (OR) 1.4.4.9 confirming that closing devices installed in the cargo transfer system and cargo piping as appropriate are satisfactory (MARPOL 90/04 Annex I regs.23 and 26);
- (OR) 1.4.4.10 confirming, as appropriate and as far as practicable, that the arrangements for the prevention of oil pollution in the event of collision or stranding are satisfactory (MARPOL 90/04/14 Annex I regs.19 to 22; Polar Code part II-A/paras.1.2.2 and 1.2.3); and
- (OR) 1.4.4.11 confirming for oil tankers of 5,000 tonnes deadweight and above that arrangements are in place to provide prompt access to shore-based damage stability and residual structural strength computerized calculation programs (MARPOL 90/04 Annex I reg.37.4).
- (OR) 1.4.5 For oil pollution prevention the completion of the renewal survey should consist of:
- (OR) 1.4.5.1 after a satisfactory survey, issuing the International Oil Pollution Prevention Certificate.
- (N) 2 GUIDELINES FOR SURVEYS FOR THE INTERNATIONAL POLLUTION PREVENTION CERTIFICATE FOR THE CARRIAGE OF NOXIOUS LIQUID SUBSTANCES IN BULK
- (NI) **2.1** Initial surveys see part "General", section 4.1
- (NI) 2.1.1 For the carriage of noxious liquid substances in bulk the examination of plans and designs (as applicable to the cargoes the ship is to be certified to carry) should consist of:
- (NI) 2.1.1.1 drawing up the list of noxious liquid substances it is proposed the ship will be certified to carry (MARPOL 90/04 Annex II reg.6);
- (NI) 2.1.1.2 examining the pumping system (MARPOL 90/04 Annex II reg.12);

- (NI) 2.1.1.3 examining the stripping system (MARPOL 90/04 Annex II reg.12);
- (NI) 2.1.1.4 examining the tank washing system and equipment (MARPOL 90/04 Annex II reg.14 and App.4);
- (NI) 2.1.1.5 examining the underwater discharge arrangements (MARPOL 90/04 Annex II reg.12);
- (NI) 2.1.1.6 examining the ventilation equipment for residue removal (MARPOL 90/04 Annex II reg.13 and App.7);
- (NI) 2.1.1.7 examining the heating system for solidifying and high viscosity substances (MARPOL 90/04 Annex II reg.14 and App.4);
- (NI) 2.1.1.8 examining the Procedures and Arrangements Manual (including cargo carriage requirements to meet Annex II regulations) (MARPOL 90/04 Annex II reg.14 and App.4 and where applicable, Polar Code part II-A/ch.2, MEPC.1/Circ.856);
- (NI) 2.1.1.9 examining the shipboard marine pollution emergency plan (MARPOL 90/04 Annex II reg.17, and where applicable, Polar Code part II-A/ch.2); and
- (NI) 2.1.1.10 examining, if applicable, the construction and arrangements of a ship certified to carry individually identified vegetable oils under exemption from the carriage requirements (MARPOL 90/04 Annex II, reg.4.1.3).
- (NI) 2.1.2 For the carriage of noxious liquid substances in bulk, the survey during construction and after installation (as applicable to the cargoes the ship is to be certified to carry) should consist of:
- (NI) 2.1.2.1 confirming that the pumping and stripping systems are satisfactory and that portable pipes or bends in sufficient number, if required, are on board (MARPOL 90/04 Annex II reg.12);
- (NI) 2.1.2.2 conducting the water test for assessing the stripping quantity, as required (MARPOL73/78/90/04 Annex II reg.12 and App.5);
- (NI) 2.1.2.3 confirming that the tank washing machines provided on board are in working order, are those described in the Procedures and Arrangements Manual and are installed in accordance with the approved plans (MARPOL 90/04 Annex II reg.14 and App.4);
- (NI) 2.1.2.4 confirming that the wash water heating system, if required, is installed in accordance with the approved plans (MARPOL 90/04 Annex II reg.14 and App.4);
- (NI) 2.1.2.5 confirming that the number and position of tank cleaning openings for portable machines are in accordance with the approved plans (MARPOL 90/04 Annex II reg.14 and App.4);
- (NI) 2.1.2.6 confirming that the underwater discharge outlet or outlets are in accordance with the approved plans (MARPOL 90/04 Annex II reg.12);

- (NI) 2.1.2.7 verifying by actual test that the discharge rate of the pumps, where a variable rate type is used, can be controlled as specified in the Procedures and Arrangements Manual (MARPOL 90/04 Annex II reg.14 and App.4);
- (NI) 2.1.2.8 confirming that the ventilation equipment for residue removal is installed in accordance with the approved plan and is in working order and that the pressure in the driving medium for portable fans for ventilation equipment for residue removal can be achieved to give the required fan capacity (MARPOL 90/04 Annex II reg.13 and App.7);
- (NI) 2.1.2.9 confirming that the heating system for solidifying and high viscosity substances is installed in accordance with the approved plan (MARPOL 90/04 Annex II reg.14 and App.4); and
- (NI) 2.1.2.10 confirming if applicable the construction and arrangements of a ship certified to carry individually identified vegetable oils under exemption from the carriage requirements (MARPOL 90/04 Annex II reg.4.1.3).
- (NI) 2.1.3 For the carriage of noxious liquid substances in bulk the check that the required documentation has been placed on board cargo ships (as applicable to the cargoes the ship is to be certified to carry) should consist of:
- (NI) 2.1.3.1 confirming that the Procedures and Arrangements Manual has been provided (MARPOL 90/04 Annex II reg.14, and where applicable, Polar Code part II-A ch.2, MEPC.1/Circ.856);
- (NI) 2.1.3.2 confirming that the Cargo Record Book or as an electronic record book⁷⁴ has been provided (MARPOL 90/04/19 Annex II reg.15, and where applicable, Polar Code part II-A ch.2); and
- (NI) 2.1.3.3 confirming that the shipboard marine pollution emergency plan is provided (MARPOL 04 Annex II, reg.17, and where applicable, Polar Code part II-A ch.2).
- (NI) 2.1.4 For the carriage of noxious liquid substances in bulk the completion of the initial survey should consist of:
- (NI) 2.1.4.1 after a satisfactory survey, issuing the International Certificate for the Carriage of Noxious Liquid Substances in Bulk.
- (NA) **2.2 Annual surveys** see part "General", section 4.2
- (NA) 2.2.1 For the carriage of noxious liquid substances in bulk the examination of current certificates and other records should consist of:
- (NA) 2.2.1.1 checking the validity, as appropriate, of the Cargo Ship Safety Equipment Certificate, the Cargo Ship Safety Radio Certificate and the Cargo Ship Safety Construction Certificate or the Cargo Ship Safety Certificate;

⁷⁴ Refer to the *Guidelines for the use of electronic record books under MARPOL*(resolution MEPC.312(74)).

- (NA) 2.2.1.2 checking the validity of the International Load Line Certificate or International Load Line Exemption Certificate;
- (NA) 2.2.1.3 checking the validity of the International Oil Pollution Prevention Certificate;
- (NA) 2.2.1.4 checking the certificates of class, if the ship is classed with a classification society;
- (NA) 2.2.1.5 checking, when appropriate, the validity of the International Certificate of Fitness for the Carriage of Liquefied Gases in Bulk;
- (NA) 2.2.1.6 checking the validity of the International Pollution Prevention Certificate for the Carriage of Noxious Liquid Substances in Bulk;
- (NA) 2.2.1.7 checking, when appropriate, the validity of the International Air Pollution Prevention Certificate;
- (NA) 2.2.1.8 confirming, when appropriate, the validity of the International Energy Efficiency Certificate (MARPOL Annex VI, regs.6.4 and 6.5);
- (NA) 2.2.1.9 confirming, when appropriate, that confirmation of compliance for the SEEMP part II is provided to and retained on board the ship (MARPOL Annex VI, reg. 5.4.5)⁷⁵;
- (NA) 2.2.1.10 confirming, when appropriate, the validity of the Statements of Compliance related to fuel oil consumption reporting (MARPOL Annex VI, regs.6.6 and 6.7);
- (NA) 2.2.1.11 checking, when appropriate, the validity of the International Sewage Pollution Prevention Certificate;
- (NA) 2.2.1.12 checking, when appropriate, the validity of the Safety Management Certificate (SMC) and that a copy of the Document of Compliance (DOC) is on board;
- (NA) 2.2.1.13 checking the validity of the International Ship Security Certificate;
- (NA) 2.2.1.14 checking, as appropriate, the validity of the Polar Ship Certificate;
- (NA) 2.2.1.15 checking that the ship's complement complies with the Minimum Safe Manning Document (SOLAS 74/00/12 reg.V/14) (SOLAS 74/88 reg.V/13(b));
- (NA) 2.2.1.16 checking that the master, officers and ratings are certificated as required by the STCW Convention;
- (NA) 2.2.1.17 checking whether any new equipment has been fitted and, if so, confirming that it has been approved before installation and that any changes are reflected in the appropriate certificate;

⁷⁵ Refer to the Sample Format of Confirmation of Compliance, Early Submission of the SEEMP part II on the ship fuel oil consumption data collection plan and its timely verification pursuant to regulation 5.4.5 of MARPOL Annex VI (MEPC.1/Circ.876).

(NA)	2.2.1.18	confirming that the Procedures and Arrangements Manual is on board (MARPOL 90/04 Annex II reg.14, and where applicable, Polar Code part II-A ch.2, MEPC.1/Circ.856);
(NA)	2.2.1.19	confirming that the Cargo Record Book or as an electronic record book ⁷⁶ is being correctly used (MARPOL 90/04/19 Annex II reg.15, and where applicable, Polar Code part II-A ch.2);
(NA)	2.2.1.20	confirming that the shipboard marine pollution emergency plan is on board (MARPOL 90/04 Annex II reg.17, and where applicable, Polar Code part II-A ch.2); and
(NA)	2.2.1.21	confirming the availability of the International Anti-fouling System Certificate (AFS 2001 annex 4 reg.2), when applicable.
(NA)	2.2.2	For the carriage of noxious liquid substances in bulk the annual survey should consist of:
(NA)	2.2.2.1	examining externally and confirming that the pumping and piping systems, including a stripping system if fitted, and associated equipment remain as approved (MARPOL 90/04 Annex II reg.12);
(NA)	2.2.2.2	examining externally the tank washing piping and confirming that the type, capacity, number and arrangement of the tank washing machines are as approved (MARPOL 90/04 Annex II reg.14 and App.4);
(NA)	2.2.2.3	examining externally the wash water heating system (MARPOL 90/04 Annex II reg.14 and App.4);
(NA)	2.2.2.4	examining externally, as far as practicable, the underwater discharge arrangements (MARPOL 90/04 Annex II reg.12);
(NA)	2.2.2.5	confirming that the means of controlling the rate of discharge of the residue is as approved (MARPOL 90/04 Annex II reg.14 and App.4);
(NA)	2.2.2.6	confirming that the ventilation equipment for residue removal is as approved (MARPOL 90/04 Annex II reg.13 and App.7);
(NA)	2.2.2.7	examining externally, as far as is accessible, the heating system required for solidifying and high viscosity substances (MARPOL 90/04 Annex II reg.14 and App.4); and
(NA)	2.2.2.8	examining any additional requirements listed on the International Certificate for the Carriage of Noxious Liquid Substances in Bulk.
(NA)	2.2.3	For the carriage of noxious liquid substances in bulk the completion of the annual survey should consist of:
(NA)	2.2.3.1	after a satisfactory survey, endorsing the International Certificate for the Carriage of Noxious Liquid Substances in Bulk; and
(NA)	2.2.3.2	if a survey shows that the condition of a ship or its equipment is unsatisfactory, see part "General" section 4.8.

⁷⁶ Refer to the *Guidelines for the use of electronic record books under MARPOL*(resolution MEPC.312(74)).

- (NIn) **2.3** Intermediate surveys see part "General", section 4.3
- (NIn) 2.3.1 For the carriage of noxious liquid substances in bulk the examination of current certificates and other records should consist of:
- (NIn) 2.3.1.1 the provisions of (NA) 2.2.1.
- (NIn) 2.3.2 For the carriage of noxious liquid substances in bulk the intermediate survey should consist of:
- (NIn) 2.3.2.1 the provisions of (NA) 2.2.2;
- (NIn) 2.3.2.2 verifying from the Cargo Record Book or as an electronic record book⁷⁷ that the pumping and stripping arrangements have been emptying the tanks efficiently and are all in working order (MARPOL 90/04 Annex II regs.12 and 15);
- (NIn) 2.3.2.3 confirming, if possible, that the discharge outlet(s) are in good condition (MARPOL 90/04 Annex II P & A Standards); and
- (NIn) 2.3.2.4 confirming that the ventilation equipment for residue removal is satisfactory and that the pressure in the driving medium for portable fans for ventilation equipment for residue removal can be achieved to give the required fan capacity (MARPOL 90/04 Annex II App.7).
- (NIn) 2.3.3 For the carriage of noxious liquid substances in bulk the completion of the intermediate survey should consist of:
- (NIn) 2.3.3.1 after a satisfactory survey, endorsing the International Certificate for the Carriage of Noxious Liquid Substances in Bulk; and
- (NIn) 2.3.3.2 if a survey shows that the condition of a ship or its equipment is unsatisfactory, see part "General", section 4.8.
- (NR) **2.4 Renewal surveys** see part "General", section 4.4
- (NR) 2.4.1 For the carriage of noxious liquid substances in bulk the examination of current certificates and other records should consist of:
- (NR) 2.4.1.1 the provisions of (NA) 2.2.1, except for the validity of the International Certificate for the Carriage of Noxious Liquid Substances in Bulk.
- (NR) 2.4.2 For the carriage of noxious liquid substances in bulk the renewal survey should consist of:
- (NR) 2.4.2.1 the provisions of (NIn) 2.3.2;
- (NR) 2.4.2.2 confirming that the pumping and stripping systems are satisfactory and that portable pipes or bends in sufficient number, if required, are on board (MARPOL 73/78/90/04 Annex II reg.12);

⁷⁷ Refer to the *Guidelines for the use of electronic record books under MARPOL*(resolution MEPC.312(74)).

- (NR) 2.4.2.3 conducting the water test for assessing the stripping quantity, as required (MARPOL 73/78/90/04 Annex II reg.12 and App.5);
- (NR) 2.4.2.4 confirming that the tank washing machines provided on board are in working order, are those described in the Procedures and Arrangements Manual and are installed in accordance with the approved plans (MARPOL 90/04 Annex II reg.14 and App.4);
- (NR) 2.4.2.5 confirming that the wash water heating system, if required, is installed in accordance with the approved plans and is in working order (MARPOL 90/04 Annex II reg.14 and App.4);
- (NR) 2.4.2.6 confirming that the number and position of tank cleaning openings for portable machines are in accordance with the approved plans (MARPOL 90/04 Annex II reg.14 and App.4);
- (NR) 2.4.2.7 confirming that the underwater discharge outlet(s) are in good condition and are in accordance with the approved plans (MARPOL 90/04 Annex II regs.12, 14 and App.4);
- (NR) 2.4.2.8 verifying by actual test that the discharge rate of the pumps, where a variable rate type is used, can be controlled as specified in the Procedures and Arrangements Manual (MARPOL 90/04 Annex II reg.14 and App.4);
- (NR) 2.4.2.9 confirming that the ventilation equipment for residue removal is installed in accordance with the approved plan and is in working order (MARPOL 90/04 Annex II regs.12, 14 and App.4); and
- (NR) 2.4.2.10 confirming that the heating system for solidifying and high viscosity substances is installed in accordance with the approved plan and is in working order (MARPOL 90/04 Annex II regs.12, 14 and App.4).
- (NR) 2.4.3 For the carriage of noxious liquid substances in bulk the completion of renewal survey should consist of:
- (NR) 2.4.3.1 after a satisfactory survey, issuing the International Certificate for the Carriage of Noxious Liquid Substances in Bulk.
- (S) 3 GUIDELINES FOR SURVEYS FOR THE INTERNATIONAL SEWAGE POLLUTION PREVENTION CERTIFICATE
- (SI) **3.1** Initial surveys see part "General", section 4.1
- (SI) 3.1.1 For sewage pollution prevention the examination of plans and designs should consist of:
- (SI) 3.1.1.1 examining as appropriate the arrangements for the provision of a sewage treatment plant, or of a sewage comminuting and disinfecting system, or of a sewage holding tank (MARPOL Annex IV reg.9);
- (SI) 3.1.1.2 if a sewage treatment plant is fitted, checking that it is type-approved by the Administration in accordance with the appropriate resolution (MARPOL Annex IV, regs.9.1.1 and 9.2.1);

- (SI) 3.1.1.3 if a sewage comminuting and disinfecting system is fitted, checking that it is approved by the Administration and that facilities for the temporary storage of sewage are provided (MARPOL Annex IV, reg.9.1.2);
- (SI) 3.1.1.4 if a sewage holding tank is fitted, checking its capacity having regard to the number of persons on board (MARPOL Annex IV, regs.9.1.3 and 9.2.2);
- (SI) 3.1.1.5 examining the arrangements for the provision of a standard discharge connection (MARPOL Annex IV reg.10); and
- (SI) 3.1.1.6 examining the arrangements for the provision of a pipeline for the discharge of sewage to a reception facility (MARPOL Annex IV reg.10).
- (SI) 3.1.2 For sewage pollution prevention the survey during construction and after installation should consist of:
- (SI) 3.1.2.1 checking externally, as applicable, the sewage treatment plant or the sewage comminuting and disinfecting system, and confirming their operation (MARPOL Annex IV, regs.4.1.1, 9.1.1, 9.1.2 and 9.2.1);
- (SI) 3.1.2.2 if a sewage holding tank is fitted, checking that it has been constructed in a satisfactory manner, and checking that the holding tank has a means to indicate visually the amount of its contents (MARPOL Annex IV, regs.9.1.3 and 9.2.2);
- (SI) 3.1.2.3 confirming that a standard discharge connection is provided (MARPOL Annex IV reg.10); and
- (SI) 3.1.2.4 confirming that a pipeline for the discharge of sewage to a reception facility is provided (MARPOL Annex IV reg.10).
- (SR) **3.2 Renewal surveys** See "General", section 4.5
- (SR) 3.2.1 For sewage pollution prevention the examination of current certificates and other records should consist of:
- (SR) 3.2.1.1 checking the validity, as appropriate, of the Cargo Ship Safety Equipment Certificate, the Cargo Ship Safety Radio Certificate and the Cargo Ship Safety Construction Certificate or the Cargo Ship Safety Certificate or Passenger Ship Safety Certificate;
- (SR) 3.2.1.2 checking the validity of the International Load Line Certificate or International Load Line Exemption Certificate;
- (SR) 3.2.1.3 checking the validity of the International Oil Pollution Prevention Certificate;
- (SR) 3.2.1.4 checking the validity of the International Air Pollution Prevention Certificate;
- (SR) 3.2.1.5 confirming, when appropriate, the validity of the International Energy Efficiency Certificate (MARPOL Annex VI, regs.6.4 and 6.5);

(SR)	3.2.1.6	confirming, when appropriate, that confirmation of compliance for the SEEMP part II is provided to and retained on board the ship (MARPOL Annex VI, reg. 5.4.5); ⁷⁸
(SR)	3.2.1.7	confirming, when appropriate, the validity of the Statements of Compliance related to fuel oil consumption reporting (MARPOL Annex VI, regs.6.6 and 6.7);
(SR)	3.2.1.8	checking the validity of the International Ship Security Certificate;
(SR)	3.2.1.9	checking the certificates of class, if the ship is classed with a classification society;
(SR)	3.2.1.10	checking, when appropriate, the validity of the International Certificate of Fitness for the Carriage of Dangerous Chemicals in Bulk or the Certificate of Fitness for the Carriage of Dangerous Chemical in Bulk;
(SR)	3.2.1.11	checking, when appropriate, the validity of the International Certificate of Fitness for the Carriage of Liquefied Gases in Bulk;
(SR)	3.2.1.12	checking when appropriate the validity of the International Pollution Prevention Certificate for the Carriage of Noxious Liquid Substances in Bulk;
(SR)	3.2.1.13	checking that the ship's complement complies with the Minimum Safe Manning Document (SOLAS 74/00/12 reg.V/14) (SOLAS 74/88 reg.V/13(b));
(SR)	3.2.1.14	checking that the master, officers and ratings are certificated as required by the STCW Convention;
(SR)	3.2.1.15	checking the validity of the Safety Management Certificate (SMC) and that a copy of the Document of Compliance (DOC) is on board, where applicable;
(SR)	3.2.1.16	checking, as appropriate, the validity of the Polar Ship Certificate;
(SR)	3.2.1.17	checking whether any new equipment has been fitted and, if so, confirming that it has been approved before installation and that any changes are reflected in the certificate; and
(SR)	3.2.1.18	confirming the availability of the International Anti-fouling System Certificate (AFS 2001 annex 4 reg.2), when applicable.
(SR)	3.2.2	For sewage pollution prevention the renewal survey should consist of:
(SR)	3.2.2.1	confirming that no change has been made nor any new equipment installed which would affect the validity of the certificate (MARPOL Annex IV reg.4.8);

⁷⁸ Refer to the Sample Format of Confirmation of Compliance, Early Submission of the SEEMP part II on the ship fuel oil consumption data collection plan and its timely verification pursuant to regulation 5.4.5 of MARPOL Annex VI (MEPC.1/Circ.876).

- (SR) 3.2.2.2 examining externally the sewage pollution prevention system and confirming, as far as practicable, its satisfactory operation (MARPOL Annex IV, reg.9);
- (SR) 3.2.2.3 confirming that a procedure for discharge of animal effluent is implemented on board (MARPOL 73/78/07 Annex IV reg.11.1.1, and where applicable, Polar Code part II-A section 4.2); and
- (SR) 3.2.2.4 confirming, for ships where a sewage holding tank is fitted as a sewage system, that an approval for the rate of discharge is available (MARPOL Annex IV, regs.9.1.3 and 11.1.1).
- (SR) 3.2.3 For sewage pollution prevention the completion of the renewal survey should consist of:
- (SR) 3.2.3.1 after a satisfactory survey the International Sewage Prevention Certificate should be issued.
- (A) 4 GUIDELINES FOR SURVEYS FOR THE INTERNATIONAL AIR POLLUTION PREVENTION CERTIFICATE AND THE NO_X TECHNICAL CODE
- (Al) **4.1** Initial surveys see part "General", section 4.1
- (AI) 4.1.1 For air pollution prevention the examination of plans and designs should consist of:
- (AI) 4.1.1.1 examining the arrangements for systems using ozone-depleting substances (MARPOL Annex VI reg.12);
- (AI) 4.1.1.2 examining the arrangements for NO_X emission control, if applicable (MARPOL Annex VI reg.13);
- (AI) 4.1.1.3 examining the arrangements for SO_X and particulate matter control, if applicable (MARPOL Annex VI reg.14);
- (AI) 4.1.1.4 examining, where applicable as an equivalence, the plans and arrangements for the exhaust gas cleaning system⁷⁹ or other technological methods (MARPOL Annex VI reg.4);
- (AI) 4.1.1.5 examining the arrangements for vapour collection systems, if applicable (MARPOL Annex VI reg.15 and MSC/Circ.585); and
- (AI) 4.1.1.6 examining the arrangements for shipboard incinerators, if applicable (MARPOL Annex VI reg.16).
- (AI) 4.1.2 For air pollution prevention the survey should consist of:
- (AI) 4.1.2.1 Ozone-depleting substances (MARPOL Annex VI reg.12):
- (AI) 4.1.2.1.1 confirming, if applicable, the satisfactory installation and operation of systems using ozone-depleting substances;

⁷⁹ Refer to the 2015 Guidelines for exhaust gas cleaning systems (resolution MEPC.259(68)).

- (AI) 4.1.2.1.2 confirming that no installation or equipment containing ozone-depleting substances, other than hydrochlorofluorocarbons, has been installed since 19 May 2005, (MARPOL Annex VI reg.12.3.1);
- (AI) 4.1.2.1.3 confirming that no installation or equipment containing hydrochlorofluorocarbons have been fitted since 1 January 2020 (MARPOL Annex VI reg.12.3.2);
- (AI) 4.1.2.2 Nitrogen oxide emissions from marine diesel engines (MARPOL Annex VI reg.13):
- (AI) 4.1.2.2.1 confirming that all marine diesel engines which are required to be certified are pre-certified in accordance with section 2.2 of the NO_X Technical Code to the required Tier and installed in accordance with the approved duty cycle.
- (AI) 4.1.2.2.1.1 If the engine parameter check method is used:
- (AI) 4.1.2.2.1.1.1 an onboard verification survey in accordance with section 6.2 of the NO_X Technical Code;
- (AI) 4.1.2.2.1.2 If the simplified method is used:
- (AI) 4.1.2.2.1.2.1 an onboard verification survey in accordance with section 6.3 of the NO_X Technical Code;
- (AI) 4.1.2.2.1.3 If the direct measurement and monitoring method is used (for existing ships only):
- (AI) 4.1.2.2.1.3.1 an onboard verification survey, in accordance with section 6.4 of the NO_X Technical Code;
- (AI) 4.1.2.2.1.4 For marine diesel engines with an output of more than 5,000 kW and a per cylinder displacement at or above 90 litres/cylinder installed on ships constructed between 1 January 1990 and 31 December 1999, check⁸⁰ (MARPOL Annex VI reg.13.7.3):
 - .1 whether an approved method exists but is not applicable;
 - .2 whether an approved method is not commercially available at this survey; or
 - .3 that an approved method is installed and, where this is the case, that there is an approved method file,

and apply the verification procedures as given in the approved method file;

.4 or that the engine has been certified, confirming that it operates within the limits set forth for Tier I, Tier II or Tier III;

⁸⁰ Refer to the 2014 Guidelines on the Approved Method Process (resolution MEPC.243(66)).

- (AI) 4.1.2.3 Sulphur oxides and particulate matter (MARPOL Annex VI reg.14):
- (AI) 4.1.2.3.1 confirming, if appropriate, that:
 - .1 satisfactory arrangements are in place for using compliant fuel as required; or
 - .2 satisfactory installation and operation of the fuel switching arrangements are in place when tanks are provided for different grades of fuel, and that a written procedure showing how the fuel oil changeover is done, is available; or
 - .3 satisfactory installation and operation of the exhaust gas cleaning system⁸¹ or other technological methods are examined (MARPOL Annex VI reg.4);
- (AI) 4.1.2.4 Volatile organic compounds (MARPOL Annex VI reg.15) (if applicable):
- (AI) 4.1.2.4.1 confirming the satisfactory installation of the vapour collection piping;
- (AI) 4.1.2.4.2 confirming the satisfactory installation and operation of the means provided to eliminate the collection of condensation in the system, such as drains in low points of the line end;
- (AI) 4.1.2.4.3 confirming the satisfactory installation and operation of the isolation valves at the vapour manifolds;
- (AI) 4.1.2.4.4 confirming that the ends of each line are properly identified as vapour collection lines;
- (AI) 4.1.2.4.5 confirming that the vapour collection flanges are in accordance with the IMO guidelines and industrial standards;
- (AI) 4.1.2.5 Shipboard incinerators (MARPOL Annex VI reg.16) (installed on or after 1 January 2000):
- (AI) 4.1.2.5.1 confirming the satisfactory installation and operation of each incinerator;
- (AI) 4.1.2.5.2 confirming that the manufacturer's name, incinerator model number/type and capacity in heat units per hour are permanently marked on the incinerator.
- (AI) 4.1.3 For air pollution prevention the check that certificates and other relevant documentation have been placed on board should consist of:
- (AI) 4.1.3.1 the provision of (AA) 4.2.2.2 as applicable except (AA) 4.2.2.2.15.
- (AI) 4.1.4 For air pollution prevention the completion of the initial survey should consist of:
- (AI) 4.1.4.1 after a satisfactory survey, issuing the International Air Pollution Prevention Certificate.

⁸¹ Refer to the 2015 Guidelines for exhaust gas cleaning systems (resolution MEPC.259(68)).

- (AA) **4.2 Annual surveys** see "General", section 4.2
- (AA) 4.2.1 For air pollution prevention the examination of current certificates and other records should consist of:
- (AA) 4.2.1.1 checking the validity, as appropriate, of the Cargo Ship Safety Equipment Certificate, the Cargo Ship Safety Radio Certificate and the Cargo Ship Safety Construction Certificate or the Cargo Ship Safety Certificate;
- (AA) 4.2.1.2 checking the validity of the Safety Management Certificate (SMC) and that a copy of the Document of Compliance (DOC) is on board, where applicable;
- (AA) 4.2.1.3 checking the validity of the International Load Line Certificate or International Load Line Exemption Certificate;
- (AA) 4.2.1.4 checking the validity of the International Oil Pollution Prevention Certificate;
- (AA) 4.2.1.5 checking, when appropriate, the validity of the International Pollution Prevention Certificate for the Carriage of Noxious Liquid Substances in Bulk;
- (AA) 4.2.1.6 checking, when appropriate, the validity of the International Sewage Pollution Prevention Certificate;
- (AA) 4.2.1.7 confirming, when appropriate, the validity of the International Energy Efficiency Certificate (MARPOL Annex VI, regs.6.4 and 6.5);
- (AA) 4.2.1.8 confirming, when appropriate, that confirmation of compliance for the SEEMP part II is provided to and retained on board the ship (MARPOL Annex VI, reg. 5.4.5);⁸²
- (AA) 4.2.1.9 confirming, when appropriate, the validity of the Statements of Compliance related to fuel oil consumption reporting (MARPOL Annex VI, regs.6.6 and 6.7);
- (AA) 4.2.1.10 checking the certificates of class, if the ship is classed with a classification society;
- (AA) 4.2.1.11 checking, when appropriate, the validity of the International Certificate of Fitness for the Carriage of Dangerous Chemicals in Bulk or the Certificate of Fitness for the Carriage of Dangerous Chemicals in Bulk;
- (AA) 4.2.1.12 checking that the ship's complement complies with the Minimum Safe Manning Document (SOLAS 74/00/12 reg.V/14) (SOLAS 74/88 reg.V/13(b));

⁸² Refer to the Sample Format of Confirmation of Compliance, Early Submission of the SEEMP part II on the ship fuel oil consumption data collection plan and its timely verification pursuant to regulation 5.4.5 of MARPOL Annex VI (MEPC.1/Circ.876).

- (AA) 4.2.1.13 checking that the master, officers and ratings are certificated as required by the STCW Convention; and
- (AA) 4.2.1.14 checking whether any new equipment has been fitted and, if so, confirming that it has been approved before installation and that any changes are reflected in the appropriate certificate.
- (AA) 4.2.2 For air pollution prevention the annual survey should consist of the following:
- (AA) 4.2.2.1 General:
- (AA) 4.2.2.1.1 confirming that no changes have been made or any new equipment installed which would affect the validity of the certificate;
- (AA) 4.2.2.2 Documentation:
- (AA) 4.2.2.2.1 confirming that there is an ozone-depleting substances record book or an electronic record book⁸³, if applicable (MARPOL Annex VI reg.12.6);
- (AA) 4.2.2.2.2 confirming that there are Engine International Air Pollution Prevention (EIAPP) Certificates for each marine diesel engine, required to be certified, as described in chapter 2.1 of the NO_x Technical Code;
- (AA) 4.2.2.2.3 confirming that there is on board an approved technical file for each marine diesel engine required to be certified;
- (AA) 4.2.2.2.4 confirming that there is a record book or an electronic record book⁸⁴ of engine parameters for each marine diesel engine required to be certified in the case where the engine parameter check method is used as a means of onboard NO_X verification (NO_X Technical Code para.6.2.2);
- (AA) 4.2.2.5 confirming that there is an approved onboard monitoring manual for each marine diesel engine required to be certified in the case where the direct measurement and monitoring method is to be used as a means of onboard NO_X verification (NO_X Technical Code para.6.4.17.1);
- (AA) 4.2.2.2.6 confirming that there is a record taking the form of a logbook or an electronic record book⁸⁵ as prescribed by the Administration, for recording the tier and on/off status (or changes) of marine diesel engines which are certified to both Tier II and Tier III or which are certified to Tier II only at entry into and exit from an emission control area (MARPOL Annex VI reg.13.5.3);
- (AA) 4.2.2.2.7 confirming that there are written procedures covering fuel changeover, where applicable;

⁸³ Refer to the *Guidelines for the use of electronic record books under MARPOL*(resolution MEPC.312(74)).

⁸⁴ Refer to the *Guidelines for the use of electronic record books under MARPOL*(resolution MEPC.312(74)).

⁸⁵ Refer to the *Guidelines for the use of electronic record books under MARPOL*(resolution MEPC.312(74)).

- (AA) 4.2.2.2.8 confirming that there is a record of fuel changeover, where applicable, and that this record should take the form of a logbook or an electronic record book⁸⁶ as prescribed by the Administration (MARPOL Annex VI reg.14.6);⁸⁷
- (AA) 4.2.2.9 confirming that there is for each exhaust gas cleaning system-SO_X (EGCS-SO_X) an approved SO_X Emissions Compliance Plan (SECP), EGC system Technical Manual (ETM-A or ETM-B). Onboard Monitoring Manual (OMM) and EGC Record Book or Electronic Logging System and additionally, if applicable, a SO_X Emissions Compliance Certificate (MARPOL Annex VI reg.4 and MEPC.259(68)) or approved documentation in respect of other technological means of achieving compliance;
- (AA) 4.2.2.2.10 confirming that there is a VOC Management Plan, if required (MARPOL Annex VI reg.15.6);
- (AA) 4.2.2.2.11 confirming that there is a transfer procedure, if required, for the VOC collection system;
- (AA) 4.2.2.2.12 confirming that there is, if required, an IMO Type Approval Certificate for each incinerator on board (MARPOL Annex VI reg.16.6.1);
- (AA) 4.2.2.2.13 confirming that there is an instruction manual for each incinerator if required (MARPOL Annex VI reg.16.7);
- (AA) 4.2.2.2.14 confirming that there are records documenting training of the crew in operating each incinerator, if required;
- (AA) 4.2.2.2.15 confirming that there are the required bunker delivery notes on board and the required fuel oil samples are kept under the ship's control (MARPOL Annex VI reg.18) or other relevant documentation;
- (AA) 4.2.2.3 Systems containing ozone-depleting substances, if fitted:
- (AA) 4.2.2.3.1 confirming that no new installation or equipment containing ozone-depleting substances except those covered by (AA) 4.2.2.3.2 have been fitted to the ship after 19 May 2005 (MARPOL Annex VI reg.12.3.1);
- (AA) 4.2.2.3.2 confirming that no installations containing hydrochlorofluorocarbons have been fitted since 1 January 2020 (MARPOL Annex VI reg.12.3.2);
- (AA) 4.2.2.3.3 examining externally any installation or equipment as far as practicable to ensure satisfactory maintenance and that there are no emissions of ozone-depleting substances;
- (AA) 4.2.2.3.4 confirming through documentary evidence that there has been no deliberate emission of ozone-depleting substance;

⁸⁶ Refer to the *Guidelines for the use of electronic record books under MARPOL*(resolution MEPC.312(74)).

⁸⁷ When not prescribed by the Administration, this information could be contained in the engine-room logbook, the deck logbook, the official logbook, the oil record book or an electronic record book, or a separate logbook solely for this purpose.

- (AA) 4.2.2.4 Nitrogen oxide emissions from each diesel marine diesel engine:
- (AA) 4.2.2.4.1 confirming that each marine diesel engine has been operated as required in accordance with its applicable NO_X emission limit(s);
- (AA) 4.2.2.4.2 confirming that no marine diesel engine has been subject to major conversion in the intervening period;
- (AA) 4.2.2.4.3 if the engine parameter check method is used:
- (AA) 4.2.2.4.3.1 reviewing engine documentation contained in the technical file and the record book or an electronic record book⁸⁸ of engine parameters to check, as far as practicable, engine rating, duty and limitation/restrictions as given in the technical file;
- (AA) 4.2.2.4.3.2 confirming that the engine has not undergone any modifications or adjustments outside the options and ranges permitted in the technical file since the last survey;
- (AA) 4.2.2.4.3.3 conducting survey as detailed in the technical file;
- (AA) 4.2.2.4.4 if the simplified method is used:
- (AA) 4.2.2.4.4.1 reviewing engine documentation contained in the technical file;
- (AA) 4.2.2.4.4.2 confirming that the test procedure is acceptable to the Administration;
- (AA) 4.2.2.4.4.3 confirming that the analysers, engine performance sensors, ambient condition measurement equipment, span check gases and other test equipment are the correct type and have been calibrated in accordance with the NO_X Technical Code;
- (AA) 4.2.2.4.4.4 confirming that the correct test cycle, as defined in the engine's technical file, is used for these onboard confirmation test measurements;
- (AA) 4.2.2.4.4.5 ensuring that a fuel sample is taken during the test and submitted for analysis;
- (AA) 4.2.2.4.4.6 witnessing the test and confirming that a copy of the test report has been submitted for approval on completion of the test;
- (AA) 4.2.2.4.5 if the direct measurement and monitoring method is used:
- (AA) 4.2.2.4.5.1 reviewing the technical file and the onboard monitoring manual to check that the arrangements are as approved;
- (AA) 4.2.2.4.5.2 the procedures to be checked in the direct monitoring and measure method and the data obtained as given in the approved onboard monitoring manual should be followed (NO_X Technical Code para.6.4.16.1);

⁸⁸ Refer to the *Guidelines for the use of electronic record books under MARPOL*(resolution MEPC.312(74)).

- (AA) 4.2.2.4.6 for a marine diesel engine with an output of more than 5,000 kW and a per cylinder displacement at or above 90 litres/cylinder installed on ships constructed between 1 January 1990 and 31 December 1999, checking:⁸⁹ (MARPOL Annex VI reg.13.7.3)
 - .1 whether an approved method exists but is not applicable;
 - .2 whether an approved method is not commercially available at this survey; or
 - .3 that an approved method is installed and where this is the case, that there is an approved method file,

and applying the verification procedures as given in the approved method file;

- .4 or that the engine has been certified, confirming that it operates within the limits set forth for Tier I, Tier II or Tier III;
- (AA) 4.2.2.5 Sulphur oxides and particulate matter:

confirming, if appropriate, that:

- .1 satisfactory arrangements are in place for using compliant fuel as required; or
- .2 satisfactory installation and operation of the fuel switching arrangements are in place when tanks are provided for different grades of fuel, including records of the changeover to and from low sulphur fuel during transit through an emission control area established for SO_X and particulate matter control; or
- .3 satisfactory installation and operation of the exhaust gas cleaning system according to approved documentation, including sensors monitoring operational or emission parameters as set out in the OMM, and confirming that all relevant parameters as set out in the SECP, ETM and OMM are recorded and presented in the form or a report (MARPOL Annex VI reg.4 and MEPC.259(68)); or
- .4 other technological methods are examined (MARPOL Annex VI reg.4);
- (AA) 4.2.2.6 Volatile organic compounds (VOCs):
- (AA) 4.2.2.6.1 confirming that the vapour collect system, if required, is maintained in accordance with its approved arrangement;
- (AA) 4.2.2.6.2 for ships carrying crude oil, confirming the VOC management plan has been implemented as appropriate;

⁸⁹ Refer to the 2014 Guidelines on the Approved Method Process (resolution MEPC.243(66)).

(AA)	4.2.2.7	Incineration:
\/ \/ \/		intointoi autori.

- (AA) 4.2.2.7.1 confirming that prohibited materials have not been incinerated;
- (AA) 4.2.2.7.2 confirming that shipboard incineration of sewage sludge or sludge oil in boilers or marine power plants is not undertaken while the ship is inside ports, harbours or estuaries;
- (AA) 4.2.2.8 Incinerators (installed on or after 1 January 2000):
- (AA) 4.2.2.8.1 confirming that operators have been trained as required;
- (AA) 4.2.2.8.2 confirming from an external examination that each incinerator is in a generally satisfactory condition and free from leaks of gas or smoke;
- (AA) 4.2.2.8.3 confirming that combustion chamber outlet temperatures have been maintained as required;
- (AA) 4.2.2.8.4 confirming that each incinerator is maintained according to its approved arrangement.
- (AA) 4.2.2.9 Fuel oil quality:
- (AA) 4.2.2.9.1 confirming that bunker delivery notes as required conform to the requirements of MARPOL Annex VI, Appendix V;
- (AA) 4.2.2.9.2 confirming that MARPOL samples as required are retained on board and labels duly completed or otherwise retained under the ship's control; and
- (AA) 4.2.2.9.3 confirming that documentation in lieu of that required by (AA) 4.2.2.9.1 or 4.2.2.9.2 is available on board.
- (AA) 4.2.3 For air pollution prevention the completion of the annual survey should consist of:
- (AA) 4.2.3.1 after a satisfactory survey, endorsing the International Air Pollution Prevention certificate; and
- (AA) 4.2.3.2 if a survey shows that the condition of the ship or its equipment is unsatisfactory see "General", section 4.8.
- (Aln) **4.3** Intermediate surveys see "General", section 4.3
- (Aln) 4.3.1 For air pollution prevention the examination of current certificates and other records should consist of:
- (Aln) 4.3.1.1 the provisions of (AA) 4.2.1.
- (Aln) 4.3.2 For air pollution prevention the intermediate survey should consist of:
- (Aln) 4.3.2.1 the provisions of (AA) 4.2.2.
- (Aln) 4.3.3 For air pollution prevention the completion of the intermediate survey should consist of:

- (Aln) 4.3.3.1 after a satisfactory survey, endorsing the International Air Pollution Prevention Certificate; and
- (Aln) 4.3.3.2 if a survey shows that the condition of the ship or its equipment is unsatisfactory see "General", section 4.8.
- (AR) 4.4 Renewal surveys see "General", section 4.5
- (AR) 4.4.1 For air pollution prevention the examination of current certificates and other records should consist of:
- (AR) 4.4.1.1 the provisions of (AA) 4.2.1 except the validity of the International Air Pollution Prevention Certificate.
- (AR) 4.4.2 For air pollution prevention the renewal survey should consist of:
- (AR) 4.4.2.1 the provisions of (AA) 4.2.2;
- (AR) 4.4.2.2 for each incinerator the renewal survey should consist of:
- (AR) 4.4.2.2.1 confirming, if necessary by simulated test or equivalent, the satisfactory operation of the alarms and safety devices.
- (AR) 4.4.3 For air pollution prevention the completion of the renewal survey should consist of:
- (AR) 4.4.3.1 after a satisfactory survey the International Air Pollution Prevention Certificate should be issued.

Annex 4

SURVEY GUIDELINES UNDER THE INTERNATIONAL CONVENTION FOR THE CONTROL AND MANAGEMENT OF SHIPS' BALLAST WATER AND SEDIMENTS, 2004

- (B) 1 GUIDELINES FOR SURVEYS FOR THE INTERNATIONAL BALLAST WATER MANAGEMENT CERTIFICATE
- (BI) **1.1** Initial surveys see part "General" section 2.1
- (BI) 1.1.1 For the control and management of ships' ballast water and sediments the examination of plans and designs should consist of:
- (BI) 1.1.1.1 examining the design and construction (BWM Convention 04, regulation B-5);
- (BI) 1.1.1.2 examining the ballast water management plan (BWM Convention 04, regulation B-1);
- (BI) 1.1.1.3 examining plans for the installation of ballast water management systems (BWMS) (BWM Convention 04/18, regulation D-3/Note: this survey requirement is relevant only when the performance standard according to regulation D-2 is applicable); and
- (BI) 1.1.1.4 if applicable, examining plans for the installation of prototype ballast water treatment technologies (BWM Convention 04, regulation D-4).
- (BI) 1.1.2 For the control and management of ships' ballast water and sediments the survey during construction and after installation should consist of:
- (BI) 1.1.2.1 confirming that the ballast water management plan has been provided (BWM Convention 04, regulation B-1);
- (BI) 1.1.2.2 confirming that the ballast water record book has been provided (BWM Convention 04, regulation B-2);
- (BI) 1.1.2.3 verifying that, if applicable, the prototype ballast water treatment technology installation has been carried out in accordance with the approved programme and that the workmanship of the installation is satisfactory (BWM Convention 04, regulation D-4);
- (BI) 1.1.2.4 confirming that the BWMS(s) installed on or after 28 October 2020 are approved in accordance with the BWMS Code, as may be amended; and confirming that the BWMS(s) installed before 28 October 2020 are approved taking into account the guidelines⁹⁰ developed by the Organization or the BWMS Code, as may be amended (BWM Convention 04/18, regulation D-3, paragraphs 1.1 and 1.2); and
- (BI) 1.1.2.5 confirming that, if applicable, a statement of compliance for a prototype ballast water treatment technology has been provided (BWM Convention 04, regulation D-4).

⁹⁰ Refer to resolutions MEPC.125(53), MEPC.174(58) or MEPC.279(70), as appropriate.

- (BI) 1.1.3 When the performance standard described in regulation D-2 is applicable, the survey after installation of BWMS (also applicable for installations on existing ships) should consist of:
- (BI) 1.1.3.1 confirming that the certificate or certificates for type approval of BWMS(s) are available (BWM Convention 04/18, regulation D-3);
- (BI) 1.1.3.2 confirming that a statement has been provided by the Administration, or by a laboratory authorized by the Administration, confirming that the electrical and electronic components of the BWMS(s) have been type-tested in accordance with the specifications for environmental testing contained in part 3 of the annex of the applicable⁹¹ Guidelines for approval of ballast water management systems (G8) (BWM Convention 04/18, regulation D-3);
- (BI) 1.1.3.3 confirming that equipment manuals for major components of the BWMS(s) in accordance with the relevant resolution^{*} have been provided (BWM Convention 04/18, regulation D-3);
- (BI) 1.1.3.4 confirming that an operations and technical manual for the BWMS(s), specific to the ship and approved, containing a technical description of the ballast water management system(s), operational and maintenance procedures, and backup procedures in case of equipment malfunction,^{*} or an operation, maintenance and safety manual of the BWMS^{*} has been provided (BWM Convention 04/18, regulation D-3);
- (BI) 1.1.3.5 confirming that installation specifications for the BWMS(s), e.g. installation drawing, piping and instrumentation diagrams, have been provided (BWM Convention 04/18, regulation D-3);
- (BI) 1.1.3.6 confirming that installation commissioning procedures for the BWMS(s) have been provided (BWM Convention 04/18, regulation D-3);
- (BI) 1.1.3.7 confirming that initial calibration procedures of the BWMS(s) have been provided⁹²(BWM Convention 04/18, regulation D-3);
- (BI) 1.1.3.8 confirming that a valid calibration certificate is provided on board in accordance with the relevant resolution;⁹³
- (BI) 1.1.3.9 confirming that sampling facilities are provided and arranged such as to collect representative samples of the ship's ballast water from the BWMS(s) intake(s), or from a location before the ballast discharge points and any other points necessary for sampling as applicable⁹⁴ (BWM Convention 04/18, regulation D-3);
- (BI) 1.1. 3.10 verifying that the BWMS installation has been carried out in accordance with the technical installation specification;

⁹⁴ Refer to resolutions MEPC.125(53), MEPC.174(58), MEPC.279(70) or the BWMS Code, where applicable.

⁹¹ Only applicable to the BWMS type-approved in accordance with resolution MEPC.125(53) or MEPC.174(58).

⁹² Only applicable to the BWMS type-approved in accordance with resolution MEPC.125(53) or MEPC.174(58).

⁹³ Applicable to the BWMS type-approved in accordance with resolutions MEPC.125(53), MEPC.174(58), MEPC.279(70) or the BWMS Code.

- (BI) 1.1.3.11 verifying that the BWMS is in conformity with the Type Approval Certificate of BWMS issued by Administration or its representative;
- (BI) 1.1.3.12 verifying that the installation of the complete BWMS has been carried out in accordance with the manufacturer's equipment specification;
- (BI) 1.1.3.13 verifying that any operational inlets and outlets are located in the positions indicated on the drawings of the pumping and piping arrangements;
- (BI) 1.1.3.14 verifying that the workmanship of the installation is satisfactory and, in particular, that any bulkhead penetrations or penetrations of the ballast system piping are to the relevant approved standards;
- (BI) 1.1.3.15 verifying in accordance with the relevant resolution⁹⁵ that the control and monitoring equipment operates correctly;
- (BI) 1.1.3.16 confirming that, if applicable, the ballast water management recording device(s) are operable and that there is a sufficient supply of consumables for the recording device(s) on board (BWM Convention 04/18, regulation D-3);
- (BI) 1.1.3.17 confirming the satisfactory installation and operation of the BWMS(s), including any audible or visual alarms (BWM Convention 04/18, regulation D-3);
- (BI) 1.1.3.18 confirming that, if applicable,⁹⁶ the suitable bypasses or overrides to protect the safety of the ship and personnel are installed and used in the event of an emergency and these shall be connected to the BWMS so that any bypass of the BWMS shall activate an alarm. The bypass event shall be recorded by the control and monitoring equipment and within the ballast water record book;
- (BI) 1.1.3.19 verifying that, if applicable,⁹⁷ installation commissioning procedures have been completed;
- (BI) 1.1.3.20 verifying that the commissioning testing of the BWMS has been carried out after installation in accordance with the applicable guidelines developed by the Organization;⁹⁸
- (BI) 1.1.3.21 confirming that, if applicable, active substances are provided on board (BWM Convention 04/18, regulation D-3); and
- (BI) 1.1.3.22 confirming that, if applicable, dosage instructions for active substances or preparations are available on board (BWM Convention 04/18, regulation D-3).
- (BI) 1.1.4 For the control and management of ships' ballast water and sediments the completion of the initial survey should consist of:

⁹⁵ Only applicable to the BWMS type-approved in accordance with resolution MEPC.125(53) or MEPC.174(58).

⁹⁶ Only applicable to the BWMS type-approved in accordance with resolution MEPC.279(70) or BWMS Code.

⁹⁷ Only applicable to the BWMS type-approved in accordance with resolution MEPC.279(70) or BWMS Code.

⁹⁸ Refer to BWM.2/Circ.70/Rev.1 on 2020 Guidance for the commissioning testing of ballast water management systems.

- (BI) 1.1.4.1 after a satisfactory survey, the International Ballast Water Management Certificate should be issued.
- (BA) 1.2 Annual surveys see part "General" section 2.5
- (BA) 1.2.1 For the control and management of ships' ballast water and sediments the examination of current certificates and other records should consist of:
- (BA) 1.2.1.1 checking the validity, as appropriate, of the Cargo Ship Safety Equipment Certificate, the Cargo Ship Safety Radio Certificate and the Cargo Ship Safety Construction Certificate or the Cargo Ship Safety Certificate or Passenger Ship Safety Certificate;
- (BA) 1.2.1.2 checking the validity of the International Load Line Certificate or International Load Line Exemption Certificate;
- (BA) 1.2.1.3 checking, where appropriate, the validity of the International Oil Pollution Prevention Certificate, International Sewage Pollution Prevention Certificate, and International Air Pollution Prevention Certificate;
- (BA) 1.2.1.4 confirming, when appropriate, the validity of the International Energy Efficiency Certificate (MARPOL Annex VI, regs.6.4 and 6.5);
- (BA) 1.2.1.5 confirming, when appropriate, that confirmation of compliance for the SEEMP part II is provided and retained on board the ship (MARPOL Annex VI, reg. 5.4.5);⁹⁹
- (BA) 1.2.1.6 confirming, when appropriate, the validity of the Statements of Compliance related to fuel oil consumption reporting (MARPOL Annex VI, regs.6.6 and 6.7);
- (BA) 1.2.1.7 checking the validity of the International Ship Security Certificate;
- (BA) 1.2.1.8 checking, when appropriate, the validity of the Safety Management Certificate (SMC) and that a copy of the Document of Compliance (DOC) is on board;
- (BA) 1.2.1.9 checking the certificates of class, if the ship is classed with a classification society;
- (BA) 1.2.1.10 checking, when appropriate, the validity of the International Certificate of Fitness for the Carriage of Dangerous Chemicals in Bulk or the Certificate of Fitness for the Carriage of Dangerous Chemicals in Bulk;
- (BA) 1.2.1.11 checking, when appropriate, the validity of the International Certificate of Fitness for the Carriage of Liquefied Gases in Bulk;
- (BA) 1.2.1.12 checking, when appropriate, the validity of the International Pollution Prevention Certificate for the Carriage of Noxious Liquid Substances in Bulk;
- (BA) 1.2.1.13 checking, when appropriate, the validity of the International Sewage Pollution Prevention Certificate;

⁹⁹ Refer to the Sample Format of Confirmation of Compliance, Early Submission of the SEEMP part II on the ship fuel oil consumption data collection plan and its timely verification pursuant to regulation 5.4.5 of MARPOL Annex VI (MEPC.1/Circ.876).

- (BA) 1.2.1.14 checking that the ship's complement complies with the Minimum Safe Manning Document (SOLAS 74/00/12 reg.V/14) (SOLAS 74/88 reg.V/13(b));
- (BA) 1.2.1.15 checking that the master, officers and ratings are certificated as required by the STCW Convention;
- (BA) 1.2.1.16 checking whether any new equipment has been fitted and, if so, confirm that it has been approved before installation and that any changes are reflected in the certificate;
- (BA) 1.2.1.17 confirming that the ballast water management plan is on board (BWM Convention 04, regulation B-1);
- (BA) 1.2.1.18 checking whether the appropriate entries have been made in the ballast record book (BWM Convention 04, regulation B-2);
- (BA) 1.2.1.19 sighting the certificate(s) for type approval for the BWMS(s) (BWM Convention 04/18, regulation D-3/Note: this survey requirement is relevant only when the performance standard according to regulation D-2 is applicable);
- (BA) 1.2.1.20 sighting the records of the recording device, if fitted (BWM Convention 04/18, regulation D-3/Note: this survey requirement is relevant only when the performance standard according to regulation D-2 is applicable);
- (BA) 1.2.1.21 confirming that a calibration check has been performed in accordance with the approved manual and the relevant resolution;¹⁰⁰
- (BA) 1.2.1.22 sighting, if applicable, the statement of compliance for a prototype ballast water treatment technology (BWM Convention 04, regulation D-4); and
- (BA) 1.2.1.23 verifying, if applicable, that there is evidence that the prototype ballast water treatment technology is continuing to be operated in accordance with the approved programme (BWM Convention 04, regulation D-4).
- (BA) 1.2.2 For the control and management of ships' ballast water and sediments, the annual survey should consist of:
- (BA) 1.2.2.1 the provisions of (BAd) 1.5 where applicable; otherwise
- (BA) 1.2.2.2 examining externally the ballast water treatment system and confirming, as far as practicable, its satisfactory operation (BWM Convention 04/18, regulation D-3/Note: this survey requirement is relevant only when the performance standard according to regulation D-2 is applicable);
- (BA) 1.2.2.3 confirming that, if applicable, active substances in accordance with the manufacturer's recommendations are provided on board (BWM Convention 04/18, regulation D-3/Note: this survey requirement is relevant only when the performance standard according to regulation D-2 is applicable);

¹⁰⁰ Applicable to the BWMS type-approved in accordance with resolutions MEPC.125(53), MEPC.174(58), MEPC.279(70) or BWMS Code.

- (BA) 1.2.2.4 confirming that, if applicable, dosage instructions for active substances or preparations are available on board (BWM Convention 04/18, regulation D-3/ Note: this survey requirement is relevant only when the performance standard according to regulation D-2 is applicable); and
- (BA) 1.2.2.5 where applicable examining externally the prototype ballast water treatment technology and confirming, as far as practicable, its satisfactory operation (BWM Convention 04, regulation D-4).
- (BA) 1.2.3 For the control and management of ships' ballast water and sediments, the completion of the annual survey should consist of:
- (BA) 1.2.3.1 after a satisfactory survey, the International Ballast Water Management Certificate should be endorsed; and
- (BA) 1.2.3.2 if a survey shows that the condition of a ship or its equipment is unsatisfactory, see part "General" section 4.8.
- (Bln) **1.3** Intermediate surveys see part "General" section 2.4
- (BIn) 1.3.1 For the control and management of ships' ballast water and sediments the examination of current certificates and other records should consist of:
- (BIn) 1.3.1.1 the provisions of (BA) 1.2.1.
- (BIn) 1.3.2 For the control and management of ships' ballast water and sediments the intermediate survey should consist of:
- (BIn) 1.3.2.1 the provisions of (BA) 1.2.2; and
- (BIn) 1.3.2.2 examining the BWMS for obvious defects, deterioration or damage including examining associated pumps, piping and fittings for wear and corrosion (BWM Convention 04/18, regulations D-3 and D-4/Note: this survey requirement is relevant only when the performance standard according to regulation D-2 is applicable).
- (BIn) 1.3.3 For the control and management of ships' ballast water and sediments the completion of the intermediate survey should consist of:
- (BIn) 1.3.3.1 after a satisfactory survey, the International Ballast Water Management Certificate should be endorsed; and
- (BIn) 1.3.3.2 if a survey shows that the condition of a ship or its equipment is unsatisfactory, see part "General" section 4.8.
- (BR) **1.4 Renewal surveys** see part "General" section 2.3
- (BR) 1.4.1 For the control and management of ships' ballast water and sediments the examination of current certificates and other records should consist of:
- (BR) 1.4.1.1 the provisions of (BA) 1.2.1, except for the validity of the International Ballast Water Management Certificate.
- (BR) 1.4.2 For the control and management of ships' ballast water and sediments the renewal survey should consist of:

- (BR) 1.4.2.1 the provisions of (BIn) 1.3.2; and
- (BR) 1.4.2.2 if applicable, confirming, if necessary by simulated test or equivalent, the satisfactory operation of the prototype ballast water treatment technology (BWM Convention 04, regulation. D-4).
- (BR) 1.4.3 For the control and management of ships' ballast water and sediments the completion of renewal survey should consist of:
- (BR) 1.4.3.1 after a satisfactory survey, the International Ballast Water Management Certificate should be issued.
- (BAd) **1.5** Additional surveys see part "General" section 2.7.
- (BAd) 1.5.1 For the control and management of ships' ballast water and sediments, the additional survey should consist of:
- (BAd) 1.5.1.1 the provisions of (BI) 1.1.1.3, (BI) 1.1.1.4, (BI) 1.1.2.3, (BI)1.1.2.4, (BI) 1.1.2.5 and (BI)1.1.3; and
- (BAd) 1.5.2 For the control and management of ships' ballast water and sediments the completion of the additional survey should consist of:
- (BAd) 1.5.2.1 after a satisfactory survey, the International Ballast Water Management Certificate should be issued.

Annex 5

SURVEY GUIDELINES UNDER MANDATORY CODES

- (D) 1 GUIDELINES FOR SURVEYS FOR THE INTERNATIONAL CERTIFICATE OF FITNESS FOR THE CARRIAGE OF DANGEROUS CHEMICALS IN BULK AND THE CERTIFICATE OF FITNESS FOR THE CARRIAGE OF DANGEROUS CHEMICALS IN BULK
- (DI) **1.1** Initial surveys see part "General" section 4.1
- (DI) 1.1.1 For compliance with the International Code for the Construction and Equipment of Ships Carrying Dangerous Chemicals in Bulk the examination of plans and designs of the structure, equipment, fittings, arrangements and materials should consist of:
- (DI) 1.1.1.1 determining the products that it is intended that the ship will be permitted to carry and noting the corresponding minimum special requirements (IBC Code 12 ch.17) and any other special requirements (IBC Code 83/90/00/04/19 ch.15);
- (DI) 1.1.1.2 examining the plans for the ship type, location of the cargo tanks, cargo containment, materials of construction, cargo temperature control, cargo tank vent systems, continuous monitoring of the concentration of flammable vapours, environmental control, electrical installations, fire protection and fire extinction, instrumentation and the provision, specification and stowage of the equipment for personnel protection (IBC Code 83/90/00 chs.2, 4, 6, 7, 8, 9, 10, 11, 13 and 14);
- (DI) 1.1.1.3 examining the plans for the freeboard and stability, discharges below the bulkhead deck and survival capability (IBC Code 83/90/00/14 ch.2);
- (DI) 1.1.1.4 examining the plans for the ship arrangements (IBC Code 83/90/00 ch.3);
- (DI) 1.1.1.5 examining the plans for the cargo transfer (IBC Code 83/90/00 ch.5);
- (DI) 1.1.1.6 examining the plans for the mechanical ventilation in the cargo area (IBC Code 83/90/00 ch.12);
- (DI) 1.1.1.7 the provisions of (NI) 2.1.1 in Annex 3;
- (DI) 1.1.1.8 examining, where applicable, the stability instrument (IBC Code 83/90/00/14 ch.2); and
- (DI) 1.1.1.9 examining, when a dispensation from carriage of a stability instrument applies, the alternative means of verification for intact and damage stability (IBC Code 83/90/00/14 ch.2).
- (DI) 1.1.2 For compliance with the International Code for the Construction and Equipment of Ships Carrying Dangerous Chemicals in Bulk, the survey during construction and after installation of the structure, equipment, fittings, arrangements and materials should consist of:

- (DI) 1.1.2.1 confirming that tanks containing cargo or residues of cargo are suitably segregated from accommodation, service and machinery spaces and from drinking water and stores for human consumption, that cargo piping does not pass through any accommodation, service or machinery space other than cargo pump-rooms or pump-rooms and that cargoes are not to be carried in either the fore or the aft peak tank (IBC Code 83/90/00 ch.3);
- (DI) 1.1.2.2 examining the air intakes and openings into the accommodation, service and machinery spaces in relation to the cargo piping and vent systems and their entrances, air inlets and openings in relation to the cargo area (IBC Code 83/90/00 ch.3);
- (DI) 1.1.2.3 examining the arrangements of the cargo pump-rooms (IBC Code 83/90/00 ch.3);
- (DI) 1.1.2.4 examining the accesses to spaces in the cargo area (IBC Code 83/90/00 ch.3);
- (DI) 1.1.2.5 examining the bilge and ballast arrangements and confirming that pumps and pipelines are identified (IBC Code 83/90/00 ch.3);
- (DI) 1.1.2.6 examining, when applicable, the bow or stern loading and unloading arrangements with particular reference to the air inlets and entrances to the accommodation, machinery and service spaces, the electrical equipment, fire-fighting arrangements and means of communication and testing the remote shut down for the cargo pumps (IBC Code 83/90/00 ch.3);
- (DI) 1.1.2.7 confirming that the cargo tank types are arranged and installed in accordance with the approved plans, internally examining the cargo tanks, water ballast tanks and other spaces in the cargo area and pressure testing the boundaries (IBC Code 83/90/00 ch.4);
- (DI) 1.1.2.8 examining the cargo transfer arrangements and confirming that any hoses are suitable for their intended purpose and, where appropriate, type-approved or marked with date of testing (IBC Code 83/90/00 ch.5);
- (DI) 1.1.2.9 examining and testing any cargo heating and cooling systems (IBC Code 83/90/00 ch.7);
- (DI) 1.1.2.10 confirming that the cargo tank vent systems have been installed in accordance with the approved plans (IBC Code 83/90/00 ch.8);
- (DI) 1.1.2.11 confirming that high-level alarms, or overflow control systems or spill valves or other equivalent means provided to control possible liquid rising in the venting system, are operating satisfactorily (IBC Code 83/90/00 ch.8);
- (DI) 1.1.2.12 confirming that suitable provision is made for drainage of vent lines and that no shut-off valves or other means of stoppage, including spectacle or blank flanges, are fitted either to the individual vents or to the header, if the vents are combined or either above or below pressure/vacuum relief valves with closed vent systems (IBC Code 83/90/00 ch.8);

(DI)	1.1.2.13	confirming that suitable provisions are made for primary and secondary means (or alternative measures) for controlled tank venting (IBC Code 83/90/00 ch.8);
(DI)	1.1.2.14	examining the location of the vent outlets in respect of the height above the weather deck or the fore and aft gangway, from the nearest air intakes or openings to accommodation, service and machinery spaces and ignition sources and confirming that any high velocity vents are of the approved type (IBC Code 83/90/00 ch.8);
(DI)	1.1.2.15	examining the cross-sectional area and height of the vent outlets for cargo tank purging with inert gas, as applicable (IBC Code 83/90/00/14 ch.8);
(DI)	1.1.2.16	examining the arrangements for environmental control, including the means of storing or generating and drying an inert gas (IBC Code 83/90/00 ch.9);
(DI)	1.1.2.17	examining the electrical installations and confirming that, when appropriate, special materials have been used and that the electrical equipment installed in hazardous locations, as permitted, is certified by a recognized authority for the cargoes to be carried (IBC Code 83/90/00 ch.10);
(DI)	1.1.2.18	confirming that independent cargo tanks are electrically bonded to the hull and that all gasketed cargo pipe joints and hose connections are electrically bonded (IBC Code 83/90/00 ch.10);
(DI)	1.1.2.19	examining the arrangements for the fire protection and fire extinction (IBC Code 83/90/00 ch.11);
(DI)	1.1.2.20	examining the fixed fire-fighting system for the cargo pump-room and confirming that the installation tests have been satisfactorily completed and that its means of operation are clearly marked (IBC Code 83/90/00 ch.11);
(DI)	1.1.2.21	checking the deck foam system for the cargo area, including the supplies of foam concentrate, and testing that the minimum number of jets of water at the required pressure in the fire main is obtained (see (EI) 1.1.3.1 in Annex 1) when the system is in operation (IBC Code 83/90/00 ch.11);
(DI)	1.1.2.22	examining the system for continuous monitoring of the concentration of flammable vapours and confirming that the installation tests have been satisfactorily completed (IBC Code 83/90/00 ch.11);
(DI)	1.1.2.23	confirming that suitable portable fire-extinguishing equipment for the cargoes to be carried is provided in the cargo area (IBC Code 83/90/00 ch.11);
(DI)	1.1.2.24	examining, and confirming the satisfactory operation of, the arrangements for the mechanical ventilation of spaces in the cargo area normally entered during cargo handling operations (IBC Code 83/90/00 ch.12) and checking in particular that:

(DI) 1.1.2.24.1 it may be controlled from outside the space;

- (DI) 1.1.2.24.2 warning notices concerning its use have been posted;
- (DI) 1.1.2.24.3 it is of the extraction type, with extraction from below the floor plates, unless the space houses electrical motors driving cargo pumps when it should be of the positive pressure type;
- (DI) 1.1.2.24.4 the ducting does not pass through accommodation, machinery and service spaces and that the exhaust ducts are clear of the ventilation inlets and openings to such spaces;
- (DI) 1.1.2.24.5 the electric motors driving ventilation fans are positioned outside the ventilation ducts and the ventilation fans and the ducts, in way of the fans only, are of non-sparking construction in hazardous locations;
- (DI) 1.1.2.25 examining, and confirming the satisfactory operation of, the arrangements for the mechanical ventilation of spaces normally entered, other than those covered by (DI) 1.1.2.24 (IBC Code 83/90/00 ch.12);
- (DI) 1.1.2.26 confirming that double bottoms, cofferdams, duct keels, pipe tunnels, hold spaces and other spaces where cargo may accumulate are capable of being efficiently ventilated to ensure a safe environment when entry into the space is necessary and that, when appropriate, permanent ducting is provided and any ventilation fans comply with (DI) 1.1.2.24.5 (IBC Code 83/90/00 ch.12);
- (DI) 1.1.2.27 examining the intrinsically safe systems and circuits used for measurement, monitoring, control and communication purposes in all hazardous locations (IBC Code 83/90/00 ch.13);
- (DI) 1.1.2.28 checking the provision of equipment for personnel protection (IBC Code 83/90/00 ch.14) and in particular that:
- (DI) 1.1.2.28.1 suitable protective clothing is available for the crew engaged in loading and discharging operations and that suitable storage is provided;
- (DI) 1.1.2.28.2 the required safety equipment and associated breathing apparatus and air supplies and, when appropriate, emergency-escape respiratory and eye protection, are provided and are properly stowed;
- (DI) 1.1.2.28.3 medical first-aid equipment, including stretchers and oxygen resuscitation equipment are provided;
- (DI) 1.1.2.28.4 arrangements have been made for the antidotes for the cargoes actually carried to be on board;
- (DI) 1.1.2.28.5 decontamination arrangements and eyewashes are operational;
- (DI) 1.1.2.28.6 the required gas detection instruments are on board and that arrangements have been made for the supply of the appropriate vapour detection tubes;
- (DI) 1.1.2.28.7 the stowage for cargo samples is satisfactory;
- (DI) 1.1.2.29 the provisions of (NI) 2.1.2 in annex 3; and

- (DI) 1.1.2.30 confirming that sampling points or detector heads are located in suitable positions in order that potentially dangerous leakages are readily detected (IBC Code 07 ch.11.1.4, BCH Code ch.III, E 3.13).
- (DI) 1.1.3 For compliance with the International Code for the Construction and Equipment of Ships Carrying Dangerous Chemicals in Bulk, the check that all the required documentation has been placed on board the ship should consist of:
- (DI) 1.1.3.1 confirming that a loading and stability information booklet, containing details of typical service and ballast conditions, provisions for evaluating other conditions of loading, a summary of the ship's survival capabilities and sufficient information to ensure that the ship is loaded and operated in a safe and seaworthy manner, is available on board (IBC Code 83/90/00/14 ch.2);
- (DI) 1.1.3.2 confirming that damage survival capability information is supplied on the basis of loading information for all anticipated conditions of loading and variations in draught and trim (IBC Code 83/90/00 ch.2);
- (DI) 1.1.3.3 confirming that a table giving the filling ratios for the cargo tanks at various densities has been provided (IBC Code 83/90/00/19 ch.16);
- (DI) 1.1.3.4 confirming that a copy of the International Code for the Construction and Equipment of Ships Carrying Dangerous Chemicals in Bulk, or the equivalent national regulations, has been provided (IBC Code 83/90/00/19 ch.16);
- (DI) 1.1.3.5 confirming that information relating to the chemical and physical properties of the products to be carried has been provided and that provision has been made for the measures to be taken in an accident (IBC Code 83/90/00/19 ch.16);
- (DI) 1.1.3.6 confirming that a manual covering procedures for cargo transfer, tank cleaning, gas freeing, ballasting, etc. has been provided (IBC Code 83/90/00/19 ch.16);
- (DI) 1.1.3.7 the provisions of (NI) 2.1.3 in annex 3;
- (DI) 1.1.3.8 confirming that compatibility information as to material of construction, protective linings and coating is provided on board (IBC Code 83/04 ch.6);
- (DI) 1.1.3.9 confirming, where applicable, that the stability instrument has been approved and is operating satisfactorily (IBC Code 83/90/00/14 ch.2); and
- (DI) 1.1.3.10 confirming, when a dispensation from carriage of a stability instrument applies, that the alternative means of verification for intact and damage stability is recorded on the Certificate of Fitness and is being applied effectively (IBC Code 83/90/00/14 ch.2).
- (DI) 1.1.4 For compliance with the International Code for the Construction and Equipment of Ships Carrying Dangerous Chemicals in Bulk the completion of the initial survey should consist of:

- (DI) 1.1.4.1 after a satisfactory survey issuing the International Certificate of Fitness for the Carriage of Dangerous Chemicals in Bulk.
- (DA) **1.2 Annual surveys** see part "General" section 4.2
- (DA) 1.2.1 For compliance with the International Code for the Construction and Equipment of Ships Carrying Dangerous Chemicals in Bulk and the Code for the Construction and Equipment of Ships Carrying Dangerous Chemicals in Bulk the examination of current certificates and other records should consist of:
- (DA) 1.2.1.1 checking the validity, as appropriate, of the Cargo Ship Safety Equipment Certificate, the Cargo Ship Safety Radio Certificate and the Cargo Ship Safety Construction Certificate or the Cargo Ship Safety Certificate;
- (DA) 1.2.1.2 checking the validity of the Safety Management Certificate (SMC) and that a copy of the Document of Compliance (DOC) is on board;
- (DA) 1.2.1.3 checking the validity of the International Ship Security Certificate;
- (DA) 1.2.1.4 checking the validity of the International Load Line Certificate or International Load Line Exemption Certificate;
- (DA) 1.2.1.5 checking the validity of the International Oil Pollution Prevention Certificate;
- (DA) 1.2.1.6 checking the certificates of class, if the ship is classed with a classification society;
- (DA) 1.2.1.7 checking, when appropriate, the validity of the International Certificate of Fitness for the Carriage of Dangerous Chemicals in Bulk or the Certificate of Fitness for the Carriage of Dangerous Chemicals in Bulk;
- (DA) 1.2.1.8 checking, when appropriate, the validity of the International Sewage Pollution Prevention Certificate;
- (DA) 1.2.1.9 checking, when appropriate, the validity of the International Air Pollution Prevention Certificate;
- (DA) 1.2.1.10 confirming, when appropriate, the validity of the International Energy Efficiency Certificate (MARPOL Annex VI, regs.6.4 and 6.5);
- (DA) 1.2.1.11 confirming, when appropriate, that confirmation of compliance for the SEEMP part II is provided to and retained on board the ship (MARPOL Annex VI, reg. 5.4.5);¹⁰¹
- (DA) 1.2.1.12 confirming, when appropriate, the validity of the Statements of Compliance related to fuel oil consumption reporting (MARPOL Annex VI, regs.6.6 and 6.7);

¹⁰¹ Refer to the Sample Format of Confirmation of Compliance, Early Submission of the SEEMP part II on the ship fuel oil consumption data collection plan and its timely verification pursuant to regulation 5.4.5 of MARPOL Annex VI (MEPC.1/Circ.876).

(DA)	1.2.1.13	checking, when appropriate, the validity of the International Ballast Water Management Certificate;
(DA)	1.2.1.14	checking that the ship's complement complies with the Minimum Safe Manning Document (SOLAS 74/00/12 reg.V/14) (SOLAS 74/88 reg.V/13(b));
(DA)	1.2.1.15	checking that the master, officers and ratings are certificated as required by the STCW Convention;
(DA)	1.2.1.16	checking whether any new equipment has been fitted and, if so, confirming that it has been approved before installation and that any changes are reflected in the appropriate certificate;
(DA)	1.2.1.17	confirming that the loading and stability information booklet, containing details of typical service and ballast conditions, provisions for evaluating other conditions of loading, a summary of the ship's survival capabilities and sufficient information to ensure that the ship is loaded and operated in a safe and seaworthy manner, is available on board (IBC Code 83/90/00/14 ch.2) (BCH Code 85/90/00/14 para.2.2.1);
(DA)	1.2.1.18	confirming that damage survival capability information is supplied on the basis of loading information for all anticipated conditions of loading and variations in draught and trim (IBC Code 83/90/00 ch.2) (No BCH Code 85/90/00 reference);
(DA)	1.2.1.19	confirming that a table giving the filling ratios for the cargo tanks at various densities has been provided (IBC Code 83/90/00/19 ch.16) (BCH Code 85/90/00 ch.III, part G);
(DA)	1.2.1.20	confirming that a copy of the International Code for the Construction and Equipment of Ships Carrying Dangerous Chemicals in Bulk or the Code for the Construction and Equipment of Ships Carrying Dangerous Chemicals in Bulk, or the equivalent national regulations, has been provided (IBC Code 83/90/00/19 ch.16) (BCH Code 85/90/00/19 ch.V);
(DA)	1.2.1.21	confirming that information relating to the chemical and physical properties of the products to be carried has been provided, and that provision has been made for the measures to be taken in an accident (IBC Code 83/90/00/19 ch.16) (BCH Code 85/90/00/19 ch.V);
(DA)	1.2.1.22	confirming that a manual covering procedures for cargo transfer, tank cleaning, gas freeing, ballasting, etc. has been provided (IBC Code 83/90/00/19 ch.16) (BCH Code 85/90/00/19 ch.V);
(DA)	1.2.1.23	confirming that the Procedures and Arrangements Manual is on board (IBC Code 83/90/00 ch.16A) (BCH Code 85/90/00 ch.VA);
(DA)	1.2.1.24	confirming that the shipboard marine pollution emergency plan is on board (MARPOL 04 Annex II, reg.17);

- (DA) 1.2.1.25 confirming that the Cargo Record Book or as an electronic record book¹⁰² is on board and being correctly used (MARPOL 04/19 Annex II, reg.15);
- (DA) 1.2.1.26 confirming that compatibility information as to material of construction, protective linings and coating is provided on board (IBC Code 83/04 ch.6) (BCH Code 85/90/00 ch.II, part G);
- (DA) 1.2.1.27 confirming the availability of the International Anti-fouling System Certificate (AFS 2001 annex 4 reg.2), when applicable.
- (DA) 1.2.1.28 confirming, where applicable, that the approved stability instrument is available on board and operating satisfactorily (IBC Code 83/90/00/14 ch.2); and
- (DA) 1.2.1.29 confirming, when a dispensation from carriage of a stability instrument applies, that the alternative means of verification for intact and damage stability recorded on the Certificate of Fitness is available on board and being applied effectively (IBC Code 83/90/00/14 ch.2).
- (DA) 1.2.2 For compliance with the International Code for the Construction and Equipment of Ships Carrying Dangerous Chemicals in Bulk and the Code for the Construction and Equipment of Ships Carrying Dangerous Chemicals in Bulk the annual survey of the structure, equipment, fittings, arrangements and materials should consist of:
- (DA) 1.2.2.1 confirming that wheelhouse doors and windows, sidescuttles and windows in superstructure and deckhouse ends facing the cargo area are in a satisfactory condition (IBC Code 83/90/00 ch.3) (BCH Code 85/90/00 ch.IIC);
- (DA) 1.2.2.2 confirming that potential sources of ignition in or near the cargo pump-room are eliminated, such as loose gear and combustible materials, that there are no signs of undue leakage and that access ladders are in a satisfactory condition (IBC Code 83/90/00 ch.3) (BCH Code 85/90/00 ch.IIC);
- (DA) 1.2.2.3 confirming that removable pipe lengths or other approved equipment necessary for cargo separation are available in the pump-room and are in a satisfactory condition (IBC Code 83/90/00 ch.3) (BCH Code 85/90/00 ch.IIC);
- (DA) 1.2.2.4 examining all pump-room bulkheads for signs of cargo leakage or fractures and, in particular, the sealing arrangements of all penetrations of pump-room bulkheads (IBC Code 83/90/00 ch.3) (BCH Code 85/90/00 ch.IIC);
- (DA) 1.2.2.5 confirming that the remote operation of the cargo pump bilge system is satisfactory (IBC Code 83/90/00 ch.3) (BCH Code 85/90 ch.IIC);

¹⁰² Refer to the *Guidelines for the use of electronic record books under MARPOL*(resolution MEPC.312(74)).

- (DA) 1.2.2.6 examining the bilge and ballast arrangements and confirming that pumps and pipelines are identified (IBC Code 83/90/00 ch.3) (No BCH Code 85/90/00 reference);
- (DA) 1.2.2.7 confirming, when applicable, that the bow or stern loading and unloading arrangements are in order and testing the means of communication and the remote shut down for the cargo pumps (IBC Code 83/90/00 ch.3) (No BCH Code 85/90/00 reference);
- (DA) 1.2.2.8 examining the cargo transfer arrangements and confirming that any hoses are suitable for their intended purpose and, where appropriate, type-approved or marked with date of testing (IBC Code 83/90/00 ch.5) (BCH Code 85/90/00 ch.IID);
- (DA) 1.2.2.9 examining, when applicable, the cargo heating or cooling systems, including any sampling arrangements, and confirming that the means for measuring the temperature and associated alarms are operating satisfactorily (IBC Code 83/90/00 ch.7) (BCH Code 85/90/00 ch.IIF);
- (DA) 1.2.2.10 examining, as far as practicable, the cargo tank vent system, including the pressure/vacuum valves and secondary means to prevent over- or under-pressure and devices to prevent the passage of flame, and the arrangements of cargo tank purging with inert gas, as applicable (IBC Code 83/90/99/00/14 ch.8,) (BCH Code 85/90/99/00 ch.IIE);
- (DA) 1.2.2.11 examining the gauging devices, high-level alarms and valves associated with overflow control (IBC Code 83/90/00 ch.8) (BCH Code 85/90/00 ch.IIE);
- (DA) 1.2.2.12 confirming that arrangements for sufficient gas to be carried or generated to compensate for normal losses, and that the means provided for monitoring ullage spaces, are satisfactory (IBC Code 83/90/00 ch.9) (BCH Code 85/90/00 ch.1IH);
- (DA) 1.2.2.13 confirming that arrangements are made for sufficient medium to be carried where drying agents are used on air inlets to cargo tanks (IBC Code 83/90/00 ch.9) (BCH Code 85/90/00 ch.IIH);
- (DA) 1.2.2.14 confirming that all electrical equipment in dangerous zones is suitable for such locations, is in satisfactory condition and has been properly maintained (IBC Code 83/90/00 ch.10) (BCH Code 85/90/00 ch.IIIB);
- (DA) 1.2.2.15 examining the fixed fire-fighting system for the cargo pump-room and the deck foam system for the cargo area and confirming that their means of operation are clearly marked (IBC Code 83/90/00 ch.11) (BCH Code 85/90/00 ch.IIIE);
- (DA) 1.2.2.16 confirming that the condition of the portable fire-extinguishing equipment for the cargoes to be carried in the cargo area is satisfactory (IBC Code 83/90/00 ch.11) (BCH Code 85/90/00 ch.IIIE);

(DA) 1.2.2.17 confirming that the system for continuous monitoring of the concentration of flammable vapours is satisfactory (IBC Code 83/90/00 ch.11);
- (DA) 1.2.2.18 examining, as far as practicable, and confirming the satisfactory operation of, the arrangements for the ventilation of spaces normally entered during cargo handling operations and other spaces in the cargo area (IBC Code 83/90/00 ch.12) (BCH Code 85/90/00 ch.IIIA);
- (DA) 1.2.2.19 confirming, as far as practicable, that the intrinsically safe systems and circuits used for measurement, monitoring, control and communication purposes in all hazardous locations are being properly maintained (IBC Code 83/90/00 ch.13) (BCH Code 85/90/00 ch.IIIC);
- (DA) 1.2.2.20 examining the equipment for personnel protection (IBC Code 83/90/00 ch.14) (BCH Code 85/90/00 ch.1IIF) and, in particular, that:
- (DA) 1.2.2.20.1 the protective clothing for crew engaged in loading and discharging operations and its stowage is in a satisfactory condition;
- (DA) 1.2.2.20.2 the required safety equipment and associated breathing apparatus and associated air supplies and, when appropriate, emergency-escape respiratory and eye protection, are in a satisfactory condition and are properly stowed;
- (DA) 1.2.2.20.3 medical first-aid equipment, including stretchers and oxygen resuscitation equipment are in a satisfactory condition;
- (DA) 1.2.2.20.4 arrangements have been made for the antidotes for the cargoes actually carried to be on board;
- (DA) 1.2.2.20.5 decontamination arrangements and eyewashes are operational;
- (DA) 1.2.2.20.6 the required gas detection instruments are on board and arrangements have been made for the supply of the appropriate vapour detection tubes;
- (DA) 1.2.2.20.7 the arrangements for the stowage of cargo samples are satisfactory;
- (DA) 1.2.2.21 the provisions of (NA) 2.2.2 in annex 3; and
- (DA) 1.2.2.22 confirming that sampling points or detector heads are located in suitable positions in order that potentially dangerous leakages are readily detected (IBC Code 07 ch.11.1.4) (BCH Code ch.IIIE 3.13).
- (DA) 1.2.3 For compliance with the International Code for the Construction and Equipment of Ships Carrying Dangerous Chemicals in Bulk and the Code for the Construction and Equipment of Ships Carrying Dangerous Chemicals in Bulk the completion of the annual survey should consist of:
- (DA) 1.2.3.1 after a satisfactory survey, endorsing the International Certificate of Fitness for the Carriage of Dangerous Chemicals in Bulk or the Certificate of Fitness for the Carriage of Dangerous Chemicals in Bulk; and
- (DA) 1.2.3.2 if a survey shows that the condition of a ship or its equipment is unsatisfactory see part "General" section 4.8.
- (DIn) **1.3** Intermediate surveys see part "General", section 4.3

- (DIn) 1.3.1 For compliance with the International Code for the Construction and Equipment of Ships Carrying Dangerous Chemicals in Bulk and the Code for the Construction and Equipment of Ships Carrying Dangerous Chemicals in Bulk the examination of current certificates and other records should consist of:
- (DIn) 1.3.1.1 the provisions of (DA) 1.2.1.
- (DIn) 1.3.2 For compliance with the International Code for the Construction and Equipment of Ships Carrying Dangerous Chemicals in Bulk and the Code for the Construction and Equipment of Ships Carrying Dangerous Chemicals in Bulk the intermediate survey of the structure, equipment, fittings, arrangements and materials should consist of:
- (DIn) 1.3.2.1 the provisions of (DA) 1.2.2;
- (DIn) 1.3.2.2 examining vent line drainage arrangements (IBC Code 83/90/00 ch.8) (BCH Code 85/90/00 ch.IIE);
- (DIn) 1.3.2.3 confirming, where applicable, that pipelines and independent cargo tanks are electrically bonded to the hull (IBC Code 83/90/00 ch.10) (BCH Code 85/90/00 ch.IIIB);
- (DIn) 1.3.2.4 generally examining the electrical equipment and cables in dangerous zones such as cargo pump-rooms and areas adjacent to cargo tanks to check for defective equipment, fixtures and wiring; the insulation resistance of the circuits should be tested and in cases where a proper record of testing is maintained, consideration should be given to accepting recent readings (IBC Code 83/90/00 ch.10) (BCH Code 85/90/00 ch.IIIB);
- (DIn) 1.3.2.5 confirming that spares are provided for cargo area mechanical ventilation fans (IBC Code 83/90/00 ch.12) (BCH Code 85/90/00 ch.IIIA); and
- (DIn) 1.3.2.6 the provisions of (NIn) 2.3.2 in annex 3.
- (DIn) 1.3.3 For compliance with the International Code for the Construction and Equipment of Ships Carrying Dangerous Chemicals in Bulk and the Code for the Construction and Equipment of Ships Carrying Dangerous Chemicals in Bulk the completion of the intermediate survey should consist of:
- (DIn) 1.3.3.1 after a satisfactory survey, endorsing the International Certificate of Fitness for the Carriage of Dangerous Chemicals in Bulk or the Certificate of Fitness for the Carriage of Dangerous Chemicals in Bulk; and
- (DIn) 1.3.3.2 if a survey shows that the condition of a ship or its equipment is unsatisfactory see part "General", section 4.8.
- (DR) **1.4 Renewal surveys** see part "General" section 4.4
- (DR) 1.4.1 For compliance with the International Code for the Construction and Equipment of Ships Carrying Dangerous Chemicals in Bulk and the

Code for the Construction and Equipment of Ships Carrying Dangerous Chemicals in Bulk the examination of current certificates and other records should consist of:

- (DR) 1.4.1.1 the provisions of (DA) 1.2.1, except the International Certificate of Fitness for the Carriage of Dangerous Chemicals in Bulk or the Certificate of Fitness for the Carriage of Dangerous Chemicals in Bulk.
- (DR) 1.4.2 For compliance with the International Code for the Construction and Equipment of Ships Carrying Dangerous Chemicals in Bulk and the Code for the Construction and Equipment of Ships Carrying Dangerous Chemicals in Bulk the renewal survey of the structure, equipment, fittings, arrangements and materials should consist of:
- (DR) 1.4.2.1 the provisions of (DIn) 1.3.3; and
- (DR) 1.4.2.2 the provisions of (NR) 2.4.2 in annex 3.
- (DR) 1.4.3 For compliance with the International Code for the Construction and Equipment of Ships Carrying Dangerous Chemicals in Bulk and the Code for the Construction and Equipment of Ships Carrying Dangerous Chemicals in Bulk the completion of the renewal survey should consist of:
- (DR) 1.4.3.1 after a satisfactory survey, issuing the International Certificate of Fitness for the Carriage of Dangerous Chemicals in Bulk or the Certificate of Fitness for the Carriage of Dangerous Chemicals in Bulk.
- (G) 2 GUIDELINES FOR SURVEYS FOR THE INTERNATIONAL CERTIFICATE OF FITNESS FOR THE CARRIAGE OF LIQUEFIED GASES IN BULK
- (GI) **2.1** Initial surveys see part "General", section 4.1.
- (GI) 2.1.1 For compliance with the International Code for the Construction and Equipment of Ships Carrying Liquefied Gases in Bulk the examination of plans and designs of the structure, equipment, fittings, arrangements and materials should consist of:
- (GI) 2.1.1.1 determining the products that it is intended that the ship will be permitted to carry and noting the corresponding minimum special requirements (IGC Code 83/90/00/14 ch.19);
- (GI) 2.1.1.2 examining the plans for the ship type, cargo containment, control of vapour space within the cargo tanks, vapour detection, gauging, personnel protection, filling limits for cargo tanks and other special requirements (IGC Code 83/90/00/14 chs.2, 4, 6, 13, 14, 15 and 17);
- (GI) 2.1.1.3 examining the plans for the freeboard, and survival capability (IGC Code 83/90/00/14 ch.2; IS Code chs.1, 2 and 3);
- (GI) 2.1.1.4 examining the plans for the ship arrangements (IGC Code 83/90/00/14 ch.3);
- (GI) 2.1.1.5 examining, where applicable, the approved documentation for the alternative design and arrangements for the segregation of the cargo area (IGC Code 83/90/00/14 ch.3; SOLAS 74/00/06 reg.II-2/17);

(GI)	2.1.1.6	examining the plans for the process pressure vessels and liquid, vapour and pressure piping systems (IGC Code 83/90/00/14 chs.5 and 6);					
(GI)	2.1.1.7	examining the plans for the cargo pressure/temperature control (IGC Code 83/90/00/14 ch.7);					
(GI)	2.1.1.8	examining the plans for the cargo tank ventilation systems (IGC Code 83/90/00/14 ch.8);					
(GI)	2.1.1.9	examining the plans for the cargo containment system atmosphere control (IGC Code 83/90/00 ch.9);					
(GI)	2.1.1.10	examining the plans for the electrical installations (IGC Code 83/90/00/14 ch.10);					
(GI)	2.1.1.11	examining the plans for fire protection and fire extinction equipment (IGC Code 83/90/00/14 ch.11);					
(GI)	2.1.1.12	examining the plans for the artificial ventilation in the cargo area (IGC Code 83/90/00/14 ch.12);					
(GI)	2.1.1.13	examining the plans for the instrumentation and automation systems (IGC Code 83/90/00/14 ch.13);					
(GI)	2.1.1.14	examining, when applicable, the plans for the use of cargo as fuel (IGC Code 83/90/00/14 ch.16);					
(GI)	2.1.1.15	examining, where applicable, the stability instrument (IGC Code 83/90/00/14 ch.2); and					
(GI)	2.1.1.16	examining, when a dispensation from carriage of a stability instrument applies, the alternative means of verification for intact and damage stability (IGC Code 83/90/00/14 ch.2).					
(GI)	2.1.2	For compliance with the International Code for the Construction and Equipment of Ships Carrying Liquefied Gases in Bulk the survey during construction and after installation of the structure, equipment, fittings, arrangements and materials should consist of:					
(GI)	2.1.2.1	confirming that the segregation in the cargo area and the arrangement of the accommodation, service and machinery spaces are in accordance with the approved plans (IGC Code 83/90/00/14 ch.3);					
(GI)	2.1.2.2	examining, where applicable, the alternative design and arrangements for the segregation of the cargo area, in accordance with the test and inspection requirements, if any, specified in the approved documentation (IGC Code 83/90/00/14 ch.3; SOLAS 74/00/06 reg.II-2/17);					
(GI)	2.1.2.3	examining the arrangements of the cargo machinery spaces and turret compartments, including their escape routes (IGC Code 83/90/00/14 ch.3);					
(GI)	2.1.2.4	confirming that the manually operated ESD (emergency shutdown) system together with the automatic shutdown of the cargo pumps and compressors are satisfactory (IGC Code 83/90/00/14 chs.5 and 18);					

- (GI) 2.1.2.5 examining the arrangement of the cargo control room (IGC Code 83/90/00/14 ch.3);
- (GI) 2.1.2.6 examining the accesses to spaces in the cargo area (IGC Code 83/90/00/14 ch.3);
- (GI) 2.1.2.7 confirming the arrangements for the air locks (IGC Code 83/90/00/14 ch.3);
- (GI) 2.1.2.8 examining the bilge, ballast and oil fuel arrangements (IGC Code 83/90/00/14 ch.3);
- (GI) 2.1.2.9 examining, when applicable, the bow or stern loading and unloading arrangements with particular reference to the air inlets and entrances to the accommodation, machinery and service spaces, the electrical equipment, fire-fighting arrangements and means of communication between the cargo control room and the shore location (IGC Code 83/90/00/14 ch.3);
- (GI) 2.1.2.10 confirming that the cargo tanks are arranged and installed in accordance with the approved plans, internally examining the cargo tanks, water ballast tanks and other spaces in the cargo area, ensuring that the appropriate non-destructive and pressure testing are carried out (IGC Code 83/90/00/14 ch.4);
- (GI) 2.1.2.11 for containment systems with glued secondary barriers, confirming that a tightness test has been carried out in accordance with the approved procedures of the system manufacturer before and after the initial cool down; where the designer's threshold values are exceeded, confirming that an investigation and additional testing, such as, thermographic or acoustic emission testing, has been carried out (IGC Code 83/90/00/14 ch.4);
- (GI) 2.1.2.12 examining during the initial cool down, loading and discharging of the first cargo, the overall performance of the cargo containment system and confirming that the system is in compliance with the design parameters; for vessels carrying liquefied natural gas, the examination includes witnessing the satisfactory operation of the following systems, if fitted:
- (GI) 2.1.2.12.1 gas detection system;
- (GI) 2.1.2.12.2 cargo control and monitoring systems such as level gauging equipment, temperature sensors, pressure gauges, cargo pumps, compressors, and proper control of cargo heat exchanges, if operating;
- (GI) 2.1.2.12.3 nitrogen generating plant and/or inert gas generator;
- (GI) 2.1.2.12.4 nitrogen pressure control systems for interbarrier, insulation and other annular spaces;
- (GI) 2.1.2.12.5 re-liquefaction plant;
- (GI) 2.1.2.12.6 equipment fitted for the burning of cargo vapours, such as boilers, multi-fuel engines or gas combustion units;
- (GI) 2.1.2.12.7 cofferdam heating systems;

- (GI) 2.1.2.12.8 on-deck cargo piping systems including expansion and supporting arrangements;
- (GI) 2.1.2.12.9 high-level alarms, by witnessing topping-off process for cargo tanks (IGC Code 83/90/00/14 ch.13);
- (GI) 2.1.2.13 examining the cargo containment system for cold spots during, or immediately following, the first loaded voyage (IGC Code 83/90/00/14 ch.4);
- (GI) 2.1.2.14 examining the cargo and process piping, including the expansion arrangements, insulation from the hull structure, pressure relief and drainage arrangements, water curtain protection as appropriate, and carrying out a leak detection test (IGC Code 83/90/00/14 ch.5);
- (GI) 2.1.2.15 confirming that the cargo system valving arrangements are in accordance with the approved plans (IGC Code 83/90/00/14 ch.5);
- (GI) 2.1.2.16 confirming that any liquid and vapour hoses are suitable for their intended purpose and, where appropriate, type-approved or marked with date of testing (IGC Code 83/90/00/14 ch.5);
- (GI) 2.1.2.17 examining the arrangements for the cargo pressure/temperature control including, when fitted, the thermal oxidation systems or any refrigeration system and confirming that any associated safety measures and alarms are satisfactory (IGC Code 83/90/00/14 ch.7);
- (GI) 2.1.2.18 confirming that the cargo tank vent systems, including the pressure relief systems and vacuum protection systems, have been installed in accordance with the approved plans, and that the PRVs are type-approved or marked with date of testing (IGC Code 83/90/00/14 ch.8);
- (GI) 2.1.2.19 examining the arrangements for the cargo containment system atmosphere control and environmental control of spaces surrounding type C independent tanks, including the means of storing or generating and drying an inert gas (IGC Code 83/90/00/14 ch.9);
- (GI) 2.1.2.20 examining the electrical installations with particular reference to the certified safe type equipment fitted in gas-dangerous spaces and zones (IGC Code 83/90/00/14 ch.10);
- (GI) 2.1.2.21 examining the arrangements for the fire protection and fire extinction (IGC Code 83/90/00/14 ch.11);
- (GI) 2.1.2.22 examining the fixed fire-fighting system for the enclosed cargo machinery spaces and the enclosed cargo motor room, and confirming that the installation tests have been satisfactorily completed and that its means of operation is clearly marked (IGC Code 83/90/00/14 ch.11);
- (GI) 2.1.2.23 examining the fire water main with particular reference to the provision of hydrants and isolation arrangements, checking that the two jets of water reach all areas of the cargo and containment area, at the required pressure and testing the remote means of starting one main fire pump (IGC Code 83/90/00/14 ch.11);

- (GI) 2.1.2.24 examining and testing the water spray system for cooling, fire protection and crew protection and confirming that its means of operation is clearly marked (IGC Code 83/90/00/14 ch.11);
- (GI) 2.1.2.25 examining and testing the dry chemical powder fire-extinguishing system for the cargo area, seeing that the fixed piping has been properly installed and has been proved clear and confirming that its means of operation is clearly marked (IGC Code 83/90/00/14 ch.11);
- (GI) 2.1.2.26 examining the appropriate fire-extinguishing system for the enclosed cargo machinery spaces for ships that are dedicated to the carriage of a restricted number of cargoes and the internal water spray system for the turret compartments, and confirming that the installation tests have been satisfactorily completed and that their means of operation is clearly marked (IGC Code 83/90/00/14 ch.11);
- (GI) 2.1.2.27 confirming the provision and examining the disposition of the fire-fighters' outfits including their self-contained compressed air breathing apparatus, and the provision of two-way portable radiotelephone apparatus of an explosion-proof type or intrinsically safe (IGC Code 83/90/00/14 ch.11) (SOLAS 74/00/12 regs.II-2/10.10; FSS Code ch.3);
- (GI) 2.1.2.28 examining, and confirming the satisfactory operation of, the arrangements for the artificial ventilation of spaces in the cargo area normally entered during cargo handling operations (IGC Code 83/90/00/14 ch.12) and checking in particular that:
- (GI) 2.1.2.28.1 it may be controlled from outside the space;
- (GI) 2.1.2.28.2 warning notices concerning its use have been posted;
- (GI) 2.1.2.28.3 it is fixed and is of the negative pressure type, permitting extraction from either the upper or lower parts of the space or from both the upper and lower parts when appropriate, for cargo compressor and pump-rooms and for cargo control rooms when considered to be in hazardous areas;
- (GI) 2.1.2.28.4 it is of the positive pressure type for spaces containing electric motors driving cargo compressors or pumps and other non-hazardous spaces within the cargo area, except those containing inert gas generators;
- (GI) 2.1.2.28.5 exhaust ducts are clear of the ventilation inlets and openings to accommodation spaces, service spaces, control stations and other non-hazardous spaces;
- (GI) 2.1.2.28.6 intakes are arranged to minimize the recycling or hazardous vapours;
- (GI) 2.1.2.28.7 ducts from hazardous areas are not led through accommodation, service and machinery spaces and control stations, except when (GI) 2.1.2.36 applies;
- (GI) 2.1.2.28.8 the electric motors driving ventilation fans are positioned outside the ventilation ducts when the carriage of flammable products is intended

and the ventilation fans and the ducts, in way of the fans only, are of non-sparking construction in hazardous areas;

- (GI) 2.1.2.29 examining, and confirming the satisfactory operation of, the arrangements for the artificial ventilation of spaces normally entered other than those covered by (GI) 2.1.2.28 (IGC Code 83/90/00/14 ch.12);
- (GI) 2.1.2.30 examining, and testing as appropriate, the liquid level indicators, overflow control, pressure gauges, high pressure and, when applicable, low pressure alarms, and temperature indicating devices for the cargo tanks (IGC Code 83/90/00/14 ch.13);
- (GI) 2.1.2.31 examining, and testing as appropriate, the permanently installed gas detection equipment (IGC Code 83/90/00/14 ch.13);
- (GI) 2.1.2.32 examining, and testing as appropriate, the oxygen-deficiency monitoring equipment (IGC Code 14 ch.13);
- (GI) 2.1.2.33 confirming that two sets of portable gas detection equipment suitable for the cargoes to be carried and a suitable instrument for measuring oxygen levels have been provided (IGC Code 83/90/00/14 ch.13);
- (GI) 2.1.2.34 examining, as appropriate, the automation systems used to provide instrumented control, monitoring/alarm or safety functions (IGC Code 14 ch.13);
- (GI) 2.1.2.35 checking the provision of equipment for personnel protection (IGC Code 83/90/00/14 ch.14) and, in particular, that:
- (GI) 2.1.2.35.1 suitable protective equipment, including eye protection, is provided for protection of crew members engaged in normal cargo operations, and properly stowed;
- (GI) 2.1.2.35.2 sufficient, but not less than three complete sets of safety equipment each permitting personnel to enter and work in a gas-filled space are provided and are properly stowed;
- (GI) 2.1.2.35.3 an adequate supply of compressed air is provided and that the spare air bottle, air compressor and charging manifold are provided and properly stowed;
- (GI) 2.1.2.35.4 a stretcher and the medical first-aid equipment, including oxygen resuscitation equipment, when available, for the products to be carried are provided;
- (GI) 2.1.2.35.5 respiratory and eye protection suitable for emergency escape purposes are provided and properly stowed;
- (GI) 2.1.2.35.6 decontamination arrangements and eyewashes are operational;
- (GI) 2.1.2.36 examining, when applicable, the arrangements for the use of cargo as fuel and testing that the gas supply to the space containing gas consumers is cut off should the double-wall concentric pipes lose the inert gas pressure or the exhaust ventilation not be functioning correctly, and that the master

gas fuel valve may be manually closed from within the space, and at least one remote location (IGC Code 83/90/00/14 ch.16).

- (GI) 2.1.3 For compliance with the International Code for the Construction and Equipment of Ships Carrying Liquefied Gases in Bulk the check that all the required documentation has been placed on board the ship should consist of:
- (GI) 2.1.3.1 confirming that a loading and stability information booklet, containing details of typical service and ballast conditions, provisions for evaluating other conditions of loading, a summary of the ship's survival capabilities and sufficient information to ensure that the ship is loaded and operated in a safe and seaworthy manner, is available on board (IGC Code 83/90/00/14 ch.2; IS Code chs.1, 2 and 3);
- (GI) 2.1.3.2 confirming that damage survival capability information is supplied on the basis of loading information for all anticipated conditions of loading and variations in draught and trim (IGC Code 83/90/00/14 ch.2);
- (GI) 2.1.3.3 confirming that, where applicable, the approved documentation for the alternative design and arrangements for the segregation of the cargo area is on board (IGC Code 83/90/00/14 ch.3; SOLAS 74/00/06 reg.II-2/17);
- (GI) 2.1.3.4 confirming that, where applicable, the evaluation certificate¹⁰³ for the adequacy of type C tank vent systems is provided (IGC Code 83/90/00/14 ch.8);
- (GI) 2.1.3.5 confirming that the approved document for the maximum allowable loading limits together with PRVs setting pressures is on board (IGC Code 83/90/00/14 ch.15);
- (GI) 2.1.3.6 confirming that necessary information for the safe carriage of the products to be carried has been provided (IGC Code 83/90/00/14 ch.18);
- (GI) 2.1.3.7 confirming that the approved cargo operations manuals, including relevant procedures for ESD system and emergency isolating operations of PRVs, has been provided (IGC Code 14 ch.18);
- (GI) 2.1.3.8 confirming that a copy of the International Code for the Construction and Equipment of Ships Carrying Liquefied Gases in Bulk, or the equivalent national regulations, has been provided (IGC Code 83/90/00/14 ch.18);
- (GI) 2.1.3.9 confirming, where applicable, that the stability instrument has been approved and is operating satisfactorily (IGC Code 83/90/00/14 ch.2); and
- (GI) 2.1.3.10 confirming, when a dispensation from carriage of a stability instrument applies, that the alternative means of verification for intact and damage stability is recorded on the Certificate of Fitness and is being applied effectively (IGC Code 83/90/00/14 ch.2).

¹⁰³ Refer to the *Guidelines for the evaluation of the adequacy of type C tank vent systems* (resolution A.829(19)).

- (GI) 2.1.4 For compliance with the International Code for the Construction and Equipment of Ships Carrying Liquefied Gases in Bulk the completion of the initial survey should consist of:
- (GI) 2.1.4.1 after a satisfactory survey, issuing the International Certificate of Fitness for the Carriage of Liquefied Gases in Bulk.
- (GA) **2.2 Annual surveys** see part "General", section 4.2.
- (GA) 2.2.1 For compliance with the International Code for the Construction and Equipment of Ships Carrying Liquefied Gases in Bulk the examination of current certificates and other records should consist of:
- (GA) 2.2.1.1 checking the validity, as appropriate, of the Cargo Ship Safety Equipment Certificate, the Cargo Ship Safety Radio Certificate and the Cargo Ship Safety Construction Certificate or the Cargo Ship Safety Certificate;
- (GA) 2.2.1.2 checking the validity of the Safety Management Certificate (SMC) and that a copy of the Document of Compliance (DOC) is on board;
- (GA) 2.2.1.3 checking the validity of the International Ship Security Certificate;
- (GA) 2.2.1.4 checking the validity of the International Load Line Certificate or International Load Line Exemption Certificate;
- (GA) 2.2.1.5 checking the validity of the International Oil Pollution Prevention Certificate;
- (GA) 2.2.1.6 checking the certificates of class, if the ship is classed with a classification society;
- (GA) 2.2.1.7 checking the validity of the International Certificate of Fitness for the Carriage of Liquefied Gases in Bulk;
- (GA) 2.2.1.8 checking, when appropriate, the validity of the International Sewage Pollution Prevention Certificate;
- (GA) 2.2.1.9 checking, when appropriate, the validity of the International Air Pollution Prevention Certificate;
- (GA) 2.2.1.10 confirming, when appropriate, the validity of the International Energy Efficiency Certificate (MARPOL Annex VI, regs.6.4 and 6.5);
- (GA) 2.2.1.11 confirming, when appropriate, that confirmation of compliance for the SEEMP part II is provided to and retained on board the ship (MARPOL Annex VI, reg. 5.4.5);¹⁰⁴

¹⁰⁴ Refer to the Sample Format of Confirmation of Compliance, Early Submission of the SEEMP part II on the ship fuel oil consumption data collection plan and its timely verification pursuant to regulation 5.4.5 of MARPOL Annex VI (MEPC.1/Circ.876).

- (GA) 2.2.1.12 confirming, when appropriate, the validity of the Statements of Compliance related to fuel oil consumption reporting (MARPOL Annex VI, regs.6.6 and 6.7);
- (GA) 2.2.1.13 checking, when appropriate, the validity of the International Ballast Water Management Certificate;
- (GA) 2.2.1.14 checking that the ship's complement complies with the Minimum Safe Manning Document (SOLAS 74/00/12 reg.V/14) (SOLAS 74/88 reg.V/13(b));
- (GA) 2.2.1.15 checking that the master, officers and ratings are certificated as required by the STCW Convention;
- (GA) 2.2.1.16 checking whether any new equipment has been fitted and, if so, confirming that it has been approved before installation and that any changes are reflected in the appropriate certificate;
- (GA) 2.2.1.17 confirming that the loading and stability information booklet, containing details of typical service and ballast conditions, provisions for evaluating other conditions of loading, a summary of the ship's survival capabilities and sufficient information to ensure that the ship is loaded and operated in a safe and seaworthy manner, is available on board (IGC Code 83/90/00/14 ch.2; IS Code chs.1, 2 and 3);
- (GA) 2.2.1.18 confirming that damage survival capability information is supplied on the basis of loading information for all anticipated conditions of loading and variations in draught and trim (IGC Code 83/90/00/14 ch.2);
- (GA) 2.2.1.19 confirming that, where applicable, the approved documentation for the alternative design and arrangements for the segregation of the cargo area is on board (IGC Code 83/90/00/14 ch.3; SOLAS 74/00/06 reg.II-2/17);
- (GA) 2.2.1.20 confirming that, where applicable, the evaluation certificate¹⁰⁵ for the adequacy of type C tank vent systems is provided (IGC Code 83/90/00/14 ch.8);
- (GA) 2.2.1.21 checking the logbook entries to confirm whether any changes were made in setting the pressure of PRVs or any emergency isolation action effected in the event of a failure of a cargo tank-installed PRV, and confirming that signs are posted in the cargo control room, if provided, and at each PRV (IGC 83/90/00/14 ch.8);
- (GA) 2.2.1.22 confirming that the approved document for the maximum allowable loading limits together with PRVs setting pressures is on board (IGC Code 83/90/00/14 ch.15);
- (GA) 2.2.1.23 confirming that necessary information for the safe carriage of the products to be carried has been provided (IGC Code 83/90/00/14 ch.18);

¹⁰⁵ Refer to the *Guidelines for the evaluation of the adequacy of type C tank vent systems* (resolution A.829(19)).

(GA)	2.2.1.24	confirming that the approved cargo operations manuals, including relevant procedures for ESD system and emergency isolating
		operations of PRVs, have been provided (IGC Code 14 ch.18);
(GA)	2.2.1.25	confirming that a copy of the International Code for the Construction and Equipment of Ships Carrying Liquefied Gases in Bulk, or the equivalent national regulations, has been provided (IGC Code 83/90/00/14 ch.18);
(GA)	2.2.1.26	confirming that there are records of the performance of the cargo containment system (IGC Code 83/90/00/14 ch.4);
(GA)	2.2.1.27	confirming the availability of the International Anti-fouling System Certificate (AFS 2001 annex 4 reg.2), when applicable;
(GA)	2.2.1.28	confirming, where applicable, the approved stability instrument is available on board and operating satisfactorily (IGC Code 83/90/00/14, ch.2); and
(GA)	2.2.1.29	confirming, when a dispensation from carriage of a stability instrument applies, that the alternative means of verification for intact and damage stability recorded on the Certificate of Fitness is available on board and being applied effectively (IGC Code 83/90/00/14, ch.2).
(GA)	2.2.2	For compliance with the International Code for the Construction and Equipment of Ships Carrying Liquefied Gases in Bulk the annual survey of the structure, equipment, fittings, arrangements and materials should consist of:
(GA)	2.2.2.1	confirming that any special arrangements to survive conditions of damage are in order (IGC Code 83/90/00/14 ch.2);
(GA)	2.2.2.2	examining, where applicable, the alternative design and arrangements for the segregation of the cargo area, in accordance with the test, inspection and maintenance requirements, if any, specified in the approved documentation (IGC Code 83/90/00/14 ch.3; SOLAS 74/00/06 reg.II-2/17);
(GA)	2.2.2.3	confirming that the wheelhouse doors and windows, sidescuttles and windows in superstructure and deckhouse ends in the cargo area are in a satisfactory condition (IGC Code 83/90/00/14 ch.3);
(GA)	2.2.2.4	examining the cargo machinery spaces and turret compartments, including their escape routes (IGC Code 83/90/00/14 ch.3);
(GA)	2.2.2.5	confirming that the manually operated ESD (emergency shutdown) system together with the automatic shutdown of the cargo pumps and compressors are satisfactory (IGC Code 83/90/00/14 ch.5 and 18);
(GA)	2.2.2.6	examining the cargo control room (IGC Code 83/90/00/14 ch.3);
(GA)	2.2.2.7	examining the gas detection arrangements for cargo control rooms and the measures taken to exclude ignition sources where such spaces are classified as hazardous areas (IGC Code 83/90/00/14 ch.3);
(GA)	2.2.2.8	confirming that the arrangements for the air locks are being properly maintained (IGC Code 83/90/00/14 ch.3);

- (GA) 2.2.2.9 examining, as far as practicable, the bilge, ballast and oil fuel arrangements (IGC Code 83/90/00/14 ch.3);
- (GA) 2.2.2.10 examining, when applicable, the bow or stern loading and unloading arrangements with particular reference to the electrical equipment, fire-fighting arrangements and means of communication between the cargo control room and the shore location (IGC Code 83/90/00/14 ch.3);
- (GA) 2.2.2.11 confirming that the sealing arrangements at the gas domes are satisfactory (IGC Code 83/90/00/14 ch.4);
- (GA) 2.2.2.12 confirming that portable or fixed drip trays or deck insulation for cargo leakage are in order (IGC Code 83/90/00/14 ch.4);
- (GA) 2.2.2.13 examining the cargo and process piping, including the expansion arrangements, insulation from the hull structure, pressure relief and drainage arrangements and water curtain protection as appropriate (IGC Code 83/90/00 ch.5);
- (GA) 2.2.2.14 confirming that the cargo tank and interbarrier space pressure and relief valves, including safety systems and alarms, are satisfactory (IGC Code 83/90/00/14 ch.5);
- (GA) 2.2.2.15 confirming that any liquid and vapour hoses are suitable for their intended purpose and, where appropriate, type-approved or marked with date of testing (IGC Code 83/90/00/14 ch.5);
- (GA) 2.2.2.16 examining the arrangements for the cargo pressure/temperature control including, when fitted, the thermal oxidation systems and any refrigeration system, and confirming that any associated safety measures and alarms are satisfactory (IGC Code 83/90/00/14 ch.7);
- (GA) 2.2.2.17 examining the cargo, bunker, ballast and vent piping systems, including PRVs, vacuum relief valves, vent masts and protective screens, as far as practicable, and confirming that the PRVs are type-approved or marked with date of testing (IGC Code 83/90/00/14 chs.5 and 8);
- (GA) 2.2.2.18 confirming that arrangements are made for sufficient inert gas to be carried to compensate for normal losses and that means are provided for monitoring the spaces (IGC Code 83/90/00/14 ch.9);
- (GA) 2.2.2.19 confirming that the use of inert gas has not increased beyond that needed to compensate for normal losses by examining records of inert gas usage (IGC Code 83/90/00/14 ch.9);
- (GA) 2.2.2.20 confirming that any air-drying system and any interbarrier and hold space purging inert gas system are satisfactory (IGC Code 83/90/00/14 ch.9);
- (GA) 2.2.2.21 confirming that electrical equipment hazardous areas is in a satisfactory condition and is being properly maintained (IGC Code 83/90/00/14 ch.10);
- (GA) 2.2.2.22 examining the arrangements for the fire protection and fire extinction and testing the remote means of starting one main fire pump (IGC Code 83/90/00/14 ch.11);

- (GA) 2.2.2.23 examining the fixed fire-fighting system for enclosed cargo machinery spaces and for the enclosed cargo motor room within the cargo area, and confirming that its means of operation is clearly marked (IGC Code 83/90/00/14 ch.11);
- (GA) 2.2.2.24 examining the water spray system for cooling, fire protection and crew protection and confirming that its means of operation is clearly marked (IGC Code 83/90/00/14 ch.11);
- (GA) 2.2.2.25 examining the dry chemical powder fire-extinguishing system for the cargo area and confirming that its means of operation is clearly marked (IGC Code 83/90/00/14 ch.11);
- (GA) 2.2.2.26 examining the appropriate fire-extinguishing system for the enclosed cargo machinery spaces for ships that are dedicated to the carriage of a restricted number of cargoes and the internal water spray system for the turret compartments, and confirming that their means of operation is clearly marked (IGC Code 83/90/00/14 ch.11);
- (GA) 2.2.2.27 confirming the provision and examining the condition of the fire-fighters' outfits including their self-contained compressed air breathing apparatus, and the provision of two-way portable radiotelephone apparatus of an explosion-proof type or intrinsically safe (IGC Code 83/90/00/14 ch.11) (SOLAS 74/00/12 regs.II-2/10.10;FSS Code ch.3);
- (GA) 2.2.2.28 examining, as far as practicable, and confirming the satisfactory operation of, the arrangements for the artificial ventilation of spaces in the cargo area normally entered during cargo handling operations (IGC Code 83/90/00/14 ch.12);
- (GA) 2.2.2.29 examining, and confirming the satisfactory operation of, the arrangements for the artificial ventilation of spaces normally entered other than those covered by (GI) 2.1.2.27 (IGC Code 83/90/00/14 ch.12);
- (GA) 2.2.2.30 examining, and testing as appropriate and as far as practicable, the liquid level indicators, overflow control, pressure gauges, high pressure and, when applicable, low pressure alarms, and temperature indicating devices for the cargo tanks (IGC Code 83/90/00/14 ch.13);
- (GA) 2.2.2.31 examining, and testing as appropriate, the gas detection equipment (IGC Code 83/90/00/14 ch.13);
- (GA) 2.2.2.32 examining, and testing as appropriate, the oxygen-deficiency monitoring equipment (IGC Code 14 ch.13);
- (GA) 2.2.2.33 confirming that two sets of portable gas detection equipment suitable for the cargoes to be carried and a suitable instrument for measuring oxygen levels have been provided (IGC Code 83/90/00/14 ch.13);
- (GA) 2.2.2.34 examining, as appropriate, the automation systems used to provide instrumented control, monitoring/alarm or safety functions (IGC Code 14 ch.13);
- (GA) 2.2.2.35 checking the provision of equipment for personnel protection (IGC Code 83/90/00/14 ch.14) and in particular that:

- (GA) 2.2.2.35.1 suitable protective equipment, including eye protection, is provided for protection of crew members engaged in normal cargo operations, and properly stowed;
- (GA) 2.2.2.35.2 sufficient, but not less than three complete sets of safety equipment each permitting personnel to enter and work in a gas-filled space are provided and are properly stowed;
- (GA) 2.2.2.35.3 an adequate supply of compressed air is provided and that the spare air bottle, air compressor and charging manifold are provided and properly stowed;
- (GA) 2.2.2.35.4 a stretcher and the medical first-aid equipment, including oxygen resuscitation equipment, when available, for the products to be carried, are provided;
- (GA) 2.2.2.35.5 respiratory and eye protection suitable for emergency escape purposes are provided and properly stowed; and
- (GA) 2.2.2.35.6 decontamination arrangements and eyewashes are operational; and
- (GA) 2.2.2.36 examining, when applicable, the arrangements for the use of cargo as fuel and testing, as far as practicable, that the gas supply to the space containing gas consumers is cut off should the double wall concentric pipes lose the inert gas pressure or the exhaust ventilation not be functioning correctly and that master gas fuel valve may be manually closed from within the space and at least one remote location (IGC Code 83/90/00/14 ch.16).
- (GA) 2.2.3 For compliance with the International Code for the Construction and Equipment of Ships Carrying Liquefied Gases in Bulk the completion of the annual survey should consist of:
- (GA) 2.2.3.1 after a satisfactory survey, endorsing the International Certificate of Fitness for the Carriage of Liquefied Gases in Bulk; and
- (GA) 2.2.3.2 if a survey shows that the condition of a ship or its equipment is unsatisfactory see part "General" section 4.8.
- (Gln) **2.3** Intermediate surveys see part "General", section 4.3.
- (GIn) 2.3.1 For compliance with the International Code for the Construction and Equipment of Ships Carrying Liquefied Gases in Bulk the examination of current certificates and other records should consist of:
- (GIn) 2.3.1.1 the provisions of (GA) 2.2.1.
- (GIn) 2.3.2 For compliance with the International Code for the Construction and Equipment of Ships Carrying Liquefied Gases in Bulk the intermediate survey of the structure, equipment, fittings, arrangements and materials should consist of:
- (GIn) 2.3.2.1 the provisions of (GA) 2.2.2;
- (GIn) 2.3.2.2 confirming, where applicable, that pipelines and independent cargo tanks are electrically bonded to the hull (IGC Code 83/90/00/14 ch.10);

(GIn)	2.3.2.3	generally examining the electrical equipment and cables in hazardous areas and zones such as cargo machinery spaces and areas adjacent to cargo tanks to check for defective equipment, fixtures and wiring; the insulation resistance of the circuits should be tested and in cases where a proper record of testing is maintained consideration should be given to accepting recent readings (IGC Code 83/90/00/14 ch.10);						
(GIn)	2.3.2.4	confirming that spares are provided for cargo area mechanical ventilation fans (IGC Code 83/90/00/14 ch.12);						
(GIn)	2.3.2.5	confirming that the heating arrangements, if any, for steel structures are satisfactory; and						
(GIn)	2.3.2.6	confirming that the high-level alarms of cargo tanks are properly working, by witnessing topping-off process for cargo tanks, at the first occasion of full loading after each dry-docking (IGC Code 14 ch.13).						
(GIn)	2.3.3	For compliance with the International Code for the Construction and Equipment of Ships Carrying Liquefied Gases in Bulk the completion of the intermediate survey should consist of:						
(GIn)	2.3.3.1	after a satisfactory survey, endorsing the International Certificate of Fitness for the Carriage of Liquefied Gases in Bulk; and						
(GIn)	2.3.3.2	if a survey shows that the condition of a ship or its equipment is unsatisfactory – see part "General" section 4.8.						
(GR)	2.4	Renewal surveys – see part "General", section 4.4.						
(GR)	2.4.1	For compliance with the International Code for the Construction and Equipment of Ships Carrying Liquefied Gases in Bulk the examination of current certificates and other records should consist of:						
(GR)	2.4.1.1	the provisions of (GA) 2.2.1, except the International Certificate of Fitness for the Carriage of Liquefied Gases in Bulk.						
(GR)	2.4.2	For compliance with the International Code for the Construction and Equipment of Ships Carrying Liquefied Gases in Bulk the renewal survey of the structure, equipment, fittings, arrangements and materials should consist of:						
(GR)	2.4.2.1	the provisions of (GIn) 2.3.2;						
(GR)	2.4.2.2	examining the insulation and means of support of the cargo tanks and confirming that the secondary barrier remains effective (IGC Code 83/90/00/14 ch.4); and						
(GR)	2.4.2.3	conducting an internal examination of selected cargo tanks (IGC Code 83/90/00/14 ch.4);						

- (GR) 2.4.2.4 confirming that the high-level alarms of cargo tanks are properly working, by witnessing topping-off process for cargo tanks, at the first occasion of full loading after each dry-docking¹⁰⁶ (IGC Code 83/90/00/14 ch.13).
- (GR) 2.4.3 For compliance with the International Code for the Construction and Equipment of Ships Carrying Liquefied Gases in Bulk the completion of the renewal survey should consist of:
- (GR) 2.4.3.1 after a satisfactory survey, issuing the International Certificate of Fitness for the Carriage of Liquefied Gases in Bulk.
- (W) 3 GUIDELINES FOR SURVEYS FOR THE POLAR SHIP CERTIFICATE ADDITIONAL TO SOLAS CERTIFICATES
- (WI) **3.1** Initial surveys see part "General" paragraph 5.13.3.1
- (WI) 3.1.1 For compliance with part I-A of the International Code for Ships Operating in Polar Waters the examination of plans and designs of the hull, machinery and equipment should consist of:
- (WI) 3.1.1.1 for ships intended to operate in low air temperature, confirming that the design service temperature of the systems and equipment required by this Code are consistent with the polar service temperature specified for the ship (Polar Code part I-A/para.1.4.2);
- (WI) 3.1.1.2 for ships operating in low air temperature, confirming that the design maximum service time of the survival systems and equipment are consistent with the maximum expected rescue time of the vessel at polar service temperature (Polar Code part I-A/para.1.4.3);
- (WI) 3.1.1.3 reviewing the operational assessment of the ship and its equipment (Add. to PI, (Polar Code part I-A section 1.5); CI, EI, RI)
- (WI) 3.1.1.4 examining the plans and designs for materials of exposed structures and (Add. to PI, scantlings of the ship, confirming that the materials and the scantlings are according to the polar service temperature and ice strengthening standards, where applicable for the category of the ship (Polar Code part I-A/paras.3.3.1 and 3.3.2);
- (WI) 3.1.1.5 examining the stability information, including the damage stability (Add. to PI, information, where applicable, and loading instrument as appropriate, CI) with icing allowance in the stability calculations (Polar Code part I-A/paras.4.3.1.1 and 4.3.2);
- (WI) 3.1.1.6 examining whether structures and installations are designed with a view (Add. to PI, to minimizing the accretion of ice (Polar Code part I-A/para.4.3.1.2.1);
 CI)

¹⁰⁶ Refer to the *Unified interpretation of paragraph 13.3.5 of the IGC Code (as amended by resolution MSC.370(93))* (MSC.1/Circ.1590).

- (WI) 3.1.1.7 examining whether the ship is equipped with efficient means for (Add. to PI, removing ice as required by the Administration (Polar Code CI) part I-A/para.4.3.1.2.2);
- (WI) 3.1.1.8 examining the means to remove or prevent ice and snow accretion around (Add. to PI, hatches and doors; for ships intended to operate in low air temperature, examining the means for prevention of freezing or excessive viscosity of liquids for hydraulically operated hatches and doors as mentioned in the PWOM (Polar Code part I-A/paras.5.3.1 and 5.3.2.1);
- (WI) 3.1.1.9 for ships intended to operate in low air temperature, confirming that the (Add. to PI, watertight and weathertight doors, hatches and closing devices, not CI) within habitable environment and requiring access while at sea, are designed to be operable by personnel wearing heavy winter clothing including thick mittens (Polar Code part I-A/para.5.3.2.2);
- (WI) 3.1.1.10 examining the means for protecting machinery installations and (Add. to PI, associated equipment against the effect of ice accretion and/or snow CI)
 CI) accumulation, ice ingestion from seawater, freezing and increased viscosity of liquids, seawater intake temperature and snow ingestion, and that seawater supplies for machinery systems are designed to prevent ingestion of ice (Polar Code part I-A/paras.6.3.1.1, 6.3.1.2 and 6.3.1.3);
- (WI) 3.1.1.11 for ships intended to operate in low air temperature, confirming that the (Add. to PI, exposed machinery and electrical installation and appliances are fit for the polar service temperature (Polar Code part I-A/para.6.3.2.1);
- (WI) 3.1.1.12 for ships intended to operate in low air temperature, confirming that (Add. to PI, means are provided to ensure that combustion air for internal combustion CI) engines driving essential machinery is maintained at a temperature in compliance with the criteria provided by the engine manufacturer (Polar Code part I-A/para.6.3.2.2);
- (WI) 3.1.1.13 for ships intended to operate in low air temperature, examining the plans (Add. to PI, for materials of the exposed machinery and foundations, confirming that the materials are approved according to applicable standards, taking into account the polar service temperature and the required ice strengthening (Polar Code part I-A/para.6.3.2.3);
- (WI) 3.1.1.14 examining the plans and design of the scantlings of propeller blades, (Add. to PI, propulsion line, steering equipment and other appendages, confirming that they are approved according to the polar service temperature if applicable and ice strengthening standards, where applicable for the category of the ship (Polar Code part I-A/para.6.3.3);
- (WI) 3.1.1.15 examining all components of fire safety systems and appliances if (Add. to PI, installed in exposed positions to ensure that they are protected from ice accretion and snow accumulation according to the operational assessment (Polar Code part I-A/para.7.2.1.1);
- (WI) 3.1.1.16 examining the design of fire safety systems and appliances for operation (Add. to PI, by persons wearing bulky and cumbersome cold weather gear including gloves, where appropriate (Polar Code part I-A/para.7.2.1.3);

- (WI) 3.1.1.17 examining the means to remove or prevent ice and snow accretion from (Add. to PI, accesses of fire safety systems and appliances, escape routes, muster EI) stations, embarkation areas, survival craft, its launching appliances and access to survival craft according to the PWOM (Polar Code part I-A/paras.7.2.1.4 and 8.3.1.1);
- (WI) 3.1.1.18 confirming that the extinguishing media is suitable for the intended (Add. to PI, operation (Polar Code part I-A/para.7.2.1.5);
 EI)
- (WI) 3.1.1.19 examining that all components of fire safety systems and appliances are (Add. to PI, designed to ensure availability and effectiveness at the polar service EI)
 EI) temperature (Polar Code part I-A/para.7.2.2.1);
- (WI) 3.1.1.20 examining that the isolating and pressure/vacuum valves in exposed (Add. to PI, locations are protected from ice accretion and remain accessible at all times (Polar Code part I-A/para.7.3.1.1);
- (WI) 3.1.1.21 examining that all two-way portable radio communication equipment is (Add. to PI, capable of operating at the polar service temperature (Polar Code EI) part I-A/para.7.3.1.2);
- (WI) 3.1.1.22 examining that the fire pumps including emergency fire pumps, water (Add. to PI, mist and water spray pumps are located in compartments maintained EI) above freezing (Polar Code part I-A/paras.7.3.2.1 and 7.3.2.2);
- (WI) 3.1.1.23 examining whether the arrangement of the fire main is such that exposed (Add. to PI, sections can be isolated and means of draining of exposed sections are provided, and, where fixed water-based fire-extinguishing systems are located in a space separate from the main fire pumps and use an own sea suction, confirming that this sea suction is capable of being cleared of ice accumulation (Polar Code part I-A/paras.7.3.2.2 and 7.3.2.4);
- (WI) 3.1.1.24 examining that the fire-fighter's outfits are stored in warm locations on (Add. to PI, the ship (Polar Code part I-A/para.7.3.2.3);
 EI)
- (WI) 3.1.1.25 examining that portable and semi-portable extinguishers are protected (Add. to PI, from freezing temperatures, and confirming that locations subject to freezing are provided with extinguishers capable of operation at the polar service temperature (Polar Code part I-A/para.7.3.3.1);
- (WI) 3.1.1.26 examining the plans for the materials of exposed fire safety systems, (Add. to PI, confirming that they are approved according to the polar service EI)
 EI) temperature and ice strengthening standards (Polar Code part I-A/para.7.3.3.2);
- (WI) 3.1.1.27 for ships constructed on or after 1 January 2017, examining that the (Add. to PI, exposed escape routes are arranged so as not to hinder passage by persons wearing suitable polar clothing (Polar Code part I-A/para.8.3.1.2);
- (WI) 3.1.1.28 for ships intended to operate in low air temperatures, examining the adequacy of embarkation arrangements, with full regard to any effect of

(Add. to PI, persons wearing additional polar clothing (Polar Code EI) part I-A/para.8.3.1.3);

- (WI) 3.1.1.29 examining the means to ensure safe evacuation of persons, including (Add. to PI, safe deployment of survival equipment, when operating in ice-covered EI) waters, or directly onto the ice, as applicable (Polar Code part I-A/para.8.3.2.1);
- (WI) 3.1.1.30 examining that life-saving appliances and arrangements as required by (Add. to PI, the Polar Code, if using devices requiring a source of power, are able to operate independently of the ship's main source of power (Polar Code part I-A/para.8.3.2.2);
- (WI) 3.1.1.31 for passenger ships, examining that a proper sized immersion suit of the insulated type or a thermal protective aid is provided for each person on board according to the operational assessment (Polar Code part I-A/paras.8.3.3.1.1 and 8.3.3.1.2);
- (WI) 3.1.1.32 for cargo ships, examining that all the immersion suits provided on board (Add. to EI) are of the insulated type (Polar Code part I-A/para.8.3.3.1.2);
- (WI) 3.1.1.33 examining that for ships intended to operate in extended periods of (Add. to PI, darkness, searchlights suitable for continuous use to facilitate identification of ice are provided for each lifeboat (Polar Code part I-A/para.8.3.3.2);
- (WI) 3.1.1.34 confirming that the lifeboats are of the partially or totally enclosed type, (Add. to PI, as appropriate (Polar Code part I-A/para.8.3.3.3.1);
 EI)
- (WI) 3.1.1.35 examining that, when personal or group survival equipment is required according to the operational assessment, personal and group survival equipment sufficient for 110% of the persons on board is stowed in easily accessible locations; that containers for group survival equipment are designed to be easily movable over the ice and floatable; and that means of ensuring that personal and group survival equipment is accessible following abandonment is provided (Polar Code part I-A/paras.8.3.3.3.2, 8.3.3.3.1 to 8.3.3.3.4);
- (WI) 3.1.1.36 examining that the survival craft and launching appliances have sufficient (Add. to PI, capacity to accommodate the additional personal and group survival equipment if required and carried in addition to persons, and that adequate emergency rations are provided for the maximum expected time of rescue (Polar Code part I-A/paras.8.3.3.3.3.5 and 8.3.3.3.4);
- (WI) 3.1.1.37 confirming that the instructions to passengers on the use of the personal (Add. to PI, survival equipment and the action to take in an emergency are provided EI) on board (Polar Code part I-A/para.8.3.3.3.3.6);
- (WI) 3.1.1.38 examining the means of receiving and displaying the information on ice (Add. to PI, conditions in the area of operation (Polar Code part I-A/para.9.3.1); EI)
- (WI) 3.1.1.39 for ships constructed on or after 1 January 2017 and ice strengthened, examining that either two independent echo-sounding devices or one

(Add. to PI, echo-sounding device with two separate independent transducers are provided (Polar Code part I-A/para.9.3.2.1.1);

- (WI) 3.1.1.40 confirming that clear view astern is achieved, and for ships built (Add. to PI, before 1 July 1998 and with a length of less than 55 m, confirming that, clear-view navigation bridge front windows are provided (SOLAS 74/00reg.V/22.1.9.4, Polar Code part I-A/ch.9.3.2.1.2);
- (WI) 3.1.1.41 where ice accretion is likely to occur, examining the means to prevent (Add. to PI, the accumulation of ice on antennas required for navigation and EI) communication (Polar Code part I-A/para.9.3.2.1.3);
- (WI) 3.1.1.42 for ice strengthened ships, examining that sensors for navigational (Add. to PI, equipment, required either by SOLAS or the Code, projecting below the EI) hull are protected against ice (Polar Code part I-A/para.9.3.2.1.4.1);
- (WI) 3.1.1.43 examining the arrangements of the bridge wings for protection of (Add. to PI, navigational equipment and operating personnel, in category A and B EI) ships constructed on or after 1 January 2017 (Polar Code part I-A/para.9.3.2.1.4.2);
- (WI) 3.1.1.44 examining the two independent non-magnetic means for heading (Add. to PI, information, and at least one GNSS compass or equivalent for ships intended to proceed to latitudes over 80 degrees, connected to the ship's main and emergency source of power (Polar Code part I-A/paras.9.3.2.2.1 and 9.3.2.2.2);
- (WI) 3.1.1.45 examining that two remotely rotatable, narrow-beam searchlights (Add. to PI, controllable from the bridge to provide lighting over an arc of 360 degrees, or other means to visually detect ice, are provided on board if the ship is not operating solely in 24 h daylight, and examining that a manually initiated flashing red light visible from astern to indicate when the ship is stopped is available, for ships involved in operations with an icebreaker escort (Polar Code part I-A/paras.9.3.3.1 and 9.3.3.2);
- (WI) 3.1.1.46 examining that the communication equipment on board has the (Add. to PI, capabilities for ship-to-ship and ship-to-shore communication, taking into account the limitations of communications systems in high latitudes and the anticipated low temperature (Polar Code part I-A/para.10.3.1.1);
- (WI) 3.1.1.47 for ships intended to provide icebreaking escort, examining the sound (Add. to PI, signalling system capable of being mounted to face astern (Polar Code RI) part I-A/para.10.3.1.2);
- (WI) 3.1.1.48 examining the means for two-way on-scene and SAR coordination (Add. to PI, communications for search and rescue purposes including aeronautical frequencies operations and that communication equipment provides for two-way voice and data communication with a Telemedical Assistance Service (TMAS) (Polar Code part I-A/paras.10.3.1.3 and 10.3.1.4);

- (WI) 3.1.1.49 for ships intended to operate in low air temperature, examining that each (Add. to PI, rescue boat and lifeboat is capable of being provided with devices for transmitting signals for distress alerting, locating and on-scene communications (Polar Code part I-A/para.10.3.2.1);
- (WI) 3.1.1.50 for ships intended to operate in low air temperature, examining the (Add. to PI, capability of all other survival craft to transmit signals for location and for communication (Polar Code part I-A/para.10.3.2.2);
- (WI) 3.1.1.51 confirming that procedures are provided on board for ensuring the (Add. to PI, availability of the mandatory communication equipment for use in survival craft, including availability of battery power for the maximum expected time of rescue (Polar Code part I-A/para.10.3.2.3); and
- (WI) 3.1.1.52 examining, where applicable, the approved documentation for the (Add. to PI, alternative design and arrangements (SOLAS 74/00/14 reg.XIV/4). CI, EI, RI)
- (WI) 3.1.2 For compliance with part I-A of the International Code for Ships Operating in Polar Waters, the survey during construction and after installation of the hull, machinery and equipment should consist of:
- (WI) 3.1.2.1 for ships intended to operate in low air temperature, checking the certificates or equivalent documents of the systems and equipment required by this Code for consistency with the polar service temperature specified for the ship (Polar Code part I-A/para.1.4.2);
- (WI) 3.1.2.2 for ships operating in low air temperature, checking the certificates or (Add. to PI, El RI)
 El RI) for ships operating in low air temperature, checking the certificates or equivalent documents of the survival systems and equipment for consistency with the maximum expected rescue time at polar service temperature (Polar Code part I-A/para.1.4.3);
- (WI) 3.1.2.3 examining the materials of exposed structures and the scantlings of the (Add. to PI, ship in accordance with the polar service temperature and ice strengthening standards, where applicable for the category of the ship (Polar Code part I-A/para.3.3.1 and 3.3.2);
- (WI) 3.1.2.4 examining whether structures and installations are designed with a view (Add. to PI, to minimizing the accretion of ice (Polar Code part I-A/para.4.3.1.2.1); CI
- (WI) 3.1.2.5 examining the means for removing ice as required by the Administration (Add. to PI, and mentioned in the PWOM (Polar Code part I-A/para.4.3.1.2.2); CI)
- (WI) 3.1.2.6 examining the means to remove or prevent ice and snow accretion (Add. to PI, around hatches and doors, and testing the function of the electric heat tracing system protecting hatches and doors from freezing as applicable; for ships intended to operate in low air temperature, examining the means for prevention of freezing or excessive viscosity of liquids for hydraulically operated hatches and doors as mentioned in the PWOM (Polar Code part I-A/paras.5.3.1 and 5.3.2.1);
- (WI) 3.1.2.7 for ships intended to operate in low air temperature, examining that the watertight and weathertight doors, hatches and closing devices, which

- (Add. to PI, are not within a habitable environment and require access while at sea
 CI) are operable by personnel wearing heavy winter clothing including thick mittens (Polar Code part I-A/para.5.3.2.2);
- (WI) 3.1.2.8 examining the means for protecting machinery installations and associated (Add. to PI, equipment against the effect of ice accretion and/or snow accumulation, ice II) ingestion from seawater, freezing and increased viscosity of liquids, seawater intake temperature and snow ingestion, and that seawater supplies for machinery systems are designed to prevent ingestion of ice (Polar Code part I-A/paras.6.3.1.1, 6.3.1.2 and 6.3.1.3);
- (WI) 3.1.2.9 for ships intended to operate in low air temperature, confirming that the (Add. to PI, exposed machinery and electrical installation and appliances are fit for CI)
 (WI) 3.1.2.9 for ships intended to operate in low air temperature, confirming that the polar service temperature (Polar Code part I-A/para.6.3.2.1);
- (WI) 3.1.2.10 for ships intended to operate in low air temperature, examining and (Add. to PI, testing the means to ensure that combustion air for internal combustion engines driving essential machinery is maintained at a temperature in compliance with the criteria provided by the engine manufacturer (Polar Code part I-A/para.6.3.2.2);
- (WI) 3.1.2.11 for ships intended to operate in low air temperature, examining the (Add. to PI, certificates or equivalent documents of the materials of the exposed CI)
 CI) machinery and foundations in accordance with applicable standards, taking into account the polar service temperature and the required ice strengthening (Polar Code part I-A/para.6.3.2.3);
- (WI) 3.1.2.12 examining the scantlings of propeller blades, propulsion line, steering (Add. to PI, equipment and other appendages in accordance with the polar service CI)
 CI) temperature if applicable and ice strengthening standards, where applicable for the category of the ship (Polar Code part I-A/para.6.3.3);
- (WI) 3.1.2.13 examining that all components of fire safety systems and appliances if (Add. to PI, installed in exposed positions are protected from ice accretion and snow accumulation according to the operational assessment (Polar Code part I-A/para.7.2.1.1);
- (WI) 3.1.2.14 examining the fire safety systems and appliances for operation by (Add. to PI, persons wearing bulky and cumbersome cold weather gear including EI) gloves, where appropriate (Polar Code part I-A/para.7.2.1.3);
- (WI) 3.1.2.15 examining the means to remove or prevent ice and snow accretion from (Add. to PI, accesses of fire safety systems and appliances, escape routes, muster EI) stations, embarkation areas, survival craft, its launching appliances and access to survival craft according to the PWOM (Polar Code part I-A/paras.7.2.1.4 and 8.3.1.1);
- (WI) 3.1.2.16 confirming that the extinguishing media is suitable for the intended (Add. to PI, operation (Polar Code part I-A/para.7.2.1.5);
 EI)
- (WI) 3.1.2.17 examining that all components of fire safety systems and appliances are (Add. to PI, designed to ensure availability and effectiveness at the polar service EI)
 EI) temperature (Polar Code part I-A/para.7.2.2.1);

- (WI) 3.1.2.18 examining that the isolating and pressure/vacuum valves in exposed (Add. to PI, locations are protected from ice accretion and remain accessible at all EI)
 EI) times (Polar Code part I-A/para.7.3.1.1);
- (WI) 3.1.2.19 examining that all two-way portable radio communication equipment is (Add. to PI, capable of operating at the polar service temperature (Polar Code EI) part I-A/para.7.3.1.2);
- (WI) 3.1.2.20 examining that the fire pumps including emergency fire pumps, water (Add. to PI, mist and water spray pumps are located in compartments maintained above freezing (Polar Code part I-A/paras.7.3.2.1 and 7.3.2.2);
- (WI) 3.1.2.21 examining whether the arrangement of the fire main is such that exposed (Add. to PI, sections can be isolated and means of draining of exposed sections are provided, and, where fixed water-based fire-extinguishing systems are located in a space separate from the main fire pumps and use an own sea suction, confirming that this sea suction is capable of being cleared of ice accumulation (Polar Code part I-A/paras.7.3.2.2 and 7.3.2.4);
- (WI) 3.1.2.22 examining that the fire-fighter's outfits are stored in warm locations on (Add. to PI, the ship (Polar Code part I-A/para.7.3.2.3);
 EI)
- (WI) 3.1.2.23 examining that portable and semi-portable extinguishers are protected (Add. to PI, from freezing temperatures, and confirming that locations subject to Fiezing are provided with extinguishers capable of operation at the polar service temperature (Polar Code part I-A/para.7.3.3.1);
- (WI) 3.1.2.24 examining the exposed fire safety systems in accordance with the polar (Add. to PI, service temperature and ice strengthening standards (Polar Code EI) part I-A/para.7.3.3.2);
- (WI) 3.1.2.25 for ships constructed on or after 1 January 2017, confirming the exposed (Add. to PI, escape routes arranged as a passage by persons wearing suitable polar EI)
 (b) Content (Content of Content of Content
- (WI) 3.1.2.26 for ships intended to operate in low air temperatures, confirming the (Add. to PI, embarkation arrangements, with full regard for persons wearing EI) additional polar clothing (Polar Code part I-A/para.8.3.1.3);
- (WI) 3.1.2.27 examining the means to ensure safe evacuation of persons, including (Add. to PI, safe deployment of survival equipment, when operating in ice-covered EI) waters, or directly onto the ice, as applicable (Polar Code part I-A/para.8.3.2.1);
- (WI) 3.1.2.28 examining life-saving appliances and arrangements as required by the Polar (Add. to PI, Code, if using devices requiring a source of power, and testing that they are able to operate independently of the ship's main source of power (Polar Code part I-A/para.8.3.2.2);
- (WI) 3.1.2.29 for passenger ships, examining that a proper sized immersion suit of the insulated type or a thermal protective aid is provided for each person on board according to the operational assessment (Polar Code part I-A/paras.8.3.3.1.1 and 8.3.3.1.2);

- (WI) 3.1.2.30 for cargo ships, examining that all the immersion suits equipped on board (Add. to EI) are of the insulated type (Polar Code part I-A/para.8.3.3.1.2);
- (WI) 3.1.2.31 for ships intended to operate in extended periods of darkness, examining (Add. to PI, and testing the searchlights provided for each lifeboat suitable for continuous use to facilitate identification of ice (Polar Code part I-A/para.8.3.3.2);
- (WI) 3.1.2.32 confirming that the lifeboats are of the partially or totally enclosed type, as (Add. to PI, appropriate (Polar Code part I-A/para.8.3.3.3.1);
 EI)
- (WI) 3.1.2.33 confirming that, when personal or group survival equipment is required (Add. to PI, according to the operational assessment, personal and group survival equipment sufficient for 110% of the persons on board is stowed in easily accessible locations; that containers for group survival equipment are designed to be easily movable over the ice and floatable; and that means of ensuring that personal and group survival equipment is accessible following abandonment is provided (Polar Code part I-A/paras.8.3.3.3.2, 8.3.3.3.1 to 8.3.3.3.4);
- (WI) 3.1.2.34 confirming that the survival craft and launching appliances have sufficient (Add. to PI, capacity to accommodate the additional personal and group survival equipment if required and carried in addition to persons, and that adequate emergency rations are provided for the maximum expected time of rescue (Polar Code part I-A/paras.8.3.3.3.3.5 and 8.3.3.3.4);
- (WI) 3.1.2.35 confirming that the instructions to passengers on the use of the personal (Add. to PI, survival equipment and the action to take in an emergency are provided EI) on board (Polar Code part I-A/para.8.3.3.3.3.6);
- (WI) 3.1.2.36 examining the means of receiving and displaying the information on ice (Add. to PI, conditions in the area of operation (Polar Code part I-A/para.9.3.1); EI)
- (WI) 3.1.2.37 for ships constructed on or after 1 January 2017 and ice strengthened, (Add. to PI, confirming that either two independent echo-sounding devices or one EI) echo-sounding device with two separate independent transducers are provided (Polar Code part I-A/para.9.3.2.1.1);
- (WI) 3.1.2.38 confirming that clear view astern is achieved, and for ships built before (Add. to PI, 1 July 1998 and with a length of less than 55 m, confirming that clear-view EI) navigation bridge front windows are provided (SOLAS 74/00 regulation V/22.1.9.4, Polar Code part I-A/para.9.3.2.1.2);
- (WI) 3.1.2.39 where ice accretion is likely to occur, examining the means to prevent (Add. to PI, the accumulation of ice on antennas required for navigation and EI) communication (Polar Code part I-A/para.9.3.2.1.3);
- (WI) 3.1.2.40 for ice strengthened ships, examining that sensors for navigational (Add. to PI, equipment, required either by SOLAS or the Code, projecting below the EI) hull are protected against ice (Polar Code part I-A/para.9.3.2.1.4.1);

- (WI) 3.1.2.41 examining the arrangements of the bridge wings for protection of (Add. to PI, navigational equipment and operating personnel, in category A and B EI) ships constructed on or after 1 January 2017 (Polar Code part I-A/para.9.3.2.1.4.2);
- (WI) 3.1.2.42 examining the two independent non-magnetic means for heading (Add. to PI, information, and at least one GNSS compass or equivalent for ships intended to proceed to latitudes over 80 degrees, connected to the ship's main and emergency source of power (Polar Code part I-A/paras.9.3.2.2.1 and 9.3.2.2.2);
- (WI) 3.1.2.43 examining and testing the two remotely rotatable, narrow-beam (Add. to PI, searchlights controllable from the bridge to provide lighting over an arc of 360 degrees, or other means to visually detect ice, for ships not operating solely in 24 h daylight, and examining and testing the manually initiated flashing red light visible from astern to indicate when the ship is stopped, for ships involved in operations with an icebreaker escort (Polar Code part I-A/paras.9.3.3.1 and 9.3.3.2);
- (WI) 3.1.2.44 examining and testing the communication equipment on board for (Add. to PI, ship-to-ship and ship-to-shore communication, taking into account the RI)
 RI) limitations of communications systems in high latitudes and the anticipated low temperature (Polar Code part I-A/para.10.3.1.1);
- (WI) 3.1.2.45 for ships intended to provide icebreaking escort, examining the sound (Add. to PI, signalling system capable of being mounted to face astern (Polar Code RI) part I-A/para.10.3.1.2);
- (WI) 3.1.2.46 examining and testing the means for two-way on-scene and SAR (Add. to PI, coordination communications for search and rescue purposes including aeronautical frequencies operations and that communication equipment provides for two-way voice and data communication with a Telemedical Assistance Service (TMAS) (Polar Code part I-A/para.10.3.1.3 and 10.3.1.4);
- (WI) 3.1.2.47 for ships intended to operate in low air temperature, examining that each (Add. to PI, rescue boat and lifeboat is capable of being provided with devices for transmitting signals for distress alerting, locating and on-scene communications (Polar Code part I-A/para.10.3.2.1);
- (WI) 3.1.2.48 for ships intended to operate in low air temperature, examining the (Add. to PI, capability of all other survival craft to transmit signals for location and for RI)
 Communication (Polar Code part I-A/para.10.3.2.2); and
- (WI) 3.1.2.49 examining, where applicable, the alternative design and arrangements (Add. to PI, for ship structure, machinery installations, fire safety/protection or CI, EI, RI) life-saving appliances and arrangements, in accordance with the test and inspection requirements, if any, specified in the approved documentation and PWOM (SOLAS 74/00/14 reg.XIV/4).
- (WI) 3.1.3 For compliance with part I-A of the International Code for Ships Operating in Polar Waters the check that all the required documentation has been placed on board the ship should consist of:

- (WI) 3.1.3.1 checking that the PWOM with the hazards identified in the operational (Add. to PI, assessment being addressed properly is placed on board (Polar Code CI, EI, RI) part I-A/paras.2.3, 4.3.1.3 and 4.3.1.4);
- (WI) 3.1.3.2 confirming that the approved stability information, damage stability (Add. to PI, information and loading instrument as appropriate, with icing allowance CI) in the stability calculations, are on board (Polar Code part I-A/paras.4.3.1.1 and 4.3.2);
- (WI) 3.1.3.3 confirming as applicable that the crew training records or other (Add. to PI, equivalent documents for the use of the personal survival equipment and group survival equipment are placed on board (Polar Code part I-A/para.8.3.3.3.7);
- (WI) 3.1.3.4 confirming that procedures are provided on board for ensuring the (Add. to PI, availability of the mandatory communication equipment for use in survival craft, including availability of battery power for the maximum expected time of rescue (Polar Code part I-A/para.10.3.2.3); and
- (WI) 3.1.3.5 confirming that, where applicable, the approved documentation for the (Add. to PI, alternative design and arrangement is on board, with the relevant CI, EI, RI) contents being entered in the PWOM (SOLAS 74/00/14 reg.XIV/4).
- (WI) 3.1.4 For compliance with part I-A of the International Code for Ships Operating in Polar Waters the completion of the initial survey should consist of:
- (WI) 3.1.4.1 after a satisfactory survey, issuing the Polar Ship Certificate and its associated Record of Equipment.
- (WA) **3.2 Annual surveys** see part "General" paragraph 5.13.3.2
- (WA) 3.2.1 For compliance with part I-A of the International Code for Ships Operating in Polar Waters the examination of current certificates and other records should consist of:
- (WA) 3.2.1.1 confirming the provision of the operational assessment and reviewing any changes thereto (Polar Code part I-A/section 1.5);
 CA, EA, RP)
- (WA) 3.2.1.2 confirming that the PWOM is on board, and checking whether any changes have been made to it since the last survey (Polar Code part I-A/section 2.3 and paras.4.3.1.3 and 4.3.1.4);
 RP)
- (WA) 3.2.1.3 confirming the availability of approved stability information, damage stability information and loading instrument as appropriate, with icing allowance in the stability calculations (Polar Code part I-A/paras.4.3.1.1 and 4.3.2);
- (WA) 3.2.1.4 confirming as applicable that the crew training records or other (Add. to EA) equivalent documents for the use of the personal survival equipment and group survival equipment are placed on board (Polar Code part I-A/para.8.3.3.3.7);

- (WA) 3.2.1.5 confirming that procedures are provided on board for ensuring the availability of the mandatory communication equipment for use in survival craft, including availability of battery power for the maximum expected time of rescue (Polar Code part I-A/para.10.3.2.3);
- (WA) 3. 2.1.6 confirming that the voyage plan has been provided on board for the voyages in polar waters since the last survey; if no trading in polar waters cA, EA, RP)
 (WA) 3. 2.1.6 confirming that the voyage plan has been provided on board for the voyages in polar waters since the last survey; if no trading in polar waters considered (Polar Code part I-A/section 11.3);
- (WA) 3.2.1.7 where applicable, checking the qualifications of the masters, chief mates, (Add. to officers and/or other persons in charge of a navigational watch on board CA, EA, ships operating in polar waters in accordance with chapter V of the STCW RP) STCW Convention and the Code (Polar Code part I-A/paras.12.3.1 and 12.3.2);
- (WA) 3.2.1.8 checking the qualification certificates (if required by the Administration) and/or familiarization records of all the crew members for their assigned duties referenced in the PWOM (Polar Code part I-A/para.12.3.4); and RP)
- (WA) 3.2.1.9 confirming that, where applicable, the approved documentation for the alternative design and arrangements is on board, with the relevant contents being entered in the PWOM (SOLAS 74/00/14 reg.XIV/4).
 RP)
- (WA) 3.2.2 For compliance with part I-A of the International Code for Ships Operating in Polar Waters the annual survey of the hull, machinery and equipment should consist of:
- (WA) 3.2.2.1 examining the means for removing ice as required by the Administration (Add. to CA)
 (Add. to CA)
- (WA) 3.2.2.2 (Add. to CA)
 (Add. to CA)
 examining the means to remove or prevent ice and snow accretion around hatches and doors, and testing the function of the electric heat tracing system protecting hatches and doors from freezing as applicable; for ships intended to operate in low air temperature, examining the means for prevention of freezing or excessive viscosity of liquids for hydraulically operated hatches and doors as mentioned in the PWOM (Polar Code part I-A/para.5.3.1 and 5.3.2.1);
- (WA) 3.2.2.3 examining the means for protecting machinery installations and associated equipment against the effect of ice accretion and/or snow accumulation, ice ingestion from seawater, freezing and increased viscosity of liquids, seawater intake temperature and snow ingestion (Polar Code part I-A/para.6.3.1.1 and 6.3.1.2);
- (WA) 3.2.2.4 for ships intended to operate in low air temperature, examining and testing the means to ensure that combustion air for internal combustion engines driving essential machinery is maintained at a temperature in compliance with the criteria provided by the engine manufacturer (Polar Code part I-A/para.6.3.2.2);

- (WA) 3.2.2.5 examining that all components of fire safety systems and appliances if (Add. to EA) installed in exposed positions are protected from ice accretion and snow accumulation according to the operational assessment (Polar Code part I-A/para.7.2.1.1);
- (WA) 3.2.2.6 examining the fire safety systems and appliances for operation by (Add. to EA) persons wearing bulky and cumbersome cold weather gear including gloves, where appropriate (Polar Code part I-A/para.7.2.1.3);
- (WA) 3.2.2.7 examining the means to remove or prevent ice and snow accretion from (Add. to EA) accesses of fire safety systems and appliances, escape routes, muster stations, embarkation areas, survival craft, its launching appliances and access to survival craft according to the PWOM (Polar Code part I-A/paras.7.2.1.4 and 8.3.1.1);
- (WA) 3.2.2.8 confirming that the extinguishing media are suitable for the intended (Add. to EA) operation (Polar Code part I-A/para.7.2.1.5);
- (WA) 3.2.2.9 examining that the isolating and pressure/vacuum valves in exposed (Add. to EA) locations are protected from ice accretion and remain accessible at all time (Polar Code part I-A/para.7.3.1.1);
- (WA) 3.2.2.10 examining that all two-way portable radio communication equipment is (Add. to EA) capable of operating at the polar service temperature (Polar Code part I-A/para.7.3.1.2);
- (WA) 3.2.2.11 examining that the fire pumps including emergency fire pumps, water (Add. to EA) mist and water spray pumps are located in compartments maintained above freezing (Polar Code part I-A/paras.7.3.2.1 and 7.3.2.2);
- (WA) 3.2.2.12 examining whether the arrangement of the fire main is such that exposed (Add. to EA) sections can be isolated and means of draining of exposed sections are provided, and, where fixed water-based fire-extinguishing systems are located in a space separate from the main fire pumps and use an own sea suction, confirming that this sea suction is capable of being cleared of ice accumulation (Polar Code part I-A/paras.7.3.2.2 and 7.3.2.4);
- (WA) 3.2.2.13 examining that the fire-fighter's outfits are stored in warm locations on (Add. to EA) the ship (Polar Code part I-A/para.7.3.2.3);
- (WA) 3.2.2.14 examining that portable and semi-portable extinguishers are protected (Add. to EA) from freezing temperatures, and confirming that locations subject to freezing are provided with extinguishers capable of operation at the polar service temperature (Polar Code part I-A/para.7.3.3.1);
- (WA) 3.2.2.15 examining the exposed fire safety systems in accordance with the polar (Add. to EA) service temperature and ice strengthening standards (Polar Code part I-A/para.7.3.3.2);
- (WA) 3.2.2.16 examining the means to ensure safe evacuation of persons, including (Add. to EA) safe deployment of survival equipment, when operating in ice-covered waters, or directly onto the ice, as applicable (Polar Code part I-A/para.8.3.2.1);

- (WA) 3.2.2.17 confirming that life-saving appliances and arrangements as required by (Add. to EA) the Polar Code, if using devices requiring a source of power, are able to operate independently of the ship's main source of power (Polar Code part I-A/para.8.3.2.2);
- (WA) 3.2.2.18 for cargo ships, examining that all the immersion suits equipped on board (Add. to EA) are of the insulated type (Polar Code part I-A/para.8.3.3.1.2);
- (WA) 3.2.2.19 for ships intended to operate in extended periods of darkness, examining (Add. to EA) and testing the searchlights suitable for continuous use to facilitate identification of ice provided for each lifeboat (Polar Code part I-A/para.8.3.3.2);
- (WA) 3.2.2.20 confirming that the lifeboats are of the partially or totally enclosed type, (Add. to EA) as appropriate (Polar Code part I-A/para.8.3.3.3.1);
- (WA) 3.2.2.21 confirming that, when personal or group survival equipment is required (Add. to EA) according to the operational assessment, personal and group survival equipment sufficient for 110% of the persons on board is stowed in easily accessible locations; that containers for group survival equipment are designed to be easily movable over the ice and floatable; and that means of ensuring that personal and group survival equipment is accessible following abandonment is provided (Polar Code part I-A/paras.8.3.3.3.2, 8.3.3.3.1 to 8.3.3.3.4);
- (WA) 3.2.2.22 confirming that the survival craft and launching appliances have sufficient (Add. to EA) capacity to accommodate the additional personal and group survival equipment if required and carried in addition to persons and that adequate emergency rations are provided for the maximum expected time of rescue (Polar Code part I-A/paras.8.3.3.3.5 and 8.3.3.4);
- (WA) 3.2.2.23 confirming that the instructions to passengers are provided on board (Add. to EA) (Polar Code part I-A/para.8.3.3.3.3.6);
- (WA) 3.2.2.24 examining the means of receiving and displaying information on ice (Add. to EA) conditions in the area of operation, with a demonstration by the crew on using the equipment and receiving the relevant information (Polar Code part I-A/para.9.3.1);
- (WA) 3.2.2.25 for ships constructed on or after 1 January 2017 and ice strengthened, (Add. to EA) confirming that either two independent echo-sounding devices or one echo-sounding device with two separate independent transducers are provided (Polar Code part I-A/para.9.3.2.1.1);
- (WA) 3.2.2.26 confirming that clear view astern is achieved, and for ships built before 1 (Add. to EA) July 1998 and with a length of less than 55 m, confirming that clear-view navigation bridge front windows are provided (SOLAS 74/00 regulation V/22.1.9.4, Polar Code part I-A/para.9.3.2.1.2);
- (WA) 3.2.2.27 where ice accretion is likely to occur, examining the means to prevent (Add. to EA) the accumulation of ice on antennas required for navigation and communication (Polar Code part I-A/para.9.3.2.1.3);

- (WA) 3.2.2.28 examining the arrangements of the bridge wings for protection of (Add. to EA) navigational equipment and operating personnel, in category A and B ships constructed on or after 1 January 2017 (Polar Code part I-A/para.9.3.2.1.4.2);
- (WA) 3.2.2.29 examining the two independent non-magnetic means for heading (Add. to EA) information, and at least one GNSS compass or equivalent for ships intended to proceed to latitudes over 80 degrees, connected to the ship's main and emergency source of power (Polar Code part I-A/paras.9.3.2.2.1 and 9.3.2.2.2);
- (WA) 3.2.2.30 examining and testing the two remotely rotatable, narrow-beam (Add. to EA) searchlights controllable from the bridge to provide lighting over an arc of 360 degrees, or other means to visually detect ice, for ships not operating solely in 24 h daylight, and examining and testing the manually initiated flashing red light visible from astern to indicate when the ship is stopped, for ships involved in operations with an icebreaker escort (Polar Code part I-A/paras.9.3.3.1 and 9.3.3.2);
- (WA) 3.2.2.31 examining and testing the communication equipment on board for ship-to-ship and ship-to-shore communication, taking into account the limitations of communications systems in high latitudes and the anticipated low temperature (Polar Code part I-A/para.10.3.1.1);
- (WA) 3.2.2.32 for ships intended to provide icebreaking escort, examining and testing the sound signalling system capable of being mounted to face astern (Polar Code part I-A/para.10.3.1.2);
- (WA) 3.2.2.33 examining and testing the means for two-way on-scene and SAR coordination communications for search and rescue purposes including aeronautical frequencies operations, and that communication equipment provides for two-way voice and data communication with a Telemedical Assistance Service (TMAS) (Polar Code part I-A/paras.10.3.1.3 and 10.3.1.4);
- (WA) 3.2.2.34 for ships intended to operate in low air temperature, examining that each rescue boat and lifeboat is capable of being provided with devices for transmitting signals for distress alerting, locating and on-scene communications (Polar Code part I-A/para.10.3.2.1);
- (WA) 3.2.2.35 for ships intended to operate in low air temperature, examining the capability of all other survival craft to transmit signals for location and for communication (Polar Code part I-A/para.10.3.2.2); and
- (WA) 3.2.2.36 examining, where applicable, the alternative design and arrangements (Add. to CA, for ship structure, machinery installations, fire safety/protection or EA, RP) life-saving appliances and arrangements, in accordance with the test, inspection and maintenance requirements, if any, specified in the approved documentation and PWOM (SOLAS 74/00/14 reg.XIV/4).
- (WA) 3.2.3 For compliance with part I-A of the International Code for Ships Operating in Polar Waters the completion of the annual survey should consist of:

(WA) 3.2.3.1 after a satisfactory survey, endorsing the Polar Ship Certificate; and (Add. to CA, EA, RP) 3.2.3.2 (WA) if a survey shows that the condition of a ship or its equipment is unsatisfactory - see part "General" section 4.8. (Add. to CA, EA, RP) (WIn) 3.3 Intermediate surveys – see part "General", paragraph 5.13.3.3 (WIn) 3.3.1 For compliance with part I-A of the International Code for Ships Operating in Polar Waters the examination of current certificates and other records should consist of: (WIn) 3.3.1.1 the provisions of (WA) 3.2.1 except (WA) 3.2.1.4 and (WA) 3.2.1.5. (Add. to Cln) (WIn) 3.3.2 For compliance with part I-A of the International Code for Ships Operating in Polar Waters the intermediate survey of the hull, machinery and equipment should consist of: (WIn) 3.3.2.1 the provisions of (WA) 3.2.2 except (WA) 3.2.2.5 to (WA) 3.2.2.36. (Add. to Cln) (WIn) 3.3.3 For compliance with part I-A of the International Code for Ships Operating in Polar Waters the completion of the intermediate survey should consist of: (WIn) 3.3.3.1 after a satisfactory survey endorsing the Polar Ship Certificate; and (Add. to Cln) (WIn) 3.3.3.2 if a survey shows that the condition of a ship or its equipment is (Add. to unsatisfactory – see part "General", section 4.8. Cln) (WP) 3.4 Periodical surveys – see part "General", paragraph 5.13.3.4 (WP) 3.4.1 For compliance with part I-A of the International Code for Ships Operating in Polar Waters the examination of current certificates and other records should consist of: (WP) 3.4.1.1 the provisions of (WA) 3.2.1 except (WA) 3.2.1.3 and (WA) 3.2.1.5. (Add. to EP) (WP) 3.4.2 For compliance with part I-A of the International Code for Ships Operating in Polar Waters the periodical survey of the life-saving appliances and other equipment should consist of:

- (WP) 3.4.2.1 the provisions of (WA) 3.2.2 except (WA) 3.2.2.1 to (WA) 3.2.2.4 and (Add. to EP) (WA) 3.2.2.32 to (WA) 3.2.2.36.
- (WP) 3.4.3 For compliance with part I-A of the International Code for Ships Operating in Polar Waters the completion of the periodical survey should consist of:
- (WP) 3.4.3.1 after a satisfactory survey, endorsing the Polar Ship Certificate; and (Add. to EP)
- (WP) 3.4.3.2 if a survey shows that the condition of a ship or its equipment is (Add. to EP) unsatisfactory see part "General", section 4.8.
- (WR) **3.5 Renewal surveys** see part "General" paragraph 5.13.3.5
- (WR) 3.5.1 For compliance with part I-A of the International Code for Ships Operating in Polar Waters the examination of current certificates and other records should consist of:
- (WR) 3.5.1.1 the provisions of (WA) 3.2.1, except the Polar Ship Certificate.
 (Add. to PR, CR, ER, RR)
- (WR) 3.5.2 For compliance with part I-A of the International Code for Ships Operating in Polar Waters the renewal survey of the hull, machinery and equipment should consist of:
- (WR) 3.5.2.1 for cargo ships, the provisions of (WA) 3.2.2;
 (Add. to CR, ER, RR)
- (WR) 3.5.2.2 for passenger ships, the provisions of (WA) 3.2.2 except (WA) 3.2.2.18; (Add. to PR)
- (WR) 3.5.2.3 for passenger ships, examining that a proper sized immersion suit of the insulated type or a thermal protective aid is provided for each person on board according to the operational assessment (Polar Code part I-A/para.8.3.3.1.1 and 8.3.3.1.2); and
- (WR) 3.5.2.4 for ice strengthened ships, examining that sensors for navigational (Add. to PR, equipment, required either by SOLAS or the Code, projecting below the ER) hull are protected against ice (SOLAS 74/00 ch.V, Polar Code part I-A/para.9.3.2.1.4.1).

- (WR) 3.5.3 For compliance with part I-A of the International Code for Ships Operating in Polar Waters the completion of the renewal survey should consist of:
- (WR) 3.5.3.1 after a satisfactory survey, issuing the Polar Ship Certificate.
 (Add. to PR, CR, ER, RR)

Appendix 1

SUMMARY OF AMENDMENTS TO MANDATORY INSTRUMENTS REFLECTED IN THE SURVEY GUIDELINES UNDER THE HSSC

The amendments to mandatory instruments reflected in annexes 1 to 5 are summarized below to facilitate amendments to the Survey Guidelines under the HSSC in the future:

SOLAS 74	up to and including the 2018 amendments (resolution MSC.436(99))
SOLAS PROT 1988	up to and including the 2015 amendments (resolution MSC.395(95))
LLC 66	up to and including the 2005 amendments (resolution A.972(24))
LL PROT 1988	up to and including the 2014 amendments (resolution MSC.375(93))
MARPOL	up to and including the 2019 amendments (resolution MEPC.316(74))
BWM Convention	up to and including the 2018 amendments (resolution MEPC.299(72))
NO _X Technical Code 2008	up to and including the 2019 amendments (resolution MEPC.317(74))
IBC Code	up to and including the 2019 amendments (resolutions MEPC.318(74)/MSC.460(101))
IGC Code	up to and including the 2018 amendments (resolution MSC.441(99))
BCH Code	up to and including the 2019 amendments (resolutions MEPC.319(74)/MSC.463(101))
LSA Code	up to and including the 2017 amendments (resolution MSC. 425(98))
FSS Code	up to and including the 2016 amendments (resolution MSC. 410(97))
IGF Code	up to and including the 2017 amendments (resolution MSC.422(98))
Polar Code	adopted by resolutions MSC.385(94)/MEPC.264(68) in 2014/2015
BWMS Code	adopted by resolution MEPC.300(72) in 2018

Appendix 2

THE HARMONIZED SYSTEM OF SURVEY AND CERTIFICATION

Years 0	1	2	3	4	5		
Months 0	9 12 15 2	3: 21 24 27	3 36 39 45	48 51	 ► 57 60		
PASSENGER	R ←→ ·	R ←→ ◀	R → ◆	R → ·	R ◀─►		
SEC	A .	A or P	P or A	A .	R ←→		
RADIO	₽	₽	► ◀	P •	R ←→		
SAFCON	A	A or In*	In*or A ► ►	A	R⁺ ◀─►		
IGC/GC	A	A or In	In or A	A	R ◀─►		
IBC/BCH	A A	A or In	In or A	A	R ←→		
LOAD LINE	A A	A A	A	A	R ◀─►		
MARPOL Annex I	A	A or In	In or A	A .	R ←→		
MARPOL Annex II	< A ►	A or In	In or A	A	R		
MARPOL Annex IV					→ R		
MARPOL Annex VI	A →	A or In ◀ ➡ ➡	In or A	A	R		
BWM Convention	A →	A or In	In or A	A	R ←→		
Code of types of survey:							
	R – P – In – A –	Rene Peric Inter Anni	ewal odical mediate ual				

DIAGRAMMATIC ARRANGEMENT

* Items associated with the cargo ship safety construction intermediate survey that are additional to the requirements of the annual survey may, alternatively, be carried out at the second or third annual survey or between these surveys.

⁺ The cargo ship safety construction renewal survey may be commenced at the fourth annual survey and may be progressed during the succeeding year with a view to completion by the fifth anniversary date. The survey items of the fourth annual survey should not be credited to the completion of the renewal survey.
ANNEX 32

DRAFT ASSEMBLY RESOLUTION

2021 NON-EXHAUSTIVE LIST OF OBLIGATIONS UNDER INSTRUMENTS RELEVANT TO THE IMO INSTRUMENTS IMPLEMENTATION CODE (III CODE)

THE ASSEMBLY,

RECALLING Article 15(j) of the Convention on the International Maritime Organization concerning the functions of the Assembly in relation to regulations and guidelines regarding maritime safety and the prevention and control of marine pollution from ships,

RECALLING ALSO that, by resolution A.1070(28), it adopted the *IMO Instruments Implementation Code* (III Code),

RECALLING FURTHER resolution A.1141(31), by which it adopted the 2019 Non-exhaustive list of obligations under instruments relevant to the IMO Instruments Implementation Code (III Code) (hereafter referred to as the "Non-exhaustive list of obligations") for guidance on the implementation and enforcement of IMO instruments, in particular including the identification of auditable areas relevant to the IMO Member State Audit Scheme, as provided in mandatory provisions of relevant IMO instruments, following successive revocation of resolutions A.1121(30), A.1105(29) and A.1077(28),

RECOGNIZING the need for the annexes to the Non-exhaustive list of obligations to be further revised to take account of the amendments to the IMO instruments relevant to the III Code which have entered into force or become effective since the adoption of resolution A.1141(31),

RECOGNIZING ALSO that parties to the relevant international conventions have, as part of the ratification process, agreed to fully meet their responsibilities and to discharge their obligations under the conventions and other instruments to which they are party,

REAFFIRMING that it is the primary responsibility of States to have in place an adequate and effective system to exercise control over ships entitled to fly their flag and to ensure that they comply with relevant international rules and regulations in respect of maritime safety, security and protection of the marine environment,

REAFFIRMING ALSO that States, in their capacity as flag, port and coastal States, have other obligations and responsibilities under applicable international law in respect of maritime safety, security and protection of the marine environment,

NOTING that, while States may gain certain benefits by becoming party to instruments aimed at promoting maritime safety, security and protection of the marine environment, these benefits can be fully realized only when all parties carry out their obligations as required by the instruments concerned,

NOTING ALSO that the ultimate effectiveness of any instrument depends, inter alia, upon all States:

- (a) becoming party to all instruments related to maritime safety, security and pollution prevention and control;
- (b) implementing and enforcing such instruments fully and effectively; and

(b) reporting to the Organization as required,

NOTING FURTHER that, by resolutions A.1083(28), A.1084(28) and A.1085(28), it adopted amendments to the International Convention on Load Lines, 1966, the International Convention on Tonnage Measurement of Ships, 1969, and the Convention on the International Regulations for Preventing Collisions at Sea, 1972, to make the use of the III Code mandatory under these conventions,

NOTING that the Maritime Safety Committee, by resolutions MSC.366(93), MSC.373(93), MSC.374(93) and MSC.375(93), adopted amendments to the International Convention for the Safety of Life at Sea, 1974, the International Convention on Standards of Training, Certification and Watchkeeping for Seafarers, 1978, the Seafarers' Training, Certification and Watchkeeping (STCW) Code, and the Protocol of 1988 relating to the International Convention on Load Lines, 1966, respectively, to make the use of the III Code mandatory under these instruments,

NOTING ALSO that the Marine Environment Protection Committee, by resolutions MEPC.246(66) and MEPC.247(66), adopted amendments to the International Convention for the Prevention of Pollution from Ships, 1973, as modified by the Protocol of 1978 relating thereto, and the Protocol of 1997 to amend the International Convention for the Prevention of Pollution from Ships, 1973, as modified by the Protocol of 1978 relating thereto, respectively, to make the use of the III Code mandatory under these instruments,

HAVING CONSIDERED the recommendations made by the Maritime Safety Committee, at its [104th] session, and the Marine Environment Protection Committee, at its [seventy-seventh] session,

1 ADOPTS the 2021 Non-exhaustive list of obligations under instruments relevant to the IMO Instruments Implementation Code (III Code), as set out in the annex to the present resolution;

2 URGES Governments of all States, in their capacity as flag, port and coastal States, to make use of the list as much as possible in implementing IMO instruments on a national basis;

3 REQUESTS the Maritime Safety Committee and the Marine Environment Protection Committee to keep the list under review and to amend it as necessary;

4 REVOKES resolution A.1141(31).

ANNEX

2021 NON-EXHAUSTIVE LIST OF OBLIGATIONS UNDER INSTRUMENTS RELEVANT TO THE IMO INSTRUMENTS IMPLEMENTATION CODE (III CODE)

(This Non-exhaustive list of obligations takes into account the amendments to the IMO instruments relevant to the III Code which have, or are expected to have, entered into force by or on 1 July 2022)

Table of contents

- Annex 1: OBLIGATIONS OF CONTRACTING GOVERNMENTS/PARTIES
- Annex 2: SPECIFIC FLAG STATE OBLIGATIONS
- Annex 3: SPECIFIC COASTAL STATE OBLIGATIONS
- Annex 4: SPECIFIC PORT STATE OBLIGATIONS
- Annex 5: INSTRUMENTS MADE MANDATORY UNDER IMO CONVENTIONS
- Annex 6: SUMMARY OF AMENDMENTS TO MANDATORY INSTRUMENTS REFLECTED IN THE NON-EXHAUSTIVE LIST OF OBLIGATIONS (ANNEXES 1 TO 4)
- Annex 7: AMENDMENTS TO IMO INSTRUMENTS EXPECTED TO BE ACCEPTED AND TO ENTER INTO FORCE BETWEEN 1 JANUARY 2022 AND 1 JULY 2022

Annex 1

OBLIGATIONS OF CONTRACTING GOVERNMENTS/PARTIES

The following table contains a non-exhaustive list of obligations, including those obligations imposed when a right is exercised.

	OBLIGATIONS OF CONTRACTING GOVERNMENTS/PARTIES	
SOURCE	SUMMARY DESCRIPTION	COMMENTS
TONNAGE 1969		
Article 1	General obligation under the Convention	
Article 5(2)	Force majeure	
Article 8	Issue of a certificate by another Government	
Article 10	Cancellation of certificate	
Article 11	Acceptance of certificates	
Article 15	Communication of information	
ANNEX III Reg.9	Verification of compliance with the provisions of this Convention	
LL 1966 and LL PROT 1988 ¹		
Article 1	General obligation under the Convention	
	General obligations	LL PROT 1988 only (Article I)
Article 7(2)	Force majeure	
Article 17	Issue or endorsement of certificates by another Government	amended by LL PROT 1988
Article 20	Acceptance of certificates	
Article 25	Special rules drawn up by agreement	
Article 26	Communication of information	

¹ When the obligation does not derive from the International Convention on Load Lines, 1966, but solely from the Protocol of 1988 relating thereto, this is indicated in the "Comments" column.

	OBLIGATIONS OF CONTRACTING GOVERNMENTS/PARTIES	
SOURCE	SUMMARY DESCRIPTION	COMMENTS
	Communication of information	LL PROT 1988 only (Article III)
ANNEX IV Reg.54	Verification of compliance with the provisions of this Convention	
COLREG 1972		
Article I	General obligations	
STCW 1978		
Article I	General obligations under the Convention	
Article IV	Communication of information	
Article XI(1)	Promotion of technical cooperation	
Reg. I/2.12	Issue of certificates of competency	
Reg. I/2.14	Maintenance of a register or registers of all certificates and endorsements	
Regs. I/2.15 and 2.16	Availability of information on the status of certificates of competency, endorsements and dispensations (as of 1 January 2017, available in English through electronic means)	
Reg. I/3	Principles governing near-coastal voyages, communication of information and incorporation of limits in the endorsements	
Reg. I/5	National provisions – impartial investigation, enforcement measures including penalties or disciplinary measures and cooperation	
Reg. I/6.1	Training and assessment – administration, supervision and monitoring	
Reg. I/6.2	Qualification of those responsible for training and assessment	
Reg. I/7	Communication of the information as referred to in article IV of the STCW Convention and A-I/7 of the STCW Code	
Regs. I/8.1 and 8.2	Quality standards system and periodical independent evaluation	

п

	OBLIGATIONS OF CONTRACTING GOVERNMENTS/PARTIES	
SOURCE	SUMMARY DESCRIPTION	COMMENTS
Reg. I/8.3	Communication of a report containing the results of the evaluation required by regulation I/8.2	
Regs. I/9.1 and 9.2	Medical standards, procedures for the issue of medical certificates and recognition of medical practitioners	
Reg. I/9.6	Procedures governing the validity of a medical certificate which expires in the course of a voyage	
Regs. I/11.5 and 11.6	Comparison of standards of competence – determination of need for appropriate refresher and updating training or assessment and formulation or promotion of the formulation of a structure of refresher and updating courses	
Reg. II/5.3	Comparison of standards of competence – determination of the need to update qualifications for able seamen to whom certificates are issued before 1 January 2012	
Reg. III/5.3	Comparison of standards of competence – determination of the need to update qualifications for ratings in engine department to whom certificates are issued before 1 January 2012	
Reg. III/6.3	Comparison of standards of competence – determination of the need to update qualifications for electro-technical officers to whom certificates are issued before 1 January 2012	
Reg. III/7.3	Comparison of standards of competence – determination of the need to update qualifications for electro-technical ratings to whom certificates are issued before 1 January 2012	
Reg. VII/3.1	Principles governing the issue of alternative certificates	
SOLAS 1974		
Article I	General obligations under the Convention	in SOLAS PROT 1978 and SOLAS PROT 1988
Article III	Communication of information	in SOLAS PROT 1978 and SOLAS PROT 1988
Article V(c)	Carriage of persons in emergencies – reporting	

	OBLIGATIONS OF CONTRACTING GOVERNMENTS/PARTIES	
SOURCE	SUMMARY DESCRIPTION	COMMENTS
Article VII	Special rules drawn up by agreement	
Article XI	Denunciation	in SOLAS PROT 1988 (Article VII)
Reg. I/13	Issue or endorsement of certificates by another Government	in SOLAS PROT 1988
Reg. I/17	Acceptance of certificates	also Reg. I/19(b)
Reg. I/21(b)	Casualties – reporting	
Reg. IV/5-1	Global maritime distress and safety system identities – ensuring suitable arrangements	
Reg. V/5	Meteorological services and warnings	
Reg. V/6	Ice Patrol Service	
Reg. V/10	Ships' routeing	
Reg. V/11	Ship reporting systems	
Reg. V/12	Vessel traffic services	
Reg. V/13	Establishment and operation of aids to navigation	
Reg. V/31.2	Danger messages – bring to the knowledge of those concerned and communicate to other interested Governments	
Reg. V/33.1-1	Distress situations: obligations and procedures – coordination and cooperation	
Reg. VI/1.2	Appropriate information on safe carriage of cargoes	
Reg. VI/2.4.2	Approval of certified method for weighing the gross mass of cargo carried in a container	
Reg. VII/2.4	Issue of instructions on emergency response, etc.	
Reg. VII/7-1	Issue of instructions on emergency response, etc.	

	OBLIGATIONS OF CONTRACTING GOVERNMENTS/PARTIES	
SOURCE	SUMMARY DESCRIPTION	COMMENTS
MARPOL		
Article 1	General obligations under the Convention	and MARPOL PROT 1978 (Article I)
Articles 4(2) and 4(4)	Violation	
Article 5(1)	Certificates and special rules on inspection of ships – acceptance of certificates	
Article 5(4)	Certificates and special rules on inspection of ships – no more favourable treatment	
Article 6(1)	Detection of violations and enforcement of the Convention – cooperation	
Article 6(3)	Detection of violations and enforcement of the Convention – furnishing evidence	
Article 7	Undue delay to ships	
Article 8	Reports on incidents involving harmful substances	
Article 11(1) (a) to (c) and (e) (f)	Communication of information	Article 11 (1) (b) modified by MARPOL PROT 1978 (Article III
Article 12(2)	Casualties to ships – information to IMO	
Article 17	Promotion of technical cooperation	
Annex I		
Reg. 8	Issue or endorsement of a certificate by another Government	
Reg. 15.7	Control of discharge of oil – investigations (machinery spaces)	
Reg. 34.7	Control of discharge of oil – investigations (cargo area)	
Regs. 38.4 and 38.6	Consultation with IMO for circulation of information regarding reception facilities by Parties participating in regional arrangements	

	OBLIGATIONS OF CONTRACTING GOVERNMENTS/PARTIES	
SOURCE	SUMMARY DESCRIPTION	COMMENTS
Annex II		
Reg. 6.3	Categorization and listing of noxious liquid substances and other substances – establish and agree on provisional assessment and notify IMO	
Regs. 9.3.1, 9.3.2, 9.3.3 and 9.3.4	Issue or endorsement of a certificate by another Government	
Reg. 13.4	Control of discharges of residues – exemption for a prewash	
Regs. 18.3 and 18.5	Consultation with IMO for circulation of information regarding reception facilities by Parties participating in regional arrangements	
Reg. 18.5	Reception facilities and cargo unloading terminal arrangements – agreeing and establishing a date, notifying IMO	
Annex III		
Reg. 1.3	Application – issue detailed requirements	
Annex IV		
Reg. 6	Issue or endorsement of a certificate by another Government	
Reg. 12.2	Consultation with IMO for circulation of information regarding reception facilities by Parties participating in regional arrangements	
Annex V		
Reg. 8.3	Consultation with IMO for circulation of information regarding reception facilities by Parties participating in regional arrangements	
Annex VI		
Reg. 7	Issue or endorsement of a certificate by another Government	Addition related to IECC by MEPC 203(62)
Regs. 9.9.3 and 9.11.2	Transfer of flag – transmitting copies of the certificate and the relevant survey report	

п

	OBLIGATIONS OF CONTRACTING GOVERNMENTS/PARTIES	
SOURCE	SUMMARY DESCRIPTION	COMMENTS
Reg. 11.1	Detection of violations and enforcement – cooperation	
Reg. 11.2	Inspection report in case of detection of violations	
Reg. 11.3	Detection of violations and enforcement – information to flag State and master on violations detected	
Reg. 11.5	Transmission of report to requesting Party	
Reg. 13.7.1	Certification of an approved method and communication to IMO	
Reg. 17.1	Adequate reception facilities	
Reg. 17.2	Consultation with IMO for circulation of information regarding reception facilities by Parties participating in regional arrangement	
ISM Code		
Para. 14.3	Extension of validity of Interim Safety Management Certificate (Interim SMC) by another Contracting Government	
1994 HSC Code		
Para. 1.8.2	Issue of certificates by another Government	
Para. 14.2.1.12	Definition of "sea area A1"	as may be defined
Para. 14.2.1.13	Definition of "sea area A2"	as may be defined
2000 HSC Code		
Para. 1.8.2	Issue of certificates by another Government	
Para. 14.2.1.13	Definition of "sea area A1"	as may be defined
Para. 14.2.1.14	Definition of "sea area A2"	as may be defined
IMDG Code		
Section 1.3.1	Training of shore-side personnel – establishment of the period of time for keeping records of training	

	OBLIGATIONS OF CONTRACTING GOVERNMENTS/PARTIES	
SOURCE	SUMMARY DESCRIPTION	COMMENTS
Section 1.5.2	Radiation protection programme – role of competent authority	
Section 1.5.3	Management system – role of competent authority	
Para. 2.0.0.1	Classification of dangerous goods where applicable – role of competent authority	
Chapter 3.3, SP356	Approval of metal hydride storage systems installed in vehicles, vessels or aircrafts or in completed components or intended to be installed in vehicles, vessels or aircrafts	
Chapter 4.1	Approval of packaging as referred to in the chapter – role of competent authority	
Section 5.1.5	General provisions for class 7 – role of competent authority	
Section 5.4.1	Information required in addition to the dangerous goods description – role of the competent authority	
Chapter 5.5	Determining the period between fumigant application and loading of fumigated cargo transport unit on board the ship	
Chapter 6.2	Approval of pressure receptacles, aerosol dispensers, small receptacles containing gas and fuel cell cartridges containing liquefied flammable gas – role of competent authority	
Para. 6.2.2.1.1	Approval of cylinder service life test programme and the results – role of competent authority	
Para. 6.2.2.5	Approval of conformity assessment system; availability of the evidence demonstrating compliance of conformity assessment system to its counterpart in a country of use; approval of inspection bodies; and availability of a current list of approved inspection bodies and their identity marks and approved manufacturers and their identity marks – role of competent authority	
Para. 6.2.2.6	Establishment of an approval system for periodic inspection and testing of pressure receptacles; availability of evidence demonstrating compliance with	

	OBLIGATIONS OF CONTRACTING GOVERNMENTS/PARTIES	
SOURCE	SUMMARY DESCRIPTION	COMMENTS
	this approval system to its counterpart in a country of use; and availability of a current list of approved periodic inspection and testing bodies and their identity marks – role of competent authority	
Section 6.2.3	The marking of salvage pressure receptacles – determination by the competent authority	
Section 6.3.2	Quality assurance programme – role of competent authority	
Section 6.3.5	Procedures for performance and frequency of tests – role of competent authority	
Chapter 6.4	Approval of package design and materials for class 7 – role of competent authority	
Section 6.5.4	Testing, certification and inspection – role of competent authority	
Chapter 6.6	Provisions for the construction and testing of large packaging – role of competent authority	
Chapter 6.7	Provisions for the design, construction, inspection and testing of portable tanks and multiple-element gas containers – role of competent authority	
Chapter 6.8	Provisions for road tank vehicles and road gas elements vehicles- role of competent authority	
Para. 7.1.4.5	Stowage of goods of class 7 – role of competent authority	
Chapter 7.9	Exemptions, approvals and certificates – notifying IMO and recognition of approvals and certificates	
IMSBC Code		
Para. 1.7	Independence of a competent authority	
Para. 1.7	Approval of the TML test procedure, such as those specified in paragraph 1 of appendix 2	
Casualty Investigation Code		

	OBLIGATIONS OF CONTRACTING GOVERNMENTS/PARTIES	
SOURCE	SUMMARY DESCRIPTION	COMMENTS
Para. 4.1	Detailed contact information of the marine safety investigation authority/authorities to IMO	
Paras. 5.1 and 5.2	Notification of a marine casualty	
Paras. 7.1 and 7.2	Agreement to conduct a marine safety investigation	
Para. 8.1	Powers provided to investigator(s)	
Para. 9.2	Coordination for parallel investigations	
Para. 10.1	Cooperation in investigating	
Para. 11.1	Investigation not to be subject to external direction	
Paras. 13.1,13.4 and 13.5	Draft marine safety investigation reports	
Paras. 14.1 and 14.2	Marine safety investigation reports – communication to IMO	
Para. 14.4	Marine safety investigation reports – available to public and shipping industry	
IBC Code		
Para. 1.5.3	Maintenance of conditions after survey	
Para. 1.5.5.1	Issue or endorsement of International Certificate of Fitness by another Government	
BCH Code		
Para. 1.6.4.1	Issue or endorsement of certificate by another Government	
IGC Code		
Para. 1.4.5	Issue or endorsement of certificate by another Government	

	OBLIGATIONS OF CONTRACTING GOVERNMENTS/PARTIES	
SOURCE	SUMMARY DESCRIPTION	COMMENTS
STCW Code, part A		
Section A-I/6.1	Training and assessment of seafarers for certification	
Section A-I/6.3	Qualifications of instructors, supervisors and assessors	
Section A-I/6.7	Training and assessment within an institution	
Section A-I/7.2	Communication of information – initial communication (within one year of entry into force of regulation I/7)	
Sections A-I/7.3, 7.4, 7.5 and 7.6	Communication of information – subsequent reports (within the periods as referred to in paragraphs 7.3, 7.4, 7.5 and 7.6)	
Sections A-I/8.1 and 8.3	National objectives and quality standards	
Section A-I/9.1	Medical standards – eyesight standards and physical and medical fitness	
Section A-1/9.4	Provisions for recognizing medical practitioners and maintenance of a register of recognized medical practitioners	
Sections A-I/9.5 and 9.6	Guidance, processes and procedures for conducting medical fitness examinations and issuing medical certificates	
Section A-I/12.1	General performance standards for simulators used in training	
Section A-I/12.2	General performance standards for simulators used in assessment of competence	
Section A-I/12.6	Simulator training objectives	
Section A-I/12.9	Qualification of instructors and assessors	
Section A-VIII/2.9	Watchkeeping at sea – directing the attention of companies, masters, chief engineer officers and watchkeeping personnel to observe the principles in parts 4-1 and 4-2	

	OBLIGATIONS OF CONTRACTING GOVERNMENTS/PARTIES	
SOURCE	SUMMARY DESCRIPTION	COMMENTS
III Code	IMO Instruments Implementation Code	
	Application of the provisions of the III Code under related mandatory IMO Instruments	

Annex 2

SPECIFIC FLAG STATE OBLIGATIONS

The following table contains a non-exhaustive list of obligations, including those obligations imposed when a right is exercised.

SPECIFIC FLAG STATE OBLIGATIONS				
SOURCE	SUMMARY DESCRIPTION	COMMENTS		
TONNAGE 1969				
Article 6	Determination of tonnages			
Article 7(2)	Issue of certificates			
Article 9	Form of certificate			
Annex I, Reg. 1(3)	Novel types of craft – determination of tonnage and communication to IMO on method used			
Annex I, Reg. 5(3)(b)	Change of net tonnage – alterations or modifications deemed by the Administration to be of a major character			
Annex I, Reg. 7	Measurement and calculation			
LL 1966 and LL PROT 1988 ²	Existing certificates	LL PROT 1988 only (Article II-2)		
Article 6(3)	Exemptions – reporting			
Article 8(2)	Equivalents – reporting			
Article 9(2)	Approvals for experimental purposes – reporting			
Article 13	Surveys and marking	amended by LL PROT 1988		
Article 14	Initial, renewal and annual survey	amended by LL PROT		
Article 16(3)	Issue of certificates			
Article 18	Form of certificates			

² When the obligation does not derive from the International Convention on Load Lines, 1966, but solely from the Protocol of 1988 relating thereto, this is indicated in the "Comments" column.

SPECIFIC FLAG STATE OBLIGATIONS				
SOURCE	SUMMARY DESCRIPTION	COMMENTS		
Article 19 Article 23	Duration and validity of certificates Casualties	amended by LL PROT 1988		
Annex I, Reg. 1	Strength of hull			
	Strength and intact stability of ships	LL PROT 1988 only (annex I, Reg. 1)		
Annex I, Reg. 2	Application – assignment of freeboard	amended by LL PROT 1988		
Annex I, Reg. 2-1	Authorization of recognized organizations	LL PROT 88 only		
Annex I, Reg. 8	Details of marking			
Annex I, Reg. 10	Stability information – approval	amended by LL PROT 1988		
Annex I, Reg. 12	Doors	amended by LL PROT 1988		
Annex I, Reg. 14	Cargo and other hatchways	amended by LL PROT 1988		
Annex I, Reg. 15	Hatchways closed by portable covers and secured weathertight by tarpaulins and battening devices	amended by LL PROT 1988		
Annex I, Reg. 16(1)	Hatchway coamings – reduced heights	amended by LL PROT 1988		
		(annex I, Reg. 14-1(2))		
Annex I, Reg. 16(4)	Securing arrangements	amended by LL PROT 1988		
		(annex I, Reg. 16(6))		
	Machinery space openings	LL PROT 1988 only (annex I, Reg. 17(4))		
Annex I, Reg. 19	Ventilators	amended by LL PROT 1988		
Annex I, Reg. 20	Air pipes	amended by LL PROT 1988		

_

SPECIFIC FLAG STATE OBLIGATIONS					
SOURCE	SUMMARY DESCRIPTION	COMMENTS		6	
	Cargo ports and other similar openings – applicable national standards	LL PROT (annex I, F	1988 Reg.	8 onl <u>9</u> 21(5	y 5))
Annex I, Reg. 22	Scuppers, inlets and discharges	amended	by	LL	PROT
Annex I, Reg. 25	Protection of the crew	amended 1988	by	LL	PROT
Annex I, Reg. 27	Freeboards – types of ships	amended	by	LL	PROT
Annex I, Reg. 28	Freeboard tables	amended 1988	by	LL	PROT
Annex I, Reg. 39	Minimum bow height and reserve buoyancy	amended 1988	by	LL	PROT
	Lashing system	LL PROT (annex I, F	1988 Reg.	3 onl <u>)</u> 44(6	y 5))
COLREG 1972					
Annex I, Para. 14	Approval of construction of lights and shapes and the installation of lights on board				
Annex III, Para. 3	Approval of construction, performance and installation of sound signal appliances on board				
STCW 1978					
Article VI	Certificates				
Article VIII(3)	Dispensation – reporting				
Article IX(2)	Equivalents – reporting				
Regs. I/2.1, 2.2, 2.7 and 2.8	Issue and endorsements of certificates of competency				
Regs. I/10.1 and 10.2	Recognition of certificates and seafarers' knowledge of maritime legislation				
Reg. I/11.7	Availability – recent changes in national and international regulations				
Reg. I/13.3	Conduct of trials – safety, security and				
Regs. I/13.5 and 13.8.1	Results of trials – communication				
Reg. I/13.7	Respect for objections to trials				

		<u>. </u>		
SPECIFIC FLAG STATE OBLIGATIONS				
SOURCE	SUMMARY DESCRIPTION	COMMENTS		
Reg. I/13.8	Respect for objections by other Parties when authorizing ships to continue to operate with the system under trial			
Reg. I/14.1	Responsibilities of companies			
Reg. IV/1.2	Application – appropriate certificates for radio operators			
Reg. V/1-1.7	Mandatory minimum requirements for the training and qualification of masters, officers and ratings on oil and chemical tankers			
Reg. V/1-2.5	Mandatory minimum requirements for the training and qualification of masters, officers and ratings on liquefied gas tankers			
Reg. V/2.1	Applicability of the requirements to domestic voyages			
Reg. V/2.10	Mandatory minimum requirements for the training and qualification of masters, officers, ratings and other personnel on passenger ships			
Reg. V/3.11	Issue of a Certificate of Proficiency to seafarers, related to IGF Code			
Reg. V/4.5	Issue of a Certificate of Proficiency in accordance with paragraphs 2 or 4 of the regulation			
Regs. VIII/1.1 and 1.2	Fitness for duty – preventing fatigue and preventing drug and alcohol abuse			
Regs. VIII/2.1 and 2.2	Watchkeeping arrangements and principles – direction and requirements			
SOLAS 1974				
Reg. I/4(b)	Exemptions – reporting			
Reg. I/5(b)	Equivalents – reporting			
Reg. I/6	Inspection and survey	in SOLAS PROT 1978 and SOLAS PROT 1988		

SPECIFIC FLAG STATE OBLIGATIONS			
SOURCE	SUMMARY DESCRIPTION	COMMENTS	
Reg. I/7	Survey of passenger ships	in SOLAS PROT 1988	
Reg. I/8	Survey of life-saving appliances and other equipment of cargo ships	in SOLAS PROT 1988	
Reg. I/9	Survey of radio installations of cargo ships	in SOLAS PROT 1988	
Reg. I/10	Survey of structure, machinery and equipment of cargo ships	in SOLAS PROT 1988	
Reg. I/12	Issue of certificates	in SOLAS PROT 1988	
	Issue and endorsement of certificates	in SOLAS PROT 1988	
Reg. I/14	Duration and validity of certificates	in SOLAS PROT 1988	
Reg. I/15	Forms of certificates and records of equipment	in SOLAS PROT 1988	
Reg. I/18	Qualification of certificates		
Reg. I/21	Casualties		
Regs. II-1/1.1.2 and 1.2	Compliance with earlier requirements		
Reg. II-1/3-1	For ships which are not designed, constructed and maintained in compliance with the requirements of a classification society recognized by the Administration, the Administration should have in place applicable national standards which provide an equivalent level of safety.		
Reg. II-1/3-2	Approval of corrosion prevention systems of seawater ballast tanks		
Reg. II-1/3-2.4	Maintenance of the protective coating		
Reg. II-1/3-3.2	Approval of means of access to tanker bows		
Regs. II-1/3-4.1.2.2 and 3-4.1.3	Approval of emergency towing arrangements on tankers		
Reg. II-1/3-6.2.3	Means of access to cargo and other spaces – satisfaction of the Administration as well as survey		

SPECIFIC FLAG STATE OBLIGATIONS				
SOURCE	SUMMARY DESCRIPTION	COMMENTS		
Reg. II-1/3-6.4.1	Approval of Ship Structure Access Manual			
Reg. II-1/3-8.3	Appropriate requirements for towing and mooring equipment			
Reg. II-1/3-9.1	Means of embarkation and disembarkation			
Reg. II-1/3-12	Determination of acceptable noise levels in machinery spaces			
Reg. II-1/4.3	Alternative methodologies – communication to IMO			
Reg. II-1/4.5	Beneficial or adverse effects of fitting structures as defined by the regulation			
Reg. II-1/5-1.1	Stability information to the Administration			
Reg. II-1/7-2.5	Acceptance of equalization devices and their controls			
Reg. II-1/13.9.2	Number and arrangements of doors with a device preventing unauthorized opening			
Reg. II-1/13.11.2	Special consideration for tunnels piercing watertight bulkheads			
Reg. II-1/15.2	Arrangement and efficiency of the means for closing any opening in the shell plating			
Reg. II-1/15.6	Special sanction for automatic ventilating side scuttles			
Reg. II-1/15.8.5	Material of pipes as referred to in the regulation			
Reg. II-1/16-1.1	Design, material and construction of all watertight closures, such as doors, hatches, side scuttles, gangway and cargo ports, etc.			

SPECIFIC FLAG STATE OBLIGATIONS			
SOURCE	SUMMARY DESCRIPTION	COMMENTS	
Reg. II-1/17-1.2	Indicators for closing appliances that could lead to flooding of a special category space or ro-ro space		
Regs. II-1/19.2 and 19.3	Damage control information – general and specific precautions		
Reg. II-1/22.3	Authorization of such a watertight door that may be opened during navigation		
Reg. II-1/26.2	Consideration of reliability of single essential propulsion components		
Regs. II-1/29.1, 29.2.1 and 29.6.3	Steering gear		
Reg. II-1/29.17.2	Adoption of regulations on rudder actuators for tankers, chemical tankers and gas carriers		
Regs. II-1/35-1.3.7.2 and 35-1.3.9	Bilge pumping arrangements		
Reg. II-1/40.2	Electrical installations – ensuring uniformity		
Reg. II-1/42.1.3	Emergency source of electrical power in passenger ships		
Reg. II-1/43.1.3	Emergency source of electrical power in cargo ships		
Reg. II-1/44.2	Approval of automatically starting emergency generating sets		
Regs. II-1/45.3.3, 45.5.3, 45.5.4, 45.9.3, 45.10 and 45.11	Precautions against shock, fire and other hazards of electrical origin		
Regs. II-1/46.2 and 46.3	Additional requirements for periodically unattended machinery space		
Reg. II-1/53.1	Special requirements for machinery, boiler and electrical installations		
Regs. II-1/55.3, 55.4.1 and 55.6	Evaluation of the alternative design and arrangements and re-evaluation due to change of conditions		

SPECIFIC FLAG STATE OBLIGATIONS			
SOURCE	SUMMARY DESCRIPTION	COMMENTS	
Reg. II-1/55.5	Alternative design and arrangements – communication to IMO		
Reg. II-2/1.2.1	Approval of fire protection arrangements in existing ships		
Regs. II-2/1.6.2.1.2 and 1.6.6	Application of requirements for tankers		
Reg. II-2/4.2.2.5.1	Approval of material for oil fuel pipes and their valves and fittings		
Reg. II-2/4.3	Approval of gaseous fuel systems used for domestic purposes		
Reg. II-2/4.5.1.4.4	Installation of cargo oil lines where cargo wing tanks are provided		
Reg. II-2/4.5.3.3	Requirements for safety devices in venting systems		
Reg. II-2/4.5.5.2.1	Requirements for inert gas systems on chemical tankers		
Reg. II-2/4.5.6.3	Arrangements for inerting, purging or gas-freeing	see Reg. II-2/4.5.5.3.1	
Reg. II-2/5.2.2.5	Positioning of controls for any required fire-extinguishing system in passenger ships	see Regs. II-2/8.3.3 and 9.5.2.3	
Reg. II-2/5.2.3.1	Special consideration to maintaining the fire integrity of periodically unattended machinery spaces		
Reg. II-2/7.3.2	Initial and periodic tests		
Reg. II-2/7.6	Protection of cargo spaces in passenger ships		
Reg. II-2/8.3.4	Release of smoke from machinery spaces – passenger ships		
Reg. II-2/9.2.2.1.5.1	Approval of equivalent means of controlling and limiting a fire on ships designed for special purposes		

SPECIFIC FLAG STATE OBLIGATIONS			
SOURCE	SUMMARY DESCRIPTION	COMMENTS	
Reg. II-2/9.2.2.3.1	Fire integrity of bulkheads and decks in ships carrying more than 36 passengers		
Regs. II-2/9.2.2.4.4, 9.2.3.3.4 and 9.2.4.2.4	Fire integrity of bulkheads and decks	see Reg. II-2/11.2	
Reg. II-2/9.3.4	Approval of structural fire protection details, taking into account the risk of heat transmission		
Reg. II-2/9.5.2.4	Protection of openings in machinery space boundaries		
Reg. II-2/10.2.1.2.1.3	Provisions for fixed water fire- extinguishing arrangements for periodically unattended machinery spaces		
Reg. II-2/10.2.1.2.2.1	Ready availability of water supply		
Reg. II-2/10.2.3.1.1	Approval of non-perishable material for fire hoses		
Reg. II-2/10.2.3.2.1	Number and diameter of fire hoses		
Reg. II-2/10.3.2.1	Arrangement of fire extinguishers		
Reg. II-2/10.6.1.1	Type approval of automatic sprinkler, fire detection and fire alarm system		
Reg. II-2/10.6.3.2	Approval of fire-extinguishing arrangement for flammable liquid lockers		
Reg. II-2/10.7.1.2	Fixed gas fire-extinguishing systems for general cargo		
Reg. II-2/10.7.1.4	Issue of an Exemption Certificate		
Reg. II-2/13.3.1.4	Provision of means of escape from, or access to, radiotelegraph stations		
Reg. II-2/13.3.2.5.1	Lighting or photoluminescent equipment to be evaluated, tested and applied in accordance with the FSS Code		

SPECIFIC FLAG STATE OBLIGATIONS			
SOURCE	SUMMARY DESCRIPTION	COMMENTS	
Reg. II-2/13.3.2.6.2	Normally locked doors that form part of an escape route – quick release mechanisms		
Reg. II-2/13.5.1	Means of escape on passenger ships from special category and open ro-ro spaces to which any passengers carried can have access		
Regs. II-2/17.4.1 and 17.6	Evaluation and approval of the engineering analysis for alternative design and arrangements for fire safety		
Reg. II-2/17.5	Alternative design and arrangements for fire safety – communication of information to IMO		
Reg. II-2/19.4	Provision of document of compliance		
Reg. II-2/20.4.1	Provision and approval of fixed fire detection and fire alarm systems		
Reg. II-2/20.6.1.4.2	Adverse effect as referred to in the regulation – approval of stability information		
Reg. II-2/21.5.2	Alternate space for medical care		
Reg. III/4	Evaluation, testing and approval of life-saving appliances and arrangements		
Reg. III/5	Production tests for life-saving appliances		
Reg. III/20.8.1.2	Approval of servicing stations		
Reg. III/20.8.5	Extension of liferaft service intervals – notifying IMO		
Regs. III/20.11	Maintenance, thorough examination, operational testing, overhaul and repair of lifeboats, rescue boats and fast rescue boats, launching appliances and release gear		
Reg. III/26.2.4	Approval of liferafts on ro-ro passenger ships		

SPECIFIC FLAG STATE OBLIGATIONS			
SOURCE	SUMMARY DESCRIPTION	COMMENTS	
Regs. III/26.3.1 and 26.3.2	Approval of fast rescue boats and their launching appliances on ro-ro passenger ships		
Reg. III/28	Approval of helicopter landing and pick- up areas on ro-ro passenger ships		
Regs. III/38.3, 38.4.1 and 38.6	Evaluation of the alternative design and arrangements and re-evaluation due to change of conditions		
Reg. III/38.5	Alternative design and arrangements – communication to IMO		
Reg. IV/3.3	Exemptions – reporting to IMO		
Reg. IV/14.1	Type approval of radio equipment		
Reg. IV/15.5	Ensuring that radio equipment is maintained		
Reg. IV/16.1	Radio personnel		
Reg. IV/17	Radio records		
Reg. V/1.4	Application – determination of the scope of application of specific regulations to certain categories of ships		
Reg. V/3.3	Exemptions and equivalents – reporting to IMO		
Reg. V/14	Ships' manning		
Reg. V/16	Maintenance of equipment		
Reg. V/17	Electromagnetic compatibility		
Reg. V/18.1	Type approval of navigational systems and equipment and voyage data recorder		
Reg. V/18.5	Requirement for quality control system at manufacturers		
Reg. V/23.3.3.1.3	Pilot transfer arrangements		
Regs. VI/3.1 and 3.2	Provision of equipment for oxygen analysis and gas detection and training of crews in their use		

٦

SPECIFIC FLAG STATE OBLIGATIONS			
SOURCE	SUMMARY DESCRIPTION	COMMENTS	
Reg. VI/5.6	Approval of Cargo Securing Manual		
Reg. VI/6	Acceptability for shipment		
Reg. VI/9.2	Grain loading information		
Reg. VII/5	Approval of Cargo Securing Manual		
Reg. VII/15.2	Warships – INF cargo		
Reg. VIII/4	Approval of design, construction and standards of inspection and assembly of reactor installations		
Reg. VIII/6	Ensuring radiation safety		
Reg. VIII/7(a)	Approval of safety assessment		
Reg. VIII/8	Approval of operating manual		
Reg. VIII/10(f)	Issue of certificates		
Reg. IX/4.1	Issue of Document of Compliance (DOC)		
Reg. IX/4.3	Issue of Safety Management Certificate (SMC)		
Reg. IX/6.1	Periodic verification of the safety management system		
Reg. XI-1/1	Authorization of recognized organizations		
Reg. XI-1/2	Enhanced surveys		
Reg. XI-1/3.5.4	Ship identification number – approval of method of marking		
Reg. XI-1/3-1.2	Registered owner identification number		
Reg. XI-1/5.3	Issue of Continuous Synopsis Record (CSR)		
Reg. XI-1/5.4.2	Amendments to CSR		

г

_

	SPECIFIC FLAG STATE OBLIGATIONS	
SOURCE	SUMMARY DESCRIPTION	COMMENTS
Reg. XI-1/5.4.3	Authorization and requirement to amend CSR	
Reg. XI-1/5.8	Requirement for former flag State to send CSR to new flag State	
Reg. XI-1/5.9	Appending previous CSR to new CSR	
Reg. XI-1/6	Investigations of marine casualties and incidents	
Reg. XII/8.1	Endorsement of booklet required by Reg. VI/7.2	
Reg. XII/9.2	Approval of bilge well high water level alarms	
Reg. XII/11.3	Loading instrument – approval of software for stability calculations	
MARPOL		
Articles 4(1) and 4(3)	Violation	
Article 6(4)	Detection of violations and enforcement of the Convention – investigations	
Article 12(1)	Casualties to ships – investigations	
Annex I		
Reg. 2.6.2	Application – an oil tanker delivered on or before 1 June 1982 engaged in specific trades: agreement with port States	
Reg. 3.3	Exemptions and waivers – reporting	
Reg. 4.3	Exceptions – discharge of substances containing oil for the purpose of combating pollution incidents	
Reg. 5.2	Equivalents – reporting	
Reg. 6	Surveys	

SOURCE	SUMMARY DESCRIPTION	COMMENTS
Reg. 7	Issue or endorsement of certificate	
Reg. 9	Form of certificate	
Reg. 10.9.3	Transfer of flag	
Reg. 12A.12	Oil fuel tank protection – approval of the design and construction of ships	
Reg. 14.3	Oil filtering equipment – volume of oil bilge holding tank	
Reg. 14.4	Oil filtering equipment – ships of less than 400 gross tonnage	
Regs. 14.6 and 14.7	Oil filtering equipment – approval	
Reg. 15.6.2	Control of discharge of oil – ships of less than 400 gross tonnage: design approval	
Reg. 17.1	Approval of an electronic record book concerning Oil Record Book Part I	
Regs. 18.8.2, 18.8.3 and 18.8.4	Requirements for product carriers of 40,000 tonnes deadweight and above – arrangement and operation, approval of oil content meter, clean ballast tank operational manual	
Reg. 18.10.1.1	Segregated ballast tanks – oil tanker delivered on or before 1 June 1982 having special ballast arrangements: approval	
Reg. 18.10.1.2	Segregated ballast tanks – oil tanker delivered on or before 1 June 1982 having special ballast arrangements: agreement with port States	
Reg. 18.10.3	Segregated ballast tanks – oil tanker delivered on or before 1 June 1982 having special ballast arrangements: communication to IMO	

SPECIFIC FLAG STATE OBLIGATIONS		
SOURCE	SUMMARY DESCRIPTION	COMMENTS
Reg. 20.8.1	Double hull and double bottom requirements for oil tankers delivered before 6 July 1996 – communication to IMO	
Reg. 21.8.1	Prevention of oil pollution from oil tankers carrying heavy grade oil as cargo – communication to IMO	
Reg. 23.3.1	Accidental oil outflow performance – calculation of mean oil outflow parameter	
Reg. 25.5	Hypothetical outflow of oil – information to IMO on accepted arrangements	
Reg. 27.3	Intact stability – approval of written procedures for liquid transfer operation	
Reg. 28.3.4	Subdivision and damage stability – sufficient stability during flooding	
Reg. 28.6.3	Subdivision and damage stability – issuing a document of approval for the stability instrument	
Reg. 29.2.1	Slop tanks – approval	
Reg. 30.6.5.2	Pumping, piping and discharge arrangement – establishment of requirements	
Reg. 30.7	Pumping, piping and discharge arrangement – positive means of loading, transporting or discharging cargo	
Regs. 31.2 and 31.4	Oil discharge monitoring and control system – approval	
Reg. 32	Oil/water interface detector – approval	
Reg. 33.1	Crude oil washing requirement – compliance with requirement	
Reg. 33.2	Crude oil washing requirements – establishment of requirements	

SPECIFIC FLAG STATE OBLIGATIONS		
SOURCE	SUMMARY DESCRIPTION	COMMENTS
Reg. 35.1	Crude oil washing operations – Operations and Equipment Manual	
Reg. 36.1	Approval of an electronic record book concerning Oil Record Book Part II	
Reg. 36.9	Oil Record Book Part II – development of Oil Record Book for ships of less than 150 gross tonnage	
Reg. 37.1	Shipboard oil pollution emergency plan – approval	
Reg. 38.9.2	Reception facilities within special areas: Antarctic area – sufficient capacity	
Reg. 38.10	Reception facilities – notification on alleged inadequacies of port reception facilities	
Reg. 39.2.2	Special requirements for fixed or floating platforms – approval of record form	
Reg. 41.1	Approval of oil tankers' ship-to-ship (STS) operations plan	
Annex II		
Reg. 3.1.3	Exceptions – approval of discharge of noxious liquid substances (NLS) for the purpose of combating pollution incidents	
Reg. 4.1.2	Exemptions – communication to IMO on relaxations	
Reg. 4.3.4	Exemptions – communication to IMO	
Reg. 4.4.5	Exemptions – communication to IMO	
Reg. 5.1	Equivalents – substitution of operational method	
Reg. 5.2	Equivalents – communication to IMO on alternatives	

SPECIFIC FLAG STATE OBLIGATIONS			
SOURCE	SUMMARY DESCRIPTION	COMMENTS	
Regs. 5.3.4 and 5.3.5	Equivalents – pumping and piping arrangement, approval of manual		
Reg. 6.3	Establishment of tripartite agreements – notifying IMO		
Reg. 8	Surveys		
Reg. 9	Issue or endorsement of certificates		
Reg. 10.7	Expiry date of existing certificate		
Reg. 10.9.3	Transfer of flag		
Reg. 11.2	Design, construction, equipment and operations – establishment of appropriate measures		
Reg. 12.5	Pumping, piping, unloading arrangements and slop tanks – approval of pumping performance test		
Reg. 13.3	Control of discharges of residues of NLS – approval of ventilation procedure		
Reg. 13.5	Control of discharges of residues of NLS – approval of tank washing procedure		
Reg. 14.1	Procedures and arrangements manual – approval		
Reg. 15.1	Approval of an electronic record book concerning Cargo Record Book		
Reg. 17.1	Shipboard marine pollution emergency plan for NLS – approval		
Reg. 18.7	Notification of alleged inadequacies of port reception facilities		

SPECIFIC FLAG STATE OBLIGATIONS		
SOURCE	SUMMARY DESCRIPTION	COMMENTS
Annex IV		
Reg. 4	Surveys	
Reg. 5	Issue or endorsement of certificates	
Reg. 7	Form of certificate	
Reg. 8.8.2	Transfer of flag	
Reg. 9.1	Approval of sewage systems	
Reg. 9.2	Approval of sewage systems (passenger ships operating in special areas)	
Reg. 11.1.1	Approval of rate of discharge	
Reg. 12.3	Notification of alleged inadequacies of port reception facilities	
Annex V		
Reg. 6.3.2	Sufficient capacity for the retention of all garbage on board ships before entering the Antarctic area	
Reg. 8.4	Notification of alleged inadequacies of port reception facilities	
Reg. 10.3	Approval of an electronic record book concerning Garbage Record Book	
Annex VI		
Regs. 3.2 and 3.3.2	Exceptions and exemptions	
Regs. 4.2 and 4.4	Equivalents and communication to IMO	
Reg. 5	Surveys and certification	
Reg. 6	Issue or endorsement of certificates and Statements of Compliance related to fuel oil consumption reporting	
Reg. 8	Form of Certificates and Statements of Compliance related to fuel oil consumption reporting	

SPECIFIC FLAG STATE OBLIGATIONS		
SOURCE	SUMMARY DESCRIPTION	COMMENTS
Regs. 9.1 and 9.10	Duration and validity of certificates	
Reg. 9.12	Duration and validity of Statement of Compliance related to fuel oil consumption reporting	
Reg. 9.9.3	Transfer of flag	
Reg. 11.4	Detection of violations and enforcement – investigations and communication to the Party and IMO	
Regs. 12.6, 13.5.3 and 14.6	Approval of an electronic record book concerning the required entries for discharge, transfers and other operations as required under Annex VI	
Regs. 13.1.1.2 and 13.1.2.2	Nitrogen oxides – acceptance of identical replacement and alternative control measures	
Reg. 13.2.2	Acceptance of installation of Tier-II engine in lieu of Tier-III engine where Tier-III engine cannot be accommodated	
Reg. 13.5.2.2	Combined nameplate diesel engine – application as referred to in the paragraph	
Reg. 13.7.2	Approved method not commercially available	
Reg. 15.5	Volatile organic compounds (VOCs) – approval of vapour collection systems	
Reg. 15.6	VOC management plan – approval	
Reg. 16.6.1	Shipboard incineration – approvals	
Reg. 17.4	Reception facilities unavailable or inadequate – communication to IMO	
Reg. 19.6	Information regarding decision to allow, suspend, withdraw or decline application of Reg. 19.4 on waiving the requirement	Refer to Reg. 19.4

SPECIFIC FLAG STATE OBLIGATIONS		
SOURCE	SUMMARY DESCRIPTION	COMMENTS
	for compliance with Regs. 20 and 21 – communication to the Organization	
Reg. 20.1	Verification of the attained Energy Efficiency Design Index (EEDI)	
Reg. 22A.7	Verification of the ship fuel oil consumption data	
Reg. 22A.9	Transfer of the reported data to the IMO Ship Fuel Oil Consumption Database	
Reg. 23	Cooperation with other parties to promote development and transfer of technology and exchange of information relating to the improvement of energy efficiency of ships	
Appendix IV, Para. 1	Type approval as referred to in the paragraph	
Appendix VI, Paras. 1.2, 2.1 and 3.1	Fuel verification procedure – management and sample delivery	
Res. MSC.133(76), as amended	Technical provisions for means of access for inspections	
Para. 3.7	Vertical or spiral ladders – acceptance	
Para. 3.9.7	Other means of access – approval and acceptance	
RO Code		
Part 1/4.2	Delegation of authority	
Part 1/5	Communication to IMO – a list of recognized organizations (ROs)	
Part 2/8	Authorization of RO	
Part 3/7.1.1	Verification that ROs fulfil the requirements of the RO Code as per part 2/2.1	
ISM Code		
Para. 13.2	Issue of Document of Compliance (DOC)	

SPECIFIC FLAG STATE OBLIGATIONS		
SOURCE	SUMMARY DESCRIPTION	COMMENTS
Para. 13.4	Annual verification (DOC)	
Para. 13.5	Withdrawal of DOC	
Para. 13.7	Issue of Safety Management Certificate (SMC)	
Para. 13.8	Intermediate verification (SMC)	
Para. 13.9	Withdrawal of SMC	
Para. 14.1	Issue of Interim DOC	
Para. 14.2	Issue of Interim SMC	
Para. 14.4	Verification required for issue of Interim SMC	
Para. 15.1	Verification – acceptance of procedures	
Para. 16	Forms of certificates	
INF Code		
Para. 1.3.2	Issue of certificate	
Para. 2.1	Damage stability (INF 1 ship)	
Para. 3.1	Fire safety measures (INF 1 cargo)	
Para. 4.1.3	Temperature control of cargo spaces (INF 1, 2 and 3 ships)	
Para. 6.2	Safe stowage and securing – approval of principles	
Para. 7.1	Electrical power supplies (INF 1 ship)	
Chapter 8	Radiological protection	
Chapter 9	Management and training	
Para. 10.2	Shipboard emergency plan – approval	
Noise Code		
Para. 1.3.1	Application of the Code	
SPECIFIC FLAG STATE OBLIGATIONS		
---------------------------------	---	----------
SOURCE	SUMMARY DESCRIPTION	COMMENTS
Para. 1.3.7	Special consideration for ships designed for and employed on voyages of short duration, ³ or on other services involving short periods of operation of the ship	
Para. 1.3.9	Repairs, alterations and modifications of a major character and outfitting of existing ships – determination of application of the Code	
Para. 2.1.1	Acceptance of equivalent standard for sound level meter	
Para. 2.1.2	Acceptance of equivalent standard for octave filter set	
Para. 3.3.9	Agreement on a process to simulate the operation of the dynamic positioning (DP) thruster system	
Para. 6.2.2	Acceptance of laboratory tests of airborne sound insulation properties	
Para. 7.4	Description of a warning notice comprising symbol and supplementary sign in working language	
FSS Code		
Ch.1, Para. 4	Use of toxic extinguishing media	
Ch.4, Para. 2	Type approval of fire extinguishers	
Ch.4, Para. 3.1.1.2	Determining equivalents of fire extinguishers	
Ch.4, Para. 3.2.2.2	Approval of foam concentrate	
Ch.5, Para. 2.1.1.4	Containers for the storage of fire-extinguishing medium, etc.	
Ch.5, Para. 2.1.2.1	System flow calculations	
Ch.5, Para. 2.1.2.3	Spare parts	

³ **Voyages of short duration:** Voyages where the ship is not generally underway for periods long enough for seafarers to require sleep, or long off-duty periods, during the voyages (Noise Code Para. 1.4.34).

	SPECIFIC FLAG STATE OBLIGATION	6
SOURCE	SUMMARY DESCRIPTION	COMMENTS
Ch.5, Para. 2.3	Steam systems	
Ch.5, Para. 2.4	Equivalent systems – approval	
Ch.6, Para. 1	Specific acceptance of fixed foam fire-extinguishing systems used for cargo pump-rooms of chemical tankers carrying liquid cargoes referred to in regulation II-2/1.6.2 of the Convention	
Ch.6, Paras. 3.1.2 and 4.1.1	Foam concentrates – approval	
Ch.6, Paras. 3.1.3 and 3.4.1	System test of fixed high-expansion foam fire-extinguishing systems	
Ch.6, Para. 4.2.1	Acceptance of means for proving effective distribution of the foam	
Ch.7, Para. 2.1	Fixed pressure water-spraying fire-extinguishing systems – approval	
Ch.7, Para. 2.2	Equivalent systems – approval	
Ch.7, Para. 2.3	Fixed pressure water-spraying fire-extinguishing systems for cabin balconies – approval	
Ch.7, Para. 2.4	Fixed water-based firefighting system for ro-ro space, vehicle spaces and special category spaces – approval	
Ch.8, Para. 2.1.2	Equivalent sprinkler systems – approval	
Ch.9, Para. 2.3.1.2	Sensitivity limits of smoke detectors in other spaces	
Ch.9, Para. 2.3.1.3	Heat detector temperature limits	
Ch.9, Para. 2.3.1.7	Fixed fire detection and fire alarm systems for cabin balconies – approval	
Ch.9, Para. 2.4.1.3	Limiting the number of enclosed spaces included in each section	
Ch.9, Para. 2.5.2	Testing on ships with self-diagnostic system – determination of requirements	

_

SPECIFIC FLAG STATE OBLIGATIONS		
SOURCE	SUMMARY DESCRIPTION	COMMENTS
Ch.10, Para. 2.1.2	Sequential scanning – overall response time	
Ch.10, Para. 2.2.2	Extractor fans – overall response time	
Ch.10, Para. 2.3.1.1	Means to isolate smoke accumulators	
Ch.11, Para. 2.1	Low-location lighting – approval	
Ch.12, Para. 2.2.2.1	Approval of heating arrangement of the diesel engine cooling water or lubricating oil system if the room for the diesel driven power source is not heated	
Ch.14, Para. 2.2.1.4	Foam concentrate supplied on board for cargoes intended to be carried – approval of additional arrangements if foam is not effective or is incompatible	
Ch.14, Para. 2.2.2.1	Medium expansion ratio foam – application rate, etc.	
Ch.15, Para. 2.2.1.1	Inert gas systems – approval	
Ch.17, Para.3.7 and 3.8	Manufacture and test of foam fire-fighting system and its components, including deck integrated foam nozzles	
2010 FTP Code		
Para. 4.2	Recognition of testing laboratories	
Paras. 5.1.1 and 5.1.2	Approval of products in accordance with established approval procedures or authorization of competent authorities to issue approvals	
Para. 5.2.2	Requirement for manufacturers to have a quality control system audited by a competent authority, or, alternatively, use of final product verification procedures as referred to in the paragraph	
Para. 7.2	Use of equivalents and modern technology – communication of information to the Organization	
Annex 1, part 3/3.3	Structural core of a material other than steel or aluminium alloy – decision on limits for rise in temperature	

SPECIFIC FLAG STATE OBLIGATIONS		
SOURCE	SUMMARY DESCRIPTION	COMMENTS
Annex 1, part 3 appendix 1, Para. 2.3.2.9	Insulation system of "A" class door – approval to the same standard as that which the door is intended to achieve	
LSA Code		
Para. 1.2.3	Determine the period of acceptability of life-saving appliances (LSAs) subject to deterioration with age	
Para. 4.4.1.2	Endorsement of lifeboat affixed approval plate	
Para. 4.5.4	Fixed two-way VHF radiotelephone apparatus – sheltered space	
Para. 5.1.1.4	Rescue boats – combination of rigid and inflatable construction	
Para. 5.1.3.8	Rubbing strips on inflated rescue boats	
Paras. 6.1.2.9 and 6.1.2.10	Lowering speed of a fully equipped liferaft	
Para. 6.2.1.2	Marine evacuation systems (MESs) – strength and construction of passage and platform	
Para. 7.2.2.1	Broadcast of messages from other places on board	
Res.MSC.402(96)	Requirements for maintenance, thorough examination, operational testing, overhaul and repair of lifeboats and rescue boats, launching appliances and release gear	
Para.7.2	Availability to the information regarding authorized service providers	
Para.7.4.1	Issuance of an authorization document upon initial audit	
Para.7.4.2	Periodic audit and if not in compliance, withdrawal of the authorization of service providers	

_

SPECIFIC FLAG STATE OBLIGATIONS		
SOURCE	SUMMARY DESCRIPTION	COMMENTS
1994 HSC Code		
Para. 1.3.5	Verification	
Para. 1.4.29	Determination of "maximum operational weight"	
Para. 1.5.1.2	Specifying intervals for renewal surveys	
Para. 1.5.4	Inspection and survey	
Para. 1.5.5	Recognized organizations and nominated surveyors	
Para. 1.5.7	Completeness of survey and inspection	
Para. 1.8.1	Issue/endorsement of certificate	
Para. 1.9.2	Issue of permit to operate	
Para. 1.11.2	Equivalents – reporting	
Para. 1.12.1	Adequate information and guidance provided to the craft by the company	
Paras. 1.13.2 and 1.13.3	Novel designs	
Para. 1.14.1	Investigation reports to IMO	
Paras. 2.7.4 and 2.14.2	Inclining and stability information – approval	
Para. 3.4	Determination of service life	
Para. 3.5	Design criteria	
Para. 4.8.3	Documentation and verification of evacuation time	
Para. 7.5.6.3	Safe outlets for exhaust fans in fuel tank spaces	
Para. 7.7.2.3.2	Sensitivity limits of smoke detectors	

Г

	SPECIFIC FLAG STATE OBLIGATIONS	
SOURCE	SUMMARY DESCRIPTION	COMMENTS
Para. 7.7.6.1.5	Additional quantity of fire-extinguishing medium	
Para. 7.7.6.1.12	Containers for the storage of fire-extinguishing medium, etc. – design	
Para. 7.7.8.5	Maximum length of fire hoses	
Para. 8.1	Approval and acceptance of LSA and arrangements	
Para. 8.9.1.2	Approval of novel LSAs or arrangements	
Para. 8.9.1.3	Notifying the Organization	
Para. 8.9.7.1.2	Approval of servicing stations	
² ara. 8.9.7.2	Deployment intervals of MES	
Para. 8.9.11	Novel LSAs or arrangements	
Para. 8.9.12	Notifying the Organization	
Para. 10.2.4.9	Flexible oil fuel pipes	
Para. 10.3.7	Internal diameters of suction branches	
Para. 12.6.2	Specified voltages to earth	
Para. 13.1.2	Navigational equipment and its installation	
Para. 13.13	Approval of systems, equipment and performance standards	
Para. 14.3.3	Exemptions – reporting	
^D ara. 14.13.1	Type approval	
^o ara. 14.14.5	Ensuring maintenance	
² ara. 14.15	Radio personnel	
Para. 14.16	Radio records	
Para. 15.3.1	Operating station – field of vision	
	I	

-

SPECIFIC FLAG STATE OBLIGATIONS		
SOURCE	SUMMARY DESCRIPTION	COMMENTS
Para. 15.7.2	Ensuring clear view through windows	
Para. 17.8	Acceleration and deceleration	
Para. 18.1.4	Determining maximum allowable distance from a base port or place of refuge	
Para. 18.2	Craft documentation	
Paras. 18.3.1 to 18.3.7	Training and qualifications	
Chapter 19	Inspection and maintenance requirements	
2000 HSC Code		
Para. 1.3.7	Verification	
Para. 1.4.37	Determination of "maximum operational weight"	
Para. 1.5.1.2	Specifying intervals for renewal surveys	
Para. 1.5.4	Inspection and survey	
Para. 1.5.5	Recognized organizations and nominated surveyors	
Para. 1.5.7	Completeness of survey and inspection	
Para. 1.7.3	Investigation to determine the need of survey	
Para. 1.8.1	Issue/endorsement of certificate	
Para. 1.9.1.1.4	Transit voyage – arrangements satisfactory	
Para. 1.9.2	Issue of permit to operate	
Para. 1.9.7	Worst intended conditions and operational limitations	
Para. 1.11.2	Equivalents – reporting	
Para. 1.12.1	Adequate information and guidance provided to the craft by the company	

SOURCE	SUMMARY DESCRIPTION	COMMENTS
Paras. 1.13.2 and 1.13.3	Novel designs	
Para. 1.14.1	Investigation reports to IMO	
Para. 2.9.3	Verification of load line marks	
Paras. 2.7.5 and 2.14.2	Inclining and stability information – approval	
Para. 3.4	Determination of service life	
Para. 3.5	Design criteria	
Para. 4.2.2	Approval of public address system	
Para. 4.8.3	Documentation and verification of evacuation time	
Para. 4.8.10	Evacuation demonstration	
Para. 7.3.3	Approval of structural fire protection details	
Para. 7.5.6.3	Safe outlets for exhaust fans in fuel tank spaces	
Para. 7.7.1.1.8	Limitation of number of enclosed spaces in each section	
Para. 7.7.1.3.2	Sensitivity limits of smoke detectors	
Para. 7.7.3.3.6	Additional quantity of fire-extinguishing medium	
Para. 7.17.1	Reduced requirements for cargo craft of less than 500 GT	
Para. 7.17.3.1.5	Water spray system – approval	
Para. 7.17.3.3	Smoke detection systems – equivalent protection	
Para. 7.17.4	Issue of Document of Compliance for craft carrying dangerous goods	

SPECIFIC FLAG STATE OBLIGATIONS		
SOURCE	SUMMARY DESCRIPTION	COMMENTS
Para. 8.1	Approval and acceptance of LSA and arrangements	
Para. 8.9.7.1.2	Approval of servicing stations	
Para. 8.9.8	Rotational deployment of marine evacuation systems	
Para. 8.9.11	Extension of life raft service intervals – notification	
Para. 8.11	Helicopter pick-up areas – approval	
Para. 10.2.4.9	Flexible oil fuel pipes	
Para. 10.3.7	Internal diameters of suction branches	
Para. 12.6.2	Specified voltages to earth	
Para. 13.1.2	Shipborne navigational systems and equipment and voyage data recorders and their installation	
Para. 13.17	Type approval	
Para. 14.3.3	Exemptions – reporting	
Para. 14.4.2	GMDSS Identities – suitable arrangements	
Para. 14.14.1	Type approval	
Para. 14.15.5	Ensuring maintenance	
Para. 14.16	Radio personnel	
Para. 14.17	Radio records	
Para. 15.3.1	Operating station – field of vision	
Para. 15.7.2	Ensuring clear view through windows	
Para. 17.8	Acceleration and deceleration	
Para. 18.1.4	Determining maximum allowable distance from a base port or place of refuge	

SPECIFIC FLAG STATE OBLIGATIONS		
SOURCE	SUMMARY DESCRIPTION	COMMENTS
Para. 18.2	Craft documentation	
Paras. 18.3.1 to 18.3.7	Training and qualifications	
Chapter 19	Inspection and maintenance requirements	
2011 ESP Code		
Annex A, part A Single-side skin bulk carriers		
Para. 3.3.4	Supervision on repair of cargo hatch securing system	
Para. 5.1.1	Cooperation on development of a specific survey programme	
Para. 5.1.4	Advice on the maximum acceptable structural diminution levels	
Para. 5.2.2	Agreement on provisions for proper and safe access	
Para. 9.1.2	Evaluation of survey report	
Para. 9.2.3	Endorsement on condition evaluation report	
Annex 4B, Para. 1	Cooperation on development of a survey	
Annex 5, Para. 3.1	Certification of a firm engaged in thickness measurement	
Annex 13, Para. 3	Approval of materials and welding	
Annex A, part B Double-side skin bulk carriers		
Para. 3.3.4	Supervision on repair of cargo hatch securing system	
Para. 5.1.1	Cooperation on development of a specific survey programme	
Para. 5.1.4	Advice on the maximum acceptable structural diminution levels	

SPECIFIC FLAG STATE OBLIGATIONS		
SOURCE	SUMMARY DESCRIPTION	COMMENTS
Para. 5.2.2	Agreement on provisions for proper and safe access	
Para. 9.1.2	Evaluation of survey report	
Para. 9.2.3	Endorsement on condition evaluation report	
Annex 4B, Para. 1	Cooperation on development of a survey programme	
Annex 5, Para. 3.1	Certification of a firm engaged in thickness measurement	
Annex 11, Para. 3	Approval of materials and welding	
Annex B, part A Double hull oil tankers		
Para. 5.1.1	Cooperation on development of a specific survey programme	
Para. 5.1.4	Advice on the maximum acceptable structural diminution levels	
Para. 5.2.1.1	Agreement on provisions for proper and safe access	
Para. 9.1.3	Evaluation of survey report	
Para. 9.2.3	Endorsement on condition evaluation report	
Annex 7B	Cooperation on development of a survey programme	
Annex 8, Para. 3.1	Certification of a firm engaged in thickness measurement	
Annex B, part B Oil tankers other than double hull oil tankers		
Para. 5.1.1	Cooperation on development of a specific survey programme	

	SPECIFIC FLAG STATE OBLIGATIONS		
SOURCE	SUMMARY DESCRIPTION	COMMENTS	
Para. 5.1.4	Advice on the maximum acceptable structural diminution levels		
Para. 5.2.1.1	Agreement on provisions for proper and safe access		
Para. 8.1.3	Evaluation of survey report		
Para. 8.2.3	Endorsement on condition evaluation report		
Annex 6B	Cooperation on development of a survey programme		
Annex 7, Para. 3.1	Certification of a firm engaged in thickness measurement		
Res. 4 of the 1997 SOLAS Conference			
Section 5	Dimension and selection of weld connections and materials		
Res. MSC.168(79)	Standards and criteria for side structures of bulk carriers of single-side skin construction		
Para. 2.1	Applicable national standards		
Para. 4.4	Applicable national standards		
Para. 4.5	Applicable national standards		
NO _X Technical Code 2008			
Chapter 1	Assumption of full responsibility for the approval of documentation as required by the Code together with the acceptance of procedures and alternatives as permitted by the Code		
Chapter 2	Issue of the Engine International Air Pollution Prevention Certificate, arrangements for the parent engine test and pre-certification of engines, usage of the engine family/engine group concepts and approval of the technical file and any subsequent amendments		

SPECIFIC FLAG STATE OBLIGATIONS		
SOURCE	SUMMARY DESCRIPTION	COMMENTS
Chapter 2, Para. 2.2.5.1	Approval and pre-certification of the combined engine/NO _x -reducing device	
Chapter 3	Acceptance of modification of engine speed at E2 test cycle 25% power mode point	
Chapter 4	Assignment of engine family/engine group status, as applicable, and selection of associated parent engine; acceptance of conformity of production arrangements; adjustment of parent engine relative to engine group reference values	
Chapter 5	Ensuring that the parent engine test and subsequent calculations are undertaken in accordance with Code requirements and that, where alternatives are applied, these meet the Code's equivalency requirements and any deviations are within the permitted margins; filing of parent engine test report	
Chapter 6	Onboard NO_X verification procedures are in accordance with the provisions of the Code and are adequate to provide verification that an engine, as so surveyed, will be in accordance with the applicable Annex VI requirements; acceptance of aspects within onboard NO_X verification procedure – simplified measurement method if applicable; approval of aspects within onboard NO_X verification procedure – direct measurement and monitoring method including the onboard monitoring manual, if applicable	
Chapter 6, Para. 6.2.2	Approval of an electronic record book concerning the entries required under the Code	
Chapter 7	Installation of approved method – amendment of IAPP Certificate	

SPECIFIC FLAG STATE OBLIGATIONS		
SOURCE	SUMMARY DESCRIPTION	COMMENTS
Appendix IV	Verification that the calibration of all necessary measurement equipment meets Code requirements	
Appendix VII	Aspects to be included within onboard NO_X verification procedure – parameter check method	
Appendix VIII	Approval of alternative exhaust gas measurement principles	
IBC CODE		
Para. 1.1.6	Prescribe preliminary suitable conditions for carriage of products not listed in chapters 17 or 18	
Para. 1.4.2	Equivalents – communication to IMO	
Section 1.5	Survey and certification	
Para. 1.5.4	Issue or endorsement of certificate	
Para. 2.2.2	Intact stability in all seagoing conditions	
Para. 2.2.3	Free surface effect in undamaged compartments	
Para. 2.2.6.3	Issuing a document of approval for the stability instrument	
Section 2.4	Conditions of loading	
Para. 2.8.1.6	Standard of damage	
Para. 2.8.2	Standard of damage – alternative measures	
Para. 2.9.2.3	Residual stability during intermediate stages of flooding	
Para. 3.4.4	Access to spaces in the cargo area	
Para. 3.7.3.5	Alternative arrangements for draining the piping	
Para. 3.7.4	Relaxation for small ships	

SPECIFIC FLAG STATE OBLIGATIONS			
SOURCE	SUMMARY DESCRIPTION	COMMENTS	
Para. 5.1.6.4	Dimensions for flanges not complying with the standards		
Para. 5.2.2	Piping fabrication and joining details		
Para. 7.1.1	Cargo temperature control – general		
Para. 8.3.6	Devices to prevent the passage of flames into cargo tanks – requirements for the design, testing and locating		
Para. 10.1.3	Electrical installations – appropriate steps for uniform implementation		
Para. 10.1.4	Electric equipment, cables and wiring that do not conform to the standard		
Para. 10.1.5	Electrical equipment in hazardous locations		
Para. 11.2.2	Approval of an appropriate fire-extinguishing system		
Para. 11.3.2	Cargo area – additional arrangements		
Para. 11.3.5.3	Cargo area – minimum capacity of monitor		
Para. 11.3.7	Minimum capacity of foam monitor for ships less than 4,000 tonnes deadweight		
Para. 11.3.13	Alternative provisions to deck foam system		
Para. 13.2.3	Exemption of toxic-vapour-detection equipment		
Para. 14.1.2	Protective equipment		
Chapter 15	Approval of special requirements for specific chemicals		
Para. 16.2.2	Cargo information – independent expert		
Para. 16.5.1	Stowage of cargo samples – approval		

SPECIFIC FLAG STATE OBLIGATIONS			
SOURCE	SUMMARY DESCRIPTION	COMMENTS	
Para. 18.2	Safety requirements – list of products to which the Code does not apply		
BCH CODE			
Para. 1.5.2	Equivalents – communication to IMO		
Section 1.6	Survey requirements		
Para. 1.6.4	Issue or endorsement of certificate		
Section 1.8	New products – establishing suitable conditions – notifying IMO		
Para. 2.2.1.2.2	Issuing a document of approval for the stability instrument		
Para. 2.2.4	Determination of the ability to survive flooding of the machinery space in Type 3 below 125 m in length		
Para. 2.2.5	Nature of alternative measures prescribed for small ships – duly noted on certificate		
Para. 2.9.5	Access to void spaces, cargo tanks, etc. – approval of smaller dimensions in special circumstances		
Section 2.10	Cargo piping systems – setting standards		
Section 2.12	Cargo hoses – setting standards		
Para. 2.14.2	High-velocity vent valves – type approval		
Para. 2.15.1	Cargo heating and cooling systems		
Para. 3.1.2(f)	Ventilation fans – approval		
Para. 3.14.1	Alternative provisions for ships dedicated to the carriage of specific cargoes		
Para. 3.14.2	Additional arrangements when foam is not effective or is incompatible		
Para. 3.14.7	Foam monitors on ships of less than 4,000 tonnes deadweight – minimum capacity		

SOURCESUMMARY DESCRIPTIONCOMMENTSPara. 3.15.2Protection of cargo pump-rooms with fire-extinguishing systems – approvalPara. 3.15.5Products evolving flammable vapours – fire-extinguishing systems – approvalChapter IVApproval of special requirements for specific chemicalsIGC CODEPara. 1.1.2.3Compliance with relevant requirements for existing shipsPara. 1.1.6.1Establishment of a Tripartite Agreement, where applicablePara. 1.1.6.3Assessment form and a proposal for a new and complete entry – submission to IMO	SPECIFIC FLAG STATE OBLIGATIONS		
Para. 3.15.2Protection of cargo pump-rooms with fire-extinguishing systems – approvalPara. 3.15.5Products evolving flammable vapours – fire-extinguishing systems – approvalChapter IVApproval of special requirements for specific chemicalsIGC CODEPara. 1.1.2.3Para. 1.1.6.1Establishment of a Tripartite Agreement, where applicablePara. 1.1.6.3Assessment form and a proposal for a new and complete entry – submission to IMO	COMMENTS	SUMMARY DESCRIPTION	SOURCE
Para. 3.15.2Protection of cargo pump-rooms with fire-extinguishing systems – approvalPara. 3.15.5Products evolving flammable vapours – fire-extinguishing systems – approvalChapter IVApproval of special requirements for specific chemicalsIGC CODEImage: Compliance with relevant requirements for existing shipsPara. 1.1.2.3Compliance with relevant requirements for existing shipsPara. 1.1.6.1Establishment of a Tripartite Agreement, where applicablePara. 1.1.6.3Assessment form and a proposal for a new and complete entry – submission to IMO			
Para. 3.15.5Products evolving flammable vapours – fire-extinguishing systems – approvalChapter IVApproval of special requirements for specific chemicalsIGC CODEImage: Compliance with relevant requirements for existing shipsPara. 1.1.2.3Compliance with relevant requirements for existing shipsPara. 1.1.6.1Establishment of a Tripartite Agreement, where applicablePara. 1.1.6.3Assessment form and a proposal for a new and complete entry – submission to IMO	ooms with pproval	Protection of cargo pump-rooms with fire-extinguishing systems – approval	Para. 3.15.2
Chapter IVApproval of special requirements for specific chemicalsIGC CODEPara. 1.1.2.3Compliance with relevant requirements for existing shipsPara. 1.1.6.1Establishment of a Tripartite Agreement, where applicablePara. 1.1.6.3Assessment form and a proposal for a 	vapours – pproval	Products evolving flammable vapours – fire-extinguishing systems – approval	Para. 3.15.5
IGC CODEPara. 1.1.2.3Compliance with relevant requirements for existing shipsPara. 1.1.6.1Establishment of a Tripartite Agreement, where applicablePara. 1.1.6.3Assessment form and a proposal for a new and complete entry – submission to 	ments for	Approval of special requirements for specific chemicals	Chapter IV
Para. 1.1.2.3Compliance with relevant requirements for existing shipsPara. 1.1.6.1Establishment of a Tripartite Agreement, where applicablePara. 1.1.6.3Assessment form and a proposal for a new and complete entry – submission to IMO			IGC CODE
Para. 1.1.6.1Establishment of a Tripartite Agreement, where applicablePara. 1.1.6.3Assessment form and a proposal for a new and complete entry – submission to IMO	quirements	Compliance with relevant requirements for existing ships	Para. 1.1.2.3
Para. 1.1.6.3 Assessment form and a proposal for a new and complete entry – submission to IMO	Agreement,	Establishment of a Tripartite Agreement, where applicable	Para. 1.1.6.1
	posal for a omission to	Assessment form and a proposal for a new and complete entry – submission to IMO	Para. 1.1.6.3
Para. 1.1.10 Implementation of the Code for ships operating in operation modes as referred to in the paragraph	 for ships as referred 	Implementation of the Code for ships operating in operation modes as referred to in the paragraph	Para. 1.1.10
Para. 1.3.2 Equivalents – communication to IMO	to IMO	Equivalents – communication to IMO	Para. 1.3.2
Paras. 1.4.1.1, 1.4.1.5, Survey and certification 1.4.6		Survey and certification	Paras. 1.4.1.1, 1.4.1.5, 1.4.6
Para. 1.4.1.3 Information of nominated surveyors or ROs – communication to IMO	irveyors or	Information of nominated surveyors or ROs – communication to IMO	Para. 1.4.1.3
Para. 1.4.4 Issue or endorsement of certificate	licate	Issue or endorsement of certificate	Para. 1.4.4
Para. 1.4.6.9.3 Flag changes – providing information regarding survey and certificate	information te	Flag changes – providing information regarding survey and certificate	Para. 1.4.6.9.3
Para. 2.2.6.3 Issuing a document of approval for the stability instrument	oval for the	Issuing a document of approval for the stability instrument	Para. 2.2.6.3
Para. 2.2.8 Damage survival capability investigation	vestigation	Damage survival capability investigation	Para. 2.2.8
Para. 2.6.2 Alternative measures – approval	val	Alternative measures – approval	Para. 2.6.2

SPECIFIC FLAG STATE OBLIGATIONS		
SOURCE	SUMMARY DESCRIPTION	COMMENTS
Para. 4.3.6	Inspection/survey plan for the cargo containment system – approval	
Para. 4.6.2.5	Inspection/test means for secondary barrier – approval	
Para. 4.14.1.3	Methods to predict accelerations due to ship motion – approval	
Para. 4.18.1.5	Allowable stress for materials other than those covered by chapter 6 – approval	
Para. 4.18.2.6.3	Load distribution and sequence for longer periods – approval	
Para. 4.19.1.6.3	Design and construction of the heating system – approval	
Para. 4.20.3.3	Determining requirements of inspection for secondary barriers	
Para. 6.4.1.1	Requirements for metallic materials as referred to in tables 6.1 to 6.5 of IGC Code	
Para. 6.6.3	Requirements for testing and inspection of secondary barriers	
Para. 8.2.6	Setting and sealing of pressure relief valves (PRVs)	
Para. 8.2.18	Adequacy of the vent system fitted on tanks	
Para. 10.2.4	Recognition of accredited testing authority or notified body	
Para. 11.4.1	Fixed dry chemical powder fire- extinguishing systems – approval	
Para. 13.6.16	Vapour concentration value at which alarm in primary space is set – approval	
Para. 15.6.1	Information on maximum allowable loading limits – approval	
Para. 17.14.7	Maximum set pressure of PRVs (Ethylene oxide) – approval	
Para. 17.18.24	Cargo handling plans – approval	

SPECIFIC FLAG STATE OBLIGATIONS		
SOURCE	SUMMARY DESCRIPTION	COMMENTS
Para. 18.2.1	Cargo system operation manuals – approval	
STCW Code, part A		
Section A-I/10.2	Withdrawal of endorsement of recognition – communication to the Party that issued the certificate	
Section A-II/4.4	Determining the requirements of training, assessment and certification where there are no tables of competence for the support level	
Section A-III/4.4	Determining the requirements of training, assessment and certification where there are no tables of competence for the support level	
Section A-VIII/1.1	Fitness for duty – consideration of the danger posed by fatigue of seafarers	
Section A-VIII/1.5	Requirement for watch schedules to be posted in a standardized form	
Section A-VIII/1.7	Requirement for maintaining records of daily hours of rest of seafarers	
Section A-VIII/1.10	Establishment of a limit of alcohol concentration for personnel performing designated duties	
Section A-VIII/2.84	Principles to be observed in keeping radio watch – directing the attention of companies, masters and radio watchkeeping personnel to comply with provisions of part 4-3 to ensure that an adequate safety radio watch is maintained while the ship is at sea	

SPECIFIC FLAG STATE OBLIGATIONS			
SOURCE	SUMMARY DESCRIPTION	COMMENTS	
Res. MEPC.94(46), as amended	Condition assessment scheme		
Para. 4.1	Issue of instructions to the recognized organization for Condition Assessment Scheme (CAS) survey		
Para. 4.3	Requirement for oil tankers to remain out of service until Statement of Compliance is issued		
Para. 7.1.3	CAS surveyors' requirements		
Para. 11	Verification of CAS		
Para. 12	Reassessment of ships that have failed		
Para. 13	Issue, suspension or withdrawal of Statement of Compliance		
Para. 14	Communication to IMO		
Res. MSC.215(82), as amended	Performance standard for protective coatings for dedicated seawater ballast tanks in all types of ships and double-side skin spaces of bulk carriers		
Para. 3.2	Inspection of surface preparation and coating processes		
Para. 3.4.1	Coating technical file		
Para. 4.4.3	Technical Data Sheet and Statement of Compliance or Type Approval Certificate – verification		
Section 5	Coating system approval		
Para. 6.1.1	Verification of equivalent qualification of coating inspector		
Section 7	Verification requirements		

SPECIFIC FLAG STATE OBLIGATIONS			
SOURCE	SUMMARY DESCRIPTION	COMMENTS	
Res. MSC.288(87), as amended	Performance standard for protective coatings for cargo oil tanks of crude oil tankers		
Para. 3.2	Inspection of surface preparation and coating processes – review		
Para. 4.6.3	Verification of the Technical Data Sheet and Statement of Compliance or Type Approval Certificate for the protective coating system		
Para. 6.1.1	Equivalent to NACE Coating Inspector Level 2 and FROSIO Inspector Level III – verification		
Para. 7	Coating verification requirement as referred to in paragraph 7		
Res. MSC.289(87)	Performance standard for alternative means of corrosion protection for cargo oil tanks of crude oil tankers		
Para. 2.2	Verification of Technical File		
Para. 4.2	Issuance of Type Approval Certificate for corrosion-resistant steel		
Para. 5	Survey(s) during the construction process to verify that approved corrosion-resistant steel has been applied to the area required		
Casualty Investigation Code			
Para. 1.3	Qualified person(s) for investigation		
Para. 6.2	Investigation into a very serious marine casualty		
IS Code, 2008	International Code on Intact Stability, 2008		
Part A, Section 1.2	Criteria demonstrating sufficient ship stability in critical stability situation in waves		
Part A, Para. 2.1.3	Stability criteria where anti-rolling devices are installed		
Part A, Section 2.3	Severe wind and stability criterion		

_

SPECIFIC FLAG STATE OBLIGATIONS		
SOURCE	SUMMARY DESCRIPTION	COMMENTS
Part A, Chapter 3	Special criteria for certain types of ships	
IMSBC Code		
Section 1.3	Conditions for the carriage of cargoes not listed in the Code	
Section 1.5	Exemptions	
Para. 7.3.2	Approval of specially constructed or fitted cargo ships for confining cargo shift	
Para. 7.3.2.1	Approval of specially constructed cargo ships with permanent structural boundaries for confining cargo shift	
Para. 7.3.2.2	Approval of plan of special arrangements and details of the stability conditions on which the design has been based	
Para. 7.3.3.2	Approval of specially constructed cargo ships for dry powdery cargoes	
Appendix 1, Schedule for Aluminium Ferrosilicon Powder, UN 1395, and Aluminium Silicon Powder, Uncoated,	Inspection and approval of gastight bulkheads between cargo spaces and engine-room	
Appendix 1, Schedule for Ferrosilicon, UN 1408, and Ferrosilicon	Inspection and approval of gastight bulkheads between cargo spaces and engine-room and approval of safety of bilge pumping arrangement	
Polar Code		
Part I-A, Paras. 1.3.4 and 1.3.7	Issue of a Polar Ship Certificate to a ship as applicable	
Part I-A, Para. 3.3.1	Approval of materials of exposed structures in ships	
Part I-A, Para. 3.3.2	Approval of scantlings of category A, B and ice-strengthened category C ships	

SPECIFIC FLAG STATE OBLIGATIONS			
SOURCE	SUMMARY DESCRIPTION	COMMENTS	
Part I-A, Para. 6.3.2.3	Approval of materials of exposed machinery and foundations for ships intended to operate in low air temperature		
Part I-A, Para. 6.3.3	Approval of propeller blades, propulsion line, steering equipment and other appendages of category A, B and ice-strengthened category C ships		
Part I-A, Para. 7.3.3	Approval of materials of exposed fire safety systems for ships intended to operate in low air temperature		
Part II-A, Para. 1.1.3	Approval of operational requirements for a category A ship in relation to discharge of oil or oily mixtures from machinery spaces, as applicable		
Part II-A, Para. 2.1.3	Approval of operational requirements for category A and B ships carrying NLS, as applicable		
Part II-A, para 4.2.3	Approval of operational requirements for category A and B ships in relation to discharge of sewage, as applicable		
IGF Code			
Para. 2.3.3	Approval of alternative design		
Para. 6.4.1.8	Approval of an inspection/survey plan for the liquefied gas fuel containment system		
Para. 6.4.4.5	Approval of the methods for periodically checking effectiveness of secondary barriers		
Para. 6.4.9.4.1.1	Approval of methods to predict accelerations due to ship motion		
Para. 6.4.12.1.1.5	Approval of allowable stresses for materials other than those covered by paragraph 7.4 of the Code		

SPECIFIC FLAG STATE OBLIGATIONS			
SOURCE	SUMMARY DESCRIPTION	COMMENTS	
Para. 6.4.12.2.5	Special consideration in the case of use of simplified dynamic loading spectra for estimation of the fatigue life		
Para. 6.4.12.2.6	Approval of load distribution and sequence for longer periods		
Para. 6.4.13.1.1.4.3	Approval of the containment system including design and construction of the heating system		
Para. 6.6.1	Certification and approval for the storage tanks to be used for compressed natural gas		
Para. 7.3.4.3	Consideration of the allowable stress for pipes made of materials other than steel		

SPECIFIC COASTAL STATE OBLIGATIONS

The following table contains a non-exhaustive list of obligations, including those obligations imposed when a right is exercised.

SPECIFIC COASTAL STATE OBLIGATIONS			
SOURCE	SUMMARY DESCRIPTION	COMMENTS	
SOLAS 1974			
Reg. IV/5	Provision of radio communication services and communication of information on such provision		
Reg. V/4	Navigation warnings		
Reg. V/7.1	Search and rescue services – necessary arrangements		
Reg. V/7.2	Search and rescue services – information to IMO		
Reg. V/8	Life-saving signals		
Reg. V/9	Hydrographic services		
Regs. VII/6.1 and 7-4.1	Reporting of incidents involving dangerous goods		
MARPOL			
Annex I			
Reg. 4.3	Exceptions – discharge of substances containing oil for the purpose of combating pollution incidents		
Annex II			
Reg. 3.1.3	Exceptions – approval of discharge of NLS for the purpose of combating pollution incidents		
Reg. 13.2.3	Control of discharges of residues of NLS – agreement and communication to IMO		

SPECIFIC PORT STATE OBLIGATIONS

The following table contains a non-exhaustive list of obligations, including those obligations imposed when a right is exercised.

SPECIFIC PORT STATE OBLIGATIONS		
SOURCE	SUMMARY DESCRIPTION	COMMENTS
TONNAGE 1969		
Article 12	Inspection	
LL 1966 AND LL PROT 1988		
Article 21	Control	amended by LL PROT 1988
STCW 1978		
Article X	Control	
Reg. I/4	Control procedures	
SOLAS 1974		
Reg. I/6(c)	Ships not allowed to sail	
Reg. I/19	Control	
Reg. VI/7.3	Agreed plan for loading, unloading and stowage of solid bulk cargoes - lodged with the authority of the port State	
Reg. VII/4.2	Special list, manifest or stowage plan copy to be made available before departure	
Reg. VII/7-2.2	Documents relating to carriage of dangerous goods in solid form	
Reg. VIII/11	Special control for nuclear ships	
Reg. XI-1/4	Port State control on operational requirements	
Reg. XIV/3.2	Port State control of ships to which the chapter applies	

SPECIFIC PORT STATE OBLIGATIONS		
SOURCE	SUMMARY DESCRIPTION	COMMENTS
MARPOL		
Article 5(2)	Certificates and special rules on inspection of ships – port State control	
Article 5(3)	Certificates and special rules on inspection of ships – denial of entry	
Article 6(2)	Detection of violations and enforcement of the Convention – inspection	
Article 6(5)	Detection of violations and enforcement of the Convention – inspection upon request – reporting	
Article 11 (1) (d)	Communication of information – a list of reception facilities including their location, capacity and available facilities and other characteristics	
Annex I		
Reg. 2.6.2	Application – an oil tanker delivered on or before 1 June 1982 engaged in specific trades: agreement with flag States	
Reg. 2.6.3	Application – an oil tanker delivered on or before 1 June 1982, engaged in specific trades: approval by port States	
Reg. 11	Port State control on operational requirements	
Reg. 17.7	Oil Record Book, Part I – inspection without undue delay	
Reg. 18.10.1.2	Segregated ballast tanks – oil tanker delivered on or before 1 June 1982 having special ballast arrangements: agreement with flag States	
Reg. 20.8.2	Denial of entry – communication to IMO	
Reg. 21.8.2	Denial of entry – communication to IMO	
Reg. 36.8	Oil Record Book, Part II – inspection without undue delay	
Regs. 38.1, 38.2, 38.3 and 38.4	Reception facilities outside special areas	
Regs. 38.5, 38.6 and 38.7	Reception facilities within special areas	
Reg. 38.9.1	Reception facilities within special areas: "Antarctic area"	

SPECIFIC PORT STATE OBLIGATIONS		
SOURCE	SUMMARY DESCRIPTION	COMMENTS
Annex II		
Reg. 4.3.3	Exemptions – approval of adequacy of reception facilities	
Reg. 13.6.1	Control of discharges of residues – endorsement of cargo record book	
Reg. 15.6	Cargo record book – inspection without undue delay	
Reg. 16.1 Regs. 16.6 and	Measures of control Measures of control – exemption granted (endorsement	
16.7	of cargo record book)	
Reg. 16.9	Port State control on operational requirement	
Regs. 18.1, 18.2, 18.4 and 18.6	Reception facilities and cargo unloading terminal arrangements	
Annex III		
Reg. 9	Port State control on operational requirements	
Annex IV		
Reg. 12.1	Provision of reception facilities	
Reg. 13.1	Provision of reception facilities for passenger ships in special areas	
Reg. 13.2	Measures taken regarding reception facilities for passenger ships in special areas – notifying the Organization	
Reg. 14	Port State control on operational requirements	
Annex V		
Reg. 6.3.1	Provision of reception facilities – all garbage from all ships departing enroute to or arriving from the Antarctic area	

SPECIFIC PORT STATE OBLIGATIONS		
SOURCE	SUMMARY DESCRIPTION	COMMENTS
Reg. 8.1	Reception facilities	
Reg. 8.2	Reception facilities within special areas	
Reg. 8.4	Measures taken regarding provision of reception facilities – notifying the Organization	
Reg. 9	Port State control on operational requirements	
Reg. 10.5	Inspection of Garbage Record Book or ship's official log-book	
Annex VI		
Reg. 5.3.3	Necessary assistance to the surveyor as referred to in the paragraph	
Reg. 10	Port State control on operational requirements – in relation to chapter 4 of Annex VI, limiting of inspection to verification that there are valid Statement of Compliance related to fuel oil consumption and International Energy Efficiency Certificate on board	
Regs. 15.2 and 15.3	Volatile organic compounds – approvals of vapour emission control systems and notifying IMO	
Reg. 17.1	Reception facilities as referred to in the paragraph	
Reg. 17.3 and 17.4	Ports and terminals where reception facilities are/are not available to manage and process the substances as referred to in regulation 17.1 – communication to IMO	
Reg. 18.1	Availability of fuel oils and communication to IMO	
Reg. 18.2.1	Ship not compliant with fuel oil standards	
Reg. 18.2.3	Action taken, including not taking control measures	
Reg. 18.2.5	Evidence of the non-availability of compliant fuel oil – communication to IMO	
Reg. 18.9	Authorities designated for register of local suppliers, bunker delivery note and sample, fuel oil quality, actions against fuel oil suppliers for non-compliance, informing the Administration of any ship receiving non-compliant fuel oil and communication to IMO of non-compliant fuel oil suppliers as referred to in the paragraph	

SPECIFIC PORT STATE OBLIGATIONS		
SOURCE	SUMMARY DESCRIPTION	COMMENTS
Reg. 18.10	Fuel oil quality – communication to Party or non-Parties and remedial action	
IBC Code		
Para. 15.8.25.3	Certification verifying that the required piping separation has been achieved	
1994 HSC Code		
Para. 1.3.5	Acceptance of the Code	
Para. 1.5.6	Providing assistance to surveyors	
Para. 1.6	Design approval	
Para. 1.9.3	Operational conditions – Permit to Operate	
Para. 1.9.4	Port State control	
Para. 18.3.8	Training and qualifications	
2000 HSC Code		
Para. 1.3.7	Acceptance of the Code	
Para. 1.5.6	Providing assistance to surveyors	
Section 1.6	Design approval	
Para. 1.9.3	Operational conditions – Permit to Operate	
Para. 1.9.4	Port State control	
Para. 18.3.8	Training and qualifications	
Grain Code		
Para. 3.4	Document of authorization	
Para. 3.5	Document of authorization	
Section 5	Exemptions for certain voyages	
Para. 7.2	Stability requirements	

SPECIFIC PORT STATE OBLIGATIONS		
SOURCE	SUMMARY DESCRIPTION	COMMENTS
IMSBC Code		
Section 1.3	Conditions for the carriage of cargoes not listed in the Code	
Section 1.5	Exemptions	
Para. 4.3.2	Concentrate of cargo which may liquefy – recognition of entity issuing the certificate or declaration of moisture content	
Para. 4.3.3	Procedures for sampling, testing and controlling – role of the competent authority as referred to in the paragraph	
IGC Code		
Para. 1.1.6.1	Establishment of a Tripartite Agreement, where applicable	
Para. 1.1.10	Implementation of the Code for ships operating in operation modes as referred to in the paragraph	
Para. 1.4.1.4	Ships not allowed to sail	

INSTRUMENTS MADE MANDATORY UNDER IMO CONVENTIONS

SOLAS 1974	Res. MSC.215(82), as amended Res. MSC.133(76), as amended Res. MSC.287(87) Res. MSC.288(87), as amended Res. MSC.289(87) Noise Code 2008 IS Code (introduction and part A) FSS Code 2010 FTP Code LSA Code Res. MSC.402(96) IMSBC Code CSS Code, sub-chapter 1.9 Grain Code IMDG Code IBC Code IBC Code ISM Code 1994 HSC Code 2000 HSC Code 2000 HSC Code RO Code 2011 ESP Code	Reg. II-1/3-2.2 Reg. II-1/3-6.2.1 Reg. II-1/3-10.3 Reg. II-1/3-11.3.1 Reg. II-1/3-11.3.2 Reg. II-1/3-12 Reg. II-1/3-12 Reg. II-2/3.22 Reg. II-2/3.23 Reg. III/3.10 Reg. III/3.25 Reg. VI/1-2 Reg. VI/1-2 Reg. VI/2.1 Reg. VI/2.1 Reg. VI/2.1 Reg. VI/8.1 Reg. VI/1.1 Reg. VII/1.1 Reg. VII/1.1 Reg. VII/1.1 Reg. VII/1.1 Reg. VII/1.1 Reg. X/1.1 Reg. X/1.2 Reg. XI-1/2 Reg. XI-1/2 Reg. XI-1/2
	Res. 4 of the 1997 SOLAS Conf. Res. MSC.169(79) Res. MSC.168(79)	Reg. XII/1.7 Reg. XII/7.2 Reg. XII/14
	III Code	Reg. XIII/2
	Polar Code (introduction and part I-A) IGF Code	Reg. XIV/3.1 Reg. II-1/57
MARPOL	RO Code	Annex I, Reg. 6 Annex II, Reg. 8
	Res. MEPC.94(46), as amended IBC Code BCH Code NO _X Technical Code 2008 III Code	Annex I, Reg. 20.6 Annex II, Reg. 1.4 Annex II, Reg. 1.4 Annex VI, Reg. 5.3.2 Annex I, Reg. 44 Annex II, Reg. 19 Annex III, Reg. 10 Annex IV, Reg. 15 Annex V, Reg. 11 Annex VI, Reg. 24
	Polar Code (introduction and part II-A)	Annex I, Reg. 47.2 Annex II, Reg. 22.2 Annex IV, Reg. 18.2 Annex V, Reg. 14.2

STCW 1978	STCW Code (part A)	Reg. I/1.2.3
LL PROT 1988	III Code 2008 IS Code RO Code	Reg. I/16.1 Annex 1, Reg. 1 Annex I, Reg. 2-1
TONNAGE 1969 COLREG 1972	III Code III Code III Code	Annex IV, Reg. 54(1) Annex III, Reg. 8 Part F, Reg. 40

SUMMARY OF AMENDMENTS TO MANDATORY INSTRUMENTS REFLECTED IN THE NON-EXHAUSTIVE LIST OF OBLIGATIONS (ANNEXES 1 TO 4 AND ANNEX 7)

The amendments to mandatory instruments reflected in annexes 1 to 4 and annex 7 are summarized below to facilitate the amendment of corresponding tables in the future.

SOLAS 1974	up to and including 2018 amendments (res. MSC.436(99) except chapter XI-2, regulation V/19-1 and ISPS Code)
Res. MSC.215(82)	up to and including the 2012 amendments (res. MSC.341(91))
Res. MSC.133(76)	up to and including the 2004 amendments (res. MSC.158(78))
Res. MSC.287(87)	as adopted
Res. MSC.288(87)	up to and including the 2012 amendments (res. MSC.342(91))
Res. MSC.289(87)	as adopted
Res. MSC.402(96)	as adopted
Noise Code	as adopted by res. MSC.337(91)
2008 IS Code	up to and including the 2018 amendments (res. MSC.443(99)/MSC.444(99))
FSS Code	up to and including the 2016 amendments (res. MSC.403(96))
2010 FTP Code	up to and including the 2018 amendments (res. MSC.437(99))
LSA Code	up to and including the 2017 amendments (res. MSC.425(98))
IMSBC Code	up to and including the 2019 amendments (res. MSC.462(101))
CSS Code, sub-chapter 1.9	up to and including the 2002 amendments (MSC/Circ.1026)
Grain Code	up to and including the 1991 amendments (res. MSC.23(59))
IMDG Code	up to and including the 2020 amendments (res. MSC.477(102))
IBC Code	up to and including the 2019 amendments (res. MSC.460(101) and res. MEPC.318(74))
IGC Code	up to and including the 2018 amendments (res. MSC.441(99))
INF Code	up to and including the 2007 amendments (res. MSC.241(83))
ISM Code	up to and including the 2013 amendments (res. MSC.353(92))
1994 HSC Code	up to and including the 2018 amendments (res. MSC.438(99))
2000 HSC Code	up to and including the 2018 amendments (res. MSC.439(99))
RO Code	as adopted by res. MSC.349(92) and MEPC.237(65)
2011 ESP Code	up to and including the 2019 amendments (res. MSC.461(101))

Casualty Investigation Code	up to and including 2014 amendments (res_MSC 390(94))
Res. 4 of the 1997 SOLAS Conf.	as adopted
Res. MSC.169(79) Res. MSC.168(79) III Code Polar Code	as adopted as adopted as adopted by res. A.1070(28) as adopted by res. MSC.385(94) and res.
	MEPC.264(68)
SOLAS PROT 1978	up to and including 2017 amendments (res.422(98)) (res. MSC.394(95))
SOLAS PROT 1988	up to and including the 2015 amendments (res. MSC.395(95))
MARPOL	up to and including the 2020 amendments (res. MEPC.324(75))
RO Code Res. MEPC.94(46), as amended	as adopted by res. MEPC.237(65) and MSC.349(92) up to and including the 2013 amendments (res. MEPC 236(65))
IBC Code	up to and including the 2019 amendments (res_MSC 460(101) and MEPC 318(74))
BCH Code	up to and including the 2019 amendments (res. MEPC, $319(74)$ and MSC $463(101)$)
NO _X Technical Code 2008	up to and including the 2019 amendments res. MEPC.317(74))
	as adopted by res. A.1070(28)
Polal Code	MSC.385(94)
STCW 1978	up to and including the 2016 amendments (res. MSC.416 (97)), except Regs. VI/5.2, 6.3 and 6.6
STCW Code, part A	up to and including the 2016 amendments (res. MSC.417(97))
III Code LL 1966	as adopted by res. A.1070(28) up to and including the 2013 amendments (res. A 1083(28))
LL PROT 1988	up to and including the 2014 amendments (res. MSC.375(93))
RO Code III Code TONNAGE 1969	as adopted by res. MSC.349(92) as adopted by res. A.1070(28) up to and including the 2013 amendments (res.
III Code COLREG 1972	as adopted by res. A.1070(28) up to and including the 2013 amendments (res A 1085(28))
III Code	as adopted by res. A.1070(28)

AMENDMENTS⁴ TO IMO INSTRUMENTS EXPECTED TO BE ACCEPTED AND TO ENTER INTO FORCE BETWEEN 1 JANUARY 2022 AND 1 JULY 2022

The following tables contain non-exhaustive lists of obligations, including those obligations imposed when a right is exercised.

Annex 1

OBLIGATIONS OF CONTRACTING GOVERNMENTS/PARTIES

OBLIGATIONS OF CONTRACTING GOVERNMENTS/PARTIES		
SOURCE	SUMMARY DESCRIPTION	COMMENTS
IMDG Code		
Chapter 3.3, SP356	Approval of metal hydride storage systems installed in vehicles, vessels, <u>machinery</u> , <u>engines</u> or aircraft or in completed components or intended to be installed in vehicles, vessels, <u>machinery</u> , <u>engines</u> or aircraft	<u>In force 1/6/2022 by</u> <u>MSC.477(102)</u>
Chapter 5.5 <u>.2.5.2</u>	Determining the period between fumigant application and loading of fumigated cargo transport unit on board the ship Determination of requirements for fumigation of cargo transport units to ensure a sufficient period has elapsed to attain a reasonable uniform gas concentration throughout the cargo in it	<u>In force 1/6/2022 by</u> <u>MSC.477(102)</u>
Chapter 6.2	Approval of pressure receptacles, aerosol dispensers, small receptacles containing gas (gas cartridges) and fuel cell cartridges containing liquefied flammable gas - role of competent authority	In force 1/6/2022 by MSC.477(102)
Para.6.2.2.1.2	Approval of tube service life test programme and the results – role of competent authority	In force 1/6/2022 by MSC.477(102)

⁴ The struck-out text indicates deletions and the underlined text shows additions or changes to the non-exhaustive list of obligations.
Annex 2

SPECIFIC FLAG STATE OBLIGATIONS

SPECIFIC FLAG STATE OBLIGATIONS									
SOURCE	SUMMARY DESCRIPTION	COMMENTS							
IMDG Code									
Chapter 3.3., SP975	Approval of long international voyages	In force 1/6/2022 by MSC.477(102)							
Chapter 3.3, SP976	Prohibition of the transport of this substance except with the approval	In force 1/6/2022 by MSC.477(102)							
MARPOL									
<u>Annex VI</u>									
Reg. 20.3	Reporting of the required and attained EEDI value and relevant information to the Organization	In force 1/4/2022 by MEPC.324(75)							

Annex 4

SPECIFIC PORT STATE OBLIGATIONS

SPECIFIC PORT STATE OBLIGATIONS								
SOURCE	SUMMARY DESCRIPTION	COMMENTS						
IMDG Code								
<u>Chapter 3.3, SP975</u>	Approval of long international voyages	<u>In force 1/6/2022 by</u> <u>MSC.477(102)</u>						
<u>Chapter 3.3, SP976</u>	Prohibition of the transport of this substance except with the approval	In force 1/6/2022 by MSC.477(102)						

DRAFT MSC-MEPC.5 CIRCULAR

MODEL AGREEMENT FOR THE AUTHORIZATION OF RECOGNIZED ORGANIZATIONS ACTING ON BEHALF OF THE ADMINISTRATION

1 The Marine Environment Protection Committee, at its [...] and the Maritime Safety Committee, at its 105th session (4 to 8 October 2021), approved the *Model agreement for the authorization of recognized organizations acting on behalf of the Administration* prepared by the Sub-Committee on Implementation of IMO Instruments at its sixth session (1 to 5 July 2019), as set out in the annex, to assist Administrations when formalizing, in writing, a delegation of authority agreement for the purpose of having statutory certification and services rendered by a recognized organization (RO) on their behalf.

2 The *IMO Instruments Implementation Code* (III Code) (resolution A.1070(28)), in paragraph 18.2, and the *Code for Recognized Organizations* (RO Code) (resolutions MSC.349(92) and MEPC.237(65)), in paragraph 8.2.1 of part 2, prescribe that the Administration should establish the legal basis on which the authorization of statutory certification and services is administered and should regulate such authorization in accordance with the applicable requirements of the international mandatory instruments. The authorization should have, as its basis, a formal written agreement between the Administration and the RO which, as a minimum, includes the elements set out in appendix 3 of the RO Code, or equivalent legal arrangements, and which may be based on the Model Agreement as set out in this circular.

3 The Model Agreement covers the elements included in appendix 3 of the RO Code. The appendix to the Model Agreement is an integral part of the Model Agreement. The Administration should specify the scope and extent of the delegated duties and authorities in an appendix to its agreements with ROs based on the appendix of the Model Agreement.

4 It is recognized that the legal system of the individual Administration as well as additional national requirements may require adaptation of the wording provided in the annex to this circular.

5 The guidance provided by the Model Agreement together with its appendix is considered to meet the minimum standard for a formal written agreement as set forth in both the III Code and RO Code. This Model Agreement, at the discretion of the Administration, may be supplemented by additional matters and/or may be formulated in more detail.

6 Member Governments are invited to use the Model Agreement when concluding a formal agreement with organizations acting on their behalf.

7 This circular supersedes MSC/Circ.710 and MEPC/Circ.307.

MODEL AGREEMENT

AGREEMENT GOVERNING THE DELEGATION OF STATUTORY CERTIFICATION AND SERVICES FOR SHIPS REGISTERED IN [STATE] BETWEEN [ADMINISTRATION] AND [RECOGNIZED ORGANIZATION]

1 Application

This Agreement delegates authority and sets forth the conditions for cooperation between [State] (hereinafter referred to as "the Administration") and the recognized organization (hereinafter referred to as "the RO") with respect to the performance of statutory certification and services on behalf of the Administration, as outlined in the *Code for Recognized Organizations* (RO Code) (resolutions MSC.349(92) and MEPC.237(65)).

2 Purpose

The purpose of this Agreement is to delegate authority to the RO to perform statutory certification and services and to define the scope, terms, conditions and requirements of that delegation.

3 General conditions

3.1 Statutory certification and services comprise the assessment of [State] registered ships and/or ship management companies in order to determine the compliance of such ships/companies with the applicable requirements of the international conventions, codes and national legislation (hereinafter referred to as "applicable instruments") and the issuance of, or in support of the issuance of, relevant certificates and documents as set out in the appendix hereto.

3.2 Insofar as the statutory certification and services covered by this Agreement are concerned, the RO agrees to cooperate with port State control authorities to facilitate the rectification of reported deficiencies or other discrepancies on behalf of the Administration when so requested, and report to the Administration.

3.3 Statutory services rendered and statutory certificates issued by the RO will be accepted as services rendered by or certificates issued by the Administration provided that the RO maintains compliance with the provisions of the RO Code and the national requirements.

4 Execution of functions under authorization

- 4.1 Functions in accordance with the general authorization:
 - .1 the RO is hereby authorized to carry out statutory certification and services on behalf of the Administration with respect to ships, registered in [State], and/or companies responsible for operating such ships, to the extent specified in the appendix to this Agreement and to require repairs to ships in accordance with the applicable instruments;
 - .2 surveys shall be carried out in accordance with applicable instruments and, where applicable, the *Survey Guidelines under the Harmonized System of Survey and Certification, 2019* (resolution A.[...](31), as may be amended); and

- .3 the RO and others acting on its behalf are authorized to take relevant control measures in accordance with their standard procedures or as explicitly advised otherwise by the Administration as is necessary to ensure that matters subject to surveys, audits and inspections correspond substantially with the particulars of the ship's certificates or the requirements of applicable instruments.
- 4.2 Functions in accordance with special (additional) authorization

Authorizations for services outside the scope of the appendix to this Agreement will be dealt with as mutually agreed on a case-by-case basis.

4.3 Relationship between the Organization's statutory and other related activities

The RO and its staff shall not engage in any activities that may conflict with their independence of judgement and integrity in relation to their statutory certification and services. The RO and its staff responsible for carrying out the statutory certification and services shall not be the designer, manufacturer, supplier, installer, purchaser, owner, user or maintainer of the item subject to the statutory certification and services, nor the authorized representative of any of these parties. The RO shall not be substantially dependent on a single commercial enterprise for its revenue.

5 Legal basis of the functions under authorization

5.1 Acts, regulations and supplementary provisions

This Agreement is made pursuant to the [legal authority] and in compliance with the RO Code and any supplementary provisions made by the Administration.

5.2 Interpretations

While interpretations of the applicable instruments, as well as the determination of equivalents or the acceptance of substitutes to the requirements of the applicable instruments are the prerogative of the Administration, the RO will cooperate in their establishment as necessary.

5.3 Deviations and equivalent solutions

5.3.1 Exemptions and deviations from, and equivalents to, the requirements of the applicable instruments are the prerogative of the Administration and must be approved by the Administration prior to issuance.

5.3.2 In instances where, temporarily, the requirements of an applicable instrument cannot be met under particular circumstances, the RO will specify such measures or supplementary equipment as may be available to permit the ship to proceed to a suitable port where permanent repairs or rectifications can be effected or replacement equipment fitted.

6 Reporting to the Administration

6.1 Procedures for reporting in the case of general authorization

The RO shall establish relevant procedures for reporting to the Administration in the case of general authorization, in accordance with the requirements of the RO Code and the additional requirements of the Administration.

6.2 **Procedures for reporting in the case of special authorization**

The RO shall establish relevant procedures for reporting to the Administration in the case of special authorization, in accordance with the requirements of the RO Code and the additional requirements of the Administration accompanying each special authorization.

6.3 Reporting on classification of ships (assignment of class, changes and withdrawals, as applicable).

6.4 Reporting of cases where a ship did not in all respects remain fit to proceed to sea without danger to the ship or persons on board, or presenting unreasonable threat of harm to the environment.

- 6.5 Other reporting:
 - .1 whenever a request is received for classification of a ship to which statutory requirements apply and which will be or has been registered in [the register], the RO shall immediately inform the Administration;
 - .2 the RO shall notify the Administration immediately upon becoming aware of a situation involving a major deficiency or serious safety-related issue that would normally be considered sufficient to detain a ship from proceeding to sea pending correction;
 - .3 the RO shall notify the Administration immediately upon becoming aware of a situation aboard a ship or within a company responsible for operating such ships involving a major non-conformity, as defined in the revised *Guidelines* on the implementation of the International Safety Management (ISM) Code by Administrations (resolution A.1118(30), as may be amended);
 - .4 the above notifications shall contain the name of the company or ship, the IMO number, the official number, if applicable, and a description of the major non-conformity, deficiency or issue;
 - .5 the RO shall inform the Administration, as soon as possible, of any dangerous occurrences, accidents, machinery or structural breakdowns, or failures, which relate to the scope of statutory certification and services as defined in the RO Code, that they are aware of on a ship;
 - .6 the RO shall report to the Administration in writing the names and official numbers, if applicable, of any ships removed from the RO's list of classed/certified ships for which the RO has performed statutory certification and services. The report should contain a description of the reason(s) for removal from class, and this should be made within thirty (30) days of the removal becoming effective; and
 - .7 in cases of transfer of the certification of the ship from one RO to another, the gaining organization shall, within one month from the issuance of the certificates, advise the Administration and the losing organization of the date of issue of the certificates and confirm the date, place and action taken to satisfy each overdue survey, overdue recommendation and overdue condition of class.

7 Development of rules and/or regulations – Information

7.1 Cooperation in connection with the development of rules and/or regulations – liaison meetings

The RO shall allow participation in the development and review of its rules, procedures and/or regulations, specifically in the review process prior to finalization, by representatives of the Administration.

- 7.2 Exchange of rules and/or regulations and information:
 - .1 the Administration will provide the RO with all necessary documentation for the purpose of the RO's provision of statutory certification and services; and notify the RO of any additions, deletions or revisions thereto in advance of their effective date and specify whether the Administration's standards go beyond convention requirements in any respect;
 - .2 the Administration shall be granted access upon request to all plans and documents including reports and records on surveys, audits and inspections on the basis of which certificates are, or have been, issued or endorsed by the RO;
 - .3 the RO should provide free electronic access to the latest editions of its rules and/or regulations applicable to ships under the scope of authorization in the appendix and listing of ships in its class; and
 - .4 the RO and the Administration, recognizing the importance of technical liaison, agree to cooperate towards this end and maintain an effective dialogue.

7.3 Language and form

The RO shall establish, publish and systematically maintain its rules and/or regulations printed and/or in electronic environment, an up-to-date version of which shall be provided in the English language, for the design, construction and certification of ships and their associated essential engineering systems.

8 Other conditions

8.1 Remuneration

Unless agreed otherwise, remuneration for statutory certification and services carried out by the RO on behalf of the Administration will be charged by the RO directly to the party requesting such services.

8.2 Rules for administrative proceedings

8.3 Confidentiality

Insofar as activities related to this Agreement are concerned, both the RO and the Administration shall be bound by confidentiality provisions to be agreed between them.

8.4 Liability and financial responsibility:

- .1 in the context of this Agreement, if a liability is finally and definitively imposed by a court of law on the State of the Administration for loss or damage which is proved in a court of law to have been caused by any wilful act, omission or gross negligence of the RO, its officers, employees or others who were acting on behalf of the RO pursuant to this Agreement, the Administration is entitled to seek from the RO compensation up to but not exceeding the amount of financial liability, as agreed bilaterally between the RO and the Administration;
- .2 while acting for the Administration under this Agreement, the RO shall be free to create contracts direct with its clients and such contracts may contain the RO's normal contractual conditions for limiting its legal liability; and
- .3 in the context of this Agreement, a professional indemnity or professional liability insurance is to be effected by the RO.

8.5 Entry into force

The agreement commences on [date].

8.6 Termination

This Agreement may be terminated by either Party by giving the other Party 12 months written notice.

8.7 Breach of agreement

If this Agreement is breached by one of the Parties, the other Party will notify the violating Party of its breach in writing to allow the notified Party the opportunity to remedy the breach within [...] days, failing which the notifying Party has the right to terminate the Agreement immediately.

8.8 Settlement of disputes

The Agreement shall be governed by and construed in accordance with [Agreed State] law. Any dispute arising in connection with this Agreement which cannot be settled by private negotiations between the Parties shall be settled finally by arbitration in [Agreed location] according to [Agreed State] law, or by the tribunal (courts).

- 8.9 Use of subcontractors:
 - .1 the RO shall perform statutory certification and services by the use of only exclusive surveyors and auditors, being persons solely employed by the RO, duly qualified, trained and authorized to execute all duties and activities incumbent upon their employer, within their level of work responsibility;
 - .2 however, if the RO finds in exceptional and duly justified cases that its own exclusive surveyor is not available, the RO shall inform the Administration, who may then nominate an exclusive surveyor of another RO;

- .3 in accordance with applicable instruments, the RO may choose to outsource any service that affects conformity to requirements or accepts work of a third party approved by the RO. The RO shall ensure that it fully controls the performance of such services through its quality management system. For the purpose of accountability to the Administration, the work performed by the subcontracted organization or service supplier constitutes the work of the RO; and
- .4 while still remaining responsible for the certification on behalf of the Administration, the RO may subcontract radio surveys to non-exclusive surveyors in accordance with the RO Code and the national requirements.

8.10 Amendments

Amendments to the Agreement and the appendix will become effective only after consultation and written agreement between the Administration and the RO.

9 Specification of the authorization from the Administration to the organization

Refer to the appendix.

10 The Administration's supervision of duties delegated to the organization

The Administration shall be entitled to satisfy itself that the RO is performing its services in compliance with the RO Code and this Agreement by establishing or participating in, as applicable, an RO oversight programme for the purpose of supervising the duties delegated to the RO under this Agreement by audits, inspections, supplementary surveys, vertical contract audits or other monitoring activities. For the purpose of this paragraph:

- .1 the Administration shall have access to the documentation of the quality management system of the RO;
- .2 the Administration shall have access to the internal instructions, circulars and guidelines of the RO;
- .3 the Administration shall have access to the RO's documentation relevant to the Administration's fleet;
- .4 the RO shall fully cooperate with the Administration's inspection and verification work;
- .5 the RO shall keep the information and statistics appropriately on cases such as, but not limited to, damage and casualties relevant to the Administration's fleet; and
- .6 the RO shall keep the information and statistics of port State control detentions and deficiencies allocated to the responsibility of the RO.

APPENDIX

of the

AGREEMENT GOVERNING THE DELEGATION OF STATUTORY CERTIFICATION AND SERVICES FOR-SHIPS REGISTERED IN [STATE] BETWEEN [ADMINISTRATION] AND [RECOGNIZED ORGANIZATION] DATED [DATE] WITH EFFECT FROM [DATE] COVERING THE:

APPLICABLE INSTRUMENTS

and

DEGREE OF AUTHORIZATION

1 The RO is hereby authorized as listed below to carry out statutory certification and services and require repairs on behalf of the Administration in accordance with the applicable instruments.

The Administration shall specify the scope of authorization granted to an RO, considering the following:

- .1 ship types and sizes;
- .2 conventions and other instruments, including relevant national legislation;
- .3 approval of drawings;
- .4 approval of materials and equipment;
- .5 surveys, audits, inspections;
- .6 issuance, endorsement and/or renewal of certificates;
- .7 corrective actions;
- .8 withdrawal or cancellation of certificates; and
- .9 reporting requirements.

The following three types of authorizations apply as noted:

- F: Full Authorization to perform plan review, carry out surveys, inspections and audits, and issue and/or revoke necessary interim and full-term certificates.
- P: Partial Authorization to perform plan review, carry out surveys, inspections and audits, and possibly issue interim certificates (specific guidance to be provided by the Administration and full-term certificate is to be issued by the Administration).
- L: Limited Authorization to account for other special categories not covered by the above, such as case-by-case authorization or geographical limitations.
- 2 The listing of instruments and type of authorization is as follows:

To be followed by a listing of the applicable instruments with specific identification of the desired authorization (F/P/L) as applicable.

PROPOSED BIENNIAL AGENDA OF THE MARITIME SAFETY COMMITTEE FOR THE 2022-2023 BIENNIUM

		MARITIME SAFETY COMMITTE				
Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ
1	1.2	Input on identifying emerging needs of developing countries, in particular small island developing States (SIDS) and least developed countries (LDCs) to be included in the ITCP	Continuous	тсс	MSC / MEPC / FAL / LEG	
1	1.3	Validated model training courses	Continuous	MSC / MEPC	III/ PPR/ CCC/ SDC /SSE/NCSR	HTW
1	1.4	Analysis of consolidated audit summary reports	Annual	Assembly	MSC/ MEPC/LEG /TCC/ III	Council
1	1.5	Non-exhaustive list of obligations under instruments relevant to the IMO Instruments Implementation Code (III Code)	Annual	MSC / MEPC		
1	1.7	Identify thematic priorities within the area of maritime safety and security, marine environmental protection, facilitation of maritime traffic and maritime legislation	Annual	тсс	MSC / MEPC / FAL / LEG	
1	1.11	Measures to harmonize port State control (PSC) activities and procedures worldwide	Continuous	MSC / MEPC	HTW / PPR / NCSR	
1	1.16	Development of measures to facilitate mandatory seagoing service required under the International Convention on Standards of Training Certification and Watchkeeping for Seafarers (STCW)	2023	MSC	111	HTW
Notes:	Target of	completion year extended to 2023 as a consequence of the postponement of	f HTW 7 and its	s planned arr	angements.	
1	1.20	Revision of the Guidelines on places of refuge for ships in need of assistance (resolution A.949(23))	2022	MSC	NCSR	

		MARITIME SAFETY COMMITTE				
Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ
1	1.22	Comprehensive review of the 1995 STCW-F Convention	2022	MSC	HTW	
1	1.28	Development of amendments to the Revised guidelines for the development, review and validation of model courses (MSC-MEPC.2/Circ.15/Rev.1)	2022	MSC	HTW	
Notes:	Target c	ompletion year extended to 2022 as a consequence of the postponement of	f HTW 7 and its	s planned arr	angements.	
1	1.29	Development of further measures to enhance the safety of ships relating to the use of fuel oil	2023	MSC		
Notes:	MSC 10	3 extended the target completion year to 2023 (MSC 103/21, paragraph 6.2	6).			
1	1.32	Implementation of the STCW Convention	Continuous	MSC	HTW	
1	1.34	Development of global maritime SAR services, including harmonization of maritime and aeronautical procedures	Continuous	MSC	NCSR	
1	1.[]	Revision of the Criteria for the provision of mobile satellite communication services in the Global Maritime Distress and Safety System (GMDSS) (resolution A.1001(25))	2023	MSC	NCSR	
1	1.[] (New)	Revision of the 1979, 1989 and 2009 MODU Codes and associated MSC circulars to prohibit the use of materials containing asbestos, including control of storage of such materials on board	2023	MSC	SDC	
	Approve	d by MSC 103				
1	1.[] (New)	Development of an entrant training manual for PSC personnel	2023	MSC		
	Approve	d by MSC 103				
1	1.[] (New)	Development of guidance in relation to IMSAS to assist in the implementation of the III Code by Member States	2023	MSC		
	Approve	d by MSC 103				

		MARITIME SAFETY COMMITTE				
Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ
1	1.[…] (New)	Development of guidance on assessments and applications of remote surveys, ISM Code audits and ISPS Code verifications	2024	MSC		
	Approve	ed by MSC 104, and MEPC was invited to consider whether it should be inclu	uded as parent	t organ for the	e output.	
2	2.1	Response to matters related to the ITU-R Study Groups and ITU World Radiocommunication Conference	Continuous	MSC	NCSR	
2	2.3	Amendments to the International Code of Safety for Ships using Gases or other Low-flashpoint Fuels (IGF Code) and development of guidelines for low- flashpoint fuels	Continuous	MSC	HTW / PPR / SDC / SSE	CCC
2	2.4	Mandatory instrument and/or provisions addressing safety standards for the carriage of more than 12 industrial personnel on board vessels engaged on international voyages	2022	MSC	SDC	
2	2.5	Safety objectives and functional requirements of the Guidelines on alternative design and arrangements for SOLAS chapters II-1 and III	2022	MSC	SSE	SDC
Notes:	Due to t	he postponement of SDC 8 to 2022, following the COVID19 pandemic, the t	arget completion	on year has b	een extended	d to 2022
2	2.6	Development of Explanatory Notes to the Interim guidelines on second generation intact stability criteria	2022	MSC	SDC	
Notes:	Due to t	he postponement of SDC 8 to 2022, following the COVID19 pandemic, the t	arget completion	on year has b	een extended	d to 2022
2	2.8	Development of guidelines for cold ironing of ships and consideration of amendments to SOLAS chapters II-1 and II-2	2022	MSC	III / HTW / SDC	SSE
Notes:	The wor	k item has been concluded, pending the input from HTW 8 and SSE 8				
2	2.10	Development of revisions and amendments to existing instruments relating to the amendments to the 1974 SOLAS Convention for modernization of the GMDSS	2022	MSC	HTW / SSE	NCSR

		MARITIME SAFETY COMMITTE				
Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ
2	2.11	Consideration of descriptions of Maritime Services in the context of e-navigation	2022	MSC	FAL / NCSR	
2	2.12	Development of generic performance standards for shipborne satellite navigation system receiver equipment	2022	MSC	NCSR	
2	2.16	Revision of SOLAS chapter III and the International Life-Saving Appliance (LSA) Code	2024	MSC	SSE	
Notes:	To remo SOLAS	ove gaps, inconsistencies and ambiguities based on the safety objectives, fu chapter III	unctional requi	irements and	expected per	formance for
2	2.17	Consideration of development of goal-based ship construction standards for all ship types	2022	MSC MEPC	/	
Notes:	As the C	Committee reviews, on a regular basis GBS audit reports, this item is quasi-c	continuous out	put		
2	2.21	Review of Formal Safety Assessment (FSA) studies by the FSA Experts' Group	Continuous	MSC		
2	2.22	Amendments to the IGC and IGF Codes to include high manganese austenitic steel and related guidance for approving alternative metallic material for cryogenic service	202320231	MSC	CCC	
Notes:	Target of	completion year extended to 2023 as requested by CCC 7.				
2	2.[] (New)	Development of amendments to VDR performance standards and carriage Requirements	2023	MSC		NCSR
2	2.[]	Development of SOLAS amendments for mandatory carriage of electronic inclinometers on container ships and bulk carriers	2022	MSC	NCSR	
2	2.[…] (New)	Development of amendments to SOLAS regulation II-1/3-4 to apply requirements for emergency towing equipment for tankers to other types of ships	2023	MSC	SDC	
Notes:	Approve	ed by MSC 103				

		MARITIME SAFETY COMMITTE				
Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ
4	4.2	Input to the ITCP on emerging issues relating to sustainable development and achievement of the SDGs	Continuous	тсс	MSC / MEPC / FAL / LEG	
5	5.2	Guidelines and guidance on the implementation and interpretation of SOLAS chapter XI-2 and the ISPS Code	Annual	MSC		
5	5.3	Consideration and analysis of reports on piracy and armed robbery against ships	Annual	MSC		
5	5.4	Revised guidance relating to the prevention of piracy and armed robbery to reflect emerging trends and behaviour patterns	Annual	MSC	LEG	
5	5.6	Development of amendments to the STCW Convention and Code for the use of electronic certificates and documents of seafarers	2022	MSC		HTW
5	5.13	IMO's contribution to addressing unsafe mixed migration by sea	2022	MSC / FAL / LEG		
Notes:	MSC 10	04 extended target completion year to 2022				-
6	6.1	Unified interpretation of provisions of IMO safety, security, environment, facilitation, liability and compensation-related conventions	Continuous	MSC / MEPC / FAL / LEG	III / PPR / CCC / SDC / SSE / NCSR	
6	6.2	Developments in GMDSS services, including guidelines on maritime safety information (MSI)	Continuous	MSC	NCSR	
6	6.4	Lessons learned and safety issues identified from the analysis of marine safety investigation reports	Annual	MSC / MEPC		
6	6.5	Identified issues relating to the implementation of IMO instruments from the analysis of PSC data	Annual	MSC / MEPC		

		MARITIME SAFETY COMMITTE				
Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ
6	6.6	Consideration and analysis of reports and information on persons rescued at sea and stowaways	Annual	MSC / FAL		
6	6.10	Amendments to the IMDG Code and supplements	Continuous	MSC	222	
6	6.13	Amendments to the IMSBC Code and supplements	Continuous	MSC	222	
6	6.15	Role of the human element	Continuous	MSC / MEPC	III/ PPR/ CCC/SDC/ SSE/ NCSR	HTW
6	6.16	Development of measures to ensure quality of onboard training as part of the mandatory seagoing service required by the STCW Convention	2023	MSC	HTW	
Notes:	Target o	completion year extended to 2023 as a consequence of the postponement of	f HTW 7 and it	s planned arr	angements.	
6	6.19	Revision of the Code of safety for diving systems (resolution A.831(19)) and the Guidelines and specifications for hyperbaric evacuation systems (resolution A.692(17))	2022	MSC	SSE	
Notes:	Target of	completion year has been consequentially updated to 2022 due to SSE 8's p	ostponement t	o 2022.		
6	6.21	Amendments to the IAMSAR Manual	Continuous	MSC	NCSR	
6	6.22	Amendments to the 2011 ESP Code	Continuous	MSC	SDC	
Notes:	Regular	updates to the 2011 ESP Code agreed by MSC 92 (MSC 92/26, paragraph	13.31)			
6	6.23	Revision of the Revised recommendations for entering enclosed spaces aboard ships (resolution A.1050(27))	2022	MSC	CCC	
Notes:	Target o	completion year is to be consequentially updated to 2022 due to the postpon	ement of this a	agenda item to	o CCC 8.	
6	6.24	Routeing measures and mandatory ship reporting systems	Continuous	MSC	NCSR	
6	6.25	Updates to the LRIT system	Continuous	MSC	NCSR	

		MARITIME SAFETY COMMITTE				
Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ
6	6.26	Verified goal-based new ship construction standards for tankers and bulk carriers	Continuous	MSC		
6	6.27	Amendments to the International Code for the Safe Carriage of Grain in Bulk (resolution MSC.23(59)) to introduce a new class of loading conditions for special compartments	2022	MSC	CCC	
Notes:	Target of	completion year is to be consequentially updated to 2022 due to the postpon-	ement of this	agenda item t	o CCC 8.	
6	6.28	Reports on unlawful practices associated with certificates of competency	Annual	MSC	HTW	
6	6.29	Reports to the MSC on information communicated by STCW Parties	Annual	MSC		
6	6.30	Updated Survey Guidelines under the Harmonized System of Survey and Certification (HSSC)	Annual	MSC MEPC	/ 111	
6	6.31	Consideration of reports of incidents involving dangerous goods or marine pollutants in packaged form on board ships or in port areas	Annual	MSC MEPC	/ 111	CCC
6	6.32	Mandatory application of the Performance standard for protective coatings for void spaces on bulk carriers and oil tankers	2022	MSC	SDC	
6	6.33	Performance standard for protective coatings for void spaces on all types of ships	2022	MSC	SDC	
Notes:	Due to t	he postponement of SDC 8 to 2022, following the COVID19 pandemic, the t	arget complet	ion year has l	been extende	d to 2022
6	6.34	Finalization of a non-mandatory instrument on regulations for non-convention ships	2022	MSC	III	
	MSC 10 Commit	2 agreed that the III should not proceed with the development of a model co tee taking into account the outcome of the work on measures to improve dor	ourse at this tir mestic ferry sa	ne, pending f afety	urther instruc	tions from the

		MARITIME SAFETY COMMITTE				
Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ
6	6.35	Requirements for onboard lifting appliances and anchor handling winches	2022	MSC	HTW	SSE
Notes:	Target o	completion year has been consequentially updated to 2022 due to SSE 8's p	ostponement to	o 2022.		
6	6.36	Review of SOLAS chapter II-2 and associated codes to minimize the incidence and consequences of fires on ro-ro spaces and special category spaces of new and existing ro-ro passenger ships	2022	MSC	HTW / SDC	SSE
Notes:	Target of	completion year has been consequentially updated to 2022 due to SSE 8's p	ostponement to	o 2022.		
6	6.37	Amendments to Guidelines for the approval of fixed dry chemical powder fire-extinguishing systems for the protection of ship carrying liquefied gases in bulk (MSC.1/Circ.1315)	2022	MSC	SSE	
Notes:	Target of	completion year has been consequentially updated to 2022 due to SSE 8's p	ostponement to	o 2022.		
6	6.38	Safety measures for non-SOLAS ships operating in polar waters	2022	MSC	NCSR	SDC
Notes:	Due to t	he postponement of SDC 8 to 2022, following the COVID19 pandemic, the t	arget completion	on year has b	een extended	d to 2022
6	6.39	New requirements for ventilation of survival craft	2022	MSC	SSE	
Notes:	Target of	completion year has been consequentially updated to 2022 due to SSE 8's p	ostponement to	o 2022.		
6	6.40	Consequential work related to the new International Code for Ships Operating in Polar Waters	2022	MSC	SSE / NCSR	SDC
Notes:	Due to t	he postponement of SSE 8 to 2022, following the COVID19 pandemic, the ta	arget completic	on year has b	een extended	to 2022
6	6.43	Revision of the Performance standards for water level detectors on bulk carriers and single hold cargo ships other than bulk carriers (resolution MSC.188(79))	2022	MSC	SSE	SDC
Notes:	Notes: MSC 103, after adoption of new SOLAS regulation II-1/25-1 on "Water level detectors on multiple hold cargo ships other than bulk carriers and tankers", agreed to extend the scope of the output and to revise the associated performance standards and change its title to "Revision of the Performance standards for water level detectors on bulk carriers and single hold cargo ships other than bulk carriers (resolution MSC.188(79))" (MSC 103/21, paragraph 3.11.1)					

		MARITIME SAFETY COMMITTE				
Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ
6	6.45	Development of amendments to the LSA Code and resolution MSC.81(70) to address the in-water performance of SOLAS lifejackets	2023	MSC	SSE	
Notes:	MSC 10	2 approved the inclusion of this item in the provisional agenda of SSE 8.				
6	6.46	Development of amendments to SOLAS chapter II-2 and MSC.1/Circ.1456 addressing fire protection of control stations on cargo ships	2023	MSC	SSE	
Notes:	MSC 10	2 approved the inclusion of this item in the provisional agenda of SSE 8.				
6	6.47	Development of provisions to prohibit the use of fire-fighting foams containing perfluorooctane sulfonic acid (PFOS) for fire-fighting on board ships	2022	MSC	SSE	
6	6.[…] (New)	Revision of ECDIS Guidance for good practice (MSC.1/Circ.1503/Rev.1) and amendments to ECDIS performance standards (resolution MSC.232(82))	2023	MSC	111	NCSR
6	6.[…] (New)	Development of amendments to SOLAS chapter II-2 and the FSS Code concerning detection and control of fires in cargo holds and on the cargo deck of containerships	2025	MSC	CCC	
	Approve	ed by MSC 103				
7	7.1	Endorsed proposals for the development, maintenance and enhancement of information systems and related guidance (GISIS, websites, etc.)	Continuous	Council	MSC / MEPC / FAL / LEG / TCC	
7	7.9	Revised documents on organization and method of work, as appropriate	2022	Council	MSC / MEPC / FAL / LEG / TCC	
OW	OW 13	Endorsed proposals for new outputs for the 2022-2023 biennium as accepted by the Committees	Annual	Council	MSC / MEPC / FAL / LEG / TCC	

		MARITIME SAFETY COMMITTE				
Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ
OW	OW 23	Cooperate with the United Nations on matters of mutual interest, as well as provide relevant input/guidance	2022	Assembly	MSC / MEPC / FAL / LEG / TCC	Council
OW	OW 24	Cooperate with other international bodies on matters of mutual interest, as well as provide relevant input/guidance	2022	Assembly	MSC / MEPC / FAL / LEG / TCC	Council
OW	OW [] (New)	Guidance on the training on and operation of emergency personal radio devices in multiple casualty situations	2022	MSC	NCSR	

BIENNIAL STATUS REPORT OF THE MARITIME SAFETY COMMITTEE FOR THE 2020-2021 BIENNIUM

Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References
1. Improve implementation	1.3	Validated model training courses	Continuous	MSC / MEPC	III / PPR / CCC / SDC / SSE / NCSR	HTW	Ongoing	Ongoing	MSC 100/20, paragraphs 10.3 to 10.6; MEPC 75/18, paragraphs 11.3 to 11.5
1. Improve implementation	1.5	Non-exhaustivelistofobligationsunderinstrumentsrelevanttoIMOInstrumentsImplementationCode(III Code)	Annual	MEPC / MSC	111			Completed	MEPC 64/23, paragraph 11.49; MSC 91/22, paragraph 10.30; MEPC 52/24, paragraph 10.15; MEPC 75/18, paragraph 11.11; MSC 104/18, paragraph 13.7
1. Improve implementation	1.7	Identify thematic priorities within the area of maritime safety and security, marine environmental protection, facilitation of maritime traffic and maritime legislation	Annual	TCC	MSC / MEPC / FAL / LEG		No work requested	Completed	MSC 104/17, MSC 104/18, paragraphs 17.9 to 17.11
1. Improve implementation	1.11	Measures to harmonize port State control (PSC) activities and procedures worldwide	Continuous	MSC / MEPC	HTW / PPR / NCSR	111	Ongoing	Ongoing	MEPC 75/18, paragraphs 11.10 and 11.11; MSC 104/18, paragraph 13.7

Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References
1. Improve implementation	1.13	Review of mandatory requirements in the SOLAS, MARPOL and Load Line Conventions and the IBC and IGC Codes regarding watertight doors on cargo ships	2021	MSC / MEPC	CCC	SDC	In progress	Completed	MSC 101/24, paragraph 21.25; MSC 102/24, paragraph 17.28; MSC 103/21, paragraphs 3.19 and 3.33
Notes:	MSC 10- amendm	4 adopted amendments to the 1 ients to MARPOL and the IGC	1988 Load Line Code will be c	e Protocol a onsidered b	nd the IGC Co y MEPC 77 (N	de (MSC 104/18 /ISC 104/18, par	3, paragraphs agraph 3.8)	3.19 to 3.21) whereas consideration of similar
1. Improve implementation	1.16	Development of measures to facilitate mandatory seagoing service required under the STCW Convention	2023	MSC		HTW		Extended	MSC 101/24, paragraphs 21.29 and 21.30
Notes:	Target c	ompletion year extended to 202	23 as a consec	quence of th	ne postponeme	ent of HTW 7 and	d its planned	arrangemen	ts
1. Improve implementation	1.20	Revision of the Guidelines on places of refuge for ships in need of assistance (resolution A.949(23))	2022	MSC	NCSR		In progress	Extended	MSC 100/20, paragraph 17.1; MSC 104/18, paragraph 15.19
Notes:	Recogni MSC 10	zing that further work on the re 4 extended the target completic	evision of the on year of this	Guidelines output to 20	on places of r)22.	efuge for ships	in need of as	ssistance (res	solution A.949(23)) was required,
1. Improve implementation	1.22	Comprehensive review of the 1995 STCW-F Convention	2022	MSC	HTW			Extended	MSC 95/22, paragraphs 19.3 and 19.4; MSC 96/25, paragraph 12.3;
Notes:	Extende	d to 2022		·	·				
1. Improve implementation	1.27	Revision of the Standardized Life-Saving Appliance Evaluation and Test Report Forms (MSC/Circ.980 and addenda)	2020	MSC	SSE		Completed		MSC 99/22, paragraphs 20.29 and 20.32; SSE 7/21, section 13

Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References
1. Improve implementation	1.28	Development of amendments to the Revised guidelines for the development, review and validation of model courses (MSC-MEPC.2/Circ.15/Rev.1)	2022	MSC	HTW		Extended	Postponed	MSC 100/20, paragraphs 17.7 and 17.8;
Notes:	Target c	ompletion year extended to 202	22 as a consec	quence of th	ne postponeme	ent of HTW 7 and	d its planned	arrangement	ts.
1. Improve implementation	1.29	Development of further measures to enhance the safety of ships relating to the use of fuel oil	2023	MSC			In progress	Extended	MSC 100/20, paragraphs 8.13 and 8.14; MSC 103/21, paragraph 6.26
Notes:	MSC 103	3 extended the target completion	on year to 2023	3 (MSC 103	3/21, paragrapl	h 6.26).			•
1. Improve implementation	1.30	Revision of the Inspection programmes for cargo transport units carrying dangerous goods (MSC.1/Circ.1442, as amended by MSC.1/Circ.1521)	2021	MSC	CCC			Completed	MSC 100/20, paragraph 17.16
1. Improve implementation	1.32	Implementation of the STCW Convention	Continuous	MSC	HTW	HTW	Ongoing	Ongoing	MSC 101/24, paragraph 15.7
1. Improve implementation	1.34	Development of global maritime SAR services, including harmonization of maritime and aeronautical procedures	2021	MSC	NCSR		In progress	Completed	MSC 104/18, paragraph 15.19
Notes:	MSC 102 "Guidelir "Develop output ui the targe	2 agreed with the request of NC nes on harmonized aeronaution oment of global maritime SAR ander SD 1. Recognizing the imp et completion year of this output	SR 7 to combi cal and mariti services, inclu ortance of the t as "continuou	ne outputs me search ding harmo continuous us", until it is	OW 28 "Furthe and rescue mization of ma consideration s decided othe	er development c procedures, incl iritime and aeror of developments rwise.	of the provisio luding SAR nautical proce s on maritime	n of global m training matt edures". MSC search and re	aritime SAR services" and OW 29 ers" under a new output called C 102 also agreed to relocate this escue, MSC 104 agreed to modify

Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References
1. Improve implementation	1.35	Review the Model Agreement for the authorization of recognized organizations acting on behalf of the Administration	2021	MEPC / MSC	111		In progress	Completed	MSC 97/22, paragraph 19.7 (output 2.0.1.6); MSC 102/24, paragraph 14.8 (new output relocated); MEPC 75/18, paragraphs 11.12 and 11.14; MSC 104/18, paragraph 13.10
Notes:	III 7 com	pleted the output and MSC 104	4 confirmed the	e completio	n, subject to M	EPC's concurre	nt decision.		
2. Integrate new and advancing technologies in the regulatory framework	2.1	Response to matters related to the ITU-R Study Groups and ITU World Radiocommunication Conference	Annual	MSC	NCSR		Completed	Completed	MSC 104/19, paragraph 15.19.
Notes:	Recogniz MSC 104	zing the importance of the conti 4 modified the target completio	nuous conside n year of this c	ration of ma output as "c	atters related to ontinuous", un	the ITU-R Study	y Groups and therwise	ITU World R	adiocommunication Conferences,
2. Integrate new and advancing technologies in the regulatory framework	2.3	Amendments to the IGF Code and development of guidelines for low-flashpoint fuels	Continuous	MSC	HTW / PPR / SDC / SSE	CCC	Ongoing	Ongoing	MSC 94/21, paragraphs 18.5 and 18.6; MSC 96/25, paragraphs 10.1 to 10.3; MSC 97/22, paragraph 19.2; PPR 6/20, paragraph 3.39; MSC 102/24, paragraph 21.4
Notes:	MSC 102	2 approved changing the target	t completion ye	ear to "conti	nuous".	•			
2. Integrate new and advancing technologies in the regulatory framework	2.4	Mandatory instrument and/or provisions addressing safety standards for the carriage of more than 12 industrial personnel on board vessels engaged on international voyages	2022	MSC	SDC		Extended	Postponed	MSC 95/22, paragraph 19.25; MSC 96/25, paragraphs 7.10 and 7.12; MSC 97/22, paragraphs 6.22 and 6.23; MSC 99/22, paragraphs 10.17 and 10.18; MSC 101/24, paragraphs 12.17 to 12.19; MSC 102/24, paragraphs 17.10 to 17.20; MSC 103/21, paragraphs 15.5 and 15.6

Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References
Notes:	Due to th	e postponement of SDC 8 to 2	022, following	the COVID	19 pandemic,	the target compl	etion year ha	is been exter	nded to 2022
2. Integrate new and advancing technologies in the regulatory framework	2.5	Safety objectives and functional requirements of the Guidelines on alternative design and arrangements for SOLAS chapters II-1 and III	2022	MSC	SSE	SDC	In progress	Postponed	MSC 82/24, paragraph 3.92; MSC 98/23, annex 38; MSC 102/24, paragraph 19.16.
Notes:	MSC 10 ²	I agreed that the remaining w	ork be limited	to SOLAS	chapter II-1.	Fransfer of outp	ut from SSE	to SDC, as	requested by SSE 7 (SSE 7/21,
	paragrap the targe	h 10.8) and agreed to by MSC t completion year has been ext	102 (MSC 102 tended to 2022	2/24, paragr	aph 19.14); Du	ie to the postpor	ement of SD	C 8 to 2022, f	following the COVID19 pandemic,
2. Integrate new and advancing technologies in the regulatory framework	2.6	Development of Explanatory Notes to the Interim guidelines on second generation intact stability criteria	2022	MSC	SDC		Extended	Postponed	MSC 85/26, paragraphs 12.7 and 23.42; MSC 102/24, paragraph 21.20 and annex 26)
Notes:	With the change t second g been ext	finalization of the second gene he output title from "Finalizatic generation intact stability criteri ended to 2022	eration intact st on of second g a"; Due to the	tability crite eneration in postponem	ria (MSC.1/Cir ntact stability on nent of SDC 8	c.1628), MSC 10 riteria" to "Deve to 2022, followir	02 agreed to elopment of E ng the COVIE	develop asso xplanatory N 019 pandemi	ociated Explanatory Notes and to lotes to the Interim guidelines on c, the target completion year has
2. Integrate new and advancing technologies in the regulatory framework	2.7	Regulatory scoping exercise for the use of Maritime Autonomous Surface Ships (MASS)	2021	MSC	FAL		In progress	Completed	MSC 98/23, paragraph 20.2.11; FAL 43/20, paragraph 17.1; MSC 103/21, paragraphs 5.45 to 5.49.
Notes:	Due to C	OVID19, MSC 102 postponed	the considerat	tion to MSC	2 103 in 2021,	thus postponed	the target co	mpletion yea	r to 2021. FAL 44 postponed the
2. Integrate new and advancing technologies in the regulatory framework	2.8	Development of guidelines for cold ironing of ships and consideration of amendments to SOLAS chapters II-1 and II-2	2022	MSC	III / SDC / HTW	SSE	Extended	Extended	MSC 98/23, paragraph 20.36;SSE 7/21, section 11; MSC 103/24, paragraphs 13.5 and 16.2 to 16.4
Notes:	The work	k item has been concluded, per	nding the input	from HTW	8 and SSE 8.				

Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References
2. Integrate new and advancing technologies in the regulatory framework	2.9	Application of the Indian Regional Navigation Satellite System (IRNSS) in the maritime field and development of performance standards for shipborne IRNSS receiver equipment	2020	MSC	NCSR		Completed		MSC 98/23, paragraphs 11.8 and 11.9; MSC 99/22, paragraph 12.7; resolution MSC.449(99); MSC 102/24, paragraph 16.5; SN.1/Circ.340
2. Integrate new and advancing technologies in the regulatory framework	2.10	Revision of SOLAS chapters III and IV for Modernization of the GMDSS, including related and consequential amendments to other existing instruments	2022	MSC	HTW / SSE	NCSR	In progress	Extended	MSC 98/23, paragraph 20.27; MSC 104/18, paragraphs 12.5 to 12.14 and 15.19, and annexes 10 to 29. COMSAR.1/Circ.37/Rev.1, COMSAR.1/Circ.50/Rev.4, MSC.1/Circ.1039/Rev.1, MSC.1/Circ.1040/Rev.2, MSC.1/Circ.1460/Rev.3, resolutions MSC.493(104) and MSC.494(104)
Notes:	Recogniz this outp Moderniz	zing that the work on the revision out as "Development of revision zation of the GMDSS", with a ta	on of certain e ons and amer arget completic	existing inst adments to on year of 2	ruments had n existing instru 022.	ot been complet uments relating	ted and that f to the amer	further work adments to t	was required, MSC 104 renamed he 1974 SOLAS Convention for
2. Integrate new and advancing technologies in the regulatory framework	2.11	Consideration of descriptions of Maritime Services in the context of e-navigation	2022	MSC	FAL / NCSR		In progress	Extended	FAL 43/20, paragraph 7.21; MSC 101/24, paragraphs 11.10 and 11.11; resolution MSC.467(101); MSC.1/Circ.1610; MSC 104/18, paragraph 15.19
Notes:	Having c context c MSC 10 ² Service	completed the work on the dev of e-navigation and recognizing 1 agreed with the request of NC Portfolios (MSPs)" as "Conside	elopment of g the need for a SR 6 to renam eration of desc	uidance on continuous e the output criptions of	definition and review proces t "Develop guic Maritime Serv	harmonization of s of maritime ser lance on definition ices in the conte	of the format vice descript on and harmo ext of e-navig	and structur ions and the nization of th gation" with a	e of Maritime Services within the harmonization of related services, e format and structure of Maritime a target completion year of 2021.

Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References
	MSC 10 ² to the de	1 also noted the decision of FA	L 43 to include	the FAL Co 104 extende	ommittee as ar ed the target c	associated orga	an for this out of this output f	put (FAL 43/2 to 2022.	20, paragraphs 7.21 to 7.23). Due
2. Integrate new and advancing technologies in the regulatory framework	2.12	Recognition of the Japanese regional navigation satellite system Quasi-Zenith Satellite System (QZSS) and development of performance standards for shipborne satellite navigation system receiver equipment	2022	MSC	NCSR		In progress	Extended	MSC 102/24, paragraph 16.6 and resolution MSC.480(102); MSC 104/18, paragraphs 12.4 and 15.19. SN.1/Circ.341
Notes:	Recognize and that MSC 104 target co	zing that the work on the recogr further work on the developm 4 renamed this output as "Dev mpletion year of 2022.	nition of the Jap ent of generic relopment of g	oanese regio performan eneric perfo	onal navigatior ce standards f ormance stand	n satellite system or shipborne sa lards for shipbo	i Quasi-Zenith tellite navigat rne satellite r	n Satellite Systion system r navigation system	stem (QZSS) had been completed eceiver equipment was required, stem receiver equipment", with a
2. Integrate new and advancing technologies in the regulatory framework	2.16	Revision of SOLAS chapter III and the LSA Code	2024	MSC	SSE		In progress	Postponed	
Notes:	To remov	ve gaps, inconsistencies and a	mbiguities bas	ed on the s	afety objective	s, functional req	uirements an	d expected p	erformance for SOLAS chapter III
2. Integrate new and advancing technologies in the regulatory framework	2.17	Consideration of development of goal-based ship construction standards for all ship types	2023	MEPC / MSC			In progress	Extended	
2. Integrate new and advancing technologies in the regulatory framework	2.21	Review of FSA studies by the FSA Experts' Group	Continuous	MSC			Postponed	Ongoing	MSC 104/18, section 10

Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References
2. Integrate new and advancing technologies in the regulatory framework	2.22	Amendments to the IGC and IGF Codes to include high manganese austenitic steel and related guidance for approving alternative metallic material for cryogenic service	2023	MSC	CCC		In progress	Extended	MSC 96/25 paragraph 23.4; MSC 98/23, annex 38; MSC 100/20 paragraph 17.21; MSC 102/24, paragraph 21.6; MSC 104/18, paragraph 15.16.
Notes:	MSC 104	4 extended the target completion	on year to 2023	3 in order to	complete the	remaining work.			
5. Enhance global facilitation and security of international trade	5.2	Guidelines and guidance on the implementation and interpretation of SOLAS chapter XI-2 and the ISPS Code	Annual	MSC			Postponed	Postponed	
5. Enhance global facilitation and security of international trade	5.3	Consideration and analysis of reports on piracy and armed robbery against ships	Annual	MSC			Postponed	Completed	MSC 104/18, section 8
5. Enhance global facilitation and security of international trade	5.4	Revised guidance relating to the prevention of piracy and armed robbery to reflect emerging trends and behaviour patterns	Annual	MSC	LEG		Postponed	Completed	MSC 104/18, section 8
5. Enhance global facilitation and security of international trade	5.6	Development of amendments to the STCW Convention and Code for the use of electronic certificates and documents of seafarers	2022	MSC		HTW		Extended	HTW 7/16, section 9; MSC 104/18, paragraph 15.17
INOTES:	IVISC 104	4 extended ICY to 2022							

Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References
5. Enhance global facilitation and security of international trade	5.13	IMO's contribution to addressing unsafe mixed migration by sea	2022	FAL / LEG / MSC			In progress	Extended	FAL41/17,paragraph 7.15;MSC 98/23,paragraph 16.14;FAL43,paragraph 10.7;MSC 101/24,paragraph 19.8;MSC 104/18, paragraph 9.5
Notes:	MSC 104 developr	4 concurred with the decision nents.	of FAL 45 to	extend the	e target comp	letion year to 2	022, with the	aim of kee	ping the Committee informed of
6. Ensure regulatory effectiveness	6.1	Unified interpretation of provisions of IMO safety, security, environment, facilitation, liability and compensation-related conventions	Continuous	MSC / MEPC / FAL / LEG	PPR / CCC / SDC / SSE / NCSR / III		Ongoing	Completed	MSC 76/23, paragraph 20.3; MSC 78/26, paragraph 22.12; PPR 7/22 section 18; MEPC 75/18, paragraphs 10.34 and 10.35; MEPC 76/15, paragraphs 4.5, 5.23 and 9.5
Notes:	A 28 exp	anded the output to include all	proposed unif	ied interpre	tations to prov	isions of IMO sa	fety, security	, and environ	ment-related Conventions.
6. Ensure regulatory effectiveness	6.2	Developments in GMDSS services, including guidelines on maritime safety information (MSI)	Continuous	MSC	NCSR		Ongoing	Ongoing	MSC 102/24, paragraphs 16.8 to 16.14; MSC.1/Circ.1364/Rev.2 and MSC.1/Circ.1635; MSC 104/18, paragraph 12.19. MSC.1/Circ.1613/Rev.1
Notes:	MSC 102 Master F maritime	2 agreed with the request of N Plan and guidelines on Maritim safety information (MSI)".	CSR 7 to com le safety inforr	bine outpu nation (MS	ts 6.2 "Develo I)", renaming o	pments in GMD output 6.2 as "D	SS satellite s evelopments	ervices" and in GMDSS	OW 6 "Updating of the GMDSS services, including guidelines on
6. Ensure regulatory effectiveness	6.4	Lessons learned and safety issues identified from the analysis of marine safety investigation reports	Annual	MSC / MEPC			Completed	Completed	MSC 92/26, paragraph 22.29; III 6/15, section 4

Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References
6. Ensure regulatory effectiveness	6.5	Identified issues relating to the implementation of IMO instruments from the analysis of PSC data	Annual	MSC / MEPC	111		Completed	Completed	MSC 96/25, paragraph 23.13
6. Ensure regulatory effectiveness	6.6	Consideration and analysis of reports and information on persons rescued at sea and stowaways	Annual	MSC / FAL			Postponed	Postponed	
6. Ensure regulatory effectiveness	6.10	Amendments to the IMDG Code and supplements	Continuous	MSC	CCC		Ongoing	Ongoing	
6. Ensure regulatory effectiveness	6.13	Amendments to the IMSBC Code and supplements	Continuous	MSC	CCC		Ongoing	Ongoing	
6. Ensure regulatory effectiveness	6.14	Amendments to paragraph 4.4.7.6.17 of the LSA Code concerning single fall and hook systems with on-load release capability	2021	MSC	SSE		Completed		SSE 7/21, section 12
6. Ensure regulatory effectiveness	6.15	Role of the human element	Continuous	MSC / MEPC	III / PPR / CCC / SDC / SSE / NCSR	HTW	Ongoing	Ongoing	MSC 89/25, paragraphs 10.10, 10.16 and 22.39 and annex 21;
6. Ensure regulatory effectiveness	6.16	Development of measures to ensure quality of onboard training as part of the mandatory seagoing service required by the STCW Convention	2023	MSC	HTW			Extended	HTW 7/16, section 10; MSC 104/18, paragraph 15.17
Notes:	Target co	ompletion year extended to 202	1 as a consequ	uence of the	e postponemer	nt of HTW 7 and i	its planned ar	rangements;	MSC 104 extended TCY to 2023.

Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References
6. Ensure regulatory effectiveness	6.17	Revision of the Guidelines for the maintenance and inspections of fixed carbon dioxide fire-extinguishing systems (MSC.1/Circ.1318)	2020	MSC	SSE		Completed		SSE 7/21, section 17
6. Ensure regulatory effectiveness	6.18	Revision of the Guidelines for Vessel Traffic Services (resolution A.857(20))	2020	MSC	NCSR		Completed		MSC 102/24, paragraph 16.7 and annex 11
6. Ensure regulatory effectiveness	6.19 (New)	Revision of the Code of safety for diving systems (resolution A.831(19)) and the Guidelines and specifications for hyperbaric evacuation systems (resolution A.692(17))	2022	MSC	SSE		In progress	Postponed	MSC 99/22, paragraph 20.26
Notes:	Target c	ompletion year has been conse	equentially upo	lated to 202	22 due to SSE	8's postponeme	nt to 2022.	L	1
6. Ensure regulatory effectiveness	6.21	Amendments to the IAMSAR Manual	Continuous	MSC	NCSR		Ongoing	Ongoing	MSC 103/21, paragraph 14.2; MSC.1/Circ.1640
6. Ensure regulatory effectiveness	6.22	Amendments to the 2011 ESP Code	Continuous	MSC	SDC		Ongoing	Ongoing	MSC 92/26, paragraph 13.31;
Notes:	Regular	updates to the 2011 ESP Code	e agreed by M	SC 92 (MS	C 92/26, parag	raph 13.31)			•
6. Ensure regulatory effectiveness	6.23	Revision of the Revised recommendations for entering enclosed spaces aboard ships (resolution A.1050(27))	2022	MSC	CCC			Extended	MSC 101/24, paragraph 21.48; MSC 104/18, paragraph 15.16 (on request CCC 7/15)
Notes:	MSC 104	4 approved target completion y	ear being exte	ended to 20	22	•	•		

Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References
6. Ensure regulatory effectiveness	6.24	Routeing measures and mandatory ship reporting systems	Continuous	MSC	NCSR		Ongoing	Ongoing	
6. Ensure regulatory effectiveness	6.25	Updates to the LRIT system	Continuous	MSC	NCSR		Ongoing	Ongoing	
6. Ensure regulatory effectiveness	6.26	Verified goal-based new ship construction standards for tankers and bulk carriers	Continuous	MSC			Ongoing	Ongoing	MSC 104/18, section 6
6. Ensure regulatory effectiveness	6.27 (New)	Amendments to the International Code for the Safe Carriage of Grain in Bulk (resolution MSC.23(59)) to introduce a new class of loading conditions for special compartments	2022	MSC	CCC		In progress	Extended	MSC 104/18, paragraph 15.16 (CCC 7/WP.2)
Notes:	MSC 104	4 approved extension of the tar	get completion	n year to 20)22			•	
6. Ensure regulatory effectiveness	6.28	Reports on unlawful practices associated with certificates of competency	Annual	MSC	HTW		Postponed	Postponed	MSC 83/28, paragraph 12.2;
6. Ensure regulatory effectiveness	6.29	Reports to the MSC on information communicated by STCW Parties	Annual	MSC			Completed	Completed	
6. Ensure regulatory effectiveness	6.30	Updated Survey Guidelines under the Harmonized System of Survey and Certification (HSSC)	Annual	MEPC / MSC			Completed	Completed	FSI 12/22, paragraph 9.4; MSC 79/23, paragraphs 9.19 and 9.20; MEPC 75/18, paragraphs 10.26, 11.11 and 11.19; MSC 104, paragraph 13.7

Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References
6. Ensure regulatory effectiveness	6.31	Consideration of reports of incidents involving dangerous goods or marine pollutants in packaged form on board ships or in port areas	Annual	MSC / MEPC	111	CCC	Completed	Postponed	CCC 7/15, section 9; MSC 102/24, paragraphs 21.2 and 21.3; MSC 102/24, paragraph 21.3
Notes:	The abov	ve output had the number OW	19 and MSC 1	02 agreed	to relocate it to	strategic directi	on 6.		
6. Ensure regulatory effectiveness	6.32 (New)	Mandatory application of the Performance standard for protective coatings for void spaces on bulk carriers and oil tankers	2022	MSC	SDC		In progress	Postponed	MSC 76/23, paragraphs 20.41.2 and 20.48; DE 50/27, section 4
Notes:	Due to th	he postponement of SDC 8 to 2	022, following	the COVID	19 pandemic,	the target comp	letion year ha	s been exter	nded to 2022
6. Ensure regulatory effectiveness	6.33 (New)	Performance standard for protective coatings for void spaces on all types of ships	2022	MSC	SDC			Postponed	MSC 76/23, paragraphs 20.41.2 and 20.48
Notes:	Due to th	ne postponement of SDC 8 to 2	022, following	the COVID	19 pandemic,	the target comp	letion year ha	s been exter	nded to 2022
6. Ensure regulatory effectiveness	6.34	Finalization of a non- mandatory instrument on regulations for non-convention ships	2022	MSC			In progress	In progress	MSC 96/25, paragraph 9.4; MSC 101/24, paragraph 21.38; MSC 102, paragraph 14.10; MSC 104, section 5 5
6. Ensure regulatory effectiveness	6.35	Requirements for onboard lifting appliances and anchor handling winches	2022	MSC	HTW	SSE	Extended	Postponed	MSC 89/25, paragraph 22.26; MSC 98/23, annex 38
Notes:	Target completion year has been consequentially updated to 2022 due to SSE 8's postponement to 2022.								

Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References	
6. Ensure regulatory effectiveness	6.36	Review of SOLAS chapter II- 2 and associated codes to minimize the incidence and consequences of fires on ro-ro spaces and special category spaces of new and existing ro-ro passenger ships	2022	MSC	HTW / SDC	SSE	In progress	Postponed	MSC 97/22, paragraph 19.19; MSC 98/23, paragraph 12.42	
Notes:	Target c	ompletion year has been conse	equentially upd	ated to 202	2 due to SSE	8's postponeme	nt to 2022.			
6. Ensure regulatory effectiveness	6.37	Amendments to Guidelines for the approval of fixed dry chemical powder fire- extinguishing systems for the protection of ship carrying liquefied gases in bulk (MSC.1/Circ.1315)	2022	MSC	SSE		In progress	Postponed	MSC 98/23, paragraph 20.37;	
Notes:	Target c	ompletion year has been conse	equentially upd	ated to 202	2 due to SSE	8's postponeme	nt to 2022.			
6. Ensure regulatory effectiveness	6.38	Safety measures for non-SOLAS ships operating in polar waters	2022	MSC	NCSR	SDC	In progress	In progress	MSC 98/23, paragraphs 10.29, 20.31.1 and 20.31.2, and annex 38; MSC 99/22, paragraphs 7.16 and 20.13.1; MSC 101/24, paragraphs 7.6 and 7.9; MSC 102/24, paragraphs 17.5 to 17.8; MSC 103/21, paragraphs 15.2 to 15.4	
Notes:	Due to the	Due to the postponement of SDC 8 to 2022, following the COVID19 pandemic, the target completion year has been extended to 2022								
6. Ensure regulatory effectiveness	6.39 (New)	New requirements for ventilation of survival craft	2022	MSC	SSE		In progress	Postponed	MSC 97/22, paragraph 19.22	
Notes:	Target c	ompletion year has been conse	equentially upd	ated to 202	2 due to SSE	8's postponeme	nt to 2022.		1	

Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References
6. Ensure regulatory effectiveness	6.40	Consequential work related to the new International Code for Ships Operating in Polar Waters	2022	MSC	NCSR / SSE	SDC	In progress	Postponed	MSC 93/22, paragraphs 10.44, 10.50 and 20.12; MSC 96/25, paragraph 3.77; MSC 97/22, paragraphs 8.32 and 19.25; MSC 101/24, paragraphs 7.9 and 11.18, and annex 31; MSC.1/Circ.1612; MSC 102/24, paragraph 19.3
Notes:	Due to the	ne postponement of SSE 8 to 2	022, following	the COVID	19 pandemic,	the target compl	etion year ha	s been exten	ded to 2022
6. Ensure regulatory effectiveness	6.41 (New)	Amendments to SOLAS chapter III, LSA Code and resolution MSC.81(70) to remove the applicability of the requirements to launch free-fall lifeboats with the ship making headway at speeds up to 5 knots in calm water	2020	MSC	SSE		Completed		SSE 7/21, section 15
6. Ensure regulatory effectiveness	6.42	Amendments to chapter 9 of the FSS Code for fault isolation requirements for cargo ships and passenger ship cabin balconies fitted with individually identifiable fire detector systems	2020	MSC	SSE		Completed		MSC 98/23, paragraph 20.34; SSE 7/21, section 8
6. Ensure regulatory effectiveness	6.43 (New)	Development of amendments to SOLAS chapter II-1 to include requirements for water level detectors on non-bulk carrier cargo ships with multiple cargo holds	2022	MSC	SSE	SDC	In progress	Extended	MSC 102/24, paragraph 17.23; MSC 103/21, paragraph 3.11.1

Reference to SD, if	Output number	Description	Target completion	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References
approuble			year				rear r		
Notes:	MSC 103 agreed to for water	 after adoption of new SOLA extend the scope of the output level detectors on bulk carrier 	S regulation II It and to revise s and single ho	-1/25-1 on the associated the	Water level de ated performar hips other than	etectors on mult ace standards ar bulk carriers (re	iple hold care of change its solution MSC	go ships othe title to "Revis 2.188(79))" (I	r than bulk carriers and tankers", ion of the Performance standards MSC 103/21, paragraph 3.11.1)
6. Ensure regulatory effectiveness	6.44 (New)	Amendments to the Explanatory Notes to SOLAS chapter II-1 subdivision and damage stability regulations (resolution MSC.429(98))	2020	MSC	SDC		Completed		MSC 101/24/Add.1, annex 31, page 23; MSC 102/24, paragraph 17.2
6. Ensure regulatory effectiveness	6.45 (New)	Development of amendments to the LSA Code and resolution MSC.81(70) to address the in-water performance of SOLAS lifejackets	2023	MSC	SSE		Postponed	Postponed	MSC 101/24, paragraph 21.6; MSC 102/24, paragraph 21.19
Notes:	MSC 102	2 approved the inclusion of this	item in the pro	ovisional ag	enda of SSE 8	3.			•
6. Ensure regulatory effectiveness	6.46 (New)	Development of amendments to SOLAS chapter II-2 and MSC.1/Circ.1456 addressing fire protection of control stations on cargo ships	2023	MSC	SSE		Postponed	Postponed	MSC 101/24, paragraph 21.3; MSC 102/24, paragraph 21.19
Notes:	MSC 102	2 approved the inclusion of this	item in the pro	ovisional ag	enda of SSE 8	3.			
6. Ensure regulatory effectiveness	6.47 (New)	Development of provisions to prohibit the use of fire- fighting foams containing perfluorooctane sulfonic acid (PFOS) for fire-fighting on board ships	2022	MSC	SSE		Postponed	Postponed	MSC 101/24, paragraph 21.27; MSC 102/24, paragraphs 19.31 and 21.19
Notes:	MSC 102 included in the provisional agenda of SSE 8 and agreed that other regulations would need to be amended or a new regulation could be necessary instead; and there could be a need for consequential amendments to other instruments e.g. the HSC Code.								
Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References
--	------------------	--	------------------------------	--------------------	------------------------------------	-----------------------	-----------------------------------	-----------------------------------	---
7. Ensure organizational effectiveness	7.1	Endorsed proposals for the development, maintenance and enhancement of information systems and related guidance (GISIS, websites, etc.)	Continuous	Council	MSC / MEPC / FAL / LEG / TCC		Ongoing	Ongoing	
7. Ensure organizational effectiveness	7.9	Revised documents on organization and method of work, as appropriate	2021	Council	MSC / MEPC / FAL / LEG / TCC		In progress	Postponed	MSC 104/18, paragraphs 1.8 and 14.2
OW. Other work	OW 13	Endorsed proposals for new outputs for the 2020-2021 biennium as accepted by the Committees	Annual	Council	MSC / MEPC / FAL / LEG / TCC		Completed	Completed	
OW. Other work	OW 23	Cooperate with the United Nations on matters of mutual interest, as well as provide relevant input/guidance	2021	Assembly	MSC / MEPC / FAL / LEG / TCC	Council	In progress	Completed	C 120/D, paragraphs 17(a).1 to 17(a).5
OW. Other work	OW 24	Cooperate with other international bodies on matters of mutual interest, as well as provide relevant input/guidance	2021	Assembly	MSC / MEPC / FAL / LEG / TCC	Council	In progress	Completed	C 120/D, paragraphs 17(a).1 to 17(a).5

ANNEX 36

POST-BIENNIAL AGENDA OF THE MARITIME SAFETY COMMITTEE

	Maritime Safety Committee (MSC)												
Number	Biennium (when the output was placed on the post- biennial agenda)	Reference to Strategic Direction, if applicable	Description	Parent organ(s)	Associated organs(s)	Coordinating organ(s)	Timescale (sessions)	References					
145	2016-2017	2	Amendments to the IMDG Code related to portable tanks with shells made of Fibre Reinforced Plastics (FRP) for multimodal transportation of dangerous goods	MSC	ccc		2	MSC 98/23, paragraph 20.11					
152	2016-2017	2	Guidelines for use of Fibre Reinforced Plastics (FRP) within ship structures	MSC	SDC		2	MSC 98/23, paragraph10.22					
180	2020-2021	2	Development of amendments to SOLAS chapters IV and V and performance standards and guidelines to introduce VHF Data Exchange System (VDES)	MSC	NCSR		2	MSC 103/21, paragraph 18.12					

	Maritime Safety Committee (MSC)												
Number	Biennium (when the output was placed on the post- biennial agenda)	Reference to Strategic Direction, if applicable	Description	Parent organ(s)	Associated organs(s)	Coordinating organ(s)	Timescale (sessions)	References					
181	2020-2021	2	Development of performance standards for a digital navigational data system (NAVDAT)	MSC	NCSR		2	MSC 103/21, paragraph 18.18					
156	2018-2019	6	Development of amendments to the LSA Code to revise the lowering speed of survival craft and rescue boats for cargo ships	MSC	SSE		2	MSC 99/22, paragraph 20.15					
158	2018-2019	6	Amendments to SOLAS chapter III and chapter IV of the LSA Code to require the carriage of self-righting or canopied reversible liferafts for new ships	MSC	SSE		2	MSC 99/22, paragraphs 20.22 and 20.23					
169	2018-2019	6	Development of design and prototype test requirements for the arrangements used in the operational testing of free fall lifeboat release systems without launching the lifeboat	MSC		SSE	2	MSC 101/24, paragraph 21.15					

Maritime Safety Committee (MSC)												
Number	Biennium (when the output was placed on the post- biennial agenda)	Reference to Strategic Direction, if applicable	Description	Parent organ(s)	Associated organs(s)	Coordinating organ(s)	Timescale (sessions)	References				
182	2020-2021	6	Revision of the Interim Explanatory Notes addressing the safe return to port (MSC.1/Circ.1369) and related circulars	MSC	HTW / SSE	SDC	1	MSC 103/21, paragraph 18.31				
183	2020-2021	6	Revision of the 2010 FTP Code to allow for new fire protection systems and materials	MSC	SSE		3	MSC 103/21, paragraph 18.28				
9	2012-2013	OW	Revision of the provisions for helicopter facilities in SOLAS and the MODU Code	MSC	SSE		1	MSC 86/26, paragraph 23.39				
168	2018-2019	OW	Development of amendments to paragraph 8.3.5 and annex 1 of the 1994 and 2000 HSC Codes	MSC		SSE	1	MSC 101/24, paragraph 21.9				
42	2012-2013	OW	Review of the 2009 Code on Alerts and Indicators	MSC	NCSR	SSE	2	MSC 89/25, paragraph 22.25				
65	2012-2013	ow	Application of amendments to SOLAS and related codes and guidelines	MSC			2	MSC 91/22, paragraphs 3.16 to 3.35				

Maritime Safety Committee (MSC)												
Number	Biennium (when the output was placed on the post- biennial agenda)	Reference to Strategic Direction, if applicable	Description	Parent organ(s)	Associated organs(s)	Coordinating organ(s)	Timescale (sessions)	References				
90	2014-2015	OW	Amendments to the LSA Code for thermal performance of immersion suits	MSC	SSE		2	MSC 92/26, paragraph 13.34				

ANNEX 37

BIENNIAL STATUS REPORT OF THE SUB-COMMITTEES

	Sub-Committee on Carriage of Cargoes and Containers (CCC)											
Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References			
1. Improve implementation	1.30	Revision of the Inspection programmes for cargo transport units carrying dangerous goods (MSC.1/Circ.1442, as amended by MSC.1/Circ.1521)	2021	MSC	CCC		In progress	Completed	MSC 100/20, paragraph 17.16 CCC 7/15, section 10			
Notes:	Due to th	ne postponement of CCC 7 to	o 2021, follow	ing the CO	VID19 pander	mic, the target c	ompletion ye	ar has been	extended to 2021.			
2. Integrate new and advancing technologies in the regulatory framework	2.3	Amendments to the IGF Code and development of guidelines for low- flashpoint fuels	Continuous	MSC	HTW / PPR / SDC / SSE	CCC	Ongoing	Ongoing	MSC 94/21, paragraphs 18.5 and 18.6; MSC 96/25, paragraphs 10.1 to 10.3; MSC 97/22, paragraph 19.2; PPR 6/20, paragraph 3.39; MSC 102/24, paragraph 21.4; CCC 7/15, section 3			
Notes:	MSC 102	2 approved changing the targ	get completion	h year to "c	ontinuous".				·			
2. Integrate new and advancing technologies in the regulatory framework	2.22	Amendments to the IGC and IGF Codes to include high manganese austenitic steel and related guidance for approving alternative metallic material for cryogenic service	2023	MSC	CCC		In progress	Extended	MSC 96/25, paragraph 23.4; MSC 98/23, annex 38; MSC 100/20, paragraph 17.21; MSC 102/24, paragraph 21.6; MSC 104/18, paragraph 15.16; CCC 7/15, section 4			
Notes:	MSC 104	4 extended the target comple	etion year to 2	:023 in orde	er to complete	the remaining v	vork.					

Sub-Committee on Carriage of Cargoes and Containers (CCC)											
Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References		
6. Ensure regulatory effectiveness	6.1	Unified interpretation of provisions of IMO safety, security, environment, facilitation, liability and compensation-related conventions	Continuous	MSC / MEPC / FAL / LEG	PPR / CCC / SDC / SSE / NCSR / III		Ongoing	Completed	MSC 78/26, paragraph 22.12; PPR 7/22, section 18; MEPC 75/18, paragraphs 10.34 and 10.35; MEPC 76/15, paragraphs 4.5, 5.23 and 9.5; CCC 7/15, section 11		
Notes:	A 28 exp	anded the output to include	all proposed u	unified inter	pretations to	provisions of IM	O safety, sec	urity, and en	vironment-related Conventions.		
6. Ensure regulatory effectiveness	6.10	Amendments to the IMDG Code and supplements	Continuous	MSC	CCC		Ongoing	Ongoing	CCC 7/15, section 6		
6. Ensure regulatory effectiveness	6.13	Amendments to the IMSBC Code and supplements	Continuous	MSC	CCC		Ongoing	Ongoing	CCC 7/15, section 5		
6. Ensure regulatory effectiveness	6.23	Revision of the Revised recommendations for entering enclosed spaces aboard ships (resolution A.1050(27))	2022	MSC	CCC			Extended	MSC 101/24, paragraph 21.48; MSC 104/18, paragraph 15.16 (on request CCC 7/15) MSC 101/24, paragraph 20.7		
Notes:	MSC 104	4 approved target completior	n year being e	xtended to	2022						
6. Ensure regulatory effectiveness	6.27 (New)	Amendments to the International Code for the Safe Carriage of Grain in Bulk (resolution MSC.23(59)) to introduce a new class of loading conditions for special compartments	2022	MSC	CCC			Extended	MSC 104/18, paragraph 15.16 (CCC 7/WP.2) MSC 101/24, paragraph 20.7; MSC 104/18, paragraph 15.16 (reference CCC 7/15)		
Notes:	MSC 104	4 approved extension of the	target comple	tion year to	2022						

Sub-Committee on Carriage of Cargoes and Containers (CCC)												
Sub-committee on carriage of cargoes and containers (CCC)												
Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References			
6. Ensure regulatory effectiveness	6.31	Consideration of reports of incidents involving dangerous goods or marine pollutants in packaged form on board ships or in port areas	Annual	MSC / MEPC	111	CCC	Completed	Completed	CCC 7/15, section 9; MSC 102/24, paragraphs 21.2 and 21.3			
Notes:	The abov	he above output had the number OW 19. However, MSC 102 agreed to relocate it to strategic direction 6.										

	Sub-Committee on Human Element, Training and Watchkeeping (HTW)											
Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References			
1. Improve implementation	1.3	Validated model training courses	Continuous	MSC / MEPC	III / PPR / CCC / SDC / SSE / NCSR	HTW	Postponed	Ongoing	MSC 100/20, paragraphs 10.3 to 10.6; MEPC 75/18, paragraphs 11.3 to 11.5; HTW 7/16, section 3			
1. Improve implementation	1.11	Measures to harmonize port State control (PSC) activities and procedures worldwide	Continuous	MSC / MEPC	HTW / PPR / NCSR	111	Postponed	No work requested	MEPC 75/18, paragraphs 11.10 and 11.11; MSC 104/18, paragraph 13.7			
1. Improve implementation	1.16	Development of measures to facilitate mandatory seagoing service required under the STCW Convention	2023	MSC		HTW	No work requested	Extended	MSC 101/24, paragraphs 21.29 and 21.30; HTW 7/16, section 11			
Notes:	Target cor	mpletion year extended to 2	023 as a con	sequence of	of the postpor	nement of HTV	7 and its pl	anned arrang	gements			
1. Improve implementation	1.22	Comprehensive review of the 1995 STCW-F Convention	2022	MSC	HTW		No work requested	Extended	MSC 95/22, paragraphs 19.3 and 19.4; MSC 96/25, paragraph 12.3; HTW 7/16, section 8			
Notes:	Extended	to 2022							·			
1. Improve implementation	1.26	Revision of MARPOL Annex IV and associated guidelines to introduce provisions for record- keeping and measures to confirm the lifetime performance of sewage treatment plants	2023	MEPC	III / HTW	PPR			MEPC 74/18, paragraphs 14.2 to 14.7; and PPR 7/22, section 16; PPR 8/13, section 7; MEPC 76/14, paragraph 12.6			

Sub-Committee on Human Element, Training and Watchkeeping (HTW)														
Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References					
Notes:	MEPC 74 implement their applic confirm th	mplementation of effluent standards and performance tests for sewage treatment plants (resolution MEPC.227(64)) to address inconsistencies in heir application" to read "Revision of MARPOL Annex IV and associated guidelines to introduce provisions for record-keeping and measures to confirm the lifetime performance of sewage treatment plants". MEPC 76 approved the request by PPR 8 to extend the TCY to 2023.												
1. Improve implementation	1.28	Development of amendments to the Revised guidelines for the development, review and validation of model courses (MSC- MEPC.2/Circ.15/Rev.1)	2022	MŚC	HTW		Extended	Extended	MSC 100/20, paragraphs 17.7 and 17.8; HTW 6/13, section 8					
Notes:	Target cor	mpletion year extended to 2	022 as a con	sequence o	of the postpo	nement of HTV	V 7 and its pl	anned arrang	gements.					
1. Improve implementation	1.32	Implementation of the STCW Convention	Continuous	MSC	HTW	HTW	Postponed	Ongoing	MSC 101/24, paragraph 15.7 HTW 7/16, section 6					
1. Improve implementation	1.33	Development of training provisions for seafarers related to the BWM Convention	2022	MEPC	HTW		No work requested	Extended	MEPC 73/19, paragraph 15.10.1; HTW 7/16, paragraph 12.2					
Notes:	Target cor	mpletion year extended to 2	022 as a con	sequence o	of the postpor	nement of HTV	V 7 and its pl	anned arrang	gements.					
2. Integrate new and advancing technologies in the regulatory framework	2.3	Amendments to the IGF Code and development of guidelines for low- flashpoint fuels	Continuous	MSC	HTW / PPR / SDC / SSE	CCC		No work requested	MSC 94/21, paragraphs 18.5 and 18.6; MSC 96/25, paragraphs 10.1 to 10.3; MSC 97/22, paragraph 19.2; PPR 6/20, paragraph 3.39; MSC 102/24, paragraph 21.4 HTW 6/13, section 12					
Notes:	MSC 102	MSC 102 approved changing the target completion year to "continuous".												

	Sub-Committee on Human Element, Training and Watchkeeping (HTW)											
Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References			
2. Integrate new and advancing technologies in the regulatory framework	2.8	Development of guidelines for cold ironing of ships and consideration of amendments to SOLAS chapters II-1 and II-2	2022	MSC	III / SDC / HTW	SSE	Postponed	Extended	MSC 98/23, paragraph 20.36; SSE 7/21, section 11; MSC 103/24, paragraphs 13.5 and 16.2 to 16.4; HTW 7/16, section 15; MSC 103/24, paragraphs 13.5 and 16.2 to 16.4			
Notes:	The work i	item has been concluded, p	ending the in	put from H	TW 8 and SS	E 8.	•		•			
4. Engage in ocean governance	4.3	Follow-up work emanating from the Action Plan to address marine plastic litter from ships	2023	MEPC	III / PPR / HTW		No work requested	Extended	MEPC 72/17, paragraphs 15.2 to 15.6; MEPC 73/19, section 8 and annex 10; MEPC 74/18, paragraph 8.37.1; PPR 7/22, section 17; MEPC 75/18, section 8; PPR 8/13, section 8; MEPC 76/15, paragraph 12.6			
Notes:	In line with MEPC/PP	n the four sessions approve R for the 2022-2023 bienniu	d to complete um	this work,	as agreed by	MEPC 74, the	e TCY should	be set to 20	23 in the biennial agenda of the			
5. Enhance global facilitation and security of international trade	5.6	Development of amendments to the STCW Convention and Code for the use of electronic certificates and documents of seafarers	2022	MSC	111	HTW	No work requested	Extended	HTW 7/16, section 9; MSC 104/18, paragraph 15.17 HTW 7/16, section 9			
Notes:	MSC 104	extended TCY to 2022					-	-				
6. Ensure regulatory effectiveness	6.15	Role of the human element	Continuous	MSC / MEPC	III / PPR / CCC / SDC / SSE / NCSR	HTW	Postponed	Ongoing	MSC 89/25, paragraphs 10.10, 10.16 and 22.39 and annex 21; HTW 7/16, section 4			

Sub-Committee on Human Element, Training and Watchkeeping (HTW)											
Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References		
6. Ensure regulatory effectiveness	6.16	Development of measures to ensure quality of onboard training as part of the mandatory seagoing service required by the STCW Convention	2023	MSC	HTW			Extended	HTW 7/16, section 10; MSC 104/18, paragraph 15.17		
Notes:	Target completion year extended to 2021 as a consequence of the postponement of HTW 7 and its planned arrangements; MSC 104 extended TCY to 2023.										
6. Ensure regulatory effectiveness	6.28	Reports on unlawful practices associated with certificates of competency	Annual	MSC	HTW		Postponed	Postponed	MSC 83/28, paragraph 12.2; HTW 6/13, section 4		
6. Ensure regulatory effectiveness	6.35	Requirements for onboard lifting appliances and anchor handling winches	2022	MSC	HTW	SSE	Postponed	No work requested	MSC 89/25, paragraph 22.26; MSC 98/23, annex 38		
Notes:	Target cor	mpletion year has been con	sequentially u	updated to	2022 due to \$	SSE 8's postpo	nement to 2	022.	•		
6. Ensure regulatory effectiveness	6.36	Review of SOLAS chapter II-2 and associated codes to minimize the incidence and consequences of fires on ro-ro spaces and special category spaces of new and existing ro-ro passenger ships	2022	MSC	HTW / SDC	SSE		No work requested	MSC 97/22, paragraph 19.19; MSC 98/23, paragraph 12.42; HTW 6/13, section 12		
Notes:	Target cor	mpletion year has been con	sequentially u	updated to	2022 due to \$	SSE 8's postpo	nement to 2	022.			

Sub-Committee on Implementation of IMO Instruments (III)												
Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References			
1. Improve implementation	1.3	Validated model training courses	Continuous	MSC / MEPC	III / PPR / CCC / SDC / SSE / NCSR	HTW			MSC 100/20, paragraphs 10.3 to 10.6; MEPC 75/18, paragraphs 11.3 to 11.5; III 6/15, section 4			
1. Improve implementation	1.4	Analysis of consolidated audit summary reports	Annual	Assembly	MSC / MEPC / LEG / TCC / III	Council	Postponed	Completed	MEPC 61/24, paragraph 11.14.1; MSC 88/26, paragraph 10.8; C 120/D, paragraphs 7.1 and 7.2; MSC 102/24, paragraph 2.2; MEPC 75/18, paragraphs 11.15 to 11.17; III 7/17, section 7			
1. Improve implementation	1.5	Non-exhaustive list of obligations under instruments relevant to the IMO Instruments Implementation Code (III Code)	Annual	MEPC / MSC	111		Postponed	Completed	MEPC 64/23, paragraph 11.49; MSC 91/22, paragraph 10.30; MEPC 52/24, paragraph 10.15; MEPC 75/18, paragraph 11.11; MSC 104/18, paragraph 13.7; III 7/17, section 9			
1. Improve implementation	1.11	Measures to harmonize port State control (PSC) activities and procedures worldwide	Continuous	MSC / MEPC	HTW / PPR / NCSR	111	Postponed	Ongoing	MEPC 75/18, paragraphs 11.10 and 11.11; MSC 104/18, paragraph 13.7 III 7/17, section 5			
1. Improve implementation	1.16	Development of measures to facilitate mandatory seagoing service required under the STCW Convention	2023	MSC	111	HTW			MSC 101/24, paragraphs 21.29 and 21.30			
Notes:	Target co	ompletion year extended to	2023 as a co	onsequence o	f the postponen	nent of HTW 7	and its plann	ed arrangeme	ents			

	Sub-Committee on Human Element, Training and Watchkeeping (HTW)											
Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References			
1. Improve implementation	1.26	Revision of MARPOL Annex IV and associated guidelines to introduce provisions for record- keeping and measures to confirm the lifetime performance of sewage treatment plants	2023	MEPC	III / HTW	PPR			MEPC 74/18, paragraphs 14.2 to 14.7; and PPR 7/22, section 16; PPR 8/13, section 7; MEPC 76/14, paragraph 12.6			
Notes:	MEPC 74 implement application lifetime p	4 agreed to expand the sco ntation of effluent standards on" to read "Revision of MA performance of sewage trea	pe of the exis and perform RPOL Annex tment plants'	sting output 1.2 nance tests for IV and associ . MEPC 76 ap	26 and amend sewage treatn iated guideline proved the req	the title of the on the title of the on the title of the of the title of the of the of the title of the of the of the of the of the of the title of the of t	output from "/ solution MEP provisions for to extend the	Amendments t C.227(64)) to record-keepir e TCY to 2023	to the 2012 Guidelines on address inconsistencies in their ng and measures to confirm the n			
1. Improve implementation	1.35	Review the Model Agreement for the authorization of recognized organizations acting on behalf of the Administration	2021	MEPC / MSC	111		In progress	Completed	MSC 97/22, paragraph 19.7 (output 2.0.1.6); MSC 102/24, paragraph 14.8 (new output relocated); MEPC 75/18, paragraphs 11.12 and 11.14; MSC 104/18, paragraph 13.10 III 7/17, section 15			
Notes:	III 7 com	pleted the output and MSC	104 confirme	d the complet	ion, subject to	MEPC's concu	rrent decisior	<u>ו</u> ז.				
4. Engage in ocean governance	4.3	Follow-up work emanating from the Action Plan to address marine plastic litter from ships	2023	MEPC	III / PPR / HTW		No work requested	Postponed	MEPC 72/17, paragraphs 15.2 to 15.6; MEPC 73/19, section 8 and annex 10; MEPC 74/18, paragraph 8.37.1; PPR 7/22, section 17; MEPC 75/18, section 8; PPR 8/13, section 8; MEPC 76/15, paragraph12.6; III 7/1/Rev.1			
Notes:	In line wi MEPC/P	th the four sessions approv PR for the 2022-2023 bienr	ed to comple nium	te this work, a	s agreed by M	EPC 74, the TC	CY should be	set to 2023 in	the biennial agenda of the			

	Sub-Committee on Human Element, Training and Watchkeeping (HTW)											
Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References			
6. Ensure regulatory effectiveness	6.1	Unified interpretation of provisions of IMO safety, security, environment, facilitation, liability and compensation-related conventions	Continuous	MSC / MEPC / FAL / LEG	PPR / CCC / SDC / SSE / NCSR / III		Postponed	Completed	MSC 76/23, paragraph 20.3; MSC 78/26, paragraph 22.12; PPR 7/22 section 18; MEPC 75/18, paragraphs 10.34 and 10.35; MEPC 76/15, paragraphs 4.5, 5.23 and 9.5; III 7/17, section 10			
Notes:	A 28 exp	anded the output to include	all proposed	d unified interp	retations to pro	ovisions of IMO	safety, secu	rity, and enviro	onment-related Conventions.			
6. Ensure regulatory effectiveness	6.4	Lessons learned and safety issues identified from the analysis of marine safety investigation reports	Annual	MSC / MEPC	111		Postponed	Completed	MSC 92/26, paragraph 22.29; III 6/15, section 4; III 7/17, section 4			
6. Ensure regulatory effectiveness	6.5	Identified issues relating to the implementation of IMO instruments from the analysis of PSC data	Annual	MSC / MEPC	111		Postponed	Postponed	MSC 96/25, paragraph 23.13; III 7/17, section 6			
6. Ensure regulatory effectiveness	6.7	Consideration and analysis of reports on alleged inadequacy of port reception facilities	Annual	MEPC	111		Postponed	Completed	MEPC 74/18, paragraphs 4.33, 4.34 and 8.22; III 6/15, section 3; III 7/17, section 3			
6. Ensure regulatory effectiveness	6.30	Updated Survey Guidelines under the Harmonized System of Survey and Certification (HSSC)	Annual	MEPC / MSC	111		Postponed	Completed	FSI 12/22, paragraph 9.4; MSC 79/23, paragraphs 9.19 and 9.20; MEPC 75/18, paragraphs 10.26, 11.11 and 11.19; MSC 104, paragraph 13.7; III 7/17, section 8			

Sub-Committee on Human Element, Training and Watchkeeping (HTW)											
Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References		
6. Ensure regulatory effectiveness	6.31	Consideration of reports of incidents involving dangerous goods or marine pollutants in packaged form on board ships or in port areas	Annual	MSC / MEPC		CCC			CCC 7/15, section 9; MSC 102/24, paragraphs 21.2 and 21.3		
Notes:	The abov	e output had the number C	W 19. Howe	ver, MSC 102	agreed to relo	cate it to strate	gic direction	6.	·		

	Sub-Committee on Navigation, Communications and Search and Rescue (NCSR)												
Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References				
1. Improve implementation	1.3	Validated model training courses	Continuous	MSC / MEPC	III / PPR / CCC / SDC / SSE / NCSR	HTW	Ongoing	Ongoing	MSC 100/20, paragraphs 10.3 to 10.6; MEPC 75/18, paragraphs 11.3 to 11.5; NCSR 7/23, section 19				
1. Improve implementation	1.20	Revision of the Guidelines on places of refuge for ships in need of assistance (resolution A.949(23))	2022	MSC	NCSR		In progress	Extended	MSC 100/20, paragraph 17.1; MSC 104/18, paragraph 15.19				
Notes:	Recogni required	zing that further work on the , MSC 104 extended the targ	revision of th et completior	e Guideline h year of th	es on places is output to 2	of refuge for sl	nips in need o	f assistance	(resolution A.949(23)) was				
1. Improve implementation	1.34	Development of global maritime SAR services, including harmonization of maritime and aeronautical procedures	2021	MSC	NCSR		In progress	Completed	MSC 104/18, paragraph 15.19				

Sub-Committee on Navigation, Communications and Search and Rescue (NCSR)											
Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References		
Notes:	MSC 102 and OW output ca agreed to rescue, I	2 agreed with the request of 29 "Guidelines on harmonize alled "Development of global o relocate this output under S MSC 104 agreed to modify th	NCSR 7 to co ed aeronautic maritime SA SD 1. Recogr ne target com	ombine out al and mar R services, nizing the ir pletion yea	puts OW 28 itime search including ha nportance of ir of this outp	Further develor and rescue pr rmonization of the continuous ut as "continuous	opment of the ocedures, incl maritime and s consideratio ous", until it is	provision of uding SAR t aeronautica n of develop decided othe	global maritime SAR services" raining matters" under a new I procedures". MSC 102 also ments on maritime search and erwise.		
2. Integrate new and advancing technologies in the regulatory framework	2.1	Response to matters related to the ITU-R Study Groups and ITU World Radiocommunication Conference	Annual	MSC	NCSR		Completed	Completed	MSC 104/18, paragraph 15.19		
Notes:	Recogniz Conferer	zing the importance of the conces, MSC 104 modified the	ntinuous con target comple	sideration etion year o	of matters re	lated to the ITI as "continuous	J-R Study Gro s", until it is de	oups and ITL cided otherv	J World Radiocommunication vise		
2. Integrate new and advancing technologies in the regulatory framework	2.9	Application of the Indian Regional Navigation Satellite System (IRNSS) in the maritime field and development of performance standards for shipborne IRNSS receiver equipment	2020	MSC	NCSR		Completed		MSC 98/23, paragraphs 11.8 and 11.9; MSC 99/22, paragraph 12.7; resolution MSC.449(99); MSC 102/24, paragraph 16.5; SN.1/Circ.340		
2. Integrate new and advancing technologies in the regulatory framework	2.10	Revision of SOLAS chapters III and IV for Modernization of the GMDSS, including related and consequential amendments to other existing instruments	2022	MSC	HTW / SSE	NCSR	In progress	Extended	MSC 98/23, paragraph 20.27; MSC 104/18, paragraphs 12.5 to 12.14 and 15.19, and annexes 10 to 29. COMSAR.1/Circ.37/Rev.1, COMSAR.1/Circ.50/Rev.4, MSC.1/Circ.1039/Rev.1, MSC.1/Circ.1040/Rev.2, MSC.1/Circ.1460/Rev.3, resolutions MSC.493(104) and MSC.494(104)		

Sub-Committee on Navigation, Communications and Search and Rescue (NCSR)												
Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References			
Notes:	Recogniz renamed Convent	zing that the work on the revi this output as "Developmen ion for Modernization of the 0	sion of certai t of revisions GMDSS", with	n existing i and amen <u>n a target c</u>	nstruments h dments to ex ompletion ye	ad not been c isting instrume ar of 2022.	ompleted and ents relating to	that further to the amendr	work was required, MSC 104 nents to the 1974 SOLAS			
2. Integrate new and advancing technologies in the regulatory framework	2.11	Consideration of descriptions of Maritime Services in the context of e-navigation	2022	MSC	FAL / NCSR		In progress	Extended	FAL 43/20, paragraph 7.21; MSC 101/24, paragraphs 11.10 and 11.11; resolution MSC.467(101); MSC.1/Circ.1610; MSC 104/18, paragraph 15.19			
Notes:	Having c within the harmonia context c associate the targe	laving completed the work on the development of guidance on definition and harmonization of the format and structure of Maritime Services vithin the context of e-navigation and recognizing the need for a continuous review process of maritime service descriptions and the larmonization of related services, MSC 101 agreed with the request of NCSR 6 to rename the output "Develop guidance on definition and larmonization of the format and structure of Maritime Services Portfolios (MSPs)" as "Consideration of descriptions of Maritime Services in the context of e-navigation" with a target completion year of 2021. MSC 101 also noted the decision of FAL 43 to include the FAL Committee as an associated organ for this output (FAL 43/20, paragraphs 7.21 to 7.23). Due to the decisions taken in view of the pandemic, MSC 104 extended he target completion year of this output to 2022.										
2. Integrate new and advancing technologies in the regulatory framework	2.12	Recognition of the Japanese regional navigation satellite system Quasi-Zenith Satellite System (QZSS) and development of performance standards for shipborne satellite navigation system receiver equipment	2022	MSC	NCSR		In progress	Extended	MSC 102/24, paragraph 16.6 and resolution MSC.480(102); MSC 104/18, paragraphs 12.4 and 15.19; SN.1/Circ.341			
Notes:	Recogniz been cor equipme system r	ring that the work on the reco npleted and that further work nt was required, MSC 104 re eceiver equipment", with a ta	ognition of the c on the deve enamed this c arget complet	e Japanese lopment of output as "E ion year of	e regional nav generic perfo Development 2022.	vigation satelli ormance stand of generic per	te system Qua lards for shipt formance star	asi-Zenith Sa oorne satellite ndards for sh	itellite System (QZSS) had e navigation system receiver ipborne satellite navigation			

		Sub-Committee	on Navigatio	n, Commu	inications ar	nd Search and	d Rescue (N	CSR)	
Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References
6. Ensure regulatory effectiveness	6.1	Unified interpretation of provisions of IMO safety, security, environment, facilitation, liability and compensation-related conventions	Continuous	MSC / MEPC / FAL / LEG	PPR / CCC / SDC / SSE / NCSR / III		Ongoing	Completed	MSC 76/23, paragraph 20.3; MSC 78/26, paragraph 22.12; PPR 7/22 section 18; MEPC 75/18, paragraphs 10.34 and 10.35; MEPC 76/15, paragraphs 4.5, 5.23 and 9.5; NCSR 7/23, section 18
Notes:	A 28 exp Convent	banded the output to include a ions.	all proposed	unified inte	rpretations to	provisions of	IMO safety, s	ecurity, and	environment-related
6. Ensure regulatory effectiveness	6.2	Developments in GMDSS services, including guidelines on maritime safety information (MSI)	Continuous	MSC	NCSR		Ongoing	Ongoing	MSC 102/24, paragraphs 16.8 to 16.14; MSC.1/Circ.1364/Rev.2 and MSC.1/Circ.1635; MSC 104/18, paragraph 12.19. MSC.1/Circ.1613/Rev.1
Notes:	MSC 102 GMDSS guideline	2 agreed with the request of Master Plan and guidelines es on maritime safety informa	NCSR 7 to co on Maritime s ation (MSI)".	ombine out safety infor	puts 6.2 "Dev mation (MSI)	velopments in ", renaming ou	GMDSS satel itput 6.2 as "D	lite services" evelopments	and OW 6 "Updating of the s in GMDSS services, including
6. Ensure regulatory effectiveness	6.18	Revision of the Guidelines for Vessel Traffic Services (resolution A.857(20))	2020	MSC	NCSR		Completed		MSC 102/24, paragraph 16.7 and annex 11
6. Ensure regulatory effectiveness	6.21	Amendments to the IAMSAR Manual	Continuous	MSC	NCSR		Ongoing	Ongoing	MSC 103/21, paragraph 14.2; MSC.1/Circ.1640
6. Ensure regulatory effectiveness	6.24	Routeing measures and mandatory ship reporting systems	Continuous	MSC	NCSR		Ongoing	Ongoing	MSC 102/24, paragraphs 16.2 to 16.4; COLREG.2/Circ.75; SN.1/Circ.339

Sub-Committee on Navigation, Communications and Search and Rescue (NCSR)												
Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References			
6. Ensure regulatory effectiveness	6.25	Updates to the LRIT system	Continuous	MSC	NCSR		Ongoing	Ongoing				
6. Ensure regulatory effectiveness	6.38	Safety measures for non- SOLAS ships operating in polar waters	2022	MSC	NCSR	SDC	In progress	In progress	MSC 98/23, paragraphs 10.29, 20.31.1 and 20.31.2, and annex 38; MSC 99/22, paragraphs 7.16 and 20.13.1; MSC 101/24, paragraphs 7.6 and 7.9; MSC 102/24, paragraphs 17.5 to 17.8; MSC 103/21, paragraphs 15.2 to 15.4			
Notes:	Due to th	ne postponement of SDC 8 to	o 2022, follow	ving the CC	VID19 pand	emic, the targe	et completion	year has bee	en extended to 2022			

Sub-Committee on Ship Design and Construction (SDC)											
Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References		
1. Improve implementation	1.13	Review of mandatory requirements in the SOLAS, MARPOL and Load Line Conventions and the IBC and IGC Codes regarding watertight doors on cargo ships	2021	MSC / MEPC	CCC	SDC	Completed		MSC 101/24, paragraph 21.25; MSC 102/24, paragraph 17.28; MSC 103/21, paragraphs 3.19 and 3.33; SDC 7/16, paragraph 12.11		
Notes:	MSC 104 considera	l adopted amendments ation of similar amendm	to the 1988 Lonents to MARF	oad Line P OL and th	rotocol and the e IGC Code wil	IGC Code (MSC I be considered	C 104/18, par by MEPC 77	agraphs 3.19 (MSC 104/1	9 to 3.21) whereas 8, paragraph 3.8)		
2. Integrate new and advancing technologies in the regulatory framework	2.4	Mandatory instrument and/or provisions addressing safety standards for the carriage of more than 12 industrial personnel on board vessels engaged on international voyages	2022	MSC	SDC			Postponed	MSC 95/22, paragraph 19.25; MSC 96/25, paragraphs 7.10 and 7.12; MSC 97/22, paragraphs 6.22 and 6.23; MSC 99/22, paragraphs 10.17 and 10.18; MSC 101/24, paragraphs 12.17 to 12.19; MSC 102/24, paragraphs 17.10 to 17.20; MSC 103/21, paragraphs 15.5 and 15.6; SDC 7/16, section 6		
Notes:	Due to th	e postponement of SD	C 8 to 2022, fo	ollowing the	e COVID19 par	idemic, the targe	et completion	year has bee	en extended to 2022		
2. Integrate new and advancing technologies in the regulatory framework	2.5	Sarety objectives and functional requirements of the Guidelines on alternative design and arrangements for SOLAS chapters II-1 and III	2022	MSC	SSE	SDC	NO WORK requested	Postponed	MSC 82/24, paragraph 3.92; MSC 98/23, annex 38; MSC 102/24, paragraph 19.16.		

	Sub-Committee on Ship Design and Construction (SDC)											
Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References			
Notes:	MSC 101 (SSE 7/2 the COVI	l agreed that the remain 1, paragraph 10.8) and ID19 pandemic, the targ	ning work be li agreed to by get completion	mited to S MSC 102 (year has	OLAS chapter I MSC 102/24, p been extended	I-1. Transfer of aragraph 19.14 to 2022	output from S); Due to the	SSE to SDC, postponeme	as requested by SSE 7 nt of SDC 8 to 2022, following			
2. Integrate new and advancing technologies in the regulatory framework	2.6	Development of Explanatory Notes to the Interim guidelines on second generation intact stability criteria	2022	MSC	SDC			Postponed	MSC 85/26, paragraphs 12.7 and 23.42; MSC 102/24, paragraph 21.20 and annex 26); SDC 5/15, section 6; SDC 6/13, section 5; SDC 7/16, section 5			
Notes:	With the Notes an Interim gr the targe	With the finalization of the second generation intact stability criteria (MSC.1/Circ.1628), MSC 102 agreed to develop associated Explanatory Notes and to change the output title from "Finalization of second generation intact stability criteria" to "Development of Explanatory Notes to the Interim guidelines on second generation intact stability criteria"; Due to the postponement of SDC 8 to 2022, following the COVID19 pandemic, the target completion year has been extended to 2022										
6. Ensure regulatory effectiveness	6.1	Unified interpretation of provisions of IMO safety, security, environment, facilitation, liability and compensation- related conventions	Continuous	MSC / MEPC / FAL / LEG	PPR / CCC / SDC / SSE / NCSR / III			Ongoing	MSC 76/23, paragraph 20.3; MSC 78/26, paragraph 22.12; PPR 7/22 section 18; MEPC 75/18, paragraphs 10.34 and 10.35; MEPC 76/15, paragraphs 4.5, 5.23 and 9.5 SDC 6/13, section 9; SDC 7/16, section 11			
Notes:	A 28 exp Conventi	anded the output to inc ons.	lude all propos	sed unified	interpretations	to provisions of	IMO safety,	security, and	environment-related			
6. Ensure regulatory effectiveness	6.22	Amendments to the 2011 ESP Code	Continuous	MSC	SDC			Ongoing	MSC 92/26, paragraph 13.31; SDC 6/13, section 7; SDC 7/16, section 10			
Notes:	Regular u	updates to the 2011 ES	P Code agree	d by MSC	92 (MSC 92/26	6, paragraph 13.	31)					

	Sub-Committee on Ship Design and Construction (SDC)											
Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References			
6. Ensure regulatory effectiveness	6.32 (New)	Mandatory application of the Performance standard for protective coatings for void spaces on bulk carriers and oil tankers	2022	MSC	SDC		In progress	Postponed	MSC 76/23, paragraphs 20.41.2 and 20.48; DE 50/27, section 4 SDC 7/16, section 8			
Notes:	Due to th	e postponement of SD	C 8 to 2022, fo	ollowing the	e COVID19 par	demic, the targe	et completion	year has be	en extended to 2022			
6. Ensure regulatory effectiveness	6.33 (New)	Performance standard for protective coatings for void spaces on all types of ships	2022	MSC	SDC			Postponed	MSC 76/23, paragraphs 20.41.2 and 20.48 SDC 7/16, section 9			
Notes:	Due to th	e postponement of SD	C 8 to 2022, fo	ollowing the	e COVID19 par	demic, the targe	et completion	year has be	en extended to 2022			
6. Ensure regulatory effectiveness	6.38	Safety measures for non-SOLAS ships operating in polar waters	2022	MSC	NCSR	SDC	In progress	Postponed	MSC 98/23, paragraphs 10.29, 20.31.1 and 20.31.2, and annex 38; MSC 99/22, paragraphs 7.16 and 20.13.1; MSC 101/24, paragraphs 7.6 and 7.9; MSC 102/24, paragraphs 17.5 to 17.8; MSC 103/21, paragraphs 15.2 to 15.4; SDC 6/13, section 8; SDC 7/16, section 4			
Notes:	Due to th	e postponement of SD	C 8 to 2022, fo	bllowing the	e COVID19 par	idemic, the targe	et completion	year has be	en extended to 2022			

			Sub-Commit	tee on Shi	ip Design and	Construction (SDC)			
Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References	
6. Ensure regulatory effectiveness	6.40	Consequential work related to the new International Code for Ships Operating in Polar Waters	2022	MSC	NCSR / SSE	SDC			MSC 93/22, paragraphs 10.44, 10.50 and 20.12; MSC 96/25, paragraph 3.77; MSC 97/22, paragraphs 8.32 and 19.25; MSC 101/24, paragraphs 7.9 and 11.18, and annex 31; MSC.1/Circ.1612; MSC 102/24, paragraph 19.3	
Notes:	Due to th	e postponement of SSE	E 8 to 2022, fo	llowing the	e COVID19 pan	demic, the targe	et completion	year has be	en extended to 2022	
6. Ensure regulatory effectiveness	6.43 (New)	Development of amendments to SOLAS chapter II-1 to include requirements for water level detectors on non-bulk carrier cargo ships with multiple cargo holds	2022	MSC	SSE	SDC	Completed	Extended	MSC 102/24, paragraph 17.23; MSC 103/21, paragraph 3.11.1; SDC 7/16, paragraph 7.10	
Notes:	MSC 103, after adoption of new SOLAS regulation II-1/25-1 on "Water level detectors on multiple hold cargo ships other than bulk carriers and tankers", agreed to extend the scope of the output and to revise the associated performance standards and change its title to "Revision of the Performance standards for water level detectors on bulk carriers and single hold cargo ships other than bulk carriers (resolution MSC.188(79))" (MSC 103/21, paragraph 3 11 1)									
6. Ensure regulatory effectiveness	6.44 (New)	Amendments to the Explanatory Notes to SOLAS chapter II-1 subdivision and damage stability regulations (resolution MSC.429(98))	2020	MSC	SDC		Completed		MSC 101/24/Add.1, annex 31; MSC 102/24, paragraph 17.2; SDC 7/16, section 3	

		Su	b-Committee	e on Ship S	Systems and	Equipment (S	SE)		
Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References
1. Improve implementation	1.27	Revision of the Standardized Life-Saving Appliance Evaluation and Test Report Forms (MSC/Circ.980 and addenda)	2020	MSC	SSE		Completed		MSC 99/22, paragraphs 20.29 and 20.32; SSE 7/21, section 13
2. Integrate new and advancing technologies in the regulatory framework	2.5	Safety objectives and functional requirements of the Guidelines on alternative design and arrangements for SOLAS chapters II-1 and III	2022	MSC	SSE	SDC	In progress	No work requested	MSC 82/24, paragraph 3.92; MSC 98/23, annex 38; MSC 102/24, paragraph 19.16; SSE 6/18, section 3; SSE 7, section 10
Notes:	MSC 10 ² paragrap pandemi	I agreed that the remaining w h 10.8) and agreed to by MS(c, the target completion year I	ork be limited C 102 (MSC 1 has been exte	to SOLAS 02/24, para ended to 20	chapter II-1. agraph 19.14) 22	Fransfer of outp ; Due to the pos	out from SSE t stponement o	o SDC, as re f SDC 8 to 2	equested by SSE 7 (SSE 7/21, 022, following the COVID19
2. Integrate new and advancing technologies in the regulatory framework	2.8	Development of guidelines for cold ironing of ships and consideration of amendments to SOLAS chapters II-1 and II-2	2022	MSC	III / SDC / HTW	SSE	Completed		MSC 98/23, paragraph 20.36; SSE 7/21, section 11; MSC 103/24, paragraphs 13.5 and 16.2 to 16.4
Notes:	The work	k item has been concluded, pe	ending the inp	ut from HT	W 8 and SSE	8.			
2. Integrate new and advancing technologies in the regulatory framework	2.16	Revision of SOLAS chapter III and the LSA Code	2024	MSC	SSE		In progress	Postponed	SSE 7/21, section 5
Notes:	To remo chapter I	ve gaps, inconsistencies and a II	ambiguities ba	ased on the	safety object	tives, functional	requirements	and expected	ed performance for SOLAS

Sub-Committee on Ship Systems and Equipment (SSE)										
Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References	
6. Ensure regulatory effectiveness	6.1	Unified interpretation of provisions of IMO safety, security, environment, facilitation, liability and compensation-related conventions	Continuous	MSC / MEPC / FAL / LEG	PPR / CCC / SDC / SSE / NCSR / III		Ongoing	Postponed	MSC 78/26, paragraph 22.12; PPR 7/22, section 18; MEPC 75/18, paragraphs 10.34 and 10.35; MEPC 76/15, paragraphs 4.5, 5.23 and 9.5; SSE 7/21, section 16	
Notes:	A 28 exp	panded the output to include al	ll proposed ur	nified interp	pretations to p	rovisions of IMC) safety, secu	rity, and env	ironment-related Conventions.	
6. Ensure regulatory effectiveness	6.14	Amendments to paragraph 4.4.7.6.17 of the LSA Code concerning single fall and hook systems with on-load release capability	2021	MSC	SSE		Completed		SSE 7/21, section 12	
6. Ensure regulatory effectiveness	6.17	Revision of the Guidelines for the maintenance and inspections of fixed carbon dioxide fire-extinguishing systems (MSC.1/Circ.1318)	2020	MSC	SSE		Completed		SSE 7/21, section 17	
6. Ensure regulatory effectiveness	6.19 (New)	Revision of the Code of safety for diving systems (resolution A.831(19)) and the Guidelines and specifications for hyperbaric evacuation systems (resolution A.692(17))	2022	MSC	SSE		In progress	Postponed	MSC 99/22, paragraph 20.26; SSE 7/21, section 14	
Notes:	Target co	ompletion year has been cons	equentially up	odated to 2	022 due to SS	SE 8's postpone	ment to 2022		•	
6. Ensure regulatory effectiveness	6.35	Requirements for onboard lifting appliances and anchor handling winches	2022	MSC	HTW	SSE	Extended	Postponed	MSC 89/25, paragraph 22.26; MSC 98/23, annex 38; SSE 7/21, section 9	
Notes:	Target co	ompletion year has been cons	equentially up	odated to 2	022 due to SS	SE 8's postpone	ment to 2022	•	•	

Sub-Committee on Ship Systems and Equipment (SSE)									
Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References
6. Ensure regulatory effectiveness	6.36	Review of SOLAS chapter II-2 and associated codes to minimize the incidence and consequences of fires on ro-ro spaces and special category spaces of new and existing ro-ro passenger ships	2022	MSC	HTW / SDC	SSE	In progress	Postponed	MSC 97/22, paragraph 19.19; MSC 98/23, paragraph 12.42; SSE 7/21, section 6
Notes:	Target c	ompletion year has been cons	equentially up	odated to 2	022 due to SS	SE 8's postpone	ment to 2022	•	•
6. Ensure regulatory effectiveness	6.37	Amendments to Guidelines for the approval of fixed dry chemical powder fire- extinguishing systems for the protection of ship carrying liquefied gases in bulk (MSC.1/Circ.1315)	2022	MSC	SSE		In progress	Postponed	MSC 98/23, paragraph 20.37; SSE 7/21, section 7
Notes:	Target c	ompletion year has been cons	equentially up	odated to 2	022 due to SS	SE 8's postpone	ment to 2022		•
6. Ensure regulatory effectiveness	6.39 (New)	New requirements for ventilation of survival craft	2022	MSC	SSE		In progress	Postponed	MSC 97/22, paragraph 19.22; SSE 7/21, section 3
Notes:	Target c	ompletion year has been cons	equentially up	odated to 2	022 due to SS	SE 8's postpone	ment to 2022		•
6. Ensure regulatory effectiveness	6.40	Consequential work related to the new International Code for Ships Operating in Polar Waters	2022	MSC	NCSR / SSE	SDC	In progress	Postponed	MSC 93/22, paragraphs 10.44, 10.50 and 20.12; MSC 96/25, paragraph 3.77; MSC 97/22, paragraphs 8.32 and 19.25; MSC 101/24, paragraphs 7.9 and 11.18, and annex 31; MSC.1/Circ.1612; MSC 102/24, paragraph 19.3; SSE 7/21, section 4
Notes:	Due to th	ne postponement of SSE 8 to	2022, followir	ig the COV	ID19 pandem	ic, the target co	mpletion year	has been e	xtended to 2022

	Sub-Committee on Ship Systems and Equipment (SSE)									
Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References	
6. Ensure regulatory effectiveness	6.41 (New)	Amendments to SOLAS chapter III, LSA Code and resolution MSC.81(70) to remove the applicability of the requirements to launch free-fall lifeboats with the ship making headway at speeds up to 5 knots in calm water	2020	MSC	SSE		Completed		SSE 7/21, section 15	
6. Ensure regulatory effectiveness	6.42	Amendments to chapter 9 of the FSS Code for fault isolation requirements for cargo ships and passenger ship cabin balconies fitted with individually identifiable fire detector systems	2020	MSC	SSE		Completed		MSC 98/23, paragraph 20.34; SSE 7/21, section 8	
6. Ensure regulatory effectiveness	6.45 (New)	Development of amendments to the LSA Code and resolution MSC.81(70) to address the in-water performance of SOLAS lifejackets	2023	MSC	SSE			Postponed	MSC 101/24, paragraph 21.6; MSC 102/24, paragraph 21.19; SSE 7/21, paragraph 20.20	
Notes:	MSC 10	2 approved the inclusion of thi	s item in the p	orovisional	agenda of SS	E 8.				
6. Ensure regulatory effectiveness	6.46 (New)	Development of amendments to SOLAS chapter II-2 and MSC.1/Circ.1456 addressing fire protection of control stations on cargo ships	2023	MSC	SSE			Postponed	MSC 101/24, paragraph 21.3; MSC 102/24, paragraph 21.19 SSE 7/21, paragraphs 20.39 and 20.40.	
Notes:	MSC 10	2 approved the inclusion of thi	s item in the p	orovisional	agenda of SS	E 8.				

Sub-Committee on Ship Systems and Equipment (SSE)										
Reference to SD, if applicable	Output number	Description	Target completion year	Parent organ(s)	Associated organ(s)	Coordinating organ	Status of output for Year 1	Status of output for Year 2	References	
6. Ensure regulatory effectiveness	6.47 (New)	Development of provisions to prohibit the use of fire- fighting foams containing perfluorooctane sulfonic acid (PFOS) for fire-fighting on board ships	2022	MSC	SSE			Postponed	MSC 101/24, paragraph 21.27; MSC 102/24, paragraphs 19.31 and 21.19	
Notes:	MSC 102 necessa	VISC 102 included in the provisional agenda of SSE 8 and agreed that other regulations would need to be amended or a new regulation could be necessary instead; and there could be a need for consequential amendments to other instruments e.g. the HSC Code.								

ANNEX 38

PROVISIONAL AGENDAS FOR THE FORTHCOMING SESSIONS OF THE SUB-COMMITTEES

PROVISIONAL AGENDA FOR CCC 8

- 1 Adoption of the agenda
- 2 Decisions of other IMO bodies
- 3 Amendments to the IGF Code and development of guidelines for low-flashpoint fuels (2.3)
- 4 Amendments to the IGC and IGF Codes to include high manganese austenitic steel and related guidance for approving alternative metallic material for cryogenic service (2.22)
- 5 Amendments to the IMSBC Code and supplements (6.13)
- 6 Amendments to the IMDG Code and supplements (6.10)
- 7 Amendments to the International Code for the Safe Carriage of Grain in Bulk (resolution MSC.23 (59)) to introduce a new class of loading conditions for special compartments (6.27)
- 8 Revision of the Revised recommendations for entering enclosed spaces aboard ships (resolution A.1050 (27)) (6.23)
- 9 Consideration of reports of incidents involving dangerous goods or marine pollutants in packaged form on board ships or in port areas (6.31)
- 10 Review of the IGC Code^{*}
- 11 Development of measures regarding the detection and mandatory reporting of containers lost at sea that may enhance the positioning, tracking and recovery of such containers*
- 12 Unified interpretation of provisions of IMO safety, security, and environment-related conventions (6.1)
- 13 Biennial status report and provisional agenda for CCC 9
- 14 Election of Chair and Vice-Chair for 2023
- 15 Any other business
- 16 Report to the Committees

^{*} Output number will be assigned after A 32.

PROVISIONAL AGENDA FOR HTW 8

- 1 Adoption of the agenda
- 2 Decisions of other IMO bodies
- 3 Validated model training courses (1.3)
- 4 Role of the human element (6.15)
- 5 Reports on unlawful practices associated with certificates of competency (6.28)
- 6 Implementation of the STCW Convention (1.32)
- 7 Development of amendments to the Revised guidelines for the development, review and validation of model courses (MSC-MEPC.2/Circ.15/Rev.1) (1.28)
- 8 Comprehensive review of the 1995 STCW-F Convention (1.22)
- 9 Development of amendments to the STCW Convention and Code for the use of electronic certificates and documents of seafarers (5.6)
- 10 Development of measures to ensure quality of onboard training as part of the mandatory seagoing service required by the STCW Convention (6.16)
- 11 Development of measures to facilitate mandatory seagoing service required under the STCW Convention (1.16)
- 12 Development of training provisions for seafarers related to the BWM Convention (1.33)
- 13 Biennial status report and provisional agenda for HTW 9
- 14 Election of Chair and Vice-Chair for 2023
- 15 Any other business
- 16 Report to the Maritime Safety Committee

PROVISIONAL AGENDA FOR III 8

- 1 Adoption of the agenda
- 2 Decisions of other IMO bodies
- 3 Consideration and analysis of reports on alleged inadequacy of port reception facilities (6.7)
- 4 Lessons learned and safety issues identified from the analysis of marine safety investigation reports (6.4)
- 5 Measures to harmonize port State control (PSC) activities and procedures worldwide (1.11)
- 6 Development of an entrant training manual for PSC personnel (1.[...])[†]
- 7 Identified issues relating to the implementation of IMO instruments from the analysis of PSC data (6.5)
- 8 Analysis of consolidated audit summary reports (1.4)
- 9 Development of guidance in relation to IMSAS to assist in the implementation of the III Code by Member States (1.[...])*
- 10 Updated Survey Guidelines under the Harmonized System of Survey and Certification (HSSC) (6.30)
- 11 Non-exhaustive list of obligations under instruments relevant to the IMO Instruments Implementation Code (III Code) (1.5)
- 12 Development of guidance on assessments and applications of remote surveys, ISM Code audits and ISPS Code verifications (1.[...])*
- 13 Unified interpretation of provisions of IMO safety, security and environment-related conventions (6.1)
- 14 Follow-up work emanating from the Action Plan to Address Marine Plastic Litter from Ships (4.3)
- 15 Biennial agenda and provisional agenda for III 9
- 16 Election of Chair and Vice-Chair for 2023
- 17 Any other business
- 18 Report to the Committees

[†] Output number will be assigned after A 32.

PROVISIONAL AGENDA FOR NCSR 9

- 1 Adoption of the agenda
- 2 Decisions of other IMO bodies
- 3 Routeing measures and mandatory ship reporting systems (6.24)
- 4 Updates to the LRIT system (6.25)
- 5 Development of generic performance standards for shipborne satellite navigation system receiver equipment (2.12)
- 6 Safety measures for non-SOLAS ships operating in polar waters (6.38)
- 7 Consideration of descriptions of Maritime Services in the context of e-navigation (2.11)
- 8 Revision of the Guidelines on places of refuge for ships in need of assistance (resolution A.949(23)) (1.20)
- 9 Development of revisions and amendments to existing instruments relating to the amendments to the 1974 SOLAS Convention for modernization of the GMDSS (2.10)
- 10 Developments in GMDSS services, including guidelines on maritime safety information (MSI) (6.2)
- 11 Revision of the Criteria for the provision of mobile satellite communication services in the Global Maritime Distress and Safety System (GMDSS) (resolution A.1001(25)) (1.[...])^{‡*}
- 12 Response to matters related to the ITU-R Study Groups and ITU World Radiocommunication Conference (2.1)
- 13 Development of global maritime SAR services, including harmonization of maritime and aeronautical procedures (1.34)
- 14 Amendments to the IAMSAR Manual (6.21)
- 15 Guidance on the training on and operation of emergency personal radio devices in multiple casualty situations (OW [...])*
- 16 Revision of ECDIS Guidance for good practice (MSC.1/Circ.1503/Rev.1) and amendments to ECDIS performance standards (resolution MSC.232(82)) (6.[...])*
- 17 Development of amendments to VDR performance standards and carriage requirements (2.[...])*

[‡] Output number will be assigned after A 32.

- 18 Development of SOLAS amendments for mandatory carriage of electronic inclinometers on container ships and bulk carriers (2.[...])^{§*}
- 19 Unified interpretation of provisions of IMO safety, security, environment, facilitation, liability and compensation-related conventions (6.1)
- 20 Validated model training courses (1.3)
- 21 Biennial status report and provisional agenda for NCSR 10
- 22 Election of Chair and Vice-Chair for 2023
- 23 Any other business
- 24 Report to the Maritime Safety Committee

[§] Output number will be assigned after A 32.

PROVISIONAL AGENDA FOR SDC 8

Opening of the session and Election of Chair and Vice-Chair for 2023

- 1 Adoption of the agenda
- 2 Decisions of other IMO bodies
- 3 Safety measures for non-SOLAS ships operating in polar waters (6.38)
- 4 Mandatory instrument and/or provisions addressing safety standards for the carriage of more than 12 industrial personnel on board vessels engaged on international voyages (2.4)
- 5 Development of Explanatory Notes to the *Interim guidelines on second generation intact stability criteria* (2.6)
- 6 Amendments to the 2011 ESP Code (6.22)
- 7 Mandatory application of the Performance standard for protective coatings for void spaces on bulk carriers and oil tankers (6.32)
- 8 Performance standard for protective coatings for void spaces on all types of ships (6.33)
- 9 Safety objectives and functional requirements of the Guidelines on alternative design and arrangements for SOLAS chapters II-1 and III (2.5)
- 10 Unified interpretation to provisions of IMO safety, security and environment-related conventions (6.1)
- 11 Revision of the 1979, 1989 and 2009 MODU Codes and associated MSC circulars to prohibit the use of materials containing asbestos, including control of storage of such materials on board^{***}
- 12 Development of amendments to SOLAS regulation II-1/3-4 to apply requirements for emergency towing equipment for tankers to other types of ships*
- 13 Revision of the Performance standards for water level detectors on bulk carriers and single hold cargo ships other than bulk carriers (resolution MSC.188(79))*
- 14 Review of the Guidelines for the reduction of underwater noise (MEPC.1/Circ.833) and identification of next steps
- 15 Biennial status report and provisional agenda for SDC 9
- 16 Election of Chair and Vice-Chair for 2023
- 17 Any other business
- 18 Report to the Maritime Safety Committee

^{*} Output number to be allocated.
PROVISIONAL AGENDA FOR SSE 8

Opening of the session

- 1 Adoption of the agenda
- 2 Decisions of other IMO bodies
- 3 New requirements for ventilation of survival craft (6.39)
- 4 Consequential work related to the new International Code for Ships Operating in Polar Waters (6.40)
- 5 Revision of SOLAS chapter III and the LSA Code (2.16)
- 6 Review of SOLAS chapter II-2 and associated codes to minimize the incidence and consequences of fires on ro-ro spaces and special category spaces of new and existing ro-ro passenger ships (6.36)
- 7 Amendments to Guidelines for the approval of fixed dry chemical powder fire-extinguishing systems for the protection of ships carrying liquefied gases in bulk (MSC.1/Circ.1315) (6.37)
- 8 Development of amendments to the LSA Code and resolution MSC.81(70) to address the in-water performance of SOLAS lifejackets (6.45)
- 9 Requirements for onboard lifting appliances and anchor handling winches (6.35)
- 10 Development of amendments to SOLAS chapter II- 2 and the FSS Code concerning detection and control of fires in cargo holds and on the cargo deck of containerships*
- 11 Development of amendments to SOLAS chapter II-2 and MSC.1/Circ.1456 addressing fire protection of control stations on cargo ships (6.46)
- 12 Development of provisions to prohibit the use of fire-fighting foams containing perfluorooctane sulfonic acid (PFOS) for fire-fighting on board ships (6.47)
- 13 Validated model training courses (1.3)
- 14 Revision of the Code of Safety for Diving Systems (resolution A.831(19)) and the Guidelines and specifications for hyperbaric evacuation systems (resolution A.692(17)) (6.19)
- 15 Unified interpretation of provisions of IMO safety, security and environment-related conventions (6.1)
- 16 Biennial status report and provisional agenda for SSE 9
- 17 Election of Chair and Vice-Chair for 2023
- 18 Any other business
- 19 Report to the Maritime Safety Committee

^{*} Output number to be allocated.

SUBSTANTIVE ITEMS FOR INCLUSION IN THE AGENDAS FOR MSC 105 AND MSC 106

105th session of the Committee (20 to 29 April 2022)

Decisions of other IMO bodies

Amendments to mandatory instruments

Measures to improve domestic ferry safety

Development of further measures to enhance the safety of ships relating to the use of fuel oil

Goal-based new ship construction standards

Development of a goal-based instrument for Maritime Autonomous Surface Ships (MASS)

Measures to enhance maritime security

Piracy and armed robbery against ships

Unsafe mixed migration by sea

Formal safety assessment

Cost implications for MSI and SAR information providers concerning the recognition of multiple GMDSS mobile satellite services

Implementation of IMO instruments (report of the seventh session of the Sub-Committee)

Carriage of cargoes and containers (report of the seventh session of the Sub-Committee)

Ship design and construction (report of the eighth session of the Sub-Committee)

Human element, training and watchkeeping (urgent matters emanating from the eighth session of the Sub-Committee)

Application of the Committee's method of work

Work programme

Any other business

106th session of the Committee (2 to 11 November 2022)*

Decisions of other IMO bodies

Amendments to mandatory instruments

Development of further measures to enhance the safety of ships relating to the use of fuel oil

Goal-based new ship construction standards

Development of a goal-based instrument for Maritime Autonomous Surface Ships (MASS)

Measures to enhance maritime security

Piracy and armed robbery against ships

Unsafe mixed migration by sea

Formal safety assessment

Human element, training and watchkeeping (report of the eighth session of the Sub-Committee)

Ship systems and equipment (report of the eighth session of the Sub-Committee)

Pollution prevention and response (matters emanating from the ninth session of the Sub-Committee)

Navigation, communications and search and rescue (report of the ninth session of the Sub-Committee)

Implementation of IMO instruments (report of the eighth session of the Sub-Committee)

Application of the Committee's method of work

Work programme

Election of Chair and Vice-Chair for 2023

Any other business

The list of items for inclusion in the agenda of MSC 106 is indicative only and depends on the outcome of MSC 105.

DRAFT ASSEMBLY RESOLUTION

COMPREHENSIVE ACTION TO ADDRESS SEAFARERS' CHALLENGES DURING THE COVID-19 PANDEMIC

THE ASSEMBLY,

RECALLING Article 15(i) of the Convention on the International Maritime Organization concerning the functions of the Assembly in relation to performing the functions of the Organization,

RECALLING ALSO resolution MSC.473(ES.2) on *Recommended action to facilitate ship crew change, access to medical care and seafarer travel during the COVID-19 pandemic,* adopted by the Maritime Safety Committee on 21 September 2020,

RECALLING FURTHER resolution MSC.490(103) on *Recommended action to prioritize COVID-19 vaccination of seafarers,* adopted by the Maritime Safety Committee on 14 May 2021,

RECALLING ALSO that the Maritime Safety Committee, at its 102nd session, approved MSC.1/Circ.1636, containing the *Industry recommended framework of protocols for ensuring safe ship crew changes and travel during the coronavirus (COVID-19) pandemic,* recognized the importance of the protocols and invited Member States and relevant stakeholders to consider them and take them into account,

NOTING resolution A.75/17 on *International cooperation to address challenges faced by seafarers as a result of the COVID-19 pandemic to support global supply chains*, adopted by the UN General Assembly on 1 December 2020,¹

NOTING ALSO resolution GB.340/Resolution (Rev.2) – *Resolution concerning maritime labour issues and the COVID-19 pandemic*, adopted by the Governing Body of the International Labour Office on 8 December 2020,²

NOTING FURTHER the Resolution concerning the implementation and practical application of the MLC, 2006, during the COVID-19 pandemic, and the Resolution concerning COVID-19 vaccination for seafarers, adopted by the ILO Special Tripartite Committee of the Maritime Labour Convention, 2006, as amended (MLC, 2006) on 23 April 2021,³

MINDFUL of the severe challenges the COVID-19 pandemic poses for international shipping and Governments' ability to respond effectively to these challenges,

MINDFUL ALSO of the importance of the health and safety of seafarers and the need to protect their mental and physical health, welfare and well-being as these elements are paramount to ensure maritime safety, security and environmental protection,

BEING CONCIOUS OF the unique and essential contribution of seafarers to international

¹ https://undocs.org/en/A/RES/75/17

² https://www.ilo.org/wcmsp5/groups/public/---ed_norm/--relconf/documents/meetingdocument/wcms_760649.pdf

³ https://www.ilo.org/global/standards/maritime-labour-convention/events/WCMS_679152/lang-en/index.htm

shipping and the world by ensuring the uninterrupted transportation of vital medical supplies, food and other basic necessities, including during the COVID-19 pandemic,

BEING AWARE OF the continuously high number of seafarers that require repatriation in accordance with the International Labour Organization's Maritime Labour Convention 2006 (MLC, 2006), and those in urgent need to join ships to replace them,

MINDFUL of the need to protect seafarers through vaccination, as soon as possible, to facilitate their safe movement across borders,

REITERATING the numerous statements by the United Nations (UN) and UN System Organizations emphasizing, inter alia, the need to designate seafarers as "key workers" and ensure their access to medical care, prioritization for COVID-19 vaccinations and facilitation of travel, thus preserving their welfare and rights,

- 1 URGES Member States and relevant national authorities to:
 - .1 designate seafarers as "key workers" in order to facilitate shore leave and safe and unhindered movement across borders, recognizing relevant documentation carried by seafarers as evidence of this status, which would entail the application of temporary measures including (where possible under relevant law) waivers, exemptions or other relaxations from any visa or documentary requirements;
 - .2 consider the implementation of the *Industry recommended framework* of protocols for ensuring safe ship crew changes and travel during the coronavirus (COVID-19) pandemic (MSC.1/Circ.1636/Rev.1, as may be revised);
 - .3 prioritize vaccination of seafarers, as far as practicable, in their national COVID-19 vaccination programmes, noting the advice of the WHO SAGE Roadmap⁴ for prioritizing the use of COVID-19 vaccines; and consider extending COVID-19 vaccines to seafarers of other nationalities, taking into account national vaccines supply;
 - .4 consider exempting seafarers from any national policy requiring proof of COVID-19 vaccination as a condition for entry, taking into account that seafarers should be designated as "key workers" and that they travel across borders frequently;
 - .5 provide seafarers with immediate access to medical care and facilitate medical evacuation of seafarers in need of urgent medical attention when the required medical care cannot be provided either on board or in the port of call;

2 REQUESTS Member States and international organizations to bring this resolution to the attention of all parties concerned.

⁴ WHO SAGE Roadmap for prioritizing the uses of COVID-19 vaccines in the context of limited supply: https://www.who.int/publications/i/item/who-sage-roadmap-for-prioritizing-uses-of-covid-19-vaccines-inthe-context-of-limited-supply

THEMATIC PRIORITIES FOR THE ITCP COVERING THE 2022-2023 BIENNIUM¹

1 Fostering the effective implementation of conventions and other mandatory instruments, with emphasis on the SAR and STCW Conventions and the ISM Code, in particular providing assistance and training to developing countries to effectively implement those instruments, as well as addressing human element-related matters and implementation of provisions related thereto.

2 Promoting the continued establishment and strengthening of effective ship and port facility security measures (SOLAS chapter XI-2 and ISPS Code), including support for LRIT implementation, the enhancement of safety and security of the ship/port interface, in accordance with the relevant IMO standards and recommendations and promoting and enhancing maritime security aspects relating to piracy and armed robbery against ships, including facilitation and effective implementation of resolution A.1025(26) on Code of Practice for the Investigation of Crimes of Piracy and Armed Robbery against Ships, as may be updated by A 31.

3 Supporting maritime administrations to strengthen their human resource capabilities in the discharge of their responsibilities as flag, port and coastal States, in particular with regard to the implementation of the IMO Instruments Implementation Code, the Casualty Investigation Code and the Code for Recognized Organizations, while promoting the global harmonization of port State control, in cooperation with PSC regimes, taking into account the analysis of the information collected through the IMO Member State Audit Scheme, as appropriate.

4 Supporting IMO Member States to strengthen their services dedicated to safety of navigation, monitoring of maritime traffic, implementation of effective ships' routeing measures, and search and rescue.

5 Supporting maritime administrations through capacity-building to strengthen their capabilities to implement the provisions of the IMDG and IMSBC Codes, to improve implementation as well as to promote good practice when dealing with these mandatory instruments.

6 Supporting maritime administrations through capacity-building and as a facilitator for international co-operation to enhance their capabilities and to share and provide them with best practices on the implementation of SOLAS chapter XIV and the Polar Code and associated recently approved guidelines.

7 Promoting the ratification and implementation of the 2012 Cape Town Agreement and the 1995 STCW-F Convention as well as proactive safety measures relating to fishing vessels and their personnel and the fight against Illegal, Unreported and Unregulated (IUU) Fishing, in cooperation with FAO and ILO, including promoting and enhancing maritime safety aspects relating to small fishing vessels.

8 Promoting and enhancing maritime safety aspects relating to ships not covered by SOLAS chapter I (so called non-convention ships), in particular in relation to all aspects of domestic ferry safety.

¹ The identification of beneficiary countries or regions for TC assistance will be based on the information provided by the Country Maritime Profiles (CMPs) in GISIS. Therefore, Member States which have not yet completed their CMPs are urged to do so at their earliest convenience. Member States are also urged to ensure that their TC needs are well defined and updated in their respective CMPs.

STATEMENTS BY DELEGATIONS AND OBSERVERS¹

AGENDA ITEM 8

Statement by the delegation of Romania

"Romania has been a member of the IMO since March 20, 1965 and continuously complies with the objectives of the IMO and enforces the international maritime rules, being strong involved in the implementation of IMO standards and regulations.

Today, Romania is one of important suppliers of seafarers, especially maritime officers. Tens of thousands of Romanian seafarers work in the international fleet, many of them being in command of the ships or in top positions, carrying out, with devotion and professionalism, the mission to safely lead ships and crews and deliver cargo around the world.

Being a seafarer is a peaceful profession. However, seafarers face countless dangers on board ships. Beyond the usual challenges - weather at sea, isolation, work accidents, burnout, lack of access to medical care - they also face dangers related to international terrorism and armed attack on merchant ships.

That happened in the recent case of a 46-year-old Romanian ship master, who tragically lost his life on July 29, 2021, in the Gulf of Oman, during he was in command of M/T Mercer Street, IMO number 9539585, Liberian flag, when his ship was attacked with a drone with explosive charges.

Romania condemns any kind of terrorist activity that endangers the safety and lives of seafarers, whose role is to participate in the wellbeing of the entire civilization. Such aggressions pose threats to the social and economic balance of all humanity.

Romanian Government call on the MSC 104 to empower IMO Secretariat to submit to UN a statement, to emphasize once again the size of terrorist threats on merchant ships, requesting the increasing of security measures in maritime areas at risk, by strengthening military patrols, expanding patrol areas, supporting and counselling coastal governments in the prevention of such situations.

Along with the Romanian captain of the ship, a British citizen, a member of the security team, also lost his life. We extend our condolences to the distinguished United Kingdom delegation for their loss.

All the distinguished delegations of this Committee, understand that any seafarer, of any nationality, can be in such a situation and our role, in the spirit promoted by this Organization from its establishment until today, is to seek solutions and prevent such incidents, to cooperate with governments and international bodies to guarantee safe waters for our seafarers.

In Romania, following this incident, a grieving family, a wife and two children, did not have the opportunity to see their father alive and that can happen anytime, anywhere in the world.

Hoping that such incidents will not occur again, Romania expresses its solidarity with all Member States, against terrorism and armed attacks on merchant ships."

Statement by the delegation of Nigeria

"This statement is on sustained measures being taken by Nigeria as a direct response to curbing acts of Piracy and Armed Robbery in the Gulf of Guinea region.

We have spoken to you at MSC 102 and MSC 103 on progress that has been made in this regard. It is my privilege to give further updates on the subsisting collaborative initiates. This progress has been with active support of the IMO, the Industry, G7++ Friends of Gulf of

¹ Statements have been included in this annex in the order in which they are listed in the report, sorted by agenda items, and in the language of submission (including translation into other languages if provided).

Guinea, NIMASA/Industry Joint Working Group, the ICC and other international partners for which Nigeria is grateful. So far, the Gulf of Guinea has witnessed continued improvement in maritime security in recent times. The IMB Report on Piracy against Ships for the 2nd Quarter of 2021 found a significant, all-time decline in the number of incidents, the lowest in 27 years. Our focus across international, regional, operational, and tactical boundaries is aimed at working towards a set of common operational objectives for the Gulf of Guinea region. Underlisted is Nigeria's three-dimensional approach. Thus:

.1 regional and global anti-piracy support:

Nigeria is working closely with regional and international partners, to ensure a reduction in piracy and criminality within the Gulf of Guinea. The Interregional Coordination Centre (ICC) Yaoundé and Nigeria are working hand-in-hand, and from this relationship the Gulf of Guinea Maritime Collaboration Forum – SHADE (GOG MCF/SHADE) has materialised. This regional architecture, based on international best practise, has within ten (10) weeks of its establishment, held two (2) plenary meetings and ten (10) different Working Groups. It is gaining momentum and offers the platform for stakeholders-collaboration with 16 regional bodies and 13 countries participating. I would therefore like to extend Nigeria's appreciation to the IMO Secretary-General and the governments of following countries – The United Kingdom; The United States; Italy, Germany, South Korea, Denmark, Poland, Japan, Brazil, Norway, Belgium, Greece and Turkey.

This International Framework is already improving shared awareness and will deconflict maritime security activities in the Gulf of Guinea (GOG) to ensure we have the best possible coverage and support to seafarers. This very week, SHADE-GOG will trial a new communications tool focussed on ensuring transparency of incidents across and the best possible chance of response to vessels in distress.

.2 industry organisations:

NIMASA and Nigerian Navy continue to work closely with industry through our jointly chaired, Joint Working Group. This forum ensures the voice of Industry, our customer, is heard and taken into consideration. Recently, we have agreed sets of common objectives on which we will focus to sustain the efforts geared towards addressing the challenges in the Gulf of Guinea region.

Accordingly, the NIMASA Joint Industry Working Group has developed a strategy on maritime security in the Gulf of Guinea and we shall keep you posted on further developments regarding this.

The IMO is also helping to develop a maritime security strategy document.

Similarly, work is ongoing on collaborative efforts between Nigeria and the UNODC and we are engaging the IMO in this respect.

The newly created Maritime Intelligence Unit was established and has made significant successes in addressing piracy from its source.

On the Legal Framework, that is the enforcement of the Suppression of Piracy and Other Maritime Offences (SPOMO Act), so far, we have secured a total of 23 convictions.

.3 national approach:

Nationally, the Administration has been engaging constructively with critical stakeholders, especially the Nigerian Navy; a critical partner who plays a pivotal role in Nigeria's counter-piracy and armed robbery prevention. The Office of

National Security Adviser (ONSA) to the President of Nigeria, continue to give moral and oversight support and coordinates relevant security agencies like the Nigerian Airforce, Army, Police (INTERPOL), Immigration Services etc to achieve this aim.

The Deep Blue Project, which was officially launched on 10 June 2021 by President Muhammadu Buhari and, goodwill message by the Secretary General was very much appreciated. I am pleased to report the maritime assets of Deep Blue are starting to make a difference, as there have been no reported incidents within the Lagos anchorage since we took control; with plans afoot to move the assets further offshore to the areas where the piracy threat is highest. To sustain the efforts in the areas of human capacity building, a regional maritime safety and security centre has been established by NIMASA with state-of-the-art facilities ranging from a 28-seater e-library, a multipurpose hall, 6 classroom blocks with 30 seating capacity each; 1 theatre room with 50 seating capacity; multipurpose hall of 350 seating capacity, all situated in the serene environment with a helipad and a jetty to enable access by sea and air.

Conclusion

Whilst there is still much to do, the early signs are positive. Increased collaboration across international and regional stakeholders, our partnership with industry and the delivery of Deep Blue Project are significant steps in the right direction. This is a team game, together we are making a difference to ensure the GOG has a secure maritime domain and seafarers can safely ply their trade using the collaborative platform."

Statement by the observer from ICS

"The cosponsors of MSC 104/8/2, ICS, BIMCO, INTERTANKO, INTERCARGO and OCIMF had not intended to comment on the submission, given your request for brevity at the opening of this session of the Committee. However, from some statements made it appears there is a misunderstanding as to the purpose of both the HRA, our proposal to define a better global mechanism to inform seafarers of threats, and how they correlate to international and national mandates and international maritime law. The co-sponsors would therefore request to make the following statement, for the sake of clarity.

The HRA and the proposed dynamic threat assessment process are very simply guidance developed by the industry for the benefit of industry. As demonstrated over many years, such guidance has provided an essential practical means for companies and ships to risk assess their voyages and ensure that measures are taken to protect them from security threats. This is at the core of the guidance provided in Best Management Practices series, guidance that has been invaluable in defending ships from security threats, and one of the key pillars in the successful fight against Somali piracy and the ongoing fight against piracy in the Gulf of Guinea.

The High-Risk Area as it currently stands is solely focussed on piracy and whilst BMP was initially developed as counter piracy guidance for a specific threat in a specific area, bitter experience has demonstrated the existence of many other maritime security threats in many other places. Indeed, the military mandates of operations in the Western Indian Ocean have expanded to account for many maritime threats beyond piracy, expansions in focus and intent which the shipping industry welcomes.

In recent years merchant ships have been subjected to attack by waterborne IEDs, limpet mines, cyber-attack, and other security threats. Furthermore, we heard yesterday, of the horrific case of the MV Mercer Street, in which a drone attack against an innocent vessel resulted in the deaths of two personnel. Such threats exist, irrespective of mandates, heedless of legal jurisdiction, and it is seafarers that find themselves in the front line.

The shipping industry takes its commitments to maritime security and the safety of its seafarers extremely seriously, and it is for this reason it has developed its own security guidance to augment international requirements such as ISPS.

The dynamic threat assessment process is a progression of this work, and will allow us to identify and mitigate against maritime security threats more consistently, wherever they are encountered, keeping companies and seafarers better informed. This work will be done in consultation with all relevant partners. As with BMP, our global threat assessment will have no legal bearing, but rather will serve to enable greater security and confidence for mariners. From its work in the Western Indian Ocean, and now with Nigeria, Industry fully understands the value of mutual cooperation to address security threats head on, and we look forward to working with all those who share our concerns and commitment to the security of maritime trade.

We are grateful for your summary at the start of our consideration of this item and support your conclusions, and we will update the committee on progress in our work at its next session."

Statement by the observer from ICS

"Document MSC 104/8/2 further updates the Committee on the review of the extent of the High-Risk area in the Western Indian Ocean, Gulf of Aden and Southern Red Sea, and advises of revisions to the area. The Committee is advised that a thorough threat assessment for the area was conducted by the BMP 5 co-authors, drawing on the threat assessments of SHADE, EUNAVFOR Atalanta, UNODC and data from UKMTO. Littoral States were also consulted as part of this process, in particular Kenya, as noted at the previous session of the Committee.

As a consequence of this threat assessment, it became clear that the threat from Somali piracy is at its lowest level since 2007, with pirate action groups and their kingpins now pursuing other criminal enterprises, and this scenario is likely to persist for reasons that are outlined in the paper. Furthermore, other threats have emerged in the region that present a greater risk to shipping, and military mandates have expanded to cover these.

The designation of a high-risk area based solely on the threat from Somali piracy is now outdated, and a more dynamic mechanism is required to account for the changed threat profile in the region. It is also noted that the designation of the area as high risk is now anachronistic, when compared to other regions such as the Gulf of Guinea in which the industry faces far more severe threats, but for which no such designation has been given.

A two-step approach is therefore being taken. Firstly, as of September 1 this year the HRA has been amended to the coordinates described on page 4 of the paper, covering all threats described in BMP 5, removing all remaining Kenyan and Omani waters from the area. This is an interim measure to allow for the continued application of BMP 5.

Secondly, the BMP co-authors are undertaking substantive work to address maritime security threats in a global context through a more dynamic threat assessment process, with a target completion of 31 December 2021, with implementation following thereafter. This process will draw on multiple information and intelligence sources to allow for the clear and consistent definition and identification of security threats wherever they may occur so that shipowners and operators can be aware of those threats, and take necessary mitigation measures according to the risk they present to seafarers and ships.

The co-authors wish to express their thanks to all those who have assisted in the threat assessment and revision of the HRA, and invite the Committee to note the information provided."

AGENDA ITEM 9

Statement by the delegation of Argentina

"Argentina has carefully gone through the rationale of the draft contained in document 9/1, as well as the background in document 9/1/1. We share the concerns of document 9/1 regarding the importance of ships receiving the necessary support from nearby coastal States when a ship provides assistance to persons in distress, in particular through the disembarkation of rescued persons. Rescuing persons in distress at sea is a cooperative exercise in which both flag states and coastal states are to be engaged. The SAR and SOLAS Conventions establish basic obligations and we share that we must promote their compliance, but there are also other obligations. As Italy clearly identifies in document 9/1/1, there are human rights aspects involved. Also, madam, it is important to bear in mind that, having raised this question, it is impossible not to also mention the applicable norms of refugee law, including the principle of 'non refoulement'.

Regarding the draft resolution, madam, Argentina favours it. However, we believe the draft should be drawn uptaking into account two issues: a) avoid incurring in interpretations or editing of the obligations of the SAR and SOLAS Conventions, and b) recognize that there are norms of international human rights law and refugee law by which IMO Members are also bound. In the understanding that it should not be difficult to do so, Argentina expresses its best willingness to work with other delegations in drafting."

Statement by the delegation of Germany

"As Denmark points out, it is of outmost importance that a ship providing assistance to persons in distress, receives the necessary support from costal states nearby. Prevention of swift disembarkation can lead to dangerous situation for crew and the persons rescued. Germany therefore fully supports the proposal made in MSC 104/9. We believe that the commitment of the governments to ensure suitable arrangements for assisting ships in their SAR activities is essential.

Italy, in MSC 104/9/1, provides its own experiences while also supporting the Danish proposal, as we understand it without asking for any amendments. Germany, however, has to give some comments, as we do not fully agree with some of the views expressed by Italy in the document. Specifically, the view that the primary responsibility for assisting the vessel in delivering survivors to a place of safety falls on the flag State and not on coastal States in the region cannot be shared by Germany. Such view contradicts the principles of the SAR Convention where the cooperation of the local coastal states is essential for providing a rapid and effective solution in rescue situations.

Furthermore, Germany wants to point out that privately owned and operated vessels engaged in search and rescue operations in the Mediterranean Sea do not act on behalf of their flag state. Therefore, they do not operate as "search and rescue vessels" of the flag state.

Germany would like this statement to be attached to the report and has forwarded it to the Secretariat."

Statement by the delegation of Malta

"My delegation wishes to thank the submitters of documents 104/9 and 104/9/1 dealing with unsafe mixed migration by sea. This is an issue of high sensitivity for Malta especially in the context of search and rescue operations in the central Mediterranean. We associate ourselves with the comment in paragraph 13 of Document 104/9 which highlights the importance of an effective and timely involvement of all governments involved in providing assistance to persons in distress at sea.

Malta would like to remind that in accordance with the applicable SAR Convention and IMO Guidelines, it is the responsibility of **ALL** Parties to cooperate with each other to provide

suitable places of safety. This is not a responsibility reserved exclusively for one entity. The reference to **All** Parties involved includes not only the Government responsible for the SAR area or the coordinating authority, but also other coastal states in the planned route of the vessel, the Flag State and the ship owners and their representatives. All stakeholders shall cooperate for disembarkation to take place within reasonable time.

In this spirit my delegation supports the draft resolution proposed by Denmark as annexed to document 104/9. We have proposed a few minor suggestions to Denmark on the working of their draft, without however altering its thrust and substance.

By the same token Malta takes issue with the point made in paragraphs 5 and 11 of document 104/9/1 that the disembarkation of persons rescued at sea be placed only on the State responsible for the SAR area in which the rescue was affected. It should be noted that Malta has not accepted the 2004 amendments to the 1974 International Convention for the Safety of Life at Sea (SOLAS), and the 1979 International Convention on Maritime Search and Rescue. The new provisions were accompanied by interpretative guidelines to the effect that the Rescue Coordination Centre (RCC) responsible for the SRR where the rescue took place also has the responsibility to identify a place of safety for the disembarkation of the persons rescued. Given that no obligation to cooperate was placed on other parties, this implies that such persons would have to be taken to a place under the responsibility of the Government of that RCC alone without applying the fundamental principle of the Convention namely cooperation between States in identifying a practicable place of safety.

Consequently, my delegation takes this opportunity to reaffirm Malta's position as a persistent objector to the 2004 amendments and does not share the view that these form part of customary international law."

Statement by the delegation of Spain

"España agradece a Dinamarca e Italia por los documentos presentados en relación a las repercusiones para el transporte marítimo de la migración mixta en condiciones peligrosas por mar.

España concuerda con los comentarios expresados por Dinamarca en el documento MSC 104/9 y apoya el proyecto de resolución que figura en el anexo.

Respecto a los comentarios contenidos en el documento MSC 104/9/1, nos gustaría realizar las siguientes observaciones de carácter general.

- .1 El Convenio SAR, fundamenta su aplicación sobre el principio de la cooperación entre los Estados para coordinar las operaciones de SAR;
- .2 Los buques destinados a ser explotados comercialmente, ajenos de inicio a realizar actividades SAR, deben ser liberados de las obligaciones derivadas de operaciones SAR, de modo que las personas auxiliadas sean desembarcadas con prontitud en un lugar Seguro
- .3 La intervención de los Estados de abanderamiento, incluso cuando participen buques de búsqueda y salvamento de su pabellón en la operaciones SAR llevadas a cabo en regiones de búsqueda y salvamento asociadas a otros Estados, quedaría supeditada a la responsabilidad primordial que le compete al Estado ribereño responsable de la región SAR donde se desarrollan las operaciones;
- .4 El Estado de abanderamiento no debería asumir una responsabilidad distinta de la prevista en el Convenio SAR, ya que para acometer los procedimientos de una operación SAR, se debe disponer de información actualizada que

solamente el Estado responsable de la región de búsqueda y salvamento dispondría;

- .5 España realiza anualmente un ingente número de operaciones SAR en las regiones de búsqueda y salvamento que tenemos asignadas. Prestamos un servicio público con medios propios de nuestra Administración Marítima, sin comprometer la participación de otros Estados como no sea de acuerdo con las disposiciones del Convenio SAR para coordinar las operaciones con los Estados vecinos, bajo la premisa de mantener relaciones cercanas de cooperación y coordinación con los servicios de salvamento de dichos Estados; y
- .6 España respalda la cooperación entre los Estados para garantizar el salvamento de las personas y su desembarco en lugar seguro, dentro del cumplimiento de los compromisos y obligaciones que se derivan de la aplicación del Convenio SAR."

AGENDA ITEM 17

Statement by the delegation of Jamaica

"Jamaica was among the first set of countries which responded to the pleas of the UN agencies led by the IMO, industry and seafarer associations to recognize their plight caused by the COVID pandemic and enacted legislation designating them as essential service workers which afforded them special rights including crew changes and repatriation.

The Maritime Authority of Jamaica developed a special portal to screen seafarers on line, and in conjunction with the Health Ministry and Immigration allowed them to use a particular corridor to move between the ships and ports, or, through our borders for transit crewmembers. Special ground transport is also arranged for movement between airports and seaports. Over 4000 seafarers have been facilitated so far."

Statement by the delegation of Panama

"Nuestra delegación agradece a la Secretaria por esta iniciativa de la implementación del proyecto de resolución examinado y agradece a INTERTANKO por su declaración.

Es por ello que la republica de Panama siendo una de las primeras en declarar a la gente de Mar como trabajadores Esenciales y comprometida con la Industria Marítima, desde el pasado mes de marzo 2020 hemos logrado de manera efectiva que más de once mil (11000) marinos de diversas nacionalidades, hayan podido retornar a sus hogares, gracias a los Instructivos internacionales y nacionales de Modalidades de Repatriación/Cambio de Tripulación, en la que se ha aportado tiempo extra a la atención de todas las solicitudes realizada durante la Pandemia, unas pudiéndose resolver inmediatamente y otras con complicaciones, pero todas atendidas en favor a nuestra gente de mar.

Cabe señalar que Panamá es signatario de la declaración de Neptuno donde buscamos reforzar nuestra postura como país y administración sería y responsable en busca del bienestar de nuestra gente de mar.

Además, quisiéramos aprovechar esta oportunidad para agradecerle infinitamente, la loable labor que realiza este grupo gestión de crisis SCAT de la OMI, a cada uno de sus integrantes y en especial al Sr. Jan De Boer por todo el esfuerzo realizado encomendado en apoyar y darle seguimiento a cada evento de sucesos acaecido de nuestra gente de mar."

Statement by the delegation of Slovenia

"The COVID-19 pandemic has represented – and continues to represent, for an unknown time – an unprecedented challenge for the international community. The international shipping community has been living up to its reputation as the backbone of international trade. Slovenia commends all that have contributed to meeting this challenge, in particular the seafarers, who have been absolutely crucial in ensuring the regular flow of trade by sea, the operation of the supply chains and the undisrupted availability of basic services, commodities, not least medical supplies, to the citizens and vital industries all over the globe.

We would also like to take this opportunity to extend our particular gratitude to the Secretary-General for his personal commitment to this cause.

We strongly support the draft Assembly resolution on comprehensive action to address seafarers' challenges during the COVID-19 pandemic as set out in MSC 104/17/1 and its adoption by the Assembly later this year."

On industry statement:

"Slovenia would like to thank our colleagues from the shipping industry for further highlighting the difficult situation of far too many seafarers.

The pandemic has left hundreds of thousands seafarers stranded in various parts of the world due to travel restrictions and reduction of travel options. Others have been unable to join the ships they were supposed to work on.

This has rightly been described as a humanitarian, safety and economic crisis.

It is first and foremost a humanitarian crisis. Seafarers remain stuck at sea or in ports, without a possibility to return home to rest and to see their loved ones. Sometimes they remain without pay and in deplorable working and living conditions. They are the collateral, and often forgotten, victims of the pandemic.

It is a safety crisis, because the physical and psychological fatigue of extended service at sea increases the risk of human errors, thus putting the safety of the ship, its crew and the environment at risk.

Finally, it is an economic crisis. If there is one thing we have learnt from the pandemic, it is the essential role of shipping and seafarers to keep the wheels of the global economy turning and to provide the citizens of the world with vital goods, not the least medicines, medical equipment and food.

The tireless efforts of the Secretariat and the personal commitment of the Secretary-General should rightly be commended.

The social partners have also worked hand in hand to address the crew change crisis.

Now it is for us, the member States, to be humble and see how we can improve our actions and increase our efforts to solve the crisis."

Statement by the observer from INTERTANKO

"This is a joint statement made on behalf of ICS, BIMCO, CLIA, ECSA, IAPH, IBIA, ICMA, IFSMA, IGP&I, IMCA, INTERCARGO, INTERFERRY, INTERMANAGER, IOGP, ITF, Nautical Institute, OCIMF, WSC and INTERTANKO.

We have all suffered during this pandemic with disruption to our lives, but few have suffered more than seafarers. They have been prevented from going home and in far too many cases refused access to urgent medical care. There have also been examples of ships being unable to repatriate seafarers that have passed away on board where member States have refused to allow this in contravention of their legal responsibilities under international law.

Notwithstanding these huge challenges, seafarers have not stopped working. They have put up with everything thrown at them and have been the true heroes of this pandemic.

The Secretariat should however be thanked. The Seafarers Crisis Action Team would be nothing without the personal commitment of the IMO Secretariat to fulfil their moral obligation to the seafarers. Their efforts have not gone unnoticed and should be applauded.

Of course, there have been some countries who have also stepped up and demonstrated leadership, waived requirements, facilitated crew change and above all else provided vaccines to foreign seafarers, however as argued at MSC 103, we would have hoped that this would have been implemented by all member States and included in the proposed Resolution.

But there have been others, too many others, who have not lived up to what can be reasonably expected. Many States have been entirely content to continue to accept the economic benefits of shipping, without making any effort to support those who make trade possible in contravention of their international obligations.

We therefore support the development and critically the implementation of any resolution to provide support for seafarers. But more importantly we urge member States that the words they say in support of this document and in support of seafarers, cannot be empty. What is said must be reflected in actions at home. Leadership must be shown.

We request member States to take the necessary steps to make life more bearable for them as they keep your economies going. As they keep your lights on. And as they bring the medicines that they are too often denied. We all know what is needed: recognition of Seafarers as key workers, vaccination, medical care and facilitation of crew changes. We need your urgent action to implement this resolution and match words with deeds."

Statement by the delegation of France

"Document MSC 103/INF.10 submitted by Mauritius at the previous session, relating to its intention to create a Particularly Vulnerable Sea Area (PSSA), has been included in the list of documents in the framework of this 104th session of the Maritime Safety Committee. However, the map annexed to the document includes the island of Tromelin in the waters under the jurisdiction of Mauritius.

The French delegation wishes to recall here that the island of Tromelin is a territory over which France exercises full sovereignty. Therefore, the map submitted to the MSC for information by Mauritius on the PSSA does not reflect the cartographic reality of the maritime spaces located around this Island and does not respect the French EEZ by implying that the EEZ and the island would fall under Mauritian jurisdiction and sovereignty. This document cannot have any legal value as it is a purely informative document.

Moreover, multilateral fora and IMO committees are not the appropriate places to raise issues in relation with national sovereignty". Finally, let me add that France maintains a dialogue with Mauritius on this subject."

Statement by the delegation of the Netherlands

"In regard to paragraph 17.32, The Netherlands notes the decision of the Committee to postpone the consideration of the mentioned documents to our next session, and we fully appreciate the reasons for doing so and we do not challenge the contents of this paragraph.

Nevertheless, we would like to make a short announcement related to the topic of preventing loss of containers at sea that is the main subject in document MSC 104/17/4 of which we are a co-sponsor.

We would just like to inform the Committee that our administration will organize a short web conference on the issue of container transport safety on 25 October 2021, from 11.00-12.15 UK Time (or 12.00-13.15 CET). It will include an introduction by our maritime director, a presentation by research institute MARIN and a short round of Q&A's and discussion.

Our delegation herewith would like to invite anyone who has not been invited personally already to contact Mr. Ringo Lakeman directly (ringo.lakeman@minienw.nl) in case you wish to participate, so that we can send you the official invitation."

Statement by the delegation of the United States

"The United States takes note of the China Maritime Safety Administration announcement on August 30, 2021, that certain vessels of foreign nationality entering the claimed territorial sea of the People's Republic of China shall report to the maritime administrations of the People's Republic of China. This new requirement appears to apply to such vessels when exercising the internationally recognized right of innocent passage in its claimed territorial sea and which do not enter or depart a port in the People's Republic of China.

Part II, section 3, of the Law of the Sea Convention reflects that all ships, regardless of cargo or means of propulsion, enjoy the right of innocent passage in the territorial sea in accordance with international law, for which neither prior notification nor authorization is required. The United States, accordingly, views this mandatory ship reporting requirement unilaterally established by the People's Republic of China as an unlawful attempt to restrict navigational rights and freedoms, wholly inconsistent with international law as reflected in the Law of the Sea Convention.

This purported reporting requirement is also concerning due to its highly unclear territorial scope. The People's Republic of China, as most delegations are aware, makes uncertain and expansive assertions of territorial sea and internal waters in areas where high seas freedoms apply as a matter of international law, including assertions in the South China Sea to which the United States and other states have objected.

The United States invites the Committee to call on the People's Republic of China to conform its regulation to international law as reflected in the Law of the Sea Convention and to the Organization's rules and procedures."

Statement by the delegation of Germany

"Germany aligns itself with the Statement presented by Slovenia and would like to take this opportunity to point out the following:

Against the background of the continuing developments in the Black Sea, Germany would like to reiterate the concern set out in the Communiqué of the G7 Foreign and Development Ministers' Meeting on 5th May 2021 in London:

'We express our deep concern over Russia's actions to block access to parts of the Black Sea, including near illegally-annexed Crimea and the Kerch Strait, impeding access to Ukraine's ports in the Sea of Azov.'"

Statement by the delegation of the United Kingdom

"The United Kingdom notes that it has been seven years since the Russian Federation's illegal annexation of Crimea which constitutes a flagrant violation of international law and the international rules-based order. The UK remains fully committed to upholding the sovereignty and territorial integrity of Ukraine within its internationally recognised borders.

In response to MSC 104/17/3, the UK is deeply concerned by the Russian Federation's decision to close parts of the Black Sea surrounding illegally annexed Crimea. This is the latest example of the Russian Federation's destabilising activity in the region, following its military build-up on the Ukrainian border earlier this year. We urge the Russian Federation to cease its provocations and immediately de-escalate the situation.

Likewise, the UK calls upon the Russian Federation not to hinder passage through the Kerch Strait. Restrictions on access to the Sea of Azov should end immediately.

The UK remains fully committed to implementing its non-recognition policy. We once again, call on UN Member States to consider non-recognition measures in line with the UNGA Resolution 74/168."

Statement by the delegation of Japan

"Regarding China's amended Maritime Traffic Safety Law, Japan takes note of the Article 121, which newly stipulates the relationship between, and application of, the international treaty to which China is a party and this law. Japan is of the view that Japan continues to pay close attention to related movements to ensure that the enforcement of the law wouldn't undermine the fair interests of the countries concerned including Japan."

Statement by the delegation of China

"First, China believes that IMO is not the appropriate platform to discuss matters of this nature. Secondly, China is a state party to the United Nations Convention on the Law of the Sea and has always strictly complied with the requirements of the Convention and respected the navigational rights of foreign ships under the United Nations Convention on the Law of the Sea.

In accordance with article 21 and 25 of the United Nations Convention on the Law of the Sea, China has formulated the Maritime Traffic Safety Law and a series of measures, aiming at providing timely knowledge and advance access to relevant information on specific ships, in particular, information on radioactive or other toxic substances carried onboard, in order to better provide maritime services and facilitate their safe, convenient and environmentally friendly access to and from the territorial waters of China.

This requirement is a precautionary measure in the territorial sea for the safety of navigation and the prevention of pollution. It is intended only as a record of information, not an authorization or approval, and does not affect the normal navigation of the ship.

In addition, the Maritime Traffic Safety Law of China is an administrative regulation regulating and adjusting maritime traffic activities. The main subjects of administration are ships, installations, and crews and shipping companies involved in maritime traffic activities that navigate, berth and operate at sea, and do not involve areas of jurisdiction. The Maritime Traffic safety law is consistent with international treaties and does not contain specific provisions for the South China Sea.

Therefore, the relevant regime established by China does not violate relevant requirements of UNCLOS or contravene the requirements of IMO Convention. As a Category A member of the International Maritime Organization (IMO), China has been actively fulfilling the requirements of various IMO conventions and working with member States to ensure the safety of navigation and the protection of the marine environment."